



2013 MAY 30 PM 5: 07 AGENDA REPORT

### TO: DEANNA J. SANTANA CITY ADMINISTRATOR

FROM: Rachel Flynn

SUBJECT: Planning Code Text Amendments

DATE: May 16, 2013

City Administrator Date Approval

COUNCIL DISTRICT: City-Wide

#### **RECOMMENDATION**

Conduct a Public Hearing and Upon Conclusion, Consider Adopting, as Recommended by the Planning Commission, An Ordinance Amending the Oakland Planning Code to:

- 1. Update References, Reduce Redundancy, And Clarify Language In Various Chapters Of The Planning Code;
- 2. Modify Truck Weight Classifications Language To Conform With Industry Standards;
- 3. Modify The Setback Requirement For The Portion of Buildings Above 55 Feet In Height In The CBD-R Zone;
- 4. Add Design Review Requirement For Residential Facilities And Historic Properties In The CIX, IG, And 10 Zones;
- 5. Eliminate The Increased Parking Requirement For Expansions Of Activities Across Zone Boundaries; and
- 6. Add Sidewalk Cafes As A Permitted Facility In The M-Industrial Zones

#### **EXECUTIVE SUMMARY**

The proposed Zoning text amendments involve mostly non-substantive and minor-substantive code clean-ups, as well as a few changes that are more substantive in nature. Examples of changes in each category are as follows:

<u>Non-Substantive changes</u> include reformatting, reorganizing and improving the internal consistency of the Planning Code and the consistency of the Code with the Land Use and Transportation Element (LUTE) of the Oakland General Plan. To achieve this, staff has simplified and clarified Code language as needed, updated Code references, and eliminated redundancies by combining and/or deleting language.

For example, there was a long-standing need to clean up Chapter 17.102 (General Regulations Applicable to All or Several Zones), which contains over 40 mostly unrelated sections. Several of these sections within the 17.102 Chapter involved special regulations that apply to certain use classifications; for ease of reference, this update is proposing to group together and place these regulations into a newly-created Planning Code Chapter: 17.103 - Special Regulations and Findings for Certain Use Classifications.

In addition, a long section of Chapter 17.102, "Mining and Quarrying", was taken out and relocated into its own new Chapter: 17.155 (Special Regulations Applying to Mining and Quarrying Extractive Activities).

The M-20, M-30 and M-40 Industrial Zones were combined into one single Chapter (Chapter 17.72) to be consistent with the way other internally-related Zoning districts are currently combined.

Other miscellaneous non-substantive changes include deleting references to Code sections that were remnants of previous code updates and are now obsolete, e.g., references to Transport and warehousing commercial activities which has been replaced by the new industrial Use Classifications adopted in 2009, and adding clarifying language to the method of calculating the maximum non-residential floor-area-ratio and residential density in the Central Business District.

<u>Minor-Substantive changes</u> include text changes to improve interpretability, clarity and flexibility of the Code. For example, staff proposes to correct the density for rooming units in the 120- and 160-foot height areas listed on Table 17.35.04 (Height, Floor Area Ratio, Density, and Open Space Regulations) to correct a typographical error. Also, the percentage of ground floor commercial façade transparency in the CBD Zones (Table 17.58.03) was increased to be consistent with similar standards in other zones. Language was added to various Code sections to codify and clarify the long-standing practice limiting evidence and testimony during appeals to evidence submitted in the appeal form itself and presented prior to the close of the written public comment period, for administrative decisions, or the close of public hearing, for Planning Commission decisions.

<u>Substantive changes</u> include changes to the standards in the Code. For example, 1) modify thresholds for truck weight classifications to conform with industry standards; 2) reduce slightly the setback requirement for the portion of buildings above 55 feet in height in the CBD-R Zone; 3) add a Design Review requirement for residential facilities and historic properties in the CIX, IG, and 10 Industrial Zones, as is currently required in the M- Industrial Zones; 4) eliminate the increased parking requirement for expansions of activities across zone boundaries; and 5) add Sidewalk Cafes as a permitted facility type in the M- Industrial Zones.

#### **OUTCOME**

Adoption of the recommended text changes to the Planning Code would clarify, improve consistency, reduce redundancy and simplify language portions of the Code to make for a more user-friendly document.

#### BACKGROUND/LEGISLATIVE HISTORY

In March of 1998, the City adopted the Land Use and Transportation Element (LUTE) of the Oakland General Plan. One of the objectives of the Land Use and Transportation Element (LUTE) is to create a 'user friendly' Planning Code document that minimizes the complexity of regulations. The City of Oakland's Planning Division staff has prepared this Code amendment package in order to correct a number of identified Code issues, update references, reduce redundancy, and clarify language in various chapters of the Planning Code.

This Planning Code amendment package is part of a series of Code amendment packages that have been approved or are in some phase of preparation, most of which are considered "cleanups", i.e., not substantive, and intended to improve consistency, reduce redundancy and simplify language. For example, in February of 2012, Council adopted amendments to the Planning Code (ZT12-012), that primarily addressed Chapter 17.108 (General Height, Yard and Court Regulations), a few portions of Chapter 17.102 (General Regulations Applicable to All or Several Zones), and minor clarifications to Chapter 17.104 (Signs). Among the substantive changes contained in that package was a modification to the square footage threshold that allows exemptions from the separation requirements for Alcohol Beverage Sales.

The Code amendments contained in this Report primarily address Chapter 17.102, (General Height, Yard and Court Regulations), but include portions of several other Planning Code Chapters.

On April 17, 2013, the Zoning Update Committee recommended that the Planning Commission hold a public hearing and forward the package to the City Council for approval (see *Attachment A*). No public testimony was presented.

On May 15, 2013, the Planning Commission recommended that the Oakland Planning Code modifications described in this report be approved by the City Council (see *Attachment A*).

#### ANALYSIS

The proposed Zoning text amendments are as follows (see *Attachment B*):

#### I. NON-SUBSTANTIVE CHANGES.

The following section summarizes the proposed Non-Substantive changes to the Planning Code. The Code language is in bold type and explanations are in normal type.

A. Staff proposes to modify the following chapters and sections to improve the clarity and internal consistency of the Code and/or its consistency with the General Plan:

17.07.030 – Purposes of the zoning regulations.

Section 17.07.030(A). To promote the achievement of the proposals, <u>policies and</u> <u>objectives</u> of the Oakland General Plan

By inserting the phrase "policies and objectives" to this Section, this amendment would bring more consistency between the objectives of the Planning Code with those of the General Plan LUTE.

17.10.280 - Fast-food restaurant commercial activities.

B. Except as may otherwise be allowed in OMC Chapters 5.49, 5.51, 8.09 and 9.52, tThe sale of ready-to-consume prepared foods from trucks, pushcarts or other movable equipment located on <u>public or private</u> property on a semi-permanent basis during hours of operation. Vehicular food vending generally has the following characteristics: For the benefit of the public, Staff proposes to insert a reference to provisions related to food vending found in the Oakland Municipal Code (OMC).

17.10.440 Transient Habitation Commercial Activities.

Transient Habitation Commercial Activities include the provision of lodging services to transient guests on a less-than-weekly basis, other than in the case of activities classified by Section 17.10.120 Semi-Transient Residential Activities or Section 17.10.125 Bed and Breakfast Commercial-Residential Activities. Examples include hotels and motels. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Staff inserted "hotels and motels" as examples of types of Transient Habitation Commercial Facilities and corrected the designation of the Bed and Breakfast classification from Commercial to Residential to be consistent with the rest of the Code.

17.10.480 - Automobile and other-light vehicle repair and cleaning commercial activities.

Automobile and Other Light Vehicle Repair and Cleaning Commercial Activities include the major repair or painting of motor vehicles that have a gross vehicle weight <u>rating of less</u> than fourteen thousand (14,000) pounds, including body work and installation of major accessories, as well as the washing and polishing of motor vehicles. This classification does not include vehicle dismantling or salvage and tire re-treading or recapping. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Staff proposes to insert the words "rating of" for consistency with other related sections and with industry standards.

1-7-10-500--Transport-and-warehousing-commercial-activities-(docs-not-apply-to-the CIX-1, CIX-2, IG, or-lO-zones).

Staff proposes to delete this section because it has been replaced by the new industrial Use Classification adopted in 2009. This classification had been kept in the Code because the classifications in the M-10 through M-40 Zones had not been updated to include the new classifications. This Zoning update includes updates to the M-10 through M-40 Zones such that the Transport and Warehousing classification is obsolete.

17.13.060 - Special regulations for mini-lot-and-planned unit developments. A.-Mini-lot-Developments.-In-mini-lot-developments, certain-regulations-that-otherwise apply-to-individual-lots-in-the-RH zones-may-be-waived-or-modified-when-and-as prescribed-in-Section-17.102.320

B.-.Planned Unit Developments. Large integrated developments shall be subject to the Planned Unit Development regulations in Chapter 17.142 if they exceed the sizes specified therein. In developments which are approved pursuant to said regulations, certain uses may be permitted in addition to those otherwise allowed in the RH zones, and certain of the other regulations applying in said zone may be waived or modified. Staff proposes to remove the reference to mini-lots from the RH (Hillside Residential) zones. This inclusion was an oversight in a previous zoning update and the change is consistent with the old R-1, R-10, and R-20 zones.

17.58 Central Business District.

Table 17.58.04, note 2: For mixed use projects in the Central Business District (CBD) zones, the allowable intensity of development shall be measured according to both the maximum nonresidential Floor Area Ratio (FAR) allowed by the zone and the maximum residential density allowed by the zone. The total lot area shall be used as a basis for computing both the nonresidential FAR and the residential density. No portion-of-lot-area-used-to-meet-the-density-requirements-for-a-Residential-Facility-shall be-used-as-a-basis for-computing, through-the-maximum-floor-area-ratio, the-maximum amount-of-floor-area-for-any-nonresidential-facility-on-the-same-lot, and-visa-versa. Staff proposes to delete the existing language regarding density calculations for mixed use projects located in the Central Business District, because the previous language regarding the method of measuring allowable intensity and density in the CBD was incorrect. This revision corrects the previous error and makes the language consistent with that used in Section 17.106.030(A).

17.106.030 - Maximum density and floor-area ratio on lots containing both Residential and Nonresidential Facilities.

A. Portion of Lot Area Used in Computing Density in the Central Business District and Jack London District. For mixed use projects in the Central Business District GBD

zones-and Jack London district, the allowable intensity of development <u>shall be</u> <u>measured according to both the maximum nonresidential Floor Area Ratio (FAR)</u> <u>allowed by the zone and the maximum residential density allowed by the zone. The total</u> <u>lot area shall be used as a basis for computing both the nonresidential FAR and the</u> <u>residential density.</u> may-be-measured-according-to-the-maximum-FAR-allowed-by-the zone-without-a-separate-residential-density-calculation, provided-the-maximum-number of-units-pursuant-to-the-residential-density-allowed-by-the General-Plan-and-Estuary</u> Policy-Fitm-is-not-exceeded--(The Central Business District is that area identified as part of the Land Use and Transportation Element Land Use Diagram of the General Plan. The Jack London district is that area identified as part of the Estuary Policy Plan and adopted as part of the General Plam).

Staff proposes to delete the existing language and insert new language clarifying density calculations for mixed use projects in the CBD and Jack London District.

17.130.060 - Obligation of applicant to defend, indemnify, and hold harmless the City of Oakland.

A. To the maximum extent permitted by law, the applicant shall defend (with counselreasonably acceptable to the City), indemnify, and hold harmless the City of Oakland, the Oakland City Council, the City of Oakland Redevelopment Agency, <u>City of Oakland Redevelopment Successor Agency</u>, the Oakland City Planning Commission and its respective agents, officers, <u>volunteers</u> and employees (hereafter collectively called City) from any liability, damages, claim, judgment, loss (direct or indirect), action, causes of action or proceeding (including legal costs, attorneys' fees, expert witness or consultant fees, City Attorney or staff time, expenses or costs) (collectively called "Action") against the City to attack, set aside, void or annul, (1) an approval by the City relating to a development-related application <u>or</u> subdivision <u>and/or a (Lease)</u> <u>Disposition and Development Agreement;</u> or (2) implementation of <u>such</u> an-approved development-related-project. The City may elect, in its sole discretion, to participate in the defense of said Action and the applicant shall reimburse the City for its reasonable legal costs and attorneys' fees.

Staff added clarifications to the above language.

# <u>17.138.015– Projects eligible and special regulations for projects with development</u> agreements

Staff proposes to relocate Section 17.102.310 to a new Section 17.138.015(A) and (B), and modify the language to add "successor to the" Redevelopment Agency and delete "major" in reference to affected permit types.

Chapter 17.142 — <u>Mini Lot and Planned Unit Development Regulations</u>. Staff added a Mini Lot section (Article II) to Chapter 17.142 to clarify applicability and basic provisions.

17.142.030 – Developments for which <u>planned unit development permit</u> approval is required <u>or requested.</u>

<u>A.</u> The following developments are permitted only upon the granting of a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140:...Other-large, integrated-developments-arc-permitted-without-such-a-permit, but shall-be-subject-to-all-regulations-generally-applying-in-the-zone-in-which-they-are located.

1A. Any planned unit development incorporating any of the bonuses set forth in Section 17.142.100.

2B. Any integrated development which is primarily designed for or occupied by Commercial Activities, which is located in any commercial zone, and which is developed under unified control, in accordance with a comprehensive plan, on a single tract with sixty thousand (60,000) square feet or more of land area, or on two or more tracts which total such area and which are separated only by a street or other right-of-way.

**B.** Unless required by the Planning Director, other large, integrated developments involving the same minimum land area thresholds of a planned unit development, as defined in Section 17.142.020, are permitted without such a permit. However, an

- applicant of such a development may request a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140, but shall be subject to all regulations generally applying in the zone in which they are located. Staff added a new subsection B to clarify that the Planning Director may require other very large integrated developments to be reviewed under the planned unit development permit procedure, where it is deemed appropriate; otherwise, an applicant of such a development may request a PUD permit.
- **B.** Staff proposes to move or combine the following code chapters and sections as noted:

Chapter-17.68-M-20-Industrial-Zones-Regulations Chapter-17.70-M-30-Industrial-Zones-Regulations

Chapter-17-72-M-40-Industrial-Zones-Regulations

Chapter + /- /2-WI-40-Huustrian-Zones-Regulations

Chapter 17.72 M-20, M-30, and M-40 Industrial Zones Regulations.

Staff proposes to combine these three Industrial zones into one chapter that will be in the same format as the most recently adopted zones. Also, Staff added sidewalk cafés as a permitted facility type in the M- Zones to correct an oversight (see 17.72 under Substantive Changes section).

17.102.070 - Application of zoning regulations to lots divided by zone boundaries Staff proposes to move this entire section to 17.154.060 Zoning maps.

<u>17.102.080---Permitted-and-conditionally-permitted-uses</u> 17.07.065 <u>– Permitted and conditionally permitted uses</u>. A. Other Uses Prohibited. Except as otherwise provided in Sections 17.102.040 <u>17.114.030</u> and 47.402.070.17.154.060, the nonconforming use regulations in Chapter 17.114, and the planned unit development regulations in Chapter 17.142, or as authorized under Section 17.102.310.17.138.015, the development agreement procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no land shall be improved or used for any activity or facility which is not listed as permitted or conditionally permitted in the applicable individual zone regulations or development control maps.

B. Relationship Between Activities and Facilities. A use must qualify under the zoning regulations both as an activity and as a facility. A permitted or conditionally permitted activity may be accommodated or served only by a permitted facility or, upon the granting of a conditional use permit, by a conditionally permitted facility; and a permitted or conditionally permitted facility may accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditionally permitted facility activity or, upon the granting of a conditionally permitted facility may accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditional use permit, a conditionally permitted activity.

Staff proposes to move Section 17.102.080 to Chapter 17.07 Title purpose and scope of zoning regulations, and correct the references as shown.

17.102.100 - Conditions for accessory parking serving activities which are not themselves allowed.

Staff proposes to move this entire section to 17.116 Off-Street Parking and Loading Requirements.

17.102.220 - Special regulations applying to mining and quarrying extractive activities. Staff proposes to move this entire section to its own new chapter (17.155), because it is so lengthy.

17.102.265 – Occupancy of a One-Family Dwelling Residential Facility by a Residential Care Residential Activity.

Staff proposes to delete this section and combine it with Section 17.09.040, definition of "Dwelling unit".

17.102.280 - Rules for determining the number of habitable rooms in Residential Facilities.

Staff proposes to move this entire section to Chapter 17.09 Definitions ("Habitable room" and "Habitable rooms, number of").

17.102.310 - Special regulations for certain projects with development agreements. Staff proposes to move this section to Chapter 17.138 Development Agreement Procedure.

17.102.320 - Conditional use permit for waiver of certain requirements in mini-lot developments.

Staff proposes to move this section to 17.142 Planned Unit Development Regulations so that both types of integrated developments, PUD's and mini-lots, are in the same chapter.

17.102.330 – Conditional use permit for waiver of certain requirements with parcel division between existing buildings.

Staff proposes to delete this section because it already exists in Chapter 17.106, General Lot, Density and Area Regulations, Section 17.106.010(B).

17.102.380 – Special regulations applying to truck-related activities in the West Oakland Community Development District.

Deleted, and added as a footnote to any zone that allows trucking activities.

17.102.400 - Special design requirements for lots that contain Residential Facilities and no Nonresidential Facilities.

Staff proposes to move the paving restrictions portion of this section to a footnote in the zoning chapters that allow residential facilities; Screening of utility meters and trash containers moved to Section 17.124.045 Landscaping and Screening standards. Exterior security bars and grills regulations would stay in this Section, 17.102.400, and the section renamed to "Exterior security bars and related devices".

17.102.420 – Special design requirements for lots located in residential and commercial zones and the OS, S-1, S-2, S-3 and S-15 zones. Deleted, because it is redundant with Section 17.108.140.

1-7-1-02-260-"Occupancy-of-a-dwelling-unit"

17.09.040 Definitions.

"Dwelling unit" means a room or suite of rooms including one and only one kitchen, except as otherwise provided in Section 17.102.270, and designed or occupied as separate living quarters for one-of-the-persons-or-groups-specified-in-Section-17-102-260, one person or family; or, where the facility occupied is a One-Family Dwelling, such family and not more than three (3) boarders, roomers, or lodgers, where access to all rooms occupied by such boarders, roomers, or lodgers is had through the main entrance of the dwelling unit.

Section 17.102.260 "Occupancy of a dwelling unit" is essentially a definition of a dwelling unit; therefore, Staff proposes to delete Section 17.102.260 and combine its language with the definition of "Dwelling unit" in Section 17.09.040.

C. Staff proposes to delete the following sections in their entirety because they no longer serve any purpose:

17,10,760-Shopping-Center-Facility.

A. Shopping-Center Nonresidential-Facility is a complex of one-or-more-retail-buildings and-related-facilities forming-a-central-retail-market-within-a-given-area and having a common-parking-area.

1-7.102.020-Supplemental-Zoning-Provisions

The definitions, special-use-classification-rules, and other-provisions-sot-forth-in Chapters-17.07, 17.09-and-17.10;-the-provisions-of-17.108.130;-the-nonconforming-use regulations-in-Chapter-17.114;-the-rezoning, variance, and other-provisions-sci-forth-in Chapters-17.130-through-17.152; and the provisions of the zoning-maps-in-Chapter 17.154-shall-apply-throughout-the-city. The-planned-unit-development-regulations-in Chapter-17.142-shall-apply-in-the-zones-and-situations-specified-in-said-chapters. The provisions-of-development-control-maps-are-in-addition-to, or-supersede-when-so specified, the-regulations-applying-in-the-zones-covering-the-same-area.

D. Staff proposes to delete Section 17.102.130 in its entirety because there is already a regulation that addresses this issue in 17.10.040 Accessory Activities.

17.102.130 Time-limit-on-operation-of-subdivision-sales-offices -- Residential-Zones:

- E. Staff proposes to move all of the following sections to a new Chapter (Chapter 17.103 -Special regulations and findings for certain use classifications), that will contain special regulations that apply to particular use classifications that are listed in Chapter 17.10:
  - 17.102.210 Special regulations applying to Convenience Markets, Fast-Food Restaurants, certain establishments selling alcoholic beverages, providing mechanical or, and electronic games.
  - 17.102.212 Special Regulations Applying to Residential Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities.
  - 17.102.290 Special regulations for Drive-Through Nonresidential Facilities.
  - 17.102.335 Standards for Sidewalk Cafes.
  - 17.102.360 Secondary Units.
  - 17.102.370 Conditional use permit for hotels and motels.
  - 17.102.390 Regulations Applying to Special Health Care Civic Activities.
  - 17.102.430 Regulations applying to check cashier and/or check cashing activity.
  - 17.102.440 Special regulations for primary collection centers in all zones.
- F. The following new code section and related code changes are proposed to: (a) codify the long-standing practice of processing minor permits with major permits; and (b) clarify that this only applies to the procedural requirements, but that the substantive requirements of minor permits (i.e., findings) still apply:

17.130.090 - Minor land use permits considered concurrently with Major permits.

A. Any Minor land use related permit and/or approval that is related to a development application that also includes any Major land use related permit and/or approval shall be considered concurrently with the Major land use related permit and/or approval, and shall follow all procedural requirements associated with City Planning Commission decisions. In this case, the entire application shall initially be considered by the City Planning Commission and may be appealed to the City Council, in accordance with the requirements for Major land use related permit and/or approval or discretionary actions.

**B.** Any Minor land use related permit referred to the City Planning Commission for initial decision in order to be considered concurrently with any Major land use related permit and/or approval shall still be considered a Minor land use related permit and/or approval, and the required findings for said Minor land use related permit and/or approval shall apply.

17.134.020 – Definition of major and minor conditional use permits. Staff proposes to delete Subsection 17.134.020(3)(h) to clarify that a minor conditional use permit referred by the Plauming Director to the Planning Commission for initial decision is still considered a minor permit.

17.134.040(B) – Procedures for consideration – Minor Conditional Use Permits. Staff proposes to add language to clarify that a minor conditional use permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major CUP procedure.

17.136.040(D) – Regular Design Review – Procedure for Consideration. Staff proposes to add language to clarify that a regular design review permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major Permit procedure.

17.148.020 – Definition of major and minor variances.

- 7.—Any-variance-application-that-requires-development-of-an-Environmental-Impact Report;
- 8.—Any-variance-application-referred-by-the-Director-of-City-Planning-to-the-City Planning-Commission-for-initial-decision-pursuant-to-Section-17-148-040(B)(1).

Staff proposes to delete Subsection 7 above to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit. Staff proposes to also delete Subsection 8 above to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit.

17.148.040 – Procedure for consideration.

Staff proposes to add language to subsection B, Minor Variances, to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major Variance procedure.

#### G. Other general clean-up:

- The word "Section" was inserted prior to the section number where appropriate (e.g., "17.105.010" changed to "Section 17.105.010").
- Numbers were placed in parentheses and inserted after spelled-out numbers where appropriate (e.g., "35 feet" was changed to "thirty-five (35) feet".
- When massage, adult entertainment; check cashing; special health care, fast food are permitted or conditionally permitted in a zone, new notes were added in the Activity Tables directing the reader to further regulations in Chapters 17.102 or 17.103.
- 17.65.030, Note L3 Staff added a missing note for General Warehousing Activities in order to fix an error during the HBX Zones text amendment.
- Staff removed references in other parts of the Code that were related to previouslydeleted sections of Chapter 17.108. (In the spring of 2012, the City Council adopted an ordinance that removed the following Code sections: 17.108.040; 17.108.050; 17.108.060; 17.108.070; 17.108.090; 17.108.100, and 17.108.110).
- Staff corrected references to the relocated sections of 17.102 in the new M-20, M-30, M-40 Chapter (17.72).
- Staff added applicable language regarding front yard paving restrictions into the code from the previous Section 17.102.400 to the new footnotes in the Zoning chapters that allow Residential facilities.
- Staff relocated Special Health Care Civic Activities from Section 17.134.020(2)(b)(iii) into a new Section 17.134.020(2)(a)(xiii), under Definition of major conditional use permits. This Activity triggering a Major CUP was inadvertently misplaced under the Facility types subsection rather than Activity types.

#### H. MINOR-SUBSTANTIVE CHANGES

The following section summarizes the proposed Minor-Substantive changes to the Plauning Code. The Code language is in bold type and explanations are in normal type.

#### 17.09.040 Definitions.

"Gross vehicle weight rating" means the vehicle weight specified by the manufacturer as the maximum loaded weight (truck plus cargo) of a single vehicle.

Staff proposes to add the above definition for "gross vehicle weight rating". This is the industry standard and creates consistency in the truck activity classifications.

Table 17.35.04 - Height, Floor Area Ratio (FAR), Density, and Open Space Regulations.

Staff proposes to correct the density for rooming units in the 120- and 160-foot height areas from "one unit per 225 sf of lot area" to "one unit per 110 sf of lot area", in order to fix a previous typographical error that reduces density for rooming units located in the tallest height areas.

 Table 17.58.03: Property Development Standards.

Staff proposes to change the required percentage of façade transparency in the CBD zones to be consistent with standards in other zones.

17.54 - C-40 Community Thoroughfare Commercial Zone Regulations.

17.56 - C-45 Community Shopping Commercial Zone Regulations.

17.65 – HBX Housing and Business Mix Commercial Zones Regulations.

17.72 - M-20, M-30 and M-40 Industrial Zones Regulations.

The updated Industrial classifications, created in 2008, were not included in the activity tables for the C-40, C-45, HBX and M-Industrial zones. Staff proposes to integrate these activities into the chapters to correct this oversight.

17-102-250-Maximum-density-and-floor-area-ratio-during-construction Staff proposes to delete Section 17.102.250 because it has never been an issue and has proven difficult to enforce.

17.102.040 17.114.030 - Effect of prior permits.

A. Fundamental Vested Rights Not Abrogated By Code Adoption or Amendment. The adoption of, or amendment to, the Planning Code (Chapter 17), including without limitation, ordinances enacted pursuant to Oakland City Charter section 213 (Emergency Ordinances) and/or ordinances enacted pursuant to Government Code section 65858 (Urgency Measure/moratoria), shall not abrogate any fundamental vested rights established pursuant to State law, including, without limitation, those established pursuant to the prior valid adoption and execution of a development agreement under Section 17.138.015 and the development agreement procedure in Chapter 17.138. Absent the prior establishment of such vested rights, any Planning Code adoption or amendment -shall apply.A.-Building-and-Sign-Permits-and-Development-Agreements. Whenever-any subsisting-building-permit-or-sign-permit-has-been-lawfully-issued-beforehand,-or-whenever a-subsisting-development-agreement-has-been-approved-beforehand-under-Section 17-102-310-and-the-development-agreement-procedure-in-Chapter-17-138, neither-the original-adoption-of-the-zoning-regulations-nor-the-adoption-of-any-subsequent-rezoning-or other-amendment-thereto-shall-prohibit-the-construction, other-development-or-change, or use-authorized-by-said-permit-or-agreement-The-uses-as-they-result-shall-be-deemed nonconforming-uses-and-subject-to-the-nonconforming-use-regulations-in-Chapter-1-7-1-14; except-to-the-extent-that-they-arc-authorized-by-a-subsisting-conditional-use-permit, development-agreement, variance, or other-special-zoning-approval.

B. Alcoholic Beverage Control Licenses. On premises for which a valid state of California Alcoholic Beverage Control license had been issued, and which premises had been used in

the exercise of the rights and privileges conferred by the license at a time immediately prior to the effective date of the applicable provisions of Section-17-102-210 <u>17.103.030</u>, the premises may hereafter be used in the exercise of the same rights and privileges without requiring a conditional use permit or having to meet the provisions of the aforesaid section. Such-use<u>The uses as they result</u> that do not conform to the zoning regulations shall be deemed a nonconforming use and subject to the nonconforming use regulations, except as otherwise provided in Sections 17.114.020 and 17.114.030. For the purposes of this subsection, the word "premises" shall mean and include only the actual space within a building devoted to the sale of alcoholic beverages on said effective date. Staff proposes to delete Section 17.102.040 above and move the text to Chapter 17.114 - Non-Conforming Uses. The resulting Section 17.114.030 above is proposed to include changes to subsection (A), Fundamental Vested Rights Not Abrogated By Code Adoption or Amendment, in order to be consistent with State law.

In addition, language was added to the following sections to codify and clarify the long-standing practice limiting evidence and testimony during appeals to evidence submitted in the appeal form itself and presented: (1) prior to the close of the written public comment period for the underlying decision being appealed, in the case of appeals based on a decision by the Zoning Administrator or other administrative decisions, or (2) prior to the close of the City Planning Commission's public hearing for the underlying decision being appealed, in the case of appeals based on decisions made by the City Planning Commission, as applicable. The language added is also intended to eliminate the requirement that, if the City Council is unable to decide an appeal, the matter appear for a vote on each regular meeting of the Council thereafter until decided.

- 17.130.050 Presentation of written and documentary evidence.
- 17.134.040 Procedures for consideration.
- 17.134.060 Appeal to Planning Commission-Minor Conditional Use Permits.
- 17.134.070 Appeal to Council-Major Conditional Use Permits.
- 17.136.040 Regular Design Review.
- 17.136.080 Appeal to Planning Commission—Regular design review.
- 17.136.090 Appeal to City Council—Regular design review.
- 17.140.030 Preliminary Planning Commission action.
- 17.140.060 Final Planning Commission action.
- 17.140.070 Appeal to Council.
- 17.144.090 Council action.
- 17.148.040 Procedure for consideration.
- 17.148.060 Appeal to Planning Commission-Minor variances.
- 17.148.070 Appeal to Council-Major variances.
- 17.156.140 Procedure for consideration of violations to performance standards.
- 17.156.150 Procedure for consideration of violations to conditions of approval.
- 17.156.160 Appeal to Planning Commission.
- 17.156.170 Appeal on the revocation of a Deemed Approved Status to the City Council.

17.157.110 - Procedure for consideration of violations to performance standards.
17.157.120 - Procedure for consideration of violations of conditions of approval.
17.157.130 - Appeal to City Planning Commission.

17.157.140 - Appeal on the revocation of a Deemed Approved Status to the City Council.

#### **III. SUBSTANTIVE CHANGES**

The following section summarizes the proposed substantive changes to the Planning Code. The Code language is in bold type and explanations are in normal type.

17.10.485 - Taxi and light fleet-based service commercial activities.

Passenger transportation services, local delivery services, and other businesses that rely on fleets of three (3) or more vehicles with a gross vehicle weight rating rated-canacity-of less than twelve-thousand-five-hundred-(12,500)-fourteen thousand (14,000) pounds. This classification includes parking, dispatching, and offices for taxicab and limousine operations, airport shuttles, medical transport, local messenger and document delivery services, janitorial services, and similar businesses. This classification does-not<u>only</u> includes towing operations when except-for-tow-truck-services-where-vehicles are taken to off-site locations and the tow trucks do not exceed the above gross vehicle weight rating. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Staff proposes the above changes to this regulation to use measurements that are consistent with industry standards for Light Vehicles (e.g., pick-up trucks, ambulances and parcel delivery trucks).

17.10.585 Trucking and truck-related industrial activities.

Trucking and Truck-Related Activities include the provision of freight handling and shipping services by trucks as well as parking, maintenance, and services for trucks and other heavy vehicles and equipment. Each classification involves the use of trucks and other heavy vehicles that have a gross vehicle weight rating greater than or equal to fourteen thousand (14,000) pounds. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Staff proposes changes to the Truck Activities regulations and use measurements in Section 17.10.585 that are consistent with industry standards.

 Table 17.58.03, note 4
 Property development standards.

In the CBD-R zone, portions of a building over fifty-five (55) feet in height shall have a setback of at least one (1) foot from the required interior side yard for every <u>five (5)</u> four (4)-<u>fcet</u> that portion is above fifty-five (55) feet. This setback, however, need not exceed forty (40) feet. Also, see Section 17.108.080 for the required interior side and rear yard setbacks on a lot containing two or more living units and opposite a legally-required living room window. See Section 17.108.130 for allowed projections into required yards.

Staff proposes to change the step-back requirement in the CBD-R zone from 1 foot for every 4 feet of height above 55 feet to 1 foot for every <u>5 feet</u> of height above 55 feet. This reduced amount of step-back is more appropriate in a downtown context.

17.72 M-20, M-30, and M-40 Industrial Zones Regulations.

Staff added sidewalk cafés as a permitted facility type in the M- Zones to correct an oversight in a previous zoning code change that added restaurants as an allowed activity type in the M-Zones, but overlooked including sidewalk cafes as a permitted facility type. (Note: The remaining M- Zones are all on the water, including Jack London Square).

17.73 - CIX-1, CIX-2, IG and IO Industrial Zone Regulations.

The following new Design Review Section (17.73.015) was added:

A. Except for projects that are exempt from design review as set forth in Section 17.136.025, no Residential Facility, Designated Historic Property, Potentially Designated Historic Property, Telecommunications Facility, Sign, or other associated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications regulations in Chapter 17.128, or the Sign regulations in Chapter 17.104.

B. No facility located within one hundred fifty (150) feet of any residential zone boundary and accommodating the following activities shall be constructed, established, or expanded in size unless plans for the proposal have been approved pursuant to the Regular Design Review procedure in Chapter 17.136:

1. Automobile and Other Light Vehicle Gas Station and Servicing Activity.

2. Automobile and Other Light Vehicle Repair and Cleaning Activity.

3. Freight/Truck Terminal.

4. Truck Yard.

5. Truck Weigh Stations.

6. Truck and Other Heavy Vehicle Sales, Rental, and Leasing.

7. Truck and Other Heavy Vehicle Service, Repair, and Refueling.

C. Establishment of a work/live unit shall only be permitted upon determination that the proposal conforms to the regular design review criteria set forth in the Regular Design Review procedure in Chapter 17.136 and to all of the additional criteria set forth in Section 17.73.040(D).

Staff proposes to add this language to provide for Design Review for Residential facilities and historic properties that are located in the CIX, IG an 10 zones and to consolidate references to activities requiring Design Review that are already included in this Chapter as part of the Activities Table. This is considered a minor change because it does not apply to non-residential buildings that are not historic.

1-7-102-1-10(F)--Increased-Off-Street-Parking--Off-street-parking-shall-be-provided-for-the proposed-development-in-an-amount-at-least-one-hundred-fifty-percent-(1-50%) of-that required-by-the-off-street-parking-requirements-in-Chapter-17-1-16-

Staff proposes to delete Section 17.102.110(F) because it seems too onerous and will generally apply in the corridors where development is encouraged.

#### **IV. OTHER RELATED CHANGES**

Various changes were made to the code that relate and correspond to the amendments described above.

#### PUBLIC OUTREACH/INTEREST

Consideration of this package of proposed Plaiming Code amendments was publicly noticed in the Oakland Tribune on Saturday, March 30, 2013; Saturday, April 27, 2013; and May 25, 2013 - a minimum of 17 days prior to public hearings before the Zoning Update Committee (ZUC) (on April 17, 2013), Planning Commission (on May 15, 2013), and the CED Committee (June 11, 2013). No comments were received during these public notice periods.

#### **COORDINATION**

The proposed Zoning text changes have been reviewed by the City Attorney's Office and the Budget Office.

#### COST SUMMARY/IMPLICATIONS

The proposed ordinance will have no direct fiscal impact on the City. Existing staffing should be sufficient to process development applications regulated by the proposed Zoning Code amendments which are the subject of this report. No additional costs are expected to be incurred by the Planning and Building Department's Development Service Fund.

#### SUSTAINABLE OPPORTUNITIES

*Economic*: The addition of Sidewalk Cafes as a permitted facility type in the M-20, M-30 and M-40 Industrial Zones, which was intended but inadvertently excluded in previous Code updates, will allow more restaurant businesses to offer outdoor dining especially along the waterfront, which will likely attract more patrons and result in increases in sales tax dollars and local jobs.

*Environmental*: The addition of a design review requirement for residential and historic properties located in the CIX, IG and 10 Industrial Zones will help to promote and preserve the quality of design of residential and historic structures throughout Oakland.

*Social Equity*: The addition of Sidewalk Cafes as a permitted facility type in the M-20, M-30 and M-40 Industrial Zones, will allow more restaurant businesses the opportunity to offer an outdoor dining amenity to their patrons, which will likely attract more business and may result in more local jobs.

#### <u>CEQA</u>

The proposal relies on the previously certified Final Environmental Impact Report (EIR) for the Land Use and Transportation Element of the General Plan (1998), the Oakland Estuary Policy Plan (1998); the EIRs for the West Oakland Central City East, Coliseum and Oakland Army Base Redevelopment Areas; the Final EIR for the 1998 Amendment to the Historic Preservation Element of the General Plan; the 2007-2014 Housing Element Final EIR (certified on November 17, 2010); and various Redevelopment Plan Final EIRs. No further environmental review is required under CEQA Guidelines sections 15162 and 15163. On a separate and independent basis, this proposal is also exempt under Section 15183 "Projects Consistent with a Community Plan, General Plan or Zoning" and Section 15061(b)(3) "Review for Exemption" of the State CEQA Guidelines.

Environmental status granted by CEDA. If applicable include the detail of the CEQA status. The appropriate citation is to be noted in the report or listed as "This report is not a project under CEQA."

For questions regarding this report, please contact Ann Clevenger, Planner III, at (510)238-6980.

Respectfully submitted,

ſ

RACHEL FLYNN Director, Department of Planning and Building

Reviewed by: Scott Miller Zoning Manager

Prepared by: Ann Clevenger, Planner 111 Planning and Zoning Division

,

# Attachments: A Planning Commission Staff Report, May 15, 2013 (with Zoning Update Committee Report, Staff Memo to ZUC, and Summary Table)

ς.

B Zoning Text Amendments

### **Oakland Planning Commission**

Case File Number ZT13-065

STAFF REPORT May 15, 2013

Proposal:	Discussion of proposed zoning text amendments to Chapter 17.102 and various
	other parts of the Planning Code, including:
· · ·	<ol> <li>Non-Substantive Changes (reformatting, reorganizing and improving the consistency of the Code);</li> </ol>
	<ol> <li>Minor-Substantive Changes (text changes to improve the interpretability, clarity, and flexibility of the Code); and</li> </ol>
· · · ·	3) Substantive Changes (changes to the standards in the Code), including: truck weight classifications (17 10.485 and 17.10.585); reducing slightly the setback for building portions above 55 feet in the CBD-R zone; adding Design Review for residential facilities and historic properties in the CIX, IG, and iO Zones; eliminating the increased parking requirement for expansions of activities across zone boundaries; adding sidewalk cafes as a permitted facility in the M-Industrial Zones.
Applicant:	Planning Commission
Planning Permits Required:	Zoning Text Amendments (ZT13-065)
General Plan:	Various
Zoning:	Various
Environmental Determination:	The proposed amendments to the Zoning Code rely on the previously certified Final Environmental Impact Reports for the Land Use and Transportation Element of the General Plan (1998); the Oakland Estuary Policy Plan (1998); the West Oakland Central City East, Coliseum and Oakland Army Base Redevelopment Areas; the
	1998 Amendment to the Historic Preservation Element of the General Plan; the 2007-2014 Housing Element Final EIR (2010); and various Redevelopment Plan Final EIRs (collectively, "EIRs"). No further environmental review is required under CEQA Guidelines sections 15162 and 15163. Each as a separate and independent basis, this proposal is also exempt from CEQA pursuant to CEQA Guidelines Sections 15183 (projects consistent with General Plan and Zoning) and 15061(b)(3)
	(no significant effect on the environment).
Service Delivery District:	Alļ
City Council District:	All
Status:	Hearing by the Planning Commission
Action to be Taken:	Recommendation to the City Council
Staff Recommendation:	Recommend adoption of the proposal to City Council
For Further Information:	Contact Ann Clevenger at (510)238-6980 or email aclevenger@oaklandnet.com

#### SUMMARY

The subject proposal includes text amendments to the City's Planning Code (see Attachment C for summary table of proposed text amendments and Attachment **D** for complete Planning Code amendment text). This is the 3<sup>rd</sup> set of amendments brought forward since the adoption of the major Planning Code amendments and citywide rezoning effective in April 2011. These proposed text amendments primarily address Section 17.102 (General Regulations Applicable to

#3

All or Several Zones), and include various changes in other Planning Code Chapters and sections. Many of the amendments are non-substantive, intended to improve consistency, reduce redundancy and simplify Code language. Staff is also proposing a few changes that are more substantive: (1) In Chapter 17.10 (Use Classifications) Change terminology and weight thresholds for vehicles for consistency with industry standards; (2) 17.58 (Central Business District Zone) – Change the building waii setback in CBD-R Zone to i foot of setback per 5 feet of height over 55, to 1 foot of setback per 4 feet; (3) 17.73 (CIX, IG and 10 Zones) – Add Design Review for Residential facilities and historic properties. (Note: This does not apply to non-residential buildings that are not historic); (4) Section 17.102.110 – Delete subsection (F) requiring increased parking in certain cases because it seems too onerous and applies in corridors where development is encouraged, and (5) adding sidewalk cafes as a permitted facility type in the M-industrial Zones.

This Zoning Update package was presented to the Zoning Update Committee (ZUC) on April 17, 2013 (see Attachment A for ZUC staff report). On Staff s recommendation, the Committee voted to:

- (1) eliminate the following two items that were originally proposed (see Attachment B for staff memo to the ZUC):
  - (a) a change limiting the time of review by advisory boards, and
  - (b) a minor change to the language in the definition of "Historic Properties", and,
- (2) refer the package as modified to the City Planning Commission.

Staff, as authorized by the ZUC, has included in this report further non-substantive, technical edits, e.g., adding language that further clarifies the method of calculating the maximum density and floor-area-ratio for mixed-use projects in the Central Business District and Jack London district, (Section 17.106.030), as well as other minor corrections and clean-ups, and has added a Summary Table (Attachment C) for ease of reference.

#### PROJECT DESCRIPTION

The proposed changes to the zoning text fall into three basic categories:

- I. Non-substantive Changes
- II. Minor-substantive Changes
- III. Substantive Changes

Non-substantive changes include reformatting, reorganizing and improving the internal consistency of the Planning Code and the consistency of the Planning Code with the General Plan Land Use and Transportation Element (LUTE). Minor substantive changes include text changes to improve the interpretability, clarity, and flexibility of the Planning Code. Finally, staff proposes substantive changes to certain codes to improve the standards in the Planning Code. The content of these proposed changes is summarized in the following report sections. Please see Attachment D for the complete zoning amendment text.

#### I. NON-SUBSTANTIVE CHANGES

The following section summarizes the proposed Non-Substantive changes to the Planning Code. The Code language is in **bold** type and explanations are in normal type.

A. Staff proposes to modify the following chapters and sections to improve the clarity and internal consistency of the Code and/or its consistency with the General Plan:

#### 17.07.030 - Purposes of the zoning regulations.

Section 17.07.030(A). To promote the achievement of the proposals, <u>policies and objectives</u> of the Oakland General Plan

By inserting the phrase "policies and objectives" to this Section, this amendment would bring more consistency between the objectives of the Planning Code with those of the General Plan LUTE.

17.10.280 - Fast-food restaurant commercial activities.

B. Except as may otherwise be allowed in OMC Chapters 5.49, 5.51, 8.09 and 9.52, tThe sale of ready-to-consume prepared foods from trucks, pushcarts or other movable equipment located on <u>public or private</u> property on a semi-permanent basis during hours of operation. Vehicular food vending generally has the following characteristics:

For the benefit of the public, Staff proposes to insert a reference to provisions related to food vending found in the Oakland Municipal Code (OMC).

17.10.440 Transient Habitation Commercial Activities.

Transient Habitation Commercial Activities include the provision of lodging services to transient guests on a less-than-weekly basis, other than in the case of activities classified by Section 17.10.120 Semi-Transient Residential Activities or Section 17.10.125 Bed and Breakfast Commercial-Residential Activities. Examples include hotels and motels. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. Staff inserted "hotels and motels" as examples of types of Transient Habitation Commercial Facilities and corrected the designation of the Bed and Breakfast classification from Commercial to Residential to be consistent with the rest of the Code.

17.10.480 - Automobile and other light vehicle repair and cleaning commercial activities. Automobile and Other Light Vehicle Repair and Cleaning Commercial Activities include the major repair or painting of motor vehicles that have a gross vehicle weight <u>rating of less</u> than fourteen thousand (14,000) pounds, including body work and installation of major accessories, as well as the washing and polishing of motor vehicles. This classification does not include vehicle dismantling or salvage and tire re-treading or recapping. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. Staff proposes to insert the words "rating of" for consistency with other related sections and with industry standards.

17.10.500—Transport-and-warehousing-commercial-nctivitics-(does-not-apply-to-the-CIX-1, CIX-2, fC, or-IQ-zones).

Staff proposes to delete this section because it has been replaced by the new industrial Use Classification adopted in 2009. This classification had been kept in the Code because the classifications in the M-10 through M-40 Zones had not been updated to include the new classifications. This Zoning update includes updates to the M-10 through M-40 Zones such that the Transport and warehousing classification is obsolete.

17.13.060 - Special regulations for mini-lot-ond-planned unit developments.

A--Mini-tot-Developments.-In-mini-hat-devetapmanas, certuin-regulations-that-otherwise-apply to individe and lots in-the-RH-zames-may-he-waived-or-modified-when-and-ns-preserihed-in Section-i-7-102-320

B.—Planned Unit Developments. Large integrated developments shall be subject to the Planned Unit Development regulations in Chapter 17.142 if they exceed the sizes specified therein. In developments which are approved pursuant to said regulations, certain uses may be permitted in addition to those otherwise allowed in the RH zones, and certain of the other regulations applying in said zone may be waived or modified.

Staff proposes to remove the reference to mini-lots from the RH (Hillside Residential) zones. This inclusion was an oversight in a previous zoning update and the change is consistent with the old R-1, R-10, and R-20 zones.

17.58 Central Business District.

Tabic 17.58.04, note 2: For mixed use projects in the Central Business District (CBD) zones, the allowable intensity of development shall be measured according to both the maximum nonresidential Floor Area Ratio (FAR) allowed by the zone and the maximum residential density allowed by the zone. The total lot area shall be used as a basis for computing both the nonresidential FAR and the residential density. No-portion-of-lot-urea-used-to-mect-the-density requirements-for a Residential-Facility-shall-be-used-us-a-basis for-computing, through the maximum-floor-area-ratio, the maximum-amount-of floor-area-for-any-non-residential facility on the same lot, and visa versa.

Staff proposes to delete the existing language regarding density calculations for mixed use projects located in the Central Business District, because the previous language regarding the method of measuring allowable intensity and density in the CBD was incorrect. This revision corrects the previous error and makes the language consistent with that used in Section 17.106.030(A).

17.106.030 - Maximum density and floor-area ratio on lots containing both Residential and Nonresidential Facilities.

A. Portion of Lot Area Used in Computing Density in the Central Business District and Jack London District. For mixed use projects in the <u>Central Business District</u> CBD-zones-and Jack London district, the allowable intensity of development <u>shall be measured according to both</u> the maximum nonresidential Floor Area Ratio (FAR) allowed by the zone and the maximum residential density allowed by the zone. The total lot area shall be used as a basis for computing both the nonresidential FAR and the residential density. mny-be-measured-according-to-the maximum-FAR-allowed-by-the-zone-without-a-separate-residential-density-calculation, provided-the-maximum-number-of-units-pursuant-to-the-residential-density-allowed-by-the Gencrnl-Plan-and-Estuary Policy-Plan is-not-exceeded-(The Central Business District is that area identified as part of the Land Use and Transportation Element Land Use Diagram of the General Plan. The Jack London district is that area identified as part of the Estuary Policy Plan and adopted as part of the General Plan-).

Staff proposes to delete the existing language and insert new language clarifying density calculations for mixed use projects in the CBD and Jack London District.

17.130.060 - Obligation of applicant to defend, indemnify, and hold harmless the City of Oakland.

A. To the maximum extent permitted by law, the applicant shall defend (with counsel reasonubly acceptable to the City), indemnify, and hold harmless the City of Oakland, the Oakland City Council, the City of Oakland Redevelopment Agency, <u>City of Oakland</u> <u>Redevelopment Successor Ageocy</u>, the Oakland City Planning Commission and its respective agents, officers, <u>volunteers</u> and employees (hereafter collectively called City) from any liability, damages, claim, judgment, loss (direct or indirect), action, causes of action or proceeding (including legal costs, attorneys' fees, expert witness or consultant fees, City Attorney or staff time, expenses or costs) (collectively called "Action") against the City te attack, set aside, void

or annul, (1) an approval by the City relating to a development-related application <u>or</u> subdivision <u>and/or a (Lease) Disposition and Development Agreement</u>; or (2) implementation of <u>such.</u> an-approved-development-related-project. The City may elect, in its sole discretion, to participate in the defense of said Action and the applicant shall reimburse the City for its reasonable legal costs and attorneys' fees.

Staff added clarifications to the above language.

17.138.015- Projects eligible and special regulations for projects with development agreements Staff proposes to relocate Section 17.102.310 to a new Section 17.138.015(A) and (B), and modify the language to add "successor to the" Redevelopment Agency and delete "major" in reference to affected permit types.

Chapter 17.142 — <u>Mini Lot and Planned Unit Development Regulations</u>. Staff added a Mini Lot section (Article 11) to Chapter 17.142 to clarify applicability and basic provisions.

17.142.030 – Developments for which <u>planned unit development permit</u> approval is required <u>or</u> requested.

<u>A.</u> The following developments are permitted only upon the granting of a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140:... Other-largc, integrated-developments-arc-permitted-without-such-a-permit, but-shall-be-subjcel to-nil-regulations-generally-applying-in-the-zone-in-which-they-are-located.

1A. Any planned unit development incorporating any of the bonuses set forth in Section 17.142.100.

2B. Any integrated development which is primarily designed for or occupied by Commercial Activities, which is located in any commercial zone, and which is developed under unified control, in accordance with a comprehensive plan, on a single tract with sixty thousand (60,000) square feet or more of land area, or on two or more tracts which total such area and which are separated only by a street or other right-of-way.

B. Unless required by the Planning Director, other large, integrated developments involving the same minimum land area thresholds of a planned unit development, as defined in Section 17.142.020, arc permitted without such a permit. However, an applicant of such a development may request a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140, but shall be subject to all regulations generally applying in the zone in which they are located.

Staff added a new subsection B to clarify that the Planning Director may require other very large integrated developments to be reviewed under the planned unit development permit procedure, where it is deemed appropriate; otherwise, an applicant of such a development may request a PUD permit.

B. Staff proposes to move or combine the following code chapters and sections as noted:

Chapter-1-7-68-M-20-Industrial-Zones-Regulations

Chapter-17-70-M-30-Industrial-Zones-Regulations

Chapter-17-72-M-10-Industrial-Zones-Regulations

Chapter 17.72 M-20, M-30, and M-40 Industrial Zones Regulations.

Staff proposes to combine these three Industrial zones into one chapter that will be in the same format as the most recently adopted zones. Also, Staff added sidewalk cafés as a permitted facility type in the M- Zones to correct an oversight (see 17.72 under Substantive Changes section).

17.102.070 - Application of zoning regulations to lots divided by zone boundaries Staff proposes to move this entire section to 17.154.060 Zoning maps.

1-7.102.080 -- Permitted-nnd-conditionally-permitted-uses

L

<u>17.07.065 - Permitted and conditionally permitted uses.</u>

A. Other Uses Prohibited. Except as otherwise provided in Sections 17.102.040-17.114.030and 17.102.070, 17.154.060, the nonconforming use regulations in Chapter 17.114, and the planned unit development regulations in Chapter 17.142, or as authorized under Section 17.102.340, 17.138.015, the development agreement procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no land shall be improved or used for any activity or facility which is not listed as permitted or conditionally permitted in the applicable individual zone regulations or development control maps.

B. Relationship Between Activities and Facilities. A use must qualify under the zoning regulations both as an activity and as a facility. A permitted or conditionally permitted activity may be accommodated or served only by a permitted facility or, upon the granting of a conditional use permit, by a conditionally permitted facility; and a permitted or conditionally permitted facility may accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditional use permit, a conditionally permitted activity.

Staff proposes to move Section 17.102.080 to Chapter 17.07 Title purpose and scope of zoning regulations, and correct the references as shown.

17.102.100 - Conditions for accessory parking serving activities which are not themselves allowed.

Staff proposes to move this entire section to 17.116 Off-Street Parking and Loading\_Requirements.

17.102.220 - Special regulations applying to mining and quarrying extractive activities. Staff proposes to move this entire section to its own new chapter (17.155), because it is so lengthy.

17.102.265 – Occupancy of a One-Family Dwelling Residential Facility by a Residential Care Residential Activity.

Deleted, and combined with Section 17.09.040, definition of "Dwelling unit".

17.102.280 - Rules for determining the number of habitable rooms in Residential Facilities. Staff proposes to move this entire section to Chapter 17.09 Definitions ("Habitable room" and "Habitable rooms, number of").

17.102.310 - Special regulations for certain projects with development agreements. Staff proposes to move this section to Chapter 17.138 Development Agreement Procedure.

17.102.320 - Conditional use permit for waiver of certain requirements in mini-lot developments.

Staff proposes to move this section to 17.142 Planned Unit Development Regulations so that both types of integrated developments, PUD's and mini-lots, are in the same chapter.

17.102.330 – Conditional use permit for waiver of certain requirements with parcel division between existing buildings.

Deleted, because it already exists in Chapter 17.106, General Lot, Density and Area Regulations, Section 17.106.010(B).

17.102.380 – Special regulations applying to truck-related activities in the West Oakland Community Development District.

Deleted, and added as a footnote to any zone that allows trucking activities.

17.102.400 - Special design requirements for lots that contain Residential Facilities and no Nonresidential Facilities.

Staff proposes to move the paving restrictions portion of this section to a footnote in the zoning chapters that allow residential facilities; Screening of utility meters and trash containers moved to Section 17.124.045 Landscaping and Screening standards. Exterior security bars and grills regulations would stay in this Section, 17.102.400, and the section renamed to "Exterior security bars and related devices".

17.102.420 - Special design requirements for lots located in residential and commercial zones and the OS, S-1, S-2, S-3 and S-1S zones.

Deleted, because it is redundant with Section 17.108.140.

17.102.260 "Occupancy-of-n-dwelling-unit"

17.09.040 Definitions.

"Dwelling unit" means a room or suite of rooms including one and only one kitchen, except as otherwise provided in Section 17.102.270, and designed or occupied as separate living quarters for one-of-the-persons or groups-specified in-Section-17.102.260, one person or family; or, where the facility occupied is a One-Family Dwelling, such family and not more than three (3) boarders, roomers, or lodgers, where access to ail rooms occupied by such boarders, roomers, or lodgers is had through the main entrance of the dwelling unit.

Section 17.102.260 "Occupancy of a dwelling unit" is essentially a definition of a dwelling unit; therefore, Staff proposes to delete Section 17.102.260 and combine its language with the definition of "Dwelling unit" in Section 17.09.040.

C. Staff proposes to delete the following sections in their entirety because they no longer serve any purpose:

17-10-760-Shopping-Center-Facility.

A.-Shopping-Center-Nonresidential-Facility-is-a-complex of one or-more-retail-buildings-and related-facilities forming-a-central-retail-market-within-a-given-area-and-having-a-common parking-area.

17-102-020-Supplemental-Zoning-Provisions

The definitions, special-use-classification-rules, and other-provisions-set-forth-in-Chapters 17-07, 17-09-and 17-10; the provisions of 17.108.130; the nonconforming-use-regulations-in Chapter-17-114; the rezoning, variance, and other-provisions-set-forth-in-Chapters 17.130 through 17.152; and the provisions of the zoning-maps-in-Chapter 17.154-shall-apply throughout-the-city. The-planned-unit-development-regulations-iu-Chapter 17.1432-shall-apply in-the zones-and-situations-specified-in-said-chapters. The provisions of development-control maps-arc-in-addition-to, or-supersede-when-so-specified, the regulations applying-in-the zones covering-the-same-area.

**D.** Staff proposes to delete Section 17.102.130 in its entirety because there is already a regulation that addresses this issue in 17.10.040 Accessory Activities.

17.102.130 Time limit on operation of subdivision sales offices Residential Zones.

- E. Staff proposes to move all of the following sections to a new Chapter (Chapter 17.103 Special regulations and findings for certain use classifications), that will contain special regulations that apply to particular use classifications that are listed in Chapter 17.10:
  - 17.102.210 Special regulations applying to Convenience Markets, Fast-Food Restaurants, certain establishments selling alcoholic beverages, providing mechanical or, and electronic games.

#### Oakland City Planning Commission

Case File Number ZT13-065

- 17.102.212 Special Regulations Applying to Residential Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities.
- 17.102.290 Special regulations for Drive-Through Nonresidential Facilities.
- 17.102.335 Standards for Sidewalk Cafes.
- 17.102.360 Secondary Units.
- 17.102.370 Conditional use permit for hotels and motels.
- 17.102.390 Regulations Applying to Special Health Care Civic Activities.
- 17.102.430 Regulations applying to check cashier and/or check cashing activity.
- 17.102.440 Special regulations for primary collection centers in all zones.
- F. The following new code section and related code changes are proposed to: (a) codify the longstanding practice of processing minor permits with major permits; and (b) clarify that this only applies to the procedural requirements, but that the substantive requirements of minor permits (i.e., findings) still apply:

17.130.090 – Minor land use permits considered concurrently with Major permits. A. Any Minor land use related permit and/or approval that is related to a development application that also includes any Major land use related permit and/or approval shall he considered concurrently with the Major land use related permit and/or approval, and shall follow all procedural requirements associated with City Planning Commission decisions. In this case, the entire application shall initially be considered by the City Planning Commission and may be appealed to the City Council, in accordance with the requirements for Major land use related permit and/or approval or discretionary actions.

**B.** Any Minor land use related permit referred to the City Planning Commission for initial decision in order to be considered concurrently with any Major land use related permit and/or approval shall still be considered a Minor land use related permit and/or approval, and the required findings for said Minor land use related permit and/or approval shall apply.

17.134.020 - Definition of major and minor conditional use permits.

Staff proposes to delete Subsection 17.134.020(3)(h) to clarify that a minor conditional use permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit.

17:134.040(B) – Procedures for consideration – Minor Conditional Use Permits. Staff proposes to add language to clarify that a minor conditional use permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major CUP procedure.

17.136.040(D) - Regular Design Review - Procedure for Consideration.

Staff proposes to add language to clarify that a regular design review permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major Permit procedure.

17.148.020 - Definition of major and minor variances.

7. Any variance application that requires development of an Environmental Impact Report;

8.—Any variance application-referred-by-the-Director-of-City-Planning-to-the-City-Planning Commission for initial decision-pursuant to Section 17.148.040(B)(1).

Staff proposes to delete Subsection 7 above to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit. Staff proposes to also delete Subsection 8 above to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit.

17.148.040 – Procedure for consideration.

Staff proposes to add language to subsection B, Minor Variances, to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major Variance procedure.

#### G. Other general clean-up:

- The word "Section" was inserted prior to the section number where appropriate (e.g., "17.105.010" changed to "Section 17.105.010").
- · Numbers were placed in parentheses and inserted after spelled-out numbers where appropriate (e.g., "35 feet" was changed to "thirty-five (35) feet".
- When massage, adult entertainment; check cashing; special health care, fast food are permitted or conditionally permitted in a zone, new notes were added in the Activity Tables directing the reader to further regulations in Chapters 17.102 or 17.103.
- 17.65.030, Note L3 Staff added a missing note for General Warehousing Activities in order to fix an error during the HBX Zones text amendment.
- Staff removed references in other parts of the Code that were related to previously-deleted sections of Chapter 17.108. (In the spring of 2012, the City Council adopted an ordinance that removed the following Code sections: 17.108.040; 17.108.050; 17.108.060; 17.108.070; 17.108.090; 17.108.100, and 17.108.110).
- Staff corrected references to the relocated sections of 17.102 in the new M-20, M-30, M-40 Chapter (17.72).
- Staff added applicable language regarding front yard paving restrictions into the code from the previous Section 17.102.400 to the new footnotes in the Zoning chapters that allow Residential facilities.
- Staff relocated Special Health Care Civic Activities from Section 17.134.020(2)(b)(iii) into a new Section 17.134.020(2)(a)(xiii), under Definition of major conditional use permits. This Activity triggering a Major CUP was inadvertently misplaced under the Facility types subsection rather than Activity types.

#### **11. MENOR-SUBSTANTIVE CHANGES**

The following section summarizes the proposed Minor-Substantive changes to the Planning Code. The Code language is in **bold** type and explanations are in normal type.

17.09.040 Definitions.

"Gross vehicle weight rating" means the vehicle weight specified by the manufacturer as the maximum loaded weight (truck plus cargo) of a single vehicle.

Staff proposes to add the above definition for "gross vehicle weight rating". This is the industry standard and creates consistency in the truck activity classifications.

Table 17.35.04 - Height, Floor Area Ratio (FAR), Density, and Open Space Regulations. Staff proposes to correct the density for rooming units in the 120- and 160-foot height areas from "one unit per 225 sf of lot area" to "one unit per 110 sf of lot area", in order to fix a previous typographical error that reduces density for rooming units located in the tallest height areas.

#### Table 17.58.03: Property Development Standards.

Staff proposes to change the required percentage of façade transparency in the CBD zones to be consistent with standards in other zones.

17.54 - C-40 Community Thoroughfare Commercial Zone Regulations.

17.56 - C-45 Community Shopping Commercial Zone Regulations.

17.65 - HBX Housing and Business Mix Commercial Zones Regulations.

17.72 - M-20, M-30 and M-40 Industrial Zones Regulations.

The updated Industrial classifications, created in 2008, were not included in the activity tables for the C-40, C-45, HBX and M-Industrial zones. Staff proposes to integrate these activities into the chapters to correct this oversight.

47-102-250-Mux4mum-density-and-Hoor-arcu-ratio-during-construction

Staff proposes to delete Section 17.102.250 because it has never been an issue and has proven difficult to enforce.

17.102.040-17.114.030 - Effect of prior permits.

A. Fundamental Vested Rights Not Abrogated By Code Adoption or Amendment. The adoption of, or amendment to, the Planning Code (Chapter 17), including without limitation, ordinances enacled pursuant to Oakland City Charter section 213 (Emergency Ordinances) and/or ordinances enacted pursuant to Government Code section 65858 (Urgency Measure/moratoria), shall not abrogate any fundamental vested rights established pursuant to State law, including, without limitation, those established pursuant to the prior valid adoption and execution of a development agreement under Section 17.138.015 and the development agreement procedure in Chapter 17.138. Absent the prior establishment of such vested rights, any Planning Code adoption or amendment shall apply.A. Building-and-Sign-Permits and Development-Agrocments. Whenever-anv-subsisting building-permit-or-sign-permit-has-been-lawfully-issued-beforehnnd-or-whenever-a-subsisting development-agreement-has-been-approved-beforehand-under-Section-1-7.102-310-und-the development-agreement-procedure in-Chapter-17-138, neithor-thc-original-od-option-of-the zoning regulations-nor-the-adoption-of-any-subsequent-rezoning-or-other-nmendment-thereto-shall prohibit-the-construction,-other-development-or-change,-or-use-authorized-hy-suid-permit-or ngreement. The uses as they result shall be deemed nonconforming uses and subject to the uoneonforming-use-regulations-iu-Chapter-17-1-14, except-to-the-extent-that-they-arc-authorized-by a-subsisting-comitional-use-pernit,-development-agreement,-vuriance,-or-other-special-zoning approval.

B. Alcoholic Beverage Control Licenses. On premises for which a valid state of California Alcoholic Beverage Control license had been issued, and which premises had been used in the exercise of the rights and privileges conferred by the license at a time immediately prior to the effective date of the applicable provisions of Section-17-102-240 <u>17.103.030</u>, the premises may hereafter be used in the exercise of the same rights and privileges without requiring a conditional use permit or having to meet the provisions of the aforesaid section. Such-use<u>The uses as thev</u> result that do not conform to the zoning regulations shall be deemed a nonconforming use and subject to the nonconforming use regulations, except as otherwise provided in Sections 17.114.020 and 17.114.030. For the purposes of this subsection, the word "premises" shall mean and include only the actual space within a building devoted to the sale of alcoholic beverages on said effective date.

Staff proposes to delete Section 17.102.040 above and move the text to Chapter 17.114 - Non-Conforming Uses. The resulting Section 17.114.030 above is proposed to include changes to subsection (A), Fundamental Vested Rights Not Abrogated By Code Adoption or Amendment, in order to be consistent with State law.

In addition, language was added to the following sections to codify and clarify the long-standing practice limiting evidence and testimony during appeals to evidence submitted in the appeal form itself and

presented: (1) prior to the close of the written public comment period for the underlying decision being appealed, in the case of appeals based on a decision by the Zoning Administrator or other administrative decisions, or (2) prior to the close of the City Planning Commission's public hearing for the underlying decision being appealed, in the case of appeals based on decisions made by the City Planning Commission, as applicable. The language added is also intended to eliminate the requirement that, if the City Council is unable to decide an appeal, the matter appear for a vote on each regular meeting of the Council thereafter until decided.

17.130.050 - Presentation of written and documentary evidence.

17.134.040 - Procedures for consideration.

17.134.060 - Appeal to Planning Commission-Minor Conditional Use Permits.

17.134.070 - Appeal to Council-Major Conditional Use Permits.

17.136.040 - Regular Design Review.

17.136.080 - Appeal to Planning Commission-Regular design review.

17.136.090 - Appeal to City Council-Regular design review.

17.140.030 - Preliminary Planning Commission action.

17.140.060 - Final Planning Commission action.

17.140.070 - Appeal to Council.

17.144.090 - Council action.

17.148.040 - Procedure for consideration.

17.148.060 - Appeal to Planning Commission-Minor variances.

17.148.070 - Appeal to Council-Major variances.

17.156.140 - Procedure for consideration of violations to performance standards.

17.156.150 - Procedure for consideration of violations to conditions of approval.

17.156.160 - Appeal to Planning Commission.

17.156.170 - Appeal on the revocation of a Deemed Approved Status to the City Council.

17.157.110 - Procedure for consideration of violations to performance standards.

17.157.120 - Procedure for consideration of violations of conditions of approval.

17.157.130 - Appeal to City Planning Commission.

17.157.140 - Appeal on the revocation of a Deemed Approved Status to the City Council.

#### **III. SUBSTANTIVE CHANGES**

The following section summarizes the proposed substantive changes to the Planning Code. The Code language is in **bold** type and explanations are in normal type.

17.10.485 – Taxi and light fleet-based service commercial activities.

Passenger transportation services, local delivery services, and other businesses that rely on fleets of three (3) or more vehicles with a gross vehicle weight rating ruted-capacity of less than twelve thousand-five-hundred (12,500) fourteen thousand (14,000) pounds. This classification includes parking, dispatching, and offices for taxicab and limousine operations, airport shuttles, medical transport, local messenger and document delivery services, janitorial services, and similar businesses. This classification does-not<u>only</u> includes towing operations when except-for-tow-truck services-where-vehicles are taken to off-site locations and the tow trucks do not exceed the above gross vehicle weight rating. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Staff proposes the above changes to this regulation to use measurements that are consistent with industry standards for Light Vehicles (e.g., pick-up trucks, ambulances and parcel delivery trucks).

17.10.585 Trucking and truck-related industrial activities.

Trucking and Truck-Related Activities include the provision of freight handling and shipping services by trucks as well as parking, maintenance, and services for trucks and other heavy vehicles and equipment. Each classification involves the use of trucks and other heavy vehicles that

have a gross vehicle weight rating greater than or equal to fourteen thousand (14,000) nounds. This classification also includes certain activities accessory to the above, as specified in Section 17,10.040.

Staff proposes changes to the Truck Activities regulations and use measurements in Section 17.10.585 that are consistent with industry standards.

 Table 17.58.03, note 4
 Property development standards.

In (he CBD-R zone, portions of a building over fifty-five (55) feel in height shall have a setback of at least one (1) foot from the required interior side vard for every five (5) four-(4)-feet that portion is above lifty-five (55) feet. This setback, however, need not exceed forty (40) feet. Also, see Section 17.108.080 for the required interior side and rear vard setbacks on a lot containing two or more living units and opposite a legally-required living room window. See Section 17.108.130 for allowed projections into required vards.

Staff proposes to change the step-back requirement in the CBD-R zone from 1 foot for every 4 feet of height above 55 feet to 1 foot for every <u>5 feet</u> of height above 55 feet. This reduced amount of step-back is more appropriate in a downtown context.

17.72 M-20, M-30, and M-40 Industrial Zones Regulations.

Staff added sidewalk cafés as a permitted facility type in the M- Zones to correct an oversight in a previous zoning code change that added restaurants as an allowed activity type in the M- Zones, but overlooked including sidewalk cafes as a permitted facility type. (Note: The remaining M- Zones are all on the water, including Jack London Square).

17.73 - CIX-1, CIX-2, IG and 10 Industrial Zone Regulations.

The following new Design Review Section (17,73.015) was added:

<u>A.</u> Except for projects that arc exempt from design review as set forth in Section 17.136.025, no Residential Facility, Designated Historic Property, Potentially Designated Historic Property, Telecommunications Facility, Sign, or other associated structure shall be constructed, established,

or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications regulations in Chapter 17.128, or the Sign regulations in Chapter 17.104.

B. No facility located within one hundred fifty (150) feet of any residential zone boundary and accommodating the following activities shall be constructed, established, or expanded in size unless plans for the proposal have been approved pursuant to the Regular Design Review procedure in Chapter 17.136:

1. Automobile and Other Light Vehicle Gas Station and Servicing Activity.

2. Automohile and Other Light Vehicle Repair and Cleaning Activity.

3. Freight/Truck Terminal.

4. Truck Yard.

5. Truck Weigh Stations.

6. Truck and Other Heavy Vehicle Sales, Rental, and Leasing.

7. Truck and Other Heavy Vehicle Service, Repair, and Refueling.

C. Establishment of a work/live unit shall only be permitted upon determination that the proposal conforms to the regular design review criteria set forth in the Regular Design Review procedure in Chapter 17.136 and to all of the additional criteria set forth in Section 17.73.040(D).

Staff proposes to add this language to provide for Design Review for Residential facilities and historic properties that are located in the CIX, IG an IO zones and to consolidate references to activities requiring Design Review that are already included in this Chapter as part of the Activities Table. This is considered a minor change because it does not apply to non-residential buildings that are not historic.

17.102.110(F),--luereased-Off-Street-Parking,--Off-street-parking-shall-be-provided-for-the proposed-development-in-an-amount-at-least-one-hundred-fifty-pcreent-(150%)-of-that-required-by the off-street-parking requirements in Chapter 17.116. Staff proposes to delete Section\_17.102.110(F)\_because it\_seems too onerous and will generally apply in the corridors where development is encouraged.

#### **IV. OTHER RELATED CHANGES**

Various changes were made to the code that relate and correspond to the amendments described above.

#### ENVIRONMENTAL DETERMINATION

The proposed amendments to the Zoning Code rely on the previously certified Final Environmental Impact Reports for the Land Use and Transportation Element of the General Plan (1998); the Oakland Estuary Policy Plan (1998); the West Oakland Central City East, Coliseum and Oakland Army Base Redevelopment Areas; the 1998 Amendment to the Historic Preservation Element of the General Plan; the 2007-2014 Housing Element Final EIR (2010); and various Redevelopment Plan Final EIRs (collectively, "EIRs"). The EIRs provide analysis of the environmental impacts of the proposed amendments and support all levels of approval necessary to implement the Zoning Code amendments. The proposed amendments to the Zoning Code would not result in any significant effect that has not already been analyzed in the EIRs, and there will be no significant environmental effects caused by the change that have not already been analyzed in the EIRs. As a result, none of the circumstances necessitating preparation of additional environmental review, as specified in CEQA and the CEQA Guidelines, including, without limitation, Public Resources Code Section 21166 and CEQA Guidelines Sections 15162 or 15163 are present in that (1) there are no substantial changes proposed in the project or the circumstances under which the project is undertaken that would require major revisions of the EIR due to the involvement of new significant environmental effects or a substantial increase in the severity of previously identified significant effects; and (2) there is no "new information of substantial importance," as defined in CEQA Guidelines Section 15162(a)(3). In addition, each as a separate and independent basis, this action is exempt from CEQA pursuant to CEQA Guidelines Sections 15183 (projects consistent with General Plan and Zoning) and 15061(b)(3) (no significant effect on the environment).

#### **KEY ISSUES AND IMPACTS**

The proposed Code amendments described in this staff report are intended to clean up and better organize the Planning Code by combining related sections where appropriate, deleting redundant and obsolete sections, making edits that clarify sections that are currently unclear, and making more substantive changes to make the Code more consistent with industry standards and good development practices.

Completing this code clean-up is a staff priority since other Code amendment initiatives that will come before the Commission and City Council this fall, such as those related to the adoption of Specific Plans, will need to incorporate and build upon these clean-up items.

#### RECOMMENDATIONS

- 1. Affirm staff's environmental determination;
- 2. Recommend that City Council approves the proposed text amendments; and
- 3. Find that existing regulations being amended or deleted are inadequate and otherwise contrary to the public interest.

Prepared by:

Ann Clevenger, Planner 111

Approved by:

Scott Miller, Zoning Manager

Approved for forwarding to the City Planning Commission by:

Rachel Flynn, Director Department of Planning and Building

LEGAL NOTICE: The decision of the City Planning Commission is final and not administratively appealable. Any party seeking to challenge such decision in court must do so within ninety (90) days of the date the decision is announced (Code of Civil Procedure Section 1094.6).

#### ATTACHMENTS:

- A. April 17, 2013 ZUC Staff Report
- B. April 17, 2013 Staff Memo to the Zoning Update Committee
- C. Summary Table of Code Changes
- D. Proposed Zoning Text Amendments

## MEMO

Date:4/17/13To:Zoning Update Committee (ZUC)From:StaffSubject:Changes to proposed code amendments presented at the 4/17/13 ZUC meeting.

The following two sets of proposed changes were determined to require further review and are therefore being pulled from the Zoning Update package. These changes will be considered and publicly reviewed in a future code amendment proposal.

#### 1. STAFF REPORT, page 9

New language was added to the following sections to require items sent to the Landmarks Board or Parks and Recreation Commission for review to automatically be sent back to either the Director or the Planning Commission for review after a 30-45 day period:

17.11.020 - Designation and mapping of parks by category

17.100B.030 - Required design review process

17.135.030 - Procedure for consideration

17.135.040 - Referral to Landmarks Preservation Advisory Board

17.136.055 - Special regulations for historic properties in the central business zones

17.136.060 - Review by Landmarks Board in certain cases

17.144.050 - Review by Landmarks Board in certain cases

17.144.055 - Review by Parks and Recreation Advisory Commission in certain cases 18.02.0140 - Appeal Procedures

#### 2. ATTACHMENT A, page 325

17.158.090 - Definitions.

Historic Property. "Historic property" are those properties that are:

A. dDesignated city landmarks: -pursuant-to-Section-17-1-02-030-of-this-code; is

B. iListed on the National Register of Historic Places

- <u>C. is-lL</u>isted as a California Registered Historical Landmark or is-a California Point of Historical Interest;
- D. is contributory to an S-7 Preservation Combining Zone pursuant to Section 17.84.010 of this code;

E. In receipt of has-received-an "A" or "B" rating by the Oakland Cultural Heritage Survey.

### Zoning Update Committee

STAFF REPORT

Michael Colbruno – Chair Blake Huntsman – Vice-Chair Vien Trnong April 17, 2013

Location:	Citywide
Item:	Discussion of proposed zoning text amendments to Chapter 17.102 and other
••	various parts of the Planning Code, including:
· .	1) Non-Substantive Changes (reformatting, reorganizing and improving the
	consistency of the Code);
	2) Minor-Substantive Changes (text changes to improve the interpretability,
	clarity, and flexibility of the Code); and,
	3) Substantive Changes (changes to the standards in the Code).
Applicant:	City of Oakland
Case File Number:	ZT13065
General Plan:	Citýwide
Zoning:	Citywide
Environmental Determination:	Several past environmental documents discussed and evaluated the
	environmental impacts associated with this proposal and reliance on them
	satisfies any requirements under the California Environmental Quality Act
	(CEQA), as specified by CEQA Guidelines Section 15183 "Projects Consistent
	with a Community Plan, General Plan or Zoning". Those environmental
	documents include, but are not limited to, the previously certified Final
	Environmental Impact Report (EIR) for the Land Use and Transportation
	Element of the Oakland General Plan, adopted in 1998.
Staff Recommendation:	Review, discuss and recommend approval to the Planning Commission
For Further Information:	Contact Ann Clevenger at 238-6980 or email <u>aclevenger@oaklandnet.com</u>

#### PROPOSED ZONING TEXT AMENDMENTS

The Land Use and Transportation Element (LUTE) of the City of Oakland General Plan contains the foilowing policy direction calling for a simple, user-friendly and easier-to-interpret Planning Code:

Since the existing Zoning Regulations are regarded as cumbersome and difficult to use, a simpler, more user-friendly code will he developed. (Pg. 170)

To comply with that directive, this report's proposed zoning text amendments are largely focused on improving consistency, reducing redundancies, and simplifying language of key chapters of the Planning Code. In particular, Chapter 17.102 - General Regulations Applicable to All or Several Zones, is the main subject of this report. In addition to cleaning up Chapter 17.102, other miscellaneous changes are included.

The proposed changes to the zoning text fall into three basic categories:

- I. Non-substantive Changes
- II. Minor-substantive Changes
- III. Substantive Changes

Non-substantive changes include reformatting, reorganizing and improving the internal consistency of the Planning Code and the consistency of the Planning Code with the General Plan Land Use and

#### Oakland Zoning Update Committee

Case File Numbers ZT13-065

Transportation Element (LUTE). Minor substantive changes include text changes to improve the interpretability, clarity, and flexibility of the Planning Code. Finally, staff proposes substantive changes to certain codes to improve the standards in the Planning Code. The content of these proposed changes is summarized in the following report. Please see Attachment A for the proposed zoning text amendments.

Staff requests authorization to make any non-substantive, technical conforming edits to the Planning Code that may have been overlooked in this update.

These proposed changes are scheduled to be presented to the City Planning Commission at a public hearing on May 15, 2013, and will ultimately be presented to the City Council for final approval.

#### I. NON-SUBSTANTIVE CHANGES

The following section summarizes the proposed Non-Substantive changes to the Planning Code. The Code language is in bold type and explanations are in normal type.

A. Staff proposes to modify the following chapters and sections to improve the clarity and internal consistency of the Code and/or its consistency with the General Plan:

17.07.030 – Purposes of the zoning regulations

Section 17.07.030(A). To promote the achievement of the proposals, <u>policies and objectives</u> of the Oakland General Plan

By inserting the phrase "policies and objectives" to this Section, this amendment would bring consistency between the objectives of the Planning Code with those of the General Plan LUTE.

17.10.2S0 - Fast-food restaurant commercial activities.

B. <u>Except as may otherwise be allowed in OMC Chapters 5.49, 5.51, and 8.09, t</u>The sale of ready-to-consume prepared foods from trucks, pushcarts or other movable equipment located on <u>public or private property on a semi-permanent basis during hours of operation</u>. Vehicular food vending generally has the following characteristics:

For the benefit of the public, Staff proposes to insert a reference to provisions related to food vending found in the Oakland Municipal Code (OMC).

17.10.440 Transient Habitation Commercial Activities

Transient Habitation Commercial Activities include the provision of lodging services to transient guests on a less-than-weekly basis, other than in the case of activities classified by Section 17.10.120 Semi-Transient Residential Activities or Section 17.10.125 Bed and Breakfast Commercinl-Residential Activities. Examples include hotels and motels. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. Staff inserted "hotels and motels" as examples of types of Transient Habitation Commercial Facilities and corrected the designation of the Bed and Breakfast classification from Commercial to Residential to be consistent with the rest of the Code.

17.13.060 - Special regulations for mini-lot-and-planned unit developments.

Staff proposes to remove the reference to mini-lots from the RH (Hillside Residential) zones. This inclusion was an oversight in a previous zoning update and the change is consistent with the old R-1, R-10, and R-20 zones.

17.58 Central Business District

Table 17.58.04, note 2: For mixed use projects in the Central Business District (CBD) zones, the allowable intensity of development shall be measured according to the maximum Floor Area
Case File Numbers ZT13-065

Ratio (FAR) allowed by the zone without a separate residential density calculation, provided the maximum number of units is not exceeded. No-portion-of-lot-area-used-to-moet-the-density requirements-for-a-Residential-Faeility-shall-be-used-as-a-basis-for-computing,-through-the maximum floor-area-rntio,-the-nuiximum-amount-of-floor-area-for-any-nonresidential-facility on-the-same-lot,-and-visa-versa.

Staff proposes to delete the existing language regarding density calculations for mixed use projects located in the Central Business District and replace it with language that is consistent with that used in 17.106.030(A).

17.130.060 - Obligation of applicant to defend, indemnify, and hold harmless the City of Oakland

A. To the maximum extent permitted by law, the applicant shall defend (with counsel reasonably acceptable to the City), indemnify, and hold harmless the City of Oakland, the Oakland City Council, the City of Oakland Redevelopment Agency, <u>Citv of Oakland</u> <u>Redevelopment Successor Agency</u>, the Oakland City Planning Commission and its respective agents, officers, <u>volunteers</u> and employees (hereafter collectively called City) from any liability, damages, claim, judgment, loss (direct or indirect), action, causes of action or proceeding (including legal costs, attorneys' fees, expert witness or consultant fees, City Attorney or staff time, expenses or costs) (collectively called "Action") against the City to attack, set aside, void or annul, (1) an approval by the City relating to a development-related application <u>or</u> subdivision <u>and/or a (Lease) Disposition and Development Agreement;</u> or (2) implementation of <u>such.</u> an-unproved-development-reloted-project-The City may elect, in its sole discretion, to participate in the defense of said Action and the applicant shall reimburse the City for its reasonable legal costs and attorneys' fees.

B. Within ten (10) calendar days of the filing of any Action as specified in subsection A above, the applicant shall execute a Letter of Agreement with the City, acceptable to the Office of the City Attorney, which memorializes the above obligations. These obligations and the Letter of Agreement shall survive termination, extinguishment or invalidation of the approval. Failure to timely execute the Letter of Agreement does not relieve the applicant of any of the obligations contained in this Section or any other requirements or conditions of approval that may be imposed by the City.

Staff added clarifications to the above language.

17.102.310—Special regulations-for-certain-projects-with-development-agreements <u>17.138.015–Projects eligible and special regulations for projects with development agreements</u> Staff proposes to relocate Section 17.102.310 to a new Section 17.138.015(A) and (B), and modify the language to add "successor to the" Redevelopment Agency and delete "major" in reference to affected permit types.

Chapter 17.142 — <u>Mini Lot and</u> Planned Unit Development Regulations Staff added a separate mini-lot section in the chapter to clarify applicability and basic provisions.

17.142.030 – Developments for which <u>planned unit development permit</u> approval is required <u>or</u> requested

Staff added a new subsection B to clarify that:

B. Unless required by the Planning Director, other large, integrated developments involving the same minimum land area thresholds of a planned unit development, as defined in Section 17.142.020, are permitted without such a permit. However, an applicant of such a development may request a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140, but shall be subject to all regulations generally applying in the zone in which they are located.

Case File Numbers ZT13-065

Page 4

B. Staff proposes to move or combine the following code chapters and sections as noted:

Chapter-17:68-M-20-Industrial-Zones-Regulations Chapter-17:70-M-30-Industrial-Zones-Regulations Chapter-17:72-M-40-Industrial-Zones-Regulations Chapter 17:72 M-20, M-30, and M-40 Industrial Zones Regulations Staff proposes to combine these three Industrial zones into one chapter that will be in the same format as the most recently adopted zones.

17.102.070 - Application of zoning regulations to lots divided by zone boundaries Staff proposes to move this entire section to 17.154.060 Zoning maps.

17.07.065 17-102.080 - Permitted and conditionally permitted uses

A. Other Uses Prohibited. Except as otherwise provided in Sections <u>17.114.030</u> 17-102-040 and <u>17.154.060</u>, 17-102-070, the nonconforming use regulations in Chapter 17.114, and the planned unit development regulations in Chapter 17.142, or as authorized under Section <u>17.138.015</u>, 17-102-3-10, the development agreement procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no land shall be improved or used for any activity or facility which is not listed as permitted or conditionally permitted in the applicable individual zone regulations or development control maps.

B. Relationship Between Activities and Facilities. A use must qualify under the zoning regulations both as an activity and as a facility. A permitted or conditionally permitted activity may be accommodated or served only by a permitted facility or, upon the granting of a conditional use permit, by a conditionally permitted facility; and a permitted or conditionally permitted facility may accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditional use permit, a conditionally permitted activity.

Staff proposes to move Section 102.080 to 17.07 Title purpose and scope of zoning regulations, and correct the references as shown.

17.102.100 - Conditions for accessory parking serving activities which are not themselves allowed

Staff proposes to move this entire section to 17.116 Off-Street Parking and Loading\_Requirements.

17.102.220 - Special regulations applying to mining and quarrying extractive activities Staff proposes to move this entire section to its own, new chapter 17.155, because it is so lengthy.

17.102.280 - Rules for determining the number of habitable rooms in Residential Facilities Staff proposes to move this entire section to Chapter 17.09 Definitions ("Habitable room" and "Habitable rooms, number of").

17.102.310 - Special regulations for certain projects with development agreements Staff proposes to move this section to Chapter 17.138 Development Agreement Procedure.

17.102.320 - Conditional use permit for waiver of certain requirements in mini-lot developments

Moved to 17.142 Planned Unit Development Regulations so that both types of integrated developments, PUD's and mini-lots, are in the same chapter.

17.102.400 - Special design requirements for lots that contain Residential Facilities and no Nonresidential Facilities

Paving restrictions moved to a footnote in the zoning chapters that allow residential facilities; Screening of utility meters and trash containers moved to Chapter 17.124 Landscaping and Screening standards. (Exterior security bars and grills regulations stay in 17.102.400).

# 17.102.260-"Occupancy-of-u-dwelling-unit"

17.09.040 Definitions

"Dwelling unit" means a room or suite of rooms including one and only one kitchen, except as otherwise provided in Section 17.102.270, and designed or occupied as separate living quarters for one-of the persons or groups-specified-in-Section-17.103.260, one person or family; or, where the facility occupied is a One-Family Dwelling, such family and not more than three (3) boarders, roomers, or lodgers, where access to all rooms occupied by such boarders, roomers, or lodgers is had through the main entrance of the dwelling unit.

Section 17.102.260 "Occupancy of a dwelling unit" is essentially a definition of a dwelling unit; therefore, Staff proposes to delete Section 17.102.260 and combine its language with the definition of "Dwelling unit" in Section 17.09.040.

C. Staff proposes to delete the following sections in their entirety because they serve no purpose:

# 17.10.760-Shopping-Center-Facility.

A. Shopping-Center-Nonresidential-Facility-is-a-complex-of-one-or-more-retail-buildings-und related-facilities-forming-a-central-retail-market-within-a-given-area-und-having-a-commoo parking-area-

#### 17.102.020-Supplemental-Zoning-Provisions

The definitions, special use classification-rules, ond-other-provisions-set-forth-in-Chapters 17.07, 17.09 and 17.10; the provisions of 17.108.130; the nonconforming-use regulations-in Chapter-17.114; the rczoning, vuriance, and other provisions-set-forth-in-Chapters-17.130 through-17.153; and the provisions of the zoning-maps-in-Chapter-17.151 shall-apply tbroughout the city, The-planned-unit-development-regulations-in-Chapter-17.1432-shall-apply in-the zones-and-situations specified-in-said-chapters. The provisions of development-control maps-arc-in-addition-to, or supersede when so specified, the regulations-applying-in-the zones covering-the-same-area.

D. Staff proposes to delete the following section in its entirety because there is already a regulation that addresses this issue in 17.10.040 Accessory Activities.

17-102-130-Time-limit-on-operation-of-sulnlivision-sales-offices---Residential-Zones.

- E. Staff proposes to move all of the following sections to a new Chapter 17.103 Special regulations and findings for certain use classifications that will contain special regulations that apply to a particular use classification that is listed in Chapter 17.10:
  - 17.102.210 Special regulations applying to Convenience Markets, Fast-Food Restaurants, certain establishments selling alcoholic beverages, providing mechanical or, and electronic games
  - 17.102.212 Special Regulations Applying to Residential Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities
  - 17.102.290 Special regulations for Drive-Through Nonresidential Facilities

17.102.335 - Standards for Sidewalk Cafes

17.102.360 - Secondary Units

Case File Numbers ZT13-065

- 17.102.370 Conditional use permit for hotels and motels
- 17.102.390 Regulations Applying to Special Health Care Civic Activities
- 17.102.430 Regulations applying to check cashier and/or check cashing activity

17.102.440 - Special regulations for primary collection centers in all zones

F. The following new code section and related code changes are proposed (a) to codify the longstanding practice of processing minor permits with major permits; and (b) to clarify that this only applies to the procedural requirements, but that the substantive requirements of minor permits (i.e., findings) still apply:

17.130.090 - Minor land use permits considered concurrently with Major permits.

- A. Any Minor land use related permit and/or approval that is related to a development application that also includes any Major land use related permit and/or approval shall be considered concurrently with the Major land use related permit and/or approval, and shall follow all procedural requirements associated with City Planning Commission decisions. In this case, the entire application shall initially be considered by the City Planning Commission and may he appealed to the City Council, in accordance with the requirements for Major land use related permit and/or approval or discretionary actions.
- B. Any Minor land use related permit referred to the City Planning Commission for initial decision in order to be considered concurrently with any Major land use related permit and/or approval shall still be considered a Minor land use related permit and/or approval, and the required findings for said Minor land use related permit and/or approval shall apply.

17.134.020 – Definition of major and minor conditional use permits. Staff deleted Subsection 17.134.020(3)(h) to clarify that a minor conditional use permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit.

17.134.040(B) – Procedures for consideration – Minor Conditional Use Permits Staff added language to clarify that a minor conditional use permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major CUP procedure.

17.136.040(D) – Regular Design Review – Procedure for Consideration Staff added language to clarify that a regular design review permit referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major Permit procedure.

17.148.020 - Definition of major and minor variances.

7. Any-variance-application-that-requires-development-of-an-Environmental-Impaet-Report;

8.—Any-variance-application-referred-by-the-Director-of-City-Planning-to-the-City-Planning Commission-for-initial-decision-pursuant-to-Section-17.148.040(B)(1).

Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.

Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit.

17.148.040 – Procedure for consideration

Staff added language to subsection B, Minor Variances, to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor permit, but shall simply be processed according to the Major Variance procedure.

Case File Numbers ZT13-065

G. Other general clean-up:

- Staff inserted the word "Section" prior to the section number where appropriate (e.g., "17.105.010" changed to "Section 17.105.010").
- Numbers were placed in parentheses and inserted after spelled-out numbers where appropriate (e.g., "35 feet" was changed to "thirty-five (35) feet".
- When massage, adult entertainment; check cashing; special health care, fast food are permitted or conditionally permitted in a zone, new notes are added in the Activity Tables directing the reader to further regulations in Chapters 17.102 or 17.103.
- 17.65.030, Note L3 Staff added a missing note for General Warehousing Activities in order to fix an error during the HBX Zones text amendment.

## 11. MINOR-SUBSTANTIVE CHANGES

The following section summarizes the proposed Minor-Substantive changes to the Planning Code. The Code language is in bold type and explanations are in normal type.

## 17.09.040 Definitions

Staff proposes to add the following definition for "gross vehicle rating":

"Gross vehicle weight rating" means the vehicle weight specified by the manufacturer as the maximum loaded weight (truck plus cargo) of a single vehicle.

This is the industry standard and creates consistency in the truck activity classifications.

Table 17.35.04 - Height, Floor Area Ratio (FAR), Density, and Open Space Regulations Staff proposes to adjust/increase the density for rooming units in the 120- and 160-foot height areas from "one unit per 225 sf of lot area" to "one unit per 110 sf of lot area", in order to correct a previous typographical error that reduces density for rooming units located in the tallest height areas.

## Table 17.58.03: Property Development Standards

Staff proposes to change the required percentage of façade transparency in the CBD zones to be consistent with standards in other zones.

The updated Industrial classifications, created in 2008, were not included in the activity table for the HBX, M-10 to M-40 zones, and the C-45 and C-40 zones. Staff proposes to integrate these activities into the chapters to correct this oversight.

17-102.250-Muximum-density-und-floor-area-ratio-during-construction Staff proposes to delete this section because it has never been an issue and has proven difficult to enforce.

## 17.114.030 17-102.040 -- Effect of prior permits.

A. Fundamental Vested Rights Not Abrogated By Code Adoption or Amendment. The adoption of, or amendment to, the Planning Code (Chapter 17), including without limitation, ordinances enacted pursuant to Oakland City Charter section 213 (Emergency Ordinances) and/or ordinances enacted nursuant to Government Code section 65858 (Urgency Measure/moratoria), shall not abrogate any fundamental vested rights established pursuant to State law, including, without limitation, those established pursuant to the prior valid adoption and execution of a development agreement under Section 17.138.015 and the development agreement procedure in Chapter 17.138. Absent the prior establishment of such vested rights, any Planning Code adoption or amendment shall apply. A.-Building-and-Sign-Permits-and-Development-Agreements. Whenever-any-subsisting

Case File Numbers ZT13-065

building permit-or-sign-permit-has been lawfully issued beforehand, or-whenever-a-subsisting development-ngreement-has been-approved beforehand-under-Section 17.102.310 and the development-agreement-procedure-in-Chapter-17.138, neither-the-original-adoption of the zoning regulations-nor-the-adoption-of-uny-subsequent-rezoning-or-other-amendment-thereto-shall prohibit the construction, other-development-or-change, or-use-authorized-by-said-permit-or agreement. The uses-us-they-result-shall-be-deemed-nonconforming-uses-and-subject-to-the nonconforming-use-regulations in-Chapter-17.11-1, except-to-the-extent-that-they-arc-authorized-by a-subsisting-conditional-use-permit, development-agreement, variance, or-other-special-zoning approval.

B. Alcoholic Beverage Control Licenses. On premises for which a valid state of California Alcoholic Beverage Control license had been issued, and which premises had been used in the exercise of the rights and privileges conferred by the license at a time immediately prior to the effective date of the applicable provisions of Section-17:102:2:10 17:103.030, the premises may hereafter be used in the exercise of the same rights and privileges without requiring a conditional use permit or having to meet the provisions of the aforesaid section. Such use The uses as they result that do not conform to the zoning regulations shall be deemed a nonconforming use and subject to the nonconforming use regulations, except as otherwise provided in Sections 17:114.020 and 17:114.030. For the purposes of this subsection, the word "premises" shall mean and include only the actual space within a building devoted to the sale of alcoholic beverages on said effective date.

Staff proposes to move the section above to Chapter 17.114 - Non-Conforming Uses, and clarify it to be consistent with State law.

In addition, language was added to the following sections to: A) codify and clarify the long-standing practice limiting evidence and testimony during appeals to evidence submitted in the appeal form itself and presented (1) prior to the close of the written public comment period for the underlying decision being appealed, in the case of appeals based on a decision by the Zoning Administrator or other administrative decisions, or (2) prior to the close of the City Planning Commission's public hearing for the underlying decision being appealed, in the case of appealed appealed, in the case of appeales based on decisions made by the City Planning Commission, as applicable; and (B) eliminate the requirement that, if the City Council is unable to decide an appeal, the matter appear for a vote on each regular meeting of the Council thereafter until decided.

17.130.050 - Presentation of written and documentary evidence

17,134.040 - Procedures for consideration

17.134.060 - Appeal to Planning Commission-Minor Conditional Use Permits

17.134.070 - Appeal to Council-Major Conditional Use Permits

17.136.040 - Regular Design Review

17.136.080 - Appeal to Planning Commission-Regular design review

17.136.090 - Appeal to City Council-Regular design review

17.040.030 - Preliminary Planning Commission action

17.040.060 - Final Planning Commission action

17.140.070 - Appeal to Council

17.144.090 - Council action

17.148.040 - Procedure for consideration

17.148.060 - Appeal to Planning Commission-Minor variances

17.148.070 - Appeal to Council-Major variances

17.156.140 - Procedure for consideration of violations to performance standards

17.156.150 - Procedure for consideration of violations to conditions of approval

17.156.160 - Appeal to Planning Commission

17.156.170 - Appeal on the revocation of a Deemed Approved Status to the City Council

Page 9

Case File Numbers ZT13-065

17.157.110 - Procedure for consideration of violations to performance standards

17.157.120 - Procedure for consideration of violations of conditions of approval

17.157.130 - Appeal to City Planning Commission

17.157.140 - Appeal on the revocation of a Deemed Approved Status to the City Council

Further, new language was added to the following sections to: require items sent to the Landmarks Board or Parks and Recreation Commission for review to automatically be sent back to either the Director or the Planning Commission for review after a 30-45 day period. This is to give certainty to the process and ensure projects do not linger with an advisory board too long without any action to schedule or review the item.

17.11.020 - Designation and mapping of parks by category

17.100B.030 - Required design review process

17.135.030 - Procedure for consideration

17.135.040 - Referral to Landmarks Preservation Advisory Board

17.136.055 - Special regulations for historic properties in the central business zones

17.136.060 - Review by Landmarks Board in certain cases

17.144.050 - Review by Landmarks Board in certain cases

17.144.055 - Review by Parks and Recreation Advisory Commission in certain cases 18.02.0140 - Appeal Procedures

#### **III. SUBSTANTIVE CHANGES**

The following section summarizes the proposed substantive changes to the Planning Code. The Code language is in **bold** type and explanations are in normal type.

17.10.485 – Taxi and light fleet-based service commercial activities.

Passenger transportation services, local delivery services, and other businesses that rely on fleets of three (3) or more vehicles with a gross vehicle weight rating rated-capacity-of less than twelve thousand-five-hundred-(12,500) fourteen thousand (14,000) pounds. This classification includes parking, dispatching, and offices for taxicab and limousine operations, airport shuttles, medical transport, local messenger and document delivery services, janitorial services, and similar businesses. This classification\_does-not<u>only</u> includes towing operations when except-for-tow-truck services-where-vehicles are taken to off-site locations and the tow trucks do not exceed the above gross vehicle weight rating. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Staff proposes the above changes to this regulation to use measurements that are consistent with industry standards for Light Vehicles (e.g., pick-up trucks, ambulances and parcel delivery trucks).

17.10.585 Trucking and truck-related industrial activities

Trucking and Truck-Related Activities include the provision of freight handling and shipping services by trucks as well as parking, maintenance, and services for trucks and other heavy vehicles and equipment. Each classification involves the use of trucks and other heavy vehicles that have a gross vehicle weight rating greater than or equal to fourteen thousand (14,000) pounds. This classification also includes certain activities accessory to the above, as specified in Section 17.10,040.

Staff proposes changes to the Truck Activities regulations and use measurements in Section 17.10.585 that are consistent with industry standards.

Case File Numbers ZT13-065

 Table 17.58.03, note 4
 Property development standards

In the CBD-R zone, portions of a building over fifty-five (55) feet in height shall have a setback of at least one (1) foot <u>from the required interior side vard for every live (5)</u> four-(1) feet that portion is above fifty-llve (55) feet. This setback, however, need not exceed forty (40) feet. Also, see Section 17.108.080 for the required interior side and rear yard setbacks on a lot containing two or more living units and opposite a legally-required living room window. See Section 17.108.130 for allowed projections into required yards.

Staff proposes to change the requirement for 1 foot of additional setback to apply to each 5 feet of height over 55 feet, rather than 4. This stepback is more appropriate in a downtown context.

## 17.73 - CIX-1, CIX-2, IG and IO Industrial Zone Regulations

The following text was added to the Design Review Section (17.73.015): <u>Except for projects that are</u> <u>exempt from design review as set forth in Section 17.136.025</u>, no Residential Facility, Designated <u>Historic Property</u>, Potentially Designated Historic Property, Telecommunications Facility, Sign, or <u>other associated structure shall be constructed</u>, established, or altered in exterior appearance, <u>unless plans for the proposal have been approved pursuant to the design review procedure in</u> <u>Chapter 17.136</u>, and when applicable, the Telecommunications regulations in Chapter 17.128, or the Sign regulations in Chapter 17.104.

Staff proposes to include this language to provide for Design Review for Residential facilities and historic properties that are located in the CIX, IG an 10 zones. This is considered a minor change because it does not apply to non-residential buildings that are not historic.

17:102.110(F).- Increased Off-Street-Parking.- Off-street-parking-shall-be-provided-for-the-proposed development-in-en-amount-nt-least-one-hundred #ifty-percent-(150%)-of-that-required-by-the-off-street parking-requirements in Chapter-17-116.

Staff proposes to delete Section\_17.102.110(F)\_because it\_seems too onerous and will generally apply in the corridors where development is encouraged.

# **IV. OTHER RELATED CHANGES**

Various changes were made to the code that relate and correspond to the amendments described above.

Case File Numbers ZT13-065

# ACTION REQUESTED OF THE ZONING UPDATE COMMITTEE

Staff recommends that the Zoning Update Committee provide comments regarding the proposed zoning text amendments, and forward the proposal to the Planning Commission.

Prepared by:

0011ANGC

Ann Clevenger, Planner III

Approved by:

Scott Miller, Zoning Manager

Approved for forwarding to the Zoning Update Committee of the City Planning Commission by:

Rachel Flynn, Director Department of Planning and Building

ATTACHMENTS: A. Proposed Zoning Text Amendments

Proposed Change	Comment
17.07.030 - Purposes of zoning regulations.	Inserting "policies and objectives" brings language
The general purposes of the zoning regulations are to protect and	consistency between the Zoning Code and the
promote the public health, safety, comfort, convenience, prosperity,	General Plan.
and general welfare and to achieve the following objectives:	
A. To promote the achievement of the proposals, <u>oolicies and</u>	
objectives of the Oakland General Plan;	
17.09.040 Definitions - "Dwelling unit"	Combined with "Occupancy of a unit"
"Dwelling unit" means a room or suite of rooms including one and	
only one kitchen, except as otherwise provided in Section	
17.102.270, and designed or occupied as separate living quarters for	
one_of_the_persons-or-groups-specified_in-Section-17-102-260-one	
person or family; or, where the facility occupied is a One-Family	
Dwelling, such family and not more than three (3) boarders, roomers,	
or lodgers where access to all rooms occuoied by such boarders,	
roomers, or lodgers is had through the main entrance of the dwelling	
unit.	
17,09.040 Definitions - "aross vehicle weight rating"	This is the industry standard and creates
Definition for "gross vehicle weight rating" was added: "Gross vehicle	consistency in the truck activity classifications.
weight rating" means the vehicle weight specified by the	
manufacturer as the maximum loaded weight (truck plus cargo) of a	
<u>sirigle vehicle.</u>	
17, 10,280 - Fast-food restaurant commercial activities.	References relevant Chapters of the OMC for the
B. Except as may otherwise be allowed in OMC Chapters 5.49, 5.51,	reader.
and 8.09 and 9.52, tThe sale of ready-to-consume prepared foods	
from trucks, pushcarts or other movable equipment located on public	
or private property on a semi-permanent basis during hours of	
operation. Vehicular food vending generally has the following	
characteristics:	·
General clean up:	Minor edits made for consistency.
<ul> <li>"Section" added prior to section number (i.e. 17.105.010</li> </ul>	
changed to Section 17.105.010)	· ,
<ul> <li>Numbers placed in parentheses and put after spelled out</li> </ul>	
numbers (i.e. "36 feet" changed to "thirty-five (35)" feet.	
17.10.440 Transient Habitation Commercial Activities	Adds examples of the term for clarity, and corrects
<ul> <li>More examples in activity classifications (i.e. hotels and motels</li> </ul>	oversight in B&B classification (residential activity
<ul> <li>added as examples in Transient Habitation Commercial</li> </ul>	type; not commercial).
Facilities).	
Correction to B&B reference (Residential Activity type).	
17.10.480 - Automobile and other light vehicle repair and cleaning	Brings consistency to Truck Activities code
commercial activities.	language.
Automobile and Other Light Vehicle Repair and Cleaning Commercial	
Activities include the major repair or painting of motor vehicles that	
have a gross vehicle weight rating of less than fourteen thousand	
(14,000) pounds, including body work and installation of major	
accessories, as well as the washing and polishing of motor vehicles.	
This classification does not include vehicle dismantling or salvage	· · · · · · · · · · · · · · · · · · ·
and tire re-treading or recapping. This classification also includes	· · ·
certain activities accessory to the above, as specified in Section	
17.10.040.	
17.10.485 - Taxi and light fleet-based service commercial activities.	Staff proposes these changes to this regulation to
Passenger transportation services, local delivery services, and other	use measurements that are consistent with industry
businesses that rely on fleets of three (3) or more vehicles with a	standards for Light Vehicles (e.g., pick-up trucks,
gross vehicle weight rating rated-caeacity-of less than twelve	ambulances and parcel delivery trucks).
thousand-five-hundred-(12,500) fourteen thousand (14,000) pounds.	anionalites and barter densely flucks/
This classification includes parking, dispatching, and offices for	
taxicab and limousine operations, airport shuttles, medical transport,	
local messenger and document delivery services, janitorial services,	
and similar businesses. This classification_does-notonly includes	
towing operations when exsect-for-tow-truck-servises-where-vehicles	
towing operations <u>when</u> exsept-for-tow-truck-servises-where-vehicles are taken to off-site locations and the tow trucks do not exceed the	
are taken to off-site locations and the tow trucks do not exceed the	
are taken to off-site locations and the tow trucks do not exceed the above gross vehicle weight rating. This classification also includes	
are taken to off-site locations and the tow trucks do not exceed the	,

Proposed Change	Comment
47:10:500 Transport and warehousing commercial activities (does nat-apply to the CIX-4, GIX-2, IG, or IQ-zones).	Staff proposes to delete this section because it has been replaced by the new industrial Use Classification adopted in 2009. This classification had been kept in the Code because the classifications in the M-10 through M-40 Zones had not been updated to include the new classifications. This Zoning update includes updates to the M-10 through M-40 Zones such that the Transport and warehousing classification is obsolete.
17.10.585 Trucking and truck-related industrial activities Trucking and Truck-Related Activities include the provision of freight handling and shipping services by trucks as well as parking, maintenance, and services for trucks and other heavy vehicles and eauipment. Each classification involves the use of trucks and other heavy vehicles that have a aross vehicle weight rating greater than or eaual to fourteen thousand (14,000) pounds. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.	Brings consistency to Truck Activities and uses measurements that are industry standards.
17.10.760 - Shopping Center Facility. A Shopping Center <u>Nonresidential Facility is a complex of one or</u> more-retail buitdings-and-related-fasilities-forming-a-centrat-retail market within a given area and having a common parking area.	This classification is never used.
17.13.060 - Special regulations for mini-lot-and-planned unit developments. AMini-lot Developments. In mini-let developments, certain regutations-that-otherwise-apply-to-iodividuat-lots-in-the-RH-zones may be waived or modified when and as prescribed in Section 17.102.320	Mini-lots removed from RH zones. This was an oversight in the zoning update and the change is consistent with the old R-1, R-10, and R-20 zones.
When massage, adult entertainment; check cashing; special health care, fast food are permitted or conditionally permitted in a zone, new notes are added in the Activity Tables directing the reader to further regulations in Chapters 17.102 or 17.103.	Reference for the reader
Table 17.35.04– density for rooming units adjusted in the 120 and 160 feet height areas from one unit per 225 sf of lot area to 110 sf of lot area	This will fix an error – density for rooming units should increase as the height area increases
17,65.030 Note L3 Added missing note for General Warehousing Activities in the HBX zones.	Fixes an error during the HBX text amendment.
Table 17.58.03: Transparency regulations in CBD Zone	Required transparency percentages were changed to be consistent with standards in other zones: CBD-P same as CN zones, CBD-C same as CC zones.
Table 17.58.04, note 2: For mixed use projects in the Central Business District (CBD) zones, the allowable intensity of development shall be measured according to both the maximum nonresidential Floor Area Ratio (FAR) allowed by the zone and the maximum residential density allowed by the zone. The total lot area shall be used as a basis for computina both the maximum nonresidential FAR and the maximum residential density. No portion of lot area used to meet the density requirements for a Residential Faoility shall be used as a basis for computing, through the maximum ftoor area-ratio, the maximom amount-ot-floor-area for any-nonresidential facility-on-the-same-tot, and visa-versa.	This clarification to the CBD density makes the regulation consistent with 17.106.030(A),
Table 17.58.04, note 4. In the CBD-R zone, portions of a building over fifty-five (55) feet in height shall have a setback of at least one (1) foot from the required interior side yard for every five (5) four-(4) feet that portion is above fifty-five (55) feet. This setback, however, need not exceed forty (40) feet. Also, see Section 17.108.080 for the required interior side and rear yard setbacks on a lot containing two or more living units and opposite a legally-required living room window. See Section 17.108.130 for allowed projections into required yards.	This stepback is more appropriate in a downtown context.
Chapter-17-68-M-20-Industrial-Zones-Regulations Chapter-17-70-M-30-Industrial-Zones-Regulations Chapter-17-72-M-40-Industrial-Zones-Regulations	These "M" Industrial zones are combined in the same chapter and with a format consistent with the updated zones.

Proposed Change	Comment
Chapter 17.72 M-20, M-30, AND M-40 INDUSTRIAL ZONES	Also, sidewalk cafés are added to the list of
REGULATIONS	pemitted facility types to correct an oversight that
	had inadvertently listed them as permitted activity
_	types.
Chapter 17.73 - CIX-1, CIX-2, IG, and 10 Zones, the following text	Adds Design Review for residential facilities and
was added to the Design Review Section: Except for projects that are	historic properties located in the CIX, IG and IO
exempt from design review as set forth in Section 17.136.025, no	zones. Does not apply to non-residential buildings
Residential Facility, Designated Historic Property, Potentially	that are not historic.
Designated Historic Property, Telecommunications Facility, Sign, or	
other associated structure shall be constructed, established, or	· · · · · · · · · · · · · · · · · · ·
altered in exterior appearance, unless plans for the proposal have	
been approved pursuant to the design review procedure in Chapter	
17.136, and when applicable, the Telecommunications regulations in	
Chapter 17.128, or the Sign regulations in Chapter 17.104. -7.4.02.020Supplemental-zoning-provisions-	Deleted because it serves no purpose
The definitions, special use classification rules, and other provisions	
set-forth in-Chapters-1-7-07-47-09 and 17-10-10-provisions of	
Section-17-108-130: the nonconforming-use-regulations-in-Chapter	
·I <del>7-144; the rezoning, varianse, and other provisions eet ferth in</del>	
Ghapters 17.130 through 17.152; and the provisions of the zoning	,
maps in <u>Chapter 17.154</u> shall apply throughout the city. The	
provisions of the performance standards in Chapter 17,120 and the	↓ ↓
planned unit development regulations in Chapter 17.142 shall apply	
in-the-zones-and-situations-specified-in-said-chaptersThe-provisions	
of development control maps are in addition to, or supersede when	
so-specified-the-regulations-applying-in-the-zones-covering-the-same	
areas.	
17-102-040-17.114 - Effect of prior pennils.	Moved 17,102.040 to the 17.114 Nonconforming
A. Fundamental Vested Rights Not Abrogated By Code	Uses Chapter and included minor clarifying edits to
Adoption or Amendment. The adoption of, or amendment to, the	be consistent with state law.
Planning Code (Chapter 17), including without limitation, ordinances	· · ·
enacted pursuant to Oakland City Charter Section 213 (Emergency	
Ordinances) and/or ordinances enacted pursuant Government Code	
Section 65658 (Urgency Measure/moratoria), shall not abrogate any	
fundamental vested rights established pursuant to State law,	
including, without limitation, those established pursuant to the prior	
valid adoption and execution of a development agreement under	
Section 17, 138,015 and the development agreement procedure in	•
Chapter 17.138. Absent the prior establishment of such vested rights, any Planning Code adoption or amendment shall apply.	
A. Building and Sign Permits and Development Agreements.	
Whenever any subsisting-building-pormit-or-sign-permit-has-been	
lawfully issued beforehand, or whenever a subsisting development	•
agreenient-has-been-approved-beforetrand-under-Section-17-102-310	
ang-the-development-agreement-procedure-in-Chapter-17-138,	
neither-the-original-adeption-of-the-zoning-regulations-nor-the	· · ·
adoption-of-any-subseguent-rezoning-or-other-amendmeot-thereto	
shall prohibit the construction, other development or change, or use	· · ·
authorized-by-said-permit-er-agreement-The-uses-as-they-result-shall	
be-deemed-nonconforming-uses-and-subject-to-the-nenconforming	-
use regulations in Chapter 17.114, except to the extent that they are	.
authorized by a subsisting-conditional-use-permit, development	
agreement, variance, or other-special-zoning-approvat-	
B. Alcoholic Beverage Control Licenses. On premises for which a	
valid state of California Alcoholic Beverage Control license had been	
issued, and which premises had been used in the exercise of the	
rights and privileges conferred by the license at a time immediately	
prior to the effective date of the applicable provisions of Section	
17-103-030 17 103.030, the premises may hereafter be used in the	
exercise of the same rights and privileges without requiring a	
conditional use permit or having to meet the provisions of the	· ·
aforesaid section. The uses as they result that do not conform to the	)
Zoning regulations Such-use-shall be deemed a nonconforming use	
and subject to the nonconforming use regulations, except as otherwise provided in Sections 17.114.020 and 17.114.030. For the	· · ·
purposes of this subsection, the word "premises" shall mean and	l
include only the actual space within a building devoted to the sale of	
alcoholic beverages on said effective date.	

1

Proposed Change	Comment
17-102-070-17 154.060 - Application of zoning regulations to lots divided by zone boundaries.	Moved to 17.154.060 Zoning maps
17-102-080-17.07.065 - Permitted and conditionally permitted uses. A. Other Uses Prohibited. Except as otherwise provided in Sections 17-102-010-17.114.040 and 17-102-070 17.154.060, the	Moved to 17.07.065 Title purpose and scope of zoning regulations
nonconforning use regulations in Chapter 17.114, and the planned unit development regulations in Chapter 17.142, or as authorized under Section $-17$ , $102$ , $310$ , $17.138.015$ , the development agreement	
procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no land shall be improved or used for any activity or facility which is not listed as permitted or conditionally permitted in the	
<ul> <li>applicable individual zone regulations or development control maps.</li> <li>B. Relationship Between Activities and Facilities. A use must qualify under the zoning regulations both as an activity and as a</li> </ul>	
facility. A permitted or conditionally permitted activity may be accommodated or served only by a permitted facility or, upon the granting of a conditional use permit, by a conditionally permitted facility; and a permitted or conditionally permitted facility may	· · ·
accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditional use permit, a conditionally permitted activity.	
17.102.100 - Conditions for accessory parking serving activities which are not themselves allowed.	Moved to 17.116 Off-Street Parking and Loading
17.192.110F. — Increased Off-Street Parking. Off-street parking shall-be-provided-for-the-proposed-development-in-an-amount-at-least one hundred-fifty-percent-(150%) of that-required by the off-street parking-requirements-in-Chapter-17-116	Deleted because seems onerous – will generally apply in the corridors where we encourage development.
17.102.138 Time limit on operation of oubdivision sales offices Residentiol zones.	Deleted because there is already a regulation in 17.10.040 Accessory Activities
17.102.210 - Special regulations applying to Convenience Markets, Fast-Food Restaurants, certain establishments selling alcoholic beverages, providing mechanical or, and electronic games.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17.102.212 - Special Regulations Applying to Residential Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
New industrial classifications added to HBX, M-10 to M-40 zones, C- 45 and C-40 zones.	Fixes long time issue. The zones were using the old use classifications that no longer exist.
17.102.220 - Special regulations applying to mining and quarrying extractive activities. 17-102-250-Maximum-density and fleer-area-ratio-during	Moved to new chapter 17.155 because it is so lengthy. Deleted because it has never been an issue and is
construction- 1-7-102-260-Oscupancy-of-a-dwelling-unit	difficult to enforce. Deleted and combined with Section 17.09.040,
47-402-265—Oceupancy-of a-One-Family-Residentiat Facility-by-a	Definition of a "Dwelling unit" Deleted and combined with Section 17.09.040,
Residential Care Residential Activity 17.102.280 - Rules for determining the number of habitable rooms in	Definition of a "Dwelling unit" Moved to 17.09 Definition of "Habitable Room" and
Residential Facilities. 17.102.290 - Special regulations for Drive-Through Nonresidential Facilities.	Habitable rooms, number of". Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17.102.310 - Special regulations for certain projects with development agreements.	Moved to 17.138 Development Agreements because it applies to Development Agreements
17.102.320 - Conditional use pennit for waiver of certain requirements in mini-lot developments.	Moved to 17.142 Planned Unit Development Regulations so that all integrated developments are in the same chapter.
17.102.330 – Conditional use permit for waiver of certain requirements with parcel division between existing buildings	Deleted, because it already exists in Chapter 17.106, General Lot, Density and Area Regulations, Section 17.106.010(B).
17.102.335 - Standards for Sidewalk Cafes.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17.102.360 - Secondary Units.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17.102.370 - Conditional use permit for hotels and motels.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.

Proposed Change	Comment
17.102.380 – Special regulations applying to truck-related activities in	Deleted, and added as a footnote to any zone that
the West Oakland Community Development District	allows trucking activities.
17.102.390 - Regulations Applying to Special Health Care Civic Activities.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17,102.400 - Special design requirements for lots that contain Residential Facilities and no Nonresidential Facilities	Paving restrictions moved to footnote in zoning chapters that allow residential facilities; Screening of utility meters and trash containers moved to 17.124 Landscaping and screening standards. Exterior bars regulations stay in 17.102.400
17,102.420 – Special design requirements for lots located in residential and commercial zones and the OS, S-1, S-2, S-3 and S- 15 Zones,	Deleted, because it is redundant with 17,108.140.
17,102,430 - Regulations applying to check cashier and/or check cashing activity.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17.102,440 - Special regulations for primary collection centers in all zones.	Moved to Chapter 17.103. This chapter contains regulations that apply to a particular use classification that is listed in 17.10.
17.106.030 – Maximum density and fioor-area ratio on lots containing both Residential and Nonresidential Facilities	Revise language in subsection (A) to clarify permitted density for mixed use projects in the CBD and Jack London District (see also Table 17.58.04, note 2).
17.108 – General Height, Yard, Court, and Fence Regulations	Staff removed references in other parts of the Code that were related to previously-deleted sections of Chapter 17.108. (In the spring of 2012, the City Council voted to remove the following Code sections: 17.108.040; 17.108.050; 17.108.060; 17.108.070; 17.108.090; 17.108.100, and 17.108.110).
17.114.040 – Effect of prior permits	Moved from 17.102.040 - Effect of prior pennits.
17.130.060 - Obligation of applicant to defend, indemnify, and hold harmless the City of Oakland A. To the maximum extent permitted by law, the applicant shall defend (with counsel-reasonably acceptable to the City), indemnify, and hold harmless the City of Oakland, the Oakland City Council, the City of Oakland Redevelopment Agency, <u>City of Oakland</u> <u>Redevelopment Successor Agency</u> the Oakland, City Planning Commission and its respective agents, officers, <u>volunteers</u> and employees (hereafter collectively called City) from any liability, damages, claim, judgment, loss (direct or indirect), action, causes of action or proceeding (including legal costs, attorneys' fees, expert witness or consultant fees, City Attomey or staff time, expenses or costs) (collectively called "Action") against the City to attack, set aside, void or annul, (1) an approval by the City relating to a development-related application, <u>or</u> subdivision <u>and/or a (Lease)</u> <u>Disposition and Development Agreement;</u> or (2) implementation of <u>such</u> an-approved-development-retoted-project. The City may elect, in its sole discretion, to participate in the defense of said Action and the applicant shall reimburse the City for its reasonable legal costs and attorneys' fees.	Clarifying language added to this section.
17.130.090 Minor land use permits considered concurrently with Major permits <u>A.</u> Any Minor land use <u>related</u> permit <u>and/or approval</u> that is related to a development application that also includes any Major land use <u>related</u> permit <u>and/or approval</u> shall be considered noncurrently with the Major land use <u>related</u> permit <u>and/or approval</u> and <u>shall</u> follow the eame <u>all</u> procedurales <u>reouirements</u> for censideration <u>associated with</u> <u>City Planning Commission decisions</u> . In this case, the entire <u>application shall initially be considered by the City Planning</u> <u>Commission and may be appealed to the City Planning</u> <u>Commission and may be appealed to the City Council, in accordance</u> <u>with the reouirements for</u> Any-elevation-of <u>a Minor</u> land-use permit-to the Major <u>land use related</u> permit <u>and/or approval or discretionary</u> <u>actions</u> . consideration-ercess-shall-be-for-consideration-ourposes only-and-shall-not-modify-any-required-findiags-for-said-Miner-land use-permit-	Clarifies issue of "packaging" Major land use related permits with Minor land use related permits.

Proposed Change	Comment
B. Any Minor land use cermit referred to the City Planning	
Commission for initial decision in order to be considered concurrently with any Major land use related permit and/or	
approval shall still be considered a Minor land use related pennit	
and/or approval and the required findings for said Minor land	· ·
use related cermit and/or approval shall apply.	
de leited of the and a approver of an appro-	
17.134.020 – Definition of major and minor conditional use permits.	Deleted Subsection 17.134.020(3)(h) to clarify that a
	minor conditional use permit referred by the
	Planning Director to the Planning Commission for
·	initial decision is still considered a minor permit.
17.134.020(2)(b)(iii), Special Health Care Civic Activifies	Moved this section into a new Section
	17.134.020(2)(a)(xiii), under Definition of major
	conditional use permits. It was inadvertently
	misplaced under the facility types section rather than
17.124.040(B) Breadway for exclusion Minor Conditional	Add language to clarify that a minor conditional use
17.134.040(B) – Procedures for consideration – Minor Conditional Use Permits	pennit referred by the Planning Director to the
, Use Permits	Planning Commission for initial decision is still
	considered a minor permit, but shall simply be
	processed according to the Major CUP.procedure.
17.136.040(D) - Regular Design Review - Procedure for	Add language to clarify that a regular design review
Consideration	permit referred by the Planning Director to the
	Planning Commission for initial decision is still
	considered a minor permil, but shall simply be
	processed according to the Major Pennit procedure.
17,138.015(A) and (B) - Projects eligible and special regulations for	Relocated from Section 17.102.310 to a new Section
projects with development agreements	17.138.015(A) and (B), and modify the language to add "successor to the" Redevelopment Agency and
· ·	delete "major" in reference to affected permit types.
Chapter 17.142 - Mini Lot and Planned Unit Development	Added separate mini-lot section (Article II) in
Regulations	Chapter 17.142 to clarify applicability and basic
Tregulations .	provisions.
17.142.030 - Development for which planned unit development	Add new subsection B to clarify that the Planning
approval is required or requested.	Director may require the PUD procedure for large,
(B) Unless required by the Plannino Director, other large, integrated	integrated developments not listed under
developments involving the same minimum land area thresholds of a	17.142.030(A).
olanned unit development, as defined in Section 17.142.020, are	
permitted without such a cermit. However, an applicant of such a	
development may request a planned unit development pennit	
pursuant to the planned unit development procedure in Chapter 17.140, but shall be subject to all regulations generally appiying in the	
zone in which they are located.	
Zone in whom they are located.	
17 149 020 Definition of mains and minor variances	
1 17.146.020 - Definition of major and minor variances.	Staff deleted the following:
17.148.020 – Definition of major and minor variances.	Staff deleted the following: 7
17.146.020 – Delinition of major and minor variances.	7. <u>Any variance application that requires</u> development—of—an—Envirenmental
17.146.020 – Definition of major and minor variances.	7. Any variance - application that requires
17.146.020 – Delinition of major and minor variances.	7. <u>Any variance application that requires</u> development—of—an—Envirenmental Impact-Report;
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. <u>Any-variance-application that-requires</u> development—of—an<u>Envirenmental</u> Impact-Report;</li> <li>8. <u>Any-variance application referred-by the</u></li> </ul>
17.146.020 – Delinition of major and minor variances.	7. <u>Any-variance-application that requires</u> development—of—an—Envirenmental Impact-Report;     8. <u>Any-variance application referred by the</u> Director-of-City-Planning-to-the-City
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. <u>Any-variance-application that-requires</u> development—of—an<u>Envirenmental</u> Impact-Report;</li> <li>8. <u>Any-variance application referred-by the</u></li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. Any variance -application that requires development—of—an—Envirenmental Impast-Report;</li> <li>8. Any-variance application referred by the Director-of—City_Planning_to_the City Planniog-Commission-for-initiat-decision pursuont-lo-Section-17,448.040(B)(1).</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>8. Any-variance application referred by the Director-of—City_Planning—to—the -City Planeiog-Commission-for-initiat-decision pursuont-Io-Section-17:448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>8. Any variance application referred by the Director—of—City_Planning—to—the—City Planniog-Commission-for-initiat-decision pursuont-Io-Section-17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>8. Any-variance application referred by the Director—of—City_Planning—to—the -City Planaiog-Commission-for-initiat-decision pursuont-Io-Section-17:448:040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>7. Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>8. Any variance application referred by the Director—of—City_Planning—to—the—City Planniog-Commission-for-initiat-decision pursuont-Io-Section-17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any variance application referred by the Director-of—City—Planning—to—the—City Planning—to—the—City Planning—commission-for-initiat-deeision pursuont-Io-Section-17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any variance application referred by the Director—of—City—Planning—to—the—City Planning—to—the—City Planning—Commission-for-initiat-deeision pursuont-Io-Section-17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> </ul>
17.146.020 – Delinition of major and minor variances.	<ul> <li>Any-variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any-variance application referred by the Director-of—City—Planning—to—the—City Planniog-Commission-for-initial-decision pursuont-Io-Section-17.448.040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor pennit.</li> </ul>
17.148.020 – Definition of major and minor variances.	<ul> <li>Any-variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any-variance application referred by the Director-of—City—Planning—to—the—City Planniog-Commission-for-initial-decision pursuont-Io-Section-17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor pennit.</li> </ul>
	<ul> <li>Any-variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any-variance application referred by the Director-of—City—Planning—to—the—City Planniog-Commission-for-initiat-decision pursuont-Io-Section-17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor pennit.</li> <li>Added language to subsection B, Minor Variances, to clarify that a minor variance referred by the</li> </ul>
	<ul> <li>Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any variance application referred by the Director—of—City—Planning—to—the—City Planning—to—the—City Planning—to—the—City Planning—commission-for-initiat-deeision pursuont-Io-Section—17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor pennit.</li> <li>Added language to subsection B, Minor Variances, to clarify that that a minor variance referred by the Planning Commission for</li> </ul>
	<ul> <li>Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any variance application referred by the Director—of—City—Planning—to—the—City Planning—to—the—City Planning—to—the—City Planning—Commission-for-initiat-deeision pursuont-Io-Section—17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor variance referred by the Planning Director to the Plan</li></ul>
	<ul> <li>Any variance -application that requires development—of—an—Envirenmental Impact-Report;</li> <li>Any variance application referred by the Director—of—City—Planning—to—the—City Planning—to—the—City Planning—to—the—City Planning—commission-for-initiat-deeision pursuont-Io-Section—17.448-040(B)(1).</li> <li>Subsection 7 was deleted to clarify that a minor variance that is simply accompanying a project that requires an EIR does not make the variance a Major permit.</li> <li>Subsection 8 was deleted to clarify that a minor variance referred by the Planning Director to the Planning Commission for initial decision is still considered a minor pennit.</li> <li>Added language to subsection B, Minor Variances, to clarify that that a minor variance referred by the Planning Commission for</li> </ul>

ATTACHMENT B

# Title 17 PLANNING

Chapters:

17.01 General Provisions of Planning Code and General Plan Conformity

17.03 City Planning Commission

17.05 Landmarks Preservation Advisory Board

17.07 Title, Purpose and Scope of the Zoning Regulations

17.09 Definitions

17.10 Use Classifications

17.11 OS Open Space Zoning Regulations

17.13 RH Hillside Residential Zones Regulations

17.15 RD Detached Unit Residential Zones Regulations

17.17 RM Mixed Housing Type Residential Zones Regulations

17.19 RU Urban Residential Zones Regulations

17.30 R-80 High-Rise Apartment Residential Zone Regulations

17.33 CN Neighborhood Center Commercial Zones Regulations

17.35 CC Community Commercial Zones Regulations

17.37 CR Regional Commercial Zone Regulations

17.54 C-40 Community Thoroughfare Commercial Zone Regulations

17.56 C-45 Community Shopping Commercial Zone Regulations

17.58 CBD Central Business District Zones Regulations

17.65 HBX Housing and Business Mix Commercial Zones Regulations

47.68-M-20-Light-Industrial-Zone-Regulations

17-70-M-30-Genoral Industrial-Zone-Regulations

17-72-M-40-Heavy-Industrial-Zone-Regulations

17.72 M-20, M-30, and M-40 Industrial Zones Regulations

17.73 CIX-1, CIX-2, IG and IO Industrial Zones Regulations

17.74 S-1 Medical Center Zone Regulations

17.76 S-2 Civic Center Zone Regulations

17.78 S-3 Research Center Zone Regulations

17.80 S-4 Design Review Combining Zone Regulations

17.82 S-6 Mobile Home Combining Zone Regulations

17.84 S-7 Preservation Combining Zone Regulations

17.90 S-10 Scenic Route Combining Zone Regulations

17.92 S-11 Site Development and Design Review Combining Zone Regulations

17.94 S-12 Residential Parking Combining Zone Regulations

17.97 S-15 Transit Oriented Development Zone Regulations

17.99 S-17 Downtown Residential Open Space Combining Zone Regulations

17.100A S-19 Health and Safety Protection Combining Zone Regulations

17.100B S-20 Historic Preservation District Combining Zone Regulations

17.101A D-WS Wood Street District Commercial Zone Regulations

17.101B D-OTN Oak To Ninth District Zone Regulations

17.101C D-BR Broadway Retail Frontage District Interim Combining Zone Regulations

17.101D D-KP Kaiser Permanente Oakland Medical Center District Zones Regulations

17.102 <u>Regulations Applicable to Certain Activities and Facilities</u> Gonoral-Rogulationo Applicable-To-All-Or-Several-Zones

17.103 Special Regulations and Findings for Certain Use Classifications

17.104 General Limitations on Signs

17.106 General Lot, Density, and Area Regulations

17.107 Density Bonus and Incentive Procedure

17.108 General Height, Yard, Court, and Fence Regulations

17.110 Buffering Regulations

17.112 Home Occupation Regulations

- 17.114 Nonconforming Uses
- 17.116 Off-Street Parking and Loading Requirements
- 17.117 Bicycle Parking Requirements
- 17.118 Recycling Space Allocation Requirements
- 17.120 Performance Standards
- 17.124 Landscaping and Screening Standards
- 17.126 Usable Open Space Standards
- 17.128 Telecommunications Regulations
- 17.130 Administrative Procedures Generally
- 17.132 Administrative Appeal Procedure
- 17.134 Conditional Use Permit Procedure
- 17.135 Special Use Permit Review Procedure for the OS Zone

17.136 Design Review Procedure

- 17.138 Development Agreement Procedure
- 17.140 Planned Unit Development Procedure
- 17.142 Mini-Lot and Planned Unit Development Regulations
- 17.144 Rezoning and Law Change Procedure
- 17.148 Variance Procedure
- 17.150 Fee Schedule
- 17.152 Enforcement
- 17.154 Zoning Maps
- 17.155 Special Regulations Applying to Mining and Quarrying Extractive Activities
- 17.156 Deemed Approved Alcoholic Beverage Sale Regulations
- 17.157 Deemed Approved Hotel and Rooming House Regulations
- 17.158 Environmental Review Regulations

17.01.050 - General Plan prevails over Planning Code and Subdivision Regulations.

Until the Planning Code is <u>fully</u>\_updated, land use designations, zoning controls and subdivision controls specified by the Planning Code and Subdivision Regulations shall apply, except where such action would expressly conflict with the Oakland General Plan. Where an express conflict does arise, the General Plan policies and land use designations shall apply. An "express conflict" shall be deemed to be any situation where a proposal clearly conforms with the General Plan but is not permitted by the Zoning and/or Subdivision Regulations, or where a proposal clearly does not conform with the General Plan but is permitted or conditionally permitted by the Zoning and/or Subdivision Regulations. The provisions of Sections 17.01.060 through 17.01.120, as applicable, shall then be followed.

17.01.080 - Appeal of Director's determination.

Within ten (10) calendar days of a written determination by the Director of City Planning B pursuant to Subsection 17.01.120.C. an appeal of such determination may be taken to the City Council by the applicant or any other interested party. Such appeal shall be accompanied by a fee as prescribed in the City master fee schedule. In event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Director and shall be filed with the City Clerk. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Planning Director or wherein his or her decision is not supported by the evidence in the record. Upon receipt of such appeal, the Council shall set the date for consideration thereof. After the hearing date is set, the Planning Director shall refer the matter to the Planning Commission for its review and advice. The Planning Commission shall consider the matter at its next available meeting. Such referral shall be only for the purpose of issue clarification and advice to the City Council. The City Clerk shall not less than seventeen (17) days prior to the Council hearing, give written notice of the date and place of the hearing on the appeal to the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate., writton-notice-of-the-date-and-place-of-tho-hearing-on-the-appeal-In considering the appeal, the Council shall determine whether the proposal conforms to the provisions of Section 17.01.120.C., and may approve or disapprove the proposed determination. The decision of the City Council shall be made by resolution and shall be final.

**17.03.010** - City Planning Commission, Landmarks Preservation Advisory Board and Board of Adjustments.

I. Removal. To assure participation of Commission and Board members, attendance by the members of the Commission and Board at ail regularly scheduled and special meetings of the Commission and Board shall be recorded, and such record shall be provided semiannually to the Office of the Mayor for review. A member may be removed pursuant to Section 601 of the City Charter. Cause for removal shall include, among other things, conviction of a felony, misconduct, incompetency, inattention to or inability to perform duties, absence from three consecutive regular meetings, or, for members of the Commission's Residential Appeals Committee, absence from three (3) consecutive regular meetings of the Commission of the Commission of the Commission of Board.

17.03.040 - Residential Appeals Committee of the City Planning Commission

There is created a Residential Appeals Committee of the City Planning Commission consisting of three (3) members of the Commission. The Committee shall decide all appeals of decisions by the Director of City Planning as set forth in the Zoning Regulations. The method for appointing Committee members and the length of Committee members' terms shall be as set forth in the Commission's Rules of Procedure.

# 17.05.010 - Creation and membership.

There is created a Landmarks Preservation Advisory Board. It shall consist of seven (7) members appointed by the Mayor subject to the affirmative vote of five or more members of the City Council. In making appointments, the Mayor may consult persons and organizations interested in landmarks or historic preservation. The members shall include at least one architect; one landscape architect or city planner; one person having extensive knowledge of Oakland history, or of relevant architectural history; and one real estate broker or other person with significant experience in the financing or management of real estate.

## 17.05.040 - Removal.

Any member of the Board may be removed for cause, after hearing, by the affirmative vote of six (6) or more members of the City Council.

# Chapter 17.07 - TITLE, PURPOSE AND SCOPE OF THE ZONING REGULATIONS Sections:

#### 17.07.010 - Title, purpose, and applicability.

17.07.020 - Title of zoning regulations.

17.07.030 - Purposes of zoning regulations.

#### 17.07.040 - Applicability of zoning regulations.

17.07.050 - Effect of development control maps.

#### 17.07.060 - Conformity with zoning regulations required.

17.07.065 - Permitted and conditionally permitted uses.

17.07.070 - Minimum requirements.

17.07.030 - Purposes of zoning regulations.

The general purposes of the zoning regulations are to protect and promote the public health, safety, comfort, convenience, prosperity, and general welfare and to achieve the following objectives:

A. To promote the achievement of the proposals, <u>policies and objectives</u> of the Oakland General Plan;

17.07.040 - Applicability of zoning regulations.

D. Private Agreements. The zoning regulations are not intended to abrogate, annul, or impair any easement, covenant, or other agreement between parties. However, where the zoning regulations impose a greater restriction or higher standard than that required by such agreement, the zoning regulations shall control, except as otherwise authorized under-Soction 47.102.310 and the development agreement procedure in Chapter 17.138.

#### 17.07.050 - Effect of development control maps.

Development control maps and all notations, references, and regulations shown therewith shall be considered part of the zoning regulations. Development control maps may include, but are not limited to, regulations intended to carry out any plan respecting location or type of activities; height, bulk, siting, or design of structures; location or design of open areas and landscaping; and other comparable regulations. In case of conflict with any other provision of the zoning regulations, the development control map shall take precedence, except as otherwise authorized under—Section—17.102.310—and the development agreement procedure in Chapter 17.138.

17.07.060 - Conformity with zoning regulations required.

Except as otherwise allowed by Section <u>17.114.030</u> <u>17.102.040</u> <u>114.030</u> and by the nonconforming use regulations in Chapter 17.114, or as authorized under Section <u>17.138.015</u>, <u>17.102.310</u>, the development agreement procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no activities or facilities shall be established, substituted, expanded, constructed, altered, moved, painted, maintained, or otherwise changed, and no lot lines shall be created or changed, except in conformity to the zoning regulations.

## 17.07.065 - Permitted and conditionally permitted uses.

- A. Other Uses Prohibited. Except as otherwise provided in Sections 17.114.030 and 17.154.060, the nonconforming use regulations in Chapter 17.114, and the planned unit development regulations in Chapter 17.142, or as authorized under Section 17.138.015, the development agreement procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no land shall be improved or used for any activity or facility which is not listed as permitted or conditionally permitted in the applicable individual zone regulations or development control maps.
- B. Relationship Between Activities and Facilities. A use must qualify under the zoning regulations both as an activity and as a facility. A permitted or conditionally permitted activity may be accommodated or served only by a permitted facility or, upon the granting of a conditional use permit, by a conditionally permitted facility, and a permitted or conditionally permitted facility may accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditional use permit, a conditionally permitted activity.

17.09.040 - Definitions.

"Dwelling unit" means a room or suite of rooms including one and only one kitchen, except as otherwise provided in Section 17.102.270, and designed or occupied as separate living quarters for one of-the-persons-or-groups opecified in Section 17.102.260, one person or family; or, where the facility occupied is a One-Family Dwelling, such family and not more than three (3) boarders, roomers, or lodgers where access to all rooms occupied by such boarders, roomers, or lodgers is had through the main entrance of the dwelling unit.

"Floor-Area Ratio (FAR)" means the number resulting from the division of the floor area on a lot by the lot area.

"Gross vehicle weight\_rating" means the vehicle weight specified by the manufacturer as the maximum loaded weight (truck plus cargo) of a single vehicle.

"Habitable room" means a space in a living unit intended for living, sleeping, eating, or cooking, including, but not limited to, living rooms, dining rooms, bedrooms, kitchens, dens, family rooms, finished recreation rooms, and enclosed porches suitable for year-round use.

To be considered an individual habitable room, a space must be either:

- 1) Entirely enclosed by partitions and connected to other rooms or spaces by doonways or open archways;
- 2) Separated from another space that both has a floor level difference of at least one (1) foot and is intended to be used for a different function; or
- 3) A kitchen area. Where there are no partitions, open archways, or a split floor level, as described in 1) and 2), the part of the kitchen space considered a habitable room includes all kitchen counters, cabinets, major appliances, and other fixtures plus the floor area within three (3) feet directly in front these items.

Specifically excluded <u>from the definition of habitable room</u> are bathrooms, water closets, hallways, foyers, storage closets, pantries, laundries, utility rooms, unfinished attics and basements, balconies, open porches, garages, and other unfinished spaces used for storage. See-Section-17-102-280-for rules-for-dotormining-the-number-of-habitable roomc-in-a-Rosidential-Facility-

"Habitable rooms, number of" means the total number of habitable rooms in a Residential Facility, except:

1. A habitable room of less than fifty (50) square feet counts as half a room.

2. A habitable room larger than four hundred (400) square feet counts as one room for each four hundred (400) square feet or fraction thereof.

<u>"Gross-vehicte-weight"-moans-the-vehicle-weight-spocified-by-the-manufacturer-as-the-maximum</u> loaded-weight-(truck-plus-cargo)-of-a-singlo-vehicle-

"Planned Unit Development <u>(PUD)</u>" means a large, integrated development adhering to a comprehensive plan and located on a single tract of land, or on two or more tracts of land which may be separated only by a street or other right-of-way.

"Potential Designated Historic Property (PDHP)" means any building or property that is determined by the City's Cultural Heritage Survey to have an existing or contingency rating of "A", "B", or "C", or to contribute or potentially contribute to an Area of Primary Importance (API) or an Area of Secondary Importance (ASI).

"Secondary unit" means a subordinate dwelling unit that is located on the same lot as a larger primary dwelling unit, is either attached or detached, and meets the standards and criteria of Section 47.102.360 17.103.080.

# Chapter 17.10 - USE CLASSIFICATIONS Sections:

#### Article I - General Classification Rules

17.10.010 - Title, purpose, and applicability.

- 17.10.020 Definitions.
- 17.10.030 Listing of activity classifications.
- 17.10.040 Accessory activities.
- 17.10.050 Classification of combinations of principal activities.
- 17.10.060 Listing of facility classifications.
- 17 10.070 Accessory facilities.
- 17.10.080 Classification of combinations of principal facilities.
- 17.10.090 Classification of unlisted uses.

#### Article II - Activity Types

Part 3 - Commercial Activity Types

Sections:

- 17,10.260 General description of commercial activities.
- 17.10.270 General food sales commercial activities.
- 17.10.272 Full service restaurant commercial activities.
- 17.10.274 Limited service restaurant and cafe.
- 17,10.280 Fast-food restaurant commercial activities.
- 17.10.290 Convenience market commercial activities.
- 17.10.300 Alcoholic beverage sales commercial activities.
- 17.10.320 Mechanical or electronic games commercial activities.
- 17.10.330 Medical service commercial activities.
- 17,10,340 General retail sales commercial activities.
- 17.10.345 Large-scale combined retail and grocery sales commercial activities.
- 17.10.350 Consumer service commercial activities.
- 17.10.360 Consultative and financial service commercial activities.
- 17.10.365 Check cashier and check cashing activity.
- 17.10.370 Consumer cleaning and repair service commercial activities.
- 17.10.375 Consumer dry cleaning plant commercial activities.
- 17.10.380 Group assembly commercial activities.
- 17.10.385 Personal instruction and improvement and small scale entertainment commercial activities.
- 17.10.390 Administrative commercial activities.

17.10.400 - Business, communication, and media service commercial activities.

- 17.10.410 Broadcasting and recording service commercial activities.
- 17.10.420 Research service commercial activities.
- 17.10.430 General wholesale sales commercial activities.
- 17.10.440 Transient habitation commercial activities.
- 17.10.450 Building material sales commercial activities.
- 17.10.460 Automobile and other light vehicle sales and rental commercial activities.
- 17.10.470 Automobile and other light vehicle gas station and servicing commercial activities.
- 17.10.480 Automobile and other light vehicle repair and cleaning commercial activities.
- 17.10.485 Taxi and light fleet-based service commercial activities.
- 17.10.490 Automotive fee parking commercial activities.
- 17-10-500 Transport and warehousing commercial activities (does not apply to the CIX-1, CIX-2, IG, or IO zones).
- 17.10.505 Animal boarding commercial activities.
- 17.10.510 Animal care commercial activities.
- 17.10.520 Undertaking service commercial activities.

## Part 2 - Nonresidential Facility Types

#### Sections:

- 17.10,710 General description of Nonresidential Facilities.
- 17.10.720 Enclosed Nonresidential Facilities.
- 17.10.730 Open Nonresidential Facilities.
- 17.10.740 Drive-In Nonresidential Facilities.
- 17.10.750 Sidewalk Geafe, Nonresidential Facilities. faoility.
- 17-10-760-Shopping-Center-Facility-
- 17.10.770 Drive-Through Nonresidential Facilities.

#### 17.10.030 - Listing of activity classifications.

All activities are classified into the following activity types, which are described in Article II of this chapter. (See Section 17.10.050 for classification of combinations of activities resembling different types.) The names of these activity types start with capital letters throughout the zoning regulations.

A. Residential Activities:

Permanent

Residential Care

Service-Enriched Permanent Housing

Transitional Housing

Emergency Shelter Semi-Transient Bed and Breakfast Civic Activities: Essential Service Limited Child-Care Community Assembly Recreational Assembly Community Education Nonassembly Cultural

Β.

Administrative -

Health Care

Special Health Care

Utility and Vehicular

Extensive Impact

C. Commercial Activities: General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Fast-Food Restaurant

Convenience Market

Alcoholic Beverage Sales

Mechanical or Electronic Games

Medical Service

General Retail Sales

Large-Scale Combined Retail and Grocery Sales

Consumer Service

Consultative and Financial Service

Check Cashier and Check Cashing

Consumer Cleaning and Repair Service

Consumer Dry Cleaning Plant

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

Administrative

Business, Communication, and Media Service

Broadcasting and Recording Service

Research Service

General Wholesale Sales

Transient Habitation

Building Material Sales

Automobile and Other Light Vehicle Sales and Rental

Automobile and Other Light Vehicle Gas Station and Servicing

Automotive and Other Light Vehicle Repair and Cleaning

Taxi and Light Fleet-Based Service

Automotive Fee Parking

Transport-and-Warehousing

Animal Boarding

Animal Care

Undertaking Service

D. Industrial Activities:

Custom Manufacturing

Light Manufacturing

General Manufacturing

Heavy/High Impact Manufacturing

Research and Development

Construction Operations

Warehousing, Storage and Distribution

Regional Freight Transportation

Trucking and Truck-Related

Recycling and Waste-Related

Hazardous Materials Production, Storage and Waste Management

E. Agricultural and Extractive Activities:

Plant Nursery

Crop and Animal Raising

Mining and Quarrying

17.10.040 - Accessory activities.

K. Temporary conduct of a real estate sales office which is necessary and incidental to, arid located on the site of, a subdivision being developed into five (5) or more lots;

17.10.060 - Listing of facility classifications.

All facilities are classified into the following facility types, which are described in Section Article III of this chapter. (See Section 17.10.080 for classification of combinations of facilities resembling different types.) The names of these facility types start with capital letters throughout the zoning regulations.

A. Residential Facilities:

**One-Family Dwelling** 

One-Family Dwelling with Secondary Unit

Two-Family Dwelling

Multifamily Dwelling

Rooming House

Mobile Home

B. Nonresidential Facilities:

Enclosed

Open

Drive-in

Sidewalk Cafe Nonresidential Facilities

C. Signs:

Residential

Special

Development

Realty

Civic

Business

Advertising

D. Telecommunications Facilities:

Micro

Mini

Macro

Monopole

Tower

17.10.110 - Permanent residential activities.

Permanent Residential Activities include the occupancy of living accommodations on a weekly or longer basis, with none of the living units under the same ownership or management on the same lot being occupied on a shorter basis; but exclude institutional living arrangements other than state-licensed residential care facilities for six (6) or fewer residents including family foster care homes. However, such residential care facilities shall be subject to the three hundred (300) foot separation requirement in Section–17-102-212\_17.103.010(B). This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

#### 17.10.112 - Residential care residential activities.

Residential Care Residential Activities include all residential care facilities that require a state license or are state licensed for seven or more residents which provide twenty-four (24) hour primarily nonmedical care and supervision. Occupancy of living accommodations by six (6) or fewer disabled persons, elderiy persons, or persons in need of support services for chemical dependency recovery; or a family foster care home; or occupancy of any facilities supervised by or under contract with the State Department of Corrections, are excluded. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. State licensed residential care facilities for six (6) or fewer residents shall be treated as Permanent Residential Activities except with regard to the three hundred (300) foot separation requirement in Section-17-102-2-12 17.103.010(B).

17.10.120 - Semi-transient residential activities.

Semi-Transient Residential Activities include the occupancy of living accommodations partly on a weekly or longer basis and partly for a shorter time period, but with less than <u>thirty</u> 30-percent (30%) of the living units under the same ownership or management on the same lot being occupied on a less-than-weekly basis; but exclude institutional living arrangements involving the provision of a special kind of care or forced residence, such as in nursing homes, orphanages, asylums, and prisons. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

#### 17.10.125 - Bed and breakfast residential activities.

The provision of lodging services to transient guests on a less-than-weekly basis, other than in the case of activities classified by another Residential Activity (Sections 17.10.100 through 17.10.120) that have each of the following characteristics:

- A. The activity occupies a One-Family Dwelling Residential Facility, One-Family Dwelling with Secondary Unit Residential Facility, or a Two-Family Dwelling Residential Facility;
- B. The activity allows no more than <u>twelve (12)</u> adult paying guests at any time and contains no more than six (6) guest units;

- C. The activity is located in a facility that is owner occupied;
- D. The activity is located in a facility on a property with an existing or contingency historic rating of "A", "B", "C", or "D" or is a Landmark according to the City of Oakland Office of Historic Preservation
- E. The facility includes incidental eating and drinking services for lodgers only that are provided from a single kitchen per bed and breakfast establishment

This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.150 - Limited child-care activities.

Limited Child-Care Civic Activities include the provision of day-care service for <u>fourteen (14)</u> or fewer children, provided, however, that care for six <u>(6)</u> or more children be provided only in facilities licensed by a state or county agency. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.170 - Recreational assembly civic activities.

Recreational Assembly Civic Activities include the provision of recreational activities, typically performed by participants within public facilities. Examples of activities in this classification include but are not limited to the following:

- · Food service and other concessions located within public parks;
- Public and parochial playgrounds and playing fields;
- Temporary nonprofit festivals;
- Basketball courts, tennis courts, handball courts, lawn bowling, <u>leisure areas</u>, and similar outdoor park and recreational facilities;
- · Community outdoor swimming and wading pools, and other water play features;
- · Picnic areas.

This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.180 - Community education civic activities.

Community Education Civic Activities include the activities typically performed by the following institutions:

- Public and private day-care centers for <u>fifteen (15)</u> or more children;
- B. Public and private nursery schools and kindergartens;
- C. Public and private elementary, junior high, and high schools;
- D. Support services provided for independent living skills development including self-improvement education, employment and job training for both on-site and off-site residents in conjunction with Service-Enriched Permanent Housing and Transitional Housing Residential Activities.

This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

Part 3 - Commercial Activity Types

17.10.260 - General description of commercial activities.

17.10.270 - General food sales commercial activities.

17.10.272 - Full service restaurant commercial activities.

17.10.274 - Limited service restaurant and cafe.

17.10.280 - Fast-food restaurant commercial activities.

17.10.290 - Convenience market commercial activities.

17.10.300 - Alcoholic beverage sales commercial activities.

17.10.320 - Mechanical or electronic games commercial activities.

17.10.330 - Medical service commercial activities.

17.10.340 - General retail sales commercial activities.

17.10.345 - Large-scale combined retail and grocery sales commercial activities.

17.10.350 - Consumer service commercial activities.

17.10.360 - Consultative and financial service commercial activities.

17.10.365 - Check cashier and check cashing activity.

17.10.370 - Consumer cleaning and repair service commercial activities.

17.10.375 - Consumer dry cleaning plant commercial activities.

17.10.380 - Group assembly commercial activities.

17.10.385 - Personal instruction and improvement and small scale entertainment commercial activities.

17.10.390 - Administrative commercial activities.

17.10.400 - Business, communication, and media service commercial activities.

17.10.410 - Broadcasting and recording service commercial activities.

17.10.420 - Research service commercial activities.

17.10.430 - General wholesale sales commercial activities.

17.10.440 - Transient habitation commercial activities.

17.10.450 - Building material sales commercial activities.

17.10.460 - Automobile and other light vehicle sales and rental commercial activities.

17.10.470 - Automobile and other light vehicle gas station and servicing commercial activities.

17.10.480 - Automobile and other light vehicle repair and cleaning commercial activities.

17.10.485 - Taxi and light fleet-based service commercial activities.

17.10.490 - Automotive fee parking commercial activities.

17-10-500-Transport-and-warehousing-commorcial activities (does-not-apply-to-the-CIX-1,-CIX-2,-IG,-or-IO-zones).

17.10.505 - Animal boarding commercial activities.

17.10.510 - Animal care commercial activities.

17.10.520 - Undertaking service commercial activities.

17.10.272 - Full service restaurant commercial activities.

Restaurants providing food or beverage services to patrons who order and are served while seated (table service), and pay after eating. Only a minor proportion, if any, of the food is sold for consumption off-premises. These restaurants have kitchens that contain equipment suitable for cooking an assortment of foods. Also, see-17-102-210 Section 17.103.030 and 17.156.070 for definitions of a full-service restaurant in relation to Alcoholic Beverage Sales. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.280 - Fast-food restaurant commercial activities.

- A. Fast-Food Restaurant Commercial Activities include the retail sale of ready-to-eat prepared foods and beverages, for on- or off-premises consumption, whenever the foods and beverages are available upon a short waiting time and are primarily served in or on disposable wrappers, containers, or plates. Fast-Food Restaurants may also exhibit other design and operating characteristics, including: (<u>1</u>A) a limited menu; (<u>2</u>B) food is typically ordered and served at a service counter; (<u>3</u>C) food is paid for prior to consumption; (<u>4</u>D) the facility in which the activity/use is occurring provides a take-out counter space and space for customer queuing. They also include certain activities accessory to the above, as specified in Section 17.10.040.
- B. Except as may otherwise be allowed in Oakland Municipal Code (OMC) Chapters 5.49, 5.51, 8.09, and 9.52, tThe sale of ready-to-consume prepared foods from trucks, pushcarts or other movable equipment located on <u>public or private</u> property on a semi-permanent basis during hours of operation. Vehicular food vending generally has the following characteristics:
  - Food is ordered and served from a take-out counter that is integral to the catering truck;
  - Food is paid for prior to consumption;
  - Catering trucks, pushcarts or other movable equipment from which the food is sold typically have a take-out counter and space for customer queuing;
  - · Food and beverages are served in disposable wrappers, plates or containers; and
  - Food and beverages are prepared and sold for off-site consumption.
  - This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.350 - Consumer service commercial activities.

Consumer Service Commercial Activities include the provision of services of a personal nature, but exclude activities more specifically classified elsewhere. Examples of activities in this classification include, but are not limited to the following:

- Barber shops;
- Beauty salons;
- Laundromats;
- Nail salons;
- Full service laundry service and dry cleaners (not including dry cleaning plants);
- Shoe shine stands;
- Tailors;
- Tanning salons;
- Tattoo parlors;

- A pharmacy that exclusively sells prescription drugs, non-prescription drugs, and other medical related products.
- Massage services.

This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.380 - Group assembly commercial activities.

Group Assembly Commercial Activities include the provision of instructional, amusement, and or similar-other services of a similar nature to group assemblages of people. This classification does not include any activity classified in <u>Section</u> 17.10.160 Community Assembly Civic Activities, <u>Section</u> 17.10.170 Recreational Assembly Civic Activities, or <u>Section</u> 17.10.180 Community Education Civic Activities. Examples of activities in this classification include, but are not limited to the following:

- Yoga, martial arts, driving school, job training, and other instructional classes in facilities with two thousand (2,000) square feet or more of classroom or instructional space;
- Drive-in theaters;
- Theaters or venues with three thousand (3,000) square feet or more of <u>performance, lobby space</u>, <u>and audience floor area;</u>
- · Temporary carnivals, fairs, and circuses;
- Cabarets, night clubs, dance halls, <u>adult entertainment</u>, and pool halls;
- Banquet halls;
- Fitness clubs with two thousand (2,000) square feet or more of floor area.

This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.385 - Personal instruction and improvement and small scale entertainment commercial activities.

The provision of informational, instructional, personal improvement, and <u>other similar</u>-services of a <u>similar</u> nature. This-classification-also-includes-theaters-where-loss-than-three-hundred-(300)-people-are viewing-an-individual-stage-or-acreen. This classification does not include any activity classified as <u>Section</u> 17.10.180 Community Education Civic Activities or <u>Section</u> 17.10.380 Group Assembly Civic Activities. Examples of activities in this classification include but are not limited to the following:

- Yoga, martial arts, driving school, job training, and other instructional classes in facilities with less than two thousand (2,000) square feet of classroom or instructional space;
- Fitness clubs with less than two thousand (2,000) square feet of floor area;
- Theaters or venues with less than three thousand (3,000) square feet of <u>performance</u>, lobby <u>space</u>, and <u>audience</u> floor area.

This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.440 - Transient habitation commercial activities.

Transient Habitation Commercial Activities include the provision of lodging services to transient guests on a less-than-weekly basis, other than in the case of activities classified by Section 17.10.120

Semi-Transient Residential Activities or Section 17.10.125 Bed and Breakfast Commorcial-<u>Residential</u> Activities. <u>Examples include hotels and motels</u>. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.460 - Automobile and other light vehicle sales and rental commercial activities.

Automobile and <u>Other Light Vehicle Light-Truck-Sales</u>, <u>and</u> Rental, and Dolivory Commercial Activities include the sale, rental, leasing and incidental cleaning, servicing, and repair of small passenger vehicles and light trucks that have a gross vehicle weight <u>rating of</u> less than fourteen thousand (14,000) pounds such as cars, sports utility vehicles, motorcycles, pickup trucks, vans, <u>light tow trucks</u>, light trucks, boats and RVs. This classification also includes the retail or wholesale sale or rental, from the premises, of any type of goods where orders are placed predominantly by telephone or mail order with delivery being provided by motor vehicle. Delivery activities that include use of more than two (2) on-site tow trucks are excluded from this classification-and-included-in-the Warehousing. Distribution-and-Storage-of Outdoor-Storage-classification. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.480 - Automobile and other light vehicle repair and cleaning commercial activities.

Automobile and Other Light Vehicle Repair and Cleaning Commercial Activities include the major repair or painting of motor vehicles that have a gross vehicle weight <u>rating of less</u> than fourteen thousand (14,000) pounds, including body work and installation of major accessories, as well as the washing and polishing of motor vehicles. This classification does not include vehicle dismantling or salvage and tire retreading or recapping. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.485 - Taxi and light fleet-based service commercial activities.

Passenger transportation services, local delivery services, and other businesses that rely on fleets of three (3) or more vehicles with a gross vehicle weight rating rated-capacity-of less than-twelve-thousand five--hundred fourteen thousand (14,000) (12,500)--pounds. This classification includes parking, dispatching, and offices for taxicab and limousine operations, airport shuttles, medical transport, local messenger and document delivery services, janitorial services, and similar businesses. This classification deep-not<u>only</u> includes towing operations when except-fer-tow-truck-sorvices-whore-vehicles are taken to off-site locations and the tow trucks do not exceed the above gross vehicle weight rating. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

17.10.500 - Transport and warehousing commercial activities (does not apply to the CIX-1, CIX-2, IG, or IO-zones).

Transport-and-Warehousing-Commercial-Activities-include-the-provision-of-warehousing-and-storage, freight-handling, shipping, and trucking-services. This-classification-does-not-apply-to-the-CIX-1, CIX-2, IG, or IO-zonos. This-classification also includes certain-activities-accessory-to-the-abovo, as-specified in Section-17.10.040.

17.10.550 - Custom manufacturing industrial activities.

Custom Manufacturing Industrial Activities include the small-scale production of artisan and/or custom products. This activity typically includes the production of finished parts or products by hand, involving the use of hand tools and small-scale equipment within enclosed buildings. Custom Manufacturing Industrial Activities do not produce noise, vibration, air pollution, fire hazard or noxious

emission that will disturb or endanger neighboring properties. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. This classification includes but is not limited to the production of:

- <u>A</u> Beverages (including alcoholic) and food (excluding the production of highly pungent, odorcausing items, such as vinegar and yeast) with less than ten thousand (10,000) square feet of floor area;
- B. Cameras and photographic equipment;

<u>C.</u> Custom sign-making;

- D. Custom clothing;
- E Custom furniture building and refinishing;
- <u>F.</u> Professional, scientific, measuring, and controlling instruments;
- G. Musical instruments;
- H. Medical, dental, optical and orthopedic instruments and appliances, and similar items;
- I. \_\_\_\_Handicraft, art objects, and jewelry.

#### 17.10.560 - Light manufacturing industrial activities.

Light Manufacturing Industrial Activities include the manufacturing, compounding, processing, assembling, packaging, or treatment of components or products, primarily from previously prepared materials, and typically within enclosed buildings. Light Manufacturing Industrial Activities do not produce noise, vibration, air pollution, fire hazard or noxious emission that will disturb or endanger neighboring properties. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. This classification includes but is not limited to the production or assembly of:

- A. Production apparel manufacturing;
- B.\_\_Computer and electronic products;
- C.\_\_\_Pharmaceutical production;
- <u>D.</u> Beverages (including alcoholic) and food (excluding the production of highly pungent, odorcausing items, such as vinegar and yeast) with ten thousand (10,000) square feet or more of floor area;
- E. Electrical equipment, appliances, and components;
- F. Furniture and related products;
- G. Pharmaceutical production;
- <u>H.</u> Sporting and athletic goods.

#### 17.10.570 - General manufacturing industrial activities.

General Manufacturing Industrial Activities include the manufacturing, compounding, processing, assembling, packaging or treatment of products from extracted, raw, recycled or secondary materials; they may have some or all activities conducted outdoors. This classification excludes all activities under Intermediate Recycling Processing Facilities. The Zoning Administrator or his/her designee may place an activity that otherwise fits this description, but does not produce noise, vibration, air pollution, fire hazard, or noxious emission that will violate standard in Chapter 17.120, or an other federal, State or local standards into the Light Manufacturing Industrial Activities classification. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. Examples of activities in this classification include but are not limited to the following:

- <u>A.</u> Chemical manufacturing (except for the chemical products listed under Heavy/High/Impact Manufacturing);
- B. Glass manufacturing;
- <u>C.</u> Metal foundries;
- D. Wood product manufacturing;
- E.\_\_Heavy equipment and manufacturing;
- F. Paper finishing;
- <u>G.</u> Pipe production facilities;
- <u>H.</u> Textile mills;
- I.\_\_\_\_Tire retreading and recapping;
- Wood product manufacturing.

17.10.580 - Heavy/high impact manufacturing industrial activities.

Heavy/High Impact Manufacturing Industrial Activities include high impact or hazardous manufacturing processes. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040. Examples of activities in this classification include but are not limited to the following:

- A. \_\_Any manufacturing use with large-scale facilities for outdoor oil and gas storage;
- B. Any biotechnology research, development or production activities involving materials defined by the National Institute of Health as Risk Group 4 or Restricted Agents (commonly known as "biosafety level 4");
- C. Battery manufacturing and storage;
- D. Lime and gypsum products manufacturing;
- E. Non-ferrous metals production, processing, smelling and refining;
- <u>F.</u> Painting, coating and adhesive manufacturing;
- G.\_\_Synthetic dye and pigment manufacturing;
- H. Urethane and other open-cell foam product manufacturing;
- Petroleum and coal products manufacturing and refining;
- J. Primary metal smelting;
- K. Vinegar, yeast and other pungent, odor-causing items production;
- L. Leather tanning;
- <u>M.</u>\_Cement and asphalt manufacturing;
- H. Explosives manufacturing;
- O. Fertilizer and other agricultural chemical manufacturing.

17.10.584 - Regional freight transportation industrial activities.

Regional Freight Transportation Activities include the provision of freight handling and shipping services by water and rail. They include the inter- and intra-regional transportation of goods. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.
- A. Seaport. The accommodation of freight service and operations by ship. This classification includes piers, wharves &and docks, marine terminals, container and break-bulk storage areas (where container storage is an accessory, rather than principal activity), related inter-modal facilities, and support services such as port and harbor operations and navigational services.
- B. Rail yard. Accommodation of freight service and operations by rail.

17.10.585 - Trucking and truck-related industrial activities.

Trucking and Truck-Related Activities include the provision of freight handling and shipping services by trucks as well as parking, maintenance, and services for trucks and other heavy vehicles and equipment. Each classification involves the use of trucks and other heavy vehicles that have a gross vehicle weight rating greater than or equal to fourteen thousand (14,000) pounds. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

- A. Freight/Truck Terminal. The accommodation of local or worldwide freight by truck. This classification includes facilities used primarily for transfer, breaking-down, and/or consolidation of freight, as well as parking and dispatch of trucks.
- B. Truck Yard. Parking, dispatch, refueling, and incidental repair of trucks, buses, or other fleets of heavy vehicles, where there is no on-site freight storage or transfer. This classification includes corporation yards operated by public and private towing operations. This classification does not include local courier and delivery services; towing operations as an accessory activity to Automotive and Other Light Vehicle Repair and Cleaning (see Section 17.10.480).
- C. Truck Weigh Stations. The weighing of commercial trucks in truck weighing facilities.
- D. Truck and Other Heavy Vehicle Sales, Rental, and Leasing. Sales, rental, and leasing of medium and heavy trucks, truck tractors, construction or agricultural equipment, buses, commercial boats, heavy equipment, and other commercial vehicles.\_\_that\_have-gross-vehicio weight-ratings-creator-than-fourteen-thousand-(14,000) pounds,.\_\_This classification includes ing the sale, installation, accessory repair and servicing of related equipment and parts. This classification does not include vehicle dismantling or salvage and tire re-treading or recapping (See Salvage/Junk Yards, Section 17.10.583).
- E. Truck and Other Heavy Vehicle Service, Repair, and Refueling. Repair, fueling, and other servicing of medium and heavy trucks, truck tractors, construction or agricultural equipment, buses, boats, heavy equipment, and similar vehicles.\_-that-generally-have-gross-vehicle-weights greater-than-fourteen-thousand (14,000)-pounds,-\_ This classification\_includesing the sale, installation, and servicing of related equipment and parts. This classification includes fueling stations, repair shops, body and fender shops, wheel and brake shops, engine repair and rebuilding, welding, major painting service, tire sales and installation, and upholstery shops for trucks and other heavy vehicles. This classification does not include vehicle dismantling or salvage (See Salvage/Junk Yards, Section 17.10.583E).

#### 17.10.586 - Recycling and waste-related industrial activities.

Recycling and Waste-Related <u>Industrial</u> Activities include recycling collection, intermediate processing, and other activities related to the storage and processing of used and waste materials. This classification also includes certain activities accessory to the above, as specified in Section 17.10.040.

B. Primary Recycling Collection Centers. An activity accepting recyclable non-hazardous materials by donation, redemption, or purchase at facilities occupying an area of more than 500 square feet that are not operated incidental to a host use and that may have a permanent building. Primary collection centers typically use power-driven equipment to sort and condense material for shipment to an intermediate processor or other user. Primary <u>Recycling C</u>collection Ccenters may have a combination of outdoor processing and storage.

Part 2 - Nonresidential Facility Types

17.10.710 - General description of Nonresidential Facilities.

17.10.720 - Enclosed Nonresidential Facilities.

17.10.730 - Open Nonresidential Facilities,

17.10.740 - Drive-In Nonresidential Facilities.

17.10.750 - Sidewalk Ceafe, Nonresidential Ffaciiitiesy.

17-10-760-Shopping-Contor-Facility-

17.10.770 - Drive-Through Nonresidential Facilities.

17.10.750 - Sidewalk Ceafe, Nonresidential Ffacilitiesy.

A <u>Sidewalk Café Nonresidential Facility</u> "sidowolk-cafe"-is either a General Food Sales Commercial, Full Service Restaurant, Limited Service Restaurant and Cafe, Fast-Food Restaurant, or Alcoholic Beverage Sales Commercial Activity located on private property with a dining area which encroaches within the sidewalk area of the public right-of-way. Such dining area shall be defined by design elements which separate the establishment from the remainder of the sidewalk.

17-10-760 -- Shopping Conter-Facility.

A-Shopping-Center-<u>Nonresidential</u>Facility-ic-a-complex-of-one-or-more-retail-buildings-and-related facilities forming o-contral-retail-market-within-a-givon-aroa-and-having-a-common-parking-aroa-

(Prior-planning-code § 2614)

17.10.810 - Development Signs.

Development Signs are temporary Signs which announce the anticipated sale, lease, rental, or character of facilities being constructed or altered, or of facilities or lots in a real estate <u>development or</u> subdivision, <u>development</u>, or which identify persons or firms engaged in the promotion, design, construction, or alteration thereof.

17.11.020 - Designation and mapping of parks by category.

A. All parks and public open space lands in the City of Oakland shall be classified using the categories listed below:

RCA	Resource Conservation Area	
RSP	Region-Serving Park	· · · · · · · · · · · · · · · · · · ·
СР	Community Park	<u></u>
NP	Neighborhood Park	······································
AMP	Active Mini-Park	<u> </u>
PMP	Passive Mini-Park	· · · · · · · · · · · · · · · · · · ·
LP	Ljnear Park	
รบ	Special Use Park	
AĘ	Athletic Field Park	<u>, , , , , , , , , , , , , , , , , , , </u>

- B. Designation of each park on the zoning maps shall be followed by the two- or three-letter abbreviation corresponding to each park type in parentheses.
- C. If a new park is developed or acquired or if an existing park is to be changed to a new category, the Parks and Recreation Advisory Commission (PRAC) shall make a recommendation on the designation to the City Council, consistent with the park type definitions contained in the Open Space Conservation and Recreation (OSCAR) Element of the Oakland General Plan. The City Council shall hold a noticed public hearing prior to making a decision on the recommendation.

Chapter 17.13 - RH HILLSIDE RESIDENTIAL ZONES REGULATIONS Sections:

17.13.010 - Title, intent, and description.

17.13.020 - Required design review process.

17.13.030 - Permitted and conditionally permitted activities.

17.13.040 - Permitted and conditionally permitted facilities.

17.13.050 - Property development standards.

17.13.060 - Special regulations for mini-lot-and-planned unit developments.

17.13.070 - Other zoning provisions.

17.13.010 - Title, intent, and description.

- A. Title and Intent. The provisions of this Chapter shall be known as the Hillside Residential (RH) regulations. The intent of the RH regulations is to create, maintain, and enhance residential areas that are primarily characterized by detached, single unit structures on hillside lots.
- B. Description of Zones. This Chapter establishes land use regulations for the following four (4) zones:
  - 1. RH-1 Hillside Residential Zone 1. The intent of the RH-1 zone is to create, maintain, and enhance areas for single-family living on lots of one acre or more, and is appropriate in portions of the Oakland Hills.
  - RH-2 Hillside Residential Zone 2. The intent of the RH-2 zone is to create, maintain, and enhance areas for single-family living on lots of at least <u>twenty-five thousand (25,000)</u> square feet, and is appropriate in portions of the Oakland Hills.
  - 3. RH-3 Hillside Residential Zone 3. The intent of the RH-3 zone is to create, maintain, and enhance areas for single-family dwellings on lots of at least <u>twelve thousand (12,000)</u> square feet and is appropriate in portions of the Oakland Hills.
  - RH-4 Hillside Residential Zone 4. The intent of the RH-4 zone is to create, maintain, and enhance areas for single-family dwellings on lots of <u>six thousand five hundred (6,500)</u> to <u>eight</u> <u>thousand (8,000)</u> square feet and is typically appropriate in already developed areas of the Oakland Hills.

#### 17.13.030 - Permitted and conditionally permitted activities.

Table 17.13.01 lists the permitted, conditionally permitted, and prohibited activities in the RH zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

#### Table 17.13.01: Permitted and Conditionally Permitted Activities

Activities		Zo	nes		Additional
	<b>RH-1</b>	RH-2	RH-3	RH-4	Regulations

Residential Activities		·····	<u>,                                     </u>		1.
Permanent	P(L-1-)	P(L4)	P(L-1-)	P(L1)	17-102-212
Residential Care	 P(L1)	 PL1)	<u>P(L1)</u>	P(L1)	<u>17.103.010</u>
Service-Enriched Permanent Housing			<u> _</u>	1	
Transitional Housing	—		<u> </u>	_	ļ
Emergency Shelter	<u> </u>	<u> </u>	<u> _</u>	<u> </u>	
Semi-Transient	<u> _</u>		<u> _</u>		
Bed and Breakfast			<u> </u>	<u> _</u>	
Civic Activities			. <b></b> .		
Essential Service	Р	P	P	Р	1
Limited Child-Care Activities	Р	Р	P	P	
Community Assembly	С	С	С	c	
Recreational Assembly	С	С	С	С	
Community Education	C	С	с	c	
Nonassembly Cultural	С	С	c	c	
Administrative	с	С	С	c	
Health Care	_		<u> _</u>		
Special Health Care		<u> _</u>	<u> _</u>		
Utility and Vehicular	С	С	С	c	
Extensive Impact	С	С	C	C	
Commercial Activities (all)	—	-	_	—	17-112
Industrial Activities (all)			_		
Agriculture and Extractive Activities					
Crop and animal raising	C(L2)	C(L2)	C(L2)	C(L2)	
Plant nursery	С	С	С	c	
Mining and Quarrying			<u>                                     </u>		
Accessory off-street parking serving prohibited activities	С	с	С	С	17-102-100 17.116.075
Additienal-a <u>A</u> ctivities that <u>are listed as oroh bited. but</u> are permitted or conditionally permitted <u>on nearby lots in an adjacent zone.</u> -on-lots-near the boundary-thereof.	С	с	С	С	17.102.110

Limitations on Table 17.13.01:

**L1.** <u>Residential Care is only permitted in a One-Family Dwelling Residential Facility.</u> No state licensed residential care facility shall be located closer than three hundred (300) feet from any other state licensed residential care facility or Residential Care, Service-Enriched Permanent Housing. Transitional Housing, or Emergency Shelter Residential Activity. See Section-17-102-212\_17.103.010 for other regulations regarding these activities.

17.13.040 - Permitted and conditionally permitted facilities.

Table 17.13.02 lists the permitted, conditionally permitted, and prohibited facilities in the RH zones. The descriptions of these facilities are contained in Chapter 17.10. The descriptions of these facilities are contained in Chapter 17.10.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

# "L" designates facilities subject to certain limitations listed at the bottom of the table.

"-" designates facilities that are prohibited.

Facilities		Z	ones		Additional
	RH-1	RH-2	RH-3	RH-4	Regulations
Residential Facilities					
One-Family Dwelling	P	Ρ.	P	P	
One-Family Dwelling with Secondary Unit	P	P	P	P	17-102-360 17,103.080
Two-Family Dwelling	_	-	-	<u> </u>	
Multifamily Dwelling	_		1	-	
Rooming House		—	1-	-	
Mobile Home	_	-	]	-	
Nonresidential Facilities		<b>_</b>	<b></b>		
Enclosed Nonresidential	P	Р	P	P	
Open Nonresidential	P	Р	Р	Р	······································
Sidewalk Cafe		<u> _</u>			
Drive-In Nonresidential	_	-	1-	<b>—</b>	
Drive-Through Nonresidential	-		1-	1_	
TelecommUnications Facilities		··	· ·· <u>·</u>		
Micro Telecommunications	c	С	С	c	17.128
Mini Telecommunications	C	С	С	C	17.128
Macro Telecommunications	C	С	С	С	17.128
Monopole Telecommunications	C(L1)	C(L1)	C(L1)	C(L1)	17.128
Tower Telecommunications		-	_	<u></u>	17.128
Sign Facilities			<u> </u>		
Residential Signs	P	P	P	P	17.104
Special Signs	P	Ρ	Р	P	17.104
Development Signs	P	Р	Р	Р	17. <b>1</b> 04
Realty Signs	P	Р	Ρ	P	17.104
Civic Signs	P	Р	Р	Р	17.104
Business Signs		-		-	17.104
Advertising Signs		_	1_	_	17.104

# Table 17.13.02: Permitted and Conditionally Permitted Facilities

# 17.13.050 - Property development standards.

A. Zone Specific Standards. Table 17.13.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "—" indicates that a standard is not required in the specified zone.

Development Standards			Zones		Additional
	RH-1	RH-2	RH-3	RH-4	Regulations

Minimum Lot Dimensions					
Width mean	. 100 ft	100 ft	90 ft	45 ft	1
Frontage	25 ft	25 ft	25 ft	25 ft	1
Lot area	43,560 sf	25,000 sf	12,000 sf	6,500 sf or 8,000 sf	1, 2, 3
Maximum Density	1 primary u	nit per lot			4
Minimum Setbacks					
Minimum front (<20% street-to-setback gradient)	25 ft	25 ft	20 ft	20 ft	<u>5, 6</u> 5
Minimum front (>20% street-to-setback gradient)	5 ft	5 ft	5 ft	5 ft	<u>5, 56, 7</u> 6
Minimum interior side <20% footprint slope	6 <sup>.</sup> ft/15%	6 ft/15%	6 ft/10%	5 ft	78,98
Minimum interior side >20% footprint slope	6 ft/15%	6 ft/15%	6 ft/10%	5 ft/10%	7 <u>8, 9</u> 6
Minimum street side	6 ft	6 ft	6 ft	5 ft	9 <u>5, 10</u>
Rear	35 ft	35 ft	25 ft	20 ft	7, 10, 11
Maximum Lot Coverage and Floor Area Ratio (FAR)	See Table 1	7.13.04			
Height Regulations for All Lots with a Footprin	t Slope of <2	0%	<u> </u>	44 (1)	
Maximum wall height primary building	25 ft	25 ft	25 ft	25 ft	4213, 4314
Maximum pitched roof height primary building	30 ft	30 ft	30 ft	30 ft	4213, 4314
Maximum height for accessory structures	15 ft	15 ft	15 ft	15 ft	42 <u>13,</u> 43 <u>14</u>
Height Regulations for all Lots with a Footprint Slope of >20%	See Table 1 footprint slo		ight regulation	s for all lots with a	
Maximum Wall Length Before Articulation Required	40 ft	40 ft	40 ft	40 ft ,	44- <u>15</u>
Minimum Parking				- <u>·</u> ···································	
Minimum parking spaces required per unit	2	2	2	2	45 <u>16</u>
Additional parking spaces required for secondary unit	1	1	1	1	45 <u>16</u> , 46 <u>17</u>

Additional Regulations for Table 17.13.03:

2. In the RH-4 Zone, for Subdivision Maps of 4 or fewer lots where each lot created has a buildable area slope of less than or equal to <u>twenty percent (20%)</u>, the minimum lot size is <u>six thousand five hundred</u> (6,500) square feet. For Subdivision Maps where any one lot buildable area slope is greater than <u>twenty percent (20%)</u> or for Subdivision Maps of 5 or more lots, the minimum lot size is increased to <u>eight</u> thousand (8,000) square feet.

a. In order to determine buildable area slope of a subdivision, each lot shown on the Subdivision Map shall indicate the buildable area in dashed lines. The buildable area slope is measured at the steepest point between the front and rear setbacks (not included within the side setbacks).

4. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section <u>17.102.360 17.103.080</u>. Also applicable are the provisions of Section <u>17.102.370</u> with respect to additional kitchens for a dwelling unit, and the provisions of Section <u>17.102.300</u> with respect to dwelling units with five or more bedrooms.

5. On lots with only residential facilities, paved surfaces within required street-fronting yards, and any unimproved rights-of-way of adjacent streets, shall be limited to fifty percent (50%) on interior lots and

thirty percent (30%) on corner lots. Exceptions: The maximum percentages of paved surfaces specified in this additional regulation may be exceeded within unimproved rights-of-way in the following cases upon issuance of a private construction of public improvements (P-iob) permit or if undertaken directly by the City or by a private contractor under contract to the City.

a. Roadway construction or widening;

b. Sidewalk construction or widening; and

c Any work pursuant to an approved final map, parcel map or final development plan pursuant to a planned unit development permit.

For purposes of this additional regulation, an unimproved right-of-way is the portion of a street or alley right-of-way that is not paved.

**67.** The minimum front setback depth required by the applicable individual zone shall be reduced to five (5) feet on any lot with a street-to-setback gradient that exceeds twenty (20)-percent (20%), provided, however, that the distance from the edge of the pavement to a garage or carport elevation containing one or more vehicular entries shall be at least eighteen (18) feet (see Illustration for Table 17.13.03, [Additional Regulation 6], below). See Section 17.108.130 for allowed projections into setbacks.

78. See Section 17.108.080 for the required interior side and rear setbacks on a lot containing two or more living units and opposite a legally-required living room window.

8<u>9</u>. The minimum interior side setback is the greater of the two listed setbacks. Also, see Section 17.108.130 for allowed projections into setbacks.

**910.** In all residential zones, on every corner lot which abuts to the rear a key lot which is In a residential zone, there shall be provided on the street side of such corner lot a side setback with a minimum width equal to one-half (½) of the minimum front setback depth required on the key lot and no less than the minimum side setback width required along an interior side lot line of the corner lot. However, such side setback shall not be required to exceed five (5) feet in width if it would reduce to less than twenty-five (25) feet the buildable width of any corner lot. Such setback shall be provided unobstructed except for the accessory structures or the other facilities allowed therein by Section 17.108.130 (see Illustration for Table 17.13.03. [Additional Regulation 9], below). See also Section 17.110.040 C for special controls on location of detached accessory buildings on such corner lots. See Section 17.108.130 for allowed projections into setbacks.

40<u>11</u>. Wherever a rear lot line abuts an alley, one-half ( $\frac{1}{2}$ ) of the right-of-way width of the alley may be counted toward the required minimum rear setback; provided, however, that the portion of the minimum rear setback depth actually on the lot itself shall not be reduced to less than ten (<u>10</u>) feet. Also, see Section 17.108.130 for allowed projections into setbacks.

12.4 For lots which abut an adjoining rear setback, the minimum rear setback depth shall be increased by an additional one-half ( $\frac{1}{2}$ ) foot of rear setback depth for each additional one foot of lot depth over one hundred (100) feet, up to a maximum rear setback depth of forty (40) feet.

42<u>13</u>. See Section 17.108.030 for allowed projections above height limits and Section 17.108.020 for increased height limits for civic buildings.

43<u>14</u>. If at least sixty (60)-percent (60%) of the buildings in the immediate context are no more than one story in height, the maximum wall height shall be fifteen (15) feet within the front twelve (12) feet of buildable area. The immediate context shall consist of the five closest lots on each side of the project site plus the ten closest lots on the opposite side of the street; however, the Director of City Planning may make an alternative determination of immediate context based on specific site conditions. Such determination shall be in writing and included as part of any approval of any variance, conditional use permit, design review, determination of exemption from design review, or other special zoning approval or, if no special zoning approval is required, part of any Planning Department approval of a building permit application.

14<u>5.</u> If the total wall length within ten (10) feet of the side lot line exceeds forty (40) feet, then the building wall shall be articulated by at least one section of additional setback. See design guidelines for more specific bulk and context standards.

165. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117. Additional parking standards apply within the S-11 Zone, as prescribed in Section 17.92.

17.6 One parking space for the Secondary Unit is required in addition to any required parking spaces for the Primary Unit. Additional parking regulations that apply to Secondary Units are provided in Section 17–102-360 17.103.080.

Regulation	Lot Size in Square Feet							
	< 5,000	> 5,000 and <12,000	> 12,000 and <25,000	> 25,000 and <43,560	> 43,560	Regulations		
Maximum FAR	0.55	0.50	0.45	0.30	0.20	1		
Maximum Lot Coverage (%)	40%	40%	30%	20%	15%	2		

#### Table 17.13.04 Floor Area Ratio (FAR) and Lot Coverage

Additional Regulations for Table 17.13.04:

1. Lots with less than <u>five thousand (5,000)</u> square feet in area may have a dwelling with at least <u>two</u> thousand (2,000) square feet, regardless of FAR listed.

2. Lots with less than <u>five thousand (5,000)</u> square feet in area may have a lot coverage of up to <u>two</u> thousand (2,000) square feet regardless of lot coverage percentage (%) listed.

C. Height. Table 17.13.05 below prescribes height standards associated with different sloped lots. The numbers in the "Additional Regulations" column refer to the regulations listed at the end of the Table.

Regulation	With a Footprint Slope of:			Ups ope Lot Height Regulations With a Footprint Slope of:	Additional Regulations	
	> 20% and < 40%	> 40% and < 60%	> 60%	> 20%		
Maximum Height for Detached Accessory Structures	15 ft	15 ft	15 ft	15 ft	1	
Maximum Wall Height Primary Building	32 ft	34 ft	36 ft	32 ft	1, 2	
Maximum Wall Height Primary Building with a CUP	36 ft	38 ft	40 ft	35 ft	1	
Maximum Pitched Roof Height Primary Building	36 ft	38 ft	40 ft	35 ft	1, 2	
Maximum Height Above Edge of Pavement	18 ft	18 ft	18 ft	N/A	1	
Maximum Height Above the Ground Elevation at the Rear Setback Line	N/A	N/A	N/A	24 ft	1	
Maximum Height from Finished or Existing Grade (whichever is Jower) Within 20' of the Front Property Line	N/A	N/A	N/A	24 ft	1, 3	

## Table 17.13.05 Height Regulations for all Lots with a Footprint Slope of >20%

Additional Regulations for Table 17.13.05:

2. On a downslope lot greater than forty (40)-percent (40%) footprint slope, the rear wall of an attached garage or carport may exceed the wall height and roof height by five (5) feet, but may not exceed eighteen (18) feet above ground elevation at edge of pavement, if the garage or carport conforms with all of the following criteria:

a. Maximum width is twenty-two (22) feet and maximum depth is twenty (20) feet; and

b. Garage or carport floor is at the same level as the edge of the street pavement resulting from the project at the center point of the driveway entrance or is at a lower level; and

c. Maximum height above the garage or carport floor is ten (10) feet for walls to the top of the plate or flat roof and twelve (12) feet for pitched roofs.

See Illustration for Table 17.13.05 [Additional Regulation 2], below.

3. The building height is measured from finished or existing grade, whichever is lower.

17.13.060 - Special regulations for mini-lot-and-planned unit developments.

A...-Mini-lot-Devolopments.-In-mini-lot--developments,--certain-regulations--that--otherwise-apply-to individual-lets--in-tho-RH--zones-may-be-waived-or-modified-when-and--as--prescribed--in-Section 17-102-320

B. Planned Unit Developments. Large integrated developments shall be subject to the Planned Unit Development regulations in Chapter 17.142 if they exceed the sizes specified therein. In developments which are approved pursuant to said regulations, certain uses may be permitted in addition to those otherwise allowed in the RH zones, and certain of the other regulations applying in said zone may be waived or modified.

# 17.13.070 - Other zoning provisions.

E. Landscaping and Screening Standards. The <u>landscaping and screening</u> regulations set forth in Ghapter 17.124 and <u>Chapter 17.102.400</u>, screening of <u>utility meters</u>, otc., shall apply in the RH zones.

1

1

17.15.010 - Title, intent, and description.

- A. Title and Intent. The provisions of this Chapter shall be known as the Detached Unit Residential (RD) regulations. The intent of the RD regulations is to create, maintain, and enhance residential areas primarily characterized by detached, single-unit structures.
- B. Description of Zones. This Chapter establishes land use regulations for the following two zones:
  - 1. RD-1 Detached Unit Residential Zone 1. The intent of the RD-1 zone is to create, maintain, and enhance areas with detached, single unit structures. A limited number of commercial uses will be permitted or conditionally permitted in existing non-residential facilities.
  - RD-2 Detached Unit Residential Zone 2. The intent of the RD-2 zone is to create, maintain, and enhance areas with detached, single unit structures, with allowances for two-family structures on lots larger than <u>six thousand (6,000)</u> square feet. A limited number of commercial uses will be permitted or conditionally permitted in existing non-residential facilities.

17.15.030 - Permitted and conditionally permitted activities.

Table 17.15.01 lists the permitted, conditionally permitted, and prohibited activities in the RD zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17,134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

#### Table 17.15.01: Permitted and Conditionally Permitted Activities

Activities	Z	ones	Additional	
	RD-1	RD-2	Regulations	
Residential Activities				
Permanent	P(L1)	P(L1)	17.102.21217.103.010	
Residential Care	<u>P(L1)</u>	G <u>P</u> (L1)	47-102-21217.103.010	
Service-Enriched Permanent Housing		C(L1)	47-102-21217.103.010	
Transitional Housing		C(L1)	47.102.21217.103.010	
Emergency Shelter				
Semi-Transient				
Bed and Breakfast	c	С	17.10.125	
CiVic Activities	·			
Essential Service	P	Р		
Limited Child-Care Activities	P	Р		
Community Assembly	c	С		
Recreational Assembly	C	С		
Community Education	C	С		
Nonassembly Cultural	C	С		
Administrative	c	С		

Health Care		,
Special Health Care	<u> </u>	
Utility and Vehicular	С	C
Extensive Impact	С	c
Commercial Activities		<u>}</u>
General Food Sales	C(L2)(L3)	C(L2)(L3)
Full Service Restaurants	C(L2)(L3)	C(L2)(L3)
Limited Service Restaurant and Cafe	· C(L2)(L3)	C(L2)(L3)
Fast-Food Restaurant	·	
Convenience Market	_	
Alcoholic Beverage Sales	—(L4)	(L4)
Mechanical or Electronic Games		
Medical Service	. – ·	
General Retail Sales	C(L2)(L3)	C(L2)(L3)
Large-Scale Combined Retail and Grocery Sales	 	
Consumer Service		
Consultative and Financial Service	·	
Check Cashier and Check Cashing	<u>`</u>	
Consumer Cleaning and Repair Service		j
Consumer Dry Cleaning Plant		·
Group Assembly		
Personal Instruction and Improvement Services		
Administrative	P(L3)(L5)	P(L3)(L5)
Business, Communication, and Media Services		·
Broadcasting and Recording Services Commercial Activities		
Research Service	.	
General Wholesale Sales		
Transient Habitation	<u>}</u>	_
Wholesale-and Professional-Building Material Sales		—
Automobile and Other Light Vehicle Sales and Rental	)—	
Automobile and Other Light Vehicle Gas Station and Servicing		—
Automobile and Other Light Vehicle Repair and Cleaning		—
Taxi and Light Fleet-Based Services		
Automotive Fee Parking	<u> </u>	
Animal Boarding		_ · _ ,
Animal Care		_
Undertaking Service		
Industrial Activities (all)		
Agriculture and Extractive Activities		
Crop and animal raising	C(L6)	C(L6)
Plant nursery	С	c

Mining and Quarrying	_		
Accessory off-street parking serving prohibited activities	С	С	17-102-100 <u>17.116.075</u>
Activities that are listed as prohibited, but are permitted or conditionally oermitted on nearby lots in an adjacent zone Additional-activities-that are permitted or conditionally-permitted in-an-adjacent zone, on-lots-near-the boundary-thereof	С	C	17.102.110

Limitations on Table 17.15.01:

L1. <u>Residential Care is only permitted in a One-Family Dwelling Residential Facility.</u> No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such activity. See Section 17-102-21217.103.010 for other regulations regarding these activities. <u>Also, Rosidential-Care is only permitted in a One-Family-Dwelling-Residential-Facility.</u>

17.15.040 - Permitted and conditionally permitted facilities.

Table 17.15.02 lists the permitted, conditionally permitted, and prohibited facilities in the RD zones. The descriptions of these facilities are contained in Chapter 17.10.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates facilities subject to certain limitations listed at the bottom of the table.

"--- " designates facilities that are prohibited.

Table 17.15.02: Permitted and Conditionally Permitted Facilities

Facilities		Zones	Additional
	RD-1	RD-2	Regulations
Residential Facilities			
One-Family Dwelling	P	P	
One-Family Dwelling with Secondary Unit	P	P	17.102.360 17.103.060
Two-Family Dwelling		C(L1)	
Multifamily Dwelling	_	-	
Rooming House	<u> </u>	_	
Mobile Home			
Nonresidential Facilities	· · ·		
Enclosed Nonresidential	Р	Р	
Open Nonresidential	P	P	
Sidewalk Cafe	P(L2)	P(L2)	17-102.335 17.103.090
Drive-In Nonresidential		—	
Drive-Through Nonresidential		_	
Telecommunications Facilities			
Micro Telecommunications	C	C	17,128
Minj Telecommunications	C	С	17.128
Macro Telecommunications	C	С	17.128

Monopole Telecommunications	C	C	17.126
Tower Telecommunications			17.128
Sign Facilities			
Residential Signs	P	P	17.104
Special Signs	Р	P	17.104
Development Signs	P	P	17.104
Realty Signs	P	Р	17.104
Civic Signs	P	P	17.104
Business Signs	P(L3)	P(L3)	17.104
AdVertising Signs		_	17.104

Limitations on Table 17.15.02:

**L2.** Sidewalk cafes are allowed only as an accessory facility to an already approved Full Service Restaurant or Limited Service Restaurant and Cafe. The sidewalk cafe may only operate within the hours of 7:00 a.m. to 10:00 p.m. No more than three (3) tables and no more than ten (10) chairs or seats are allowed. If more tables or chairs are requested, a Conditional Use Permit (CUP) is required (see Chapter 17.134 for the CUP procedure). See-17-102-335\_Section\_17.103.090 for other regulations regarding Sidewalk Cafes; however, the regulations in this section supersede any contradicting regulations in Section-17-102-335\_17.103.090.

17.15.050 - Property development standards.

A. Zone Specific Standards. Table 17.15.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "—" indicates that a standard is not required in the specified zone.

Development Standards		Zones	Additional
1	RD-1	RD-2	Regulations
Minimum Lot Dimensions			
Width mean	45 ft	45 ft	1
Frontage	25 ft	25 ft	1 .
Lot area	5,000 sf	5,000 sf	1, 2
Maxìmum Density	·		
Permitted density	1 primary Unit per lot	1 primary Unit per lot	3, 4
Conditionally permitted density		2 units on lots 6,000 sf or greater	3, 5
Mİŋimum Setbacks	annanna - 1976-1977		
Minimum front (<20% street-to-setback gradient)	20 ft	20 ft	6
Minimum front (>20% street-to-setback gradient)	5 ft .	5 ft	6, 7 <u>, 8</u>
Minimum interior side <20% footprint slope	5 ft	5 ft	8 <u>9</u> , 9 <u>10</u>
Minimum interior side >20% footprint slope	5 ft/10%	5 ft	8 <u>9</u> , 9 <u>10, 1011</u>
Minimum street sjde	5 ft	5 ft	8 <u>, 9</u> , 41 <u>12</u>
Rear	20 ft	15 ft	8 <u>9,</u> 42 <u>13,</u> 13 <u>14</u>
Side and Rear Setbacks for Smaller Lots	See Table 17.15.04 for	setbacks for smaller lots	

# Table 17.15.03: Property Development Standards

Floor Area Ratio (FAR) and Lot Coverage	See Table 17.	15.05	
Height Regulations for All Lots With a Footprint Slope of <20%			
Maximum wall height primary building	25 ft	25 ft	14, 45- <u>16</u>
Maximum pitched roof height primary building	30 ft	30 ft	14, 45- <u>16</u>
Maximum height for accessory structures	15 ft	15 ft	14, 46- <u>16</u>
Height Regulations for all Lots With a Footprint Slope of >20%	See Table 17. footprint slope		
MaximUm Wall Length Before Articulation Required	40 ft	40 ft	46- <u>17</u>
Minimum Parking	÷		
Minimum parking spaces required per unit	2	1.5	17- <u>18</u>
Additional parking spaces required for secondary unit	1	1	1 <del>7,</del> 18 <u>, 19</u>
Minimum Open Space			
Group open space per unit	N/A	300 sf .	19 <u>20</u>
Group open space per unit when private open space substituted	N/A	100 sf	49 <u>20</u>

Additional Regulations for Table 17.15.03:

**4.** A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section-17-102-360 17.103.080.

**5.** A minimum lot size of <u>six thousand (6,000)</u> square feet is required in order to apply for a conditional use permit for a second primary dwelling in the RD-2 Zone. A conditional use permit for a Two-Family Dwelling Residential Facility or for two (2) dwelling units on a lot may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the Conditional use Permit (CUP) procedure in Chapter 17.134 and to all of the following additional use permit criteria:

a. That the proposed development will not adversely affect adjoining property, nor the surrounding neighborhood, with consideration to be given to density; to the availability of neighborhood facilities and play space; to the generation of traffic and the capacity of surrounding streets; arid to all other similar, relevant factors;

b. That the site design and landscaping and the scale, height, length and width, bulk, coverage, and exterior treatment of structures are in harmony with neighborhood character and with facilities on nearby lots;

c. That the shape and siting of the facilities, and especially of any portions thereof which exceed one story in height, are such as to minimize blocking of views and direct sunlight from nearby lots and from other Residential Facilities in the surrounding neighborhood;

d. That the design and site planning of the buildings, open areas, parking and service areas, and other facilities provide a convenient, attractive, and functional living environment; and that paths, stairways, accessways, and corridors are designed to ensure privacy;

e. That lot shape, size, and dimensions allow a development which will provide satisfactory internal living conditions without adversely affecting the privacy, safety, or residential amenity of adjacent residences.

8. On lots with only residential facilities, paved surfaces within required street-fronting yards, and any unimproved rights-of-way of adjacent streets, shall be limited to fifty percent (50%) on interior lots and

thirty percent (30%) on corner lots. Exceptions: The maximum percentages of paved surfaces specified in this additional regulation may be exceeded within unimproved rights-of-way in the following cases upon issuance of a private construction of public improvements (P-job) permit or if undertaken directly by the City or by a private contractor under contract to the City:

<u>a.</u> Roadway construction or widening;

b. Sidewalk construction or widening; and

c. Any work pursuant to an approved final map, parcel map or final development plan pursuant to a planned unit development permit.

For purposes of this additional regulation, an unimproved right-of-way is the portion of a street or alley right-of-way that is not paved.

89. See additional reduced side, and rear setbacks for smaller lots, Table 17.15.04. See Section 17.108.130 for allowed projections into setbacks.

**910.** See Section 17.108.080 for the required interior side and rear setback on a lot containing two or more living units and opposite a legally-required living room window.

40<u>11</u>. For RD-1, the minimum interior side setback is the greater of the two listed setbacks, either five (5) feet or ten (40)-percent (10%) of the lot width, whichever is greater.

1412. In all residential zones, on every corner lot which abuts to the rear a key lot which is in a residential zone, there shall be provided on the street side of such corner lot a side setback with a minimum width equal to one-half (½) of the minimum front setback depth required on the key lot and no less than the minimum side setback width required along an interior side lot line of the corner lot However, such side setback shall not be required to exceed five (5) feet in width if it would reduce to less than twenty-five (25) feet the buildable width of any corner lot. Such setback shall be provided unobstructed except for the accessory structures or the other facilities allowed therein by Section 17.108.130 (see Illustration for Table 17.15.03 [Additional Regulation 11], below). See also Section 17.110.040 C for special controls on location of detached accessory buildings on such corner lots.

42<u>13</u>. Wherever a rear lot line abuts an alley, one-half ( $\frac{1}{2}$ ) of the right-of-way width of the alley may be counted toward the required minimum rear setback; provided, however, that the portion of the minimum rear setback depth actually on the lot itself shall not be so reduced to less than ten feet. Also, see Section 17.108.130 for allowed projections into setbacks.

43<u>14</u>. In the RD-1 zone, for lots which abut an adjoining rear setback, the minimum rear setback depth shall be increased by an additional one-half (½) foot of rear setback depth for each additional one foot of lot depth over one hundred (100) feet, up to a maximum rear setback depth of forty (40) feet.

44<u>15.</u> See Section 17.108.030 for allowed projections above height limits and Section 17.108.020 for increased height limits for civic buildings.

**4516.** If at least sixty (60)-percent (60%) of the buildings in the immediate context are no more than one story in height, the maximum wall height shall be fifteen (15) feet within the front twelve (12) feet of buildable area. The immediate context shall consist of the five closest lots on each side of the project site plus the ten closest lots on the opposite side of the street; however, the Director of City Planning may make an alternative determination of immediate context based on specific site conditions. Such determination shall be in writing and included as part of any approval of any variance, conditional use permit, design review, determination of exemption from design review, or other special zoning approval or, if no special zoning approval is required, part of any Planning Department approval of a building permit application.

**16**<u>17</u>. If the total wall length within ten (10) feet of the side lot line exceeds forty (40) feet, then the building wall shall be articulated by at least one section of additional setback. See design guidelines for more specific bulk regulations and context standards.

47<u>18</u>. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117. Also, additional parking standards apply within the S-11 and S-12 Zones, as prescribed in Section 17.92 and Section 17.94.

48<u>19</u>. One parking space for the Secondary Unit is required in addition to any required parking spaces for the Primary Unit. Additional parking regulations that apply to Secondary Units are provided in Section 17-102-360 17.103.080.

4920. Each square foot of private usable open space equals two (2) square feet towards the total usable open space requirement, except that actual group space shall be provided in the minimum amount of one hundred (100) square feet per dwelling unit. All usable open space shall meet the standards contained in Chapter 17.126.

B. Setbacks for Smaller Lots. Table 17,15.04 below prescribes reduced setback standards for lots less than <u>four thousand (4,000)</u> square feet. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table.

Regulation	Lot Size							
	< 4,000 sf or < 40 feet wide	< 3,000 sf or < 35 feet wide	Additional Regulations					
Minimum Setbacks			·····					
Minimum interior side	4 ft	3 ft	1					
Minimum street side	4 ft .	3 ft	1					
Rear	15 ft	15 ft	1					

#### Table 17.15.04 Setbacks for Smaller Lots

#### Table 17.15.05 Floor Area Ratio (FAR) and Lot Coverage

Regulation	Lot Size in Square Feet							
	< 5,000	> 5,000 and < 12,000	> 12,000 and <25,000	> 25,000 and < 43,560	> 43,560	Additional Regulations		
Maximum FAR for Lots with a Footprint Slope > 20%	0.55	0.50	0.45	0.30	0.20	1		
Maximum Lot Coverage (%)	40%	40%	30%	20%	15%	2		

Additional Regulations for Table 17.15.05:

1. FAR only applies to lots that have a footprint slope of greater than <u>twenty percent (20%)</u>. Lots less than 5,000 square feet may have a dwelling with a minimum of 2,000 square feet of floor area, regardless of FAR listed.

D. Height. Table 17.15.06 below prescribes height standards associated with different sloped lots. The numbers in the right-hand column refer to the additional regulations listed at the end of the Table.

Regulation		ot Height Reg print Slope of:	ulations	Upslope Lot Height Regulations With a Footprint Slope of:		
	>20% and <40%	>40% and <60%	>60%	>20%	Additional Regulations	
Maximum Height for Detached Accessory	15 ft	15 ft	15 ft	15 ft	1	

Table 17.15.06 Height Regulations for all Lots with a Footprint Slope of >20%

Structures			· ·		
Maximum Wall Height Primary Building	32 ft	34 ft	36 ft	32 ft	1, 2
Maximum Wall Height Primary Building with a CUP	36 ft	38 ft	40 ft	35 ft	1
Maximum Pitched Roof Height Primary Building	36 ft	38 ft	40 ft	35 ft	1, 2
Maximum Height Above Edge of Pavement	la ft	18 ft	18 ft	N/A	1
Maximum Height Above the Ground Elevation at the Rear Setback Line	N/A	N/A	N/A	24 ft	1
Maximum Height from Finished or Existing Grade (whicheVer is greater) Within 20' of the Front Property Line	N/A	N/A	N/A	24 ft	1, 3

Additional Regulations for Table 17.15.06:

2. On a downslope lot greater than forty (40)-percent (40%) footprint slope, the rear wall of an attached garage or carport may exceed the wall height and roof height by five (5) feet, but may not exceed eighteen (18) feet above ground elevation at edge of pavement, if the garage or carport conforms with all of the following criteria:

a. Maximum width is twenty-two (22) feet and maximum depth is twenty (20) feet; and

b. Garage or carport floor is at the same level as the edge of the street pavement resulting from the project at the center point of the driveway entrance or is at a lower level; and

c. Maximum height above the garage or carport floor is ten (10) feet for walls to the top of the plate or flat roof and twelve (12) feet for pitched roofs (see Illustration for Table 17.15.06 [Additional Regulation 2], below).

17.15.060 - Special regulations for mini-lot and planned unit developments.

- A. Mini-Lot Developments. In mini-lot developments, certain regulations that otherwise apply to individual lots in the RD-1 zones may be waived or modified when and as prescribed in-Section 17-102-320 Chapter 17.142.
- B. Planned Unit Developments. Large integrated developments shall be subject to the Planned Unit Development regulations in Chapter 17.142 if they exceed the sizes specified therein. In developments which are approved pursuant to said regulations, certain uses may be permitted in addition to those otherwise allowed in the RH zones, and certain of the other regulations applying in said zone may be waived or modified.

17.15.070 - Other zoning provisions.

- A. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- B. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- E. Landscaping and Screening Standards. <u>The landscaping and screening regulations set forth in</u> <u>Chapter 17.124 shall apply in</u> The regulations-sot-forth-in-Chapter-17-124-and-Chapter-17-102-400, screening-of-utility-meters, etc., shall apply-in-the RD zones.

17.17.030 - Permitted and conditionally permitted activities.

Table 17.17.01 lists the permitted, conditionally permitted, and prohibited activities in the RM zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

Activities	Primary Zo	ones			Combining Zone*	Additional Regulations
	RM-1	RM-2	RM-3	RM-4	c	
Residential Activities				<u> </u>		······································
Permanent	P(L1)	P(L1)	P(L1)	P(L1)	P(L1)	17-102-21217.103.010
Residential Care	PG(L1)	PC(L1)	PG(Li)	PC(L1)	PC(L1)	47-102-24217.103.010
Service-Enriched Permanent Housing	C(L1)	C(L1)	C(L1)	C(L1)	C(L1)	17-102-212 <u>17.103.010</u>
Transitional Housing	C(L1)	C(L1)	C(L1)	C(L1) ·	C(L1)	17-102-21217.103.010
Emergency Shelter						
Semi-Transient						
Bed and Breakfast	С	С	С	С	Р	17.10.125
Civic Activities						
Essential Service	Р	Р	Р	Р	Р	
Limited Child-Care Activities	P	Р	Р	Р	Р	
Community Assembly	С	с	С	C.	С	
Recreational Assembly	С	С	С	С	С	
Community Education	c ·	C .	С	С	C	
Nonassembly Cultural	С	C .	c	С	С	
Administrative	С	С	С	С	С	
Health Care	С	С	c	С	C	
Special Health Care		— <u> </u>		-	<u> </u>	
Utility and Vehjcular	С	с	C	С	С	}
Extensive Impact	С	С	C	С	С	
Commercial Activities						
General Food Sales	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	
Full Service Restaurants	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	
Limited Service Restaurant and Cafe	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	
Fast-Food Restaurant			[ <u> </u>		-	
Convenience Market		-	-		— .	

Table 17.17.01: Permitted and Conditionally Permitted Activities

Alcoholic Beverage Sales	—(L5)	(L5)	(L5)	(L5)	(1_5)	1
Mechanical or Electronic Games	<u>i</u>	_	1		<u> _</u>	
Medical Sen/ice	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	GP(L6)	
General Retail Sales	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	· · · · · · · · · · · · · · · · · · ·
Large-Scale Combined Retail and Grocery Sales	_	_	<u> </u>	-	_	
Consumer Service	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	PC(L6)	
Consultative and Financial Service	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	[
Check Cashier and Check Cashing	[	-	<u> </u>			1
Consumer Cleaning and Repair Service	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	
Consumer Dry Cleaning Plant	_		_	-	_	
Group Assembly	_	—	–	—	C(L6)(L7)	
Personal Instruction and Improvement Services	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	C(L2)(L3)	P(L4)	
Administrative	P(L2)(L3) (L <u>8</u> 7)	P(L2)(L3) (L <u>8</u> 7)	P(L2)(L3) (L <u>8</u> 7)	P(L2)(L3) (L7 <u>L8</u> )	P(L4)	
Business, Communication, and Media Services	-	-	C(L2)(L3)	C(L2)(L3)	P(L4)	
Broadcasting and Recording Services Commercial Activities	-	-	-	—		
Research Service	_	—	-	_	_	
General Wholesale Sales	-	—	—	<u> </u>	<b>—</b> .	
Transient Habitation	_	_	_	—		
-Wholesale-ond-Professional Building Material Sales	-	—	_	—	_	
Automobile and Other Light Vehicle Sales and Rental	_	_	_	_	_	
Automobile and Other Light Vehicle Gas Station and Servicing	_	-	_	-		
Automobile and Other Light Vehicle Repair and Cleaning	_	-	-		-	
Taxi and Light Fleet-Based Services	_	-			—	
Automotive Fee Parking	-	_	—	—	—	
Animal Boarding	-	—		—	—	
Animal Care	_	_	-	—	- ·	
Undertaking Service						
Industrial Activities (all)	_	—	<b></b>	<b> </b>		
Agriculture and Extractive Activities						
Crop and animal raising	C(L8 <u>L9</u> )	C(L <u>9</u> 8)	C(L <u>9</u> 8)	C(L <u>9</u> 8)	C(L <u>9</u> 8)	
Plant nursery	c	С	С	С	C	
Mining and Quarrying	_					
Accessory off-street parking serving prohibited activities	-	-		<u> </u>		17-102-100 17.116.075
Activities that are listed as orohibited but are permitted or conditionally permitted on nearby lots in an adjacent zone Additienat	c	С	C	C	С	17.102.110

activities that are permitted or sonditionally	•			
permitted in an adjacent-zone, en lots near		1	•	
the boundory-thereof				

Limitations on Table 17.17.01:

\* If a base zone (RM-1, RM-2, RM-3, or RM-4) also has the C combining zone, the C regulations supersede the base zone.

L1. Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when not located in a One-Family Dwelling Residential Facility. No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such activity. See Section-17-102-212\_17.103.010 for other regulations regarding these activities. <u>Also, Rosidontial-Care-is only permitted-upon-the-granting-of-a</u> Conditional-Use-Permit (see Chapter-17-134-for-tho-CUP procedure) when not-located in a One-Family Dwolling Residential Focility.

L4. These activities may only be located on the ground floor of an existing or new nonresidential facility and may only operate within the hours of 7:00 a.m. and 10:00 p.m.; a Conditional Use Permit (CUP) is required if the facility exceeds <u>three thousand (3,000)</u> square feet (see Chapter 17.134 for the CUP procedure).

#### L7. Adult entertainment activities are prohibited.

**L78**. These activities may only be located in an existing ground floor of nonresidential facility that was both built prior to the effective date of this chapter (April 15, 2011) and not original used for a Civic Activity. For the purposes of this limitation, a facility is considered built if it received its certificate of occupancy or passed its final building inspection on its building permit Also, these activities may only operate within the hours of 7:00 a.m. and 10:00 p.m.; a Conditional Use Permit (CUP) is required if the ground floor nonresidential facility exceeds 1,500 square feet (see Chapter 17.134 for the CUP procedure).

**L8<u>L9</u>**. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the criteria contained in Section 17.134.050, this activity must meet the following use permit criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.17.040 - Permitted and conditionally permitted facilities.

Table 17.17.02 lists the permitted, conditionally permitted, and prohibited facilities in the RM zones. The descriptions of these facilities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates facilities subject to certain limitations listed at the bottom of the table.

"---" designates facilities that are prohibited.

Table 17.17.02: Permitted and Conditional	y Permitted Facilities
---	------------------------

Facilities	Zones					Additional	
	RM-1	RM-2	RM-3	RM-4	C+	Regulations	
Residential Facilities	 _					· · ·	
One-Family Dwelling	Р	P	P	Р	Р		
One-Family Dwelling with Secondary Unit	Р	Ρ	P	Р	P	17-102-360 17.103.080	
Two-Family Dwelling	C(L1)	P	P	P	Same as underlying zone		
Multifamily Dwelling		C(L1)	C(L1)	G <u>P(</u> L1)	Same as underlying zone		
Rooming House		-		1	<b>—</b>		
Mobile Home	-	<b> </b>	_	-			
Nonresidential Facilities			•				
Enclosed Nonresidential	Р	Р	Р	Р	Р		
Open Nonresidential	Р	Р	Р	Р	C.		
Sidewalk Cafe	P(L2)	P(L2)	P(L2)	P(L2)	P(L2)	17-102-335 17.103.090	
Drive-In Nonresidential		-	[		-		
Drive-Through Nonresidential	1-	-		-	-		
Telecommunications Facilities							
Micro Telecommunications	С	С	c	С	С	17.128	
Mini Telecommunications	c	С	C	С	С	17.128	
Macro Telecommunications	C	С	С	С	С	17.128	
Monopole Telecommunications	С	С	C	С	C	17.128	
Tower Telecommunications	-	]_		· ·	-	17.128	
Sign Facilities	•						
Residential Signs	P	Р	Р	Р	Р	17.104	
Special Signs	Р	Р	Р	Р	Р	17.104	
Development Signs	Р	Р	Р	Р	Р	17.104	
Realty Signs	Р	Р	P	P	Р	17.104	
Civic Signs	Р	Р	Р	Р	Р	17.104	
Business Signs	P(L3)	P(L3)	P(L3)	P(L3)	P(L4)	17.104	
Advertising Signs	_	1_	1—		_	17.104	

# Limitations on Table 17.17.02:

\* If a base zone (RM-1, RM-2, RM-3, or RM-4) also has the C combining zone, the C regulations supersede the base zone.

L1. See Table 17.17.03, Property Development Standards, for additional regulations on this conditionally permitted density.

1

L2. Sidewalk cafes are allowed only as an accessory facility to an already approved Full Service Restaurant or Limited Service Restaurant and Cafe. The sidewalk cafe may only operate within the hours of 7:00 a.m. to 10:00 p.m. No more than three (3) tables and no more than ten (10) chairs or seats are allowed. If more tables or chairs are requested, a Conditional Use Permit (CUP) is required (see Chapter 17.134 for the CUP procedure). See–17:102:335 Section 17:103:090 for other regulations regarding Sidewalk Cafes; however, the regulations in this section supersede any contradicting regulations in 17:102:335 Section 17:103:090.

17.17.050 - Property development standards.

A. Zone Specific Standards. Table 17.17.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "—" indicates that a standard is not required in the specified zone.

Development Standards	Zones				Additional
	RM-1	RM-2	RM-3	RM-4	Regulations
Minimum Lot Dimensions		<u></u>	<u> </u>	· · · · · · · · · · · · · · · · · · ·	
Width mean	45 ft	45 ft	25 ft ·	25 ft	1
Frontage	25 ft	25 ft	25 ft	25 ft	1
Lot area	5,000 sf	5,000 sf	4,000 sf	4,000 sf	1
Maximum Density					
Permitted density	1 primary unit per lot	1 unit on lots less than 4,000 sf; 2 units on lots 4,000 sf or greater	1 unit on lots less than 4,000 sf; 2 units on lots 4,000 sf or greater	1 unit on lots less than 4,000 sf; For 1 — 4 units, 1 unit per 1,100 sf of lot area; only on lots 4,000 sf or greater	2
Conditionally permitted density (only for lots 4,000 sf or greater)	2 units	For 3 or more units, 1 unit per 2,500 sf of lot area	For 3 or more units, 1 unit per 1,500 sf of lot area	For 5 or more units, 1 unit per 1,100 sf of lot area;	2, 3
Minimum Setbacks					
Minimum front (<20% street-to-setback gradient)	20 ft	20 ft	15 ft	15 ft	<u>4, 5</u> 4
Minimum front (>20% street-to-setback gradient)	5 ft	5 ft	5 ft	5 ft	<u>4, 5</u> 4, <u>6</u> 5
Minimum interior side	5 ft	5 ft	4 ft	4 ft	<u>7</u> 6, <u>8</u> 7, <u>9</u> 8
Minimum street side	5 ft	5 ft	4 ft	4 ft	<u>4, 7</u> 6, <u>8</u> 7, 9 <u>10</u>
Rear	15 ft	15 ft	15 ft	15 ft	110
Side and Rear Setbacks for Smaller Lots	See Table	17.17.04 for setbacks f	or smaller jots		
Floor Area Ratio (FAR) and Lot Coverage for 1 or 2 Units	See Table	17.17.05 for FAR and r	naximum lot coVerage	for 1 or 2 units	
Maxirnum Lot Coverage for 3 or More Units	N/A	40%	50%	N/A	
Height Regulations for Al	Lots with a	Footprint Slope of <	20%		
Maximum wall height primary building	25 ft	25 ft	30 ft	35 ft	1 <u>2</u> 4, 1 <u>3</u> 2, 1 <u>4</u> 3
Maximum pitched roof	30 ft	30 ft	30 ft	35 ft	1 <u>2</u> 4, 1 <u>3</u> 2,

Table 17.17.03: Property Development Standards

height primary building					1 <u>4</u> 3
Maximum height for accessory structures	15 ft	15 ft	15 ft	15 ft	. 1 <u>2</u> 1
Height Regulations for all Lots With a Footprint Slope of > 20%	See Tab	le 17.17.06 for He	ight regulations for all	ots with a footprint slope of >	20%
Maximum Walt Length Before Articulation Required	40 ft	40 ft	40 ft	40 ft	1 <u>5</u> 4
Minimum Parking					
Minimum parking spaces required per unit	1.5	1.5	1	1.	1 <u>6</u> 6, 46 <u>17</u>
Additional parking spaces required for secondary unit	1	1	1	1	1 <u>6</u> 5, 47 <u>18</u>
Minimum Parking Spaces for Nonresidential Activities	See Cha	pter <b>1</b> 7.116 for au	tomobile parking and C	Chapter 17.117 for bicycle pa	rking
Minimum Open Space			•		····
Group open space per unit	300 sf	300 sf	200 sf	175 sf	48 <u>19</u>
Group open space per unit when priVate open space substituted	100 sf	100 sf	85 sf	70 sf	48 <u>19</u>
Courtyard Regulations	See Sec	ion 17.108.120			

#### Additional Regulations for Table 17.17.03:

2. See Chapter 17.107 for affordable and senior housing incentives. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section-17-102-360\_17.103.080. Also applicable are the provisions of Section 17.102.270 with respect to additional kitchens for a dwelling unit, and the provisions of Section 17.102.300 with respect to dwelling units with five or more bedrooms.

3. A Conditional Use Permit (CUP) for density may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the CUP procedure in Chapter 17.134. In addition, the CUP in the RM-1 and RM-2 Zones may only be granted upon determination that the proposal conforms to the following additional use permit criteria. In the RM-3 Zone, this CUP may only be granted upon determination that the proposal conforms to the proposal conforms to the the proposal conforms to the following additional use permit criteria.

a. That the proposed development will not adversely affect adjoining property, nor the surrounding neighborhood, with consideration to be given to density; to the availability of neighborhood facilities and play space to the generation of traffic and the capacity of surrounding streets; and to all other similar, relevant factors;

b. That the site design and landscaping and the scale, height, length and width, bulk, coverage, and exterior treatment of structures are in harmony with neighborhood character and with facilities on nearby lots;

c. That the shape and siting of the facilities are such as to minimize blocking of views and direct sunlight from nearby lots and from other Residential Facilities in the surrounding neighborhood;

d. That the design and site planning of the buildings, open areas, parking and service areas, and other facilities provide a convenient, attractive, and functional living environment; and that paths, stairways, accessways, and corridors are designed to minimize privacy impacts;

e. That lot shape, size, and dimensions allow a development which will provide satisfactory internal living conditions without adversely affecting the privacy, safety, or residential amenity of adjacent residences.

5. On lots with only residential facilities, paved surfaces within required street-fronting vards, and any unimproved rights-of-way of adjacent streets, shall be limited to fifty percent (50%) on interior lots and thirty percent (30%) on corner lots. Exceptions: The maximum percentages of paved surfaces specified in this additional regulation may be exceeded within unimproved rights-of-way in the following cases upon issuance of a private construction of public improvements (P-iob) permit or if undertaken directly by the City or by a private contractor under contract to the City:

a Roadway construction or widening;

b. Sidewalk construction or widening; and

c. Any work pursuant to an approved final map, parcel map or final development plan pursuant to a planned unit development permit.

For purposes of this additional regulation, an unimproved right-of-way is the portion of a street or alley right-of-way that is not paved.

**56.** In all residential zones the minimum front setback depth otherwise required by the applicable individual zone regulations shall be reduced to five (5) feet on any lot with a street-to-setback gradient that exceeds twenty (20)-percent (20%), provided, however, that the distance from the edge of the pavement to a garage or carport elevation containing one or more vehicular entries shall be at least eighteen (18) feet (see Illustration for Table 17.17.03 [Additional Regulation 5], below). See Section 17.108.130 for allowed projections into setbacks.

**67.** No front or side setbacks are required for commercial facilities in the C combining zone except wherever an interior side lot line of any lot located in the C combining zone abuts an interior side lot line of any lot that is not located in an C combining zone or commercial zone, there shall be provided on the former lot, along the abutting portion of its side lot line, a side yard with a minimum width of five (5) feet (Where it abuts a rear lot line, no yard is required). Section 17.108.080 still applies. Also, see Section 17.108.130 for allowed projections into setbacks.

78. See additional reduced side, and rear setbacks for smaller lots, Table 17.17.04. See Section 17.108.130 for allowed projections into setbacks.

89. See Section 17.108.080 for the required interior side and rear setbacks on a tot containing two or more living units and opposite a legally-required living room window.

**910.** In all residential zones, on every corner lot which abuts to the rear a key lot which is in a residential zone, there shall be provided on the street side of such corner lot a side setback with a minimum width equal to one-half ( $\frac{1}{2}$ ) of the minimum front setback depth required on the key lot and no less than the minimum side setback width required along an interior side lot line of the corner lot. However, such side setback shall not be required to exceed five (5) feet in width if it would reduce to less than twenty-five (25) feet the buildable width of any corner lot. Such setback shall be provided unobstructed except for the accessory structures or the other facilities allowed therein by Section 17.108.130. This does not apply to lots within the C combining zone (see illustration for Table 17.17.03 [Additional Regulation 9], below). See also Section 17.110.040(C) for special controls on location of detached accessory buildings on such corner lots.

40<u>11</u>. Wherever a rear lot line abuts an alley, one-half of the right-of-way width of the alley may be counted toward the required minimum rear setback; provided, however, that the portion of the minimum

rear setback depth actually on the lot itself shall not be reduced to less than ten (10) feet. Also, see Section 17.108.130 for allowed projections into setbacks.

124. See Section 17.108.030 for allowed projections above height limits and Section 17.108.020 for increased height limits for civic buildings.

132. In the RM-1 and RM-2 Zones if at least sixty (60)-percent (60%) of the buildings in the immediate context are no more than one story in height, the maximum wall height shall be fifteen (15) feet within the front twelve (12) feet of buildable area. The immediate context shall consist of the five closest lots on each side of the project site plus the ten closest lots on the opposite side of the street; however, the Director of City Planning may make an alternative determination of immediate context based on specific site conditions. Such determination shall be in writing and included as part of any approval of any variance, conditional use permit, design review, determination of exemption from design review, or other special zoning approval or, if no special zoning approval is required, part of any Planning Department approval of a building permit application.

14.3 In the RM-2 Zone, the maximum pitched roof height may be increased to thirty-five (35) feet and maximum wall height may increase to thirty (30) feet upon the granting of a Conditional Use Permit (CUP) (see Chapter 17.134 for the CUP procedure). An increased wall height shall only be permitted in conjunction with a project with a pitched roof (a "pitched roof," as used in this section, is defined as having a vertical to horizontal ratio of a minimum of four in twelve (4:12) slope). In addition to the criteria contained in <u>Section</u> 17.136.050, any proposed increase in roof height must also meet the following use permit criteria:

' a. The additional pitched roof height is required to accommodate a roof form that is consistent with the historic context in the neighborhood; and

b. In conjunction with an increased pitched roof height, the additional wall height is required to accommodate a wall height that is consistent with the historic context in the neighborhood.

15.4 If the total wall length within ten (10) feet of the side lot line exceeds forty (40) feet, then the building wall shall be articulated by at least one section of additional setback. See design guidelines for more specific bulk regulations and context standards.

166. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117. Also, additional parking standards apply within the S-11 and S-12 Zones, as prescribed in <u>Chapters Section-17.92</u> and Section-17.94.

16<u>7.</u> In the RM-2 Zone when the lot is less than 4,000 square feet in size or forty-five (45) feet in width only one (1) parking space is required per unit

187. One parking space for the Secondary Unit is required in addition to any required parking spaces for the Primary Unit Additional regulations that apply to Secondary Units are provided in Section 17-102-360 17 103.080.

198. Each square foot of private usable open space equals two (2) square feet towards the total usable open space requirement, except that actual group space shall be provided in the minimum amount specified in the table per dwelling unit. All usable open space shall meet the standards contained in Chapter 17.126.

B. Setbacks for Smaller Lots. Table 17.17.04 below prescribes reduced setback standards for lots less than <u>four thousand (4,000)</u> square feet. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table.

Additional Regulations for Table 17.17.04:

1. See Section 17.108.130 for allowed projections into setbacks.

C: Floor Area Ratio (FAR) and Lot Coverage for One and Two-Family Dwelling Units Only. Table 17.17.05 below prescribes FAR and lot coverage standards associated with lot sizes. The numbers in the "Additional Regulations" column refer to the regulations listed at the end of the Table.

Table 17.17.05 Floor Area Ratio (FAR) and Lot Coverage Regulations for One and Two-Family Dwelling Units Only

Regulation	Lot Size	Lot Size in Square Feet							
	< 5,000	> 5,000 and < 12,000	> 12,000 and < 25,000	> 25,000 and < 43,560	> 43,560	Regulations			
Maximum FAR for Lots with a Footprint Slope >20%	0.55	0.50	0.45	0.30	0.20	1, 2			
Maximum Lot Coverage (%)	40%	40%	30%	20%	15%	2			

## Additional Regulations for Table 17.17.05:

1. Floor Area Ratio (FAR) only applies to lots that have a footprint slope of greater than twenty (20) percent (20%). Lots less than five thousand (5,000) square feet may have a dwelling with a minimum of two thousand (2,000) square feet of floor area, regardless of FAR listed.

**3.** Lots less than <u>five thousand (5,000)</u> square feet may have a lot coverage of up to <u>two thousand</u> (2,000) square feet regardless of lot coverage percentage (%) listed.

D. Height Table 17.17.06 below prescribes height standards associated with different sloped lots. The numbers in the "Additional Regulations" column refer to the regulations listed at the end of the Table.

# Table 17.17.06 Height Regulations for all Lots With a Footprint Slope of >20%

Regulation		ot Height Reg rint Slope of:	ulations	Upslope Lot Height Regulations With a Footprint Slope of:	Additional Regulations	
	> 20% and < 40%	> 40% and < 60%	> 60%	> 20%		
Maximum Height for Detached Accessory Structures	15 ft	15 ft	15 ft	15 ft	1	
Maximum Wall Height Primary Building	32 ft	34 ft	36 ft	32 ft	1, 2	
Maximum Wall Height Primary Building with a CUP	36 ft	38 ft	40 ft	35 ft	1	
Maximum Pitched Roof Height Primary Building	36 ft	38 ft	40 ft	35 ft	1, 2	
Maximum Height Above Edge of Payment	18 ft	18 ft	18 ft	N/A ·	1	
Maximum Height Above the Ground Elevation at the Rear Setback Line	N/A	N/A	N/A	24 ft	1	
Maximum Height from Finished or Existing Grade (whichever is lower) Within 20' of the Front Property Line	N/A	N/A	N/A	24 ft	1, 3	

#### Additional Regulations for Table 17.17.06:

2. On a downslope lot greater than forty (40)-percent (40%) footprint slope, the rear wall of an attached garage or carport may exceed the wall height and roof height by five (5) feet, but may not exceed

eighteen (18) feet above ground elevation at edge of pavement, if the garage or carport conforms with all of the following criteria:

a. Maximum width is twenty-two (22) feet and maximum depth is twenty (20) feet; and

b. Garage or carport floor is at the same level as the edge of the street pavement resulting from the project at the center point of the driveway entrance or is at a lower level; and

c. Maximum height above the garage or carport floor is ten (10) feet for walls to the top of the plate or flat roof, and twelve (12) feet for pitched roofs (see Illustration for Table 17.17.06 [Additional Regulation 2], below).

17.17.060 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mIni-lot developments, certain regulations that otherwise apply to individual lots in the RM zones may be waived or modified when and as prescribed in–Section 17.102-320 Chapter 17.142.

17.17.070 - Other zoning provisions.

- A. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- B. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- E. Landscaping and Screening Standards. <u>The landscaping and screening regulations set forth in</u> <u>Chapter 17.124 shall apply in</u> Tho-regulations-set-forth-in-Chapter-17-124-and-Chapter-17-102-400, scrooning-of-utility-meters, otc.-shall-apply-in the RM zones.
- F. Buffering. All uses shall be subject to the applicable requirements of the buffering regulations in Chapter 17.110 with respect to screening or location of parking, loading, storage areas, control of artificial illumination, and other matters specified therein.

17.19.030 - Permitted and conditionally permitted activities.

Table 17.19.01 lists the permitted, conditionally permitted, and prohibited activities in the RU zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

Activities	Zones					Additional
	RU-1	RU-2	RU-3	RU-4	RU-5	Regulations
Residential Activities						· · · · · · · · · · · · · · · · · · ·
Permanent	P(L1)	P(L1)	P(L1)	P(L1)	P(L1)	17.102.212 17.103.010
Residential Care	PG(LI)	€ <u>P</u> (L1)	G <u>P</u> (L1)	GP(1_1)	CP(LI)	17-102-212 17.103.010
Service-Enriched Permanent Housing	C(L1)	C(L1)	C(L1)	C(L1)	C(L1)	17-102-21-2 17.103.010
Transitional Housing	C(L1)	C(L1)	C(L1)	C(L1) .	C(L1)	17-102-212 17.103.010
Emergency Shelter	—	C(L1)	C(L1)	C(L1)	C(L1)	17-102-212 17.103.010
Semi-Transient	<u> </u>	-		C(L1)	C(L1)	
Bed and Breakfast	С	С	С	C	C	17.10.125
Civic Activities						
Essential Service	P	Р	P	P	Р	
Limited Child-Care Activities	P	Р	Р	P	Р	
Community Assembly	С	С	С	С	C .	
Recreational Assembly	Р	р	P	P	Р	1
Community Education	С	С	c	P(L2)	P(L2)	
Nonassembly Cultural	P(L2)	P(L2)	P(L2)	P(L2)	P(L2)	
Administrative	С	С	С	P(L2)(L3)	P(L2)(L4)	
Health Care	С	С	C .	P(L2)(L3)	P(L2)(L4)	
Special Health Care	<u> -</u>	-	<u> _</u>			
Utility and Vehicular	С	С	C	c	c	
Extensive İmpact	С	С	c	C	c	
Commercial ActiVities						
General Food Sales	C(L5)(L6)	C(L5)(L6)	C(L5)(L6)	P(L2)(L3)	P(L2)(L4)	
Full Service Restaurants	C(L5)(L6)	C(L5)(L6)	C(L5)(L6)	P(L2)(L3)	P(L2)(L4)	
Limited Service Restaurant and Cafe	C(L5)(L6)	C(L5)(L6)	C(L5)(L6)	P(L2)(L3)	P(L2)(L4)	

Table 17.19.01: Permitted and Conditionally Permitted Activities

Fast-Food Restaurant	—	_	-	-	C(L4)	17-102 <del>-210</del> 17. <u>103.030</u>
Convenience Market	_	-	-	_	C(L4)	47-102-2-10 17.103.030
Alcoholic Beverage Sales	—(L7)	(L7)	(L7)	C(L3)	C(L4)	4-7-1 <del>02-210</del> <u>17.103.030</u> and 1-7-102-046 <u>17.114.030</u>
Mechanical or Electronic Games	<b> </b>	–	-	<u> </u>	C(L4)	17.102.210
Medical Service	P(L6)(L8)	P(L6)(L8)	P(L6)(L8)	P(L2)(L3)	P(L2)(L4)	
General Retail Sales	P(L6)(L8)	P(L6)(L8)	P(L6)(L8)	P(L2)(L3)	P(L2)(L4)	
Large-Scale Combined Retail and Grocery Sales	–	-	-		-	
Consumer Service	C(L5)(L6)( <u>L9)</u>	P(L6)(L8) (L9)	P(L6)(L8) <u>(L9)</u>	P(L2)(L3) (L9)	P(L2)(L4) (L9)	
Consultative and Financial Service	P(L6)(L8)	P(L6)(L8)	P(L6)(L8)	P(L2)(L3)	P(L2)(L4)	
Check Cashier and Check Cashing		-	<u> </u>	<u> </u>	1—	
Consumer Cleaning and Repair Service	P(L6)(L8)	P(L6)(L8)	P(L6)(L8)	P(L2)(L3)	P(L2)(L4)	
Consumer Dry Cleaning Plant	]	_	–	C(L3)	C(L4)	
Group Assembly			C(L5)(L6)(L10)	C(L3)(L10)	C(L4)(L10)	
Personal Instruction and Improvement Services	P(L6)(L8)	P(L6)(L8)	P(L6)(L8)	P(L2)(L3)	P(L2)(L4)	
Administrative	P(L6)(L9 <u>L11</u> )	P(L6)(L9 <u>L11</u> )	P(L6)(1=9 <u>L11</u> )	P(L2)(L3)	P(L2)(L4)	
Business, Communication, and Media Services	P(L6)(L8)	P(L6)(L8)	P(L6)(L8)	P(L2)(L3)	P(L2)(L4)	
Broadcasting and Recording Services Commercial Activities	-	—	— .	P(L2)(L3)	P(L2)(L4)	
Research Service	[	-	—	P(L2)(L3)	P(L2)(L4)	
General Wholesale Sales	_	-	—		·	
Transient Habitation	-	—	_		[ <u>`</u>	
Wholesaie-and-Prefessioaat-Building Material Sales	_	<b>_</b> ·	—	—	_	
Automobile and Other Light Vehicle Sales and Rental		—	_	_	-	
Automobile and Other Light Vehicle Gas Station and Servicing	<b>–</b>	-	<b>—</b>	–		
Automobile and Other Light Vehicle Repair and Cleaning	-	<b>—</b> ·	-	_	[—	
Taxi and Light Fleet-Based Services			—	-	<b>—</b>	
Automotive Fee Parking	-			<b> </b>		•
Animal Boarding		<b>—</b>	[	<u> </u>		
Animal Care		—	<u> </u>	<u> </u>	—	
Undertaking Service		-			[ <b>_</b>	
Industrial Activities (all)		—		-		
Agriculture and Extractive Activities						
Crop and animal raising	C(L10L12)	C(L-10L12)	C(L10 <u>L12</u> )	C(L-10 <u>L12</u> )	C(L10 <u>L12</u> )	
			······································			

5

Plant nursery	C		C	С	c	С	
Mining and Quarrying	<b> </b> -	``	]_				
Accessory off-street parking serving prohibited activities	С		С	. c	с	с	17.102.100 17.116.075
Activities that are listed as prohibited, but are permitted or conditionally permitted on nearby lots in an adjacent zone. Additional aetivities-that-are-permitted-or oortditionally-permitted in-an-adjacent zone,-on-lots-near-the-boundary thereof	C .		C	c	c	c ~	17.102.110

Limitations on Table 17.19.01:

L1. Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when not located in a One-Family Dwelling Residential Facility. No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such activity. See Section-17-102-212 17.103.010 for other regulations regarding these activities. Also, Rosidential Cere is only-permitted-upon-the-granting-of-a-Conditional-Use-Permit-(see-Chaptor-17-134-for-the-CUP procedure) when-not-losated in a One-Family Dwelling Residential Facility.

**L8.** These activities may only be located in an existing ground floor of nonresidential facility that was both built prior to the effective date of this chapter (April 14, 2011) and not originally used for a Civic Activity. For the purposes of this limitation, a facility is considered built if it received its certificate of occupancy or passed its final building inspection on its building permit. These activities may only operate within the hours of 7:00 a.m. and 10:00 p.m.; a Conditional Use Permit (CUP) is required if the facility exceeds three thousand (3,000) square feet (see Chapter 17.134 for the CUP procedure).

L9. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five-hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L10. Adult entertainment activities are prohibited.

**L119.** These activities may only be located in an existing ground floor of nonresidential facility that was both built prior to the effective date of this chapter (April 14, 2011) and not originally used for a Civic Activity. These activities may only operate within the hours of 7:00 a.m. and 10:00 p.m.; a Conditional Use Permit is required if the facility exceeds <u>one thousand five hundred (1,500)</u> square feet (see Chapter 17.134 for the CUP procedure).

L120. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the criteria contained in Section 17.134.050, this activity must meet the following use permit criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.19.040 - Permitted and conditionally permitted facilities.

Table 17.19.02 lists the permitted, conditionally permitted, and prohibited facilities in the RU zones. The descriptions of these facilities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates facilities subject to certain limitations listed at the bottom of the table.

"---" designates facilities that are prohibited.

Table 17.19.02: Permitted and Conditionally Permitted Facilities

Facilities	Zones		•			Additional	
	RU-1	RU-2	RU-3	RU-4	RU-5	Regulations	
Residential Facilities		•					
One-Family Dwelling	Р	Р	Р	-(L1)	-(L1)		
One-Family Dwelling with Secondary Unit	Р	Р	Р	—(L1)	-(L1)	17-102-360 <u>17.103.080</u>	
Two-Family Dwelling	Р	Р	Р	Ρ.	Р		
Multifamily Dwelling	Р	Р	Р	Р	Р		
Rooming House		С	С	Р	Р		
Mobile Home				_	_		
Nonresidential Facilities							
Enclosed Nonresidential	Р	Р	Р	P(L2)	P(L3)		
Open Nonresidential	P	Р	Р	Р	P		
Sjdewalk Cafe	P(L4)	P(L4)	P(L4)	Р	Р	47-102-335 17.103.090	
Drive-In Nonresidential		<b> </b>	<u> </u> _	-	-		
Drive-Through Nonresidential	·			-	-		
Telecommunications Facilities					,		
Mjcro Telecommunications	C	С	C	С	C	17.128	
Mini Telecommunications	С	С	С	С	С	17.128	
Macro Telecommunications	с	С	c	С	c	17.128	
Monopole Telecommunications	С	С	c	С	c	17.128	
Tower Telecommunications						17.128	
Sign Facilities							
Residential Signs	Р	Р	Р	Р	Р	17.104	
Special Signs	Р	Р	Р	Р	Р	17.104	
Development Signs	Р	Р	Р	Р	Р	17.104	
Realty Signs	Р	Р	Р	Р	Р	17.104	
Civic Signs	P	Р	Р	Р	P	17.104	
Business Signs	P(L5)	P(L5)	P(L5)	Р	Р	17.104	
Advertising Signs						17.104	

#### Limitations on Table 17.19.02:

L4. Sidewalk cafes are allowed only as an accessory facility to an already approved Full Service Restaurant or Limited Service Restaurant and Cafe. The sidewalk cafe may only operate within the hours of 7:00 a.m. to 10:00 p.m. No more than three (3) tables and no more than ten (10) chairs or seats are allowed. If more tables or chairs are requested, a Conditional Use Permit (CUP) is required (see Chapter 17.134 for the CUP procedure). See–17–102-335 Section 17.103.090 for other regulations regarding Sidewalk Cafes; however, the regulations in this section supersede any contradicting regulations in 17-102-335 Section 17.103.090.

L5. Business Signs are only allowed on existing nonresidential facilities built prior to the effective date of this chapter (April 14, 2011); otherwise Chapter 17.104 applies. For the purposes of this limitation, a facility is considered built if it received its certificate of occupancy or passed its final building inspection on its building permit. The maximum aggregate area of display surface of all business, civic, and residential signs on any one lot shall be 0.5 square foot for each one foot of lot frontage in the case of an interior lot, or 0.25 square feet for each one foot of lot frontage in the case of a corner lot. The aggregate shall include only one face of a double-faced sign. The total amount of aggregate sign area shall not exceed <u>one hundred (100)</u> square feet on any one property. See Chapter 17.104 for other regulations regarding Business Signs; however, the regulations in this section supersede any contradicting regulations in Chapter 17.104.

17.19.050 - Property development standards.

A. Zone Specific Standards. Table 17.19.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "—" indicates that a standard is not required in the specified zone.

Development Standards	Zones	Zones						
	RU-1	RU-2	RU-3	RU-4	RU-5	Regulations		
Minimum Lot Dimensions	1	·.				,		
Width mean	25 ft	25 ft	25 ft	25 ft	25 ft	1		
Frontage	25 ft	25 ft	25 ft	25 ft	25 ft	1		
Lot area	4,000 sf	4,000 sf	4,000 sf	4,000 sf	4,000 sf	1		
Maximum Density		•						
Pennitted density for regular dwelling units	1 unit per 1,100 sf	1 unit per 800 sf	1 unit per 450 sf	See Table 17.19.04	See Table 17.19.04	2		
Permitted density for rooming units	N/A	1 unit per 800 sf	1 unit per 450 sf	See Table 17.19.04	See Table 17.19.04			
Minimum Setbacks					·····			
Minimum front (<20% street-to-setback gradient) for residential facilities	15 ft	10 ft	10 ft	5 ft	O ft	3, <u>4, 5</u> 4, <u>6</u> 5		
Minimum front (>20% street-to-setback gradjent) for residentjal facilities	5 ft	5 ft	5 ft	5 ft	5 ft	3, <u>4, 5</u> 4, <u>6</u> 5		
Minimum front for commercial facilities	15 ft	10 ft	10 ft ~	0 ft	0 ft	3		
Minimum interior side	4 ft	4 ft	0 ft	0 ft	0 ft	3, <u>7</u> 6, <u>8</u> 7		
Minimum street side	4 ft	4 ft	4 ft	0 ft	0 ft	3, <u>4, 7</u> 6, <u>9</u> 8		
Rear (Residential Facilițies)	15 ft	15 ft	15 ft	10/15 ft	10/15 ft	3, <u>7</u> 6, <u>10</u> 9, 1 <u>1</u> 0		

Table 17.19.03: Property Development Standards

Rear (Nonresidential Facilities)	15 ft	15 ft	15 ft	0/10/15 ft 🢡	0/10/15 ft	3, 9 <u>10, 11</u> 40
Setbacks for Smaller Lots < 3,000 sf or < 38	ft Wide			······		
Minimum interior side	3 ft	3 ft	0 ft	NA	NA	3, 67
Minimum street side	3 ft	3 ft	3 ft	NA	NA	3 <u>. 4</u> , <u>8</u> 7
Height Regulations					<u> </u>	-
Minimum height of ground ftoor nonresidential facilities	0 ft	0 ft	0 ft	12 ft	12 ft	1 <u>2</u> 4
Minimum separation between the grade and ground floor living space	0 ft	0 ft	0 ft	2.5 ft	2.5 ft	1 <u>3</u> 2
Maximum height primary building	40 ft	50 ft	60 ft	See Table 17.19.04	See Table 17.19.04	13 <u>4</u> , 1 <u>5</u> 4 ;
Maximum height for accessory structures	15 ft	15 ft	15 ft	See Table 17.19.04	See Table 17.19.04	
Parking Requirements						
Minimum Parking Spaces Required per Regular Residential Unit	1	1	1	1	1	1 <u>6</u> 5
Additional Parking Spaces Required for Secondary Unit	1	1	1	1	1	1 <u>6</u> 5, 1 <u>7</u> 9
Parking and driveway location requirements	No	No	No	Yes	Yes	17 <u>8</u>
Minimum Parking Spaces for Nonresidential Activities	See Chap bicycle pa		r automobile	parking and Cha	apter 17.117 for	
Minimum Usable Open Space						· · · · · · · · · · · · · · · · · · ·
Group usable open space per regular unit	175 sf	175 sf	150 sf	See Table 17.19.04	See Table 17.19.04	<u>19</u> 48
Group usable open space per regular unit when private open space is substituted	50 sf	30 sf	30 sf	See Table 17.19.04	See Table 17.19.04	<u>19</u> 18
Group usable open space per rooming unit	85 sf	85 sf	75 sf	See Table 17.19.04	See Table 17.19.04	<u>19</u> 48
Group usable open space per rooming unit when private open space substituted	15 sf	15 sf	15 sf	See Table 17.19.04	See Table 17.19.04	1 <u>9</u> 8
Courtyard Regulations	See Sect	ion 17.108.12	20			

## Additional Regulations for Table 17.19.03:

1. See Sections 17.106.010 and 17.106.020 for exceptions to lot area, width mean and street frontage regulations.

2. See Chapter 17.107 for affordable and senior housing incentives. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section-17-102-360 17.103.080. Also applicable are the provisions of Section 17.102.270 with respect to additional kitchens for a dwelling unit, and the provisions of Section 17.102.300 with respect to dwelling units with five (5) or more bedrooms.

3. See Section 17.108.130 for allowed projections into setbacks.

4. On lots with only residential facilities, paved surfaces within required street-fronting vards, and any unimproved rights-of-way of adjacent streets, shall be limited to fifty percent (50%) on interior lots and thirty percent (30%) on corner lots. Exceptions: The maximum percentages of paved surfaces specified in this additional regulation may be exceeded within unimproved rights-of-way in the following cases upon

issuance of a private construction of public improvements (P-job) permit or if undertaken directly by the City or by a private contractor under contract to the City:

a Roadway construction or widening;

b. Sidewalk construction or widening; and

c. Any work pursuant to an approved final map, parcel map or final development plan pursuant to a planned unit development permit.

For purposes of this additional regulation, an unimproved right-of-way is the portion of a street or alley right-of-way that is not paved.

54. In the RU-1 zone, if adjacent lots abutting the side lot lines of the subject lot both contain principal Residential Facilities that have front setbacks with a depth of less than fifteen (15) feet, the minimum front setback may be reduced for buildings and other structures on the subject lot up to a line parallel to the front lot line and extended from the most forward projection of the principal Residential Facility on the adjacent lots having the deeper front setback depth, provided such projection is enclosed, has a wall height of at least eight (8) feet, and has a width of at least five (5) feet. In the case of a corner lot or lot that has a vacant parcel next to it, this same principal may apply if the two lots adjacent to the corner lot or lot along its front lot line have less than a fifteen (15) foot front setback (see Illustration for Table 17.19.03 [Additional Regulation 4], below). Also, see Section 17.108.130 for allowed projections into setbacks.

**<u>65.</u>** In all residential zones, the minimum front setback otherwise required by the applicable individual zone regulations shall be reduced to five (5) feet on any lot with a street-to-setback gradient that exceeds twenty (20)-percent (20%), provided, however, that the distance from the edge of the pavement to a garage or carport elevation containing one or more vehicular entries shall be at least eighteen (18) feet (see illustration for Table 17.19.03 [Additional Regulation 5], below). See Section 17.108.130 for allowed projections into setbacks.

**67**. See Section 17.108.080 for the required interior side and rear setbacks on a lot containing two or more living units and opposite a legally-required living room window.

87. Wherever an interior side lot line of any lot located in the RU-3, RU-4, or RU-5 zone abuts an interior side lot line of any lot located in an RH or RD zone, the setback of the abutting portion of its side lot line is ten (10) feet. In the case where an interior side lot line of any lot located in the RU-3, RU-4, or RU-5 zone abuts an interior side lot line in an RM zone, the setback of the abutting portion of its side lot line is five (5) feet. In the case where an interior side lot line in an RU-3, RU-4, or RU-5 zone abuts a side yard of an RU-1 or RU-2 lot, a side setback of four (4) feet is required (see Illustration for Table 17.19.03 [Additional Regulation 7] below).

<u>98.</u> When the rear yard of a reversed corner lot abuts a key lot that is in an RH, RD, or RM zone or the RU-1 zone, the required street side yard setback in the rear twenty (20) feet of the reversed corner lot is one-half of the minimum front yard required on the key lot (see Illustration for Table 17.19.03 [Additional Regulation 8], below)

8<u>10</u>. Wherever a rear lot line abuts an alley, one-half ( $\frac{1}{2}$ ) of the right-of-way width of the alley may be counted toward the required minimum rear setback; provided, however, that the portion of the minimum rear setback actually on the lot itself shall not be so reduced to less than ten (10) feet. Also, see Section 17.108.130 for allowed projections into setbacks.

<u>1140.</u> When a rear lot line in the RU-4 or RU-5 zone is adjacent to an RH, RD, or RM zone or the RU-1 zone, the required rear setback for both residential and nonresidential facilities is ten (10) feet if the lot depth is one-hundred (100) feet or less and fifteen (15) feet if the lot depth is more than one-hundred (100) feet. When a rear lot line of a lot in these zones is not adjacent to an RH, RD, or RM zone or the RU-1 zone, the required rear setback is ten (10) feet for residential facilities and there is no required setback for nonresidential facilities.
<u>1241</u>. This height is only required for new principal buildings and is measured from the sidewalk grade to the ground floor ceiling.

<u>13</u>42. This regulation only applies to new residential facilities and ground floor living space located within fifteen (15) feet of a street frontage.

<u>14</u>13. Buildings in the RU-1, RU-2, RU-3, RU-4, and RU-5 zones shall have a thirty (30) foot maximum height at the setback line associated with any rear or interior side lot line that abut a lot in an RU-1 zone or an RH, RD, or RM zone; this maximum height may increase one foot for every foot of distance from this setback line (see Illustration for Table 17.19.13 [Additional Regulation 11], below). Also, see Section 17.108.030 for allowed projections above height limits and Section 17.108.020 for increased height limits for civic buildings.

154. In the RU-2 and RU-3 zone, a building may only exceed forty (40) feet in height up to the maximum height if each portion above forty (40) feet is: Set back from the inner line of each of the minimum side setbacks, if any, required by Section 17.28.150(C)(1) a minimum horizontal distance equal to one (1) foot for each four (4) feet by which it extends above the height of forty (40) feet; and set back from the inner line of the minimum rear yard required by Section 17.28.150D a minimum horizontal distance equal distance equal to one (1) foot for each two (2) feet by which it extends above the height of forty (40) feet, provided, however, that such setback from the inner line of the minimum rear yard need not exceed forty (40) feet (see Illustration for Table 17.19.03 [Additional Regulation 14], below).

165. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117. Also, additional parking standards apply within the S-11 and S-12 zones, as prescribed in <u>Chapters</u> Section-17.92 and Soction-17.94.

176. One parking space for the Secondary Unit is required in addition to any required parking spaces for the Primary Unit. Additional regulations that apply to Secondary Units are provided in Section 17-102-360 17 103.080.

187. For the new construction of principal buildings in the RU-4 and RU-5 zones, access to parking and loading facilities through driveways, garage doors, or other means shall not be from the principal street when alternative access is feasible from another location such as a secondary frontage or an alley. Where this is not feasible, every reasonable effort shall be made to share means of vehicular access with abutting properties. Open parking areas shall not be located between the sidewalk and a principal building.

198. Each square foot of private usable open space equals two (2) square feet towards the total usable open space requirement, except that actual group space shall be provided in the minimum amount specified in the table per dwelling unit. All usable open space shall meet the standards contained in Chapter 17.126.

B. Height, Floor Area Ratio (FAR), Density, and Open Space for the RU-4 and RU-5 Zones Only. Table 17.19.04 below prescribes height, FAR, intensity, and open space standards associated with the Height Areas described in the Zoning Maps. The number designations in the "Additional Regulations" column refer to regulations below the table.

Table 17.19.04 Heiglit, Floor Area Ratio (FAR), Density, and Open Space Regulations for the RU-4 and RU-5 Zones Only

Regulation	Heigh	Height Area						
	35	45	60	75	90	120	Regulations	
Maximum Height	35 ft	45 ft	60 ft	75 ft	90 ft	90 ft	1, 2	
Height Minimum		•						
Pennitted height minimum	0 ft	0 ft	35 ft	35 ft	35 ft	35 ft	3	

Conditionally permitted height minimum	NA	NA	25 ft	25 ft	25 ft	25 ft	3
Maximum Residential Density (square feet of lot area required per unit)	550 sf	450 sf	375 sf	275 sf	225 sf	225 sf	4, 5
Maximum Nonresidential FAR	2.0	2.5	3.0	4.0	4.0	4.0	4, 5
Maximum Number of Stories (not including UndergroUnd construction)	3	4	5	7	8	11	
Minimum Usable Open Space		·			-	<u> </u>	· · · · ·
Group usable open space per regular unit	150 sf	150 sf	150 sf	150 sf	100 sf	100 sf	6
Group usable open space per regular unit when private open space substituted	30 sf	30 sf	30 sf	30 sf	20 sf	20 sf	6
Group usable open space per rooming unit	75 sf	75 sf	75 sf	75 sf	50 sf	50 sf	6
Group usable open space per rooming unit when private open space is substituted	15 sf	15 sf	15 sf	15 sf	I0 st	10 sf	6

## Additional Regulations for Table 17.19.04:

3. This minimum height requirement only applies to the new construction of a principal building that is located on parcels adjacent to a street right-of-way that is <u>one hundred (100)</u> feet wide or more. Buildings constructed to accommodate Essential Service, Utility and Vehicular, or Extensive Impact Civic Activities are exempted from the height minimum regulation. The allowed projections into the height limits contained in <u>Section</u> 17.108.030 are not counted towards the height minimum.

**4.** See Chapter 17.107 for affordable and senior housing incentives. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section-17-102-360\_17.103.080. Also applicable are the provisions of Section 17.102.270 with respect to additional kitchens for a dwelling unit, and the provisions of Section 17.102.300 with respect to dwelling units with five (5) or more bedrooms.

**5.** No portion of lot area used to meet the residential density requirements shall be used as a basis for computing the maximum nonresidential FAR unless the total nonresidential floor area on the lot is less than three thousand (3,000) square feet.

17.19.060 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain regulations that apply to individual lots in the RU zones may be waived or modified when and as prescribed in-Section-17-102-320 Chapter 17.142.

17.19.070 - Other zoning provisions.

- A. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- B. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- E. Landscaping and Screening Standards. <u>The landscaping and screening regulations set forth in Chapter 17.124 shall apply in the Tho-regulations-set-forth-in-Chapter 17.124 and Chapter 17.102-400-scrooning of utility-meters, etc., shall apply in the RU zones.</u>

17.30.060 - Conditionally permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A. Residential Activities:

Residential Care, except when occupying a One-Family Dwelling Residential Facility

Service-Enriched Permanent Housing

Transitional Housing

Emergency Shelter

B. Civic Activities:

Administrative

Health Care

Utility and Vehicular

Extensive Impact

C. Commercial Activities:

General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Convenience Market

Alcoholic Beverage Sales

Medical Service

Consumer Service (see Section 17.102.170 for special regulations relating to massage services and Section 17.102.450 for special regulations related to laundromats)

Consultative and Financial Service, but limited to the provision of advice, designs, information, or consultation of a professional nature

D. Agricultural and Extractive Activities:

Crop and Animal Raising

- E. Off-street parking serving activities other than those listed above or in Section 17.30.050, subject to the conditions set forth in Section-17-102-100\_17.116.075.
- F. <u>Activities that are listed neither as permitted nor conditionally permitted, but are permitted or conditionally permitted on nearby lots in an adjacent zone Additional activities which are permitted or conditionally permitted in an adjacent zone, on lots near tho boundary thereof, subject to the conditions set forth in Section 17.102.110.</u>

17.30.070 - Permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Facilities:

One-Family Dwelling

One-Family Dwelling with Secondary Unit, subject to the provisions specified in Section 17-102-360 17.103.080

Two-Family Dwelling

Multifamily Dwelling

Rooming House

B. Nonresidential Facilities:

Enclosed

Open

C. Signs:

Residential

Special

Development

Realty

Civic

17.30.090 - Special regulations applying to certain commercial activities.

All General Food Sales, Full Service Restaurant, Limited Service Restaurant and Cafe, Convenience Market, Alcoholic Beverage Sales, and Consumer Service Commercial Activities shall, except for offstreet parking and loading and maintenance of accessory landscaping and screening, be conducted entirely within enclosed portions of Multifamily Dwelling or Rooming House Residential Facilities, with customer access only through the lobby of such a facility. The maximum floor area devoted to such activities by any single establishment shall be <u>one thousand five hundred (1,500)</u> square feet No Business Signs or display windows shall be provided for such activities. See also Section-17-102-2-10 <u>17.103.030</u>.

17.30.110 - Use permit criteria for commercial activities.

A. General Food Sales, Full Service Restaurant, Limited Service Restaurant and Cafe, Convenience Market, Fast-Food Restaurant, Alcoholic Beverage Sales, or Consumer Service. A conditional use permit for General Food Sales, Full Service Restaurant, Limited Service Restaurant and Cafe, Convenience Market, Fast-Food Restaurant, Alcoholic Beverage Sales, or Consumer Service Commercial Activities may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, and that the proposed activities are intended primarily to serve residents of the facility within which the activities are to be located. See also Section-17-102-210 17.103.030.

#### 17.30.130 - Minimum lot area, width, and frontage.

Every lot shall have a minimum lot area of <u>four thousand (4,000)</u> square feet and a minimum lot width of <u>twenty-five (25)</u> feet, except as a lesser area or width is allowed by Section 17.106.010. Every lot shall have a minimum frontage of <u>twenty-five (25)</u> feet upon a street, except as this requirement is modified by Section 17.106.020.

#### 17.30.140 - Maximum residential density.

The maximum density of Residential Facilities shall be as set forth below, subject to the provisions of Section 17.106.030 with respect to maximum density on lots containing both Residential and Nonresidential Facilities. Also applicable are the provisions of Section 17.102.270 with respect to additional kitchens for a dwelling unit, and the provisions of Section 17.102.300 with respect to dwelling units with five or more bedrooms. No residential facility shall be permitted to have both an additional kitchen as provided for in Subsection 17.102.270B and a Secondary Unit

- Permitted Density, One regular dwelling unit is permitted for each three hundred (300) square A. feet of lot area, provided that one extra such unit is permitted if a remainder of 200 square feet or more is obtained after division of the lot area by three hundred (300) square feet. One efficiency dwelling unit is permitted for each two hundred (200) square feet of lot area. provided that one extra such unit is permitted if a remainder of one hundred fifty (150) square feet or more is obtained after division of the lot area by two hundred (200) square feet. One rooming unit is permitted for each one hundred fifty (150) square feet of lot area, provided that one extra such unit is permitted if a remainder of 100 square feet or more is obtained after division of the lot area by one hundred fifty (150) square feet. For a combination of different types of living units, the total required lot area shall be the sum of the above requirements for each. The number of living units permitted heretofore may be exceeded by ten percent on any corner lot, and may also be exceeded by ten percent on any lot which faces or abuts a public park at least as wide as the lot. A One-Family Dwelling or a One-Family Dwelling with Secondary Unit is permitted on any lot which gualifies under Section 17.106.010 as an existing buildable parcel and that contains no other dwelling units.
- B. Conditionally Permitted Density. The number of living units permitted by Subsection A. of this Section may be increased by not to exceed <u>fifty 50</u>-percent (50%) upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, in each of the following situations:
  - 1. In the case of a Residential Facility with more than four (4) stories containing living units, subject to the provisions of Section 17.106.040.
  - Upon the acquisition of development rights from nearby lots, subject to the provisions of Section 17.106.050.

The number of living units may also be increased, as prescribed in Section 17.106.060, in certain special housing.

17.30.150 - Maximum floor-area ratio.

The maximum floor-area ratio of any facility shall be as set forth below, subject to the provisions of Section 17.106.030 with respect to maximum floor-area ratio on lots containing both Residential and Nonresidential Facilities.

- A. Permitted Floor-Area Ratio (FAR). The maximum permitted floor-area ratio is 3.50, except that this ratio may be exceeded by ten percent (10%) on any corner lot and may also be exceeded by ten percent (10%) on any lot which faces or abuts a public park at least as wide as the lot
- B. Conditionally Permitted Floor-Area Ratio (FAR). The floor-area ratio permitted by Subsection A. of this Section may be increased by not to exceed <u>fifty 50</u>-percent (50%) upon the granting of a

conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, in each of the following situations:

- 1. In the case of a Residential Facility with more than four (4) stories containing living units, subject to the provisions of Section 17.106.040.
- 2. For any facility, upon the acquisition of development rights from nearby lots, subject to the provisions of Section 17.106.050.

17.30.170 - Minimum yards and courts.

The following minimum yards and courts shall be provided unobstructed except for the accessory structures or the other facilities allowed therein by Section 17.108.130:

- A. Front Yard. The minimum front yard depth on every lot shall be ten (10) feet. oxcopt-as-a lesser-depth is allowed-by-Section-17-108-050-on-steep-lots-
- B.—Side-Yard-Street-Sido-of-Corner-Lot.-A-side-yard-shall-bo-provided-on-the-street-side-of-a corner-lot-when and as-proscribed-in-Soction-17+108-060
- CB. Side Yard---Interior Lot Line. No side yard is generally required along an interior side lot line except as follows:
  - A side yard shall be provided, when and as prescribed in Section 17.108.080, opposite a living room window which faces an interior side lot line and which is located on a lot containing Residential Facilities with a total of two or more living units.
  - 2.----A-side-yard-shall-be-provided, as-proscribed-in-Section-17.108.090, along-an-interior-side lot-lino-lying-elong-a-boundary-of-any-of-certain-other-zones.
- DC. Rear Yard. The minimum rear yard depth on every lot shall be ten (10) feet., except-as-a-lessor depth-is-allowed-by-Section-17-108-140
- ED. Courts. On each lot containing Residential Facilities with a total of two or more living units, courts shall be provided when and as required by Section 17.108.120.

17.30.180 - Minimum usable open space.

On each lot containing Residential Facilities with a total of two or more living units, group usable open space shall be provided for such facilities in the minimum amount of <u>one hundred fiftv (150)</u> square feet per regular dwelling unit plus <u>one hundred (100)</u> square feet per efficiency dwelling unit plus <u>seventy-five (75)</u> square feet per rooming unit. Private usable open space may be substituted for such group space in the ratio prescribed in Section 17.126.020. All required space shall conform to the standards for required usable open space in Chapter 17.126.

17.30.200 - Special regulations for mini-lot developments, planned unit developments, and largescale developments.

- A. Mini-Lot Developments. In mini-lot developments, certain of the regulations otherwise applying to individual lots in the R-80 zone may be waived or modified when and as prescribed in-Soction 17-102-320 Chapter 17.142.
- C. Large-Scale Developments. No development which involves more than <u>one hundred thousand</u> (100,000) square feet of new floor area, or a new building or portion thereof of more than <u>one hundred twenty (120)</u> feet in height, shall be permitted except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134. This requirement shall not apply to developments where a valid planned unit development permit is in effect.

17.33.030 - Permitted and conditionally permitted activities.

Table 17.33.01 lists the permitted, conditionally permitted, and prohibited activities in the CN zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

Activities	Zones	Additional			
	CN-1	CN-2	CN-3	CN-4	Regulations
Residential Activities				· · · · · · · · · · · ·	
Permanent	P(L1)(L2)(L3)	P(L1)(L2)(L3)	<b>P</b> (L1)(L3)	P(L1)(L3)	1
Residential Care	P(L1)(L2)(L3)	P(L1)(L2)(L3)	P(L1)(L3)	P(L1)(L3)	17-102.212 17.103.010
Service-Enriched Pennanent Housing	C(L1)(L3)(L4)	C(L1)(L3)(L4)	C(L1)(L3)(L4)	C(L1)(L3).	17-102-212 17.103.010
Transitional Housing	C(L1)(L3)(L4)	C(L1)(L3)(L4)	C(L1)(L3)(L4)	C(L1)(L3)	. 17.102.212 17.103.010
Emergency Shelter	C(L1)(L3)(L4)	C(L1)(L3)(L4)	C(L1)(L3)(L4)	C(L1)(L3)	17.102.212 17,103.010
Semi-Transient	. –		-		
Bed and Breakfast	C(L3)(L4)	C(L3)(L4)	C(L3)(L4)	C(L3)	17-10-125
Civic Activities					
Essential Sen/ice	Р	Р	Р	Р	
Limited Child-Care Activities	P(L2)	P(L2)	P(L5)	P(L5)	
Community Assembly	C(L4)	C(L4)	C(L4)	С	
Recreational Assembly	P(L2)	P(L2)	P(L5)	P(L5)	
Community Education	C(L4)	C(L4)	C(L4)	С	
Nonassembly Cultural	P(L5)	P(L5)	P(L5)	P(L5)	
Administrative	P(L2)	P(L2)	P(L5)	P(L5)	
Health Care	C(L4)	C(L4)	C(L4)	C	
Special Health Care	C(L4) <u>(L6)</u>	C(L4) <u>(L6)</u>	C(L4) <u>(L6</u> )	C( <u>L6)</u>	17-102-390 17.103.020
Utility and Vehicular	C(L4)	C(L4)	C(L4)	С	
Extensive Impact	C(L4)	C(L4)	C(L4) .	С	
Commercial Activities					
General Food Sales	P(L5)	P(L5)	P(L7)	P(L7)	
Full Service Restaurants	C(L4)	P(L5)	P(L5)	P(L5)	
Limited Service Restaurant and Cafe	C(L4)	P(L5)	P(L5)	P(L5)	· .
Fast-Food Restaurant	C(L4)	C(L4)	C(L4)	С	17-102-210

Table 17.33.01: Permitted and Conditionally Permitted Activities

······································					<u>17.103.030</u> and 8.09
Convenience Market	C(L4)	C(L4)	C(L4)	С	17.102.210 17.103.030
Alcoholic Beverage Sales	C(L4)	C(L4)	C(L4)	c	1 <del>7.102.210</del> 17.103.030 and 17.102-040 17.114.030
Mechanical or Electronic Games	C(L4)	C(L4)	C(L4)	С	17-102-2-10
Medical Service	P(L2)	P(L6 <u>L7</u> )	P(L <u>7</u> 6)	P(L5)	· · · · · · · · · · · · · · · · · · ·
General Retail Sales	P(L5)	P(L5)	P(L5L8)	P(L5 <u>L8</u> )	j
Large-Scale Combined Retail and Grocery Sales	_	<u> </u>	-	-	
Consumer Service	P(L5)( <u>L9</u> )	P(L5)(L9)	P(L5) <u>(L9)</u>	P(L5)( <u>L9)</u>	· · ·
Consultative and Financial Service	P(L2)	P(£8 <u>L10</u> )	P(L5)	P(L5)	
Check Cashier and Check Cashing	<u> </u>	<u> </u>	_	-	1
Consumer Cleaning and Repair Service	P(L5)	P(L5)	P(L5)	P(L5)	<u> </u>
Consumer Dry Cleaning Plant	C(L4)	C(L4)	C(L4)	с	
Group Assembly	C(L4)(L11)	C(L4)(L11)	C(L4)(L11)	C(L11)	<u></u>
Personal Instruction and Improvement Services	P(L2)	P(L5)	P(L5)	P(L5)	
Administrative	P(L2)	P(L2)	P(L5)	P(L5)	
Business, Communication, and Media Services	P(L2)	P(L2)	P(L5)	P(L5)	
Broadcasting and Recording Services	P(L2)	P(L2)	P(L5)	P(L5)	
Research Service	C(L4) .	C(L4)	P(L5)	P(L5)	
General Wholesale Sales	-	-	_	_	
Transient Habitation	<u> </u>	<u> </u>	<b> _</b>	<u> </u>	
Wholesale-and-Professional-Building Material Sales		_	_	-	- -
Automobile and Other Light Vehicle Sales and Rental	_	-	-		
Automobile and Other Light Vehicle Gas Station and Servicing	-	—	—	С	
Automobile and Other Light Vehicle Repair and Cleaning	_	—	_	_ ·	
Taxi and Light Fleet-Based Services	<b>—</b>	<u> </u>	-	-	
Automotive Fee Parking	C(L4)	C(L4)	C(L4)	С	
Animal Boarding	-	1_		-	[
Animal Care	C(L4)	C(L4)	P(L5)	P(L5)	
Undertaking Service	-	_	-	-	
Industrial Activities					
Custom Manufacturing	C(L4)(L9 <u>12</u> )	C(L4)(LL129)	C(L <u>12</u> 4)	С	17,102.040
Light Manufacturing	<b>—</b>		<u> _</u>	Τ	
General Manufacturing	-	_		-	
Heavy/High Impact	-		-	1-	1
Research and Development	1_	<u> </u> _	<b> </b> _	<b> </b> _	

۰.

.

1

3

	T	T	•	1	1
Construction Operations	<u> -</u>	<del></del>	<del></del>		]
Warehousing, Storage, and Distribution		<u> </u>			
A. General Warehousing, Storage and Distribution		·	-	-	
B. General Outdoor Storage					
C. Self- or Mini-Storage	-				
D. Container Storage	-	· ·	-		
E. Salvage/Junk Yards	-	-	-		
Regional Freight Transportation	-		-		
Trucking and Truck-Related		· ·		<u> </u>	
Recycling and Waste-Related					
A. Satellite Recycling Collection Centers	-		-	-	
B. Primary Recycling Collection Centers	]		—		
Hazardous Materials Production, Storage, and Waste Management	-		- '	-	
Agriculture and Extractive Activities	·	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	·
Crop and animal raising	C(L-I-0 <u>L13</u> )	C(L10L13)	C(L10L13)	C(L10L13)	:
Plant nursery	C(L4)	C(L4)	C(L4)	с	
Mining and Quarrying					
Accessory off-street parking serving prohibited activities	C(L4)	C(L4)	C(L4)	С	17-102-100 17.116.075
Activities that are listed as prohibited, but are oermitted or conditionally permitted on nearby lots in an adjacent zone. Additional activities that are permitted or conditionally permittod in an adjacent zone, on lots near the boundary-thereof	C(L4)	C(L4)	C(L4)	с	17.102.110

Limitations on Table 17.33.01:

L1. <u>Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when not located in a One-Family Dwelling Residential Facility.</u> No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such activity. See Section-17-102-212\_17.103.010 for other regulations regarding these activities.

L6. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2,500) feet from any other such activity or five hundred (500) feet from any K-12 school or Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity. See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

**L6**<u>L7</u>. A Medical Service Commercial Activity that occupies more than thirty-five (35) feet of frontage facing the principal street is not permitted except upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). All window space facing the principal street shall be clear, non-reflective, and allow views into the indoor space.

**L87.** The total floor area devoted to these activities on the ground floor by any single establishment may only exceed fifteen\_-thousand (15,000) square feet upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in Section 17.134.050, these conditionally permitted ground floor activities must also meet the criteria contained in L4, above.

L9. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L108. With the exception of retail bank branches, these activities are only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when located on the ground floor of a building. Incidental pedestrian entrances that lead to one of these activities elsewhere in the building are exempted from this Conditional Use Permit requirement. In addition to the CUP criteria contained in Section 17.134.050, these conditionally permitted ground floor activities must also meet the criteria contained in L4, above. The size limitation described in L5, above, shall apply to retail bank branches.

L11. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

L129. Not permitted on the gGround fEloor-oxcept-when-associated-with-a-fuil-sorvice-restaurant-or-retail store.

L1<u>3</u>0. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in <u>Section</u> 17.134.050, this activity must meet the following use permit criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.33.040 - Permitted and conditionally permitted facilities.

Table 17.33.02 lists the permitted, conditionally permitted, and prohibited facilities in the CN zones. The descriptions of these facilities are contained in Chapter 17.10.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates facilities subject to certain limitations listed at the bottom of the table.

"-" designates facilities that are prohibited.

Table 17.33.02: Permitted and Conditionally Permitted Facilities

Facilities	Zones		Additional		
	CN-1	CN-2	CN-3	CN-4	Regulations
Residential Facilities					<u> </u>
One-Family Dwelling	(L1)	(L1)	(L1)	(L1)	
One-Family Dwelling with Secondary Unit	(L1)	(L1)	(L1)	(L1)	17-102-360 17-103.080
Two-Family Dwelling	P(L2)	P(L2)	P(L3)	Р	
Multifamily Dwelling	P(L2)	P(L2)	P(L3)	Р	
Rooming House	P(L2)	P(L2)	P(L3)	P	

Mobile Home	-	-	-	-	
Nonresidential Facilities	<u> </u>				
Enclosed Nonresidential	P	Р	Р	Р	
Open Nonresidential	C(L4)	C(L4)	C(L4)	C(L4)	
Sidewalk Cafe	P	Р	Р	P	47.102-335 17.103.090
Drive-In	<u> </u>		-	С	
Drive-Through	<u> </u>	1	1	C(L5)	17.103.100
Telecommunications Facilities			۰		
Micro Telecommunications	P(L <u>6</u> 5)	P(L <u>6</u> 5)	P(L <u>6</u> 5)	P(L <u>6</u> 5)	17.128
Mini Telecommunications	P(L <u>6</u> 5)	P(L <u>6</u> 5)	P(L <u>6</u> 5)	P(L <u>6</u> \$)	17.128
Macro Telecommunications	c	C ·	С	С	17.128
Monopole Telecommunications	с	c	С	C	17.128
Tower Telecommunications			-	Ţ <u></u>	17.128
Sign Faci ities	<u> </u>		```` <u>`````````````````````````````````</u>		
Residential Signs	P	Р	Р	Р	17.104
Special Signs	P ·	P	Р	Р	17.104 .
Development Signs	P	Р	Р	Р	17.104
Realty Signs	P	Р	Р	Р	17.104
CNic Sighs	Ρ.	Р	Р	Р	17. t04
Business Signs	- P	Р	Ρ.	Р	17.104
Advertising Signs		<u> </u>	<u> </u>	<u> </u>	17.104

Limitations on Table 17.33.02:

L4. No conditional use permit is required for <u>Open Nonresidential Facilities to accommodate either</u> seasonal sales and <u>or</u> special event activities.

L5. No new or expanded Fast-Food Restaurants with Drive-Through Nonresidential Facilities shall be located closer than five hundred (500) feet of an elementary school, park, or playground. See Sections 17.103.030 and 17.103.100 for further regulations regarding Drive-Through Nonresidential Facilities.

L65. See Section 17.128.025 for restrictions on Telecommunication Facilities near residential or HBX zones.

17.33.050 - Property development standards.

A. Zone Specific Standards. Table 17.33.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "---" indicates that a standard is not required in the specified zone.

Development Standards	Zones	Zones					
	CN-1	CN-2	CN-3	CN-4	Regulations		
Minimum Lot Dimensions							
Width Mean	25 ft	25 ft	25 ft	25 ft	1		
Frontage	25 ft	25 ft	25 ft	25 ft	1		

### Table 17.33.03: Property Development Standards

Lot area	4,000 sf	4,000 sf	4;000 sf	4,000 sf	1
Minimum/Maximum Setbacks	•	· · · · · · · · · · · · ·			
Minjmum front	0 ft ·	0 ft	0 ft	0 ft	2
Maximum front	10 ft	10 ft	10 ft	None	3
Minimum interior side	0 ft	0 ft	O ft	0 ft	4, 5
Minimum street side	0 ft	O ft	Oft ,	0 ft	6
Rear (Residentjal Facilities)	10/15 ft	10/15 ft	10/15 ft	10/15 ft	7, 8
Rear (Nonresidential Facilities)	0/10/15 ft	0/10/15 ft	0/10/15 ft	0/10/15 ft	8
Design Regulations	da				
Minimum ground floor nonresidential facade transparency	65%	65%	65%	None	9
Minimum height of ground floor nonresidential facilities	12 ft	12 ft	12 ft	12 ft	10
Minimum separation between the grade and ground floor living space		_	2.5 ft	2.5 ft	11
Parking and driveway location requirements	Yes	Yes	Yes	No	12
Ground floor active space requirement	Yes	Yes	Yes	No	13
Height, Floor Area Ratio, Density, and Open Space Regulations	See Table 17.33.04				
Minimum required parking	See Chapter 17.116 for automobile parking and Chapter 17.117 for bicycle parking				
Courtyard regulations	See Section	17.108.120			

Additional Regulations for Table 17.33.03:

1. See Sections 17.106.010 and 17.106.020 for exceptions to lot area, width mean, and street frontage regulations.

2. If fifty (50)-percent (50%) or more of the frontage on one side of the street between two intersecting streets is in any residential zone and all or part of the remaining frontage is in any commercial or industrial zone, the required front setback of the commercially or industrially zoned lots is one-half of the minimum front setback required in the residential zone. If <u>fifty</u> 50-percent (50%) or more of the total frontage is in more than one residential zone, then the minimum front setback on the commercially or industrially zoned lots is one-half of that required in the residential zone with the lesser front setback (see Illustration for Table 17.33.03 [Additional Regulation 2]). Also, see Section 17.108.130 for allowed projections into setbacks.

Illustration for Table 17.33.03 [Additional Regulation 2] \*for illustration purposes only

3. The following notes apply to the maximum front yard requirement:

a. The requirements only apply to the construction of new principal buildings.

b. The requirements do not apply to lots containing Recreational Assembly, Community Education, Utility and Vehicular, or Extensive Impact Civic Activities or Automobile and Other Light Vehicle Gas Station and Servicing Commercial Activities as principal activities.

c. Maximum yards apply to seventy-five (75)-percent (75%) of the street frontage on the principal street and fifty (50)-percent (50%) on other streets, if any. All percentages, however, may be reduced to fifty (50)-percent (50%) upon the granting of regular design review approval (see Chapter 17.136 for the design review procedure). In addition to the CUP criteria contained in Section 17.136.035, the proposal to reduce to fifty (50)-percent (50%) must also meet each of the following criteria:

i. The additional yard area abutting the principal street is designed to accommodate publicly accessible plazas, cafes, or restaurants;

1i. The proposal will not impair a generally continuous wall of building facades;

iii. The proposal will not weaken the concentration and continuity of retail facilities at groundlevel, and will not impair the retention or creation of an important shopping frontage; and

iv. The proposal will not interfere with the movement of people along an important pedestrian street.

9. This percentage of transparency is only required for principal buildings that include ground floor nonresidential facilities and only apply to the facade facing the principal street. The regulations only apply to facades located within twenty (20) feet of a street frontage. The area of required transparency is between two (2) feet and nine (9) feet in height of the ground floor and must be comprised of clear, nonreflective windows that allow views out of indoor commercial space, residential space, or lobbies (see Illustration for Table 17.33.03 [Additional Regulation 9], below). Areas required for garage doors shall not be included in the calculation of facade area (see Note 12 for limitations on the location of parking access). Glass block does not qualify as a transparent window. Exceptions to this regulation may be allowed by the Planning Director for unique facilities such as convention centers, gymnasiums, parks, gas stations, theaters and other similar facilities.



Facade facing Principal Street

3. This minimum height requirement only applies to the new construction of a principal building that is located on parcels adjacent to a street right-of-way that is <u>one hundred (100)</u> feet wide or more. Buildings constructed to accommodate Essential Service, Utility and Vehicular, or Extensive Impact Civic Activities or Automobile and Other Light Vehicle Gas Station and Servicing Commercial Activities may be exempted from the height minimum regulation by the Planning Director. The allowed projections into the height limits contained in <u>Section</u> 17.108.030 are not counted towards the height minimum.

**4.** See Chapter 17.107 for affordable and senior housing incentives. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section <u>17.102.360</u> <u>17.103.080</u>. Also applicable are the provisions of Section 17.102.270 with respect to additional kitchens for a dwelling

unit, and the provisions of Section 17.102.300 with respect to dwelling units with five (5) or more bedrooms.

5. No portion of lot area used to meet the residential density requirements shall be used as a basis for computing the maximum nonresidential <u>Floor Area Ratio (FAR)</u> unless the total nonresidential floor area on the lot is less than <u>three thousand (3,000)</u> square feet.

17.33.060 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain regulations that apply to individual lots in the CN zones may be waived or modified when and as prescribed in-Section-17-102-320 Chapter 17.142.

17.33.070 - Other zoning provisions.

- A. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- B. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- E. Landscaping and Screening Standards. <u>The landscaping and screening regulations set forth in</u> <u>Chapter 17.124 shall apply in the</u>The\_regulations\_sot\_forth\_in\_Chapter\_17-124\_and\_Chaptor 17-102-100, screening-of-utility-motors, etc., shall apply in the CN zones.

ſ

17.35.030 - Permitted and conditionally permitted activities.

Table 17.35.01 lists the permitted, conditionally permitted, and prohibited activities in the CC zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

ł

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

Activities	Zones			Additional
	CC-1	CC-2 ·	CC-3	Regulations
Residential Activitjes			<u> </u>	· · · · · · · · · · · · · · · · · · ·
Pennanent	_ P(L1)(L2)(L3)	P(L1)(L2)(L3)	C(L1)(L3)	
Residential Care	P(L1)(L2)(L3)	P(L1)(L2)(L3)	C(L1)(L3)	47-102-212 17.103.010
Service-Enriched Permanent Housing	C(L1)(L3)	C(L1)(L3)	C(L1)(L3)	47-102.212 17.103.010
Transitional Housing	C(L1)(L3)	C(L1)(L3)	C(L1)(L3)	47-102-212 17.103.010
Emergency Shelter	C(L1)(L3)	C(L1)(L3)	C(L1)(L3)	47-102-212 17.103.010
Semi-Transient				
Bed and Breakfast	C	С	С	17-10-125
Civic Activities	· · · · · · ·			
Essential Service	P	Р	Р	
Limited Child-Care Activities	P	Р	С	-
Community Assembly	c	С	С	
Recreational Assembly	P	Р	Р	
Community Education	P	P	P	
Nonassembly Cultural	P	Р	Р	
Admjnistrative	P	Р	Р	
Health Care	P	Р	Р	
Special Health Care	C <u>(L4)</u>	C( <u>L4)</u>	C <u>(L4)</u>	47.102.390 17.103.020
Utility and Vehjcular	C	С	С	
Extensive Impact	C	С	С	
Commercial Activities				
General Food Sales	P	Р	Р	
Full Service Restaurants	Р	Р	Р	
Limited SerVice Restaurant and Cafe	P	Р	Р	
Fast-Food Restaurant	C	С	С	17.102.210 <u>17.103.030</u> and 8.09
Convenience Market	C	C	С	17,102-210 17,103.03
Alcoholic Beverage Sales	c	c	С	1-7-102-210_17.103.03( and 17-102-040

Table 17.35.01: Permitted and Conditionally Permitted Activities

				17.114.030
Mechanical or Electronic Games	С	С	c	47-102-210
Medical Service	Р	P	P	
General Retail Sales	P	P	·P	
Large-Scale Combined Retail and Grocery Sales	-	_		
Consumer Service	P <u>(L<del>5</del>)</u>	P <u>(L5)</u>	P(L <u>5</u> )	
Consultative and Financial Service	P	P	Р	
Check Cashier and Check Cashing	C <u>(L6)</u>	C(L6)	C <u>(L6)</u>	47.102.430 17.103.040
Consumer Cleaning and Repair Service	Р	Ρ.	Р	
Consumer Dry Cleaning Plant	Р	С	Р	
Group Assembly	C <u>(L7)</u> .	C(L7)	C( <u>L7</u> )	
Personal Instruction and Improvement Services	Р	P	Р	· ·
Administrative	P '	Р	Р	
Business, Communication, and Media Services	Р	P	P	
Broadcasting and Recording Services	P	P	P	
Research Service	Р	P	P	
General Wholesale Sales	7		P	· · · · · · · · · · · ·
Transient Habitation	-	-	<u> </u>	
Wholesale-ond-Professional-Building Material Sales	<b> </b> _	—	Р	·····
Automobile and Other Light Vehicle Sales and Rental	C ·	P(L4L8)	P	
Automobile and Other Light Vehicle Gas Station and Servicing	P	С	Р	•
Automobile and Other Light Vehicle Repair and Cleaning	C(L5 <u>L9</u> )	C( <del>L5<u>L9</u>)</del>	Р	
Taxi and Light Fleet-Based Services	1		C	
Automotive Fee Parking	С	С	C	
Animal Boarding	С	C	C	
Animal Care	Ρ.	P	P ·	-
Undertaking Service	-	<u> </u>	-	
Industrial Activities				<u></u>
Custom Manufacturing	С	С	Р	47-102.040
Light Manufacturing	_	С	P(16L10)	17-102-040
General Manufacturing	_		_	
Heavy/High Impact	<b> </b> _		· _	
Research and Development	_	_	_	
Construction Operations				
Warehousing, Storage, and Distribution			```	
A. General Warehousing, Storage and Distribution			P	[
B. General Outdoor Storage	-	<u> </u>		
C. Self-or Mini Storage	·			
D. Container Storage			<u> </u>	-
E. Salvage/Junk Yards			_	
Regional Freight Transportation				·.

Oakland, California, Code of Ordinances .

Page 73

.

.

Trucking and Truck-Related	-	<u> </u>	-	
Recycling and Waste-Related	· · · · · · · · · · · · · · · · · · ·			·
A. Satellite Recycling Collection Centers	-	-		17-10.040
B. Primary Recycling Collection Centers	<b>—</b>	_	_	<u> </u>
Hazardous Materials Production, Storage, and Waste Management	-	-	-	
AgricUlture and Extractive Activities				
Crop and animal raising	C( <u>L7L11</u> )	C( <del>L7<u>L11</u>)</del>	C(t-7L11)	
Plant nursery	С.,	С	С	
Mining and Quarrying	<u> </u>			· · · ·
Accessory off-street parking serving prohibited activities	С	С	С	17.102.100116.075
Activities that are listed as orohibited, but are oermitted or conditionally permitted on nearby lots in an adjacent zoneAd ditional-activities-that-are-permitted-or conditionally-permitted-in-an-adjacent-zone-on-lots-near the-boUndary-thereof	С	C	C .	17.102.110

Limitations on Table 17.35.01:

L1. <u>Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when not located in a One-Family Dwelling Residential Facility.</u> No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such activity. See Section-17-102-212\_17.103.010 for other regulations regarding these activities.

L4. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2,500) feet from any other such activity or five hundred (500) feet from any K-12 school or Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity. See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

L5. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L6. No new or expanded Check Cashier and Check Cashing Commercial Activity shall be located closer than one thousand (1,000) feet from any other such activity or five hundred (500) feet from any Community Education, Community Assembly, or Recreational Assembly Civic Activity; state or federally chartered bank, savings association, credit union, or industrial loan company; or certain Alcoholic Beverage Sales Commercial Activities. See Section 17.103.040 for further regulations regarding Check Cashier and Check Cashing Commercial Activities.

L7. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

<u>L8</u>. Automobile and Other Light Vehicle Sales and Rental is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when located on Telegraph Avenue between 28<sup>th</sup> Street and I-580.

**L5L9.** This Conditional Use Permit may only be granted upon determination that the proposal conforms to the general use permit criteria set forth in the Conditional Use Permit procedure in Chapter 17.134 and to the following additional use permit criteria:

1. That all repair and servicing is performed in an enclosed building;

**2.** That a minimum six (6) foot tail masonry or decorative screening wall is provided at all parcels lines adjacent to an RH-, RD- or RM-zone;

3. That a landscape buffering is at all parcels lines adjacent to an RH-, RD- or RM-zone; and

4. That no auto repair activities shall be conducted before 7:00 a.m. or after 9:00 p.m. on any day of the week.;

**L6L10.** This activity is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when located on a lot that is within <u>three hundred (300)</u> feet of an RH, RD, or RM zone.

t=7<u>L11</u>. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in Section 17.134.050, this activity must meet the following use permit criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.35.040 - Permitted and conditionally permitted facilities.

Table 17.35.02 lists the permitted, conditionally permitted, and prohibited facilities in the CC zones. The descriptions of these facilities are contained in Chapter 17.10.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17 134 for the CUP procedure).

"L" designates facilities subject to certain limitations listed at the bottom of the Table.

"-" designates facilities that are prohibited.

Table 17.35.02: Permitted and Conditionally Permitted Facilities

Facilities	Zones		Additional		
	CC-1	CC-2	CC-3	Regulations	
Residentja  Facilities			<u>_</u>		
One-Family Dwelling	. (L1)	(L1)	-(L1)		
One-Family Dwelling with Secondary Unit	(L1)	-(L1)	(L1)	17.102.360 17.103.080	
Two-Family Dwelling	P(L2)	P(L3)	_		
Multifamily Dwelling	P(L2)	P(L3)	-		
Rooming House	P(L2)	P(L3)			
Mobile Home			<b>—</b>		
Nonresidential Facilities		,			
Enclosed Nonresidential	P	P	P		
Open Nonresidential	Р	P	P		
Sidewalk Cafe	P	P	P	17-102-335 <u>17.103.090</u>	

Drive-In	С	C	С	
Drive-Through	C( <u>L4)</u>	C(L41	C(L4)	
TelecommUnications Facilities	•			
Mjcro Telecommunjcations	P(L4L5)	P(L4L5)	P(L4 <u>L5</u> )	17.128
Mini Telecommunications	P(L <u>5</u> 4)	P(L <u>5</u> 4)	P(L <u>5</u> 4)	17.128
Macro Telecommunications	С	С	С	17.128
Monopole Telecommunications	c	C	c	17.128
Tower Telecommunications			<u> </u>	17.128
Sign Facilities				
Residential Signs	. P	P	P	17.104
Special Signs	P	P	P	17.104
Development Signs	P	Р	P	17.104
Realty Signs	P	P	P	17.104
Civic Signs	P	P	P .	17.104
Business Signs	P	P	Р	17.104
Advertising Signs			-	17.104

## Limitations on Table 17.35.02:

L4. No new or expanded Fast-Food Restaurants with Drive-Through Nonresidential Facilities shall be located closer than five hundred (500) feet of an elementary school, park, or playground. See Sections 17.103.030 and 17.103.100 for further regulations regarding Drive-Through Nonresidential Facilities.

L54. See Section 17.128.025 for restrictions on Telecommunication Facilities near residential or HBX zones.

# 17.35.050 - Property development standards.

A. Zone Specific Standards. Table 17.35.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "—" indicates that a standard is not required in the specified zone.

Development Standards	Zones	Zones				
, ,	CC-1 CC-2		CC-3	Regulations		
Minimum Lot Dimensions		;				
Width mean	50 ft	25 ft	25 ft	1		
Frontage	50 ft	25 ft	25 ft	1		
Lot area	7,500 sf	4,000 sf	4,000 sf	1 '		
Minimum/MaximUm Setbacks			<u> </u>			
Minimum front	0 ft	0 ft	Oft	2		
Maximum front	N/A	10 ft	N/A	3		
Minimum interior side	0 ft	. Oft	Oft .	4, 5		
Minimum street side	0 ft	0 ft	Oft	6		

Table 17.35.03: Property Development Standards

Oakland, California, Code of Ordinances

Page 76

Rear (Residential Facilities)	10/15 ft	10/15 ft	10/15 ft	7, 8
Rear (Nonresidential Facilities)	0/10/ <b>1</b> 5 ft	0/10/15 ft	0/10/15 ft	8
Design Regulations			<u> </u>	
Minimum ground floor nonresidential facade transparency	N/A	55%	N/A	9
Minimum height of ground floor nonresidential facilities	N/A	12 ft	N/A	10
Parking and driveway location requirements	No	Yes	No	11
Ground floor active space requirement	No	Yes No		12
Height, Floor Area Ratio, Density, and Open Space Regulations	See Table 17.35.04			
Minimum Required Parking	See Chapter 17.116 for automobile parking and Chapter 17.117 for bicycle parking			
Courtyard Regulations	See Section 17.108.120			

## Additional Regulations for Table 17.35.03:

**2.** If fifty (50)-percent (50%) or more of the frontage on one side of the street between two intersecting streets is in any residential zone and all or part of the remaining frontage is in any commercial or industrial zone, the required front setback of the commercially or industrially zoned lots is one-half of the minimum front setback required in the residential zone. If <u>fifty 50</u>-percent (50%) or more of the total frontage is in more than one residential zone, then the minimum front setback on the commercially or industrially zoned lots is one-half of that required in the residential zone with the lesser front setback (see Illustration for Table 17.35.03 [Additional Regulation 2], below).

3. The following notes apply to the maximum front yard requirement:

a. The requirements only apply to the construction of new principal buildings.

b. The requirements do not apply to lots containing the following principal activities: Recreational Assembly, Community Education, Utility and Vehicular, or Extensive Impact Civic Activities or Automobile and Other Light Vehicle Gas Station and Servicing Commercial Activities as principal activities.

c. Maximum yards apply to seventy-five (75)-percent (75%) of the street frontage on the principal street and fifty (50)-percent (50%) on other streets, if any. All percentages, however, may be reduced to fifty (50)-percent (50%) upon the granting of Regular Design Review approval (see Chapter 17.136 for the design review procedure). In addition to the CUP criteria contained in Section 17.136.035, the proposal to reduce to fifty (50)-percent (50%) must also meet each of the following criteria:

**i.** The additional yard area abutting the principal street is designed to accommodate publicly accessible plazas, cafes, or restaurants;

ii. The proposal will not impair a generally continuous wall of building facades;

iii. The proposal will not weaken the concentration and continuity of retail facilities at groundlevel, and will not impair the retention or creation of an important shopping frontage; and

iv. The proposal will not interfere with the movement of people along an important pedestrian street.

8. When a rear lot line is adjacent to an RH, RD, RM, or RU-1 zone, the required rear setback for both residential and nonresidential facilities is ten (10) feet if the lot depth is one-hundred (100) feet or less and fifteen (15) feet if the lot depth is more than one\_-hundred (100) feet. When a rear lot line is not adjacent to an RH, RD, RM, or RU-1 zone, the required rear setback is ten (10) feet for residential facilities and there is no required setback for nonresidential facilities.

12. For the new construction of principal buildings in the CC-2 zone, ground level parking spaces, locker areas, mechanical rooms, and other non-active spaces shall not be located within <u>thirty (30)</u> feet from the front of the principal building except for incidental entrances to such activities elsewhere in the building. Exceptions to this regulation may be permitted by the Planning Director for utilities and trash enclosures that cannot be feasibly placed in other locations of the building. Driveways, garage entrances, or other access to parking and loading facilities may be located on the ground floor of this area as regulated by Note 11, above.

C. Height, Floor Area Ratio (FAR), Density, and Open Space. Table 17.35.04 below prescribes height, FAR, density, and open space standards associated with the Height Areas described in the Zoning Maps. The number designations in the "Additional Regulations" column refer to regulations below the table.

Regulation	Heig	ht Are	a					Additional
	35	45	60	75	90	120	160	Regulations
Maximum Height	35 ft	45 ft	60 ft	75 ft	90 ft	120 ft	160 ft	1. 2
Height Minimum			<u> </u>					
Pennjtted height minimum	0 ft	0 ft	35 ft	35 ft	35 ft	35 ft	35 ft	3
Conditionally pemiitted height minimum	NA	NA	25 ft	25 ft	25 ft	25 ft	25 ft	3
Maximum Residential Density (square feet of lot area required per dw	elling	Unit)						
Regular units	550	450	375	275	225	225	225	4, 5
Rooming units	275	225	185	135	110	<u>110</u> 225	<u>110</u> 225	4, 5
Maximum Nonresidential FAR	2.0	2.5	3.0	4.0	4.5	5.0	5.0	4, 5
MaximUm nUmber of stories (not including Underground construction)	3	4	5	7	8	11	15	
Minimum Usable Open Space			,	<u></u>			-	
Group usable open space per regular unit	150	150	150	150	100	100	100	6
Group usable open space per regular unit when private open space substituted	30	30	30	30	20	20	20	6
Group usable open space per rooming unit	75	75	75	75	50	50	50	6
Group usable open space per rooming unit When private open space is substituted	15	15	15	15	10	10	10	6

Table 17.35.04 Height, Floor Area Ratio (FAR), Density, and Open Space Regulations

Additional Regulations for Table 17.35.04:

3. This minimum height requirement only applies to the new construction of a principal building that is located on parcels adjacent to a street right-of-way that is <u>one hundred (100)</u> feet wide or more. Buildings in the CC-1 zone and buildings constructed to accommodate Essential Service, Utility and Vehicular, or Extensive Impact Civic Activities or Automobile and Automobile and Other Light Vehicle Sales and

Rental, Automobile and Other Light Vehicle Gas Station and Servicing or Automobile and Other Light Vehicle Repair and Cleaning Commercial Activities may be exempted from the height minimum regulation by the Planning Director. The allowed projections into the height limits contained in Section 17.108.030 are not counted towards the height minimum.

4. See Chapter 17.107 for affordable and senior housing incentives. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section <u>47-102-360 17.103.080</u>. Also applicable are the provisions of Section <u>17.102.270</u> with respect to additional kitchens for a dwelling unit, and the provisions of Section <u>17.102.300</u> with respect to dwelling units with five <u>(5)</u> or more bedrooms.

**5.** No portion of lot area used to meet the residential density requirements shall be used as a basis for computing the maximum nonresidential FAR unless the total nonresidential floor area on the lot is less than <u>three thousand (3,000)</u> square feet.

17.35,060 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain regulations that apply to individual lots in the CC zones may be waived or modified when and as prescribed in-Section-17-102-320 <u>Chapter</u> <u>17.142.</u>

17.35,070 - Other zoning provisions.

- A. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- B. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- D. Recycling Space Allocation Requirements. The regulations'set forth in Chapter 17.118 shall apply in the CN zones.
- E. Landscaping and Screening Standards. <u>The landscaping and screening regulations set forth in</u> <u>Chapter 17.124 shall apply in the</u>Tho-rogulations-sot-forth in-Chapter-17-124-and-Chapter 47-102-400-screening-of-utility-meters, otc., shall apply in the CC zones.

17.37.030 - Permitted and conditionally permitted activities.

Table 17.37.01 lists the permitted, conditionally permitted, and prohibited activities in the CR-1 zone. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates activities subject to certain limitations or notes listed at the bottom of the table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

Activities Zone Additional Regulations **CR-1 Residential Activities** Permanent **Residential Care** \_\_\_ Service-Enriched Permanent Housing 17.102.212 17.103.010 Transitional Housing C(L1) 17-102-212 17.103.010 Emergency Shelter C(L1) Semi-Transient C(L1) Bed and Breakfast 17.10.125 **Civic Activities** Ρ Essential Service Ρ Limited Child-Care Activities Р Community Assembly Ρ Recreational Assembly С Community Education Р Nonassembly Cultural Р Administrative С Health Care Special Health Care C(L2) 17.102.390 17.103.020 Ç Utility and Vehicular С Extensive Impact Commercfat Activities General Food Sales Ρ Р Full Service Restaurants P Limited Service Restaurant and Cafe с. Fast-Food Restaurant 17.102.210 17.103.030 and 8.09 С 17.102.210 17.103.030 Convenience Market С 17-102-210 17.103.030 Alcoholic Beverage Sales and 17, 102,040

Table 17.37.01: Permitted and Conditionally Permitted Activities.

		17.114.030
Mechanical or Electronic Games	C	17.103.210
Medical Service	P	
General Retail Sales	· P	
Large-Scale Combined Retail and Grocery Sales		
Consumer Service	P <u>(L3)</u>	
Consultative and Financial Service	P	
Check Cashier and Check Cashing	<u> </u>	
Consumer Cleaning and Repair Service	P(I <u>-2L4</u> )	
Consumer Dry Cleaning Plant	c	
Group Assembly	C <u>(L5)</u>	
Personal Instruction and Improvement Services	P	
Administrative	P	
Business, Communication, and Media Services	P	
Broadcasting and Recording Senvices	· P	ĺ
Research Service	P	
General Wholesale Sales	c	
Transient Habitation	С	47.102.370 17.103.050
Whotesale-and-Professional-Building Material Sales	P( <u>L2L4</u> )	
Automobile and Other Light Vehicle Sales and Rental	• P(L2 <u>L4</u> )	· · ·
Automobile and Other Light Vehicle Gas Station and Servicing	P(L2 <u>L4</u> )	
Automobile and Other Light Vehicle Repair and Cleaning	P(L2 <u>L4</u> )	
Taxi and Light Fleet-Based Services	P(L3 <u>L4</u> )	
Automotive Fee Parking	—(L3 <u>L6</u> )	
Animal Boarding	С	
Animal Care	P	
Undertaking Service		
Industrial Activities		
Custom Manufacturing	P(L2L4)	17.102.040
Light Manufacturing	P(L2L4)	17.102.040
General Manufacturing	C( <u>L-2L4</u> )	<del>i7.102.</del> 040
Heavy/High Impact		
Research and Development	P	
Construction Operations		
Warehousing, Storage, and Distribution	<u></u>	
A. General Warehousing, Storage and Distribution	P(L2L4)	-
B. General Outdoor Storage	C(L2 <u>L4</u> )	
C. Self-or Mini Storage	C(L2 <u>L4</u> )	
D. Container Storage	C(L2L4)	
E. Salvage/Junk Yards	–	
Regional Freight Transportation	C(L2 <u>L4</u> )	
Trucking and Truck-Related	C(L2L4)	

Recycling and Waste-Related		
A. Satellite Recycling Collection Centers	C(L2L4)	17-10-040
AB. Primary Recycling Collection Centers	C(L2 <u>L4</u> )	17.103.060
Hazardous Materials Production, Storage, and Waste Management		<u>, , , , , , , , , , , , , , , , , , , </u>
Agriculture and Extractive Activities	<u> </u>	·
Crop and animal raising	C(L2L4)(L4L7)	
Plant nursery	C(1_2_4)	<u></u>
Mining and Quarrying	-	<u></u>
Accessory off-street parking serving prohibited activities	P	17-102-100 17.116.075
Activities that are listed as prohibited but are committed or conditionally permitted on nearby lots in an adjacent zoneAdditional-activities-that-are-permitted-or eonditionally-permitted-in-an-adjacent-zono, on lots-noar-tho-boUndary-thereof	C	17,102,110

Limitations on Table 17.37.01:

L1. No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three\_hundred (300) feet from any other such activity. See Section-17-102-212\_17\_103.010 for other regulations regarding these activities.

L2. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2.500) feet from any other such activity or five hundred (500) feet from any K-12 school or Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity. See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

L3. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L42. These activities are not permitted within <u>three hundred (300)</u> feet of a lot line adjacent to the Hegenberger Road right-of-way.

L5. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

L<u>6</u>3. Existing fee parking lots may be reconfigured to increase the number of parking spaces and make more efficient use of the existing parking area. Expansion of existing facilities to include structured parking or expanding the size of the parcel with the parking constitutes an expansion of a nonconforming use and is not permitted.

L<u>7</u>4. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in Section 17.134.050, this activity must meet the following criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.37.040 - Permitted and conditionally permitted facilities.

Table 17.37.02 lists the permitted, conditionally permitted, and prohibited facilities in the CR-1 zone. The descriptions of these facilities are contained in Chapter 17.10.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).

"L" designates facilities subject to certain limitations listed at the bottom of the Table.

"-" designates facilities that are prohibited.

Facilities	Zone	Additional
	CR-1	Regulations
Residential Facilities		
One-Family Dwelling	(L1)	
One-Family Dwelling with Secondary Unit	—(L1)	
Two-Family Dwelling	(L1)	
Multifamjiy Dwelling	(L1)	· ·
Rooming House	(L1)	
Mobile Home	(L1)	
Nonresidential Facilities		
Enclosed Nonresidential	P	and a second second second second second second second second second second second second second second second
Open Nonresidential	P	· ·
Sidewalk Cafe	P	1-7-102-335 17.103.090
Drive-In	c	
Drive Through	С	
TelecommUnications Facilities		
Micro Telecommunications	P(L2)	17.128
Mini Telecommunications	P(L2)	17.128
Macro Telecommunications	C .	17.128
Monopole Telecommunications	C	17.128
Tower Telecommunications		17.128
Sign Facilities	· · · · · · · · · · · · · · · · · · ·	
Residential Signs	· P	17.104
Special Signs	P	17.104
Development Signs	P	17.104
Realty Signs	P	17.104
Civic Signs	P	17.104
Business Signs	P	17.104

Oakland, California, Code of Ordinances

Page 83

Adventising organis	 17.104	

17.37.050 - Property development standards.

A. Zone Specific Standards. Table 17.37.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table.

DeVelopment Standards	CR-1 zone	Additional Regulations
Minimum Lot Dimensions		
Width mean	50 ft	1
Frontage	50 ft	1
Lot area	7,500 sf	1
Minimum/Maximum Setbacks		
Minimum front	20 feet on parcels facing a right of way of 100 ft or more; 10 feet on parcels facing a right of way that is less than 100 feet wide.	2
Minimum interior side	O ft	3
Minimum street side	O ft	4
Rear	0/10/15 ft	5
Height and Floor Area Ratio Regulations	See Table 17.37.04	
Minimum Required Parking	See Chapter 17.116 for automobile parking and Chapter 17.117 for bicycle parking	

Table 17.37.03: Property Development Standards

Additional Regulations for Table 17.37.03:

1. See Sections 17.106.010 and 17.106.020 for exceptions to lot area, width mean, and street frontage regulations.

2. Hegenberger Road, 98th Avenue, and Edgewater **D**rive each have a right-of-way width of one\_hundred (100) feet or more. This minimum front yard setback area shall, except for necessary driveways, walkways, and allowable signs, be developed as open landscaped areas with lawn, ground cover, garden, shrubs, trees, or decorative paving materials, subject to the standards for required landscaping and screening in Chapter 17.124. Further, if fifty (50)-percent (50%) or more of the frontage on one side of the street between two intersecting streets is in any residential zone and all or part of the remaining frontage is in any commercial or industrial zone, the required front setback of the commercially or industrially zoned lots is one-half of the minimum front setback required in the residential zone. If fifty 50 percent (50%) or more of the total frontage is in more than one residential zone, then the minimum front setback on the commercially or industrially zoned lots is one-half of the total frontage is one-half of that required in the residential zone with the lesser front setback.

17.37.060 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain regulations that apply to individual lots in the CR zone may be waived or modified when and as prescribed in-Soction-17-102.320 Chapter 17.142.

17.37.070 - Other zoning provisions.

- A. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- B. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- E. Landscaping and Screening Standards. <u>The landscaping and screening regulations set forth in</u> <u>Chapter 17.124 shall apply in the The</u>-regulations-set-forth-in-Chaptor-17-12-1-and-Chapter 17.102.400, screening-of-utility-meters, etc., shall apply in the CR zone.

17.54.050 - Permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Activities:

Permanent

Residential Care occupying a One-Family Dwelling Residential Facility

Semi-Transient

B. Civic Activities:

**Essential Service** 

Limited Child-Care

Community Assembly

Recreational Assembly

Community Education

Nonassembly Cultural

Administrative

Health Care

Utility and Vehicular, but excluding communications equipment installations and exchanges

C. Commercial Activities:

General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Medical Service

General Retail Sales

Consumer Service

Consultative and Financial Service

Consumer Cleaning and Repair Service

**Consumer Dry Cleaning Plant** 

Administrative

Business, Communication, and Media Service

Broadcasting and Recording Service

Research Service

General Wholesale Sales

**Building Material Sales** 

Automobile and Other Light Vehicle Sales and Rental

Automobile and Other Light Vehicle Gas Station and Servicing

Automotive and Other Light Vehicle Repair and Cleaning

Automotive Fee Parking

D. Industrial Activities:

Custom Manufacturing

E. Off-street parking serving activities other than those listed above, subject to the conditions set forth in Section-17.102-100\_17.116.075.

17.54.060 - Conditionally permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A. Residential Activities:

Residential Care, except when occupying a One-Family Dwelling Residential Facility

Service-Enriched Permanent Housing

Transitional Housing

Emergency Shelter

B. Civic Activities:

Extensive Impact

Utility and Vehicular (communications equipment installations and exchanges, only)

Special Health Care Civic Activities

C. Commercial Activities:

Check Cashier and Check Cashing

Fast-Food Restaurant

Convenience Market

Alcoholic Beverage Sales

Mechanical or Electronic Games, subject-to-the-provisions-of-Section-17-102-210C

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

**Transient Habitation** 

Animal Care

Animal Boarding

Undertaking Service

D. Industrial Activities:

Light

E. Agricultural and Extractive Activities:

Plant Nursery

Crop and Animal Raising (see Section 17.54.090)

F. <u>Activities that are listed neither as permitted nor conditionally permitted, but are permitted or conditionally permitted on nearby lots in an adjacent zone.</u> Additional-activities-which-are permitted-or-conditionally-permitted-in-an-adjacent-zone, on-lots-near-the-boundary-thereof, subject to the conditions set forth in Section 17.102.110

17.54.070 - Permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Facilities:

One-Family Dwelling

One-Family Dwelling with Secondary Unit, subject to the provisions specified in Section <u>47-102-360 17.103.080</u>

Two-Family Dwelling

Multifamily Dwelling

Rooming House

B. Nonresidential Facilities:

Enclosed

Open

Drive-In

Sidewalk Cafes, subject to the provisions of Section-47-102.335 17.103.090

C. Signs:

Residential

Special

Development

Realty

Civic

Business

D. Telecommunications Facilities:

Micro, except when a Major Conditional Use Permit is required by Section 17.128.025

Mini, except when a Major Conditional Use Permit is required by Section 17.128.025

17.54.090 - Special regulations applying to certain activities.

- A. Fast-Food Restaurants, Convenience Markets, and Certain Establishments Selling Alcoholic Beverages-or-Providing-Mechanical-or-Electronic-Gamos. See Section-17-102-2-10.17.103.030.
- C. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in <u>Section</u> 17.134.050, this activity must meet the following use permit criteria:
  - 1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;
  - Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and
  - 3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.54.130 - Maximum residential density.

- B. Density Bonuses. The number of living units allowed by subsection A of this section may be increased by not to exceed fifty (50)-percent (50%) upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, in each of the following situations:
  - 1. In the case of a Residential Facility with more than four stories containing living units, subject to the provisions of Section 17.106.040.
  - 2. Upon the acquisition of development rights from nearby lots, subject to the provisions of Section 17.106.050.

The number of living units may also be increased, as prescribed in Section 17.106.060, in certain special housing.

17.54.180 - Buffering and landscaping.

B. Landscaping for Automobile and Other Light Vehicle Gas Station and Servicing Commercial Activities. One and one-half percent of the lot area devoted to Automobile and Other Light Vehicle Gas Station and Servicing Commercial Activities shall be developed with lawn, ground cover, garden, or shrubs, and one street tree shall be provided for each <u>one hundred (100)</u> feet of street line abutting the lot, subject to the standards for required landscaping and screening in Chapter 17.124.

17.54.190 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain of the regulations otherwise applying to individual lots in the C-40 zone may be waived or modified when and as prescribed in-Section -17-102-320 Chapter 17.142. 17.54.200 - Other zoning provisions.

- A. Parking and Loading. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116.
- B. Bicycle Parking. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117.
- C. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- D. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.

Chapter 17.56 - C-45 COMMUNITY SHOPPING COMMERCIAL ZONE REGULATIONS Sections:

17.56.010 - Title, purpose, and applicability.

17.56.040 - Required design review process.

17.56.050 - Permitted activities.

17.56.060 - Conditionally permitted activities.

17.56.070 - Permitted facilities.

17.56.080 - Conditionally permitted facilities.

17.56.090 - Restriction on accessory parking and loading within 75 feet of front lot line.

17.56.095 - Special regulations regarding crop and animal raising.

17.56.100 - Special regulations applying to Fast-Food Restaurants, Convenience Markets, and certain establishments selling alcoholic beverages.-or-providing-mochanical-or-olectronic-gamec-

17.56.105 - Reserved.

17.56.110 - Special regulations applying to the demolition of a facility containing rooming units or to the conversion of a living unit to a nonresidential activity.

17.56.120 - Limitations on Signs.

17.56.130 - Minimum lot area, width, and frontage.

17.56.140 - Maximum residential density.

17.56.150 - Maximum floor-area ratio.

17.56.160 - Maximum height.

17.56.170 - Minimum yards and courts.

17.56.180 - Minimum usable open space.

17.56.190 - Buffering.

17.56.200 - Special regulations for mini-lot and planned unit developments.

17.56.210 - Other zoning provisions.

17.56.060 - Conditionally permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A. Residential Activities:

Residential Care, except when occupying a One-Family Dwelling Residential Facility

Service-Enriched Permanent Housing

Transitional Housing

Emergency Shelter

B. Civic Activities:

Utility and Vehicular

Special Health Care Civic

Extensive Impact Civic

C. Commercial Activities:

Check Cashier and Check Cashing

Fast-Food Restaurant

Convenience Market

Alcoholic Beverage Sales

Mechanical or Electronic Games, subject-to-the-provisions of Section 17.102.210C

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

General Wholesale Sales

Transient Habitation

Automobile and Other Light Vehicle Sales and Rental

Automobile and Other Light Vehicle Gas Station and Servicing

Automotive and Other Light Vehicle Repair and Cleaning

Automotive Fee Parking

Animal Care

Animal Boarding

Undertaking Service

- D. Industrial Activities: Light Manufacturing
- E. Agricultural and Extractive Activities:

Plant Nursery

Crop and Animal Raising (see Section 17.56.095)

- F. Off-street parking serving activities other than those listed above or in Section 17.56.050, subject to the conditions set forth in Section-17-102-100\_17.116.075.
- G. Activities that are listed neither as permitted nor conditionally permitted, but are permitted or conditionally permitted on nearby lots in an adjacent zone, subject to the conditions set forth in Section 17.102.110.

G.—.-Additional-activities which are permitted or conditionally permitted in an adjacent zone, on lots near-tho-boundary-thereof, subject to tho conditions set forth in Soction 17,102,110

17.56.070 - Permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Facilities:

One-Family Dwelling

One-Family Dwelling with Secondary Unit, subject to the provisions specified in Section 17-102-360 17 103 080

Two-Family Dwelling

Multifamily Dwelling

Rooming House

B. Nonresidential Facilities:

Enclosed

Sidewalk Cafes, subject to the provisions of Section-17-102-335 17.103.090

C. Signs:

Residential

Special

Development

Realty

Civic 1

Business

D. Telecommunications Facilities:

Micro, except when a Major Conditional Use Permit is required by Section 17.128.025

Mini, except when a Major Conditional Use Permit is required by Section 17.128.025

17.56.090 - Restriction on accessory parking and loading within 75 feet of front lot line.

Accessory off-street parking and loading activities, areas, and driveways shall not be located within <u>seventy-five (75)</u> feet from the front lot line of the lot on which they are located, except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.

17.56.100 - Special regulations applying to Fast-Food Restaurants, Convenience Markets, and certain establishments seliing alcoholic beverages-or-providing-mechanical-or-electronic games.

See Section <u>17-102-210</u> <u>17.103.030</u>.

ί.
17.56.140 - Maximum residential density.

- B. Conditionally Permitted Density. The number of living units permitted by subsection A of this section may be increased by not to exceed fifty (50)-percent (50%) upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, in each of the following situations:
  - 1. In the case of a Residential Facility with more than four (4) stories containing living units, subject to the provisions of Section 17.106.040.
  - 2. Upon the acquisition of development rights from nearby lots, subject to the provisions of Section 17.106.050.

The number of living units may also be increased, as prescribed in Section 17.106.060, in certain special housing.

### 17.58.010 - Title, purpose, and applicability.

- A. Intent The provisions of this Chapter shall be known as the CBD Central Business District regulations. The intent of the CBD regulations is to:
  - 1. Encourage, support, and enhance the Central Business District as a high density, mixed use urban center of regional importance and a primary hub for business, communications, office, government, urban residential activities, technology, retail, entertainment, and transportation.
  - 2. Encourage, support, and enhance a mix of large-scale offices, commercial, urban high-rise residential, institutional, open space, cultural, educational, arts, entertainment, services, community facilities, and visitor uses.
  - 3. Enhance the skyline and encourage well-designed, visually interesting, and varied buildings.
  - 4. Encourage and enhance a pedestrian-oriented streetscape.
  - 5. Encourage vital retail nodes that provide services, restaurants, and shopping opportunities for employees, residents, and visitors.
  - 6. Preserve and enhance distinct neighborhoods in the Central Business District.
- B. Description of zones. This Chapter establishes land use regulations for the following four zones:
  - CBD-R Central Business District Residential Zone. The intent of the CBD-R zone is to create, maintain, and enhance areas of the Central Business District appropriate for residential development with small-scaled compatible ground-level commercial uses.
  - CBD-P Central Business District Pedestrian Retail Commercial Zone. The intent of the CBD-P zone is to create, maintain, and enhance areas of the Central Business District for ground-level, pedestrian-oriented, active storefront uses. Upper story spaces are intended to be available for a wide range of office and residential activities.
  - 3. CBD-C Central Business District General Commercial Zone. The intent of the CBD-C zone is to create, maintain, and enhance areas of the Central Business District appropriate for a wide range of ground-floor office and other commercial activities. Upper-story spaces are intended to be available for a wide range of residential and office or other commercial activities.
  - 4. CBD-X Central Business District Mixed Commercial Zone. The intent of the CBD-X zone is to designate areas of the Central Business District appropriate for a wide range of upper story and ground level residential, commercial, and compatible light industrial activity.
- C. Description of Combining Zone. This Chapter establishes interim land use regulations for the following combining zone:

# 17.58.030 - Conditional use permit for targe projects.

No development that involves more than two hundred thousand (200,000) square feet of new floor area, or a new building or portion thereof of more than two hundred fifty (250) feet in height, shall be permitted except upon the granting of a conditional use permit (see Chapter 17.134 for the CUP

procedure). This requirement shall not apply to developments that have been approved according to the planned unit development procedure (see Chapter 17.140 for the PUD procedure).

17.58.040 - Permitted and conditionally permitted activities.

Table 17.58.01 lists the permitted, conditionally permitted, and prohibited activities in the CBD-R, CBD-P, CBD-C and CBD-X zones. The descriptions of these activities are contained in Chapter 17.10.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134 for the CUP procedure) in the corresponding zone.

"L" designates activities subject to certain limitations or notes listed at the bottom of the Table.

"---" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

Activities	Primary Zo	nes		•	Combining Zones*	Additional
	CBD-R	CBD-P	CBD-C	CBD-X	CBD-P/CH	Regulations
Residential Activities		· · ·····		·		
Permanent	P	P(L1)	P(L1)	P	P(L1)	
Residential Care	P(L2)	P(L1)(L2)	P(L1)(L2)	P	P(L1)(L2)	17-102-212 17.103.010
Service-Enriched Permanent Housing	C(L2)	C(L1)(L2)	C(L1)(L2)	C	C(L1)(L2)	47.102.212 17.103.010
Transitional Housing	C(L2)	C(L1)(L2)	C(L1)(L2)	С	C(L1)(L2)	17-102-212 17.103.010
Emergency Shelter	C(L2)	C(L2)	C(L2)	C	C(L2)	<del>17.102.212</del> <u>17.103.010</u>
Semi-Transient	C	C(L1)	C(L1)	C	C(L1)	
Bed and Breakfast	P	P	P	P	P	17.10.125
Civic Activities		· · ·		4	··	• <u> </u>
Essential Service	P	P	P ·	P	P	
Limited Child-Care Activities	P(L3)	P(L5)	P	P	P	
Community Assembly	Č	C(L6)	C	C	C	
Recreational Assembly	P(L3)(L4)	P(L5)	P	P	P	
Community Education	P(L3)(L4)	P(L5)	P	P	P	
Nonassembly Cultural	P(L3)(L4)	P(L4)	P	P	P	
Administrative	P(L4)(L7)	P(L5)	P	P	P	
Health Care	P(L3)(L4)	P(L4)(L5)	P	P	P(L4)	
Special Health Care		C(L6)(L8)	C <u>(L8)</u>	C <u>(L8)</u>	C(L6) (L8)	17.102.390 17.103.020
Utility and Vehicular	C	C	C	C	С	
Extensive Impact	Ċ	C	C	С	C	
Commercial ActiVities	, <b></b> ,	1				L
General Food Sales	P(L4)(L7)	P(L4)	P	Р	P	
Full Service Restaurants	P(L4)(L7)	P	P	P	Р	· ·
Limited SerVice Restaurant and Cafe	P(L4)(L7)	P	P	P	P	
Fast-Food Restaurant		С	С	C	С	47-102-240 17.103.030 and 8.09
Convenience Market	C(L7)	С	С	С	С	47-102-240 17.103.030
Alcoholic Beverage Sales	C(L7)	С	C .	С	c	17-102-210 17.103.030 and 17-102-010 17.114.030
Mechanical or Electronic Games		С	C	С	С	
Medical Service	P(L4)(L7)	P(L5)	Р	P	P	
General Retail Sales	P(L4)(L7)	P	P	P	P	
Large-Scale Combined Retail and Grocery S	ales —			]	1	]

Table 17.58.01: Permitted and Conditionally Permitted Activities

Consumer Service	P(L4)(L7)	P(L4)(L9)	P(L9)	P(L9)	P(L9)	<u> </u>
	(L9)		• 1==1	1.757	1.1201	ļ
Consultative and Financial Service	P(L4)(L7)	P(L5)	P	P	P	
Check Cashier and Check Cashing	-	C <u>(L1<del>0</del>)</u>	C <u>(L10)</u>	C <u>(L10)</u>	C <u>(L10)</u>	47 <del>.102.430</del> 17.103.040
Consumer Cleaning and Repair Service	P(L4)(L7)	P(L5)	Р	P	P	
Consumer Dry Cleaning Plant	C(L7)	С	С	C	C	
Group Assembly	C(L7)(L11)	P(L4)(Li1)	P(L11)	P <u>(L11)</u>	P(L4)(L11)	
Personal Instruction and Improvement Services	P(L4)(L7)	P(L5)	P	P	P	
Administrative	P(L4)(L7)	P(L5)	P	P	P	
Business, Communication, and Media Services	P(L4)(L7)	P(L5)	Р	P	P	
Broadcasting and Recording Services Commercial Activities	-	P(L5)	P	P(L4)	P	
Research Service	P(L4)(L7)	P(L5)	P .	P	P	
General Wholesale Sales				C.	<u> </u>	
Transient Habitation	C(L8L12)	C(L6)	P	C,	C	17-102-370 17.103.050
Building Material Sales	<u> </u>	_		† <u> </u>	† <u> </u>	11.100.000
Automobile and Other, Light Vehicle Sales and	1	<u> </u>	† <u> </u>	c	†- <u></u>	
Rental						
Automobile and Other Light Vehicle Gas Station and Servicing		_	C(L <u>L13</u> 9)	C(L <u>L1</u> 39)	† <b>-</b>	· · ·
Automobile and Other Light Vehicle Repair and	_		-	<u> </u>		
Cleaning	1					
Taxi and Light Fleet-Based Services		<b>—</b> .	C(L <u>L13</u> 9)	C( <u>LL1</u> 39)	_	
Automotive Fee Parking	C( <u>L</u> 1 <u>L14</u> θ)	C(L4L140)	C <u>(L1L14</u> 9)	C( <u>L1L</u> 140)	C( <u>L4L14</u> 0)	
Animal Boarding		—		_		
Animal Care	I —	C(L6)	С	С	C	
Undertaking Service		_	С	С	<u> </u>	
Industrial Activities	· ····					• • • • • • • • • • • • • • • • • • • •
Custom Manufacturing	-(4_1_1)	(L-11)	( <u>L-1-1</u> )	C( <u>L9L</u> 13)	( <u>L</u> 11)	17.102.040
Light Manufacturing	-(L11)	-(L-1-1)	<b>(</b> [ <b>-</b> 1-1-)	C(L9L 13)	-(L11)	17.102.040
General Manufacturing	-(L-1-1)	—(L-1-1)	-(L-14-)		(L44)	17,102,040
Heavy/High Impact	· ·	_		<u> </u>	1	1
Research and Development	-	-	C(L9 <u>L13</u> )	C( <u>L9L</u> 13)	-	
Construction Operations	<b>—</b>		_		-	
Warehousing, Storage, and Distribution	J					·
A, General Warehousing, Storage and		1		C(L9L	Τ	T
Distribution	1		1	13)		
B. General Outdoor Storage	-			/	† <u> </u>	
C. Self- or Mini Storage	<b> </b>	<b> </b>	1	1_	1	
D. Container Storage	·····	1		1	1	<u> </u>
E. Salvage/Junk Yards	1	1=	-	1	1_	
Regional Freight Transportation	1_	<u> </u>			†	† <u> </u>
Trucking and Truck-Related	<b> </b>	- <u>-</u>	1=	1	1-	1 ·
Recycling and Waste-Related					±	<u> </u>
A. Satellite Recycling Collection Centers		I	C	C	1-	17,10.040
B. Primary Recycling Collection Centers		-		1	1	
Hazardous Materials Production, Storage, and Waste Management	-					
Agriculture and Extractive Activities		l	<u>I</u>	· · · · ·	L	L
Crop and animal raising	1 —	i —	1-	1 -		
Plant nursery	-	-	-			
Mining and Quarrying	_	-	<b>—</b>	- 1	-	
Accessory off-street parking serving prohibited activities	c	с	C	C	c	47 <del>,102,100</del> 17,116.075
Activities that are listed as orohibited, but are permitted or conditionally oermitted on nearby lots in an adjacent zoneAdditiooal activitiee-that are-permitted-or-conditionally-permitted in an adjacent-zone, on-lots-near-the-boundary	c	C	C	С	C	17.102.110

thereof ·

Limitations:

\* If the base zone (CBD-P) also has the CH Combining Zone, the CH regulations supersede the base zone.

L1. These activities may not be located within <u>thirty (30)</u> feet of the front lot line on the ground floor of the principal building with the exception of incidental pedestrian entrances that lead to one of these activities elsewhere in the building.

L2. <u>Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when not located in a One-Family Dwelling Residential Facility.</u> No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than <u>three hundred (300)</u> feet from any other such Activity or Facility.

L4. With the exception of parcels facing Broadway, Telegraph Avenue, and 14th Street, the total floor area devoted to these activities on the ground floor by any single establishment may only exceed <u>seven</u> thousand five hundred (7,500) square feet upon the granting of a conditional use permit (see Chapter 17.134 for the CUP procedure).

L5. If located both on the ground floor of a building and within <u>thirty (30)</u> feet from any street-abutting property line, these activities are only permitted upon the granting of a conditional use permit (see Chapter 17.134 for the CUP procedure). Incidental pedestrian entrances that lead to one of these activities elsewhere in the building are exempted from this conditional use permit requirement. In addition to the criteria contained in Section 17.134.050, these conditionally permitted ground floor proposals must also meet each of the following criteria:

a. The proposal will not impair a generally continuous wall of building facades;

b. The proposal will not weaken the concentration and continuity of retail facilities at ground-level, and will not impair the retention or creation of an important shopping frontage; and

c. The proposal will not interfere with the movement of people along an important pedestrian street.

L6. These activities are only permitted upon the granting of a conditional use permit (see Chapter 17.134 for the CUP procedure). In addition to the criteria contained in Section 17.134.050, when these activities are located within <u>thirty (30)</u> feet of the front of the ground floor of the principal building (with the exception of incidental pedestrian entrances that lead to one of these activities elsewhere in the building) the proposed activities must also meet the criteria contained in Note L5, above.

L7. These activities may only be located on or below the ground floor of a building with the following exceptions:

a) If the floor area devoted to the activity is less than <u>two thousand (2,000)</u> square feet and the activity takes place in a Local Register property, then the activity is permitted above the ground floor upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP process).

b) An activity located on the ground floor may extend to the second floor of a building if each: 1) the floor area devoted to nonresidential activities in the building is less than the floor area devoted to residential activities; 2) the activity on the second floor is the same as, or accessory to, the ground floor activity and part of the same business or establishment; and 3) there is a direct internal connection between the ground floor and the second story activities.

L8. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2,500) feet from any other such activity or five-hundred (500) feet from any K-12 school or Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity, See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

L9. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L10. No new or expanded Check Cashier and Check Cashing Commercial Activity shall be located closer than one thousand (1,000) feet from any other such activity or five hundred (500) feet from any Community Education, Community Assembly, or Recreational Assembly Civic Activity; state or federally chartered bank, savings association, credit union, or industrial loan company; or certain Alcoholic Beverage Sales Commercial Activities. See Section 17.103.040 for further regulations regarding Check Cashier and Check Cashing Commercial Activities.

L11. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

t=8L12. The total floor area devoted to these activities by any single establishment shall not exceed three thousand (3,000) square feet

**L**9<u>L13</u>. These activities, including accessory activities, are only allowed to be performed indoors. This requirement includes, but is not limited to: vehicles stored before and after servicing, general storage, vehicle and other repair, and automotive cleaning. This requirement excludes parking for customers currently at the business and automotive fueling.

L140. Auto fee parking is permitted upon the granting of conditional use permit (see Chapter 17.134 for the CUP procedure) if it is located in either a parking structure that is at least three (3) stories high or in a below grade parking lot. Auto fee parking is otherwise prohibited.

L11.— Thoso-activitios-are-not-allowed\_as-a-principal-activity\_but\_are-permittod-as-an-accossory-activity subject-to-the-regulations-contained-in-Subsoction-17-10-040E.

17.58.050 - Permitted and conditionally permitted facilities.

Table 17.58.02 lists the permitted, conditionally permitted, and prohibited facilities in the CBD-R, CBD-P, CBD-C, and CBD-X zones. The descriptions of these facilities are contained in Chapter 17.10.

"P" designates permitted facilities in the corresponding zone.

"C" designates facilities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134 for the CUP procedure) in the corresponding zone.

"L" designates facilities subject to certain limitations listed at the bottom of the Table.

"---" designates facilities that are prohibited

### Table 17.58.02: Permitted and Conditionally Permitted Facilities

Activities		Additional			
	CBD-R	CBD-P	CBD-C	CBD-X	Regulations
Residential Facilities		<b>*</b>			
One-Family Dwellings	(L1)	_			
One-Family Dwelling with Secondary Unit	P	—		1_	17.102.360 17.103.080
Two-Family Dwelling	· P			-	

Multifamily Dwelling	Р	Р	Р	P	
Rooming House	P	P.	Р	· P	
Mobile Home	-	-	-		
Nonresidential Facilities					
Enclosed Nonresidential	P	Р	P	Р	
Open Nonresidential	c	C	С	C ·	
Sidewalk Cafe	P	Р	Р	Р	17-102-335 17.103.090
Drive-in Nonresidential	. –	-		С	
Drive-Through Nonresidential	1-	-		C(L2)	17.103.100
Telecommunications Facilities			· ·		
Micro Telecommunications	С	Р	Р	P	17.128
Mini Telecommunications	c	Р	P	Р	17.128
Macro Telecommunications	c	c	c	С	17.128
Monopole Telecommunications	С	С	С	С	17.128
Tower Telecommunications	<u> </u> _	]	]—	-	17.128
Sign Facilities					
Residential Signs	P	P ·	Р	Р	17.104
Special Signs	P	Р	Р	Р	17.104
Development Signs	Р	Р	Р	Р	17.104
Realty Signs	. P	Р	·P	Р	17.104
Civic Signs	P	Р	P	P	17.104
Business Signs	Р	Р	Р	Р	17.104
Advertising Signs		_	—	_	17.104

Limitations:

L1. See Chapter 17 114, Nonconforming Uses, for additions and alterations to legal nonconforming One-Family Dwellings.

L2. No new or expanded Fast-Food Restaurants with Drive-Through Nonresidential Facilities shall be located closer than five hundred (500) feet of an elementary school, park, or playground. See Sections 17.103.030 and 17.103.100 for further regulations regarding Drive-Through Nonresidential Facilities.

17.58.060 - Property development standards.

A. Zone Specific Standards. Table 17.58.03 below prescribes development standards specific to individual zones. The number designations in the right-hand column refer to the additional regulations listed at the end of the Table.

Development Standards	Zones	Zones					
	CBD-R	CBD-P	CBD-C	CBD-X	Regulations		
Minimum Lot Dimensions	· · · · · · · · · · · · · · · · · · ·						
Width	25 ft	25 ft	50 ft	50 ft	1		
Frontage	25 ft	25 ft	50 ft	50 ft	1		

# Table 17.58.03: Property Development Standards

Lot area	4,000 sf	4,000 sf	7,500 sf	7,500 sf	1
Minimum/Maximum Setbacks		··			<u>.</u>
Minimum front	0 ft	0 ft	0 ft	0 ft	2
Maximum front and street side for the first story	None	5 ft	5 ft	10 ft	3
Maximum front and street side for the second and third stories or 35 ft, whatever is lower	None	5 ft	5 ft	None	3
Minimum interior side	0 ft	0 ft	0 ft	0 ft	4
Minimum corner side	0 ft	0 ft	0 ft	0 ft	
Rear	10 ft	0 ft	0 ft	0 ft	5
Design Regulations		<u> </u>	· · · · · · · · · · · · · · · · · · ·	<u></u>	
Ground floor commercial facade transparency	<u>55</u> 50%	<u>65</u> 70%	<u>55</u> 60%	5550%	6
Minimum height of the ground floor	15 ft	15 ft	15 ft	15 ft	7
Minimum separation between the grade and ground floor living space	2.5 ft	Not Applicable	Not Applicable	2.5 ft	8

### Additional Regulations:

3. The following notes apply to the maximum yard requirements:

a. The requirements only apply to the construction of new principal buildings and to no more than two property lines. One of these property lines shall abut the principal street.

b. The requirements do not apply to lots containing Recreational Assembly, Community Education, Utility and Vehicular, or Extensive Impact Civic Activities or Automobile and Other Light Vehicle Gas Station and Servicing Commercial Activities as principal activities.

c. In the CBD-P, CBD-C, and CBD-X zones, these maximum yards apply to seventy-five percent (75%) of the street frontage on the principal street and fifty percent (50%) on other streets, if any. All percentages, however, may be reduced to fifty percent (50%) upon the granting of regular design review approval (see Chapter 17.136 for the design review procedure). In addition to the criteria contained in Section 17.136.035, the proposal must also meet each of the following criteria:

i. <u>Any</u> The-additional yard area abutting the principal street is designed to accommodate publicly accessible sidewalk cafes and restaurants;

ii. The proposal will not impair a generally continuous wall of building facades;

iii. The proposal will not weaken the concentration and continuity of retail facilities at groundlevel, and will not impair the retention or creation of an important shopping frontage; and

iv. The proposal will not interfere with the movement of people along an important pedestrian street.

d. The maximum yard requirements above the ground floor may be waived upon the granting of a conditional use permit (see Chapter 17.134 for the CUP procedure). In addition to the criteria contained in <u>Section</u> 17.134.050, the proposal must also meet each of the following criteria:

i. It infeasible to both accommodate the use proposed for the space and meet the maximum yard requirement;

ii. The proposal will not weaken the street definition provided by buildings with reduced setbacks; and

iii. The proposal will not interrupt a continuity of 2<sup>nd</sup> and 3<sup>rd</sup> story facades on the street that have minimal front yard setbacks.

**4.** In the CBD-R zone, portions of a building over fifty-five (55) feet in height shall have a setback of at least one (1) foot from the required interior side yard for every <u>five (5)</u> four-(4)-feet that portion is above fifty-five (55) feet. This setback, however, need not exceed forty (40) feet. Also, see Section 17.108.080 for the required interior side and rear yard setbacks on a lot containing two or more living units and opposite a legally-required living room window. See Section 17.108.130 for allowed projections into required yards.

5. In the CBD-R zone, portions of a building over fifty-five (55) feet shall setback at least one (1) foot from the required rear yard for every five (5) four (4)-feet that portion is above fifty-five (55) feet. This regulation shall not apply when the rear yard faces a street. This setback, however, need not exceed forty (40) feet. The following other minimum rear yard setback regulations apply in all CBD zones.

a. A minimum ten (10) foot rear yard setback is required whenever a rear lot line abuts any portion of a lot in a residential zone; and

b---Soo-Section-17-108-110-for-reduced-required-rear-yards-eotbacks-noxt-to-an-alley- and

be. See Section 17.108.130 for allowed projections into required yards.

**6.** This percentage of transparency is only required for principal buildings that include ground floor nonresidential facilities and only apply to the facade facing the principal street. On all other street facing facades, the requirement is one-half the standard for the facade facing the principal street. The area of required transparency is between two (2) feet and <u>nine (9)</u> ten-(10)-feet in height of the ground floor and must be comprised of clear, non-reflective windows that allow views out of indoor commercial space, residential-space, or lobbies. The bottom-of-any-window-used-to-satisfy-this-requirement-may-not-bo-more than-four-(4)-feet-above-the-adjacent-sidowalk. Glass block does not quality as a transparent window. Exceptions to this regulation may be allowed by the Planning Director for unique facilities such as convention centers, gymnasiums, parks, gas stations, theaters and other similar facilities.

Regulation				Height/Bulk/Inte	ensity Area	rea		Notes
	1	2	3	4	5	6	7	1
Maximum Dens	ity (Square	Feet of Lot	Area Required	Per Unit)	······································			
Dwelling unit	300	200	90	90	90	90	90	1,2
Rooming unit	150	100	45	45	45	45	45	1,2
Maximum Floor Area Ratio	4.5	6.0	8.0	14.0	17.0	20.0	20.0	2
Maximum Heigi	ht					<u></u>		
Building base	55 ft	85 ft	55 ft	85 ft	85 ft	85 ft	120 ft	3
Total	No tower permifted	No tower permitted	170 ft	275 ft	400 ft	No height limit	No height limit	3
Minimum Heigh	nt		·	<u> </u>			<u>. ,</u>	
New principal buildings	None	None	None	45 ft	45 ft	45 ft	45 ft	4
Maximum Lot C	overage		· · ·					
Building base (for each story)	Not applicable	Not applicable	100% of site area	100% of site area	100% of site area	100% of site area	100% of site area	

Table 17.58.04 Height, Density, Bulk, and Tower Regulations

Average per storv lot coverage above the base	Not applicable	Not applicable	50% of site area or 7,500 sf, whichever is greater	75% of site area or 10,000 sf, whichever is greater	75% of site area or 10,000 sf, whichever is greater	75% of site area or 10,000 sf, whichever is greater	85% of site area or 10,000 sf, whichever is greater	5
Tower Regulati	ons	· · · · · · · · · · · · · · · · · · ·	·f······	· · · · · · · · · · · · · · · · · · ·	· · · ·			
Maximum average area of floor plates	Not applicable	Not applicable	10,000 sf	15,000 sf	20,000 sf	25,000 sf	No maximum	6
Maximum building length	Not applicable	Not applicable	115 ft	150 ft	175 ft	195 ft	No maximum	7
Maximum diagonal length	Not applicable	Not applicable	145 ft	180 ft	210 ft	235 ft	No maximum	~
Minimum distance between towers on the same lot	Not applicable	Not applicable	40 ft	40 ft	40 ft	40 ft	No minimum	-

### Notes:

1. See Chapter 17.107 for affordable and senior housing density incentives.

2. For mixed use projects in the Central Business District (CBD) zones, the allowable intensity of development shall be measured according to both the maximum nonresidential Floor Area Ratio (FAR) allowed by the zone and the maximum residential density allowed by the zone. The total lot area shall be used as a basis for computing both the maximum nonresidential FAR and the maximum residential density. No-portion-of-lot-area-used-to-meet-the-donsity-requirements-for-a-Residential-Facility-shall-be used-as-a-basis-for-computing, through the maximum-floor-area-ratio, the-maximum-amount-of-floor-area for-any-nonrosidontial-facility-on-the-same-lot, and-visa-versa-

3. In Height Areas 4, 5, and 6, lots having frontage on Broadway, San Pablo Avenue, or Telegraph Avenue where the width of the right of way is greater than eighty-five (85) feet shall have a maximum base height equal to the width of that right of way. Also, see Section 17.108.030 for allowed projections above height limits and Section 17.108.020 for increased height limits for civic buildings.

**4.** This minimum height excludes the height of the allowed projections into the height limit contained in 17.108.030.

5. The average floor area of the stories above the base cannot exceed this percentage of lot area, with the following qualifications:

a. When a project contains more than one tower above the base, the floor area of a story is calculated by adding the square footages of the equivalent story in each tower. For example, if there, are two towers above the base and the 5<sup>th</sup> story of one tower is fifteen thousand (15,000) square feet and the 5<sup>th</sup> story of the other tower is twenty thousand (20,000) square feet, then the total floor area of the 5<sup>th</sup> story is thirty-five thousand (35,000) square feet.

b. To allow a variety of articulation in a building, the floor area of an individual story can be as much as fifteen percent (15%) greater than the maximum average per story floor area above base.

c. A story that is more than fifteen percent (15%) less than the maximum average floor area is not included in the average per story floor area above the base.

6. The average floor plate of an individual tower cannot exceed this area, with the following gualifications:

a. The floor area of an individual tower floor plate cannot be more than fifteen percent (15%) greater than the maximum average tower floor plate.

b. An individual tower floor plate that is more than fifteen percent (15%) less than the maximum average tower floor plate is not included in the maximum average tower floor plate area calculation.

**7.** The following regulation applies to lots that both: 1) are designated as Special Area A on Map 17.58; and 2) have either a west or east side property line that is more than ninety (90) feet in length: the cumulative building length of the east or west elevation of all towers on such a lot shall be no more than two-thirds (2/3) the length of any east or west side property line.

17.65.030 - Permitted, conditionally permitted, and prohibited activities.

The following table lists the permitted, conditionally permitted, and prohibited activities in the HBX<sub> $\tau$ </sub> zones. The descriptions of these activities are contained in Chapter 17.10. A legally constructed facility shall be allowed to contain or be converted to contain any activities listed as permitted in the table below if they meet all applicable regulations.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134) in the corresponding zone.

"L" designates activities subject to certain limitations listed at the bottom of the table.

"---" designates uses that are prohibited in the corresponding zone.

Activity		Reg	gulations	Additional
	HBX-1	HBX-2	HBX-3	Regulations
Residential Activities		······································		
Permanent Residential	Р	Р	P	
Residential Care-occupying-a One-Family Dwelling Residential Facitity	P <u>(L1)</u>	P <u>(L1)</u>	P <u>(L1)</u>	17.102.212 <u>17.103.010</u>
Residential-Care-not-occupying a-One-Pamily-Dwelling Residontial-Facility	e	c	e	17.102.212
Service-Enriched Permanent Housing	С	С	c	17.102.212 <u>17.103.010</u>
Transitional Housing	С	c	c	17.102.212 <u>17.103.010</u>
Emergency Shelter	С	С	c	17.102.24217.103.010
Semi-Transient Residential	С	С	С	17-102-212 <u>17.103.010</u>
Bed and Breakfast		_	-	17.12.125
Civic Activities				
Essential Service	Р	P	Р	
Limited Child-Care	Р	P	Р	
Community Assembly	P(L1L2)	P(1_1_1_2)	P(나 <u>L2)</u>	
Recreational Assembly	P( <u>L1L2</u> )	P(L2 <u>L3</u> )	P( <u>+</u> 3 <u>+4</u> )	
Community Education	С	С	С	
Nonassembly Cultural	P( <b></b> <u></u>	P( <u></u> 42 <u>13</u> )	P( <u>L2L3</u> )	
Administrative	P(L <u>2L3</u> )	P(L2L3)	P(L2 <u>L3</u> )	
Health Care	С	С	c	· · · · · · · · · · · · · · · · · · ·
Special Health Care	C( <u>L5)</u>	C( <u>L5)</u>	C <u>(L5)</u>	4 <del>7.102.39</del> 0 <u>17.103.020</u>
Utility and Vehicular	С	c	С	
Extensive Impact	С	c	c	·
Commercial Activities				
General Food Sales	P(L3)	P(L3)	P(L3)	
Full Service Restaurant	P( <u></u> L3 <u>L4</u> )	P(L3 <u>L4</u> )	P( <u></u> <u></u> 43 <u></u> <u></u> 4)	
Limited Service Restaurant and Cafe	P(L3 <u>L4</u> )	P(L3 <u>L4</u> )	P(L3 <u>L4</u> )	

1-	-		17-102-210 <u>17.103.030</u>
c	С		17-102-210 17.103.030
c	С		17.102.210 17.103.030
c	с		17-102-210
P(L2 <u>L3</u> )	P(t_2 <u>L3</u> )		
P	P		
_	-		
P <u>{L6}</u>	P <u>(L6)</u>		}
P(L2 <u>L3</u> )	P(L2 <u>L3</u> )		
_	_		17.102.430 <u>17.103.010</u>
c	С		
С	C		
C <u>(L7)</u>	C( <u>L7)</u>		······
C	C		
P( <del>L2<u>L3</u>)</del>	P( <u>L2L3</u> )		
P	P		· · · · · · · · · · · · · · · · · · ·
P	P		
<u>8)</u> P(L2 <u>L3)(L4L8</u> )	P(L2L3)(L4L8)		
P( <u>L2L3</u> )	P( <u>L2L3</u> )		
_	_	· · · · · · · · · · · · · · · · · · ·	17-102-370 <u>17.103.050</u>
P(L5 <u>L9</u> )	P(L6 <u>L9</u> )		
		<u> </u>	
-	-	, <b></b>	
-	]_		
P(L7 <u>L3</u> )	P(L7 <u>L3</u> )		
	_		
P(i_7)	P(L7)		
	=		· · · · · · · · · · · · · · · · · · ·
		<u> </u>	
	<u> </u>	1	<u>l</u>
	<u> </u>	17-102-210	<u></u>
			.,
$P(\downarrow 2 \downarrow 3)$	P(L2L3)		17.120
			17.120
	P( <u>L2_3)</u> B) P(L2 <u>L3)(L4_8)</u>		

•

Oakland, California, Code of Ordinances

÷

General Manufacturing		<b></b>	······································	
Heavy <u>/High Imoact</u> Manufacturing		 		· · · · · · · · · · · · · · · · · · ·
Research and Development	P(L3)(L8)	P(L3)(L8)	P(L3)(L8)	
Construction Operations	P(L9)	P(L9)	<u>, P(L9)</u>	
Warehousind, Storage, and Distribution				
<u>A. General</u> <u>Warehousing, Storage and</u> <u>Distribution</u>	<u>P(L3)</u>	<u>P(L3)</u>	<u>P(L3)</u>	
B. General Outdoor Storage	C	C	C	
C. Self- or Mini-Storage	<u>C(L111</u>	<u>C(L11)</u>	<u>C(L11)</u>	,
D. Container Storage				
E. Salvage/Junk Yards				•
Regional Freight Transportation	=	=		7
Trucking and Truck-Related	=	<u> </u>	=	
Recycling and Waste-Related	<u> </u>			
A. Satellite Recyclino Collection Centers	=	=	<b>=</b>	
B. Primary Recycling Collection Centers	=	=	=	
<u>Hazardous Materials</u> Production, Storage, and Waste Management	=	=		
-Small-Scale-Transfer-and Storage-Hazardous-Waste Management	-		-	
Industrial-Transfer/Storage Hazardous-Waste-Management		-		
—Residuals-Repositories Hazardous-Waste-Management				
Agricultural and Extractive Activ	ities		<u></u>	
Plant Nursery	С	С	С	
Crop and Animal Raising			-C(L8L12)	
Mining and Quarrying Extractive		<u> </u>	I <u></u>	
Accessory off-street parking serving prohibited activities	C	С	c	17-102-110- <u>17.118.175</u>
Additional activities which are permitted or conditionally permitted in an adjacent zone, on lots near the boundary . thereof	С	c	C	17.102.110

# Limitations:

L1. Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when not located in a One-Family Dwelling Residential Facility. No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such Activity or Facility.

L24. The total floor area devoted to these activities by a single-ostablishment-shall only exceed ten thousand (10,000) square feet upon the granting of a conditional use permit (see Chapter 17.134).

 $\pm 2$ <u>L3</u>. The total floor area devoted to these activities by a single-establishmont-shall only exceed twentyfive thousand (25,000) square feet upon the granting of a conditional use permit (see Chapter 17.134).

t=3----The-total-floor-area-devoted-to-a-grocery-store-shall-only-exceed-twenty-five-thousand-(25,000) square-feet-upon-the-granting-of-a-conditional-use-permit-(see-Chaptor-17-134)-

<u>L4.</u> The total floor area devoted to a <u>Full-Service or Limited-Service restaurant-Restaurant Commercial</u> <u>Activity</u> shall only exceed three thousand (3,000) square feet upon the granting of a conditional use permit (see Chapter 17.134).

L5. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2,500) feet from any other such activity or five hundred (500) feet from any K-12 school or Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity. See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

L6. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L7. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

**L4L8.** Not-including-accessory<u>Outdoor principal</u> activities, this-activity-shall-take-place-entirely-within-an enclosed-building. Other-outdoor-activitios-shall-are only be permitted upon the granting of a conditional use permit (see Chapter 17.134).

**L5**<u>L9</u>. This activity shall-is only permitted upon the granting of a conditional use permit (see Chapter 17.134) if it is the principal activity-on a lot that is twenty-five thousand (25,000) square feet or larger or covers twenty-five thousand (25,000) square feet or more of lot area.

1=6<u>L10</u>. Except on Lowell Street, a nonconforming Automobile and Other Light Vehicle Gas Station and Servicing or Automotive and Other Light Vehicle Repair and Cleaning Commercial Activity in the HBX-1 zone may be extended, and the facilities accommodating or serving such activity may be altered or otherwise changed upon the granting of a conditional use permit (see Chapter 17.134) and approval pursuant to the regular design review procedure (see Chapter 17.136). This conditional use permit and regular design review approval may be granted only upon determination that the proposal is adequately buffered from the street and surrounding residential activities through landscaping and fencing. See Section 17.114 for general regulations regarding nonconforming uses.

L7--Warohousing-is-permittod-if-the-total-floor-area-by-a-single-establishment-does-not-exceed-twentyfive-thousand-(25,000)-square-feet.-Floor-areas-over-twenty-five-thousand-(25,000)-square-feet-are-only permitted-upon-the-granting-of-a-conditional-use-pormit-(seo-Chaptor-17-134). Outdoor-storage-as-a principal-activity-is-only-permitted-upon-the-granting-of-a-conditional-use-pormit-(see-Chaptor-17-134). Container-storage, oil-and-gas-storage, freight-terminals, corporation-yards, truck-terminals, and-truck sonvices as-primary-activities-are-not-permitted.-Also, see-Section-17-65-050-for-special-regulations regarding-self-storage-establishments-

L11. All facilities containing Self- or Mini Storage activities shall meet the following requirements:

- 1. No more than twenty percent (20%) of the total fleor area on a lot shall be occupied by facilities containing Self- or Mini-Storage activities.
- 2. No facility that includes a Self- or Mini-Storage activity shall be within the front twenty (20) feet of a building.
- 3. Projects that include self storage establishments shall have a minimum fifty percent (50%) of lot frontage occupied by Convenience Sales and Service, General Food Sales, General Retail

Sales, and/or General Personal Service Commercial Activities on the ground floor. These ground floor activities shall not be directly associated with the self storage establishment at the site.

L8L12. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in <u>Section</u> 17.134.050, this activity must meet the following use permit criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, secunty, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.65.040 - Permitted and conditionally permitted facilities.

The following table lists regulations relating to certain facilities. The descriptions of these facilities are contained in Chapter 17.10.

"P" indicates that the facility is permitted in the corresponding zone.

"C" indicates that the facility is only permitted upon the granting of a conditional use permit (see Chapter 17.134) in the corresponding zone.

Facility Types		Zones		Additional
	HBX-1	HBX-2	HBX-3	Regulations
Residential Facilities				den de la companya de la companya de la companya de la companya de la companya de la companya de la companya de I
One-Family Dwellings	P	Ρ.	Р	
One-Family Dwelling with Secondary Unit	P	Р	Р	<del>17.102.360 <u>17.103.080</u></del>
Two-Family Dwelling	Р	Р	P	
Multifamily Dwelling	Р	Р	Р	
Rooming House	Р	Р	Р	· · · · · · · · · · · · · · · · · · ·
Mobile Home			<b> </b>	
Nonresidential Facilities	· · · · · · · · · · · · · · · · · · ·			·
Enclosed Nonresidentjal	Р	P	Р	
Open Nonresidential	C	С	С	
Sidewalk Cafe	Р	Р	Р	47-102-335 <u>17.103.090</u>
Drive-In Nonresidential	P	Р	Р	
Drive-Through Nonresidential	C <u>(L1)</u>	C <u>(L1)</u>	C(L1)	17-102-290 17.103.100
Shopping Center	-		-	
TelecommUnications Facilities				······································
Micro Telecommunications	C	С	С	17.128
Mini Telecommunications	C	С	C	17.128
Macro Telecommunications	c	С	С	17.128
Monopole Telecommunications	С	C	С	17.128

"---" designates uses that are prohibited in the corresponding zone.

Tower Telecommunications	-		_	17.128	
Sign Facilities				***************************************	
Residential Signs	P	Р	P	17.104	
Special Signs	P	Р	Р	17.104	
Development Signs		P	P	17.104	
Realty Signs	P	P	P	17.104	
Civic Signs	P	P	P	17.104	
Business Signs	P	P	P	17.104	
Advertising Signs	i	<u> </u>	-	17.104	

Limitation: 17-65-050 - Special regulations for self storage facilities.

A.....For the purposes of this Chapter, a "self storage establishment" moans an establishmont that provides storage in small individual spaces that are exclusively ond directly accessible to a specific tenant, offered on a monthly or other limited basis, and available to the general public. Generally, the individual storage spaces are four hundred (400) square foot or less.

B. ....-No-more-than-twenty-porcont-(12%)-of-the-total-floor-area-on-a-lot-shall-be-occupied-by-self storage establishments.

C. \_\_\_\_No\_project\_that-includes\_a\_solf\_storage\_establishment\_shall-have\_any\_floor-aroa\_devoted-to-self storage-within-twenty-(20)-foet-of-tho-building-frontage.

D. Projects that include solf-storogo ostablishments shall have a minimum fifty percent (50%) of lot frontage occupied by Convenience Sales and Service, General Food Sales, General Retail Sales, and/or General Personal Service Commercial Activities on the ground floor. These ground floor activities shall not be directly associated with the self storage establishment at the site.

L1. No new or expanded Fast-Food Restaurants with Drive-Through Nonresidential Facilities shall be located closer than five hundred (500) feet of an elementary school, park, or playground. See Sections 17.103.030 and 17.103.100 for further regulations regarding Drive-Through Nonresidential Facilities.

17.65.070 - Max1mum density.

The following table contains the maximum number of living units allowed per lot for the zones in this Chapter.

	· · ·	Zone	
Living Unit Type	HBX-1	HBX-2	HBX-3
Dwelling Unit	1,000 sf of lot area per unit	930 sf of lot area per unit	730 sf of lot area per unit
Rooming Unit	500 sf of lot area per unit	465 sf of lot area per unit	365 sf of lot area per unit

### Notes:

1. See (1) <u>Section Chapter-102.360-17.103.070</u> for regulations regarding secondary units; (2) Chapter 17.107 for affordable housing density incentives; and (3) Section 17.106.060 for increased density for senior housing.

2. New construction on a vacant lot that is greater than <u>five thousand (5,000)</u> square feet shall only result in a total of one unit on the lot upon the granting of a conditional use permit (see Chapter 17.134) in

the HBX-2 and HBX-3 zones. This requirement does not apply to the expansion of the floor area or other alteration of an existing Single Family Dwelling.

17.65.080 - Maximum floor area ratio.

17.65.090 - Maximum density and floor-area ratio for mixed use projects.

- A. This Section shall only apply to the following two types of mixed use projects:
  - 1. Projects that have at least <u>twenty 20-percent (20%)</u> of its total floor area devoted to nonresidential facilities or
  - 2. Projects that:
    - a. Are on lots that are <u>fifty (50)</u> feet wide or less and
    - b. Have a minimum <u>fifty 50</u>-percent (50%) of lot frontage occupied by Ground Floor Convenience Sales and Service, General Food Sales, General Retail Sales, and/or General Personal Service Commercial Activities. This commercial floor area must be at least <u>twenty</u> (20) feet deep measured from the building frontage and be within an enclosed building. Projects on through lots require this minimum <u>fifty 50</u>-percent (50%) on only the longest lot frontage to qualify as a mixed use project for this Section.
- B. For projects described in Subsection A, the maximum number of units permitted on a lot shall not be affected by the nonresidential floor area provided on the same lot. Conversely, for these projects the maximum floor area allowed on a lot shall not be affected by the number of living units provided on the same lot. For projects described in Subsection A, this Subsection supersedes the requirements in Section 17.106.030.

### 17.65.130 - Landscaping, paving, and buffering.

- A. Submittal and approval of a landscaping and buffering plan for the entire site is required for the establishment of a new building facility (see Section 17.09.040 for definition), excluding secondary units of <u>five hundred (500)</u> square feet or less, and for additions to existing building facilities of over <u>five hundred (500)</u> square feet.
  - 1. Landscaping and buffering that is consistent with the "Design Guidelines for the HBX Zones" as adopted by the City Council;
  - 2. An automatic system of irrigation for all landscaping shown in the plan;
  - 3. A minimum of one 15-gallon tree, or substantially equivalent landscaping as approved by the Director of City Planning, for every <u>twenty-five (25)</u> feet of street frontage or portion thereof. On streets with sidewalks where the distance from the face of the curb to the outer edge of the sidewalk is at least six and one-half feet, the trees shall be street trees to the satisfaction of the City's Tree Division.
  - 4. At least one 15-gallon tree in the parking lot for every six (6) parking spaces for projects that involve new or existing parking lots of three thousand (3,000) square feet or greater.
  - 5. A minimum of five feet of landscaping shall be required adjacent to the front and street side property lines for parking lots of <u>three thousand (3,000)</u> square feet or greater. Where parking stalls face into this required area, the width of the required landscaping area shall be increased by two feet unless wheel stops are installed.

### 17.65.140 - Outdoor storage.

The outdoor storage of materials shall not exceed <u>sixteen (16)</u> feet in height on a lot. Further, outdoor storage may not be higher than eight feet if both: 1) the storage is within <u>fifteen (15)</u> feet from any property line of a lot containing residential activities, and 2) the storage faces any windows of a residential facility. Outdoor storage may also not be higher than eight (8) feet if it is within <u>fifteen (15)</u> feet from the front property line. The height of all outdoor storage shall also be restricted according to the Fire Code regulations. Sites with outdoor storage shall be screened in conformance to the "Design Guidelines for the HBX zones" as adopted by the City Council.

17.65.150 - Special regulations for HBX work/live units.

- A. Definition. An "HBX work/live unit" means a room or suite of rooms that are internally connected maintaining a common household that includes: (1) cooking space and sanitary facilities that satisfy the provisions of other applicable codes, and (2) adequate working space reserved for, and regularly used by, one or more persons residing therein. An HBX work/live unit accommodates both residential and nonresidential activities but emphasizes the accommodation of commercial activities. An HBX work/live unit meets all applicable regulations contained in this section.
- D. Activity, parking, bicycle parking, loading, open space, and unit size standards. The following table contains the activities allowed in an HBX work/live unit; the minimum size of an HBX work/live unit; and the parking, loading, and open space required for each HBX work/live unit:

Standard	Requirement					
Activities allowed in an HBX work/live unit	Same permitted and conditionally permitted activities as described in Section 17.65.030 and any activity that would qualify as a home occupation in a residential facility (See Chapter 17.112					
Required parking	One parking space per unit <u></u> pk/s-ene HBX-work/live-units	ne parking space per unit, pkis-ene-additional-unassigned-visiter-or-employee parking-space-per-five 3X-work/live-units				
Required bicycle	With private garage for each unit;	Without private garage for each unit:	2			
parking	One short-term space for each 20 dwelling units; minimum requirement is two short-term spaces.	One long-term space for each four dwelling units; minimum requirement is two long-term spaces. One short-term space for each 20 dwelling units; minimum requirement is two short-term spaces.				
Required loading	Square feet of facility Requirement		3			
	Less than 25,000 square feet No berth required		1			
	25,00069,999 square feet	One berth				
	70,000130,000 square feet	Two berths				
	Each additional 200,000 square feet One additional berth					
Residential Density	Not applicable because HBX work/live units are nonresidential facilities.					
Required usable open space	75 square feet of usable open space per unit					
Minimum size of unit	No individual unit shall be less than e	ight hundred (800) square feet of floor area	Ī			

### Notes:

- 1. See Chapter 17.116 for other off-street parking standards.
- 2. See Chapter 17.117 for other bicycle parking standards.

3. Each square foot of private usable open space equals two square feet towards the total usable open space requirement. Also, Aall required usable open space shall meet the usable open standards contained in Chapter 17.126, except that all usable open space for HBX work/live units may be provided above ground. Further, each-square-foot-of-private-usable-open-space-equals-two square-feet-towards tho-total-usable open-space-requiroment.

Unit Type	Maximum residential floor area (percent of total floor area)	Special requirements	Separation between residential and nonresidential floor area
Type 1	One-third	None	Nonresidential floor area and residential floor area shall be located on separate floors (including mezzanines), or be separated by an interior wall (see Note 1, below, for an exception for kitchens).
Туре 2	45 percent	There must be two entrances into the unit, one adjacent to the residential space, the other adjacent to the nonresidential space; the nonresidential entrance must be clearly designated as a business entrance separate from the residential entrance and be directly accessible by the public.	Nonresidential floor area and residential floor area shall be located on separate floors (including mezzanines), or be separated by an interior Wall (see Note 1, below, for an exception for kjtchens).
Туре 3	55 percent	<ol> <li>The majority of the nonresidential floor area for the unit must be at a public street level and directly accessible to the street;</li> <li>The unit must have no residential floor area at the ground level; and</li> <li>The ground floor entrance must be clearly designated as a business entrance.</li> </ol>	Nonresidențial floor area and residențial floor area shall be located on separate floors (including mezzanines).

E. Each new HBX work/live unit shall qualify as at least one of the following Unit Types:

### Notes:

1. In Types 1 and 2, a kitchen may be open to non-residential floor area if-the-kitchen\_it is adjacent to and directly accessible from a-residential floor area or stairs that lead to residential floor area.\_In these-unpartitioned-kitchens,-the-kitchen\_is-only-required-to-bo-soparatod-from-the-nonresidential floor-area-by-a-partition-that-can-be-opened--and-closed---The-c\_Counters, cabinets, sink and appliances in-the-area-that-will-function-as-a-kitchen-and the floor area that is four (4) feet in front of these items shall be considered residential floor area.

2. See <u>Section\_17.102.190</u> for regulations regarding converting facilities originally designed for industrial or commercial occupancy to joint living and working quarters.

- F.— All-required-plans-for-the-creation-of-HBX-work/livo-units-shall (1) deline ate-areas-designated-to contain-residential-activities-and-areas-designated-to-contain-nonresidential-activities-and-(2)-contain a-table-showing-the-square-footage-of-each-unit-devoted-to-residential-and-nonresidential-activities.
- GE. For HBX work/live units, residential and nonresidential floor areas shall be designated according to the following standards:
  - 1. Residential floor area shall be considered areas containing bedrooms, sleeping areas, and kitchens (not including kitchenettes).
  - 2. Nonresidential floor area shall include floor areas designated for working.
  - The floor area of stairs and-balconios-shall not be considered floor area for the purpose of this subsection.

- 4. The floor area between residential rooms that will commonly be used for residential activities and foot traffic such as the corridors and areas between bedrooms, kitchens, residentially designated bathrooms, and other similar areas shall be considered residential floor area.
- 5. The floor area of bathrooms shall be counted <u>as residential or nonresidential based on whether</u> which area\_of\_tho\_unit\_it is most conveniently accessed from the residential or nonresidential portion of the unit\_according-to-tho-following-rules:
- a.—If there is only one bathroom in the unit, half of the bathroom shall be considered residential floor area and half shall be considered nonresidential floor area;
- b.--If-there-is-moro-than-one-bathroom-in-a-unit-the-rules-in-the-following-table-shall-apply-

Bathroom-access	Floor-area-calculation-of-bathroom
Bathreem-can only be accessed-through residential-floor-area	All of bathroom shall be considered residential floor-area
Bathroom can be directly accossed-from-both nonresidential-and residential-floor-aroa	Half of bathroom shall be considered residential flaor area, the other half nonresidential floor area.
Bathroom can only be accessed through nonresidential floor area	All of bathroom shall be considered nonresidential floor area. However, if all bathrooms in the unit require access through nonresidential floor area, then at least one bathroom shall be considered residential floor area. In this case, the bathroom that is closest to or most conveniently accessed from residential floor area shall be designeted as maidential floor area.

- 6. In unpartitioned kitchens, <u>counters</u>, <u>cabinets</u>, <u>sink and appliances and the floor area that is four</u> (4) feet in front of these items shall be considered residential floor area.
  - \_(seo--footnote\_1\_of\_the\_table\_contained\_in\_subsection\_E), tho\_counters, cabinets, sink\_and appliances\_in\_the\_area\_that-will\_function\_as\_a\_kitchen\_and\_the\_floor\_area\_that\_is\_four-(4)-feet in-front-of-those-items\_shall\_bo-considered\_residential\_floor\_area.
- 7. If any part of a loft or mezzanine is designated as residential space, -according-to-rules-above, then the entire loft or mezzanine space shall be considered residential floor area.
- 8. The Planning Director shall determine the <u>floor area</u> designation of tho floor area when the above standards do-not-clearly-do-sodo not clearly apply.
- H.— Each-HBX\_work/live\_unit-shall-contain\_no-more-than-one-fully-equipped-kitchen. An HBX-work/live unit-may-contain-a-second-kitchenette-to-serve-the-nonresidential-floor-aroa. For-the-purpocos-of-this soction-a-kitchenette-shall-bo-considered-a-space-with-a-counter-that-ic-no-more-than-twenty-(20) square-foot;-a-sink;-and-an-area-for-a-refrigerator. No-stovetop-or-oven-(oxcluding-microwave-ovens) shall-be-permitted-in-a-kitchenette.
- <u>GI</u>. Each HBX work/live unit shall have at least one public entrance that is directly adjacent to nonresidential floor area. A visitor traveling through this business entrance shall not be required to pass through any residential floor area in order to enter into the nonresidential area of the unit.
- HJ. Each unit shall contain at least one tenant that operates a business within that unit. That tenant shall possess a valid and active City of Oakland Business Tax Certificate to operate a business out of the unit.
- IK. For any HBX work/live unit, a statement of disclosure shall be: (1) provided to prospective owners or tenants before a unit or property is rented, leased, or sold; and (2) in any covenant, conditions, and restrictions associated with a facility. This statement of disclosure shall contain the following acknowledgments:

- 1. The unit is in a nonresidential facility that allows commercial and/or light industrial activities that may generate odors, truck traffic, vibrations, noise and other impacts at levels and during hours that residents may find disturbing.
- 2. Each unit shall contain at least one tenant that operates a business within that unit. This tenant must possess an active City of Oakland Business Tax Certificate for the operation out of the unit.
- Jt. Each building with an HBX work/live unit shall contain a sign that: (1) is permanently posted; (2) is at a common location where it can be frequently seen by all tenants such as a mailbox, lobby, or entrance area; (3) is made of durable material; and (4) has a minimum dimension of nine by eleven inches and lettering at least one-half an inch tall. This sign shall contain the following language; "This development contains work/live units. As such, please anticipate the possibility of odors, truck traffic, noise or other impacts at levels and hours that residents may find disturbing." Further, City of Oakland regulations require that each unit have a tenant that: (1) operates a business from that unit, and (2) possesses an active City of Oakland Business Tax Certificate for this business.
- MK. HBX work/live units are nonresidential facilities and counted towards the nonresidential floor area ratio, not the residential density.
- N.— The-dovolopment-of-HBX-work/live-units-in-an-HBX-zone-shall-not-be-considered-adding-housing units-to-tho-City's-rental-supply-and-doos-not-create-"conversion-rights"-under-the-City's-condominium conversion-ordinanco, Chapter-16.36... The-dovolopment-standards-for-HBX-work/live-units-are-not intended-to-be-a-circumvention-of-the-requirements-of-the-City's-condominium-conversion-ordinance, Chapter-16.36
- OL. Regular Design Review Criteria. Regular design review approval for HBX work/live units may be granted only upon determination that the proposal conforms to the regular design review criteria set forth in the design review procedure in Chapter 17.136 and to all of the following additional criteria:
  - That the exterior of a new building containing primarily HBX work/live units has a commercial or industrial appearance. This includes, but is not necessarily limited to, the use of nonresidential building styles or other techniques.
  - That a building containing HBX work/live units has nonresidential activities and nonresidential floor area on the ground floor or level and at street fronting elevations. These-units-shall-have-a significant--ground-floor--street--presence.--The--floor--aroa--facing--the--streets--shall--contain nonresidential-activities-and-a-depth-of-at-least-twenty-(20)-foot-for-lots-more-than-thirty-five (35) feet-wide,-fifteen-(15)-feot-othonwiso.--This-ground-level-shall-be-either-part-of-a-larger--HBX work/live-Typo-3-unit-or-its-own-independent-commercial-space;
  - That units on the ground floor or level of a building have nonresidential floor area that is directly accessible from and oriented towards the street.
  - 4. That units on the ground floor or level of a building have a business presence on the street. This includes, but is not necessarily limited to, providing storefront style windows, <u>roll-up doors</u>, intorior-space-visible-to-the-street, a business door that-is-oriented towards the street, a sign or other means that identifies the business on the door and elsewhere, a prominent ground floor height, or other techniques\_<u>i</u>
  - 5. That the layout of nonresidential floor areas within a unit provides a functional and-bona-fide open area for working activities.
  - 6. That the floor and site plan for the project include an adequate provision for the delivery of items required for a variety of businesses. This may include, but is not necessarily limited to, the following:
    - a. Service elevators designed to carry and move oversized items,
    - b. Stairwells wide and/or straight enough to deliver large items,
    - c. Loading areas located near stairs and/or elevators and

- d. Wide corridors for the movement of oversized items.
- .7. That the floor and site plan for the project provide units that are easily identified as businesses and conveniently accessible by clients, employees, and other business visitors.

17.65.160 - Special regulations for HBX live/work units.

- H. Each building with an HBX live/work unit shall contain a sign that: (1) is permanently posted; (2) is at a common location where it can be frequently seen by all tenants such as a mailbox, lobby, or entrance area; (3) is made of durable material; and (4) has a minimum dimension of nine by eleven inches and lettering at least one-half an inch tall. This sign shall contain the following language: "This development contains live/work units. As such, please anticipate the possibility of odors, truck traffic, noise or other impacts at levels and hours that residents may find disturbing.
- I. HBX live/work units are residential facilities, shall be counted towards the residential density, not the nonresidential floor area ratio, and may create "conversion rights" under the City's condominium conversion ordinance, Chapter 16.36. The same requirements contained in the City's condominium conversion ordinance that relate to residential units shall apply to HBX live/work units.

17.65.170 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain regulations otherwise applying to individual lots in the HBX-1, HBX-2 and HBX-3 zones may be waived or modified when and as prescribed in Section-17-102-320 Chapter 17.142.

17.65.180 - Other zoning provisions.

- A. Parking and Loading. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116.
- B. Bicycle Parking. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117.
- C. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- D. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.

Chapter 17.72 M-20, M-30, AND M-40 INDUSTRIAL ZONES REGULATIONS

Sections:

<u>17.72.010</u>	Title, Intent, and Description
17.72.020	Required Design Review Process
17.72.030	Permitted and Conditionally Permitted Activities
17.72.040	Permitted and Conditionally Permitted Facilities
<u>17.72.050</u>	Property Development Standards
17.72.060	Special Regulations for Mini-lot and Planned Unit Developments
<u>17.72.070</u>	Other Zoning Provisions

# 17.72.010 Title, Intent, and Description

- A. Title and Intent. The provisions of this chapter shall be known as the M-20, M-30, and M-40 Industrial Zones regulations. This chapter establishes regulations for the (M-20) Light Industrial, (M-30) General Industrial, and (M-40) Heavy Industrial. The intent of the regulations is to create, preserve and enhance areas containing manufacturing, industrial, and related establishments.
- B. Description of Zones. This Chapter establishes land use regulations for the following three zones:
  - 1. M-20 Light Industrial Zone. The M-20 zone is intended to create, preserve, and enhance areas containing manufacturing, industrial and related establishments with limited external impact within an open and attractive setting, and is typically appropriate to locations adjacent to residential communities.
  - 2. M-30 General Industrial Zone. The M-30 zone is intended to create, preserve, and enhance areas containing a wide range of manufacturing, industrial, and related establishments, and is typically appropriate to areas providing a wide variety of sites with good rail or highway access.
  - 3. M-40 Heavy Industrial Zone. The M-40 zone is intended to create, preserve, and enhance areas containing manufacturing, industrial, or related establishments that are potentially incompatible with most other establishments, and is typically appropriate to areas which are distant from residential areas and which have extensive rail or shipping facilities.

# 17.72.020 Required Design Review Process

A Except for projects that are exempt from design review as set forth in Section 17.136.025, no Residential Facility, Designated Historic Property, Potentially Designated Historic Property, Telecommunications Facility, Sign, or other associated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications regulations in Chapter 17.128, or the Sign regulations in Chapter 17.104.

- No facility located within one hundred fifty (150) feet of any residential zone boundary and Β. accommodating the fellowing activities shall be constructed, established, or expanded in size unless plans for the proposal have been approved pursuant to the Regular Design Review procedure in Chapter 17.136:
  - Automobile and Other Light Vehicle Gas Station and Servicing Activity. 1.
  - 2. Automobile and Other Light Vehicle Repair and Cleaning Activity.
  - 3. Freight/Truck Terminal.
  - 4. Truck Yard,
  - 5. Truck Weigh Stations.
  - 6. Truck and Other Heavy Vehicle Sales, Rental, and Leasing.
  - Truck and Other Heavy Vehicle Service, Repair, and Refueling. 7.

### 17.72.030 Permitted and Conditionally Permitted Activities

Table 17.72.01 lists the permitted, conditionally permitted, and prohibited activities in the M-20, M-30, and M-40 zones. The descriptions of these activities are contained in Chapter 17.10. Section 17.10.040 contains permitted accessory activities.

- "P" designates permitted activities in the corresponding zone.
- "C" designates activities that are permitted only upon the granting of a Conditional Use permit (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).
- "L" designates activities subject to certain limitations or notes listed at the bottom of the table.
- "\_" designates activities that are prohibited except as accessory activities according to the regulations contained in Section 17.010.040.

<u>Activities</u>	Zones			Additional Regulations
	<u>M-20</u>	<u>M-30</u>	<u>M-40</u>	,
Residential Activities				
Permanent .	· ÷	=	=	•
Residential Care	=	=	=	
Service-Enriched Permanent Housing	=	=		
Transitional Housing	=	=	=	
Emergency Shelter	=	=	=	
Semi-Transient	=	=	=	
Bed and Breakfast	=	=	=	

Essential Service	<u> </u>	P	<u>P</u>	
Limited Child-Care Activities	<u>P</u> -	<u>P</u>	<u>P</u>	
Community Assembly	<u>C</u>	<u> </u>	<u>c</u>	· ·
Recreational Assembly	<u> </u>	<u>c</u>	<u>C</u>	
Community Education	<u>c</u>	<u>C</u>	<u>C</u>	
Nonassembly Cultural	P	=	P	,
Administrative	<u>P(L1)</u>	P(L1)	<u>P(L1)</u>	· · · · · · · · · · · · · · · · · · ·
Health Care	. <u>C</u>	=	<u> </u>	
Special Health Care	<u>C(L2)</u>	<u>C(L2)</u>	<u>C(L2)</u>	17.103.020
Utility and Vehicular	<u>c</u>	P(L3)	P(L3)	
Extensive Impact	<u> </u>	<u>C</u>	<u>C</u>	
Commercial Activities	L	_!	L,	· · · · · · · · · · · · · · · · · · ·
General Food Sales	P	P	P	· · · · · · · · · · · · · · · · · · ·
Full Service Restaurants	<u>P</u>	P	P	
Limited Service Restaurant and Café	<u>P</u>	<u>P</u>	<u> </u>	
Fast-Food Restaurant	<u><u> </u></u>	<u>C</u>	<u>c</u>	17.103.030 and 8.09
Convenience Market	<u>c</u>	<u>P</u>	<u>P</u>	<u>17.103.030</u>
Alcoholic Beverage Sales	<u>C</u>	<u>C</u>	<u>c</u>	17.103.030 and 17.114.030
Mechanical or Electronic Games	P	P	<u>P</u>	
Medical Service	P(t 1)	<u> </u>	<u>P</u>	
General Retail Sales	<u>P(L4)</u>	<u>P(L4)</u>	<u>P(L4)</u>	
Large-Scale Combined Retail and Grocery Sales	=		=	
Consumer Service	P(L4)(L5)	P(L4)(L5)	P(L4)(L5)	1
Consultative and Financial Service	<u>P(L1)</u>	<u>P(L1)</u>	<u>P(L1)</u>	· · · · · · · · · · · · · · · · · · ·
Check Cashier and Check Cashina	=	=	=	
Consumer Cleaning and Repair Service	<u>c</u>	<u>c</u>	<u>C</u>	
Consumer Dry Cleaning Plant ,	<u>C</u>	<u>C</u>	<u> </u>	//
Group Assembly	=	<u>C(L6)</u>	<u>C(L6)</u>	
Personal Instruction and Improvement Services	=	<u> </u>	<u>C</u>	
Administrative	P(L1)	<u>P(L1)</u>	<u>P(L1)</u>	
Business, Communication, and Media Services	P	P	P	

Oakland, California, Code of Ordinances

Page 119

	·		r
<u> </u>	<u>P</u>	<u>P</u>	
P	P	P	
••••••••••••••••••••••••••••••••••••••	=	=	
<u>C</u>	P	P	· · ·
P	<u>C</u>	<u>P</u>	
<u>C(L7)</u>	<u>P(L7)</u> <sup>.</sup>	<u>P(L7)</u>	
<u>P(L7)</u>	<u>P(L7)</u>	<u>P(L7)</u>	
Ċ	P	P	
P		P	
<u>C</u>	<u>C</u>	<u>C</u>	
<u>C</u>	<u>C</u>	<u>C</u>	
=	=	= .	
			<b></b>
<u>P(L8)</u>	P	P	
<u>P(L9)</u>	. <u>P</u>	<u>P</u>	
<u>C(L9)</u>	<u>C(L9)</u>	<u>C(L9)</u>	
=	=	<u>C</u>	
<u><u>c</u></u>	<u>C</u>	P	·
<u>C</u>	P	P	
<u>C</u>	<u>P</u>	P	
<u>c</u>	<u> </u>	P	
<u>P</u>	<u>P</u>	P	
<u>C</u>	<u>P</u>	<u>P</u>	
<u>c</u>	<u>c</u>	P(L10)	
· <u>C</u>	<u>c</u>	<u>C</u> -	· · · ·
C	P(L7)(L11 )	<u>P(L7)(L11</u> )	
<u>C</u>	<u>P(L7)(L11</u> )	<u>P(L7)(L11</u> )	
	• •		•
		P       P         P       P         IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P(L7)       P(L7)       P(L7)         P(L7)       P(L7)       P(L7)         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P       P         P       P

Oakland, California, Code of Ordinances

Page 120

······································				
		1	)	
D. Truck and Other Heavy Vehicle Sales, Rental and Leasing	<u>c</u>	P(L7)(L11 )	<u>P(L7)(L11</u> )	
<u>E Truck and Other Heavy Vehicle</u> Service, Repair and Refueling	C	<u>P(L7)(L11</u> )	<u>P(L71(L11</u> )	
Recycling and Waste-Related				
A. Satellite Recycling Collection Centers	Ç	<u>C</u>	<u>c</u>	
B. Primary Recycling Collection Centers	Ξ	<u>C</u>	P	<u>17.103.060</u>
Hazardous Moterials Production, Storage, and Waste Management	<u>C(L12)</u>	<u>C(L12)</u>	<u>C(L12)</u> ,	
Aariculture and Extractive Activities				
Plant Nursery	. <u>C</u>	P	P	
Crop and Animal Raisind	<u>C(L13)</u>	<u>C(L13)</u>	<u>C(L13)</u>	
Mining and Quarrying	<u>C</u>	ç	<u>C</u>	
Accessory off-street oarking serving prohibited activities	<u>P</u>	<u>P</u>	<u>P</u>	<u>17.116.075</u>
Additional activities that are permitted or conditionally permitted in an adjacent zone, on lots near the boundary thereof.	<u>C</u>	<u>c</u>	<u>.</u>	17.102.110

# Limitations on Table 17.72.01:

L1. The total floor area devoted to these activities on any single lot may only exceed fifteen thousand (15,000) square feet upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure).

L2. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2,500) feet from any other such activity or five hundred (500) feet from any K-12 school or Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity. See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

L3. Communications equipment installation and exchanges are only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure).

L4. The total floor area devoted to these activities on any single lot may only exceed three thousand (3,000) square feet upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure).

L5. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than five hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L6. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

L7. No facility accommodating either of these activities that is located within one hundred fifty (150) feet of any residential zone boundary shall be constructed, established, or altered in exterior appearance,

unless the proposal have been approved pursuant to the Design Review Procedure (see Chapter 17,136 for the Design Review Procedure).

L8. These activities are only permitted upon the granting of a Conditional Use Permit if located within one hundred fifty (150) feet of a residential zone (see Chapter 17,134 for the CUP procedure).

L9. Electroplating activities are prohibited.

L10. Salvage/Junk Yards Industrial Activities in the M-40 zone are only permitted upon the granting of a Conditional Use Permit if located within four hundred (400) feet of any zone except the M-30 zone (see Chapter 17.134 for the CUP procedure). The following regulations shall apply to all Salvage/Junk Yards Industrial Activities in the M-40 zone that do not require for a conditional use permit: Except for accessory off-street parking, landscaping, and screening, said activities shall be conducted entirely within an enclosed building or behind a solid lumber, masonry, or sheet metal fence or wall not less than ten (10) feet high, subject to the standards for required landscaping and screening in Chapter 17.124. All openings in such fence or wall shall be equipped with solid gates or doors of the same height as the fence or wall, and said gates or doors shall be kept securely closed at such times as the establishment is not open for business. Open storage of vehicles and other scrap material shall not exceed twenty (20) feet in height.

L11. Only permitted upon the granting of a Conditional Use Permit (see Section 17.134 for the CUP procedure) in the West Oakland Community Development District, defined to include all areas between Interstate 980 to the east, 3rd Street to the south, Interstate 880 to the west, and Interstate 580 to the north.

L12. These activities are prohibited when located within two thousand (2,000) feet from a residential facility. When not within two thousand (2,000) feet of a residential facility, Hazardous Materials Production, Storage, and Waste Management is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in Section 17.134.050, this activity must meet the following use permit criteria:

- 1. That the project is not detrimental to the public health, safety or general welfare of the community;
- 2. That the project is or will be adequately served by roads and other public or private service facilities;
- 3. That the project is consistent with the regional fair-share facility needs assessment and siting criteria established in the Alameda County Hazardous Waste Management Plan;
- 4. That the cumulative effects of locating the project within the proposed area have been analyzed and where applicable, measures have been incorporated into the project.

L13. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 fer the CUP procedure). In addition to the CUP criteria contained in Section 17.134.050, this activity must meet the following use permit criteria:

- The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;
- 2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and
- 3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.72.040 Permitted and Conditionally Permitted Facilities

Table 17.72.02 lists the permitted, conditionally permitted, and prohibited facilities in the M-20, M-30, and M-40 zones. The descriptions of these facilities are contained in Chapter 17.10.

- "P" designates permitted facilities in the corresponding zone.
- <u>"C" designates facilities that are permitted only upon the granting of a Conditional Use Permit</u> (CUP) in the corresponding zone (see Chapter 17.134 for the CUP procedure).
- "L" designates facilities subject to certain limitations listed at the bottom of the Table.
- "--" designates facilities that are prohibited.

Table 17.72.02. Permitted and Conditional					
Facilities	<u>Zones</u>	<u>Zones</u>			
	<u>M-20</u>	<u>M-30</u>	<u>M-40</u>		
Residential Facilities	•				
One-Family Dwelling	=	<b>=</b>	=		
One-Family Dwelling with Secondary Unit	=	=	· =	<u>17.103.080</u>	
Two-Family Dwelling	=		=		
Multifamily Dwelling	- <u>-</u>	=	=		
Rooming House	=	=	=		
Mobile Home	=`	=			
Nonresidential Facilities	•			· · ·	
Enclosed Nonresidential	P	P	P		
Open Nonresidential	<u>C(L1)</u> ,	<u>P(L1)</u>	<u>P(L1)</u>		
Sidewalk Café	<u>P</u>	<u></u> P	<u>P</u>	<u>17.103.090</u>	
Drive-In	<u>C</u>	P	P		
Drive-Through	. <u>C</u>	` <u>c</u>	<u>C</u>	<u>17.103.100</u>	
Telecommunications Facilities					
Micro Telecommunications	P	<u>P</u> .	P	<u>17.128</u>	
Mini Telecommunications	<u>P</u>	<u>P</u>	, <u>Р</u>	<u>17.128</u>	
Macro Telecommunications	P	P	P	<u>17.128</u>	
Monopole Telecommunications	<u>C</u>	P	P	<u>17.128</u>	
Tower Telecommunications	=	<u>`c</u>	<u>C</u>	<u>17.128</u>	
Sign Facilities					
Residential Signs	P	<u>P</u> .	P	<u>17.104</u>	
Special Signs	<u>P</u>	P	P	<u>17.104</u>	
Development Signs	P	<u>`</u> <u>P</u>	P	<u>17.104</u>	
Realty Sians	P	P	P	<u>17.104</u>	
<u>Civic Signs</u>	P	P	<u>P</u> .	<u>17.104</u>	
Business Signs	<u>P</u>	P	P	17.104	
Advertising Signs	=		=	<u>17.104</u>	

### Limitations on Table 17.72.02:

L1. Off-street parking facilities serving fifty (50) or more vehicles are only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure).

L2. No new or expanded Fast-Food Restaurants with Drive-Through Nonresidential Facilities shall be located closer than five hundred (500) feet of an elementary school, park, or playground. See Sections 17.103.030 and 17.103.100 for further regulations regarding Drive-Through Nonresidential Facilities.

### <u>17.72.050</u> Property Development Standards

A. Zone Specific Standards. Table 17.72.03 below prescribes development standards specific to individual zones. The number designations in the "Additional Regulations" column refer to the regulations listed at the end of the Table. "--" indicates that a standard is not required in the specified zone.

Table:17.72.03: Procenty Development Standards	A NO DEC			
- <u>Development/Standards</u>				Additional Reaulations
	<u>M-20</u>	<u>M-30</u>	<u>M-40</u>	
Minimum Lot Frontage	25 feet	25 feet	<u>25 feet</u>	<u><u>1</u></u>
Minimum/Maximum Setbacks				
Minimum front	<u>5 ft</u>	<u>0 ft</u>	<u>0 ft</u>	<u>2, 3</u>
Minimum interior side	<u>0 ft</u>	<u>0 ft</u>	<u>0 ft</u>	<u>4</u>
Minimum street side	<u>0 ft</u>	<u>0 ft</u>	<u>0 ft</u>	<u>5</u>
Rear (Nonresidential Facilities)	<u>0/10/15 ft</u>	<u>0/10/15 ft</u>	<u>0/10/15 ft</u>	<u>6. 7</u>
Maximum Hejaht	None	None	None	<u>8</u>
Minimum Required Parking	See Chapter 1 Chapte			
CourtVard Regulations	<u>S</u> e			

Additional Regulations for Table 17.72.03:

- 1. See Section 17.106.020 for exceptions to street frontage regulations.
- 2. If fifty percent (50%) or more of the frontage on one side of the street between two intersecting streets is in any residential zone and all or part of the remaining frontage is in any commercial or industrial zone, the required front setback of the commercially or industrially zoned lets is one-half of the minimum front setback required in the residential zone. If fifty percent (50%) or more of the total frontage is in more than one residential zone, then the minimum front setback on the commercially or industrially zoned lots is one-half of that required in the residential zone with the lesser front setback (see Illustration for Table 17.72.03[Additional Regulation 8], below).
- 3. In the M-20 zone, this minimum front vard shall, except for accessory driveways and walkways, be developed with lawn, ground cover, harden, shrubs, or trees, subject to the standards for required landscaping and screening in Chapter 17.124.

### Illustration for Table 17.72.03[Additional Regulation 8]



### \*for illustration purposes only

4. Wherever an interior side lot line abuts an interior side lot line of any lot located in an RH or RD zone, the setback of the abutting portion of its side lot line is ten (10) feet. In the case where an interior side lot line is five (5) feet. In the case where an interior side lot line abuts a side yard of an RU-1 or RU-2 lot, a side setback of four (4) feet is required (see Illustration for Table 17.68.03 [Additional Regulation 91, below). Also, see Section 17.108.130 for allowed projections into setbacks.

### Illustration for Table 17.72.03[Additional Regulation 9]



### \*for illustration purposes only

5. When the rear yard of a reversed corner lot abuts a key lot that is in an RH, RD, or RM zone or the RU-1 zone, the required street side yard setback in the rear twenty (20) feet of the reversed corner lot is one-half (1/2) of the minimum front yard required on the key lot (see Illustration for Table 17.33.03[Additional Regulation 11], below). Also, see Section 17.108.130 for allowed projections into the setbacks.

### Illustration for Table 17.72.03fAdditional Regulation 11]

\*for illustration purposes only



- 6. Wherever a rear lot line abuts an alley, one-half (1/2) of the right-of-way width of the alley may be counted toward the required minimum rear setback; provided, however, that the portion of the minimum rear setback actually on the lot itself shall not be so reduced to less than ten (10) feet. Also, see Section 17.108.130 for allowed projections into setbacks.
- 7. When a rear lot line is adjacent to an RH, RD, RM, or RU-1 zone, the required rear setback for both residential and nonresidential facilities is ten (10) feet if the lot depth is one hundred (100) feet or less and fifteen (15) feet if the lot depth is more than one hundred (100) feet. When a rear lot line is not adjacent to an RH, RD, RM, or RU-1 zone, the required rear setback is ten (10) feet for residential facilities and there is no required setback for nonresidential facilities.

8. Buildings shall have a thirty (30) foot maximum height at the setback line associated with any rear or interior side lot line that abut a lot in an RH. RD. RM. or RU-1 zone; this maximum height shall increase one foot for every foot of distance away from this setback line(see Illustration for Table 17.35.04 [Additional Regulation 14], below). Also, see Section 17.108.030 for allowed projections above height limits and Section 17.108.020 for increased height limits for civic buildings.



# 17.72.060 Special Regulations for Mini-lot and Planned Unit Developments

- A. Mini-lot Developments. In mini-lot developments, certain regulations that apply to individual lots in the M-20, M-30, and M-40 zones may be waived or modified when and as prescribed in Chapter 17.142.
- B. Planned Unit Developments, Large integrated developments shall be subject to the Planned Unit Development regulations in Chapter 17.142 if they exceed the sizes specified therein. In developments which are approved pursuant to said regulations, certain uses may be permitted in addition to those otherwise allowed in the M-20, M-30, and M-40 zones, and certain of the ether regulations applying in said zone may be waived or modified.

# 17.72.070 Other Zoning Provisions

- A. Performance Standards.
  - <u>1. In the M-20 zone, all Commercial and Industrial Activities shall be subject to the applicable provisions of the performance standards in Chapter 17 120.</u>
  - In the M-30 zone, all Commercial and Industrial-Activities which are located within four hundred (400) feet from any boundary of a residential zone shall be subject to the applicable provisions of the performance standards in Chapter 17.120.

- B. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- C. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- D. General Provisions. The general exceptions and other regulations set forth in Chapters 17.102, 17.103, 17.104, 17.106, and 17.108 shall apply in the M-20, M-30, and M-40 zones.
- E. Recycling Space Allocation Requirements. The regulations set forth in Chapter 17.118 shall apply in the M-20, M-30, and M-40 zones.
- F. Landscaping and Screening Standards. The regulations set forth in Chapter 17.124 shall apply in the M-20, M-30, and M-40 zones.
- <u>G. Buffering. All uses shall be subject to the applicable requirements of the buffering regulations in</u> <u>Chanter 17.110 with respect to screening or location of parking, loading, storage areas, control of</u> <u>artificial illumination, and other matters specified therein.</u>
### Ghapter 17.68 - M-20-LIGHT INDUSTRIAL ZONE REGULATIONS

Sections:

17.68.010 - Titlo, purpose, and applicability.

17.68.020 - Required design review process.

17.68.030 -- Permittod activities.

17.68.040 - Conditionally permitted activitios.

17.68.050 Permitted facilities.

17.68.060 -- Conditionally permitted facilities.

17.68.070 - Special regulations applying to certain activities.

17.68.071-Special-regulations-applying to Hazardous Waste Management Activities.

17.68.080 - Special regulations applying to the demolition of a facility containing rooming units or to the conversion of a living unit to a nonrosidential activity.

17.68.085 -- Reserved..

17.68.090 - Performance-standards-for Commercial-and Industrial activities.

17.68.100 - Limitations on Signs.

17.68.110 - Minimum lot frontage.

17.68.120 - Maximum height.

17.68.130-Minimum yards.

17.68.140 - Buffering and londscaping-

17:68:150 - Other-zoning-provisions-

17.68.010 - Title, purpose, and applicability.

The provisions of this Chapter shall be known as the M-20 light industrial zone regulations. The M-20 zone is intended to create, preserve, and enhance areas containing manufacturing and related ostablishmonts with limited external impact within an open and attractive sotting, and is typically appropriate to locations adjocont to residential communities. These regulations shall apply in the M-20 zone.

(Prior planning code § 5600)

17.68.020 -- Required design-review-process.

A., - Except for projects that are exempt from design review as set forth in Section 17.136.025, no Designeted Historic Property, Potentially Designated Historic Property, Residential Facility, Mixed Use Development, Telecommunications Facility, Sign, or other acsociated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications in Chapter 17.128, or the Sign regulations in Chapter 17.104

B..... No facility occommodating an Automotive Servicing or Automotive Repair and Cloaning Commercial Activity that is located within 150 feet of any residential-zone-boundary-shall be constructed, established, or altered in exterior appearance, unless plans for the proposal-have boon approved pursuant to the design review procedure in Chapter 17.136

47-68.030 - Permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, arc permitted:

General Food Sales Full Service Roctaurant Limited Service Restaurant and Cafe Mechanical or Electronic Gamos, subject to the provisions of Soction 17.102.210C Medical Service General Retail Sales Consumer Service Consultative and Financial Service Administrative Business, Communication, and Media Service Broadcasting and Recording Service Rosearch Service General Wholesale Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones Light Manufacturing—when located further than one hundred fifty (150) foot from residential zones D.Off-street parking serving activities other than those listed above, subject to the condition cot fo in Soction 17.102.100 8.040Conditionally permitted activities, The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134:	A	Civic Activities:
Nenassembly-Cultural Administrative BCommercial Activities General Food Sales Full Service Rostaurant Limited Service Restaurant and Cafe Mechanical or Electronic-Gamos, subject to the provisions of Soction 17.102.210C Medical-Service General-Retail-Sales Consumer-Service Consultative and Financial-Service Administrative Business, Communication, and Media-Service Broadcasting and Recording Service General-Wholesale-Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zones Light Manufacturing—when-located further than one-hundred fifty (150) foot from residential zones Light Manufacturing = econditional	Essential Service	
Administrative BCommercial Activities General Food-Sales Full Service Rostaurant Limited-Service Restaurant and Cafe Mechanical-or Electronic Games, subject to the provisions of Soction 17: 102.210C Medical-Service General Retail Sales Consumer-Service Consultative and Financial-Service Administrative Business, Communication, and Media-Service Rescarch Service General-Wholesale-Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing—when-located further than one hundred-fifty (150) foot from residential zenes Light Manufacturing serving activities. The following activities, as des	Limited Child-Caro	
B. Commercial Activities General Food Sales Full Service Rectaurant Limited Service Restaurant and Cafe Mechanical or Electronic Games, subject to the provisions of Section 17.102.210C Medical-Service General-Retail Sales Consumer Service Consultative and Financial Service Administrative Business, Communication, and Media-Service Broadcasting and Recording Service Research Service General-Wholesale-Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking C. Industrial Activities Light Manufacturing—when located further than one hundred fifty (150) foot from residential zenes Light Manufacturing—when located further than one hundred fifty (150) foot from residential zenes D.Off-street parking serving activities other than these listed above, subject to the condition cot fo in Section 17.102.100 Be040 - Conditionally permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the Conditional use pormit procedure in Chapter 17.134. A. Civic Activities Community Assembly Recreational Assembly Community-Education Utility and-Vehicular	Nonassembly Cultural	
Community Assembly Community Assembly Community Education Utility and Vohicular Community Education	Administrative	
Full Service Roctaurant         Limited Service Roctaurant and Cafe         Mechanical or Electronic Gamos, subject to the provisions of Soction 17.102.210C         Medical Service         General-Retail-Sales         Consumer Service         Consultative and Financial-Service         Administrative         Business, Communication, and Media-Service         Broadcasting-and Recording-Service         Research-Service         General-Wholesale-Sales         Automotive and Other Light Vehicle Repair and Cleaning         Automotive Fee Parking         C	B	
Limited Service Restaurant and Cafe Mechanical or Electronic Games, subject to the provisions of Soction 17.102.210C Medical Service General-Retail Sales Consumer Service Consultative and Financial Service Administrative Business, Communication, and Media Service Broadcasting and Recording Service Research Service General-Wholesale-Sales Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zenes Light Manufacturing—when located further than one hundred fifty (150) feet from residential zenes Light Manufacturing—when located further than one hundred fifty (150) feet from residential zenes D.Off street parking serving activities other than those listed above, subject to the condition cot fe in Soction 17.102.100 8.040—Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use-permit pursuant to the conditional use permit procedure in Chapter 17.134: ACivic Activities Community-Assembly Recreational-Assembly Community-Education Utility and Vehicular	General Food Sales	
Mechanical or Electronic Gamos, subject to the provisions of Section 17.102.210C Medical-Service General Retail-Sales Consumer-Service Consultative and Financial-Service Administrative Business, Communication, and Media Service Broadcasting and Recording Service Research Service General-Wholesale-Sales Automotive Fee Parking C	Full-Service Roctaurant	
Medical-Service         General-Retail-Sales         Consumer-Service         Administrative         Business, Communication, and Media-Service         Broadcasting and Recording-Service         Research-Service         General-Wholesale-Sales         Automotive and Other Light Vehicle Repair and Cleaning         Automotive Fee Parking         C.       Industrial Activities         Custom Manufacturing—when located further than one hundred fifty (150) foet from residential zenes         Light Manufacturing—when located further than one hundred fifty (150) feet from residential zenes         D.Off-street parking serving activities other than those listed above, subject to the conditions cot for in Socian 17.102.100         8.040Conditionally-permitted activities.         The following activitios, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit-pursuant to the conditional use pormit procedure in Chapter 17.134:         A.       Civic Activities         Community-Assembly       Recreational Assembly         Community-Education       Utility and Vehicular	Limited Service Restaurant and Cafe	
General-Retail-Sales Consumer-Service Consultative and Financial-Service Administrative Business, Communication, and Media-Service Broadcasting and Recording Service Rosearch-Service General-Wholesale-Sales Automotive and Other Light-Vehicle Repair and-Cleaning Automotive and Other Light-Vehicle Repair and-Cleaning Automotive Fee Parking C	Mechanical or Electronic Gamos, subject to the provisions of Soction	<del>17.102.210C</del>
Consultative and Financial Service Administrative Business, Communication, and Media-Service Broadcasting and Recerding Service Rosearch Service General Wholesale Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) feet from residential zenes Light Mamifacturing—when located further than one hundred fifty (150) feet from residential zenes D.Off street parking serving activities other than those listed above, subject to the condition cot fe in Section 17.102.100 8.040Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134: ACivic Activities Community Assembly Recreational Assembly Community-Education Utility and Vehicular	Medical-Service	
Consultative and Financial Service Administrative Business, Communication, and Media-Service Broadcasting-and-Recording-Service Rosearch-Service General-Wholesale-Sales Automotive-and-Other-Light Vehicle Repair and-Cleaning Automotive-Fee-Parking CIndustrial Activities Custom Manufacturing—when-located further than one-hundred-fifty (150) foot from residential zenes Light Mamifacturing—when-located further than one-hundred-fifty (150) foot from residential zenes D.Off-street parking serving activities, other than those listed above, subject to the condition cot fc in Section 17.102.100 8.040Conditionally-permitted-activities. The following-activities, as-described in the use classifications in Chapter 17.10, may be permitted upon-the-granting of a conditional-use-permit pursuant to the conditional-use pormit procedure in Chapter 17.134: AClivic Activities Community-Assembly Rocroational Assembly Community-Education Utility-and-Vehicular	General-Retail Sales	``
Administrative         Business, Communication, and Media Service         Broadcasting and Recording Service         Research Service         General Wholesale Sales         Automotive and Other Light Vehicle Repair and Cleaning         Automotive Fee Parking         C	Consumer Service	
Business, Communication, and Media Service Broadcasting and Recording Service Research Service General Wholesale Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones Light Manufacturing—when located further than one hundred fifty (150) foot from residential zones D.Off-street parking serving activities other than those listed above, subject to the conditions cot for in Section 17.102.100 8.040 - Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community-Assembly Recreational Assembly Community-Education Utility and Vehicular	Consultative and Financial Service	
Broadcasting and Recording Service Research Service General Wholesale Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones Light Manufacturing—when located further than one hundred fifty (150) feet from residential-zones D.Off-street parking serving activities other than those listed above, subject to the conditions cet for in Soction 17.102.100 8.040—Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use-permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community Assembly Recreational Assembly Community Education Utility-and Vehicular	Administrative	
Research Service         General Wholesale Sales         Automotive and Other Light Vehicle Repair and Cleaning         Automotive Fee Parking         C.       Industrial Activities         Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones         Light Manufacturing—when located further than one hundred fifty (150) feet from residential zones         D.Off-street parking serving activities other than those listed above, subject to the conditions cot for in Section 17.102.100         8.040 - Conditionally-permitted activities.         The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:         A.       Civic Activities         Community Assembly       Community Assembly         Genmunity-Education       Utility-and Vehicular	Business, Communication, and Media Service	
General Wholesale Sales Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones Light Manufacturing—when located further than one hundred fifty (150) feet from residential zones D.Off-street parking serving activities other than those listed above, subject to the conditions cot fe in Section 17.102.100 8.040 - Conditionally permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community Assembly Recreational Assembly Gemmunity Education Utility-and Vehicular	Broadcasting and Recording Service	
Automotive and Other Light Vehicle Repair and Cleaning Automotive Fee Parking CIndustrial Activities Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones Light Mamufacturing—when located further than one hundred fifty (150) feet from residential zones D.Off-street parking serving activities other than those listed above, subject to the conditions cot fo in Soction 17.102.100 8.040 - Conditionally permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Cemmunity Assembly Recreational Assembly Cemmunity Education Utility and Vohicular	Rosoarch Service	
Automotive Fee Parking       Industrial Activities         C.       Industrial Activities         Custom Manufacturing—when located further than one hundred fifty (150) foot from residential zones       Light Mamifacturing—when located further than one hundred fifty (150) feet from residential zones         D.Off-street parking serving activities other than those listed above, subject to the conditions cot for in Section 17.102.100       8.040Conditionally-permitted activities.         The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134:         A.       Civic Activities         Community Assembly       Recreational Assembly         Community-Education       Utility and Vehicular	General Wholesale Sales	
C. Industrial Activities Custom Manufacturing — when located further than one hundred fifty (150) foot from residential zones Light Mamifacturing — when located further than one hundred fifty (150) feet from residential zones D.Off-street parking serving activities other than those listed above, subject to the conditions cot fe in Soction 17.102.100 8.040Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: A. Civic Activities Community Assembly Recreational Assembly Community Education Utility and Vohicular	Automotive and Other Light Vehicle Repair and Cleaning	· · ·
Custom Manufacturing — when located further than one hundred fifty (150) foot from residential zones Light Mamufacturing — when located further than one hundred fifty (150) feet from residential zones D.Off-street parking serving activities other than those listed above, subject to the conditione cot feet in Section 17.102.100 8.040 Conditionally permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community Assembly Recreational Assembly Community Education Utility and Vohicular	Automotive Fee Parking	·
zones Light Mamifacturing—when located further than one hundred fifty (150) feet from residential zones D. Off-street parking serving activities other than those listed above, subject to the conditions out for in Section 17.102.100 8.040 Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community Assembly Recreational Assembly Community-Education Utility-and Vohicular	C	Industrial Activities:
D.Off-street parking serving activities other than those listed above, subject to the conditions cot for in Section 17.102.100 8.040 Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community-Assembly Recreational Assembly Community-Education Utility and Vohicular	-	150) foot from residential
in Section 17.102.100 8.040 Conditionally-permitted activities. The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community-Assembly Recreational Assembly Community-Education Utility and Vohicular	Light Mamifacturingwhen located further than one hundred-fifty (150	) feet from residential zones
The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134: ACivic Activities Community Assembly Recreational Assembly Community Education Utility and Vohicular		ubject to the conditions cot for
upon the granting of a conditional use-permit pursuant to the conditional use-pormit procedure in Chapter 17.134: ACivic Activities Community Assembly Rocreational Assembly Community Education Utility and Vohicular	8.040 - Conditionally-permitted activities.	
Community-Assembly Rocreational Assembly Community-Education Utility-and Vohicular	upon the granting of a conditional use permit pursuant to the condition	oter 17.10, may be permitted Pal-use pormit procedure in
Rocreational Assembly Community-Education Utility-and Vohicular	A	
Community-Education Utility-and Vohicular	Community-Assembly	
Utility and Vohicular	Rocreational Assembly	
	Community-Education	
	Utility and Vohicular	· · ·
	Extonsive Impact	

۰.	~ ~	 	<b>ι Λ</b> .	-+1.	itios:
	$\Delta \mathbf{n}$				11.12

Fast-Food-Restaurant

Convenience Market

Alcoholic Beverage Sales

Consumer Cleaning and Repair Service

Consumer-Dry-Cleaning-Plant

Wholesale Professional Building Material Sales

Automobile and Other Light Vehicle Gas Station and Servicing

Taxi and Light Fleet Based Service

Transport and Warehousing

Animal-Care

Animal Boarding

<del>C.</del>

R.

-Industrial Activitiec:

Non-second ended the elitic en

Custom Manufacturing—when located within one-hundred-fifty (150) feet of residential-zones

Light-Manufacturing—when located within one hundred fifty (150) feet of residential zones

General Manufacturing, provided that electroplating activities shall also be subject to the previsions of Section 17:102:340

Warehousing, Storage, and Distribution—Automotive Salvage/Junk Yards

Small Scale Transfer and Storage Hazardous Waste Management when located a minimum of 2,000 foot from a Residential Facility; such facilities when located within 2,000 feot of a Residential Facility . are not permitted

----- Agricultural and Extractive Activities:

Plant Nursery

Crop and Animal Raising (see Section 17.68.070)

Mining and Quarrying

€...

Ð.

Additional activities which are permitted or conditionally pormittod in an adjacent zone, on lots near the boundary thereof, subject to the conditions cot forth in Section 17.102.110

### 17.68.050 - Permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, are permitted:

A	· · · · · · · · · · · · · · · · ·		-Nonresidentia-i	achitles
Enclosed				
B				<u>—Signs</u> :
Special		· .		
Development				
Realty				
Civie				
Business				

C	 	
Micro	<i>,</i>	•

Mini

Масго

17,68.060 - Conditionally permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A	Nonrosidential Facilitios:
Open	
Drive-In	
Drive-Through	
B	Telecommunications:
••	

Monopole

47.68.070 - Special regulations applying to certain activities.

A. Administrative Activities, Medical Service, and Consultative and Financial Service. The total floor area devoted to Administrative Civic Activities or Administrative, Medical Service, or Consultative and Financial Service Commorcial Activities on any single lot shall not exceed fifteen thousand (15,000) square feet, except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134

B.General Retail-Sales and Consumer Service. The total floor area devoted to General Retail-Sales or Consumer Service Commercial Activities by any single establishment shall not exceed three thousand (3,000) square feet, except upon the granting of a conditional uso permit pursuant to the conditional uso permit procedure in Chapter 17.134

C. Fast-Food Restaurants, Convenience Markets, and Certain Establishments Selling Alcoholic Beverages or Providing Mechanical or Electronic Games. See Soction 17-102.210

D. Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 fer the CUP procedure). In addition to the CUP criteria contained in 17.134.050, this activity must most the following use permit criteria:

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. Tho soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

Editor's-note-

17.68.071--Special rogulations applying to Hazardous Waste Management Activitios.

A conditional use permit for a Hazardous Waste Management Activity may be granted only upon determination that the proposed development conforms to the general uso permit criteria sot forth in the conditional use permit procedure at Chapter 17.134 and to all of the following use permit criteria:

A.That the project is not detrimental to the public health, safety or general-welfare of the community;

B. That the project ic or will be adequotely served by reads and other public or private service facilities;

C. --- That-the project is consistent with the regional fair-share focility needs assessment and siting criteria established in the Alameda County Hazardous Waste Monagement Plan;

D. That the cumulative effects of locating the project within the proposed oroa have been analyzed and where applicable, measures have been incorporated into the project.

17.68.080 - Special regulations applying to the domeiltlon of a facility containing resoning units or to the conversion of a living unit to a nonrosidential activity.

Soo Section 17.102.230.

(Prior planning code § 5609)

17.68-085 - Reeerved-

Editor's note---

17.68.090 - Performance standards for Commercial and Industrial activities.

All Commercial and Industrial Activities shall be subject to the applicable provisions of the performance standards in Chaptor 17.120.

Editor's-note-

17.68.100 - Limitations on Signs.

A.— General Limitations. All Signs shall be subject to the applicable limitations set forth in Section 17.104.020

17.68.110 - Minimum lot-frontago.

Every lot shall have a minimum frontage of 25 feet upon a street, except as this requirement is modified by Soction 17.106.020

(Prior planning code § 5614)

17.68.120 - Maximum height.

Except as provided in Chapter 17.128, no general maximum height is prescribed, except that the height of facilities shall be limited, as prescribed in Soction 17.108.010, on lots lying along a boundary of any of certain other zones. But see Soction 17.68.100 for maximum height of Signs, and Section 17.108.130 for maximum height of facilities within minimum yards and courts.

#### 17.68.130 - Minimum yards.

No yards are generally required except as indicated below. The following minimum yards shall be provided unobstructed except for the accessory structures or the ether facilities allowed therein by Section 17.108.130. See also Section 17.108.020 for greater yard requirements applying to certain facilities which exceed the general maximum height prescribed in Section 17.68.120.

A. Front-Yard. The minimum front yard depth en-every lot shall be five foot, except that a greater depth shall be required, as prescribed in Section 17.108.040, in certain situations where part of the frontage on the same side of a block is in a residential zone.

B. Side-Yard Street Side of Corner Lot. A side yard shall be provided, as prescribed in Section 17.108.070, on the street side of a corner lot in certain situations where a lot to the rear of the corner lot is in a residential zone.

C. Side Yard—Intertor Lot Line. A side yard shall be provided, as prescribed in Section 17.108.090, along an interior side lot lino lying along a boundary of any of certain ether zonoc.

D.Rear Yard. A rear yard shall be provided, as prescribed in Section 17.108.100, along a boundary of any of certain other zones.

(Prior planning code § 5620)

17.68.140 - Buffering and landscaping.

A: General Roquirements. All uses shall be subject to the applicable requirements of the buffering regulations in Chapter 17.110 with respect to screening or location of parking, loading, and storage areas; control of artificial-illumination; and other matters specified therein.

B-I=andscaped-Front Yard.—Tho minimum front yard required by Sections 17.68.130A and 17.108.040 shatl, except for necessary driveways and walkways, be developed with lawn, ground cover, garden, shrubc, or trooc, subject to the standards for required landscaping and screening in Chapter 17.124

(Prior planning code § 5622)

17.68.150 - Other zoning provisions.

A: Parking-and Loading. Off-street parking and loading-shall be provided as proscribed in the offstreet-parking and loading requiremente in Chapter 17.116

B.Bicycle Parking. Bicycle-parking-shall be provided as prescribed in the bicycle parking rogulations in Choptor 17.117

C.—\_\_\_\_\_Nonconforming Usos. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.

D.General Provisions. The general exceptions and ether regulations set forth in Chapter 17.102 shall apply in the M-20-zone.

E.Recycling Space Allocation Requirements. The regulations set forth in Chapter 17.118 shall apply in M-20 zone.

### Chapter 17.70 -- M-30-GENERAL INDUSTRIAL ZONE REGULATIONS

Sections:

17.70.010 - Title, purpose, and applicability.

17.70.020 - Required design review process.

17-70-030 - Permittod activities-

17-70-040 - Conditionally pormitted activities.

17.70.050 - Permitted-facilities-

17.70.060 - Conditionally permitted facilities.

17.70.070 - Special rogulations applying to certain Commercial and Civic Activities.

17.70.080 - Special regulations applying to the demolition of a facility containing reeming units or to the conversion of a living unit to a nonresidential activity.

17-70-081 -- Special regulations applying to Hazardous Waste Management Activities-

17-70-085 - Reserved-

17-70.090 - Performance standards for Commorcial and Industrial Activities within four hundred feet of residential zone.

17.70.100 - Limitations on Signs.

17-70.110 - Minimum-lot frontage.

17-70-120 - Maximum height.

17.70.130 - Minimum yards-

17.70.140 - Buffering-

17-70-150 - Other zoning previsions.

17.70.010 - Title, purpose, and applicability.

The provisions of this Chaptor shall be known as the M-30 general industrial zone regulations. The M-30 zone is intended to create, preservo, and enhance areas containing a wide-range of manufacturing and related ostablishments, and is typically appropriate to areas providing a wide variety of sites with good rail or highway access. These regulations shall apply in the M-30 zono.

(Prior planning code § 5700)

17.70.020 - Required design-review-precess-

A. Except for projects that are exempt from design review as set forth in Section 17.136.025, no Designated Historic Property, Potentially Designated Histehs Property, Residential Facility, Mixed Use-Development, Telecommunications Facility, Sign, or other associated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications regulations in Chapter 17.128, or the Sign regulations in Chapter 17.101

B.------No facility accommodating an Automotive Servicing or Automotive Repair and Cloaning Commercial Activity that is located within 150 feet of any residential-zone-boundary shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136

17.70.030 - Permitted astivities.

The following activities, as described in the use classifications in Chapter 17.10, are permitted:

essential-Service imited Child-Care	
imited Child-Care	
lenassombly Cultural	
dministrative	
Itility and Vehicular, but excluding communications equipment installa	stion and oxchanges
·	Commercial Activities
Senerol Food Sales	
ull-Sonvice-Restaurant	
mited-Service-Restaurant-and-Cafe	
envenience-Market	· .
leehanical or Electronic Games, subject to the provisions of Section 1	17.102.210C
ledical Service	
eneral Retail Sales	
oncumer Service	
onsultative and Financial Service	
dminictrative	
usinosc, Communication, and Media Service	
roadcosting and Recording Service	
esearch Service	
eneral Wholesale Sales	
uilding Material Sales	
utomobile and Other Light Vehicle Gas Station and Servicing	
utomotive-and Other-Light-Vehicle-Repair-and Cleaning	
axi-and-Light Fleet-Based-Service	
utomotive Fee Parking	
ransport-and Warehousing	
·	
Sustom-Manufacturing	
ight-Manufacturing	
Seneral-Manufacturing, except electroplating activities	
Agricult	tural-and Extractive-Activities
lant Nursery	
rep-and-Animal-Raising	

17.70.040 - Conditionally permitted activities.

Oakland, California, Code of Ordinances 🔗

.

Α	······						Activities
Communit	y-Assembly						
Recreation	al Assembly		-				
Communit	y-Education						
Extensive	Impact						
Utility and	Vohicular (commi	nications equ	ipment instal	ations-and-e	xchanges,	-only)	
Special He	alth-Care-Civic-A	tivities					-
₿		nden <b>2</b> 74 (20, 2000)	<u>1,</u>		C	ommercial	-Activities
Fast-Food	Restaurant						
Alcoholic [	everage-Sales						
Consumer	Cleaning and Re	oir-Service					
Consumer	Dry Cleaning Pla	ŧ					
Group Ase	emb <del>ly</del>						
Personal I	nstruction-and-Imp	rovement and	d Small Scale	-Entertainme	nŧ		
Automobil	and-Other-Light-	√ehicle					
Sales and	Rental				•		
Animal Ca	Fe ·						
Animal Bo	arding						
<del>C</del>	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·				Industrial	-Activities
General-M	anufacturing, eloc	troplating acti	vitios subject	to the provisi	ions-of-So	ction 17-10	<del>2.340</del>
Warehous	n <del>g, Storage</del> -and-I	)istribution/	Automotive Sa	alvage/Junk `	¥ards		
	e Transfer and St Residential Facil mitted						
	<del>Fransfer/Storoge Ι</del> sidential Facility; ε ed						
Ð.		<u>.                                    </u>		Agric	ultural and	l Extractive	-Activities
Mining and	l-Qu <del>orryi</del> ng						
E.Addition the bound	al-activities-which ary-thereof, subje	are-permitted t to the condi	⊢ <del>or conditiona</del> tions-set forth	Illy-permitted ⊢in Section-1	in-an-adja 7102.110	icent-zone,	on lots n
0.050 - Pe	mitted-facilities.						
The follow	ing-facilities, as d	scribod in the	e uso classific	atiens in Cha	opter 17 1	). aro-porm	ittod:
110101010						nresidentia	

Oakland, California, Code of Ordinances

.

<del>Drive-In</del>		
B		Signs:
Special		· ·
Development		
Realty		,
Civic		
Business		
<b>C</b>	<u></u>	Telecommunications:
Micro		
Mini		
Масто		· · · · · ·
Monopóle		
Tower		
		•

17.70.060 - Conditionally pormitted facilities.

The following-facilities, os described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A. \_\_\_\_\_Nonresidential Focilitios: Drive-Through

(Prior planning oode § 5706)

17.70.070 - Special regulations applying to certain Commercial and Civic Activities.

A. Administrative Activities, Medical Service, and Consultative and Financial Service. The total floor area devoted to Administrative Civic Activities or Administrative, Medical Service, or Consultative and Financial Service Commercial Activities on any single lot shall not exceed 15,000 square feet, except upon the granting of a conditional use permit-pursuant to the conditional use permit procedure in Chapter 17,134

B.General-Retail Sales and Consumer-Service. The total floor aroo devoted to Gonoral Retail Sales or Consumer Service Commercial Activities by ony single establishment chall not exceed 3,000 cquare feet, except upon the granting of a conditional use permit pursuant to the conditional use pormit procedure in Chapter 17.134

C.Certain Establishmontc-Selling Alcoholic Beverages or Providing Mechanical or Electronic Games. See Section 17.102.210

17.70.080 - Special regulations applying to the demolition of a facility containing rooming units or to the sonversion of a living unit to a nonrosidential activity.

See-Section 17.102.230.

(Prior planning code § 5709)

17.70.081 -- Special regutations-applying to Hazardous Waste Management Activities-

A conditional use pormit for a Hazardous Waste Management Activity may be granted only upon determination that the proposed development conforms to the general use pormit criteria set forth in the conditional use pormit procedure in Chapter 17.134 and to all of the following use permit criteria:

A That the project is not detrimental to the public health, safety or generol welfare of the community;

B. That the project is or will be adequately served by reads and other public or private service facilities;

C. That the project is consistent with the regional fair-share facility needs accessment and siting criteria established in the Alameda County Hazardous Waste Management Plan;

D.-That the cumulative effects of locating the project within the proposed area have been analyzed and where applicable, measures that minimize adverce imposts to the surrounding community have been incorporated into the project.

17.70.085 -- Reserved-

Editor's note-

17.70.090 - Performance standards for Commercial and Industrial Activities within four hundred feet of residential zono.

All Commercial and Industrial Activitios which are located within four hundred (400) feet from any boundary of a residential zono chall be cubject to the applicable provisions of the performance standards in Chapter 17.120.

17.70.100 - Limitations on Signs.

A.---- General Limitations. All Signs chall be subject to the applicable limitations set forth in Section 17.104.020

17-70-110-Minimum-tot-frontage.

Every lot shall have a minimum frontage of 25 feet upon a street, except as this requirement is modified by Section 17.106.020.

(Prior planning code §-5714)

17.70.120 - Maximum height.

Except as provided in Chapter 17.128, no general maximum height is prescribed, except that the height of facilities chall be limited, as prescribed in Section 17.108.010, on lots lying along a boundary of any of certain other zones. But see Section 17.70.100 for maximum height of Signs, and Section 17.108.130 for maximum height of facilities within minimum yards and courts.

17.70.130-Minimum-yards.

No yards are generally required except as indicated below. The following minimum yords shall be provided unobstructed except for the accessory structuros or the other facilities allowed therein by Soction 17.108-130:

A. Front Yard. A front yard shall be provided, as prescribed in Section 17.108.040, in certain situations where part of the frontage on the same side of a block is in a residential zone.

B.— Side Yard—Street Side of Corner Lot. A side yard shall be provided, as prescribed in Section 17.108.070, on the street side of a corner lot in certain situations where a lot to the rear of the corner lot is in a residential zone.

C. Side Yard — Interior Lot Line. A side yard shall be provided, as prescribed in Section 17.108.090, along an interior cide lot-line lying along a boundary of any of certain other zonos.

D.Rear-Yard. A rear-yard shall be provided, as prescribed in Section 17.108.100, along a boundary of any of certain-other-zones.

(Prior-planning-code § 5720)

### 17.70.140 - Buffering.

All uses shall be subject to the applicable requirements of the buffering regulations in Chaptor 17,110 with respect to screening or location of parking, loading, and storage areas; control of artificial illumination; and ether mattors specified therein.

(Prior planning code § 5722)

17.70.150 - Other zoning-provisions.

A.— Parking and Loading. Off-street-parking-and loading-shall be-provided as-proccribed-in-the-offstreet-parking and loading requirements in Chapter 17.116

B.Bicycle Parking. Bicycle parking shall be provided as prescribed in the bicycle parking rogulations in Chapter 17.117

C.------- Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114

D.General Provisions. The general exceptions and ether regulations set forth in Chaptor 17.102 shali apply in the M-30 zone.

E.Recycling Space Allocation Requirements. The regulations set forth in Chaptor 17.118 shall-apply in the M-30-zono.

# Chapter 17.72 - M-40 HEAVY INDUSTRIAL ZONE REGULATIONS

Sections:

17.72.010 - Title, purpose, and applicability.

17.72.020 - Required design review process.

17.72.030 - Permitted activities

17.72.040 - Conditionally permitted activities.

17.72.050 - Pormittod facilities.

17.72.060 - Conditionally permitted facilities.

17.72.070 - Special regulations applying to certain Commercial and Civic Activitios.

17.72.080 - Special regulations applying to the demolition of a facility containing rooming units or to the conversion of a living unit to a nonresidential activity.

17.72.081--Special regulations applying to Hazardous Wacto Management Activities.

17.72.085 - Reserved

17.72.090 - Limitations on Signs

17.72.100 - Minimum lot frontage.

17.72.110 - Maximum height.

17.72.120 - Minimum yards.

17.72.130 - Buffering.

17.72.140 - Other zoning provisions

17.72.010 - Title, purpose, and applicability.

The previsions of this Chapter shall be known as the M-40 heavy industrial zone regulations. The M-40 zone is intended to create, preserve, and enhance areas containing manufacturing or related establishments which are potentially incompatible with most other establishments, and is typically appropriate to areas which are distant from residential areas and which have extensive rail or shipping facilities. These regulations shall apply in the M-40-zone.

(Prior planning code § 5800)

17.72.020 - Required design review process.

A. Except for projects that are exempt from design review as set forth in Section 17.136.025, no Docignated Historic Property, Potentially Dosignated Historic Property, Residential Facility, Mixed Use Development, Telecommunications Facility, Sign, or other associated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved purcuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications in Chapter 17.128, or the Sign regulations in Chapter 17.104

B............No facility accommodating an Automotive Servicing or Automotive Repair and Cleaning Commercial Activity that ic located within 150 feet of any recidential zone boundary shail-be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chaptor 17.136

17.72.030 - Pormittod activitiee.

The following activities, ac described in the use classifications in Chaptor 17.10, are permitted:

Civic Activities:

**Essential Service** Limited Child-Care Nonacsombly Cultural Administrative Utility and Vehicular, but oxcluding communications equipment installation and exchanges 8. Commercial Activities: General Food Sales Full-Service-Restaurant Limited Service Restaurant and Cafe Convenience-Market Mechanical or Electronic Games, subject to the provisions of Section 17.102.210C Medical-Service General-Retail Sales Consumer Service Consultative and Financial Service Administrative Business, Communication, and Media Service Broadcasting-and-Recording-Service Resoarch Service General Wholesale-Sales **Building Material Sales** Automobile and Other Light Vehicle Sales and Rental Autemebile and Other Light Vehicle Gas Station and Servicing Automotive-and-Other Light Vehicle-Repair and Cleaning Taxi and Light Fleet-Based Service Automotive Fee Parking Tronsport and Warehousing G. Industrial Activities: Gustom Manufacturing Light Manufacturing General Manufacturing, except electroplating activities Ð. Agricultural and Extractive Activities: **Riant Nursery** Grop-and-Animal Raising E-Off-street parking serving activities other than these listed above, subject to the conditions set forth in-Soction 17,102,100

17.72.040 - Conditionally pormitted activities.

The following activitios, ac described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A	Civic Activities:
Community Assembly	
Recreational Assembly	· · · ·
Community Education	
Extensive Impact	
Utility-and Vehicular (communications equipment ins	tallatiens-and-exchanges, only)
Special Health Care Civic Activities	
B	Commercial-Activities:
Fast-Food-Restaurant	
Alcoholic Beverage Sales	
Consumer Cloaning and Repair Service	
Censumer Dry Cleaning Plant	
Group Assembly	
Personal Instruction and Improvemont and Small Sca	ale-Entertainment
Animal-Care	
Animal Boarding	
Q	Industrial Activities:
General, electroplating activities subject to the provis	ions of Section 17.102.340
Heavy/High Impact Manufacturing	
Small-Scale Transfer and Storage Hazardous Waste feot from a Residential Facility; such facilities when h are not permitted	
Industrial Transfer/Storage Hazardous Waste Manag from a Residential Facility; such facilities when-locate net permitted	ement when located a minimum of 2,000 foet ed within 2,000 foet of a Residential Focility are
D	
Mining and Quarrying	
E.Additional activitios which are permitted or condition the beundary thereof, subject to the conditions sot for	
17.72.050 Permitted fasilities.	
The following facilities, as described in the use class	ifications in Chapter 17.10, are permitted:
A	Nonresidential Facilities:
Enelesed	· · ·
Open	
Ðrive-In	
B	Signs:

Special			
Development			
Realty			
Civic			•
Business			
G	 <u>.</u>		Telecommunications:
Miere			
Mini			
Масго			́с.,
Monopole Towor	`		

17.72.060 - Conditionally pormittod facilities.

The following facilities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

Nonresidential Facilities:

Drive-Through

17.72.070 -- Special regulations applying to certain Commorcial and Civic Activitios.

A. Administrative Activities, Medical Service, and Consultative and Financial-Service. The total-floor area devoted to Administrative Civic Activities or Administrative, Medical Service, or Consultative and Financial Service Commercial Activities on any single lot shall not exceed 15,000 square feet, except upon the granting of a conditional use-permit-pursuant to the conditional use permit procedure in Chapter 17,134

B.General Retail Sales and Consumer Service. The total floor area devoted to General Retail-Sales or Consumer Service Commercial by any single establishment shall not exceed 3,000 square feet, except upon the granting of a conditional use permit-pursuant to the conditional use permit procedure in Chapter 17.134

C.Certain Establishments Selling Alcoholic Beverages or Providing Mechanical or Electronic Games. See Section 17.102.210

Ð...

з

----Scrap-Operations.

1.No Scrap Operation Commercial Activity shall be located within 400 feet from the boundary of any other zone except the M-30-zone, except upon the granting of a conditional-use permit pursuant to the conditional-use permit procedure in Chopter 17.134

2. The following regulations shall apply to all Scrap Operation Commercial Activities in the M-40-zone which are not subject to the requirement for a conditional use permit. Except for accessory off-street parking, landscaping, and screening, said activities shall be conducted entirely within an enclosed building or behind a solid lumber, masonry, or sheet metal fence or wall not less than ten feet high, subject to the standards for required landscaping and screening in Chapter 17.124. All openings in such fence or wall shall be equipped with solid gates or doors of the same height as the fence or wall, and said gates or doors shall be kept securely closed at such times as the establishment is not open for business. Open storage of vehicles and other scrap matchal shall not exceed 20 feet in height.

See Soctions 17.72,090 and 17,102,210F.

17.72.080 - Special rogulations applying to the domolition of a facility containing rooming units or to the conversion of a living unit to a nonrocidential activity.

See-Section 17.102.230.

(Prior planning code § 5809)

17.72.081 - Special rogulations applying to Hazardous Waste Management Activitios.

A conditional uso permit for a hazardouc waste management activity may be granted only upon determination that the proposed development conforms to the general use permit criteria sot forth in the conditional uso permit procedure at Chapter 17.134 and to all of the following use permit criteria:

A. That the project is not detrimental to the public health, safety or general welfare of the community;

B. \_\_\_\_\_That the project is or will be adequately cerved by roads and other public or private service facilities;

C. That the project is consistent with the regional fair-share facility-noodo assessment and siting criteria established in the Alameda County Hazardous Waste Management Plan;

D. That the cumulative effects of locating the project within the proposed area have been analyzed and where applicable, measures that minimize adverse impacts to the surrounding community have been incorporated into the project.

### 17.72.085 - Rosorved.

Editor's note-

17.72.090 - Limitations on Signs.

A. - Gonoral Limitations. All Signs-shall be subject to the applicable-limitations set-forth in Sections 47.104.020 and 47.104.060

17.72.100 -- Minimum lot frontago.

Every lot shall have a minimum frontage of 25 feot upon a street, except ac this requirement is modified by Section 17.106.020.

(Prior-planning-code-§-5814)

17.72.110 - Maximum height.

Except as provided in Chapter 17.128, no general maximum height is prescribed, except that the height of facilities shall be limited, as proscribed in Section 17.108.010, on lots lying along a boundary of any of certain other zones. But see Section 17.72.090 for maximum height of Signs, and Section 17.108.130 for maximum height of facilities within minimum yards and courts:

### 17.72.120 - Minimum-yards.

No yards are generally required except as indicated below. The following minimum yards shall be provided unobstructed except for the accessory structures or the other facilities allowed therein by Sestion 17.108.130.

A.—\_\_\_\_Front-Yard. A front-yard-shall-be provided, as prescribed in Section 17.108.040, in certain situations where part of the frontage on the same side of a block is in a residential-zone.

B.---Sido Yard-Street Sido of Corner Lot. A side-yard-shall-be provided, as proscribed in Section 17.108.070, on the street side of a corner lot in certain-situations where a lot to the rear of the corner lot is in a residential zone.

C. --Side Yard--Interior Lot Line. A side-yard-shall be provided, as in Soction 17.108.090, along an interior side-lot tine tying along a boundary of any of certain other zones.

D. Roar Yard: A rear yard-shall be provided, as in Section 17.108.100, along an interior side lot line lying along a boundary of any of certain ether zones.

(Prior-planning code § 5820)

17.72.130 - Bufforing.

All-uses-chall bo subject to the applicable requiromonts of the buffering regulations in Chapter 17.110 with respect to screening or location of parking, loading, and-storage areas; control of artificial illumination; and other matters specified therein. See also Section 17.72.070(D).

(Prior-planning-code-§ 5822)

17.72.140 - Other zoning provisions.

A. Parking and Loading. Off-street parking and loading-shall be provided as prescribed in the offstreet parking and loading requirements in Chaptor 17.116

B.Bicycle-Parking. Bicycle-parking-shall be provided as proccribed in the bicycle-parking regulations in Chapter 17.117

C. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming uso regulations in Chapter 17.114

D.General Provicions. The general exceptions and other rogulations set forth in Chapter-17-102 shall apply in the M-40-zone.

E.Recycling-Space Allocation Requirements. The regulations set forth in Chapter 17:118 chall apply in M-40-zone.

# Chapter 17.73 - CIX-1, CIX-2, IG AND IO INDUSTRIAL ZONES REGULATIONS

Sections:

17.73.010 - Title, purpose, and applicability.

17.73.015 - Required Design Review Process

17.73.020 - Permitted and conditionally permitted uses and facilities.

17.73.030 - Property Development Standards.

17.73.035 - Special regulations for <u>Recycling and Waste-Related Industrial Activities -- Primary</u> <u>Recycling Collection Centers sprimary-collection-contors-in the CIX, IG, and IO</u> industrial zones.

17.73.040 - Special regulations for work/live units in the CIX, IG, and IO industrial zones

17.73.050 - Parking and loading dock restrictions.

17.73.060 - Referral to other applicable regulations.

17.73.070 - Other zoning provisions.

17.73.010 - Title, purpose, and applicability.

The provisions of this chapter shall be known as the <u>CIX, IG, and IO</u> industrial zones regulations. This chapter establishes regulations for the (CIX-1) Commercial Industrial Mix-1, (CIX-2) Commercial Industrial Mix-2, (IG) General Industrial, and (IO) Industrial Office.

These <u>CIX, IG, and IO</u> industrial zoning districts are intended to create, preserve, and enhance areas for industrial uses, including manufacturing, scientific and product-related research and development, construction, transportation, warehousing/storage/distribution, recycling/waste-related activities, clean technology, and similar uses. The primary purposes of these areas are to support Oakland's economic base and to provide employment opportunities. The specific purposes of these industrial districts are to:

- 1. Provide a diversified economic base and a wide range of employment opportunities;
- 2. Maximize Oakland's regional role as a transportation, distribution, and communications hub;
- 3. Support Port operations and expansion by providing land for Port services such as trucking, warehousing, and distribution;
- Preserve areas with good freeway, rail, seaport, and/or airport access for business and industrial uses;
- 5. Prohibit residential uses and limit commercial uses in General Industrial (IG) areas so that a maximum amount of the City's land base is preserved for industrial uses, and so that industrial uses may operate without impacting those activities;
- 6. Locate high impact industrial uses away from residential areas; and
- Allow heavy-impact or large scale commercial retail uses on sites with direct access to the regional transportation system.
- A. CIX-1 Commercial Industrial Mix 1 Zone. The CIX-1 zone is intended to create, preserve, and enhance the-industrial areas, including but not limited to of-West Oakland, that are appropriate for a wide variety of businesses and related commercial and industrial establishments. This zone is intended to accommodate existing older industries and provide flexibility in order to anticipate new technologies. Large-scale commercial and retail uses will be limited to sites with direct access to the regional transportation system.
- B. CIX-2 Commercial Industrial Mix 2 Zone. The CIX-2 zone is intended to create, preserve, and enhance <u>industrial</u> areas, <u>including but not limited to</u> of the Central and Eastern portions of the City, that are appropriate for a wide variety of heavy commercial and industrial establishments.

Uses with greater off-site impacts may be permitted provided they meet specific performance standards and are buffered from residential areas.

- C. IG General Industrial Zone. The IG zone is intended to create, preserve and enhance areas of the City that are appropriate for a wide variety of businesses and related commercial and industrial establishments that may have the potential to generate off-site impacts such as noise, light/glare, odor, and traffic. This zone allows heavy industrial and manufacturing uses, transportation facilities, warehousing and distribution, and similar and related supporting uses. Uses that may inhibit such uses, or the expansion thereof, are prohibited. This district is applied to areas with good freeway, rail, seaport, and/or airport access.
- D. 10 Industrial Office Zone. The IO zone is intended to create and support areas of the City that are appropriate for a wide variety of businesses and related commercial and industrial establishments in a campus-style setting. Development and performance standards in this district are more restrictive and accommodate large-parcel development in an attractive, welllandscaped setting. Future development shall reflect large-scale office, research and development, light industrial, wholesaling and distribution, and similar and related supporting uses.

# 17.73.015 Required Design Review Process

- A. Except for projects that are exempt from design review as set forth in Section 17.136.025, no Residential Facility, Designated Historic Property, Potentially Designated Historic Property, Telecommunications Facility, Sign, or other associated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the Telecommunications regulations in Chapter 17.128, or the Sign regulations in Chapter 17.104.
- B. No facility located within one hundred fifty (150) feet of any residential zone boundary and accommodating the following activities shall be constructed, established, or expanded in size unless plans for the proposal have been approved pursuant to the Regular Design Review procedure in Chapter 17.136:
  - 1. Automobile and Other Light Vehicle Gas Station and Servicing Activity.

2. Automobile and Other Light Vehicle Repair and Cleaning Activity.

3. Freight/Truck Terminal.

4. Truck Yard.

5. Truck Weigh Stations.

- 6. Truck and Other Heavy Vehicle Sales, Rental, and Leasing.
- 7. Truck and Other Heavy Vehicle Service, Repair, and Refueling.
- C. Establishment of a work/live unit shall only be permitted upon determination that the proposal conforms to the regular design review criteria set forth in the Regular Design Review procedure in Chapter 17.136 and to all of the additional criteria set forth in Section 17.73.040(D).

17.73.020 - Permitted and conditionally permitted uses and facilities.

The following table lists the permitted, conditionally permitted, and prohibited uses and facilities in the CIX-1, CIX-2, IG and IO zones. The descriptions of these uses are contained in Chapter 17.10.

"P" designates permitted uses and facilities in the corresponding zone.

"C" designates uses and facilities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134) in the corresponding zone.

"L" designates uses and facilities subject to certain limitations listed at the bottom of the Table.

"--- " designates uses and facilities that are prohibited in the corresponding zone.

Uses Zones Additional Regulations CIX-1 10 CIX-2 IG Residential Uses All residential uses prohibited in each zone Civic Uses Essential Service GΡ GР CP GP Limited Child-Care \_ \_ \_ Р С С Community Assembly \_ Recreational Assembly Р С С Р С Community Education С Р Nonassembly Cultural С С P С С Administrative \_ P Health Care \_\_\_\_\_ \_ С See Section 17.103.020 Special Health Care С \_ Utility and Vehicular Ρ С С С С С С С Extensive Impact . Commercial Uses Ρ C(L1) P(L1) General Food Sales C(L1) Р P(L1) Full Service Restaurant C(L1) C(L1) Limited Service Р C(L1) P(L1) C(L1) Restaurant and Cafe С С С Fast Food Restaurant . See Section-17-102-210 17.103.030 С С С Convenience Market C See Sections 17,103.030 and 17,114.030 Alcoholic Beverage Sales <u>L2</u> \_ \_ Mechanical or Electronic L3 = \_ -----<u>Games</u> Р Medical Service С С Р General Retail Sales In the CIX-2 and 10 zones. Retail allowed as an accessory use only per Section 17.10.040 Large-Scale Combined \_ \_ Retail and Grocery Sales Consumer Service P Ρ \_\_\_\_ С See Section 17.102.170 for special reaulations relating to massage services and Section 17.102,450 for special regulations related to laundromats. Consultative and P **Financial Service** Check Cashier and Check Cashing Consumer Cleaning and Р С Repair Service Р С Consumer Dry Cleaning \_ Plant Group Assembly P(L7) C(L8) C(L8) C(L8) Personal Instruction and P(L7) C(L8) C(L8) C(L8) Improvement and Small

Table 17.73.020: Permitted and Conditionally Permitted Uses and Facilities

Scale Entertainment	<u> </u>			1	
Administrative	P	P	L9	P	
Business, Communication, and Media Service	P	P	P	P	· ·
Broadcasting and Recording Service	P	P	Ρ	Р	
. Research Service	P	Р	С	P	· · · · · · · · · · · · · · · · · · ·
General Wholesale Sales	P	Р	Ρ	P	No retail ancillary activities for this use allowed in IG or IO.
Transient Habitation	-		-	<b>—</b> ,	
Building Material Sales	L3	L3		-	
Automobile and Other Light Vehicle Sales and Rental	P	c	-	С	
Automobile and Other Light Vehicle Gas Station and Servicing	P	Ρ	P	-	Requires Regular Design Review approval lif located within 150 feet of any residential zone boundary-is pursuant-to-the-design-roview-procedure-in(see Chapter 17.136 for the Design Review procedure).
Automotive and Other Light Vehicle Repair and Cleaning	L3	L3	P		Requires Regular Design Review approval if located within 150 feet of any residential zone (see Chapter 17.136 for the Desian Review procedure). If located within (150) feet-of-any-residential-zone-boundary is pursuant-to-the-design-review-procedure-in-Chapter 17.136
Automotive Fee Parking	L3	P	Р	Р	· · · · · · · · · · · · · · · · · · ·
Animal Care	L4	с	с	—	· · · · · · · · · · · · · · · · · · ·
Animal Boarding	L4	c	с	<b> </b>	
Undertaking Service	P	С	с	-	
Industrial Uses	······································	·		·	
Custom Manufacturing	P	Р	P .	Р	
Light Manufacturing	P	Р	Р	Р	· · · · · · · · · · · · · · · · · · ·
General Manufacturing	L3	L3	Р	—	,
Heavy Manufacturing		<u> </u>	С		
Research and Development	P	Ρ	P	Р	
Construction Operations	L3	L3	L3	С	
Warehousing, Storage and	Distribution		· · · · · · · · ·		
A. General Warehousing, Storage and Distribution	P	Ρ	Р	P	No retail component of this use allowed in IG or IO. Also, Ssee Sections 17.73.060
B. General Outdoor Storage	C	L3	Ρ	Ρ.	· · · · · · · · · · · · · · · · · · ·
C. Self or Mini Storage	С	С	_	c	
D. Container Storage	-	L3	Р	<u> </u>	
E. Automotive Salvage and Junk Yards	-	-	L3	-	
Regional Freight & and Tran	sportation:				
A. Seaport		-	P	с	

Oakland, California, Code of Ordinances

,

Trucking ∧ Trucking-R	related-Activit	ies:			<u></u>
A. Freight/Truck Terminal	L5	L3	P	1-	The establishment of new or expanded trucking and
B. Truck Yard	L5	С	P	С	truckino-related activities requires Regular Design Review approval if located within 150 feet of any
C. Truck Weigh Stations	<u> </u>	Р	Р		residential zone (see Chapter 17, 136 for the Design
D. Truck & <u>and</u> Other Heavy Vehicle Sales, Rental & <u>and</u> Leasing	L6	P .	P	P	Review procedure). If leeated-within-(150)-feet-of-any residential-zone boundary is pursuant to the design review-procedure-in-Chapter-17-136
E. Truck & <u>and</u> Other Heavy Vehicle Service, Repair, and Refueling	L5	P	Р	<u></u>	
Recycling & and Waste-Was	ste-Related A	ctivities	· · · · · ·		
A. Satellite Recycling Collection Centers	С	С	C .	c	-
B. Primary <u>Recyclina</u> Collection Centers	L10	L10	L11	-	
Hazardous Materials Production, Storage & Waste <u>and Waste</u> -Related Activities					L12 See also Health & <u>and</u> Safety Protection Zone (S-19)
A. Small Scale Transfer and Storage	<u> </u>	С	С	-	
B. Industrial Transfer/Storage	<u> </u>	-	С	-	
C. Residuals Repositories	_	1-	С		
D. Oil and Gas Storage	<u> </u>	<u> _</u>	L2	-	
Agricultural and Extractive	uses				
Plant nursery	P	P	P		· · · · · · · · · · · · · · · · · · ·
Crop and animal raising		-C(L13)	C(L13)	1-	See-Section-17-102.220
Mining and Quarrying Extractive	-	-	С	-	See Chapter 17.155

Facility Types	Zones				Additional	
	CIX-1	CIX-2	IG	10	Regulations	
Residential Facilities	All residential uses prohibited in each zone					
Nonresidential Facilities				····	·	
Enclosed Nonresidential	P	Р	Р	Р	· · · · · · · · · · · · · · · · · · ·	
Open Nonresidential	P	P	Р	Р		
Sidewalk Cafe	c	C `	-	7_	See Section-1-7-1-02-335 17.103.090	
Drive-In Nonresidential			-	]_		
Drive-Through Nonresidential	C	С	С	C	See Section-17-102-200 17.103.100	
Shopping Center Facility			-	_		
Telecommunications						
Micro Telecommunications	P	Р	P	Р	See Chapter 17.128	
Mini Telecommunications	P	Р	P	P		
Macro Telecommunications	C	С	P	Р		

Oakland, California, Code of Ordinances

٢

Monopole Telecommunications	C	c	P	ĺΡ	
Tower Telecommunications	_	<del>.</del>	Р	Р	
Signs			•		
Residential Signs		—			See Chapter 17.104
Special Signs	P	P	Р	Р	
Development Signs	P	P			
Realty Signs	P	Р	Р	Р	
Civic Signs	Р	Р	Р	Р	
Business Signs	P	Р	Р	P	
Advertising Signs		<u> </u>			

# Limitations:

L1. Limited to location on a ground floor in CIX-2, IG and IO. Over <u>five thousand (5,000)</u> sf floor area requires a conditional use permit in CIX-2, IG, and IO.

L2. Prohibited within <u>three hundred (300)</u> feet of a residential zone and requires a conditional use permit elsewhere throughout the zone. (Conditional use permit is required in CIX-2).

L3. A conditional use permit is required if within <u>three hundred (300)</u> feet of a residential zone; Permitted if beyond <u>three hundred (300)</u> feet of a residential zone.

L5. Prohibited within six hundred (600) feet of a residential zone. A conditional use permit is required elsewhere throughout the zone. Also, only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) in the West Oakland Community Development District, defined to include all areas between Interstate 980 to the east, 3rd Street to the south. Interstate 880 to the west and Interstate 580 to the north.

L6. A conditional use permit is required: a) if within <u>three hundred (300)</u> feet of a residential zone, and b) if located anywhere in the district when outdoor repair and service activity exceeds <u>fifty percent (50%)</u> of site area. Also, only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) in the West Oakland Community Development District, defined to include all areas between Interstate 980 to the east, 3rd Street to the south, Interstate 880 to the west, and Interstate 580 to the north.

**L7.** A conditional use permit is required for entertainment uses. <u>Also, No new or expanded adult</u> <u>entertainment activity shall be located closer than one thousand (1.000) feet to the boundary of any</u> residential zone or three hundred (300) feet from any other adult entertainment activity. See Section <u>17.102.160</u> for further regulations regarding adult entertainment activities.

L8. Entertainment, educational and athletic services are not permitted.

L10. Prohibited within <u>three hundred (300)</u> feet of a residential zone; a conditional use permit containing requirements no less stringent than the performance standards set out in <u>Section</u> 17.73.035 is required if beyond <u>three hundred (300)</u> feet of a residential zone boundary.

L11. Prohibited within <u>three hundred (300)</u> feet of a residential zone, permitted outright beyond <u>three hundred (300)</u> feet with a standard set of performance standards that would apply to existing, new or expanded uses, as detailed in Section 17.73.035.

**L13.** Crop and Animal Raising is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure). In addition to the CUP criteria contained in <u>Section</u> 17.134.050, this activity must meet the following use permit criteria:

1. The proposal will not adversely affect the livability or appropriate development of abutting properties and the surrounding neighborhood in terms of noise, water and pesticide runoff, farming equipment operation, hours of operation, odor, security, and vehicular traffic;

2. Agricultural chemicals or pesticides will not impact abutting properties or the surrounding neighborhood; and

3. The soil used in growing does not contain any harmful contaminants and the activity will not create contaminated soil.

17.73.030 - Property Development Standards.

Table 17.73.030 contains the property development standards for all zones within this Chapter.

# Table 17.73.030

# Property Development Standards

Development Standards	Zones				
	CIX-1	CIX-2	lG	10	Regulations
Mjnjmum Lot Frontage	25 ft	25 ft	25 ft	100 ft	1
Minimum Lot Width	25 ft	25 ft	25 ft	100 ft	1
Minimum Lot Area (square feet)		10,000 sf	10,000 sf	25,000 sf	1
Floor-Area Ratio (FAR)					
Greater than 300 feet		4.0	2.0	4.0	
Within 300 feet of a residential zone boundary		2.0	1.0	2.0	2
Maximum Height	None	55 ft	None	55 ft	3,4
Minimum Front Yard Setback	0 ft	0 ft	0 ft	20 ft	5
Minjmum Rear Yard Setback	0 ft		5		
Minimum Interior Side Yard setback	O ft			5	
Minimum Street Side Yard Setback of a Corner Lot	10 ft	10 ft	10 ft	20 ft	5
Site Landscaping (% of lot area)	5%	5%	5%	15%	6, 7
Parking Lot Landscaping (%of lot area)	10%	10%	10%	10%	8
Street Trees	Required	Required	See also note 10	Required	9
Site and DriVeway Access - Minimum Distance from any residential or open space boundary	50 ft	50 ft	50 ft	50 ft	11
Drjveway Width Maximum	35 ft	35 ft	35 ft	35 ft -	12
Pedestrian Walkway	Required	Required	Required	Required	13
Minimum Fence Height in Yards adjacent to Residential or Open Space Zones	8 ft	8 ft	8 ft -	8 ft	14
Maximum Fence Height in Yards adjacent to Residential or Open Space Zones	15 ft	15 ft	15 ft	15 ft	14, 15

### Additional Regulations Noted in Table 17.73.030

2. A conditional use permit to exceed the permitted Floor Area Ratio (FAR) may be allowed, as shown in parentheses above in Table 17.73.030, upon determination that the proposal conforms to the conditional use permit criteria set forth in the conditional use permit procedure in Chapter 17.134; and to all of the following additional criteria:

a. Additional intensity does not subject residentially zoned areas within <u>three hundred (300)</u> feet to significant adverse impacts related to: truck traffic; nighttime operations; noise; vehicular traffic; hazardous materials exposure and risk; air emissions; blockage of sunlight to private open space areas; or other such environmental impacts;

b. The site is located on a major arterial, freeway, rail line or other location that has adequate capacity to handle the intensity and type of traffic volume.

c. If adjacent to a residential or open space zone boundary the proposed development has a step back of one foot to every one foot of height, beginning with a maximum height of <u>thirty (30)</u> feet at all required yard setbacks; and

d. All new development activities meet the Performance Standards in 17.120.

5. See <u>Section\_17-108.040,-17-108.070,-17-108.090,-17-108-100,-17-108-110, and</u>-17.108.130 for minimum front, side, and rear yards in commercial and industrial zones which may be across from, abut or be adjacent to a residential zone or alley. Accessory structures or other facilities allowed within the yards and setbacks <u>are is-in sSections 17.108.130</u>.

**6.** All new projects which involve the construction of a new building, or the expansion or replacement of existing building footprint by more than twenty percent (20%) such that the floor area to site ratio exceeds thirty\_-five percent (35%), shall comply with the landscape requirements. Landscaping shall consist of pervious surface with lawn, ground cover, shrubs, permeable paving materials, and/or trees and which is irrigated and maintained. See also <u>Chapter</u> 17.124 Landscaping and Screening Standards.

8. Parking Lot Landscaping applies only to lots associated with new construction with more than <u>twenty-five thousand (25,000)</u> sf floor area. Shade trees shall be provided at a ratio of 1 tree for every 10 spaces through the parking lot. A minimum of <u>ten 40-percent (10%)</u> of a surface parking lot shall be landscaped accompanied by an irrigation system that is permanent, below grade and activated by automatic timing controls which may be provided entirely in permeable surfacing in lieu of irrigated landscaping if approved through <u>the Regular</u> Design Review <u>process</u> (see <u>Chapter 17.136</u>). Parking lots located adjacent to a public right-of-way shall include screening consisting of a minimum of five (5) foot deep planted area or a three (3) foot tall opaque, concrete, or masonry wall. Chain link, cyclone, and barbed wire fencing is prohibited in all cases.

**9.** For all projects requiring a building permit, street trees are required. In addition to the general landscaping requirements set forth above, a minimum of one fifteen-gallon tree, or substantially equivalent landscaping consistent with City policy and as approved by the Director of City Planning, shall be provided for every twenty (20) feet of street frontage or portion thereof and, if a curbside planting strip exists, for every twenty-five (25) feet of street frontage. On streets with sidewalks where the distance from the face of the curb to the outer edge of the sidewalk is at least six and one-half  $(6\frac{1}{2})$  feet, the trees to be provided shall include street trees to the satisfaction of the Tree Division.

11. Applies to new development; or expansion of industrial or commercial buildings by more than <u>twenty</u> 20-percent (20%) floor area; or b) addition or expansion of an existing building so that the building to land ratio exceeds <u>thirty-five\_35</u>-percent (35%), which ever is greater; and all new driveway projects. This requirement may be waived administratively if such distance requirement will impede direct access to a rail line.

12. Driveway shall not exceed <u>thirty-five</u> 35–(35) feet in width without obtaining approval from the Engineering Department of Building Services through the Driveway Appeal Process.

13. A cleariy defined and lighted walkway, at least four (4) feet wide, shall be provided between the main building entry and a public sidewalk for all new development. On-site walkways shall be separated from on-site automobile circulation and parking areas by landscaping, a change in paving material, or a change in elevation.

14. Applies to all property lines in industrial zones, except those fronting a public street, which directly abut a residential or open space zone. All buffering Requirements apply to: a) new development; or expansion of an industrial or commercial building by more than twenty 20-percent (20%) floor area, or b) addition or expansion of an existing building so that the building to land ratio exceeds thirty-five\_35 percent (35%), whichever is greater.

17.73.035 - Special regulations for <u>Recycling and Waste-Related Industrial Activities -- Primary</u> <u>Recycling Collection Centers primary-collection-oonters in the CIX, IG, and IO industrial zones.</u>

- A. Applicability. This Section applies to <u>Recycling and Waste-Related Industrial Activities --</u> Primary Collection Centers, as-defined in 17:10.586."Recycling and Waste-Related Industrial Activities," that are located in the Commercial Industrial Mix-1 (CIX-1), Commercial Industrial Mix-2 (CIX-2) or General Industrial (IG) zone. Conditional use permits issued for operations In CIX zones must contain conditions no less stringent than the performance standards set out in this Section. Where there is any apparent conflict between these regulations and regulations contained elsewhere in Title. 17 of the Oakland Municipal Code, and/or with conditions of approval, the more stringent shall govern.
- B. Performance Standards. In addition to the performance standards set forth in Chapter 17.120, the following minimum performance standards shall be uniformly applied, as applicable, to all Primary Recycling Collection Centers.
  - 1. Site Design and Layout For new and expanded uses, submittal and approval of the following plans, and implementation of approved plans shall be required:
    - Site and floor plans, which shall include designated areas for separation and disposal of materials, as well as required fencing/walls, to the Planning and Zoning and Building Services Divisions;
    - b. Building plans to the Fire Services Division;
    - c. Fire safety/emergency plan to the Fire Services Division.
  - Signage. For existing, new or expanded uses: identification, directional and informational signs shall be provided on site in conformance with Chapter 17.104 General Limitation on Signs and with the small project design review procedure in Chapter 17.136. At a minimum, the following information shall be posted near the entrance(s) and/or perimeter of the facility:
    - a. Business Identification, 24-hour contact information of facility operator;
    - b. Hours of operation;
    - c. Signage prohibiting the delivery or drop off of material to be recycled after-hours;
    - d. Signage prohibiting illegal dumping, littering loitering or sleeping in proximity of the site's perimeter,
    - A map of authorized truck routes to the facility posted at the office or scale house (and available to customers);
    - f. A list of accepted and/or non-accepted materials for recycling.

- 3. Appearance and Design.
  - a. Landscaping.
    - i) For existing, new or expanded uses, all required planting shall be permanently maintained in good growing condition and, whenever necessary, replaced with new plant materials to ensure continued compliance with applicable landscaping requirements. All required irrigation systems shall be permanently maintained in good condition, and, whenever necessary, repaired or replaced.
    - ii) For new or expanded uses, submittal and approval of a landscape and irrigation maintenance plan and/or street tree plan, and implementation of approved plan for new and expanded uses, as required by the Planning Director or his/her designee;
  - b. Screening. For existing, new and expanded uses, screening by a solid fences and/or walls shall be required around the entire site;
  - c. Lighting.
    - i) For new or expanded uses, submittal and approval of lighting plans, and implementation of approved plans, to the Planning and Zoning Division and the Electrical Services Division of the Public Works Agency shall be required. The proposed lighting fixtures shall be adequately shielded to a point below the light bulb and reflector so as to prevent unnecessary glare onto adjacent properties or public streets.
    - For existing uses, lighting shall comply with the performance standards of Section 17.120.100 of the Oakland Planning Code relating to glare. Lighting shall be so operated as to not adversely affect nearby properties or public streets.
- 4. Noise, Vibration and Other Applicable Health and Safety Regulations. For existing, new or expanded uses:
  - a. Noise levels from the activity, property, or any mechanical equipment on site shall comply with the performance standards of Section 17.120 of the Oakland Planning Code and Section 8.18 of the Oakland Municipal Code. If noise levels exceed these standards, the activity causing the noise shall be abated until appropriate noise reduction measures have been installed and compliance verified by the Planning and Zoning and Building Services Divisions;
  - b. Vibration levels from the activity, property, or any mechanical equipment on site shall comply with the performance standards of Section 17.120 of the Oakland Planning Code. If vibration levels exceed these standards, the activity causing the vibration shall be abated until appropriate vibration reduction measures have been installed and compliance verified by the Planning and Zoning Division and Building Services;
  - c. The project operator/applicant shall comply with State and other regional bodies and/or applicable regulations including, but not limited to, the federal Clean Water Act and Occupational Safety and Health Administration (OSHA), the California Penal Code Section 496. (a), the Environmental Protection Agency (EPA), the Bay Area Air Quality Management District (BAAQMD) and Best Management Practices (BMP) for stormwater.
- 5. Litter, Debris, Graffiti and Cleanliness. For existing, new or expanded uses:
  - a. The site shall be maintained in a clean and orderly condition, free of vectors, and free of standing water and any odiferous waste;
  - b. The public right-of-way shall not be used for storage or processing of materials;
  - c. Graffiti shall be removed within seventy-two (72) hours of application;
  - d. A cleanliness/litter management and control plan shall be developed, implemented and maintained, such that it is ready for inspection. The plan shall include provisions for the disposal of recycling related litter and debris in the public right-of-way within the area

comprised of all streets adjacent to the premises, and the one-block extension of those streets to the north and south, and east and west, respectively (See Figure 17.73.01). This would not include material illegally dumped that is not related to the recycling operation, including but not limited to hazardous material, containers of paint or unidentified liquids, tree trimmings, residential, commercial and/or industrial waste or dumping of materials net accepted by the Primary <u>Recycling</u> Collection Center. In addition, the Primary <u>Recycling</u> Collection Center shall produce a notice to distribute to customers that states that all illegal dumping shall be reported to City authorities.

- e. A site/immediate neighborhood shopping cart management plan shall be developed, implemented and maintained, such that it is ready for inspection. If the Primary <u>Recycling</u> Collection Center accepts materials from the public brought by means of a shopping cart, it shall be responsible for the retrieval of all shopping carts within the area comprised of all streets adjacent to the premises, and the one-block extension of those streets to the north and south, and east and west, respectively (See Figure 17.73.01). Additionally, a Primary <u>Recycling</u> Collection Center shall post signage that includes contact information to report abandoned shopping carts in the vicinity of the facility; if called or notified by a member of the public about abandoned shopping carts located within a two-block radius of the premises, a Primary <u>Recycling</u> Collection Center shall street sides of the twenty-five- (25) square block area that includes the block on which the premises is located (as the center block of the twenty-five (25) square block area) See Figure 17.73.01).
- f. A loitering deterrence plan shall be developed, implemented and maintained, such that is ready for inspection;
- 6. Circulation. For new or expanded uses submittal and approval of the following plans, and implementation of approved plans are required:
  - a. A circulation plan that shows ingress and egress, parking both on-site and off-street, as well as includes provisions for any needed staff to monitor on-site traffic operations, submitted to the Transportation Services Division;
  - b. A plan showing rail loading and unloading within site shall be required (as applicable) submitted to the Transportation Services Division.
- 7. Equipment and Facilities. For existing, new or expanded uses:
  - a. There shall be no exterior pay telephones located at the site;
  - b. All equipment shall be maintained and kept in good working order;
    - c. After business hours, all facility-owned vehicles shall be stored within the facility or at an appropriate alternative off-street location.
- 8. Operations. All existing, new or expanded uses:
  - a. Shall have a representative attend Neighborhood Crime Prevention Council meetings—a minimum of two meetings per year or more frequently if items pertaining to their facility are on the agenda—for their community policing beat with the sole purpose of addressing and responding to community complaints. For the purposes of this provision said representative will mean a site or company manager with sufficient authority to address the concerns of neighbors;
  - b. Shall maintain a 24<sub>2</sub>-hour "hotline" where neighbors can log complaints regarding nuisance activity associated with or emanating from the recycling facility. Complaints logs shall be maintained and made available to the City for inspection/copying upon reasonable nofice;



Figure 17.73.01: Illustration of Extent of Area Primary <u>Recycling</u> Collection Centers are responsible for Litter/Garbage/Debris Removal and Shopping Cart Retrieval

c. Shall provide staff and training for traffic operations needed on-site, as required by the Transportation Services Division as part of any circulation plan;

- d. Shall develop, implement and maintain a plan for the disposal and containment of nonrecoverable materials that is ready for inspection; submittal and approval of such a plan prior to operation shall be required for new or expanded uses;
- e. Shall keep all entrance gates closed and locked when the <u>Pprimary Rrecycling C</u>collection facility is not open to the public;
- f. Shall not burn insulation from copper wire as a means to increase the material's value or for any other purpose.
- Relief from Performance Standards. Any person who owns or operates, or who has applied to C. construct, expand, modify or establish an activity or facility that involves Primary Recycling Collection Centers which would be affected by the performance standards required, and who contends that the performance standards as applied to him or her would be unlawful under Federal, State, or local law or regulation, may submit a written application to the Planning Director requesting relief from the performance standards within ten (10) 10-(ten)-days of being initially notified of the performance standards. For purposes of this section, notice to a predecessor in interest shall constitute such initial notice to subsequent owners/operators. The written request for relief from these performance standards must: (a) identify the name and address of the applicant and business: (b) the affected application number; (c) specifically state how the performance standards as applied to him or her would be unlawful under Federal. State, or local law or regulation; and (d) include all appropriate legal and factual support for the request for relief. Within thirty (30) days of receipt of the completed request for relief, the Planning Director, or his/her designee, shall mail to the applicant a written determination. The applicant may appeal such determination pursuant to the provisions in Oakland Planning Code chapter 17.132,

17.73.040 - Special regulations for work/live units in the CIX, IG, and IO industrial zones

- C. Conditional use permit required.
  - Establishment of a work/live unit for new construction is only permitted upon determination that the proposal conforms to the conditional use permit criteria set forth in the conditional use permit procedure in Chapter 17.134 on lots that are both: (1) in the CIX-1 or CIX-2 zones, and (2) within three\_hundred (300) feet of a residential zone.
  - Establishment of a work/live unit through the conversion of an existing building originally designed for commercial or industrial activities is permitted in all industrial zones with the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134 provided there are existing artist and/or artisan residents who meet the requirements of Zoning Code Bulletin regarding "live/work" (issued August 29, 2001 and amended August 23, 2004).
- K. The development of work/live units In the industrial zones shall not be considered adding housing units to the City's rental supply, nor does it create "conversion rights" under the City's condominium conversion ordinance, O.M.C. Chapter 16.36, nor are the development standards for work/live units intended to be a circumvention of the requirements of the City's condominium conversion ordinance, O.M.C. Chapter 16.36.

17.73.050 - Parking and loading dock restrictions.

- A. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116.
- B. Parking for new development shall be located at the rear of the site or at the side of the building in the CIX-1, CIX-2, and IO except for drop-off areas, which may be at the entry, except where access to existing loading docks and/or rail lines is required. nNew truck loading docks shall not be located closer than fifty (50) feet from property line as measured from the subject dock to any property boundary if located within three hundred (300) feet of a residential zone, unless such a distance

requirement will impede direct access to a rail line. Truck docks shall be located such that trucks do not encroach into the public right of way. All existing loading docks are not subject to this requirement.

# 17.73.060 - Referral to other applicable regulations.

The following table contains referrals to other regulations that may apply:

Subject	Section	
Required number, dimensions, and location of parking spaces; maneuvering aisle dimensions, and related regulations		
Sign regulations		
Buffering regulations, including the buffering of parking, loading, glare, and storage from other properties	17.110	
Landscaping and screening, including street trees	17.66.140A 17.103.040 17.124 17.68.130	
Recycling space requirements	17.118	
Nonconforming uses and facilities	17.114	
Joint living and working quarters	17.102.190	
Performance standards regarding the control of noise, odor, smoke, and other objectionable impacts	17.120	
The demolition of living units and the conversion of a living unit to a Nonresidential Use	17.102.230	
Accessory Uses	17.10.040	
Fence and retaining wall standards, including location, height, and materials	·I <del>7.</del> I08.040	
Expanding a use into adjacent zones	17.102.110	
Application of zoning regulations to lots divided by zone boundaries	1-7-102-070 17.154.060	
Landmarks	17.05	
Special-Restrictiono-on-Establishmentc-Solling-Alcoholic-Beverages	17-102-210	
Regulations applying to tobacco-oriented activities		
Microwave dishes and energy production facilities		
Special regulations applying to adult entertainment activities		
Special regulations applying to massage service activities		
Buffering regulations for lots with three or more required parking space. This includes the screening of parking, loading, glare, and storage from residential properties and zones		
Buffer Regulations for commercial and industrial uses next to residential and open space zones		
Special regulations applying to electroplating activities		
S-19 Health and Safety Protection Overlay Zone	17.100A	

17.74.040 - Conditionally permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134;

A. Residential Activities:

Residential Care, except when occupying a One-Family Dwelling Residential Facility

Service-Enriched Permanent Housing

Transitional Housing

Emergency Shelter

B. Civic Activities:

Administrative

Utility and Vehicular

Community Education

Special Health Care Civic Activities (see Section 17.103.020)

C. Commercial Activities:

General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Fast Food Restaurant

Convenience Market

Alcoholic Beverage Sales

Consumer Service (see Section 17,102.170 for special regulations relating to massage services and Section 17,102.450 for special regulations related to laundromats)

Consultative and Financial Service

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

Administrative

Business, Communication, and Media Service-

Broadcasting and Recording Service

Research Service

Automotive Fee Parking

Animal Care

Animal Bearding

- D. Off-street parking serving activities other than those listed above or in Section 17.74.030, subject to the conditions set forth in Section-17.102.100 17.116.075.
- E. Activities that are listed neither as permitted nor conditionally permitted, but are permitted or conditionally permitted on nearby lots in an adjacent zone, subject to the conditions set forth in Section 17.102.110.

\_E...-Additional-activities-which-aro-permitted-or-conditionally-permitted-in-an-adjacent-zone,-on-lots near-the-boundary-thereof,-subject-to-tho-conditions-set-forth-in-Section-17-102-110

17.74.050 - Permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Facilities:

One-Family Dwelling

One-Family Dwelling with Secondary Unit, subject to the provisions specified in Section 47-102-360 17.103.080

Two-Family Dwelling

Multifamily Dwelling

Rooming House

- B. Nonresidential Facilities: Enclosed
- C. Signs:

Residential

Special

Development

Realty

Civic

Business

D. Telecommunications:

Micro, except when a Major Conditional Use Permit is required by Section 17.128.025

Mini, except when a Major Conditional Use Permit is required by Section 17.128.025

17.74.070 - Special regulations applying to certain Commercial Activities.

All conditionally permitted Commercial Activities other than Automotive Fee Parking shall, except for accessory off-street parking and loading and maintenance of accessory landscaping and screening, be

conducted entirely within enclosed buildings which are primarily occupied by permitted activities. See also Section-17-102-2-10\_17\_103.030.

17.74.090 - Use permit criteria for Commercial Activities.

B. That the proposed activity will not create or aggravate traffic congestion or interfere with the movement of traffic generated by permitted activities.

See also Section-17-102-210 17.103.030.

17.74.120 - Maximum residential density.

- B. Conditionally Permitted Density. The number of living units permitted by subsection A of this section may be increased by not to exceed fifty (50)-percent (50%) upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, in each of the following situations:
  - 1. In the case of a Residential Facility with more than four stories containing living units, subject to the provisions of Section 17.106.040.
  - Upon the acquisition of development rights from nearby lots, subject to the provisions of Section 17.106.050.

The number of living units may also be increased, as prescribed in Section 17.106.060, in certain special housing.

17.74.150 - Minimum yards and courts.

- B. Side Yard—Street Side of Corner Lot. The minimum side yard width on the street side of every corner lot shall be ten (10) feet.
- D. Rear Yard. The minimum rear yard depth on every lot shall be ten (10) feet. except-as a lesser depth-is-allowed-by-Soction-17-108-110
- E. Courts. On each let containing a Residential Facility, courts shall be provided when and as required by Section 17.108.120.

17.74.180 - Special regulations for mini-lot and planned unit developments.

A. Mini-Lot Developments. In mini-lot developments, certain of the regulations otherwise applying to individual lots in the S-1 zone may be waived or modified when and as prescribed in-Section .17-102-320 Chapter 17.142.

17.76.060 - Conditionally permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134:

A. Residential Activities:

Residential Care, except when occupying a One-Family Dwelling Residential Facility

Service-Enriched Permanent Housing

Transitional Housing

Emergency Shelter

B. Civic Activities:

Health Care

Utility and Vehicular

Extensive Impact

- C. Commercial Activities:
  - General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Fast-Food Restaurant

Convenience Market

Alcoholic Beverage Sales

General Retail Sales

Consumer Service (see Section 17.102.170 for special regulations relating to massage services and Section 17.102.450 for special regulations related to laundromats)

**Consumer Cleaning and Repair Service** 

Consumer Dry Cleaning Plant

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

Business, Communication, and Media Service

Broadcasting and Recording Service

**Research Service** 

Transient Habitation (see Section 17.103.050)

Automobile and Other Light Vehicle Gas Station and Servicing
Automotive Fee Parking

D. Off-street parking serving activities other than those listed above or in Section 17.76.050, subject to the conditions set forth in Section-17-102-100 17.116.075.

E. Activities that are listed neither as permitted nor conditionally permitted but are permitted or conditionally permitted on nearby lots in an adjacent zone, subject to the conditions set forth in Section 17.102.110. E. Additional activities which are permitted or conditionally permitted in an adjacent-zone, on lots near-the-boundary-thereof, subject-to-the-conditions sot-forth-in-Section 17.102.110

17.76.070 - Permitted facilities.

The following facilities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Facilities:

One-Family Dwelling

One-Family Dwelling with Secondary Unit, subject to the provisions specified in Section 47-102-360\_17.103.080

Two-Family Dwelling

Multifamily Dwelling

Rooming House

B. Nonresidential Facilities:

Enclosed

C. Signs:

Residential

Special

Development

Realty

Civic

Business

D. Telecommunications:

Micro, except when a Major Conditional Use Permit is required by Section 17.128.025

Mini, except when a Major Conditional Use Permit is required by Section 17.128.025

17.76.090 - Special regulations applying to certain Commercial Activities.

All conditionally permitted Commercial Activities other than Automotive Fee Parking shall, except for open-air dining facilities, accessory off-street parking and loading, and maintenance of accessory landscaping and screening, be conducted entirely within enclosed buildings. See also Section-17-102-210 17.103.030.

17.76.140 - Maximum residential density.

- B. Conditionally Permitted Density. The number of living units permitted by subsection A of this section may be increased by not to exceed fifty (50)-percent (50%) upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, in each of the following situations:
  - 1. In the case of a Residential Facility with more than four stories containing living units, subject to the provisions of Section 17.106.040.
  - Upon the acquisition of development rights from nearby lots, subject to the provisions of Section 17.106.050.

The number of living units may also be increased, as prescribed in Section 17.106.060, in certain special housing.

17.76.200 - Special regulations for mini-lot developments, planned unit developments, and large-scale developments.

- A. Mini-Lot Developments. In mini-lot developments, certain of the regulations otherwise applying to individual lots in the S-2 zone may be waived or modified when and as prescribed in-Section 17-102-320 Chapter 17.142.
- C. Large-Scale Developments. No development which involves more than <u>one hundred thousand</u> (100,000) square feet of new floor area, or a new building or portion thereof of more than <u>one hundred twenty</u> (120) feet in height, shall be permitted except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134. This requirement shall not apply to development where a valid planned unit development permit is in effect.

17.76.210 - Other zoning provisions.

- A. Parking and Loading. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116.
- B. Bicycle Parking. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117.
- C. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- D. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.

#### 17.78.120 - Maximum height.

Except for the projections allowed by Section 17.108.030, and except as provided in Chapter 17.128, no building shall exceed <u>forty-five (45)</u> feet in height unless the building, or that portion thereof which exceeds <u>forty-five (45)</u> feet in height, is set back from the inner line of each of the rhinimum side yards required by Section 17.78.130C a minimum horizontal distance equal to one (1) foot for each two (2) feet by which it exceeds <u>forty-five (45)</u> feet in height. See Section 17.78.090 for maximum height of Signs, and Section 17.108.130 for maximum height of facilities within minimum yards.

17.78.130 - Minimum yards.

- A. Front Yard. The minimum front yard depth on every lot shall be fifty (50) feet
- B. Side Yard—Street Side of Corner Lot. The minimum side yard width on the street side of every corner lot shall be <u>fifty (50)</u> feet
- C. Side Yard—Interior Lot Line. The minimum width of the side yard along any single interior side lot line of any lot shall be <u>twenty (20)</u> feet. The minimum combined width of both such side yards shall be <u>fifty (50)</u> feet.
- D. Rear Yard. The minimum rear yard depth on every lot shall be <u>thirty (30)</u> feet, except as a lessor depth is allowed by Section 17-108-110 and except that the minimum rear yard depth shall be <u>fifty (50)</u> feet along any portion of a rear lot line which abuts a lot in any residential zone.

17.78.140 - Buffering and landscaping.

C. Landscaping Coverage. A minimum of <u>forty 40</u>-percent (40%) of the lot area of each lot shall be developed with lawn, ground cover, garden, shrubs, or trees, subject to the standards for required landscaping and screening.

#### 17.84.030 - Required design review process.

- A. Except for projects that are exempt from design review as set forth in Section 17.136.025, no <u>Building Facility</u>. Designated Historic Property, Potentially Designated Historic Property, Building Facility, Telecommunications Facility, Sign, or other associated structure shall he constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the additional provisions in Sections 17.84.040, 17.84.050, and 17.84.060; the Telecommunications regulations in Chapter 17.128; or the Sign regulations in Chapter 17.104.
- B. Section 17.136.075 contains design review criteria for the demolition or removal of Designated Historic Properties (DHPs) and Potentially Designated Historic Properties (PDHPs).

### 17.92.060 - Limitations on residential density.

- A Overall Density. The maximum overall number of dwelling units within any development shall be whichever of the following is applicable and lowest
  - 1. The number of dwelling units implied by the applicable basic zone's minimum lot area requirement, and detined in the same manner as prescribed in subsection A of Section 17.142.110 and the tirst three sentences of subsection B of Section 17.142.110.
  - 2. In the case of those properties for which the Site Development Map of the North Oakland Hill Area Specific Plan depicts siting of dwelling units, the number of dwelling units indicated by that map.
  - In the case of those properties where dwelling units are not shown on the Site Development Map of the North Oakland Hill Area Specific Plan, the lowest number of dwelling units derived from:
    - a. Dividing the street frontage of the property by the minimum lot width requirement in the respective residential zone; and
    - b. Counting the number of legally platted lots within the proposed development area; and
    - c. Analyzing the project under the regular design review process to affirm or lower the maximum theoretical density pursuant to Sections 17.92.030 and 17.92.050.

One through three above shall not be deemed to preclude such additional secondary units in the S-11 zone as may be approved in accordance with the standards, criteria and conditions in Section 17-102-360\_17.103.080.

17.94.040 - Off-street parking regulations---Residential Activities.

A. Basic Requirement. One off-street parking space shall be provided for each three habitable rooms in the facility, as determined by the definition of "habitable rooms, number of" in accordance-with Section-17-102-280 17.09.040 and rounded to a whole number in accordance with the rules of Section 17.116.050, or the same number of spaces as required by the underlying base zone pursuant to Section 17.116.060 without regard to the provisions of the S-12 regulations, whichever is greater. Such parking shall be designated and permanently maintained for the use of residents of the facility.

#### 17.94.050 - Standards for required parking facilities.

The standards for required parking and leading facilities set forth in Article IV of Chapter 17.116 of the Oakland Planning Code, and the provisions of Chapters 12.04, 12.08, and 12.52 of the Oakland Municipal Code, shall apply in the S-12 zone, except as specified in Sections 17.94.060 through 17.94 100. All required parking spaces and associated maneuvering aisles, driveways, and other related features shall be of such design and arrangement as to provide motor vehicles with adequate ingress to and egress from all required parking spaces, and to provide pedestrians with adequate access to parked vehicles.

17.94.100 - On-street parking regulations.

- C. Definition of Parking Obstructions. For purposes of this section, parking obstructions are any features, other than posted fime limitations, which preclude or restrict the parallel on-street parking of an automobile, including, but not limited to, the following:
  - 1. Any existing driveway opening plus one and one-half  $(1\frac{1}{2})$  feet on either side;
  - 2. Any existing or required fire hydrant plus five (5) feet on either side;
  - 3. Any marked or unmarked crosswalk, plus a distance on either side to be determined by the City Traffic Engineer;
  - 4. Any red, yellow, green, white, blue, or other colored curb established by the City Traffic Engineer;
  - 5. Any area posted by the City Traffic Engineer for "No Parking Any Time";
  - Any posted bus stop, the length of which shall be determined by the City Traffic Engineer if the curb is not marked;
  - 7. Any handicapped curb cut, plus a distance on either side to be determined by the City Traffic Engineer;
  - 8. Any metered parking space established by the City Traffic Engineer;
  - 9. Any parking space signed or marked by the City Traffic Engineer for angle parking;
  - 10. At approximate right-angle intersections, the curb return plus the area between the curb return and a point the following distance from the intersection of the curb lines projected: twenty (20) feet on the near side of the intersection, or ten feet on the far side of the intersection, measured in the normal direction of vehicular travel. If no curb exists, the edge of the roadway where such curb return and the area specified above would be located;
  - 11. Any section of curb or roadway edge located between any two parking obstructions as defined in subsections (C)(1) through (C)(10) of this section, that is currently, and that will remain, too short to be a usable on-street parking space as defined in subsection D of this section;

12. Any section of curb or roadway edge along which the City Traffic Engineer determines that it is unsafe to park an automobile.

**17.97.030** - Special regulations applying to mixed-use developments on Bay Area Rapid Transit (BART) stations on sites with one acre or more land area.

- D. Performance Bonds. The City Planning Commission or, on appeal, the City Council may, as a condition of approval of any said development, require a cash bond or surety bond for the completion of all or specified parts of the development deemed to be essential to the achievement of the purposes set forth in Section 17.97.010. The bond shall be in a form approved by the City Attorney, in a sum of <u>one hundred tifty 450</u>-percent (150%) of the estimated cost of the work, and conditioned upon the faithful performance of the work specified within the time specified. This requirement shall not apply if evidence is provided to the city which indicates that alternative bonding or other assurances have been secured by the Bay Area Rapid Transit District.
- 17.97.040 Permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, are permitted:

A. Residential Activities:

Permanent

B. Civic Activities:

Essential Service

Limited Child Care

Community Assembly

Recreational Assembly

Community Education

Nonassembly Cultural

Administrative

Health Care

C. Commercial Activities:

General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Medical Service

General Retail Sales

Consumer Service (see Section 17.102.170 for special regulations relating to massage services and Section 17.102.450 for special regulations related to laundromats)

Consultative and Financial Service

Administrative

Business, Communication, and Media Service

Broadcasting and Recording Service

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

17.97.050 - Conditionally permitted activities.

The following activities, as described in the use classifications in Chapter 17.10, may be permitted upon the granting of a conditional use permit pursuant to Section 17.97.100 and the conditional use permit procedure in Chapter 17.134:

A. Residential Activities:

**Residential Care** 

Service-Enriched Permanent Housing

Transitional Housing

B. Civic Activities:

Utility and Vehicular

Extensive Impact

C. Commercial Activities:

Fast Food Restaurant

Convenience Market

Consumer Cleaning and Repair Service

Consumer Dry Cleaning Plant

Transient Habitation and Commercial Activities (see Section 17.103.050)

Alcoholic Beverage Sales

Mechanic or Electronic Games

Animal Care

Animal Boarding

Automotive Fee Parking subject to the additional criteria contained in Subsection 17.97.100 F.

D. Industrial Activities:

Custom Manufacturing

E. Off-street parking serving nonresidential activities listed in Sections 17.97.040 and 17.97.050

F. Activities that are listed neither as permitted nor conditionally permitted, but are permitted or conditionally permitted on nearby lots in an adjacent zone, subject to the conditions set forth in Section 17.102.110.

E----Additional-astivitios-which-are-permitted-or-conditionally-permitted-in-an-adjacent-zone, on-lots near-the-boundaty-thereof, subject-to-tho-conditions-set-forth-in-Section-17-102.110

17.97.080 - Special regulations applying to certain Commercial and Industrial Activities.

A. Fast-Food Restaurants. Convenience Markets, and Certain Establishments Selling Alcoholic Beverages. See Section-17-102-210\_17.103.030.

17.97.100 - Use permit criteria.

- D. That no front yard parking, loading area, or driveway shall connect or abut directly with the principal commercial street unless the determination can be made:
  - That vehicular access cannot reasonably be provided from a different street or other way;
  - 2. That every reasonable effort has been made to share means of vehicular access with abutting properties:
  - 3. That the proposal is enclosed or screened from view of the abutting principal street by the measures required in Section 17.110.040B.;
- F. In addition to the foregoing criteria and any other applicable requirements, auto fee parking within this zone shall be subject to the following use permit criteria:
  - 1. Auto fee parking shall be part of a larger development that contains a significant amount of commercial and/or residential facilities;
  - Auto fee parking may only be contained in a structured parking facility of at least three stories that replaces an existing at grade parking facility;
  - 3. The new parking structure shall represent no more than a <u>seventy-tive</u> 75-percent (75%) increase of existing parking at the site;
  - Auto fee parking at the site shall be specifically designated by a city sponsored plan or study designed to promote a transit oriented district as detined by the general plan;
  - 5. The facility or facilities containing the residential and/or commercial activities shall be adjacent to the principal street(s) and the auto fee parking shall be behind and substantially visually obstructed from the principal Street(s) by the residential and/or commercial facility or facilities; and
  - 6. The project shall be consistent with the general plan's goals, objectives, and policies that promote transit oriented development and districts.

For purposes of this subsection 17.97.100(F) "principal street" means the street or streets on which the development is most primarily oriented and that is appropriately designated in the general plan to accommodate the amount of trips proposed. On an interior lot the principal street shall be the street in front of the development. On a corner lot, the principal streets shall be both the streets adjacent to the development. On a lot that has frontage on three or more streets, at least two streets shall be designated as principal streets.

17.97.130 - Height, floor area ratio (FAR), density, and open space.

Table 17.97.01 below prescribes height, FAR, density, and open space standards associated with the Height Areas described in the Zoning Maps. The number designations in the "Additional Regulations" column refer to regulations below the table.

Regulation	Height Area						Additional	
	35	45	60	75	90	120	20 160 Regu	
Maximum Height	35 ft	45 ft	60 ft	75 ft	90 ft	120 ft	160 ft	1, 2
Height Minimum					- <u>i</u>			
Pennitted height minimum	0 ft	0 ft	35 ft	35 ft	35 ft	35 ft	35 ft	3
Conditionally permitted height minimum		NA	25 ft	25 ft	25 ft	25 ft	25 ft	3
Maximum Residential Density (square feet of lot area required per du	velling	unit)			4 <u></u>		:	
Regular units	550	450	375	275	225	225	225	4, 5
Rooming units	275	225	185	135	110	<u>110</u> 2 <del>2</del> 5	<u>110</u> 225	4, 5
Maximum Nonresidential FAR	2.0	2.5	3.0	4.0	4.5	5.0	5.0	4, 5
Maximum number of storjes (not including undergroUnd construction)		4	5	7	8	11	15	
Minimum Usable Open Space							<u> </u>	
Group usable open space per regular unit	150	150	150	150	100	100	100	6
Group usable open space per regular unit when private open space substituted		30	30	30	20	20	20	6
Group usable open space per rooming unit	75	75	75	75	50	50	50	6
Group usable open space per rooming unit when private open space is substituted	15	15	15	15	10	10	10	6

## Table 17.97.01 Height, Floor Area Ratio (FAR), Density, and Open Space Regulations

### Additional Regulations for Table 17.97.01:

3. This minimum height requirement only applies to the new construction of a principal building that is located on parcels adjacent to a street right-of-way that is <u>one hundred (100)</u> feet wide or more. Buildings in the CC-1 zone and buildings constructed to accommodate Essential Service. Utility and Vehicular, or Extensive Impact Civic Activities or Automobile and Automobile and Other Light Vehicle Sales and Rental, Automobile and Other Light Vehicle Gas Station and Servicing or Automobile and Other Light Vehicle Repair and Cleaning Commercial Activities may be exempted from the height minimum regulation by the Planning Director. The allowed projections into the height limits contained in Section 17.108.030 are not counted towards the height minimum.

4. See Chapter 17.107 for affordable and senior housing incentives. A Secondary Unit may be permitted when there is no more than one unit on a lot, subject to the provisions of Section-17-102-360\_17.103.080. Also applicable are the provisions of Section 17.102.270 with respect to additional kitchens for a dwelling unit, and the provisions of Section 17.102.300 with respect to dwelling units with tive or more bedrooms.

5. No portion of lot area used to meet the residential density requirements shall be used as a basis for computing the maximum nonresidential FAR unless the total nonresidential floor area on the let is less than three thousand (3,000) square feet

17.97.190 - Special regulations for mini-lot developments.

In mini-lot developments, certain of the regulations otherwise applying to individual lots in the S-15 zone may be waived or modified when and as prescribed in-Section-17-102-320 Chapter 17.142.

17.97.210 - Other zoning provisions.

- A. Parking and Loading. Off-street parking and loading shall be provided as prescribed in the off-street parking and loading requirements in Chapter 17.116.
- B. Bicycle Parking. Bicycle parking shall be provided as prescribed in the bicycle parking regulations in Chapter 17.117.
- C. Home Occupations. Home occupations shall be subject to the applicable provisions of the home occupation regulations in Chapter 17.112.
- D. Nonconforming Uses. Nonconforming uses and changes therein shall be subject to the nonconforming use regulations in Chapter 17.114.
- E. General Provisions. The general exceptions and other regulations set forth in Chapter 17.102 shall apply in the S-15 zone.

### 17.100B.030 - Required design review process.

- A. Except for projects that are exempt from design review as set forth in Section 17.136.025, no <u>Building Facility (see Section 17.09.040 for definition)</u>, Designated Historic Property, Potentially Designated Historic Property, Building Facility, (see Section 17.00-010 for definition), Telecommunications Facility, Sign, or other associated structure shall be constructed, established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136, and when applicable, the additional provisions in Sections 17.100B.050, 17.100B.060, and 17.100B.070, the Telecommunications regulations in Chapter 17.128; or the Sign regulations in Chapter 17.104.
- B. See Section 17.136.075 for design review criteria for the demolition or removal of Designated Historic Properties and Potentially Designated Historic Properties.
- C. Landmarks Referral. If an application is for regular design review in the S-20 zone, and the Director of City Planning determines that a proposed addition or alteration will have a significant effect on the property's character-defining elements that are visible from a street or other public area, the Director may, at his or her discretion, refer the project to the Landmarks Preservation Advisory Board for its recommendations. "Character-defining elements" are those features of design, materials, workmanship, setting, location, and association that identify a property as representative of its period and contribute to its visual distinction or historical significance. An addition or alteration is normally considered "visible from a street or other public area" if it affects a street face or public face of the facility or is otherwise located within the "critical design area," defined as the area within forty (40) feet of any street line, public alley, public path, park or other public area.

17.101C.060 - Permitted and conditionally permitted activities.

Table 17.101C.01 lists activities permitted, conditionally permitted, and prohibited in the D-BR zone. The descriptions of these activities are contained in Chapter 17.10.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134) in the corresponding zone.

"L" designates activities subject to certain limitations listed at the bottom of the table.

"---" designates activities that are prohibited in the corresponding zone.

Table 17.101C.01 Permitted and Conditionally Permitted Activities

· ·	<u>D-BR</u> Zone	
Permanent Residential	C(L1)	 
Residențial Care-occupying-a-One-Family-Dwelling-Residențial-Facility	GP(L1)(L2)	47.102.21217.103.01
	C(L1)	47.40 <u>2.212</u>
Service-Enriched Pemanent Housing	C(L1)	17-102-21217.103.01
Transitional Housing	C(L1)	17-102-21217.103.01
		17-102-21217.103.01
Emergency Shelter		<u></u>
Semi-Transjent Residentia		17-10 <del>2.2</del> 12 <u>17.103.01</u>
Bed and Breakfast		<u> </u>
Civic Activities		
Essentja  Service	P	
Limited Child-Care	C(L1)	
Community Assembly	C(L1)	
Recreational Assembly	C(L1)	
Community Education	P(L1)	
Nonassembly Cultural ,	P	
Administrative	P(L1)	
Health Care	c	
Special Health Care	·	1-7-102-390 17.103.020
Utility and Vehicular		
Extensive Impact		
Commercial Activities		
General Food Sales	P	· · · · · · · · · · · · · · · · · · ·
Full Service Restaurant	P	]
Limited Service Restaurant and Cafe	P	i
Convenience Market	c	1-7-102-210 17.103.030
Fast-Food Restaurant	C	
Alcoholic Beverage Sales	c	17.102.210 17.103.030 and

		<u>17.114.030</u>	
Mechanical or Electronic Games	<u> </u>	17.102.210	
Medical Service	P(L32)		
General Retail Sales	P		
Large-Scale Combined Retail and Grocery Sales	-		
Consumer Service	P(L4)		
Consultative and Financial Service	P		
Check Cashier and Check Cashing			
Consumer Cleaning and Repair Service	Р		
Consumer Dry Cleaning Plant	С	•	
Group Assembly	C <u>(L5)</u>		
Personal Instruction and Improvement and Small Scale Entertainment	С		
Administrative	P(L <u>1</u> 4)		
Business, Communication, and Media Service	P		
Broadcasting and Recording Service	P		
Retail Business Activity	c		
Research Center	_		
General Wholesale Sales	-		
Transient Habitation	-	17-102.370 17.103.050	
Building Material Sales	-		
Automobile and Other Light Vehicle Sales and Rental	c		
Automobile and Other Light Vehicle Gas Station and Servicing	(L <u>6</u> 3)	17.114.050 (A)	
Automotive and Other Light Vehicle Repair and Cleaning	(L <u>6</u> 3)	17.114.050 (A)	
Taxi and Light Fleet-Based Service	-		
Automotive Fee Parking	c		
Transport-and-Warehousing	-		
Animal Care	c		
Animal Boarding	c		
Undertaking Service	-		
Scrap Operation	<u> _</u>	17.102.210	
ndustrial Activities	All Industrial Activities prohibited these zones		
Agricultural and Extractive Activities	All Agricultural and Extractive Activities prohibited in these zones		
Off-street parking serving activities other than those listed above or in Section 7.74.030, subject to the conditions set forth in Section.47.102.109_17.116.075	С 	17.102.100 17.116.075	
Activities that are listed as prohibited, but are permitted or conditionally oermitted on pearby lots in an adjacent zone. Additional activities which are permitted or conditionally permitted in an adjacent zone, on lots near the boundary thereof, subject to the conditions set forth in Section 17.102.110	C	17.102.110	

### Limitations:

L2. Residential Care is only permitted upon the granting of a Conditional Use Permit (see Chapter 17.134 for the CUP procedure) when net located in a One-Family Dwelling Residential Facility. No

<u>Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter</u> <u>Residential Activity shall be located closer than three hundred (300) feet from any other such Activity or</u> <u>Facility.</u>

<u>L2-L3</u>. These activities may only be located on the ground floor of a building on a lot that has a property line abutting the Broadway or 27<sup>th</sup> Street right of way upon the granting of a conditional use permit (see Chapter 17.134), and shall conform to the additional criteria contained in Section 17.25.030. However, incidental pedestrian entrances that lead to one of these activities in stories above the ground floor are permitted without the granting of a conditional use permit.

L4. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than tive hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

L5. No new or expanded adult entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

L36. Reestablishment of a discontinued, legal non-conforming Automobile and Other Light Vehicle Gas Station and Servicing activity and/or an Automotive and Other Light Vehicle Repair and Cleaning activity may only occur no later than six (6) months after discontinuation of such a activity, per Section 17,114.050(A).

17.101D.030 - Permitted and conditionally permitted activities.

Table 17.101D.01 lists the permitted, conditionally permitted, and prohibited activities in the D-KP-1, D-KP-2, and D-KP-3 zones. The descriptions of these activities are contained in Chapter 17.10.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134) in the corresponding zone.

"---" designates uses that are prohibited in the corresponding zone.

## Table 17.101D.01 Permitted and Conditionally Permitted Activities

Activities ·	Regulati	Additional		
	D-KP-1	D-KP-2	D-KP-3	Regulations
Residential Activities				
Pennanent	P	P	Р	]
Residential Care	c	C	c	<del>17.102.212</del> 17.103.010
Service-Enriched Pennanent Housing	C	C	С	1-7-102.212 17.103.010
Transitional Housing	c .	c	C ·	17-102.212 17.103.010
Emergency Shelter	c	c	C	17.102.212 17.103.010
Semi-Transient	C	С	С	17.102.212 17.103.010
Bed and Breakfast	C	С	С	17. <b>1</b> 0.125
Civic Activities				
Essentjal Service	Р	Р	Р	
Limited Child-Care	Р	P	P	1
Community Assembly	P	Р	Р	1
Recreational Assembly	Р	P	P	
Community Education ·	P	Р	P .	
Nonassembly Cultural	P	Р	P	
Administrative	С	C	С	· .
Health Care	P	Р	P	
Special Health Care	C <u>(L1)</u>	C( <u>L1</u> )	C <u>(L1)</u>	<del>17.102.380</del> <u>17.103.020</u>
Utility and Vehicular	С	c	c	
Extensive Impact	C	c	C	·
Commercial Activities			1	
General Food Sales	P	Р	Р	
Full Service Restaurants	P	Р	P	
Limited Service Restaurants and Cafe	P	Р	P	
Fast-Food Restaurant	С	c	C	17 <u>-102.2</u> 10 <u>17.103.030</u> and 8.09
Convenience Market	P	С	С	17-10 <u>2-2</u> 10 17.103.030

Alcoholic Beverage Sales	c	С	C	17 <u>-102-240</u> <u>17.103.030</u> and 17 <u>-102-040</u> <u>17.114.030</u>
Mechanical or Electronic Games	c	c	С	1-7-10 <del>2-2</del> 40
Medical Service	P	P	р	<u> </u>
General Retail Sales	P	Р	P	······
Large-Scale Combined Retail and Grocery Sales	-	- <b>_</b>	İ	·
Consumer Service	P(L2)	P(L2)	P <u>(L2)</u>	Í
Consultative and Financial Service	c	С	C.	
Check Cashier and Check Cashing	1	[- <u>-</u>	-	
Consumer Cleaning and Repair	p	P	P	1
Consumer Dry Cleaning Plant	С	c	с	
Group Assembly	C(L3)	C(L3)	C(L3)	]
Personal Instruction and Improvement	P	P	P	4
Administrative	C	c	с	j
Business, Communication, and Media Service	С	С	c	· <u>·····</u>
Broadcasting and Recording Services	С	С	c	
Research Service	C	С	с	<u>;                                     </u>
General Wholesale Sales		-	-	<u> </u>
Transient Habitation	-	-		; <u> </u>
Wholesale-and-Professional-Building Material Sales		-	·	<u> </u>
Automobile and Other Light Vehicle Sales and Rental	1	[		
Automobile and Other Light Vehicle Gas Station and Servicing		-		
Automobile and Other Light Vehicle Repair and Cleaning		-	-	<u></u>
Taxi and Light Fleet-Based Services		-	-	
Automotive Fee Parking	c	С	С	j
Animal Boarding	C	С	с	
Animal Care	c	c	с	
Industrial Activities	All Industrial Activities prohibited in these zones		· · · · · · · · · · · · · · · · · · ·	
Agricultural and Extractive Activities	All Agricultural and Extractive Activities prohibited in these zones			
Off-street parking serving activities other than those listed above or in Section 17.74.030, subject to the conditions set forth in Section 17.402.100 <u>116.075</u>	c	С	С	17.74.030 17.102-100 <u>17.116.075</u>
Activities that are listed as prohibited, but are permitted or conditionally permitted on nearby lots in an adjacent zone Additional-activities-which are-permitted-or-conditionally-permitted-in-an-adjacent-zone, on-lots-hear the-boundary-thereof, subject-to-the-conditions-set-forth-in-Section 47-102-140	c	с	с	17.102.110

### Limitations:

L1. No new or expanded Special Health Care Civic Activity shall be located closer than two thousand five hundred (2,500) feet from any other such activity or five hundred (500) feet from any K-12 school or

Transitional Housing, Enriched Housing, or Licensed Emergency Shelters Civic Activity. See Section 17.103.020 for further regulations regarding Special Health Care Civic Activities.

L2. See Section 17.102.170 for special regulations relating to massage services. Also, no new or expanded laundromat shall be located closer than tive hundred (500) feet from any existing laundromat. See Section 17.102.450 for further regulations regarding laundromats.

<u>L3.</u> No new or expanded aduit entertainment activity shall be located closer than one thousand (1,000) feet to the boundary of any residential zone or three hundred (300) feet from any other adult entertainment activity. See Section 17.102.160 for further regulations regarding adult entertainment activities.

17.101D.040 - Permitted and conditionally permitted facilities in the D-KP-1, D-KP-2, and D-KP-3 zones.

Table 17.101D.02 lists the permitted, conditionally permitted, and prohibited facilities in the D-KP-1, D-KP-2, and D-KP-3 zones. The descriptions of these activities are contained in Chapter 17.10.

"P" designates permitted activities in the corresponding zone.

"C" designates activities that are permitted only upon the granting of a conditional use permit (see Chapter 17.134) in the corresponding zpne

"---" designates uses that are prohibited in the corresponding zone

Table 17.101D.02 Permitted and Conditionally Permitted Activities

Facility Types	Zone		Additional	
	D-KP-1	D-KP-2	D-KP-3	Regulations
Residential Facilities				
One-Family Dwelling	P	P	Р	
One-Family Dwelling with Secondary Unit	P	P	P	<del>17.102.360<u>17.103.080</u></del>
Two-Family Dwelling	P	P	Р	· · · · · · · · · · · · · · · · · · ·
Multifamily Dwelling	P	P	P	
Rooming House	P	Р	Р	
Mobile Home	—			
Nonresidential Facilities				- <u></u>
Enclosed nonresidential facilities	P	Р	Р	
Open nonresidential facilities	С	С	С	
Sidewalk Cafe	Р	Р	Р	17-102-335_17.103.090
Drive-In	С	С	С	
Drive-Through	C <u>(L1)</u>	G <u>(L1)</u>	C <u>(L1)</u>	17-10 <del>2-290</del> 17.103.100
Telecommunications Facilities				
Micro Telecommunications	P	P	Р	
Mini Telecommunications	С	C	С	
Macro Telecommunications	c	c	C	
Monopole Telecommunications	С	C	С	
Tower Telecommunications	_	_		

Oakland, California, Code of Ordinances

Codo of Ordin

Residential Signs	P	Р	Р	17.104	
Special Signs	P	P	P	17.104	
Development Signs	· P	P	P	17.104	
Realty Signs	P	Р	P	17.104	
Civic Signs	P	P	P	17.104	
Business Signs	P	Р	P	17.104	
Advertising Signs		—		17.104	<b></b>

### -Limitation:

L1. No new or expanded Fast-Food Restaurants with Drive-Through Nonresidential Facilities shall be located closer than five hundred (500) feet of an elementary school, park, or playground. See Sections 17.103.030 and 17.103.100 for further regulations regarding Drive-Through Nonresidential Facilities.

#### 17.101D.060 - Design review.

- B. Exemptions from Design Review. The following changes to existing nonresidential buildings are exempt from design review:
  - Any alteration or addition of existing floor area or footprint area determined by the Director of City Planning to be not visible from the street or from other public areas. An alteration or addition will normally be considered "not visible from the street or from other public areas" if it does not affect any street face or public face of a building or is located more than forty (40) feet from any street line, public path, park or other public area;
  - 2. Alterations or additions of floor area or footprint that are determined by the Director of City Planning to be visible from the street or from other public areas, but which comprises less than ten percent (10%) of the total floor area, or anything under <u>twenty-five thousand (25,000)</u> square feet, whichever is smaller;
  - A change of sign face copy or new sign face so long as the structure and framework of the sigh remain unchanged and the new sign face duplicates the colors of the original or, in the case of an internally illuminated sign, the letter copy is light in color and the background is dark;
  - 4. Any alteration or addition not normally exempt which is used as a loading dock, recycling area, utility area, porch, deck or similar open structure addition that is no higher than six (6) feet above finished grade, less than five hundred (500) square feet in floor or footprint area, and has no significant visual or noise impact to neighboring properties or from a public street. Exemptions only permitted where the proposal conforms with all buffering requirements in Chapter 17.110 and all performance standards in Chapter 17.120;
  - 5. The alteration or addition is on a roof and does not project above the parapet walls.
- D. Regular Project Design Review. Unless determined exempt or subject to small project design review pursuant to Section 17.101D.040 B or C above, no building, sign or other facility shall be constructed or established or altered in such a manner as to substantially affect its exterior appearance unless plans for such proposal have been approved pursuant to the following Regular Design Review procedures:
  - 1. Reviewing Body:
    - a. If the project requires preparation of a Subsequent or Supplemental Environmental Impact Report, or involves twenty-five thousand (25,000) square feet or more of floor area, or includes a proposed skybridge, the Director of City Planning shall refer the application to the City Planning Commission for an initial decision.

- b. All other applications for regular design review shall be considered by the Director of City Planning. However, the Director may, at his or her discretion, refer the application to the City Planning Commission for an initial decision rather than acting on it himself or herself.
- 2. Procedure for Consideration of Design Review: Applications for design review shall be considered by the Director of City Planning or the Planning Commission according to the following procedures:
  - a. Decisions by the Planning Commission shall be made at a public hearing. At his or her discretion, the Director of City Planning may hold an administrative hearing for projects under his or her review.
  - b. Notice of public and/or administrative hearings shall be given by posting notices thereof within three hundred (300) feet of the property involved in the application; notice shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the City within three hundred (300) feet of the property involved. Notice shall also be given by e-mail, mail or delivery to all persons previously requesting to be notified of actions related to the Kaiser OMC Campus through public workshops, community meetings or other direct requests to the Planning Department. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing, if such is to be held, or, if not, for decision on the application by the Director or the Commission, as the case may be.
  - c. The Director or the Commission may seek the advice of outside design professionals and/or refer the matter to the City's Landmark's Preservation Advisory Board if Historic Resources may potentially be affected.
  - d. The Director or the Commission, as the case may be, shall determine whether the proposal conforms to the applicable design review criteria and also is in substantial conformance to the Kaiser Permanente Oakland Medical Center Master Plan, and may approve or disapprove the proposal or require such changes therein or impose such reasonable conditions of approval as are in his or her or its judgment necessary to ensure conformity to said criteria.
  - e. A determination by the Director shall become final ten (10) days after the date of decision unless appealed to the City Planning Commission in accordance with the procedures in Section 17.136.080. The decision of the Planning Commission on appeal is final and is itself not appealable.
  - f An initial decision of the Commission shall become final ten (10) days after the date of decision unless appealed to the City Council in accordance with the procedures in Section 17.136.090.
- F. Adherence to Approved Plans. A design review approval shall be subject to the plans and other conditions upon the basis of which it was granted and shall terminate in accordance with Section 17.136.100.
- G. Revocation/Enforcement. In the event of a violation of any of the provisions of the zoning regulations, or in the event of a failure to comply with any prescribed condition of approval, or if the activity causes a public nuisance, the City may, after holding a public hearing, revoke any design review approval or other approval or take other enforcement actions in accordance with the procedures in Chapter 17.152.
- H. Review by Landmarks Board. A design review application may be subject to review by the Landmarks Preservation Advisory Board in accordance with Section 17.136.040.
- I. Design Review and Other Approvals. Whenever design review approval is required for a proposal also requiring a conditional use permit, or planned unit development permit or variance, the application for design review shall be included in the application to said permit and shall be processed and considered as part of same, in accordance with Section 17.136.120.

# 17.101D.130 - Signs.

A. If a comprehensive sign program is adopted as part of the Kaiser Permanente Oakland Medical Center Master Plan, the provisions of the comprehensive sign program shall govern and shall supersede the provisions of Chapter 17.104.

### CHAPTER 17.102 - GSNSRAL-REGULATIONS APPLICABLE TO <u>CERTAIN ACTIVITIES AND</u> FACILITIES ALL-OR-SEVERAL-ZONES

#### Sections:

17.102.010 - Title, purpose, and applicability.

- 17-102-020 -- Supplemental-zoning provisions.
- 17-102.040 Effect of prior permits.
- 17.102.070 Application of zoning regulations to lots divided by zone boundaries.
- 17.102.080 Permitted and conditionally permitted uses.
- 17.102.090 Conditional use permit-for-sShared access facilities.
- 17-102-100---Conditions-for-accessory parking serving activities-which are not-themselves allowed.
- 17.102.110 EConditions-for-eExpansion of use into adjacent zones.
- 17.102.120 Restriction on rRemoval of dirt or other minerals—Residential and S-1, S-2, S-3 and OS zones.
- 17.102.130---Time limit on operation of subdivision sales offices -- Residential zones.
- 17.102.140 Special-regulations-applying to pPrivate stables and corrals.
- 17.102.160 Special regulations applying to aAdult entertainment activities.
- 17.102.170 Special regulations applying to mMassage activities.
- 17.102.180 Restriction on vVertical location of activities in buildings containing both Residential and Nonresidential Activities---Commercial zones.
- 17.102.190 Joint Living and Work Quarters.
- 17.102.195 Residentialiy-Oriented Joint Living and Working Quarters.
- 17.102.200 Conditional use permit-required for pPedestrian bridges constructed over city streets.
- 17-102-210 -- Special regulations applying to Convenience Markets, Fast-Food Restaurants, certain establishments seiling alcoholic beverages, providing mechanical or, and electronic games.
- 17.102.212--- Special Regulations Applying to Residential Care, Service Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities.
- 17.102.220 --- Special regulations applying to mining and quarrying extractive activities.
- 17.102.230 Special regulations applying to the dDemolition of a facility containing rooming units or to the conversion of a living unit to a Nonresidential Activity—Nonresidential zones.
- 17-102-240 --- Special-regulations-applying-to-microwave-dishes-in-or-near-residential-zones-
- 17.102.250- Maximum density and floor-area-ratio-during-construction.
- 17-102.260 Occupancy-of a dwelling unit.
- 17-102-265---Occupancy-of-a One-Family Dwelling Residential-Faollity-by-a-Residential-Care Residential-Activity-
- 17.102.270 An-aAdditional kitchen for a single dwelling unit.
- 47.102.280 Rules for determining the number of habitable rooms in Residential Facilities.
- 17.102.290 Special regulations for Drive-Through Nonresidential Facilities.
- 17.102.300 Conditional-use permit for dDwelling units with five or more bedrooms.
- 17-102-310-Special-regulations-for-certain-projects-with-development-agreements-
- 17.102.320 --- Conditional-use-permit for waiver of certain requirements in mini-lot developments.

17.102.330 - Conditional uso permit for waiver of cortain requirements with parcel-division between existing buildings.

17.102.335 - Standards for Sidewalk Cafes.

- 17.102.340 Special-regulations applying-to-eElectroplating activities in the M-20,-M-30, and M-40 zonesIndustrial Zones.
- 17.102.350 Regulations applying to tTobacco-oriented activities.

17.102.360 - Secondary Units.

17.102.370 - Conditional use permit for hotels and motels.

17.102.380 - Special regulations applying to truck-related activities in the West Oakland Community Development District.

17.102.390 - Regulations Applying to Special Health Care Civic Activities.

17.102.400 - Exterior security bars and related devices. Special design-requirements for lote-that contain Residential Facilities and no-Nonresidential Facilities.

17.102.420---Special design-requirements-for lots-located in a residential and commercial zones and the OS, S-1, S-2, S-3 and S-15 zones.

17-102-430 --- Regulations applying to check cashier and/or check cashing activity.

17.102.450 - Special regulations applying to ILaundromats.

### 17.102.010 - Title, purpose, and applicability.

The previsions of this chapter and Chapters 17.104 through 17.108 shall be known as the general regulations <u>applicable to certain activities and facilities</u>. The purpose of these provisions is to set forth certain of the regulations which apply throughout the city or in several zones. These regulations shall apply in the zones and situations specified hereinafter.

(Prior planning code § 7000)

17.102.020 -- Supplemental zoning provisions.

The detinitions, special use classification rules, and other provisions set forth in Chapters 17.07, 17.09 and 17.10; the provisions of Section 17.108.130; the nonconforming use regulations in Chapter 17.114; the rezoning, variance, and other provisions set forth in Chapters 17.130 through 17.152; and the provisions of the zoning maps in Chapter 17.154 shall apply throughout the city. The provisions of the performance standards in Chapter 17.120 and the planned unit development regulations in Chopter 17.142 shall apply in the zones and situations specified in said chapters. The provisions of development control maps are in addition to, or cuporsede when so specified, the regulations applying in the zones covering the same areas.

17.102.040 -- Effect of prior permits-

A.....Building ond Sign-Permits ond Development Agroomonts. Whenever any subsisting building permit or sign-permit has been lawfully issued beforehand, or whenever a subsisting development agreement has been approved beforehand under Section 17.102.310 and the development agreement procedure in Chapter 17.138, neither the original adoption of the zoning regulations nor the adoption of ony subsequent rezoning or other amendment thereto shall prohibit the construction, other development or change, or use authorized by said permit or agreement. The uses as they result-shall be deemed nonconferming uses and subject to the nonconforming-use regulations in Chepter 17.114, except to the extent that they ore authorized by a subsisting conditional use-permit, development-agreement, variance, or other special zoning approval.

B. Alcoholic Beverage Control Licenses. On premises for which a valid state of California Alcoholic Beverage Control licence had been issued, and which premises had been used in the exercise of the rights and privileges conferred by the licence at a time immediately prior to the offective date of the applicable provisions of Section 17.102.210, the premises may hereafter be used in the exercise of the oame rights and privileges without requiring a conditional use permit or having to meet the provisions of the aforesaid section. Such use shall be deemed a nonconforming use and subject to the nonconforming use regulations, except as otherwise provided in Sections 17.114.020 and 17.114.030. For the purposes of this subsection, the word "premises" shall mean and include only the actual space within a building devoted to the sale of alcoholic beverages on said effective date.

(Prior planning code §-7003)

\_17.102.070 -- Application-of-zoning-regulations-to-lots-divided-by-zono-boundaries-

Wherever it is found, after applying the rules set forth in Section 17.154.050 for interpretation of zone bsundaries, that any let is divided by a boundary between zones, the provisions of the zoning regulations shall apply as follows to such let:

A. Application of All Rogulations of One-Zone to Existing Let If Boundary Is Near Let Line. If the let was on the effective date of the zoning regulations, or of a subsequent rezoning or other amendment thereto resulting in division of the let by a zone boundary, and the owner or developer of such let, or of a portion or combination of such let or lets, may at his or her option assume that all of the regulations applying in any zone covering fifty percent (50%) or more of the let area apply to the entire let or lets. However, this option shell not apply unless the entire let or all such lets or parcel of land could be included in such zone by shifting the affected zone boundary by not-more than thirty (30) feet, as measured perpendicularly to said boundary at any point.

B.\_\_\_\_\_Application of Regulations Where subsection A Is Inoperative. Wherever the provisions of subsection A of this section do not apply or the option provided therein is not exercised:

1.— No activity type or facility type is allowed on any-portion of the lot located in a zone where such type is not generally allowed, except for the accessory uses ellowed by subsections (B)(2) and (B)(3) of this section.

2. Accessory off-street parking and leading may be located on the lot without regard for zone beundaries; provided that no parking or leading shall be located on any portion of the lot located in a zone where the principal activity served is not generally allowed, except as such parking is specifically allowed by the appliceble individual zone regulations subject to the conditions set forth in Section 17.102.100; and further provided that parking and leading shall be subject to a conditional ese-permit requirement or other special controls on any portion of the lot located in any zone where such controls generally apply to parking or leading. The total amount of required parking and leading shall be calculated separately on the basis of the amount of the served use and the requirements applying in each zone; provided that the minimum size for which any parking or leading is required shall be deemed to be exceeded if it is exceeded by the total of such use on the entire lot.

3.——Accessory landscaping, fences, screening or retaining walls, and usable open space may be located on the lot without regard for zone boundaries. The total amount of required usable open space shall be calculated separately on the basic of the number of living units, or amount of floor area, and the usable open space requirements in each zone, provided that where reference is made to the total number of living units on a let, the number on the entire lot shall be considered.

1. The maximum permitted or conditionally permitted number of living units or fleor-aroa ratio, if any, en the lot chall be calculated separately on the basis of the amount of lot area and the donsity ratio and tioer-aroa ratio applying in each zone. The resulting maximum permitted or conditionally permitted total number of living units or amount of tioor area may be distributed on the lot without regard for zone boundaries, except as otherwise provided in subsection (B)(1) of this section and except that the number of living units and amount of tioor area within each zone shall not exceed the number or amount which would be allowed on the entire lot if it were completely within such zone. 5. The minimum lot area, width, and frontage requirements of the zone which covers the greater or greatest portion of the let area of the lot shall apply to the ontiro lot. If the lot area is divided equally between two or more zones, the owner or developer of the lot may assume that the minimum lot area, width, and frontage requirements of either or any of cuch zones apply to the entire lot.

6.Ati regulations not covered above shall apply separately to the portion of the lot within each zone, provided that where reference is mode in such regulation to the total quantity of living units or other unif ef measurement on a lot, the quantity on the entire lot shell be considered.

C-Wherover a lot is divided by a boundary between height areas, the height line may be moved up to thirty (30) feet in any direction upon the granting of Regular Design Review approval (see Chopter 47.436 for the Regular Design Review process) to accommodate the site plan of a proposed development project. In addition to the general Design Review Criteria contained in Chapter 17.136, the proposal must mooting the following criteria:

4. The height line adjustment creates a mere successful site plan in terms of open space, parking, or building location; and

2. Appropriate-height-transitions are incorporated into the building design and site plan to adjacent lower donsity-residential properties that either share a parcol line or are across the street-from the prepesal.

17.102.080 -- Permitted and conditionally-permitted uses.

A-Other Uses Prohibited. Except as otherwise provided in Soctions 17.102.040 and 17.102.070, the nonconforming use regulations in Chapter 17.114, and the planned unit development regulations in Chapter 17.142, or as authorized under Section 17.102.310, the development agreement procedure in Chapter 17.138, or the variance procedure in Chapter 17.148, no land shall be improved or used for any activity or facility which is not-listed as permitted or conditionolly permitted in the applicable individual zono regulations or development control maps.

B. Relationship Between Activitios and Facilities. A use must qualify under the zoning regulations both as an activity and as a facility. A permitted or conditionally-permitted activity may be accommodated or served only by a permitted facility or, upon the granting of a conditional use permit, by a conditionally permitted facility; and a permitted or conditionally permitted facility may accommodate or serve, or be designed to accommodate or serve, only a permitted activity or, upon the granting of a conditional use pormit, a conditionally permitted activity or, upon the granting of a conditional use permitted activity or a permitted activity.

17.102.090 - Conditional-uso-permit-for-sShared access facilities.

A. Use Permit Required. A shared access facility shall be allowed only upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.

47:402-100 - Conditions-for-accossory-parking-serving-activitios-which are not-thomsolvos allowed-

The following rogulations shall apply to parking serving principal activities which are not themselves permitted, wherever such parking is listed in the applicable individual zono regulations as permitted or conditionally permitted subject to the conditions cot forth in this section:

A. General Conditions. In all-zones, such parking facilities shall be ucod for accossory porking only, with no sales, dead storage, repair work, dismantling, or servicing of any kind.

S.—— Conditions in Residential Zones. In all residential zones:

1. Such parking shall not in any case be located farther than one hundred fifty (150) feet, excluding the width of any intervening street, from the nearest-boundery of any nonresidential zone, as measured perpendicularly from said boundary at any point; and

2. Such-parking-shall-not-be-co-located-ac-to-extend-along-any-one-side-of-any-street-farther-into-any residential-zone-than any-residentially-zoned-lot-which-is-in-separate-ownership-and-which-hoc-frontago on-tho-same-side-of-the-same-street-ac-said-parking, ether-than-a-lot-developed-only-for-parking; and 3. Such-parking facilities-shall be-open-only; and

4.—\_\_\_\_All Signs serving such parking shall be cubject to the limitations set forth in Section 17.104.010(G)(3).

(Prior-planning code § 7011)

17.102.110 - EConditions-for-eExpansion of use into adjacent zones.

The following regulations standards and criteria shall apply when the applicable individual zone regulations conditionally permit activities that are prohibited in the subject zone, but permitted or conditionally permitted on nearby lots in an adjacent zone.

- A. A conditional use permit for such a use may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134 and the following additional use permit criteria:
  - <u>1.</u>to-activitios-which-are-conditionally-permitted-by-the-applicable-individual-zone-regulations near-a-zone-boundary-and-subject-to-the-conditions-set-forth-in-thie-section:
  - A.— Substantial Improvement in, or Superior, Environment. A conditional-use pormit-for-such-a use-may-be-granted-only-upon-determination-that-the-proposal-conforms-to-the-general use-permit-criteria-sot-forth-in-the-conditional-use-permit-proceduro-in-Chapter-17.134-and
  - -t<u>T</u>hat the location, size, design, and other characteristics of the entire use as proposed will substantially improve or provide superior environmental relationships among all uses in the immediate vicinity.
  - B2. Preservation of Privacy. A conditional use permit-for such a use may be granted only upon determination tThat the design and site planning of all buildings, open areas, parking, service areas, paths, stairways, accessways, corridors, and balconies will be so designed as to not adversely affect the privacy, safety, or environmental amenities of adjacent properties.
  - <u>3G</u>. Retention of Natural and Topographic Features.-A-conditional use-permit for-such a uco may be granted only upon determination tThat within the expansion area every reasonable effort will be undertaken to preserve natural grades, topographic features, watercourses, and significant landscape features.
- BD. Expansion of Use on Abutting Lot. The following standards shall also apply:
  - <u>1.</u> Such uses shall be allowed only when they constitute an expansion of or are a part of an existing or proposed activity or facility which is located in or partially located in the adjacent zone, and is permitted or conditionally permitted therein. Such uses shall be allowed only on a lot, or one of a series of lots under one ownership, directly contiguous to the lot in the adjacent zone, with no intervening streets.

- E. Maximum Distance from Zone Boundary. Such uses shall not extend more than one hundred fifty (150) feet into the zone, as measured perpendicularly from the zone boundary at any point
- F----Increased--Off-Street--Parking--Off-street--parking--shall--bo--provided--for--the--proposed development-in-an-amount-at-least-one-hundred-fifty-percent-(150%)-of-that-required-by-the off-street-parking-requirements-in-Chapter-17-116
- <u>3.</u>G. Height Within the area of the allowed expansion, the maximum height of any building or facility shall not exceed the maximum height permitted on abutting lots.

<u>4.H</u> Increased Yard Areas. The minimum yard depth or width, as the case may be, for buildings within the expansion area shall be no less than one hundred fifty percent (150%) of the yard depth or width, if any, required for uses on those properties abutting the expansion area.

- <u>5.1</u>-Screening and Buffering. The exterior perimeter of the expansion area shall be provided with screening and buffering devices including, but not limited to, established trees.
- 6.4 Maximum Density. The number of living units on any lot or series of lots involved in the expansion of use shall be calculated separately on the basis of the amount of lot area and the density rafio applying in each of the affected zones. The maximum number of living units allowed in the proposed development shall not exceed the accumulative total resulting from adding the density calculations for each of the lot areas and zones involved in the expansion.

17.102.120 - Rostriction-on-rRemoval of dirt or other minerals-Residential and S-1, S-2, S-3 and OS zones.

In all residential zones and in the S-1, S-2, S-3 and OS zones, no grading or excavation shall involve the removal of any soil, rock, sand, or other material for purposes of sale, fill, building, or other construction usage off the premises,\_\_from-which-removed, unless a conditional use permit-for-such removal is granted pursuant to the conditional use permit procedure in Chapter 17.134. However, excavations in any street, alley, or other public place and excavations for foundations, basements, or cellars for the erection of any buildings for which a building permit has been issued shall be exempt from the above restriction.

17.102.130-Time-limit-on-operation-of-subdivision-sales-offices-Residential-zones-

In-all-residontial-zones, the conduct-and-maintenonce-of-any-real-estate-sales-office-which-ic-incidental-to the-development-of-a-subdivision-shall-bo-limited-to-a-one-year-period-unless-a-conditional-use-permit-for a-longer-time-period-is-granted-pursuant-to-the-conditional-uso-permit-procedure-in-Chapter-17-134.

(Prior-planning-code § 7014)

17.102.140 - Special-regulations-applying-to-pPrivate stables and corrals.

The following regulations shall apply in all zones to private stables, corrals, and similar facilities and to the keeping or training of horses, mules, or donkeys as an accessory activity:

- A. Conditional Use Permit Requirement Such uses are permitted only upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
- B. Maximum Number of Animals. No more than three (3) such horses, mules, or donkeys shall be kept or trained on any single lot.

- C. Minimum Lot Area. Such uses shall not in any case be located on any lot having a lot area of less than twenty-five thousand (25,000) square feet.
- D. Location on Lot. No such stable, corral, or paddock shall be located within thirty (30) feet from any lot line.
- E. Screening. All open portions of such facilities shall be screened from abutting lots, streets, alleys, and paths, and from the private ways described in Section 17.106.020, by dense landscaping not less than five and one-half (5½) feet high and not less than three (3) feet wide or by a decorative screening fence or wall not less than five and one-half (5½) feet high, subject to the standards for required landscaping and screening in Chapter 17.124 and the exceptions stated in said chapter.
- F. See Oakland Municipal Code (OMC) Chapter 6.04 for additional regulations for animal guarters.

17.102.160 - Spocial-regulations-applying-to-aAdult entertainment activities.

- A. Conditional Use Permit Requirement. Adult entertainment activities are not permitted in any zone except upon the granting of a conditional use permit pursuant to the criteria in subsection B of this section (which supersedes the general criteria in Section 17.134.050) and the conditional use procedure in Chapter 17.134.
- B. Conditional Use Permit Criteria. A conditional use permit for an adult entertainment activity shall only be granted upon a determination that all of the following conditions are present notwithstanding any conflicting requirements contained elsewhere in the zoning regulations:
  - 1. The requested use at the proposed location will not adversely affect the use of churches, temples or synagogues; public, parochial or private elementary, junior high or high schools; public parks and recreation centers; public or parochial playgrounds; residences; child care facilities; elderly residential care facilities; hospitals; medical clinics; colleges; or libraries, all within a tive hundred (500) foot radius by engendering sounds, activities, visual depictions or advertisements that create an exterior atmosphere which unreasonably interferes with the operations of such surrounding uses.
  - 2. The requested use at the proposed location is sufficiently buffered in relation to residentially zoned areas within the immediate vicinity such that any obtrusive or distracting environmental factors which may emanate from the use do not adversely affect said areas.
  - 3. The exterior appearance of the structure will not be conspicuously of a lesser quality (i.e., with respect to such elements as building facade, lighting, and signage materials) than the exterior appearance of commercial structures already constructed or under construction within the immediate neighborhood or cause a substantial diminution or impairment of property values within the neighborhood.
  - 4. The proposed use will not be inconsistent with the adopted general plan for the area.
  - 5. The proposed site is adequate in size and shape to accommodate the parking and loading facilities, landscaping and other development features prescribed in the planning code or other City regulations or as is otherwise required in order to integrate said use with the uses in the surrounding area.
  - 6. The proposed site is adequately served:
    - a. By highways or streets of sufficient width and capacity to carry the kind and quantity of traffic and to accommodate the parking demand such use would generate; and
    - b. By other public or private service facilities such as tire protection or trash collection services as are required.
- C. Location.

- 1. No adult entertainment activity shall be located within, nor closer than one thousand (1,000) feet to, the boundary of any residential zone.
- 2. No adult entertainment activity shall be closer than three hundred (300) feet to any other adult entertainment activity except that this restriction shall not apply to any adult entertainment activity in an establishment devoted exclusively and on a full-time basis to such activity, which establishment was in existence on December 21, 1976 and operating under a valid City regulatory permit, where such a permit is required.
- D. Discontinuance of Nonconforming Activities. See Section 17.114.090.
- 17.102.170 Special-regulatione-applying-te-mMassage activities.

Massage activities as defined in the Oakland Municipal Code shall be subject to the regulations contained in the Oakland Municipal Code Section 5.36 as may be amended by the Oakland City Council.

17.102.190 - Joint Living and Work Quarters.

- A. General Provisions. Joint living and work quarters are permitted in all zones where Residential Activities are permitted or conditionally permitted. In all zones where Residential Activities are not otherwise allowed by the applicable individual zone regulations, joint living and work quarters may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
- B. Definition. Joint living and work quarters means residential occupancy by not more than four persons, maintaining a common household of one or more rooms or floors in a building originally designed for industrial or commercial occupancy which includes: (1) cooking space and sanitary facilities which satisfy the provisions of other applicable codes; and (2) adequate working space reserved for, and regularly used by, one or more persons residing therein.
- C. Use Permit Criteria. A conditional use permit for joint living and work quarters may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure and to both of the following use permit criteria:
  - 1. That the workers and others living there will not interfere with, nor impair, the purposes of the particular zone; and
  - 2. That the workers and others living there will not be subject to unreasonable noise, odors, vibration, or other potentially harmful environmental conditions.

17.102.195 - Residentially-Oriented Joint Living and Working Quarters.

- A. Area of Applicability. The provisions of Section 17.102.195 apply to the area bounded by Highway 980/Brush Street, the Estuary shoreline, the Lake Merritt/Estuary channel, the western shore of Lake Merritt, and 27th Street.
- B. Definition. Residenfially-Oriented Joint Living and Working Quarters means residential occupancy by one or more persons maintaining a common household of one or more rooms in a building originally designed for non-residential occupancy which includes cooking space and sanitary facilities which satisfy the provisions of other applicable municipal codes. A Residentially-Oriented Joint Living and Working Quarter consists of a designated residential area and a designated work area. However, the definitions applied by City Council Resolution Number 68518 C.M.S that apply to "Joint Live/Work Space" including criteria that define space requirements are not applicable to Residentially-Oriented Joint Living and Working Quarters.
- C. Conditions for Conversion.

- 1. In the area prescribed in Subsection (A), an existing building or portion of a building that was originally designed for non-residential occupancy can be converted to Residentially-Oriented Joint Living and Working Quarters as long as each of the following standards is met
  - a. The total number of Residentially-Oriented Joint Living and Working Quarter units on the subject property after the conversion will not exceed the maximum number of residential units permitted by the underlying zone.
  - b. All existing on-site parking spaces are retained for use by the residents, unless existing onsite parking exceeds required parking for all activities on the lot, in which case the number of parking spaces shall not be reduced below the number of spaces prescribed in Chapter 17.116 for all activities on the lot
  - c. All open space associated with the building is retained for use by the residents, unless existing open space exceeds the requirement for of the applicable zone or zones.
  - d. All existing ground-ticor commercial space is retained for commercial activities.
- 2. New tioor area may be created that is entirely within the existing building envelope; however, in no case shall the height, footprint, wall area, or other aspect of the exterior of the building proposed for conversion be expanded to accommodate Residentially-Oriented Joint Living and Working Quarters, except for dormers not exceeding the existing roof height and occupying no more than ten percent (10%) of the roof area, and incremental appurtenances such as elevator shafts, skylights, rooftop gardens, or other facilities listed in Section 17.108.130.
- 3. If a project is located within the S-7 zone and involves exterior alterations, the design review requirements of that zone shall apply (see Sections 17.84.030 and 17.84.040).
- In any zone, projects involving exterior alterations shall be subject to the design review procedure in Chapter 17.136.
- D. Conditional use permit required in certain instances. In the area prescribed in Subsection A, a project that involves the conversion of an existing building or portion of a building that was originally designed for non-residential occupancy to Residentially-Oriented Joint Living and Working Quarters and does not meet one or more of the requirements of Subsection (C)(1) above may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134. A conditional use permit may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in conditional use permit procedure in Chapter 17.134 and to any and all applicable additional use permit criteria set forth in the particular individual zone regulations.
- E. Non-applicability of certain requirements pertaining to dwelling units. In the area prescribed in Subsection (A), the conversion to Residentially-Oriented Joint Living and Working Quarters of a building or portion of a building that was originally designed for non-residential occupancy is not subject to the requirements for eff-street parking in Section 17.116.020 (New Parking to Be Provided for New Living Units in Existing Facilities) and is not subject to the open space requirements for new residential dwelling units contained in the applicable zoning district or districts, but is subject to the requirements of subsection (C)(i) above for retention of existing parking and open space.
- 17.102.200 Conditional-use-permit-required-for-pPedestrian bridges constructed over city streets.

In all zones, pedestrian bridges are permitted over city streets only upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.

47.402.210 -- Spocial-regulations-applying-to-Convenienco-Markets, Fast-Food-Restaurants, certain establishments-selling-alcoholic-beverages, providing-mochanical-or, and olectronic-games.

A. Use Permit-Criteria-fer-Fast-Foed-Restaurants, Cenveniense Markets, end-Establishments-Seiling Alcehelic-Beverages. A-cenditional-use-permit-for-any-cenditionally-permitted-Pest-Feed-Restaurant, Convenience Market, or Alcoholic Beverage Sales Commercial Activity may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, to any and all opplicable use permit criteria set forth in the particular individual zone regulations, and to all of the following additional use permit criteria:

1.—That-the-propesal will-net-contribute to undue-preliferation of such-uses in an area-where additional-ones-would-be undecirable, with consideration to be given to the area's function and character, problems of crime and loitering, and traffic problems and capacity;

2. That the propesal will not adversely affect adjacent or nearby churches, temples, or synagogues; public, parochial, or private elementary, junior high, or high schools; public parks or recreation centers; or public or parochial playgrounds;

3. That the proposal will not interfore with the movement of peeple along an important pedestrian street;

4. That the proposed development will be of an architoctural and visual quality and character which harmonizes with, or where appropriate enhances, the surrounding area;

5. That the design will avoid unduly large or obtrusive Signs, bleak unlandscaped parking areas, and an overall garish impression;

6. That adequate litter receptacles will be provided whore appropriate;

7.--That whore the proposed use is in close proximity to residential uses, and especially to bedroom windows, it will be limited in hours of operation, or designed or operated, so as to avoid disruption of residents' sleep between the hours of ten (10)-p.m. and seven (7) a.m. The same criteria shall apply to all conditional uso permits required by subsection. B of this section for sale of alcoholic beverages at full service restaurants;

8. That proposals for new Fast-Food-Restaurants must substantially comply with the provisions of the Oakland City Planning Cemmission "Fast-Food-Restaurant—Guidelines for Development and Evaluation" (OCPD 100-18).

B. Special-Restrictions on Establishments Selling Alcoholic Beverages.

1. No Alcoholic Beverage Saloc Commorciol Activity shall be located closer then one thousand (1,000) feet to any ether Alcoholic Beverage Sales Commercial Activity, oxcept.

a. On-sale retail licenses lecated in the Central District (defined for the purposes of this Chapter) as within the boundaries of I-980 and Brush street to the west; both sides of 27th Street to the north; Harrison Street/Lake Merritt and the Loke Merritt Channel to the east; and the Estuary to the south); er

b. Off-sale-retail-licenses-located in the Jack London district-(defined for the purposes-of this Chapter as within the bounderies of Martin Luther King Jr. Way to the west, I-880 to the north; the Lake Merritt Channel to the east: and the Estuary to the south); or

c. If the acfivity is in conjunction with a Full-Service Restaurant; or

d.-Establishments-with-twenty-five (25) or more-full-fime-equivalent (FTE) employees-or-a-total floor area of twelve thousand (12,000) square feet or more.

2. Sale of alcoholic beverages in conjunction with a Full Service Restaurant and located within any of the following restricted street areas applied to a depth of two hundred (200) feet on each side of the identified streets and portions of streets, as moocured perpendicularly from the right-of-way line thereof; International Boulevard; Foothill Boulevard; MacArthur Boulevard and West-MacArthur Boulevard; that portion of San Poble Avonue lying betwoon Highway I-980 and I-580; that portion of Edes Avonue lying between Clara Street...and Bergodo Drive, shall require a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134

3. In addition to the criteria prescribed elcowhoro in the zening regulations, a land use permit for an Alcoholic Beverage Sales. Commercial Activity-located within an Alcoholic Beverage Sales license ovorconcentrated area chall enly bo granted, and a finding of Public Convonience or Necessity made, if the proposal conforms to all of the following three criteria:

a.—That a community-need for the project is-clearly demonstrated. To demonstrate community-need, the applicant shall document in writing, specifically how the project would serve an unmet or underserved need or population within the overall-Oakland community or the community in which the project is located, and how the proposed project would enhance physical accessibility to needed goods or services that the project would provide, including, but not limited to alcohol; and

b. That the overall project will have a positive influence on the quality of life for the community in which it is located, providing economic benefits that outweigh anticipated negative impacts, and that will not result in a significant increace in calls for police service; and

c. That alcohol soles are typically a part of this type of business in the City of Oakland (for example and not by way of limitation, alcohol sales in a laundromat would net meet this critoria).

4. In addition to the above criteria, projects outside the Central District and Hegenberger Corridor shall meet all of the following criteria to make a finding of Public Convenience or Necessity, with the exception of those projects that will result in twenty-five (25) or more full time equivalent (FTE) employees and will result in a total floor area of twelve thousand (12,000) square feet or more:

a. The proposed-project is not within one thousand (1,000) feet of another alcohol outlet (excopt full service roctaurants), school, licenced day caro center, public pork or playground, churches, senior citizen facilities, and licensed alcohol or drug treatment facilities; and

b. Police department calls for service within the "beat" where the project is located do not excood by twenty-percent (20%), the average of calls for police service in-police beats Citywide during the preceding tweive (12) months.

C. Spocial-Restrictions on Provision of Mechanical or Electronic Games in Certain Cases. The following rogulations shall apply to the prevision of pinball machines, video game devices, or other mechanical or electronic games can be played or operated by the public or by customerc; provided, however, that thece regulations shall not apply to the provision of a total of fewer than three mechanical or electronic games in ony single place of business, except where the games provide the main or primary source of income for the proprietor; and further provided that these regulations shall not apply to the provision of a total of games provide the main or primary source of income for the proprietor; and further provided that these regulations shall not apply to the provision of any number of such games in any pool or billiard room or bowling alley for which a permit is required pursuant to Chapter 5.02 of the Oakland Municipal Code and from which persons under eighteen (18) are barred at all times by the owner or operator, nor in any premises which are licensed by the State Department of Alcoholic Beverage Control for on sale consumption of alcoholic beverages and which do not lawfully allow minors.

1. It shall not be located within three hundred (300) foot from any lot in o residential zone; or within one-thousand (1,000) foot from the nearest regular entrance to or exit from any public playground or public, parochial, or private elementary, junior high, or high school.

Those distances sholl be measured horizontally in the most direct pedestrian route along or across any-ctreet or streets, alloys, or paths, or private weys described in Section-17,106.020, leading to the elosest regular entrance to the actual space devoted to said games.

D. Special Restrictions Applying to Fast-Food Restaurants.

1.--No Fast-Food Restaurant Commorcial Activity shall be located within a one thousand (1,000) foot radius of an existing or approved Fast-Food Restaurant, as measured from the center of the front property line of the proposed site, except in the Central District (defined for the purposes of this Chaptor as within the boundaries of I-980 and Brush Street to the west; both sides of 27th Street to the North; Harricon Street/Lake Merritt and the Lake Merritt Channel to the oast; and the Estuary to the south), within the main building of Shopping Center Facilities, and in the CR-1 Regional Commercial zone.

2. Fast-Food Restaurants with Drivo-Through Facilities shall not be located within five hundred (500) feet of a public or private elementary school, pork, or playground, measured perpendicularly from the street right of way.

3. Access. Ingress and ogress to Fast Food Facilities shall be limited to commercial arterial streets rather than rosidential sfroots. No direct access sholl be provided to adjacent residential streets which are

less than thirty-two (32) feet in pavement width. Exceptions to either of the requirements may be obtained where the City Traffic Engineer determines that compliance would deteriorate local circulation or jeopardize the public safety. Any such determination shall be stated in writing and shall be supported with findings. Driveway locations and widths and entrances and exits to Fast-Food Facilities shall be subject to the approval of the City Traffic Engineer.

4.--Trash-and-Litter. Disposable containers, wrappers and napkins utilized by Fast-Food Restaurants shall be imprinted with the restaurant name or logo.

5. Vacatod/Abondoned Fast-Food Facilities. The project sponsor of a proposed Fast-Food Facility shall be required to obtain a performance bond, or other security acceptable to the City Attorney, to cover the cost of cocuring and maintaining the facility and site if it is abandoned or vacated within a prescribed high-risk period. As used in this code, the words "abandoned" or "vacated" shall mean e facility that has not-been operational for a period of thirty (30) consecutive deys, except where nonoperation is the result of mainfenance or renevation activity pursuant to valid City-permits. The defined period of coverage is four (4)-years following the obtaining of an occupancy permit. The bend may be renewed annually, and proof of renewal shall be forwarded to the Director of City Planning. The bend amount shall be determined by the City's Risk Manager and shall be adequate to defray expenses associated with the requirements outlined below. Monitoring and enforcement of the requirements set forth in this section shall be the responsibility of the Building Official, pursuant to Chapter 8.24 of the Oakland Municipol Code and those sections of the Oakland Building Code which are applicable.

If a Fact-Food Facility-has been vacated or abandoned for more than thirty (30)-consecutive days, the project sponsor shall be required to comply with the following requirements, pursuant to the relevant cited City, county and state codes:

a. Enclose the property with a security fence and secure the facility;

b. Post-signc indicating that vehicular parking and storage aro prohibited on the site (10.16.070 O.T.C. and 22658 C.V.C), and that violators will be sited, and vehicles towed at the owner's expense, and that-it is unlawful to litter or dump waste on the site (Sections 374b.5 C.P.C. and 374b-C.P.C.). All signs shall conform to the limitations on signs for the specific-zone and shall be weatherproof and of appropriate-size and standard design for the particular function;

c. Install and maintain-security lighting as appropriate and required by the Oakland Police Department;

d. Keep the site free of handbills, posters and graffiti and clear of litter and debris pursuant to Section 8:38:160 of the O.M.C.;

e. Maintain existing landscaping and keep the site free of overgrown vegetation.

### 17.102.212 - Special Regulations Applying to Residential Care, Service-Enriched Permanent Housing, Transitional Hensing, and Emergency-Shelter-Residential Activities.

A. Additional Use Permit Criteria. A conditional use permit for any conditionally permitted Residential Care, Service-Enriched Permanent-Housing, Transifional-Housing, or Emergency Shelter Residential Activity may only be granted upon determination that the proposal conforms to the general uso permit criteria set forth in the conditional uso permit procedure in Chapter 17.134, to ony and oll applicable use permit criteria set forth in the particular individual zone regulations, and to all of the following additional use permit criteria:

That staffing of the facility is in compliance with ony State Licensing Agency requirements;

2.—That if located in a residential zone, the operation of buses and vane to transport residents to and from off-site activities does not generate vehicular traffic substantially greater than that normally generated by Residential Activities in the surrounding area;

3.--That if located in a residential zone, the on-street-parking demand generated by the facility due to visitors is not-substantially greater than that normally generated by the surrounding-Residential Activities;

4. That-if-located in a residential zone, arrangements for delivery of goods are made within the hours that are compatible with and will-not adversely affect the livability of the surrounding-properties;

5. That the facility's program does not generate noise at levels that will adversely affect the livability of the surrounding properties.

B. Restriction on Overconcentration of Resident-Core, Sorvico-Enriched Permanent Housing, Transitional Heusing, and Emergency Shelter Residential Activities. No Residential Care, Service-Enriched Permanent-Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such Activity or Facility.

17.102.220 - Special regulations applying to mining and quarrying extractive activities.

Sections:

4.0 Purpose-and Intent.   2.0 Definitions.   2.0 Incorporation-by-Reference.   3.0 Incorporation-by-Reference.   4.0 Scope.   6.0 Scope.   6.0 Process.   7.0 Standards-for-Reelamation.   8.0 Statement of Responsibility.   9.0 Findings for Appreval.   10.0 Fittansiel-Assurances.   11.0 Interm Management Plans.	·	· · · · · · · · · · · · · · · · · · ·
3.0 Incorporation by Reference:   4.0 Scope:   6.0 Vested Rights:   6.0 Process:   7.9 Standards for Reelamation:   8.0 Statement of Responeibility:   9.0 Findings for Approval.   10.0 Fittansial Assurances:	<del>1.0</del>	Purpose and Intent.
3.0 Incorporation by Reference:   4.0 Scope:   6.0 Vested Rights:   6.0 Process:   7.9 Standards for Reelamation:   8.0 Statement of Responeibility:   9.0 Findings for Approval.   10.0 Fittansial Assurances:		
4.0 Scope.   5.0 Vested Rights.   6.0 Process.   7.0 Standards for Reelamation.   8.0 Statement of Responeibility.   9.0 Findings for Approval.   10.0 Fittansial Assurances.	<del>2.0</del>	
5.0 Vested-Rights.   6.0 Processe.   7.0 Standards for Reelamation.   8.0 Statement of Responeibility.   9.0 Findings for Approval.   10.0 Firtansial-Assurances.	<del>3.0</del> 	Incorporation by Reference.
6.0 Process.   7.0 Standards for Reelamation.   8.0 Statement of Responsibility.   9.0 Findings for Åpproval.   10.0 Firtansial Assurances.	4 <del>.0</del>	Scope.
7.0   Standards for Reelamation.     8.0   Statement of Responeibility.     9.0   Findings for Approval.     10.0   Firtansial Assurances.	5.0	<del>Vested Rights.</del>
7.0   Standards for Reelamation.     8.0   Statement of Responeibility.     9.0   Findings for Approval.     10.0   Firtansial Assurances.	6 <del>.0</del>	Process.
8.0 Statement of Responeibility.   9.0 Findings for Approval.   10.0 Firtansial Assurances.		
10.0 Firtansial Assurances.	8.0	Statement of Responeibility.
	<del>9.0</del>	
	1 <del>0.0</del>	-
12.0 Annual Report Requirements.	<del>12.0</del>	
43.0 Inspections.	43.0	

14.0	Violations-and Penalties.
<b>15-0</b>	Appeals.
46.0	Fees
<del>17.</del> 0	Severability.
48 <del>.0</del>	EffectiVe-Date.

#### §-1.0 Purpose and Intent.

The City of Oakland recognizes that, historically, the extraction of minerals hac benefited the sconomic well-being of the city and the needs of society and that the reclamation of mined lands is necessary to prevent or minimize adverse effects on the environment and to protect the public health and safety. The city also recognizes that surface mining within the city occurs in a diverse, established, urban environment, which presents unique health, safety and welfare issues and where geologic, topographic, climatic, biological, and other conditions are significantly different than in less urbanized areas. Therefore, reclamation operations and the appropriate to the surrounding conditions.

Tho\_purpose\_and\_intent\_of\_this\_section\_ic to regulate\_surface\_mining\_operations\_as\_required\_by California's Surface Mining and Reclamation Act of 1975 (Public Resources Code Sections 2710 et soq.), ac amended, hereinafter referred to ac "SMARA", Public Resources Code (PRC) Section 2207 (relating to annual reporting requirements), and State Mining and Geology Board regulations (hereinafter referred to as "State regulations") for surface-mining and reclamation practice (California Code of Regulations [CCR], Titie 14, Division 2, Chapter 8, Subchapter 1, Sections 3500 et seq.), to ensure that:

(a) Reclamation activities eliminate hazards to public health and safety and restore mined lands to a standard that is safe, stable, and usable for development of rouses that will enhance the community;

(b) Adverse-onvironmentol effects are prevented or minimized in accordance with CEQA and other applicable requirements;

(c) Reclamation activitios further adopted city goals, plans, policies, objectives and regulations, including, without limitation the city's General Plan;

(d) Reclamation activities appropriately consider values relating to recreation, watershed, wildlife, range and forage, and aesthetic enjoyment.

### § 2.0 Definitions.

The definitions set forth in this soction shall govern the construction of this chapter.

"Area of Regional-Significonce" means an area designated by the State Mining and Geology-Board which is known to contain a deposit of minerals, the extraction of which is judged to be of prime importance in meeting future-needs for minerals in a particular region of the State within which the minerals are-located and which, if prematurely developed for alternato incompatible-land uses, could result in the premature loss of minerals that are of more than local significance.
"Area of Statewide Signiticance" means an area designated by the beard which is known to contain a deposit of minorals, the extraction of which is judged to be of prime importence in meeting future needs for minerals in the State and which, if premoturely developed for alternate incompatible land uses, could result In the permanent loss of minerals that are of more than local or regional cigniticance.

"Approved Plan" moans a land-use and/or dovelopmont plan and all conditions of approval and adopted mitigation measures, as approved by the city pursuant to Chapter 17 of the Oakland Municipal Cede.

"Borrow Pits" mean excavations created by the curface mining of rock, unconsolidated geologic doposits or soil to provide material (borraw) for fill elcowhere.

"City" means City of Oakland.

"City-Council" means City Counsil of the City of Oakland-

"Compatible Land Uses" means land uses inhorently compatible with mining and/or-that-require a minimum public or private investment in structuros, land improvements, and which may allow mining because of the relative economic vakie of the land and its improvements. Examples of such uses may include, but shall not be limited to, very low density rocidential, goographisally extensive but low impact industrial, recreational, agricultural, silvicultural, grazing, and open space.

"General Plan" means the General Plan of the City of Oakland.

"Haul-Road" means a road along which material is transported from the area of excevation to the processing plant or ctock pile area of the surface mining operation.

"Idle"-means-surface-mining-operations-curtailed-for-a-period-of-one-year-or-mere, by more than ninety (90) percent of the operation's previous maximum annual mineral production, with the intent-to resume those surface mining operations at a future-date.

"Incompatible-Land Uses"-means-land-uses-inherently-incompatible-with-mining-and/or-that-require public or private-investment-in-structures, land-improvements, and landcoaping-and-that-may-prevent mining because of the greater economic value of the land and its-imprevements. Examples of such uses may-include, but shall not be limited to, high density residential, low density-residential-with high-unit value, public facilitios, geographically limited but impact intensive industriol, and commercial.

"Minod Lands" moon the surface, subsurface, ond ground water of an area in which surface mining operations will be, are being, or have been conducted, including private ways and roads appurtenant to any such area, land excavations, workings, mining woste, and areas in which structures, facilities, equipment, machines, tools or other materials or property which result from, or are used in, surface mining operations are located.

"Minerals" means any naturally occurring chemical element or compound, or groups of elements and eompounds, formed from inorganic processes and organic substances, including, but not limited to, coal, peat, and bituminous rock, but excluding geothermol resources, natural gas, and petroleum.

"Operator" means any person who is engaged in surface mining operations, or who contracts with others to conduct operations on his or her behalf, except a person who is engaged in surface mining operations as an employee with wages as his or her sole compensation.

"Reotamation" means-the combined-process of land treatment that minimizes water-degradation, air pollution, damage to aquatic or wildlife habitat, tiooding, erosion, and other adverse effects from surface mining operations, including adverse surface effects incidental to underground mines, so that mined lands are reclaimed to a usable oenditien whish is readily adaptable for alternate land usee and create ne danger te-public health or safety. The procecc may oxtond to affected lands surrounding mined lands, and may require basktilling, grading, reselling, revegetation, soil compaction, stabilization, or other measures.

"Reclamation-Plan" means a plan for reclamation of minod-lands as specified by SMARA.

"Stroam Bed-Skimming" means excavation of sand and gravet from stream bod deposits above the mean-summer water level or stream bottom, whichever is higher.

"Surface Mining Operations" moons all, or any part of, the process involved in the mining of minerals on mined lands by removing overburden and mining directly from the mineral deposits, open-pit mining of minorols naturally exposed, mining by the auger method, dredging ond quarrying, or surfoce work incident to an underground mine. Surface mining operations include, but are not limited to, in place distillation or retorting or leaching, the production and disposal of mining waste, prospecting and exploratory activities, borrow pitting, streambed skimming, and segregation and stockpiling of mined materials (and recovery of same).

"Use-Permit" means a conditional use permit or other land use permit for mining octivities.

§-3-0-Incorporation by Reference.

The previsions of SMARA (PRC § 2710 et soq.), PRC Section 2207, and State regulations CCR § 3500 et seq., ac those previsions and regulations may be amended from time to time, are made a part this section by reference with the came force and effect as if the previsions therein were specifically and fully set-out herein, excepting that when the provisions of this section are more restrictive than correlative State provisions, this section shall prevail.

#### §-4-0-Scope

Except as provided in this section, no person shall conduct surface mining operations unless a Reclamation Plan and financial assurances for roclamation have first been approved by the city. Any applicable exemption from this requirement does not automatically exempt or limit a project or activity from the application of other regulations, ordinances or policies of city, including but-not limited to, the application of CEQA, the requirements of other dedications and exactions as may be permitted under the law. The provisions of this section shall apply to oil lands within the city, public and private.

This section shall not apply to the following activities, subject to the above-referenced exceptione:

(a) Exsavations or grading conducted for farming or on-site construction or for the purpose of restoring land following a flood or natural disaster.

(b) Qnsite excavation and onsite earthmoving activities which are an integral and necessary part of an approved construction project that are-undertaken to prepare a site for construction of structures, landscaping, or other land improvements, including the related excavation, grading, compaction, or the creation of fills, road cuts, and embankments, whether or not surplus materials are exported from the site, subject to all of the following conditions:

(1) All-roquired permits for the construction, landscaping, or related land improvements have been approved by a public agency or agencies in accordance with applicable provisions of state law and locally adopted plans and ordinances, including, but not limited to, the California Environmental Quality Act ("CEQA", Public Resources Code, Division 13, §-21000 et seq.).

(2) The city's approval and CEQA review (if applicable) of the construction project-included the onsite excavation and onsite earthmoving activities.

(3) The approved construction project is consistent with the General Plan and zoning of the site.

(4) Surplus-materials shall not be exported from the site unless and until-actual construction work has commenced and shall cease if the city determines, in its discretion, that construction activitios have terminated, have been indefinitely suspended, or are no longer-being actively pursued.

(c) Permitted oporation of a plant-site-used for mineral processing, including ascociated onsite structures, equipment, machines, tools, or other materials, including the onsite stockpiling ond onsite recovery of mined materials, subject to all of the following conditions:

(1) The plant site is located on lands designated for industrial or commercial uses in the city's general plan.

(2) The plant site is located on lands-zoned industrial or commercial, or are contained within a zoning category intended exclusively for industrial activities by the city.

(3) None-of the minerals being processed is being extracted onsite.

Oakland, California, Code of Ordinances

(4) All-reclamation-work-has been completed pursuant to the approved Roclamation Plan for any mineral extraction activities that occurred ensite after January 1, 1976.

(d) Prospecting for, or the extraction of, minerals for commercial purposes and the removal of everburden in total amounts of less than one thousand (1,000) cubic yards in any one location of one acre or less.

(e) Surface mining operations that are required by federal law in order to protect a mining claim, as specified in Public Resources Code section 2714(e).

(f)-Any-other surface mining operations that the State-Mining and Geology Board detorminos to be of an infrequent nature and which involve only minor-surface disturbances.

(g) The solar evaporation of sea water or bay water for the production of salt and related minorals.

(ii) Emergency excavations or grading conducted by the Department of Water Resources or the Reclamation Board for the purpose of averting, alleviating, repairing, or restoring damage to property due to imminent or recent floods, disasters or other omergencies.

(i) Road construction and maintenance for timber or forest operations, as specified in Public Resources Code section 2714(j)(1); and

(j) Excavation, grading, or other earthmoving activitios in an oil or gas field, ac specified in Public Resources Code section 2714(k).

#### § 5.0 Vested Rights.

No person who obtained a vested-right to conduct surface mining operations prior to January 1, 1976, shall be required to secure a permit to mine, so long as the vested right confinuos and as long as no substantial changes have been made in the mining operation except in accordance with SMARA, State regulations, this section and any other applicable requirements. Where a person with vested rights has continued surface mining in the same area cubsequent to January 1, 1976, he shall obtain city approval of a Reclamation Plan covoring the minod lands disturbed by such subsequent curface mining. In these cases where an overlap exists (in the horizontal and/or vertical sense) between pre- and pest-Act mining, the Reclamation Plan shall call for reclamation proportional to that dicturbance caused by the mining after the effective date of the Act (January 1, 1976), as determined by the city to be necessary or appropriate to accommodate rouse of the proposed site according to city plans, policies, ordinances, and other applicable requirements.

All-other requirements of State-law, this-soction or an approved plan shall apply to vested mining operations.

§ 6.0 Process.

(a) Applications under the requirement for an Approved Plan or Reclamation Plan for surface mining or land reclamation projects shall include, at a minimum, each of the elements required by SMARA (§ 2772-2773) and State regulations, and any other requirements determined, in the discretion of the Planning-Director or designee, to be noceccary or appropriate to facilitate an evaluation of the proposed Reclamation Plan.

(b) Within thirty (30) days of the accoptance of a complete application for a Reclamation Plan or as a requirement of an Approved Plan for surface mining operations and/or a Reclamation Plan, the Planning Department shall notify the State Department of Conservation of the filing of the opplication(s). Whenever mining operations are proposed in the one-hundred (100) year-fieod plain of any stream, as shown in Zone A of the Flood Insurance Rate Maps iscued by the Federal Emergency Management-Agency ("FEMA"), and within one-milo, upstream or downstream, of ony-state-highway bridge, the Planning Department shall also notify the State Department of Transportation ("Caltrans") that the application nas been received.

(c) Tho-Planning-Department-shall-procece the application(s) in accordance with the California Environmental Quality-Act (Public-Resourcec-Code Sections 21000 et sog.) and the City's environmental review guidelines.

(d) Subsequent to tho appropriato onvironmental review, the Planning Department shall prepare a staff-report-with recommendations for consideration by the City Planning Commission. The City Planning Commission shall hold at least one noticed public hearing on Use Permit and/or Roclamation Plan. Notice shall be given by mail or delivery to all persons shown on the last available equalized ascossmont role as owning-real property in the city limits within three hundred feet (300 feet) of the property involved. All such notices chall be given not less than sovonteen (17) days prior to the date-set for the hearing. At the conclusion of such hearing or hearings, the Planning Commission sholl recommend to the City Council that it should approve, approve with changes, or deny the subject-Reclamation Plan and/or Use-Permit.

(e) The City Council shall hold at least one noticed public hearing on a Uso Permit and/or Reslamation Plan. Nofice shall be given by mail or delivery to all persons shown on the last available equalized assessment role as owning real property in the city limits within three hundred feet (300 feet) of the property involved. All such nofice shall be given net less than seventeen (17) days prior to the date set for the hearing.

(f) Prior to final approval of a Roclamation Plan, financial ascuranoss (as provided in this Chapter), any amendments to the Roclamation Plan, existing financial assurances, or those financial assurances required as part of an Approved Plan, the City Council shall certify to the State Department of Conservation that the Roclamation Plan and/or financial assurance compiles with the applicable requirements of State law, and submit the plan, assurance, or amendments to the State Department of Censorvation for review.

Pursuant to PRC § 2774(d), the State Department of Conservation shall be given thirty (30) days to review and commont on the Reclamation Plan and forty-five (45) days to review and comment on the financial assurance. The Planning Department shall evaluate written comments received, if any, from the State Department of Conservation during the comment periods. Staff chall prepare a response describing the disposition of the major issues raised by the State for the City Council's approval. In particular, when the Planning Deportment's position is at variance with the recommendations and objections raised in the State's comments, the response shall address, in detail, why specific comments and suggestions were not accepted. Copies of any written commonts received and responses prepared by the Planning Department shall be promptly forwarded to the operator/applicant.

(g) The City-Council chall-then take action to approve, conditionally approve, or deny Use Permit and/or-Reclamation Plan, and to approve the financial assurances pursuant to PRC-§ 2770(d) or any other requirement of an Approved Plan.

(h)-The Planning Department-shall-forward a copy of each approved Use Permit-for mining operations, an Approved Plan and/or approved Reclamation Plan, and a copy of the approved financial ascurances to the State Department of Conservation. By July 1 of each year, the Planning Department shall submit to the State Department of Conservation for each active or idle mining operation o copy of the Approved Plan, or Reclamation Plan amendments, as applicable, or a statement that there have been no changes during the previous year.

#### §-7-0-Standards for Reolamation.

(a) All Reclamation Plans shall comply with the provisions of SMARA (§ 2772 and § 2773) and State regulations (CCR § 3500-3505). Reclamation Plane approved after January 15, 1993, Reclamation Plans for proposed new mining operations, and any substantial amondments to previously approved Reclamation Plans, shail also somply with the requirements for reclamation performance standords (CCR § 3700-3713).

(b) Tho-city-may-impoce additional-perfornance-standards-as-developed-either in review-of-individual projects, as-warranted, through the formulation-and adeptien-of-citywide performance-standards-or through an Approved Plan.

(o)-Reclamation\_activities\_shall\_be\_initiated\_at\_the\_earliest\_possible\_time\_on\_those\_portions\_of\_the mined-lands-that-will\_not\_be\_subject\_to\_further\_disturbance.-Interim\_reclamation\_may\_alsc\_be\_required\_for minod-lands-that have been disturbed\_and-that may\_be disturbed again in future operations: Reclamation may be-done-on an annual-basis, in stages compatible with continuing operationc, or on completion of all excavation, removal, or fill, as approved\_by-the\_city. Each\_phase\_of\_reclamation\_shall\_be\_specifically

described in the Reclamation Plan and shall-include (a) the beginning and expected ending dates for each phase; (b) all reclamation activities required; (c) criteria for measuring completion of spesific roclamation activities; and (d) estimated costs for completion of each phase of reclamation.

(d) The information, analysis and other cpocifications submitted as part of the Reclamation Plan shall demonstrate that the improvements and financial assurances are sufficient to roclaim the site in a condition that meets all applicable state and sity otandards, and that is approprioto for the proposed reuse of the site ond consistent with the land use and other applicable policies of the General Plan.

§ 8.0 Statement of Responsibility.

The person submitting the Roclamation Plan shall sign a statement accopting responsibility for reclaiming the mined lands in accordance with the Reclamation Plan. Said statement shall be kept by the Planning Department in the mining operation's permanent record. Prior to sale or transfer of the operation, the new operator shall submit a signed statement of responsibility as well as evidence required to demonstrate the financial assurance requirement cot forth in this section or the Planning Department record.

§-9.0 Findings for Approval.

(a) Approved Plans. In addition to any findings required by the Approved Plan or for surface mining operations, a finding-shall be included that the project complies with the provisions of SMARA and State regulations.

(b) Reclamation Plans. For Reclamation Plans, the following-findings shall be required:

(1) That the Reclamation Plan complies with SMARA Sections 2772 and 2773, and any other applicable provisions;

. (2) That-the-Reclamation-Plan-complies with applicable requirements of State-rogulations (CCR-§ 3500-3505, and § 3700-3713).

(3) That the Reclamation Plan and potential-use of reclaimed land pursuant to the plan are consistent with this section, the city's General Plan and any applicable resource plan, element or an Approved Plan.

(4) That the Reclamation Plan has been reviewed pursuant to CEQA-and the City's environmental review guidelines, and oil significant adverse impacts from roclamation of the curfooo mining operations ore mitigated to the maximum extent foasiblo.

(5) That the land-and/or resources such as water-bodies to be reclaimed will be restored to a sondition that is compatible with, and blends in with, the surrounding natural environment, topography, and other resources, or that suitable off-site development-will compensate for related disturbance to resource-values.

(6) That the Reclamation-Plan will restore the mined lands to a safe, stable and usable condition which is readily adaptable for alternative land uses consistent with the General Plan, and other city Approved Plans, policies, ordinances and regulations.

(7) That a written response to the State Department of Concervotion has been prepared, describing the disposition of major issues raised by that department. Where the city's position is at variance with the recommendations and objections raised by the State Department of Conservation, said-response shall address, in detail, why specific comments and suggestions were not accepted.

§ 10.0 Financial Assurances.

(a) To ensure that reslamation will proceed in accordance with the approved Reclamation Plan, the city shall require as a condition of approval security which will be released upon satisfactory performance. The applicant may pose sesurity in the form of a surety bond, trust-fund, irrevocable letter of credit from an accredited financial institution, or other method acceptable to the city and the State Mining and Geology Board as specified in State regulations, and which the city determines are adequate to perform reclamation in accordance with the surface mining operation's approved Reclamation. Plan and/or on Approved Plan. Financial assurances shall be made payable to city and the State Department of Conservation.

(b) Financial-assurances-will-bp-required-te-ensure-compliance-with-elements-of-the Reclamation Pian, including-but-not limited-to, revegetation-ond-landscaping-requirements, restoration-of-aquatic-or wiidiife-habitat, restoration of water-bodioc-and-water-quality, slope stability-and-erosion and-drainage control, disposal of-hazardous-materials, and other measures, if determined-necessary-by-the Planning Department to comply-with-the-requirements of an Approved Plan.

(c) Cost-estimates-for-the-financial-assurance-shali-be-submitted to the Planning Department-for review and approval-prior-to-the operator-cocuring financial-assurances. The Planning-Director-shall forword a copy of the cost-estimates, together with any documentation received supporting the amount of the-cost-estimates, to the Department of Conservation for review. If the State Department of Conservation does not comment-within forty-five (45) days of receipt of those estimates, it shall be assumed that the eest-estimates are adequate, unless the City has reason to determine that additional costs may be incurred. The Planning Director sholl have the discretion to approve the financial assurance if It-meets the requirements of this Chaptor, SMARA, State regulations and any requirements of an Approved Plan.

(d) The amount of the financial assurance shall be based upon the estimated costs of reclamation to a-safer-stable-and-usable-condition-in-accordonco-with an Approved Plan for the years or phases stipulated-in-the-approved-Reclamation-Plan, including-any maintenance of reclaimed-oreas as may be required, subject to adjustment for the actual amount required to reclaim lands disturbed by surface mining-activities-since-January 1, 1976, and now lands to be disturbed in the upcoming year. Cost estimates\_should\_be~prepared\_by\_a\_Colifornia\_registered\_Professional\_Engineer~ond/or\_other\_similarly tieensed and qualified professionals retained by the operator ond approved by the Planning Director. The estimoted amount of the financial assurance shall be based on an analysis of physical activities necossary to imploment the approved Reclamation Plan in accordance with an Approved Plan for the site, the unit costs for oach of thoco activities, the number of units of each of these activities, and the actual-administrative-costs. Financial-assurances-to-ensure-Reclamation-Plan-implementation-and compliance-with-revegetation, restoration of water bodies, restoration of aquatic or wildlife habitat, and any other-applicable element of the approved Roolamafion Plon chall be bosed upon cost estimates that include, but may not be limited to, labor, equipment, materialc, mobilization of equipment, administration. monitoring, inspections and roosonoble profit by a commorcial operator other than the permittee. A contingency factor of ton porcont (10%) shail be odded to the cost of financial assurances.

(e) In--projecting\_the\_costs--of\_financial\_acsurances, it\_shall\_be\_assumed\_without-projudico\_or insinuation-that-the-surface mining operation-could be abandoned by the operator and, consequently, the city or State Department of Conservation may need to contract with a third-party commercial company for reclamation of the site.

(f)-The-financial-ascurances-shall-remain-in-effect for the duration of the surfoce mining-operation and any-additional-period-until reclamation is completed (including any maintenance-required).

(g) The amount of financial assurances required of a surface mining operation for any one-year-shall be adjusted annually to account for new lands disturbed by surface mining operations, inflation, and reclamation of lands accomplished in accordance with the approved Reclamation Plan. The financial assurances shall include estimates to cover reclamation for existing conditions and anticipated activities during the upcoming year, excepting that the permittoo may not claim credit for reclamation scheduled for completion during the coming year.

(h) Revisiens to financial assurances shall be submitted to the Planning Diroctor each year-prior to the anniversary date for approval of the financial assurances. The financial assurance shall cover the cost of existing disturbance and antieipoted activities for the next calendar year, including any required interim reclamation. If revisions to the financial assurances are not required, the operator shall explain, in writing, why revisions are not required.

§-11.0 Interim-Management Plans.

(a) Within ninety (90) days of a surface mining operation becoming idle, the operator shall-submit to the Planning Department a proposed Interim Management Plan (IMP). The proposed IMP shall-fully comply with the requirements of SMARA, including but not limited to all Approved Plan conditions, and shall-provide measures the operator will implement to maintain the site in a stable condition, taking into

concideration public health and safety. The proposed IMP chall be submitted on forms provided by the Planning Department, and shall be processed as an amendment to the Reclamation Plan. IMPs shall not be sonsidered a project for the purposes of environmental review.

(b) Financiol assurances for idle-operations shall be maintained as though the operation were active, or as otherwiso approved through the idle mine's IMP.

(c) Upon receipt of a complete proposed IMP, the Planning Department sholl forword the IMP to the State Department of Conservation for review. The IMP shall be submitted to the State Department of Conservation of least thirty (30) days prior to approval by the Planning Director.

(d) Within sixty (60) days of receipt of the proposed IMP, or a longer-period mutually agreed upon by the Planning Director ond the operator, the Planning Director shall review and approve or dony the IMP in accordance with this Chapter. The operator sholl have thirty (30) days, or o longer period mutually agreed upon by the operator and the Planning Director, to submit a revised IMP. The Planning Director shall approve or dony the revised IMP within sixty (60) days of receipt. If the Planning Director denies the revised IMP, the operator may appeal that action to the City Council. The decision of the City Council shall be final.

(e) The IMP may remain in effect for a poriod not to exceed five years, at which time the City Council may renew the IMP for another period not to exceed five years, or require the surface mining operator to commence reclamation in accordance with its approved Reclamation Plan.

#### §-12-0-Annual Report Requirements.

Surface mining operaters shall forward an annual surface mining report to the State Department of Conservation and to the City Planning Department on a date established by the State Department of Conservation, on forms furnished by the State Mining and Geology Board. New mining operations shall file on initial surface mining report and any applicable filing fees with the State. Department of Conservation within thirty (30) days of permit approval, or before commencement of operations, whichever is sooner. Any applicable foes, together with a copy of the annual inspection report, shall be forwarded to the State Department of Conservation at the time of filing the annual surface mining report.

#### § 13.0 Inspections.

The Planning Director, through the Building Department Inspection Services Division or other agency or other designee, sholl arrange for inspectien of a surface mining operation within six months of receipt of the Annual Report required in Section 12, to determine whether the surface mining operation is in compliance with applicable requirements, including, without-limitation, the Approved Plan, Reclamation Plan, approved financial assurances, and State regulations. In no event shall less than one inspection be conducted in any calendar year. Said inspections may be mode by a state-registered geologist, state registered civil onginoor, state-licensed landscope orshitoct, or state-registered forester, who is experienced in land roclomation and who has net boon employed by the mining operation in ony capacity during the previous twelve (12) months, or other qualified specialists, as selected by the Planning Director. All inspections shall be conducted using o form approved and provided by the State Mining and Geology-Board.

The Planning Department shall notify the State Department of Conservation within thirty (30) days of completion of the inspection that said inspection has been conducted, and shall forward a copy of said inspection notice ond any supporting documentation to the mining operator. The operator shall be solely respensible for all costs of Inspections required by the city in furtherance of this section in accordance with the city master fee schedule or other applicable foe agroomonts or requirements.

#### § 14.0 Violations and Penalties.

If the Planning Director, through the Building Department Inspection Services Division or other designee, based upon an annual inspection or otherwice confirmed by an inspection of the mining operation, detorminos that a surface mining operation is not in sompliance with this section, the Approved Plan, the Reclamation Plan or other applicable requirements, the city-shall follow the proceduros set forth in Public Recources Code, Sections 2774.1 and 2774.2 concerning violations and penaltios.

§ 15.0-Appeals.

Oakland, California, Code of Ordinances

A decision-by-the-City-Council-to either approve or deny a Reclamation-Plan pursuant to this section shall be considered a tinal agency action.

#### §-16.0 Fees.

The-city-shall-establish-such-fees-as-it-deoms-necessary-to-cover-the-reasonable-costs-incurred-in implementing\_this\_section\_and\_the\_State\_regulations,\_including\_but-net\_limited\_to, processing\_of applications,\_annual-reports,\_inspections,\_monitoring,\_enforcement\_and\_compliance.\_These-fees-may-be set-forth-in-the-city-master-fee-schedule; however,-failure-to-include-such-fees-in-the-master-fee-schedule shall\_not-limit\_the\_city's\_ability\_to-impose\_fees\_it\_determines\_aro\_necessary\_or\_desirable\_to-fulfill\_the purposes-of-this\_section,\_State-regulations\_and-other-applicable-requirements.\_Such-fees\_shall\_be-paid-by the-operator,-as-required-by-the-city,\_at-the-time-of-tiling-of-the-Reclamation\_Plan-application,\_as-a-part-of-a fee\_agreement\_through\_an\_Approved\_Plan\_or-at-such\_other\_times\_as-are\_determined\_by-the-city\_to-be appropriate\_in-order\_to-ensure\_that\_all-reasonable\_costs\_of-implementing\_this\_soction\_aro\_borne\_by-the mining-operator.

17.102.230 - Special regulations-applying-to-the-dDemolition of a facility containing rooming units or to the conversion of a living unit to a Nonresidential Activity—Nonresidential zones.

- A. Conditional Use Permit Requirement. The demolition of a facility containing, or intended to contain, rooming units, or the conversion of a living unit from its present or last previous use by a Permanent Residential Activity, a Semi-Transient Residential Activity, or a Transient Habitation Commercial Activity to its use by a nonresidential activity other than Transient Habitation Commercial is only permitted in a nonresidential zone upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134. The only exceptions to this requirement are conversions in the HBX zones, and units in a One-Family or Two-Family Residential Facility. Such permit may be granted only upon determination that the proposed demolition or conversion conforms to the general use permit criteria set forth in the conditional use permit procedure and to at least one of the following additional use permit criteria:
  - 1. That the facility proposed for demolition or the living unit proposed for conversion is unoccupied and is, or is situated in, a residential building that has been found, determined, and declared to be substandard or unsafe pursuant to Section 15.08.350(B) of the Oakland Municipal Code; or
  - 2. That a replacement rental unit, comparable in affordability and type to each unit proposed for demolition or conversion, will be added to the City's housing supply prior to the proposed demolition or conversion taking place; or
  - 3. That the benefits to the City resulting from the proposed demolition or conversion will outweigh the loss of a unit from the City's housing supply; or
  - 4. That the conversion will be an integral part of a rehabilitation project involving both residential and nonresidential activities, and that the rehabilitation project would not be economically feasible unless some nonresidential activity were permitted within it.
- B. Tenant Assistance. Upon the granting of a conditional use permit for the demolition of a facility containing rooming units or for the conversion of a living unit to a nonresidential activity, the actual demolition or conversion cannot take place until the following have occurred:
  - If a dwelling unit is to be converted, the tenant has been given a one hundred twenty (120) day written notice of the conversion. If a rooming unit is to be demolished or converted, the tenant, if a permanent tenant, has been given a seventy-tive (75) day written notice of the demolition or conversion. All such written notices shall comply with the legal requirements for service by mail.
  - 2. If a dwelling unit is to be converted, the tenant has been provided with a relocation allowance equal to one month's rent or five hundred dollars (\$500.00), whichever is greater. If a rooming unit is to be demolished or converted, the owner of the building containing the unit to be demolished or converted has referred the tenant (if a permanent tenant) to a comparable, available unit; if a comparable unit is not available, the permanent tenant has been provided

with a relocation allowance equal to one month's rent or five hundred dollars (\$500.00), whichever is greater.

3. The Director of City Planning has been provided with proof that the above actions have been taken.

(As used in this section, a permanent tenant of a rooming unit is defined as a tenant maintaining occupancy for six (6) months or more at a hotel or motel where the innkeeper does not retain a right of access and control of the unit and where the hotel or motel does not provide or offer all of the following services to all of the residents: safe deposit boxes for personal property; central telephone service; central dining; maid, mail, room, and recreational service; and occupancy for periods of less than seven (7) days.)

17.102.240 - Speciat-regulations applying-to-mMicrowave and satellite dishes over three (3) feet one (1) meter-in diameter located in or near residential zones.

The following regulations shall apply to microwave and satellite dishes which are over three (3) feet one-(1)-meter-in diameter, which are located in any residential zone or within one hundred fifty (150) feet from the nearest boundary of any residential zone, as measured perpendicularly from said boundary at any point:

17.102.250 - Maximum density and floor-area ratio during construction.

Whenever-a-new-Residential Facility is constructed on any lot-upon-which-there-presentiy-exists a Residential Facility, and euch-existing-facility is retained and occupied-temporarily-pending completion of the new-residential structure, the maximum density and floor-area-ratio proscribed for euch-lot-shall be computed upon-the basis of the new-fecility-only. However, such existing facility shall be vecetod and demolished or removed within one (1) year after commencement of construction of the new-facility unless the existing end-new-facility together shall conform to said maximum density and floor-area-ratio proscribed.

17-102-260 - Occupancy-of-a-dwelling-unit-

A-Residential-Facility, or-portion-thereof, shall-be-deemed-to-constitute-a-single-dwelling-unit-only-if-it is-occupied-by-a-single-family-or, whero-the-facility-occupied-is-a-One-Family-Dwelling, such-family and-not-more-than-three-(3)-boarders, roomers, or-lodgers-where-access-to-all-rooms-occupied-by such-boarders, roomers, or-lodgers-is-had-through-the-main-entrance-of-the-dwelling-unit-

17-102.265 - Ossupancy-of-a One-Family-Dwelling Residential Facility-by-a Residential-Care Rosidential-Astivity-

A Residential-Cere-Residential Activity-shall-be-deemed-to-occupy a-One-Family Dwolling Residential-Facility-if-it-operates-as-a-single-housekeeping-unit, as-defined-in-Sec. 17.09.040, and tho facility-meets-all-of-the-characteristics-of-a-One-Family-Dwelling-Residential-Facility-as-defined-in Section-17.10.640.

17.102.270 - An additional kitchen for a single dwelling unit.

An additional kitchen for a single dwelling unit in any Residential Facility may be permitted, without thereby creating an additional dwelling unit, upon the granfing of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, and upon determination that all of the following conditions set forth below exist:

 $\wedge$ 

- A. That the additional kitchen shall will be be located within the same residential structure as the existing kitchen and solely constitute an additional service facility for the resident household, family or its temporary guests.
- B. That the additional kitchen shall-will not serve as a basis for permanent habitation of an extra household or family on the premises, or the creation of an additional dwelling unit on the premises.
- G That the additional kitchen is necessary to render habitable a living area occupied by one or more persons related by blood, marriage, or adoption to the resident family or collective household occupying the main portion of the dwelling unit

However, a conditional use permit under this subsection shall not be granted in the RH zones or the RD-1 zone if the lot contains two (2) or more dwelling units.

#### 17.102.280 -- Rules for determining the number of habitable rooms in Residential Facilities.

The total number of habitable rooms in o Rosidontial Facility shall be determined by odding together all rooms in all dwelling units in the facility, in accordance with the rules of subsections A through F of this section. In a case where application of these rules results in more than one possible interpretation of the total number of rooms, or where those rules appoor to contradict oach other, the interpretation resulting in the greatest number of rooms shall be used. For purposes of this section, a "kitchon" shall be doomed to include the fiour area within three (3) foet directly in front of all kitchen counters, cabinets, major appliances, and other fixtures.

A.- Except as specified in subsections B through F of this section, a space which meets the definition of "habitable-room" at Section 17.09.040, which is entirely enclosed by floor to coiling partitions, and which is connected to other rooms or spaces by doorways or open archways shall count as one room.

B. A habitable room of less than fifty (50) square feot shall count as half a room.

C. A habitable room larger than four hundred (400)-cquare feet shall count as one room for each four hundred (400) square feet or fraction thereof.

D.-Spaces which are not separated by fioor to ceiling partitions but whose floor levels differ by more than one feet and which are intended to be used for different functions shall count as separate rooms.

E.-A kitchon area of a least-fifty (50) squaro-feet which is not entirely enclosed by floor to ceiling partitions shall count as a separate room.

F. A kitchon area of less than fifty (50) square feet-whose fioor perimeter is at least-fifty percent (50%) enclosed by any combination of partitionc, counters, cabinets, major appliances, and other similar space dividers shall count as half a room; if not co enclosed, it shall not count as a separate room.

#### 17.102.250 -- Special-regulations for Drive-Through Nonresidential Facilities.

The following regulations shall apply to Drive Through Nonrosidential Facilities wherever permitted:

A. General Provisions/Use Permit Criteria. A Conditional Use Pormit for a Drive-Through Nonresidential Facility may be granted only pursuant to the conditional uco permit procedure in Chapter 17.134, and upon determination that the proposal, in addition to the general uso permit criteria in that chapter, conforms to the additional use permit criteria set forth below:

1. That the proposed facility will not impair a generally continuous wall of building focados;

2. That-the proposed facility will not result in weakening the concentration and continuity of retail focilitios of ground level, and will not impair the retention or creation of a chopping frontage;

 That the proposed focility will not directly result-in-a-significant reduction in the circulation-level-of service-of-adjacent-streets. B. Standards. A driveway-serving os o vehicle stacking or queuing lone for a drive-through window shall be separated from parking areas and shall not be the only entry or exit lone on the premises. Such facility shall be so situated that any vehicle overflow from it shall not spill onto public streets or the major circulation aisles of any parking lot. Such facility shall have durable, all-weather surface; chall have reasonable disposal of surface waters by grading and drainage; and shall be permanently maintained in good condition.

C. Dimonsions- Eoch-vehicle space comprising a stocking or queuing lone-for a drivo-through window sholl be o minimum of ten (10) feet in width by twenty (20) feot in length. Such a stacking or queuing lone sholl hove o moximum capacity of eight (8) vohicles.

17.102.300 - Conditional use permit for dDwelling units with five or more bedrooms.

- A. Use Permit Required. No existing Residential Facility shall be altered, through additions, division of existing rooms, or other means, so as to create a total of five (5) or more bedrooms in any dwelling unit except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
- C. Use Permit Criteria. A conditional use permit under this section may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134 and to all of the following additional use permit criteria:
  - 1. That off-street parking for residents of the entire facility, including any existing facility and any proposed alteration or addition, is provided as specified in the zone or zones in which the facility is located, as set forth in Section 17.116.060.
  - 2. That a minimum of one (1) off-street visitor parking space is provided for the entire facility;
  - 3. That the parking spaces provided in accordance with criteria 1 and 2, and all associated driveways, maneuvering aisles, and other related features, comply with the standards for required parking and loading facilities applicable in the base zone in which the facility is located, as set forth in Sections 17.116.170 through 17.116.300\_
  - 4. That no required parking spaces are located other than on approved driveways between the front lot line and the front wall of the facility or its projection across the lot.
  - 5. That the applicable requirements of the buffering regulations in Chapter 17.110 are met.

47.102.310 -- Special regulations-for-certain-projects-with-development-agreements,

Any-person having-a legal or equitable interest in the real property involved may, upon approval pursuant to the dovelopment agreement procedure in Chapter 17-138, enter into a development agreement-with-the-City-for-any-specific-development-project-which-involves-a-total-of-at-least-four-(4) acros of land area or five hundred-thousand (500,000) square feet of floor area and is a project intended to-be developed in stages, or which involves land sold or loosed by the Redevelopment Agency of the City-and-is-to-bo-carried-out-by agreement-with the Redevelopment Agency.- The development agreement stall not be approved unless the project has received, or simultaneously roooives, whatever major conditional-use-permit, preliminary-planned-unit-development-plan-approval, and major variance-it-may otherwise-roquire. For-tho-duration-of-the-particular-agreement, and unless otherwise-provided-in-the terms-thereof, there-shall be a contractual guarantee that the project covered by the agreement may be pursued under the applicable-procedural criteria, if any, and other-zoning regulations, and plans or other decuments-referred-to-by ony-such-criteria, as they existed when the agreement-was approved and notwithstanding any-subsequent changes in said zoning regulations or documents. However, the agreement may also subject the proposal to special conditions to benefit or protect the City-for entering into the development agreement. The conditions may include, but are not limited to, supplemental restrictions on kinds of usos, floor area ratio, or density; special conditions or critorio for required subsequent zoning approvals, if ony, and requirements for the reservation, dedication, or improvement of land for public purposes or occossible to the public.

Oakland, California, Code of Ordinances

17.102.320 -- Conditional uso permit for waiver of certain requirements in mini-lot developments.

A. Basic Provisions. Subject to the provisions of subsections B and C of this section, the maximum height and minimum yard, lot area, width, and frontage requirements otherwise applying to individual lots may be waived or modified within a mini-lot development, ond floor area, parking, and ether facilities may be located within said development without reference to lot lines, upon the gronting of a conditional use pennit pursuant to the conditional use permit procedure in Chapter 17.134 and upon determination:

1. That there is adequate provision for mointenance of the open space and other facilities within the development; and

2. That the total development meets all the requirements that would apply to it if it wore a single lot

B. Zones in Which-Requirements May-Bo Waived. A conditional use-pormit-pursuant to subsection A of this section may be granted only in the S-1 or S-2-zone or in any residential or commercial zone other than RH-zones or the RD-1-zone.

C. Maximum Size for Which-Requirements May Be Waived. A conditional use permit pursuant to subsection. A of this soction may be granted only if the total land area of the mini lot development is less than sixty thousand (60,000) square feet.

17-102.330 - Conditional-uso-pormit-for-waivor-of-certain-roquiromonts-with parcel division between existing-buildings-

Whore any parcol containing two or more-oxicting principal-buildings is divided in accordance-with the conditions stated in Section 17.106.010, those-requirements specified there which would otherwise apply to the divided lots may be waived or modified upon the granfing of a conditional use permit pursuont to the conditionol use permit procedure in Chopter 17.134. Granting of any such pormit shall be subject to the use permit critoria prescribed by Section 17.106.010.

17-102.335 - Standards for Sidewalk Cafes-

A. Procedures for Construction of Sidewalk Cafe Facilities.

1. Not-withstanding-any design-review requirement of the particular zone, Sidewalk Cafes that have a maximum of five (5) tablec and no more than fifteen (15) chairs and/or will not have any permanent structuros in the public right of way, are allowed by right subject to the standards required in subsection B of this section.

2. Sidewalk Cafes that have more than five (5) tables/fiftoen (15) chairs and/or have a permanent structure in the public right of way are oubject to small project design review in Section 17.136.030

B. Standards for Sidewalk Cafes.

1.—Sidewalk-Cafes chall net encroach upon any public right of way unless a minimum of five (5) feet of unobstructed improved cidewalk remains available for pedestrian purposes. The minimum distance shall be measured from the portion of the Sidewalk Cafe encroachment which is nearest to any obstruction within the sidewalk area. For purposes of the minimum clear path, parking meters, traffic signs, trees and all similar obstacles shall constitute obstruction.

2.--Operotors/owners-of-Sidewalk-Cafes-shall-obtain-on-oncroachment-permit-from the-city's Building Servicos-Division, and chail-comply with all-requirements-imposed-by-other-affected-departments. The encroochment-permit-shall-include-language-that-a-woste-receptacle-be-placod-outside, all-garbage/litter associated-with-Sidewalk-Cafes-must-be-removed-within-twenty-four-(24) hours, and a requirement-to obtain-liability-insurance. The city-shall-be-named\_as-an-additional-insured\_and-the-amount-of-tho insurance-shall-be-determined by the-city's Risk Manager.

3.- The operators/owners of Sidewalk-Cafes chall defend, indemnify, ond hold harmless the City of Oakland its agents, officers, and employees from ony claim, action, or proceeding (including legal costs

and otterney's feos) against the City of Oakland, its ogents, officers or employees to attock, sot aside, void or annul, on approval by the City of Oakland, the City Planning Department, Planning Commission, or City Council. The city shall promptly notify the applicant of any claim, action or proceeding and the city shall cooperate fully in such defense. The city may elect, in its sole discretion, to porficipote in the defense of said claim, action, or proceeding.

4. The operator/owners of Sidewalk Cafes shall continually bus tables and provide a final cleanup at the end-of the business day that will include litter pickup one hundred (100) foot in each direction from the site.

17.102.340 - Special regulations applying to eElectroplating activities in the M-20, M-30, and M-40 no end of the M-20, M-30, and M-40 no end of the M-20 no end of t

A. Distance Standards. No electroplating activity shall be located nor expanded within one thousand (1,000) feet from the boundary of any other zone except the <u>CIX, IG,</u> M-20, M-30, or M-40 zone, nor from any area designated "Resource Conservation Area" or "Park and Urban Open Space" in the Oakland General Plan.

17.102.350 - Regulations applying to type bacco-oriented activities.

- A. Conditional Use Permit Requirement for Tobacco-Oriented Activities. Such uses are permitted only upon the granting of a conditional use permit pursuant to Sections 17.134 and to the following additional use permit criteria:
  - No tobacco-oriented activity shall be located within, nor closer than one\_-thousand (1,000) feet to the boundary of a residential zone, school, public library, park or playground, recreation center or licensed daycare facility.

17-102-360 - Secondary Units-

- A.— Development Standards. The following rogulations shall apply to the construction, establishmont or alteration of Secondary Units wherever pormitted or conditionally permitted, as specified in each individual zone:
- Other Uses on Property. A Secondary Unit shall only be permitted on a lot that contains only one other primary dwelling unit A Secondary Unit may be approved and constructed at the same time or after the approval and sonstruction of the primary dwelling unit.
- Sale of Unit A-Secondary Unit shall-not be sold separately from the primary dwelling on the same iet.
- 3— Owner Occupancy. The legal owner shall-occupy either the primary dwelling or the Secondary Unit. Prior to issuance of a building permit for a Secondary Unit the applicant shall record as o deed restriction in the Alameda County Recordor's Office, notise of this requirement, in a form proscribed by the Diroctor of City Planning.
- 4. Maximum Permitted Floor Area. The floor area of a Secondary Unit-shall not exceed nine-hundred (900) square feet or fifty percent (50%) of the floor area of the primory dwolling, whichever is loss, oxcept that Secondary Units of up to five hundred (500) square foot in floer area are permitted regardless of the size of the primory dwelling.
- 5. Fire Flow and Water Pressure. A Sesendary Unit may be permitted only if the fire flow and water pressure in the adjoining street meets the minimum requirements as dotormined by the Fire Marshal.
- 6. Emergency Assess—Multiple Vehicular Outlets. A Secondary Unit may be permitted only on a lot which has frontage on a through street, or a dood-end street that has a total length of less than three hundred (300) feet. For the purposes of this subsection, the total length of a dead-end street choll be the distance from the intersection with the neorest through street to the farthest opposite end of the

street-right-of-way, or-private-access easement-(os-defined-by-Section-16.32.010 of tho-Oakland Municipal Code) if the private-access easement-is connected to said-dead-end-street.

- 7. Emergency Access Minimum Pavement Width. A Secondary Unit may be permitted only if all streets connecting the let to the nearest orterial street (oc designated by the City of Oakland General Plan Land Use and Transportation Element) have a minimum pavement width of at least twenty four (24) feet. The minimum pavement width limitation may be reduced to a minimum of twenty (20) feet, upon the granting of a conditional uce permit pursuant to the criteria in subsection B of this section, ond the conditional use permit procedure in Chapter 17.134
- 8:----Public-Sonitary Sower--A-Secondary-Unit-may-be-permitted only if it is served by o-public sonitary sewer-
- 9.— Architectural Compatibility. The Secondary Unit shall be clearly subordinate to the primary dwelling unit in size and location. Also, the architectural design and materials of a Secondary Unit shall match or be visually compatible with that of the primary dwelling, including the architectural style, siding material, roof shape, roofing material, trim motorial and design, window types, window trim, and window sill detail.
- 10.- Compliance with Building and Fire-Codes. All Secondary Units shall comply with all other-code and permit-requirements-imposed by all other offocted departments, including but not limited to fire separation, sound ceparation, egress, utility access, and the requirement for a building permit.
- 41. Review Procedure. An application for a Secondary Unit-of up to five hundred (500) square foot shall be granted ministerial approval as specified in Section 17.136.025 upon confirmation of compliance with all applicable zoning regulations, including but not limited to, all provisions in this Section. The five hundred (500) square-foot floor area threshold for a Secondary Unit may only be exceeded, up to a maximum of nine hundred (900) square foot or fifty-percent (50%) of the floor area of the primary dwelling, whichever is less, upon the granting of small-project design review, pursuant to the small project design review procedure in Section 17.136.030
- B. Use permit criteria for Secondary-Units accessed via narrow-streets. A conditional use-permit for a Secondary Unit-accessed from the nearest orterial street-via a street with a minimum pavement width of between twenty (20) and twenty-four (24) feet may only be granted upon determination that the proposal conforms to the general use permit criteria cot forth in the general use permit procedure in Chapter 17.134 and to all of the following additional use permit criteria:
- 1.---That there is adequate emergency accoss to the lot as determined by the Fire Marshall.
- 2.--- That the portions of the street that have a pavement width of less than twonty-four-(24) feet are not located on a dead-end street.
- 3.— That if en-street-parking is permitted on portions of the street that have a pavement width of less than-twenty-four (24) feet that there exist a level and hard surface shoulders with a combined additional width of at least eight (8) feet
- 4.— Thot-if on-street parking is prohibited en-portions of the street that have o pavement width of less than twenty-four (24) feet, that the restricted parking areas are clearly marked with official City installed no-parking signs and/or rod curbs, pursuant to the provisions of the Oakland Traffic Code (Title 10 of the Oakland Municipal Code).

#### 17-102.370 - Conditional-use-pormit for-hotels and motels.

A.— Use Permit Criteria for Hotel and Motel Uses. A conditional use permit for hotel ond motel uses may be granted only upon determination that the proposal conforms to the generol use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, to any and all applicable use permit criteria set forth in the particular individual zone regulations; and to all of the following additional use permit criteria:

- 1. That the proposal is located in downtown, along the waterfront, near the airport, or along the I-880 freeway, and/or in an area with a concentration of amenities for hotel patrons, including restaurant, retail, recreation, open space and exercise facilities, and is well-served by public transit,
- That the proposal considers the impact of the employees of the hotel or motel on the demand in the city for housing, public transit, and social services;
- 3: That the proposal-is-concistent-with the goal-of-attracting first-class, luxury-hotels-in-downtown, along the waterfront, near-the-airport, or-along the I-880 freewoy-which provide:
- a. A minimum of one hundred (100) sleeping rooms;
- b. A full service restaurant providing three meals per-day; and
- On-site-recreational-amenities, which may include an exercise reem, swimming-pool, and/or tennis eeurts.
- 4. That the proposed development will be of an architectural and visual quality-and charaster which harmonizes and enhances the surrounding area, and that cuch design includes;
- Site-planning that insures appropriate access and circulation, locates building entries which face the primary street, provides a consistent development pattern along the primary street, and insures a design that promotes safety for its users;
- Landscaping that creates a pleasant visual corridor along the primary streets with a variety of local species and high quality landscape materials;
- c. Signage that is integrated and consistent with the building design and promotes the building entry, is consistent with the desired character of the area, and does not detract from the overall streetscape;
- d. The majority of the parking to the rear of the site ond where appropriate is provided within a structured parking focility that is consistent, compatible and integrated into the overall development,
- Appropriate design-treatment for-ventilation of room-units as well as structured parking areas; and prominent entry features that may include attractive ports cocheres;
- f.— Building design that enhances the building's quality with strong arohitectural stotements, high quality materials particularly at the pedestrian level and appropriate attention to detail; and
- g. Lighting standards for hotel buildings, grounds and parking lots shall not be overly bright and shali direst, the downward placement of light.
- 5: That-the proposed development provides adoquatoly buffored loading areas and to the extent possible, are located on secondary streets;
- 6. The proposed operator of the facility shall be identified as part of the project description at the time of application.

17.102.380 - Spoclat regulations applying to truck-related activities in the West-Oakland Community Development-District.

- A. Use Pormit Required. No Truck and Truck related activity as described in Sections 17.10.470, 17.10.480, 17.10.490, and 17.10.500 shall be established or expanded in the West Oakland Community Development District except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.131
- B.---"West-Oakland-Community-Development-District"-is-defined to-include-all-areas-between Interstate 980 to the oast, 3rd-Street to the south, Interstate-880-to the west, and Interstate-680 to tho north.
- C.— The term "Truck" shall be defined as a "Commercial Vehicle" having a "Manufasturer's Gross Vehicle Weight Rating" exseeding ten thousand (10,000) pounds or a "Trailer," as those terms are defined in the California Vehicle Code.

#### 17.102.390 - Regulations Applying to Special Health Care Civic Activities.

- A. -- Additional Uso Permit Criteria.- A conditional use permit for any conditionally permitted Special Health Care Civic Activities-may only be granted upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure. In Chapter 17.134, to any and all applicable use permit criteria set forth in the partisular individual zone regulations, and to alt of the following use permit criteria:
- 1. -- That each Special-Health-Care-Civic-activity be ne-located no-less than 2,500 feet from the nearest Special-Health-Care-Civic-activity within the City of Oakland.
- 2.--- That each Special Hoalth Core Civic activity be a minimum of 500 feat from the following activities:
- -Sehaols-K--12
- Lisensed Transitional Housing
- -Licensed Service Enrished Housing
- -Licensed Emergency Shelters
- 3.— That each Special Heolth Care Civic activity meet the following Performance Standards and that these performance standards be included as standard conditions of approval. These performance standords moy be amended or expanded by the Planning Commission as they are applied to individual locations and projects:
- A-lighting-plan-sholl-be-reviewed-and-opproved by the-Zening Administrator.-Exterior-lighting-shell-be provided on olt frontages. Such lighting shall be designed to illuminate persons standing-outside such that they can be identified 50 feet away. Exterior lighting chall be designed so as not to cost glare offsite.
- -----A-plan\_for\_any\_exterior\_changes\_and\_signage\_shall-bo\_reviewed\_and\_approved\_by\_the\_Zoning Administrator.
- Storefrents-chall have glass or transparent-glazing in the windows. No more than 30-percent of any window area shall be covered by signs, banners, or opaque coverings of any kind.
- Final-floor-plans-shall be-reviewed and approved by the Zoning Administrator-prior to issuance of building permits.
- Fenced yards shall be fonced with fencing a minimum of six feet tall. Such fencing shall be of an open design.
- -One-nen-flammable-trash-container and ashtray shall be located in front of the facility for smokers.
- Days and hours of operation shall be Monday through Friday, 8:00 a.m. to 6:00 p.m. Clients shall be dissouraged from loitering prior to or aftar hours. At least one-no-loitering sign with letters at least two inchee talt shall be installed and maintained where it will be visible to pedestriane in front of the property. These performance standards and any conditions of approval, days and hours of operation, phone contact, and after hours phone contacts shall be posted where visible to the public-24 hours a day.
- -Graffiti shall be removed within 72 hours of application. No exterior pay telephones shall be installed.
- Litter-shalt-remeved-at-least-two-times-daily-or-as-needed-from-in-front-of-and-for-20-feet-beyend-the building-along-adjacent-street(s).-All-"street furniture" such-as-orates-or-mattresses-shall-bo-removed daily-or-as-needed-
- ---Prior to issuance of building-permits or commoncement of uso, applicant shall submit a needle-retrieval plan for all Special Health-Care Civic activitios that previde needle-exchange services on site.-The

plan shall, at a minimum, detail the protocol for the exchange of clean needles for dirty noodlos and for retrieving used needles within 300 feet of the site on a regular basis.

17.102.400 - Exterior security bars and related devices. Special design requirements for lets that contain Residential Facilities and no Nonrecidential Facilities

The provisions of this section apply to lots containing Residential Facilities and no Nonrosidential Facilities.

A. \_\_\_\_Limitations\_on\_Paving\_in\_Street-Fronting\_Yards.\_Pavod\_surfaces\_within\_roquired\_street-fronting\_yards, and any-unimprevod-rights-of-way of adjasent-streets, sholl be limited to the following:

1. .....All lots other than corner lots: fifty percent (50%) maximum paved surface;

2. Corner lots: thirty percent (30%) maximum paved surface.

Exceptions: Tho maximum percentages of paved surfaces specified in this subcection A may be exceeded within unimproved rights of way in the following coses upon issuance of a private construction of public improvements (P-job) pormit or if undortakon directly by the City or by a private controctor under contract to the City:

a. Roadway-construction or widening;

b. Sidewalk construction or widening; and

c. \_\_\_\_Any work-pursuont to on-approved finol-mop, parcel mop or final dovelopmont-plan-pursuant to a planned unit development pennit

For-purpocos of this subsection A, an unimproved right of way is the portion of a street or alloy right of way that is not paved.

B. Screening-of-Utility-Meters. All-utility-motors-chall-either-be-located-within-a-box-set-within-a building, located on a non-street facing elevation, or screened with vegetation.

C. Screening of Trash Containors. All trash containers shall be located in a storage oroa that is screened from the stroot and adjacent properties by a wall, fence, or dense landscaping with a minimum height of four (4) foot.

D.——Restrictions on-oxterior socurity bars and related devices. Exterior security bars and grills are not permitted on windows, doors, or perch enclosures that are located on a street-facing elevation of primary Residential Facilities unless the Director of City Planning determines that the proposed bars or grills are consistent with the architectural style of the building. Removal of such bars or grills shall be a condition of the granting of all conditional use permits, variances, design reviews and other special zoning approvals involving changes to the elevation on which the bars or grills are located unless the bars or grills have been shown to be architecturally consistent with the architectural style of the building.

17.102.420 - Speeial design-requirements for lote-located in a residential and commercial zones and the OS. S-1. S-2. S-3 and S-15 zones.

- The provisions of this section apply to any lot located in a residential or commercial zone, or are in the OS, S 1, S 2, S 3, or S-15 zone.
- A. Rostriction on Barbed Wiro and Razor Wire. In any location visible from the public right of way, no barbed wiro or razor wire may be attached to the exterior of any building or similar facility, Other restrictions on barbed wire and razor wire are cpecified in Section 17.108.140

17.102.430 - Regutations applying to sbesk cashior and/or chock cashing activity.

A. Additional Use Permit Criteria. A conditional use permit for any conditionally permitted check cashier and or chock cashing astivity as defined in Section 17.10.365 of the Oakland Planning-Code may

Oakland, California, Code of Ordinances

only bo granted-upon determination that the proposal conforms to the general use permit criterio set forth in the conditional-use permit procedure in Chapter 17.134, to any and oll opplicable-use permit criteria sot forth-in the particular individual zone regulations, and to all of the following performance standards:

- That each check cashier and or check cashing activity be located not less than one thousand (1,000) feet from the nearest check coshier and chock cashing activity within the city of Ooklond.
- That oach chock cashier and or check cashing octivity be a minimum of five-hundred (500) feet from the following activities, which en the dote of an application for check cashier and/or check coshing activity had a vosted right under California law to operate:
- a. Community-education-civic-activities,
- b.—State or federally chartered bank, savings association, credit union, or industrial loan company,
- c. Community assembly civic activities, or
- d. Recreational-assembly-civic-activities-or
- e. Alcoholic beverage sales commercial activities, excluding full service restaurants and alcoholic beverage sales commercial activities with twenty-five (25) or more full time equivalent (FTE) employees and a total fleer area of twenty thousand (20,000) square feet or more. (Note that this precludes combining check cashing with alcoholic beverage sales commercial activities because alcoholic beverage sales commercial contivities are alcoholic beverage sales commercial activities and therefore subject to this distance standard).
- 3. That each check cashier and or check coshing activity meets the following performance standards and that theso performance standards are included as standard conditions of approval. These performance standards may be amended or expanded by Staff and/or the Planning Commission as they are applied to individual locations and projects:
- a. A lighting-plan-chall-be-reviewed-and approved by the-Zoning-Administrator-prior-to-issuing-building permits-and installed prior to establishing-the activity. Exterior lighting shall be provided-on-all frontages. Such lighting shall be designed to illuminate persons standing outside such that they can be identitied fifty (50) feet away. Exterior lighting shall be designed so as not-to cast glare offsite.
- b. Storefronts-sholl-hove glass or transparent-glazing in the windows and doors. No more than ton (10) percent of ony window or door area shall be covered by signs, benners, or opaque coverings of any kind so that law enforcement personnel will have clear view of the entire public area in the promices from the public sidewalk.
- c. Days-and-hours of-operation shall-be, no earlier than 7:00 AM nor-later than 7:00 PM Monday through Saturday. Patrons shall be discouraged from loitering prior to, during and/or after hours. At least one no-loitering sign with letters at least two inches tall shall be installed and mointoined where it will be most visible to pedostrians on each side of the building in which the activity is located including, but not limited to, ctroot frontages and parking lots.
- d. Graffiti shall be removed within seventy-two (72) hours of application.
- c. No-exterior-pay-tolephenes-shall-be-permitted-
- f. Litter shall be removed at least two times daily or as needed from in front of and for twenty (20) feet beyond the building olong adjocent street(s). All "street furniture" such as crates or mattresses shall be removed daily or as needed.
- g.— The applicant shall post at least one certitied uniformed security guard on duty at all times the bucinoss is open. The security guard shall patrol the interior and all extorior portions of the property under control of the owner or lessee including, but not limited to, porking lots and any open public spaces such as lobbies.

17-102-440 - Special regulations-for-primary-colloction-contors-in-all-zonos.

- A:—Applicability:-This-Section-applies-to-Primary-Colloction-Centers, as-defined-in-Section-17:10.585 "Recycling-and-Waste-Related-Industriol-Activities," that-aro-located in-any-zone.-Where-there is any apparent-conflict-between-these-regulations-and-regulations-contained-olsewhore-in-this-Tifie-17, and/or with conditions-of-approval, the more stringent shall-govern.
- B.— Performance-Standards. In-addition-to-the-performance-standards-sot-forth-in-Chapter-17-120, the performance-standards-specified-in-Subsoction-17-73-035-B.—shall--be--uniformly-applied, as applicable, and-tho-relief-from-the-performance-standards-in-Subsoction-17-73-035-C.-shall-apply-to all-Primary-Colloction-Centers.

17.102.450 - Special-regulations-applying-to-ILaundromats.

The following regulations shall apply in all zones to the Consumer Service Commorcial Activity of laundromats:

- A. Conditional Use Permit Required. All new or expanded uses-laundromats shall be required to obtain a Conditional Use Permit as specified in Chapter 17.134
- B. Restriction on Over-Concentration of Laundromats. No new or expanded laundromat use shall be located closer than <u>five hundred (500)</u> feet from any existing laundromat as measured by <u>the</u> closest radial distance between buildings.
- C. Standards. The following standards shall apply to all new or expanded Laundromat uses:
  - 1. On-Site Attendant An employee shall be on the premises during all business hours.
  - 2. Security Cameras. Security cameras shall be operated on the premises during all business hours and recordings shall be maintained for a minimum of seven days.
  - 3. When located adjacent to or below a dwelling unit the following shall be minimized:
    - a. Noise shall not exceed the limits set forth in Chapter 17.120, Performance Standards.
    - b. Vibrations shall not exceed the limits set forth in Chapter 17.120, Performance Standards.
    - c. Venting shall be directed away from residential dwelling units.
  - 4. Transparency.
    - a. A minimum of <u>sixty</u> 60-percent (60%) of the building facade along a street or streets shall be glass (windows and/or doors).
    - b. Window Clarity. Ninety percent (90%) of area of windows shall remain clear to allow views into the commercial space.
  - Exterior Illumination. Outdoor lighting shall be attached to the exterior of the facility containing the laundromat establishment and operated after dusk so that the exterior of the premises are discernible.
  - 6. Off-Site Impacts.
    - a. Litter and debris shall be cleared from the premises and the adjacent right-of-way and sidewalks of the property at least once daily or as needed to maintain a litter free environment
    - b. Graffiti shall be removed from the exterior of the building within 72 hours of application.
    - c. At least two "No Loitering" signs shall be posted on the building facade and other visible locations around the site. Signs shall be of a permanent nature and have letters a minimum of two inches in height. The owner, manager, and employees of this

establishment shall make appropriate efforts to discourage loitering from the premises including calling the police to ask that they remove loiters who refuse to leave. Persons loitering in the vicinity of the exterior of the establishment with no apparent business for more than ten minutes shall be asked to leave. Techniques discussed in the manual entitled "Loitering: Business and Community Based Solutions" shall be used.

## Chapter 17.103 – SPECIAL REGULATIONS AND FINDINGS FOR CERTAIN USE CLASSIFICATIONS

Sections:

#### Article I – Residential Activities

<u>17.103.010</u> - Residential Care; Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities.

#### Article II - Civic Activities

17.103.020 - Special Health Care Civic Activities.

## Article III - Commercial Activities

<u>17.103.030 - Convenience Market and Fast-Food Restaurant, and Alcoholic Beverage Sales</u> <u>Commercial Activities.</u>

17.103.040 - Check Cashier and Check Cashing Commercial Activities.

17.103.050 - Transient Habitation Commercial Activities.

#### Article IV - Industrial Activities

<u>17.103.060 - Recycling and Waste-Related Industrial Activities -- Primary Recycling Collection</u> Centers.

#### Article V – Agricultural and Extractive Activities

17.103.070 - Mining and Quarrying Extractive Activities.

#### Article VI\_Residential Facilities

17.103.080 - One-Family Dwelling with Secondary Unit Residential Facilities.

#### Article VII -Nonresidential Facilities

17.103.090 - Sidewalk Café Nonresidential Facilities.

17.103.100 - Drive-Through Nonresidential Facilities.

#### Article IX – Sign Facilities

17.103.110 - Sign Facilities.

#### Article X – Telecommunications Facilities

#### 17.103.120 - Telecommunications Facilities.

## Article I - Residential Activities

# <u>17.103.010 - Residential Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities.</u>

- A. Additional Use Permit Criteria. A conditional use permit for any conditionally permitted Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity may only be granted upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, to any and all applicable use permit criteria set forth in the particular individual zone regulations, and to all of the following additional use permit criteria:
  - 1. That staffing of the facility is in compliance with any State Licensing Agency requirements;
  - 2. That if located in a residential zone, the operation of buses and vans to transport residents to and from off-site activities does not generate vehicular traffic substantially greater than that normally generated by Residential Activities in the surrounding area;
  - 3. That if located in a residential zone, the on-street parking demand generated by the facility due to visitors is not substantially greater than that normally generated by the surrounding Residential Activities;
  - 4. That if located in a residential zone, arrangements for delivery of goods are made within the hours that are compatible with and will not adversely affect the livability of the surrounding properties;
  - 5. That the facility's program does not generate noise at levels that will adversely affect the livability of the surrounding properties.
- B. Restriction on Overconcentration of Resident Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities. No Residential Care, Service-Enriched Permanent Housing, Transitional Housing, or Emergency Shelter Residential Activity shall be located closer than three hundred (300) feet from any other such Activity or Facility.

#### Article II - Civic Activities

#### 17.103.020 - Special Health Care Civic Activities.

- A. Additional Use Permit Criteria. A conditional use permit for any conditionally permitted Special Health Care Civic Activities may only be granted upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, to any and all applicable use permit criteria set forth in the particular individual zone regulations, and to all of the following use permit criteria:
  - 1. That each Special Health Care Civic activity be no located no less than two thousand five hundred (2,500) feet from the nearest Special Health Care Civic activity within the City of Oakland.

- 2. That each Special Health Care Civic activity be a minimum of five hundred (500) feet from the following activities:
  - a) Schools K-12
  - b) Licensed Transitional Housing
  - c) Licensed Service Enriched Housing
  - d) Licensed Emergency Shelters
- 3. That each Special Health Care Civic activity meets the following Performance Standards and that these performance standards be included as standard conditions of approval. These performance standards may be amended or expanded by the Planning Commission as they are applied to individual locations and projects:
  - a) A lighting plan shall be reviewed and approved by the Zoning Administrator. Exterior lighting shall be provided on all frontages. Such lighting shall be designed to illuminate persons standing outside such that they can be identified fifty (50) feet away. Exterior lighting shall be designed so as not to cast glare offsite.
  - b) A plan for any exterior changes and signage shall be reviewed and approved by the Zoning Administrator.
  - c) Storefronts shall have glass or transparent glazing in the windows. No more than thirty percent (30%) of any window area shall be covered by signs, banners, or opaque coverings of any kind.
  - d) Final floor plans shall be reviewed and approved by the Zoning Administrator prior to issuance of building permits.
  - e) Fenced vards shall be fenced with fencing a minimum of six (6) feet tall. Such fencing shall be of an open design.
  - f) One non-flammable trash container and ashtray shall be located in front of the facility for smokers.
  - g) Clients shall not be allowed to loiter outside the building on the sidewalk or street. Clients waiting to be served shall be accommodated inside the building.
  - h) Days and hours of operation shall be Monday through Friday, 8:00 a.m. to 6:00 p.m. Clients shall be discouraged from loitering prior to or after hours. At least one no loitering sign with letters at least two (2) inches tall shall be installed and maintained where it will be visible to pedestrians in front of the property. These performance standards and any conditions of approval, days and hours of operation, phone contact, and after hours phone contacts shall be posted where visible to the public 24 hours a day.
  - i) Graffiti shall be removed within 72 hours of application. No exterior pay telephones shall be installed.
  - j) Litter shall be removed at least two times daily or as needed from in front of and for twenty (20) feet beyond the building along adjacent street is). All "street furniture" such as crates or mattresses shall be removed daily or as needed.
  - k) Prior to issuance of building permits or commencement of use, applicant shall submit a needle retrieval plan for all Special Health Care Civic activities that provide needle exchange services on site. The plan shall, at a minimum, detail the protocol for the exchange of clean needles for dirty needles and for retrieving used needles within three hundred (300) feet of the site on a regular basis.

## Article III - Commercial Activities

## 17.103.030 - Fast-Food Restaurant, Convenience Market, and Alcoholic Beverage Sales Commercial Activities.

A. Use Permit Criteria for Fast-Food Restaurants, Convenience Markets, and Establishments Selling Alcoholic Beverages. A conditional use permit for any conditionally permitted Fast-Food Restaurant, Convenience Market, or Alcoholic Beverage Sales Commercial Activity may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, to any and all applicable use permit criteria set forth in the particular individual zone regulations, and to all of the following additional use permit criteria:

- 1. That the proposal will not contribute to undue proliferation of such uses in an area where additional ones would be undesirable, with consideration to be given to the area's function and character, problems of crime and loitering, and traffic problems and capacity;
- That the proposal will not adversely affect adjacent or nearby churches, temples, or synagogues; public, parochial, or private elementary, junior high, or high schools; public parks or recreation centers; or public or parochial playgrounds;
- 3. That the proposal will not interfere with the movement of people along an important pedestrian street;
- 4. That the proposed development will be of an architectural and visual quality and character which harmonizes with, or where appropriate enhances, the surrounding area;
- 5. That the design will avoid unduly large or obtrusive Signs, bleak unlandscaped parking areas, and an overall garish impression;
- 6. That adequate litter receptacles will be provided where appropriate:
- 7. That where the proposed use is in close proximity to residential uses, and especially to bedroom windows, it will be limited in hours of operation, or designed or operated, so as to avoid disruption of residents' sleep between the hours of ten (10) p.m. and seven (7) a.m. The same criteria shall apply to all conditional use permits required by subsection B of this section for sale of alcoholic beverages at full-service restaurants;
- That proposals for new Fast-Food Restaurants must substantially comply with the provisions of the Oakland City Planning Commission "Fast-Food Restaurant—Guidelines for Development and Evaluation" (OCPD 100-18).
- B. Special Restrictions on Establishments Selling Alcoholic Beverages.
  - No Alcoholic Beverage Sales Commercial Activity shall be located closer than one thousand (1,000) feet to any other Alcoholic Beverage Sales Commercial Activity measured between closest building walls, except:
    - a. On-sale retail licenses located in the Central District (defined for the purposes of this Chapter) as within the boundaries of I-980 and Brush street to the west; both sides of 27th Street to the north; Harrison Street/Lake Merritt and the Lake Merritt Channel to the east and the Estuary to the south); or
    - b. Off-sale retail licenses located in the Jack London district (defined for the purposes of this Chapter as within the boundaries of Martin Luther King Jr. Way to the west, I-880 to the north; the Lake Merritt Channel to the east and the Estuary to the south); or
    - c. If the activity is in conjunction with a Full-Service Restaurant Commercial Activity; or
    - d. Establishments with twenty-five (25) or more full time equivalent (FTE) employees or a total floor area of twelve thousand (12,000) square feet or mere.
  - 2. Sale of alcoholic beverages in conjunction with a Full Service Restaurant Commercial Activity and located within any of the following restricted street areas applied to a depth of two hundred (200) feet on each side of the identified streets and portions of streets, as measured perpendicularly from the right-of-way line thereof: International Boulevard; Foothill Boulevard; MacArthur Boulevard and West MacArthur Boulevard; that portion of San Pablo Avenue lying between Highway I-980 and I-580; that portion of Edes Avenue lying between Clara Street and Bergedo Drive, shall require a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
  - 3. In addition to the criteria prescribed elsewhere in the zoning regulations, a land use permit for an Alcoholic Beverage Sales Commercial Activity located within an Alcoholic Beverage Sales

license overconcentrated area shall only be granted, and a finding of Public Convenience or Necessity made, if the proposal conforms to all of the following three criteria:

- a. That a community need for the project is clearly demonstrated. To demonstrate community need, the applicant shall document in writing, specifically how the project would serve an unmet or underserved need or population within the overall Oakland community or the community in which the project is located, and how the proposed project would enhance physical accessibility to needed goods or services that the project would provide, including, but not limited to alcohol; and
- b. That the overall project will have a positive influence on the quality of life for the community in which it is located, providing economic benefits that outweigh anticipated negative impacts, and that will not result in a significant increase in calls for police service; and
- c. That alcohol sales are customarily associated with, and are appropriate, incidental, and subordinate to, a principal activity on the lot.
- 4. In addition to the above criteria, projects outside the Central District and Hegenberger Corridor shall meet all of the following criteria to make a finding of Public Convenience or Necessity, with the exception of those projects that will result in twenty-five (25) or more full time equivalent (FTE) employees and will result in a total floor area of twelve thousand (12,000) square feet or more:
  - a. The proposed project is not within one thousand (1,000) feet of another alcohol outiet (not including Full Service Restaurant Commercial Activities), school, licensed day care center, public park or playground, churches, senior citizen facilifies, and licensed alcohol or drug treatment facilities; and
  - b. Police department calls for service within the "beat" where the project is located do not exceed by twenty percent (20%), the average of calls for police service in police beats Citywide during the preceding one calendar year.
- 5. See Chapter 17.156 for Deemed Approved Alcoholic Beverage Sale regulations.
- C. Special Restrictions Applying to Fast-Food Restaurants.
  - 1. No Fast-Food Restaurant Commercial Activity shall be located within a one thousand (1,000) foot radius of an existing or approved Fast-Food Restaurant as measured from the center of the front property tine of the proposed site, except in the Central District (defined for the purposes of this Chapter as within the boundaries of I-980 and Brush Street to the west both sides of 27th Street to the North; Harrison Street/Lake Merritt and the Lake Merritt Channel to the east and the Estuary to the south), within the main building of Shopping Center Facilities, and in the CR-1 Regional Commercial zone.
  - 2. Fast-Food Restaurants' with Drive-Through Facilities shall not be located within five-hundred (500) feet of a public or private elementary school, park, or playground.
  - 3. Access. Ingress and egress to Fast-Food Facilifies shall be limited to commercial arterial streets rather than residenfial streets. No direct access shall be provided to adjacent residential streets which are less than thirty-two (32) feet in pavement width. Exceptions to either of the requirements may be obtained where the City Traffic Engineer determines that compliance would deteriorate local circulation or jeopardize the public safety. Any such determination shall be stated in writing and shall be supported with findings. Driveway locations and widths and entrances and exits to Fast-Food Facilities shall be subject to the approval of the City Traffic Engineer.
  - 4. Trash and Litter. Disposable containers, wrappers and napkins utilized by Fast-Food Restaurants shall be imprinted with the restaurant name or logo.
  - 5. Vacated/Abandoned Fast-Food Facilities. The project sponsor of a proposed Fast-Food Facility shall be required to obtain a performance bond, or other security acceptable to the City Attorney, to cover the cost of securing and maintaining the facility and site if it is abandoned or

vacated within a prescribed high-risk period. As used in this code, the words "abandoned" or "vacated" shall mean a facility that has not been operational for a period of thirty (30) consecutive days, except where nonoperation is the result of maintenance or renovation activity pursuant to valid City permits. The defined period of coverage is four (4) years following the obtaining of an occupancy permit. The bond may be renewed annually, and proof of renewal shall be forwarded to the Director of City Planning. The bond amount shall be determined by the City's Risk Manager and shall be adequate to defray expenses associated with the requirements outlined below. Monitoring and enforcement of the requirements set forth in this section shall be the responsibility of the Building Official, pursuant to Chapter 8.24 of the Oakland Municipal Code (O.M.C.) and those sections of the Oakland Building Code which are applicable.

If a Fast-Food Facility has been vacated or abandoned for more than thirty (30) consecutive days, the project sponsor shall be required to comply with the following requirements, pursuant to the relevant cited City, county and state codes:

a. Enclose the property with a security fence and secure the facility;

- b. Post signs indicating that vehicular parking and storage are prohibited on the site (10.16.070 O.T.C. and 22658 C.V.C), and that violators will be cited, and vehicles towed at the owner's expense, and that it is unlawful to litter or dump waste on the site (Sections 374b.5 C.P.C. and 374b C.P.C.). All signs shall conform to the limitations on signs for the specific zone and shall be weatherproof and of appropriate size and standard design for the particular function;
- c. Install and maintain security lighting as appropriate and required by the Oakland Police Department
- d. Keep the site free of handbills, posters and graffiti and clear of litter and debris pursuant to Section 8.38,160 of the O.M.C.:
- e. Maintain existing landscaping and keep the site free of overgrown vegetation.

17.103.040 – Check Cashier and Check Cashing Commercial Activities.

- A. Additional Use Permit Criteria. A conditional use permit for any conditionally permitted Check Cashier and Check Cashing Commercial activity may only be granted upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134, to any and all applicable use permit criteria set forth in the particular individual zone regulations, and to all of the following performance standards:
  - That each Check Cashier and Check Cashing Commercial activity be located not less than one thousand (1,000) feet from the nearest Check Cashier and Check Cashing Commercial activity within the city of Oakland.
  - 2. That each Check Cashier and Check Cashing Commercial activity be a minimum of five hundred (500) feet from the following activities, which on the date of an application for Check Cashier and Check Cashing Commercial activity had a vested right under California law to operate:
    - a. Community education civic activities,
    - b. State or federally chartered bank, savings association, credit union, or industrial loan company,
    - c. Community assembly civic activities, or
    - d. Recreational assembly civic activities or
    - e. Alcoholic beverage sales commercial activities, excluding full service restaurants and alcoholic beverage sales commercial activities with twenty-five (25) or more full time

J

equivalent (FTE) employees and a total floor area of twenty thousand (20,000) square feet or more. (Note that this precludes combining Check Cashier and Check Cashing Commercial activity with alcoholic beverage sales commercial activities because alcoholic beverage sales commercial activities are always considered a primary activity and therefore subject to this distance standard).

- 3. That each Check Cashier and Check Cashing Commercial activity meets the following performance standards and that these performance standards are included as standard conditions of approval. These performance standards may be amended or expanded by Staff and/or the Planning Commission as they are applied to individual locations and projects:
  - a. A lighting plan shall be reviewed and approved by the Zoning Administrator prior to issuing building permits and installed prior to establishing the activity. Exterior lighting shall be provided on all frontages. Such lighting shall be designed to illuminate persons standing outside such that they can be identified fifty (50) feet away. Exterior lighting shall be designed so as not to cast glare offsite.
  - b. Storefronts shall have glass or transparent glazing in the windows and doors. No more than ten percent (10%) of any window or door area shall be covered by signs, banners, or opaque coverings of any kind so that law enforcement personnel will have clear view of the entire public area in the premises from the public sidewalk.
  - c. Days and hours of operation shall be, no earlier than 7:00 AM nor later than 7:00 PM Monday through Saturday. Patrons shall be discouraged from loitering prior to, during and/or after hours. At least one no loitering sign with letters at least two (2) inches tall shall be installed and maintained where it will be most visible to pedestrians on each side of the building in which the activity is located including, but not limited to, street frontages and parking lots.
  - d. Graffiti shall be removed within seventy-two (72) hours of application.
  - e. No exterior pay telephones shall be permitted.
  - t. Litter shall be removed at least two times daily or as needed from in front of and for twenty. (20) feet beyond the building along adjacent street(s). All "street furniture" such as crates or mattresses shall be removed daily or as needed.
  - g. The applicant shall post at least one certified uniformed security guard on duty at all fimes the business is open. The security guard shall patrol the interior and all exterior portions of the property under control of the owner or lessee including, but not limited to, parking lots and any open public spaces such as lobbies.

## 17.103.050 Transient Habitation Commercial Activities.

- A. A Conditional Use Permit for any Transient Habitation Commercial Activity may only be granted upon determination that the proposal conforms to the general use permit criteria (see Section 17.134.050) and to all of the following additional use permit criteria:
  - That the proposal is located in downtown, along the waterfront, near the airport, or along the I-880 freeway, and/or in an area with a concentration of amenities for hotel patrons, including restaurant, retail, recreation, open space and exercise facilities, and is well-served by public transit.
  - 2. That the proposal considers the impact of the employees of the hotel or motel on the demand in the city for housing, public transit, and social services;
  - 3. That the proposal is consistent with the goal of attracting first-class, luxury hotels in downtown, along the waterfront, near the airport, or along the I-880 freeway, which provide:
    - a. A minimum of one hundred (100) sleeping rooms;

- b. A full service restaurant providing three meals per day; and
- c. On-site recreational amenities, which may include an exercise room, swimming pool, and/or tennis courts;
- 4. That the proposed development will be of an architectural and visual quality and character which harmonizes and enhances the surrounding area, and that such design includes:
  - a. Site planning that insures appropriate access and circulation, locates building entries which face the primary street, provides a consistent development pattern along the primary street, and insures a design that promotes safety for its users;
  - b. Landscaping that creates a pleasant visual corridor along the primary streets with a variety of local species and high quality landscape materials;
  - c. Signage that is integrated and consistent with the building design and promotes the building entry, is consistent with the desired character of the area, and does not detract from the overall streetscape;
  - d. The majority of the parking located to the rear of the site and where appropriate is provided within a structured parking facility that is consistent, compatible and integrated into the overall development
  - e. Appropriate design treatment for ventilation of room units as well as structured parking areas; and prominent entry features that may include attractive porte-cocheres;
  - f. Building design that enhances the building's quality with strong architectural statements, high quality materials particularly at the pedestrian level and appropriate attention to detail;
  - g. Lighting standards for hotel buildings, grounds and parking lots that are not be-overly bright and direct the downward placement of light.
- 5. That the proposed development provides adequately buffered loading areas and to the extent possible, are located on secondary streets;
- 6. The proposed operator of the facility shall be identified as part of the project description at the time of application.
- B. See Chapter 17.157 for Deemed Approved Hotel regulations.

## Article IV -Industrial Activities

## <u>17.103.060 - Recycling and Waste-Related Industrial Activities -- Primary Recycling Collection</u> Centers.

- A. Applicability. This Section applies to Recycling and Waste-Related Industrial Activities -- Primary Collection Centers that are located in any zone. Where there is any apparent conflict between these regulations and regulations contained elsewhere in this Title 17, and/or with conditions of approval, the more stringent shall govern.
- B. Performance Standards. In addition to the performance standards set forth in Chapter 17.120, the performance standards specified in Subsection 17.73.035.B. shall be uniformly applied, as applicable, and the relief from the performance standards in Subsection 17.73.035.C. shall apply to all Primary Collection Centers.

## Article V – Agricuttural and Extractive Activities

## 17.103.070 - Mining and Quarrying Extractive Activities.

See Chapter 17,155 for special regulations regarding Mining and Quarrying Extractive Activities.

## Article VI – Residential Facilities

#### 17.103.080 - One-Famify Dwelling with Secondary Unit Residential Facilifies,

- A. Development Standards. The following regulations shall apply to the construction, establishment or alteration of Secondary Units wherever permitted or conditionally permitted, as specified in each individual zone:
  - 1. Other Uses on Property. A Secondary Unit shall only be permitted on a lot that contains only one other primary dwelling unit. A Secondary Unit may be approved and constructed at the same time or after the approval and construction of the primary dwelling unit.
  - 2. Sale of Unit A Secondary Unit shall not be sold separately from the primary dwelling on the same lot
  - 3. Owner Occupancy. The legal owner shall occupy either the primary dwelling or the Secondary Unit. Prior to issuance of a building permit for a Secondary Unit, the applicant shall record as a deed restriction in the Alameda County Recorder's Office, notice of this requirement, in a form prescribed by the Director of City Planning.
  - 4. Maximum Permitted Floor Area. The floor area of a Secondary Unit shall not exceed nine hundred (900) square feet or fifty percent (50%) of the floor area of the primary dwelling, whichever is less, except that Secondary Units of up to five hundred (500) square feet in floor area are permitted regardless of the size of the primary dwelling.
  - 5. Fire Flow and Water Pressure. A Secondary Unit may be permitted only if the fire flow and water pressure in the adjoining street meets the minimum requirements as determined by the Fire Marshal.
  - 6. Emergency Access—Multiple Vehicular Outiets. A Secondary Unit may be permitted only on a lot which has frontage on a through street or a dead-end street that has a total length of less than three hundred (300) feet. For the purposes of this subsection, the total length of a dead-end street shall be the distance from the intersection with the nearest through street to the farthest opposite end of the street right-of-way, or private access easement (as defined by Section 16.32.010 of the Oakland Municipal Code) if the private access easement is connected to said dead-end street.
  - 7. Emergency Access—Minimum Pavement Width. A Secondary Unit may be permitted only if all streets connecting the lot to the nearest arterial street (as designated by the City of Oakland General Plan Land Use and Transportation Element) have a minimum pavement width ef at least twenty-four (24) feet. The minimum pavement width limitation may be reduced to a minimum of twenty (20) feet, upon the granting of a conditional use permit pursuant to the criteria in subsection B of this section, and the conditional use permit procedure in Chapter 17.134
  - 8. Public Sanitary Sewer. A Secondary Unit may be permitted only if it is served by a public sanitary sewer.
  - 9. Architectural Compatibility. The Secondary Unit shall be clearly subordinate to the primary dwelling unit in size and location. Also, the architectural design and materials of a Secondary Unit shall match or be visually compatible with that of the primary dwelling, including the

architectural style, siding material, roof shape, roofing material, trim material and design, window types, window trim, and window sill detail.

- 10. Compliance with Building and Fire Codes. All Secondary Units shall comply with all other code and permit requirements imposed by all other affected departments, including but not limited to fire separation, sound separation, egress, utility access, and the requirement for a building permit.
- 11. Review Procedure. An application for a Secondary Unit of up to five hundred (500) square feet shall be granted ministerial approval as specified in Section 17.136.025 upon confirmation of compliance with all applicable zoning regulations, including but not limited to, all provisions in this Section. The five hundred (500) square-foot floor area threshold for a Secondary Unit may only be exceeded, up to a maximum of nine hundred (900) square feet or fifty percent (50%) of the fioor area of the primary dwelling, whichever is less, upon the granting of small project design review, pursuant to the small project design review procedure in Section 17.136.030.
- B. Use permit criteria for Secondary Units accessed via narrow streets. A conditional use permit for a Secondary Unit accessed from the nearest arterial street via a street with a minimum pavement width of between twenty (20) and twenty-four (24) feet may only be granted upon determination that the proposal conforms to the general use permit criteria set forth in the general use permit procedure in Chapter 17.134 and to all of the following additional use permit criteria:
  - 1. That there is adequate emergency access to the lot as determined by the Fire Marshall.
  - 2. That the portions of the street that have a pavement width of less than twenty-four (24) feet are net located on a dead-end street.
  - 3. That if on-street parking is permitted on portions of the street that have a pavement width of less than twenty-four (24) feet that there exist a level and hard surface shoulders with a combined additional width of at least eight (8) feet.
  - 4. That if on-street parking is prohibited on portions of the street that have a pavement width of less than twenty-four (24) feet that the restricted parking areas are clearly marked with official City installed no-parking signs and/or red curbs, pursuant to the provisions of the Oakland Traffic Code (Title 10 of the Oakland Municipal Code).

Article VIII - Nonresidential Facilities

#### 17.103.090 - Sidewalk Café Nonresidential Facilities.

- A. Procedures for Construction of Sidewalk Cafe Facilities.
  - Not withstanding any design review requirement of the particular zone. Sidewalk Cafes that have a maximum of five (5) tables and no more than fifteen (15) chairs and/or will not have any permanent structures in the public right of way, are allowed by right subject to the standards required in subsection B of this section.
  - Sidewalk Cafes that have more than five (5) tables/fifteen (15) chairs and/or have a permanent structure in the public right of way are subject to small project design review in Section 17.136.030.
- B. Standards for Sidewalk Cafes.
  - Sidewalk Cafes shall not encroach upon any public right-of-way unless a minimum of five and one-half (5½) feet of unobstructed improved sidewalk remains available for pedestrian purposes. The minimum distance shall be measured from the portion of the Sidewalk Cafe encroachment which is nearest to any obstruction within the sidewalk area. For purposes of the minimum clear path, parking meters, traffic signs, trees and all similar obstacles shall constitute obstruction.

- 2. Operators/owners of Sidewalk Cafes shall obtain an encroachment permit from the city's Building Services Division, and shall comply with all requirements imposed by other affected departments. The encroachment permit shall include language that a waste receptacle be placed outside, all garbage/litter associated with Sidewalk Cafes must be removed within twenty-four (24) hours, and a requirement to obtain liability insurance. The city shall be named as an additional insured and the amount of the insurance shall be determined by the city's Risk Manager.
- 3. The operators/owners of Sidewalk Cafes shall defend, indemnify, and hold harmless the City of Oakland its agents, officers, and employees from any claim, action, or proceeding (including legal costs and attorney's fees) against the City of Oakland, its agents, officers or employees to attack, set aside, void or annul, an approval by the City of Oakland, the City Planning Department, Planning Commission, or City Council. The city shall promptly notify the applicant of any claim, action or proceeding and the city shall cooperate fully in such defense. The city may elect, in its sole discretion, to participate in the defense of said claim, action, or proceeding.
- 4. The operator/owners of Sidewalk Cafes shall continually bus tables and provide a final cleanup at the end of the business day that will include litter pickup one hundred (100) feet in each direction from the site.

## 17.103.100 - Drive-Through Nonresidential Facilities.

- A. Conditional Use Permit for any Drive-Through Nonresidential Facility may only be granted upon determination that the proposal conforms to the general use permit criteria (see Section 17.134.050) and to all of the following additional use permit criteria:
  - 1. That the proposed facility will not impair a generally continuous wall of building facades;
  - 2. That the proposed facility will not result in weakening the concentration and continuity of retail facilities at ground level, and will not impair the retention or creation of a shopping frontage;
  - 3. That the proposed facility will not directly result in a significant reduction in the circulation level of service of adjacent streets.
- B. A driveway serving as a vehicle stacking or queuing lane for a drive-through window in a Drive-Through Nonresidential Facility shall be separated from parking areas and shall not be the only entry or exit lane on the premises. Such facility shall be so situated that any vehicle overflow from it shall not spill onto public streets or the major circulation aisles of any parking lot. Such facility shall have durable, all-weather surface; shall have reasonable disposal of surface waters by grading and drainage; and shall be permanently maintained in good condition.
- C. Each vehicle space comprising a stacking or queuing lane for a drive-through window drive-through window in a Drive-Through Nonresidential Facility shall be a minimum of ten (10) feet in width by twenty (20) feet in length. Such a stacking or queuing lane shall have a minimum capacity of eight (8) vehicles.

Article IX -Sign Facilities

## 17.103.110 - Sign Facilities.

See Chapter 17.104 for special regulations regarding Sign Facilities.

# Article X – Telecommunications Facilities

## 17.103.120 - Telecommunications Facilities.

See Chapter 17.128 for special regulations regarding Telecommunications Facilities.

Oakland, California, Code of Ordinances

17.104.020 - General limitations on signs—RU-4 and RU-5 zones, and all Commercial and Industrial zones and the RU-4 and RU-5 zones.

The following limitations shall apply to the specified signs in the RU-4 and RU-5 zones and all Commercial and Industrial zones, except as otherwise provided herein, and are in addition to the limitations, if any, prescribed for signs in the applicable individual zone regulations and development control maps:

- A. Design Review. No business, civic, or residential sign shall be constructed or established, or altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136.
- B. Permitted Aggregate Sign Area.
  - 1. In the RU-4 and RU-5 zones and all Commercial zones, the maximum aggregate area of display surface of all business, civic, and residential signs on any one lot shall be one square foot for each one foot of lot frontage in the case of an interior lot, or 0.5 square feet for each one foot of lot frontage in the case of a corner lot. The aggregate shall include only one face of a double-faced sign. The total amount of aggregate sign area shall not exceed two hundred (200) square feet on any one property. Exceptions to the total amount of aggregate sign area normally allowed on any one property may be approved pursuant to the regulations in Subsection B(3) below and to the small project design review procedure in Chapter 17.136.
  - 2. In all industrial zones, the maximum aggregate area of display surface of all business, civic and residential signs on any one lot shall be one square foot for each one foot of lot frontage in the case of an interior lot, or 0.5 square feet for each one foot of lot frontage in the case of a corner lot. The aggregate shall include only one face of a double-faced sign. The total amount of aggregate sign area shall not exceed three\_hundred (300) square feet on any one property. Exceptions to the total amount of aggregate sign area normally allowed on any one property may be approved pursuant to the regulations in Subsection B(3) below.
  - Exception to Aggregate Sign Area Limits. The following exceptions to the aggregate sign area limits may be approved:
    - a. In cases in which the maximum aggregate sign area for a property is already being utilized by a portion of the existing tenant spaces in a multi-tenant building or complex, <u>twenty (20)</u> square feet of sign area for each tenant space in the multi-tenant building or complex without existing signage on site is allowed if approved pursuant to the small project design review procedure in Chapter 17.136.
    - b. Signs conforming to a Master Sign Program approved pursuant to Section 17.104.070.
- F. Development Signs. In RU-4 and RU-5 zones and all Commercial and Industrial zones, the maximum aggregate area of display surface of all development signs on any one lot shall be either seventy-five (75) square feet or one square foot for each two (2) feet of street line abutting the lot, whichever is greater. However, a greater area of display surface may be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
- H. Signs Within One thousand (1,000) Feet of Rapid Transit Routes. Signs within one thousand (1,000) feet of the centerline of rapid transit routes shall be subject to the applicable limitations set forth in Sections 17.104.040 and 17.114.150.
- J. Temporary Business Signs.
  - 1. Size Allowed. Temporary signs are allowed in addition to permanent signs. The size of the temporary signs may not exceed the allowed square footage for permanent signs:

- 2. Allowed Time Limits.
  - a. Grand Opening Signs. Temporally signs for the purpose of grand openings of a new business can be in place for a maximum of thirty (30) days. The installation date of the sign shall be placed on the sign to verify compliance with this regulation.
  - b. Special Event Signs. Temporary signs for the purpose of special events may be placed on site a maximum of four (4) times per calendar year and a maximum of five (5) consecutive days per event.
- 3. Placement of Signs.
  - a. Signs are allowed on private property only. Signs shall not he placed in public rightsof-way or at off-site locations.
  - b. Signs must be affixed to a permanent structure.
- 4. Temporary signs shall not be illuminated.
- 5. Durable Materials Required. Signs shall be constructed of durable, rigid material suitable to the location and purpose. Only interior window signs may he made of nonrigid (e.g. paper) material.
- 6. Removal of Signs. Temporary signs and their components shall be promptiv removed at the expiration of the time limits set forth above.

17.104.030 - General limitations on signs—S-1, S-2, S-3 and S-15 zones.

The following limitations shall apply to the specified signs in the S-1, S-2, S-3 and S-15 zones, and are in addition to the limitations, if any, prescribed for signs in the applicable individual zone regulations or development control maps:

- B. Permitted Aggregate Sign Area. S-1, S-2, S-3 and S-15 Zones. The maximum aggregate area of display surface of all business, civic, and residential signs on any one lot shall be one square foot for each one foot of lot frontage in the case of an interior lot, or 0.5 square feet for each one foot of lot frontage in the case of a corner lot. The aggregate shall include only one face of a double-faced sign. The total amount of aggregate sign area shall not exceed two hundred (200) square feet on any one property. Exceptions to the total amount of aggregate sign area normally allowed on any one property may be approved pursuant to the regulations in Subsection B(1) below.
  - 1. Exception to Aggregate Sign Area Limits. The following exceptions to the aggregate sign area limits may be approved:
    - a. In cases in which the maximum aggregate sign area for a property is already being utilized by a portion of the existing tenant spaces in a multi-tenant building or complex, <u>twenty (20)</u> square feet of sign area for each tenant space in the multi-tenant building or complex without existing signage on site is allowed if approved pursuant to the small project design review procedure in Chapter 17.136.
    - b. Signs conforming to a Master Sign Program approved pursuant to Section 17.104.070.
- F. Signs within One Thousand (1,000) Feet of Rapid Transit Routes. Signs within one thousand (1,000) feet of the centeriine of rapid transit routes shall be subject to the applicable limitations set forth in Sections 17.104.040 and 17.114.150.

17.104.040 - Limitations on Signs within one thousand (1,000) feet of rapid transit routes.

The following limitations shall apply in all zones, within one thousand (1,000) feet of the centeriine of every rapid transit route, after the date of official determination thereof and except where the route is

underground. The distance shall be measured perpendicularly from said centerline, i.e., at right angles to said centerline. These provisions shall not prohibit a sign identifying an on-premises business or naming the product manufactured thereon, except to the extent of requiring design review approval.

- A. Design Review for Certain New or Altered Signs the Advertising Material of Which Is Primarily Viewable from the Transit Route.
  - 1. No sign the advertising material of which is or has become primarily viewable by the passengers on the transit route shall be constructed, established, reoriented, changed as to illumination, or otherwise altered or painted a new color unless plans for such Sign have been approved pursuant to the regular design review procedure in Chapter 17.136.
  - 2. The Director of City Planning shall determine which signs are or have become primarily viewable by the passengers on the transit route, subject to appeal pursuant to the administrative appeal procedure in Chapter 17.132.

#### 17.104.070 - Master Sign Programs.

- C. Review of individual signs upon approval of a Master Sign Program. Once a Master Sign Program is approved for any multi-tenant building or complex, the following provisions shall apply:
  - 1. Sign applications determined to conform to the provisions of an approved Master Sign Program shall be exempt from design review as is otherwise specified in Chapter 17.136.
  - 2. Sign applications determined to not conform to an approved Master Sign Program may only be granted upon approval of a revision to the original Master Sign Program conditional use permit

17.106.030 - Maximum density and floor-area ratio on lots containing both Residential and Nonresidential Facilities.

- A. Portion of Lot Area Used in Computing Density in the Central Business District and Jack London District For mixed use projects in the Central Business District CBD-zones-and Jack London district, the allowable intensity of development shall be measured according to both the maximum nonresidential Floor Area Ratio (FAR) allowed by the zone and the maximum residential density allowed by the zone. The total lot area shall be used as a basis for computing both the maximum nonresidential FAR and the maximum residential density. may-be-measured-according-to-the maximum-FAR-allewed-by-the-zone-witheut-a-separate-residential density-calculation, provided-the maximum-number-ef-units-pursuant-to-the-residential-density-allowed-by-the-General-Plan-and Estuary-Pollsy-Plan-is-not-exceeded. (The Central Business District is that area identitied as part of the Land Use and Transportation Element Land Use Diagram of the General Plan. The Jack London district is that area identified as part of the Estuary Policy Plan and adopted as part of the General Pian-).
- B. Portion of Lot Area Used in Computing Density in Areas other than the Central Business District and Jack London District For mixed use projects located in areas other than the Central Business District CBD-zones and Jack London district in which a maximum Floor Area Ratio (FAR) is generally prescribed for Nonresidential Facilities, no portion of lot area used to meet the density requirements for a Residential Facility shall be used as a basis for computing, through such Floor Area Ratio, the maximum amount of floor area for any Nonresidential Facility on the same lot unless the total nonresidential floor area on the lot is (ess than three thousand (3,000) square feet.
- C. Different Floor-Area Ratios. In all zones in which the maximum Floor-Area Ratio (FAR) generally prescribed for Residential Facilities is different from that for Nonresidential Facilities, the overall maximum Floor-Area Ratio of any lot containing both Residential and Nonresidential Facilities shall be the greater of the two prescribed Floor-Area Ratios. However, the total floor area actually devoted to each class of facility shall not exceed the maximum ratio prescribed for that class.

**17.106.050** - Use permit criteria for increased density or floor-area ratio with acquisition of abutting development rights.

A conditional use permit for an increase in the number of living units or Floor-Area Ratio (FAR) upon acquisition of nearby development rights, wherever such increase is provided for in the applicable individual zone regulations, may be granted only upon determination that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134 and to all of the following additional use permit criteria:

- A. That the applicant has acquired development rights from the owners of abutting lots, restricting the number of living units or the amount of tioor area which may be developed thereon so long as the facilities proposed by the applicant are in existence;
- B. That the owners of all such abutting lots shall prepare and execute an agreement, approved as to form and legality by the City Attorney and tiled with the Alameda County Recorder, incorporating such restriction;
- C. That the resultant reduction in potential number of living units or amount of floor area on the abutting lots is sufficient in amount and is so located as to cause the net effect upon the surrounding neighborhood to be substantially equivalent to that of the development which would be allowable otherwise.
## 17.107.020 - Definitions.

- A. Affordable Housing. "Affordable housing" shall mean that the relevant housing is available on terms such that the housing costs are less than a specified percentage of the gross income of households within a particular income category (adjusted for household size, depending on the number of bedrooms in the living unit) as determined for the Oakland Primary Metropolitan Statistical Area (PMSA). For a rental unit housing costs include rent and a reasonable allowance for utilities. For a for-sale unit, housing costs include loan principal, loan interest, property and mortgage insurance, property taxes, home owners' association dues and a reasonable allowance for utilities.
  - Where units are targeted as being affordable to low income households, housing costs for rental units must be equal to or less than thirty percent (30%) of the gross monthly income, adjusted for household size, of sixty percent (60%) of the median income for the Oakland PMSA. Housing costs for for-sale units must be equal to or less than thirty (30)-percent (30%) of the gross monthly income, adjusted for household size, of seventy percent (70%) of the median income.
  - 2. Where units are targeted as being affordable to very low income households, housing costs for rental units and for for-sale units must be equal to or less than thirty percent (30%) of the gross monthly income, adjusted for household size, of fifty percent (50%) of the median income for the Oakland PMSA.
  - 3. Where units are targeted as being affordable to moderate income households, housing costs for rental units must be equal to or less than thirty percent (30%) of the gross monthly income, adjusted for household size, of one hundred twenty percent (120%) of the median income for the Oakland PMSA. Housing costs for for-sale units must be equal to or less than thirty-five percent (35%) of the gross monthly income, adjusted for household size, of one hundred twenty percent (120%) of the median thirty-five percent (120%) of the gross monthly income, adjusted for household size, of one hundred twenty percent (120%) of the median income.
- E. Moderate, Low and Very Low Income Households. "Moderate, low and very low income households" means those households whose income matches levels determined periodically by the U.S. Department of Housing and Urban Development based on the Oakland Primary Metropolitan Statistical Area (PMSA) median income levels by family size, under which:
  - 1. "Moderate income" is defined as greater than eighty percent (80%) to one\_-hundred\_-twenty (120) percent of median income.
  - 2. "Low income" is defined as greater than fifty percent (50%) to eighty percent (80%) of median income.
  - 3. "Very low income" is defined as less than fifty percent (50%) of median income.

### 17.107.040 - Findings required.

## A. Density Bonús.

Whenever action is taken on an application for design review of a housing development of at least five units which also seeks approval of a density bonus, the city shall grant the applicant a density bonus and, unless findings justifying the denial of an incentive are adopted, one density incentive, as set forth in Section 17.107.020(C), if the applicant proposes to build one of the following. Nothing in this section shall preclude the requirement for design review as provided for in the individual zone regulations:

- 1. Where the request is for a density bonus of twenty-five (25)-percent (25%), or less if requested by the applicant and the applicant proposes that
  - a. Twenty (20)-percent (20%) of the total housing units shall be affordable to low income households; or

- b. Ten (10) percent (10%) of the total housing units shall be affordable to very low income households; or
- c. Fifty (50)-percent (50%) of the total housing units shall be affordable to qualifying residents as defined in Section 51.3 of the Civil Code (senior citizens); or
- d: Fifty (50)-percent (50%) of the total housing units are affordable to moderate income households and an additional ten percent of the total housing units are affordable to low income households and the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134 except that the density bonus cannot exceed the maximum density in the General Plan.
- 2. Where the request is for a density bonus of ten (10)-percent (10%), or less if requested by the applicant, and the project is a residential condominium development, and twenty (20)-percent (20%) of the total housing units are and will continue to be affordable to moderate income households.
- 3. Where the request is for a density bonus of greater than twenty-five (25)-percent (25%), but not more than one hundred (100)-percent (100%), the reviewing body shall find that the proposal conforms to the general use permit criteria set forth in the conditional use permit procedure in Chapter 17.134 and that the proposal provides additional housing units that are affordable to very low income, low income or moderate income households, beyond the minimum requirements described above, proportional to the additional density bonus. Proposals for senior citizen housing projects that conform to the requirements of Section 17.106.060 may request a total density bonus, over the allowable base density, of up to one hundred (100) percent

# Chapter 17.108 - GENERAL HEIGHT, YARD, AND COURT REGULATIONS

Sections:

17.108.010 - Height restrictions on lots abutting property in an RH, RD, or RM zone.

17.108.020 - Different maximum height in certain situations.

17.108.030 - Allowed projections above height limits.

17.108.080 - Minimum side yard opposite living room windows.

17.108.120 - Minimum court between opposite walls on same lot.

17.108.130 - Exceptions to required openness of minimum yards and courts.

17.108.140 - Fences, dense hedges, barrier, and similar freestanding walls.

17.108.150 - Retaining walls

17.108.030 - Allowed projections above height limits.

The height restrictions prescribed for facilities in the applicable individual zone regulations and development control maps and in Sections 17 108.010 and 17 108.020 may be exceeded in accordance with the following table. However, facilities within required minimum yards and courts shall also be subject to the applicable provisions of Section 17 108.130.

Facilities Allowed Above the Prescribed	Restrictions on Facility, or Portion Thereof, Above the Prescribed Height			
Height .	Maximum Aggregate Coverage of the Building's Horizontal Area (If on a Building)	MaximUm Vertical Projection Above the Prescribed Height	MinjmUm Horizontal Distance from any Abutting Residentially Zoned Lot	
A. Chimneys, ventilators, plumbing vent stacks, water tanks, cooling towers, machinery rooms, and other equipment and appurtenances which are not provided for elsewhere in this section. (For screening around these, see below.)	Ten <u>percent</u> (10%) percent, minus any percentage covered pursuant to subsection B of this section.	Ten (10) feet, except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134	Fifteen (15) feet, except upon the granting of a conditional use permit; but no restriction if the vertical projection above the prescribed height does not exceed four (4) feet.	
B. Elevator or stair towers; penthouses, excluding those containing any living unit; stage or scenery lofts in theatres or performance venues; skylights and dormer windows located on principal and accessory Nonresidential Facilities; and rooftop fenced or walled spaces which do not qualify elsewhere in this section.	Ten percent (10%) persent, minus any percentage covered pursuant to subsection A of this section.	Twelve (12) feet, except upon the granting of a conditional use permit.	Ten (10) feet, except upon the granfing of a conditional use pennit; but no restriction if the vertical projection above the prescribed height does not exceed four (4) feet.	
C. Skylights, donners and gable ends up to fifteen (15) feef in width located on principal and accessory Residential Facilities, except accessory facilities permitted in minimum yards or courts pursuant to Section 17.108.130K.	Ten <u>percent</u> (10%) percent, minus any percentage covered pursuant to subsection A of this section.	Ten (10) feet for donners and gable ends and one foot for skylights, but in all cases, no higher than the maximum height of the roof section on which they are located, except that skylights on a flat roof (slope 1:12 or less) may extend one (1) foot above the roof	Ten (10) feet, except upon the granting of a conditional use pennit; but no restriction if the vertical projection above the prescribed height does not exceed four (4) feet.	
D. Decorative features such as spires, bell towers, domes, cupolas, obelisks, and monuments.	Ten <u>percent (10%)</u> percent, minus any percentage covered pursuant to	Fifteen (15) feet, except upon the granting of a conditional use permit.	Fifteen (15) feet, except upon the granting of a conditional use permit; but no restriction if the vertical	

	subsection A or B of this section.		projection above the prescribed height does not exceed four (4) feet.
E. Fire escapes, catwalks, and open railings required by law.	No restriction.	No restriction.	No restriction.
F. Rooftop recreational, observation, seating, outdoor dining, clothesline, and parking facilities, unroofed themselves except for incidental sunshades, wind- screens, and similar devices; rooftop landscaping, other than trees; and unroofed open stairs and rooftop open fencing which do not qualify elsewhere in this section.	No restriction.	Ten (10) feet, except upon the granting of a conditional use permit.	Fifteen (15) feet, except upon the granting of a conditional use permit; but no restriction if the vertical projection above the prescribed height does not exceed four (4) feet.
G. Eaves, awnings, balconies, open stairs, and similar lateral extensions of a building, where the prescribed height is expressed as a ratio to some horizontal setback.	No restriction.	Four (4) feef in the case of Section 17.108.010 and eight (8) feet othenwise	No restriction.
H. Microwave and satellite dishes which are over one-(1)-meter <u>three (3) feet</u> in diameter and located in any residential zone or within one hundred fifty (150) feet from the nearest boundary of any residential zone, subject where applicable to the provisions of Section 17.102.240	Ten <u>percent (10%)</u> percent-minus any percentage covered pursuant to subsection A or B of this section.	Seven (7) feet, except upon the granting of a conditional use permit.	Ten (10) feet, except upon the granting of a conditional use pennit.
1. Radio and television masts antennas, other than micro-wave and satellite dishes.	No restriction.	Fifteen (15) feet, except upon the granting of a conditional use permit.	Five (5) feet, except upon the granting of condifional use pennit.
J. Trees; flagpoles; weather vanes; microwave and satellite dishes which are one (1) meter or less in diameter; and utility poles and lines.	No restriction.	No restriction.	No restriction.
K. Special Signs; and other Signs if flat against the surface of a facility authorized above.	No special restriction, but subject to the regular height and other limitations applicable to Signs.	No special restriction, but subject to the regular height and other limitations applicable to Signs.	No special restriction, but subject to the regular height and other limitations applicable to Signs.

Any conditional use permit under subsection H of this section shall be subject to the same use permit criteria as are prescribed in Section 17.102.240.

## 17.108.080 - Minimum side yard opposite living room windows.

On each lot containing Residential Facilities with a total of two or more living units, except in the case of a One-Family Dwelling with Secondary Unit, a side yard with the minimum width prescribed hereinafter shall be provided opposite any legally required window of a living room in a Residential Facility wherever such window faces any interior side lot line of such lot, other than a lot line abutting an alley, path, or public park. The side yard prescribed by this section is not required on other lots or in other situations. Such yard shall have a minimum width of eight (8) feet, plus two (2) feet for each story at or above the level of the aforesaid window; provided, however, that such side yard width shall not be required to exceed ten percent (10%) percent-of the lot width in the RU-3, RU-4, RU-5, R-80, ON, CC, C-40, C-45, CBD, S-1, S-2, S-15, and D-KP zones and fifteen percent (15%) of the lot width in all other zones, except that in no case shall such side yard width be less than five (5) feet. The side yard required by this section shall be provided opposite the legally required window and opposite.that portion of the wall containing

such window, or of any extension of such wall on the same lot, for a distance of not less than eight (8) feet in both directions from the centeriine of such legally required window, and at and above finished grade or the floor level of the lowest story containing such a window, whichever level is higher. Such yard shall be provided unobstructed except for the accessory structures or the other facilities allowed therein by Section 17.108.130.

### 17.108.130 - Exceptions to required openness of minimum yards and courts.

Every part of each required minimum yard and court shall be open and unobstructed from finished grade, or where applicable from such other specified level at which the yard or court is required, to the sky except for the facilities allowed in the yard or court by the following table. Furthermore, in no case shall more than fifty percent (50%) of the horizontal area of any required minimum rear yard be covered by any facilities, other than trees, which extend more than six (6) feet above the level at which the rear yard is required. Wherever a yard is required only for a particular facility, it may be provided at the level of the lowest story containing such facility; provided that where such facility is a Residential Facility, such level shall be that of the lowest story, or portion thereof, containing any living unit. Where the height of facilities within minimum yards or courts is not specifically further limited by the following table, the facilities within minimum yards and courts shall also be subject to any applicable screening requirements or other controls prescribed by the buffering regulations in Chapter 17.110 or by the pertinent development control maps or individual zone regulations, which in some zones require that minimum front yards, or side yards on the street side of a corner lot, be landscaped.

Facilities	Allowed Projection Into or Location Within Minimum Required Yard or Court, Subject to the Further Restrictions Indicated in This Section's First Paragraph (Blanks indicate that facility is not allowed.)				
	Front Yard	Side Yard on Street Side of Comer Lot	Side Yard Along Interior Side Lot Line	Rear Yard (But see coverage limit in first paragraph.)	Court
T. Retaining walls; and earthen mounds, embankments, and other fill.	in any y 1 <del>7.102.</del> 4		such facilities comply with	the provision of <u>Section 17</u>	<u>.108.150.</u>

17.108.140 - Fences, dense hedges, barrier, and similar freestanding walls.

- B. Residential zones and Residential Facilities. The provisions of this section apply to all properties located in <u>all</u> residential zones, and to all properties located in any zone containing Residential Facilities.
  - 1. Height. In the locations specified below, the height of any fence, dense hedge, or barrier or similar freestanding wall, but excluding retaining walls, shall not exceed the following:
    - a. In any minimum front yard, or any minimum side yard on the street side of a corner lot: forty two (42) inches, except that six (6) feet is permitted in the following cases:
      - i. In the portions of street side yards located within the greater of the following distances, from the rear lot line:
        - a) Thirty-five (35) feet from the rear lot line
        - b) The distance between the rear lot line and a line that is perpendicular to the street side lot line and that extends to the rearmost enclosed portion of the primary building on the lot; or
      - ii. Upon the granting of small project design review pursuant to the small project design review procedure in Chapter 17.136.

- b. In any minimum rear yard if within ten (10) feet of a street line that abuts the lot: six (6) feet.
- c. In any other minimum yard or court: eight (8) feet; and
- d. One entry gateway, trellis or other entry structure may be permitted in the required front setback area of each lot provided the maximum height or width of the facility does not exceed ten (10) feet\_;
- 2. Materials. The following materials are restricted in constructing or rebuilding walls or fences:
  - a. Barbed wire or razor wire is not allowed to be used in fences.\_, see also Section 17:102.420.
  - b. Chain link fencing is not allowed to exceed forty-two (42) inches in height in the following locations:
    - i. Street-fronting yards; or
    - ii. Interior side yards if closer to the front lot line than the front wall of the primary Residential Facility.
  - c. Plain concrete blocks are not allowed as a fencing material unless capped and tinished with stucco or other material approved by the Director of City Planning.
- C. Commercial zones and in the <u>OS</u>, S-1, S-2, S-3, and S-15 zones. The provisions of this subsection apply to <u>all properties</u> fences, donse-hedges, barrier and similar freestanding-walls, but-oxcluding retaining-walls, located with in all commercial zones and in the <u>OS</u>, S-1, S-2, S-3, and S-15 zones.
  - 1. Heighti
    - a. The height of any fence, dense hedge, barrier or similar freestanding wall located within ten (10) feet of any abutting property located in a residential zone shall not exceed eight (8) feet. A fence higher than eight (8) feet but no more than ten (10) feet may only be permitted in these locations upon the granting of small project design review pursuant to the small project design review procedure in Chapter 17.136.
    - b. The maximum height of any fence, dense hedge, barrier, or similar freestanding wall elsewhere on a lot shall be ten (10) feet.
  - 2. Restricted materials. In any location visible from the adjacent public right of way, no barbed wire or razor wire shall be permitted as part of or attached to fences or walls, or attached to the exterior of any building or similar facility, -soo-aiso-Section-17.102.420.
    - a. Exceptions: Fences enclosing the following activities shall be exempted from the above limitation on barbed wire and razor wire where the Director of City Planning determines that trespassing could present a public safety hazard and/or disruption of public utility, transportation, or communication services:
      - Public utility installations, including but not limited to electrical substations and gas substations.
      - ii. Rights ef way and transit routes.
- D. Industrial zones. The provisions of this subsection apply to <u>all properties\_fences, dense-hedgos</u>, <u>barrier-and-similar-freestanding-walls-located-with in all industrial zoning districts</u>.
  - 1. Height:
    - a. The maximum height of any fence, dense hedge, barrier or similar freestanding wall located within ten (10) feet of any abutting property located within a residential zone shall be eight (8) feet. A fence higher than eight (8) feet but no more than ten (10) feet may only be permitted in these locations upon the granting of small project design review pursuant to the small project design review procedure in Chapter 17.136.

# 17.110.040 - Special buffering requirements.

- A. Open Storage Areas on Same Lot as Residential Facility—Screening Required Within Three Years. In all zones, on any lot which contains both a Residential Facility and any area devoted to open storage or display of goods or materials, said open storage or display area shall be screened from all abutting lots, streets, alleys, and paths, and private streets or other ways described in Section 17.106.020, by dense landscaping not less than five and one-half (5½) feet high and not less than three (3) feet wide, or by a decorative screening fence or wall not less than tive and one-half (5½) feet high, subject to the standards for required landscaping and screening in Chapter 17.124 and the exceptions stated in said chapter. Existing open storage and display areas on such lots shall either be removed or provided with the above prescribed screening within three years after the effective date of the zoning regulations.
- C. Location of Detached Accessory Buildings on Corner Lot Abutting a Key Lot in a Residential Zone. In ail zones, on any reversed corner lot which abuts a key lot located in any residential zone, no detached accessory building shall be located within tive (5) feet from the abutting side lot line of the key lot. No detached accessory building on such lot shall be located closer to the street line on which the key lot fronts than a distance equal to the minimum front yard depth required on the key lot, unless the accessory building is at least thirty-five (35) feet from the side lot line of the key lot. An accessory building shall be considered detached from any principal building on the same lot if the only roofed attachment thereto consists of a breezeway or similar structure exceeding neither twelve (12) feet in height nor eight (8) feet in width.
- D. Other Provisions. Also applicable are the special provisions, if any, set forth in the applicable individual zone regulations and development control maps with respect to landscaping and screening and controls on parking, loading, and other specified uses; the requirements set forth in Section 17.102.140 for stables, corrals, and similar facilities; and the screening and other standards prescribed for required usable open space in the standards for required usable open space in Chapter 17.126.

Chapter 17.114 - NONCONFORMING USES Sections:

Article I - General Provisions

17.114.010 - Title, purpose, and applicability.

17.114.020 - Definitions.

17.114.030 - Effect of Prior permits.

17.114.040 - Right to continue nonconforming use, subject to limitations.

Article II - Nonconforming Activities

17.114.050 - Nonconforming activity—Discontinuance.

17.114.060 - Nonconforming activity—Damage or destruction.

17.114.070 - Nonconforming activity—Allowed substitutions and other changes in activity.

17.114.080 - Nonconforming activity—Allowed alterations and extensions.

17.114.090 - Nonconforming massage service and adult entertainment activities— Discontinuance required within one year.

17.114.100 - Nonconforming scrap operation commercial activities—Discontinuance required within one year.

Article III - Nonconforming Facilities

17.114.110 - Nonconforming facility—Allowed alterations.

17.114.120 - Nonconforming facility—Damage or destruction.

17.114.130 - Nonconforming open storage on same lot as Residential Facility—Screening required within three years.

17.114.140 - Nonconforming open storage in CN, CR-1 and M-20 zones—Screening required within three years.

17.114.150 - Nonconforming Sign within 1,000 feet of, and primarily viewable from, rapid transit route—Removal required for certain categories.

17.114.160 - Reserved.

17.114.170 - Nonconforming Signs in CN-1 zone—Removal required.

17.114.180 - Nonconforming Signs in CR-1 zone—Removal required for certain categories.

## Article I - General Provisions

## 17.114.030 - Effect of Prior permits.

- A. Fundamental Vested Rights Not Abrogated By Code Adoption or Amendment. The adoption of, or amendment to, the Planning Code (Chapter 17), including without limitation, ordinances enacted pursuant to Oakland City Charter Section 213 (Emergency Ordinances) and/or ordinances enacted pursuant Government Code Section 65858 (Urgency Measure/moratoria), shall not abrogate any fundamental vested rights established pursuant to State law, including, without limitation, those established pursuant to the prior valid adoption and execution of a development agreement under Section 17.138.015 and the development agreement procedure in Chapter 17.138. Absent the prior establishment of such vested rights, any Planning Code adoption or amendment shall apply.
- B. Alcoholic Beverage Control Licenses. On premises for which a valid state of California Alcoholic Beverage Control license had been issued, and which premises had been used in the exercise of the rights and privileges conferred by the license at a time immediately prior to the effective date of the applicable provisions of Section 17.103.030, the premises may hereafter be used in the exercise of the same rights and privileges without requiring a conditional use permit or having to meet the provisions of the aforesaid section. The uses as they result that do not conform to the zoning regulations shall be deemed a nonconforming use and subject to the nonconforming use regulations, except as otherwise provided in Sections 17.114.020 and 17.114.030. For the purposes of this subsection, the word "premises" shall mean and include only the actual space within a building devoted to the sale of alcoholic beverages on said effective date.

A-\_Building-and-Sign-Permits-and-Development-Agreements--As-specified-in-Sections-17-102-040 and-17-102-310,-uses-may-in-certain-cases-bo-established,-constructed,-alterod,-extended, sobstitoted,-moved,-or-otherwise-changed-on-the-basis-of-building-or-sign-permits-or-development agreements-although-the-zoning-regulations-or-a-rezoning-or-other-amendment-thereto-would otherwise-prohibit-such-use,-development,-or-change-

B. Alcoholic-Beverage-Controi-Licenses.-Notwithstanding-tho-provisions-of-the-nonconforming-use regulations, said-provisions-shall-not-apply-to-the-extent-that-they-would-preclude-the-exercise-of-the same-rights-and-privileges-as-those-conforrod-by-a-valid-state-of-California-Alcoholic-Beverage Control-license-for-premises-which-had-been-usod-in-tho-exercise-of-such-rights-and-privileges-at-a time-immediately-prior-to-the effective-date-of-the-applicable-previsions-of-Sectien-17.102.210. For the-purposes-of-this-Subsoction, the-word "premises" shall-mean-and-include-only-the-actual-spaee within-a-building-dovotod-to-tho-sale-of-alcohelie-beverage.

(Prior planning code § 7402)

### 17.114.040 - Right to continue nonconforming use, subject to limitations.

- A. Right to Continue. A nonconforming use which is in existence on the effective date of the zoning regulations or of any subsequent rezoning or other amendment thereto which makes such use nonconforming, and which existed lawfully under the previous zoning controls, or which is subsequently developed or changed pursuant to Section 17.114.030, may thereafter be continued and maintained indefinitely, and the rights to such use shall run with the land, except as otherwise specified in the nonconforming use regulations. However, no substitution, extension, or other change in activities and no alteration or other change in facilities is pennitted except as otherwise provided in Section 17.114.030 and except as specifically provided hereinafter.
- B. <u>Limitation on Right to Continue Nonconforming Auto and Truck Related Activities in All Districts. As used in regards to all such nonconforming auto and truck related activities, the word "activity" refers solely to the unique function or operation occurring on the affected property, and does not refer to any other activity within an activity type with which that activity is grouped. Any right to substitute,</u>

extend or alter an existing auto or truck related activity refers solely to the specific existing function or operation, and does not provide any right to substitute, extend or alter that activity with any other type of activity within the activity type with which the activity is grouped.

## Article II - Nonconforming Activities

17.114.050 - Nonconforming activity-Discontinuance.

- A. Activity Nonconforming Because It Is Not a Permitted Activity. Other than: 1) an Alcoholic Beverage Sales Commercial Activity, 2) the sale of alcoholic beverages at any full-service restaurant in a location described by Section—17-102-210\_17.103.030(B), or 3) an Automotive Servicing or Automotive Repair and Cleaning Activity in the D-BR\_S-5-Zone, whenever an activity that is nonconforming wholly or partly because it is not itself a permitted activity where it is located, occupies four hundred (400) square feet or more of fioor area and hereafter discontinues active operation for a continuous period of one year, or occupies less than four hundred (400) square feet of floor area and hereafter discontinues active operation for a continuous period of six months, and the facilities accommodating or serving such activity are not utilized for another activity during such period, said facilities may thereafter be utilized only for a normally permitted or conditionally permitted activity pursuant to Section 17.114.070A, except the former activity may be resumed after a longer period upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
- B. Whenever an Alcoholic Beverage Sales Commercial Activity, or sale of alcoholic beverages at any full-service restaurant in a location described by Section-17-102-210 17.103.030(B), discontinues active operation for more than <u>ninety (90)</u> days or ceases to be licensed by the State Department of Alcoholic Beverage Control, it may be resumed only upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134. However, if another activity has replaced it, the former activity may thereafter be resumed if and only if such resumption would constitute an allowable change under Section 17.114.070A. Section 17.114.060 shall also apply.
- C. Whenever an Automotive Servicing or Automotive Repair and Cleaning Activity in the D-BR Zone discontinues active operation for more than six (6) months, it may be resumed only upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134. However, if another activity has replaced it, the former activity may thereafter be resumed if and only if such resumption would constitute an allowable change under Section 17.114.070(A). Section 17.114.060 shall also apply.

## 17.114.060 - Nonconforming activity-Damage or destruction.

A. Nonconforming Nonresidential Activities. Facilities accommodating or serving any nonconforming nonresidential activity which are damaged or destroyed to the extent of not more than seventy-five percent (75%) may be restored to their prior condition and occupancy. If such damage or destruction exceeds seventy-five percent (75%), the facilities may thereafter only be restored to accommodate or serve the prior nonconforming activity upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.

17.114.070 - Nonconforming activity---Allowed substitutions and other changes in activity.

A. Activity Nonconforming Because It Is Not a Permitted Activity. The activities specified in the following table may be substituted for any of the indicated activities which is nonconforming wholly or partly because it is not itself a permitted activity where it is located:

Provide the second seco		T
Zone	Prior Nonconforming Activity	Activity Which May be Substituted for Prior

·	<u></u>	Activity, Subject to the Provisions Listed Below This Table
Any zone.	Any such activity.	Any activity othenvise pennitted or, upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, any activity othenwise conditionally pennitted in the same location.
Any Residential zone or S-1, S-2,	Any such Industrial Activity where it is not a permitted or conditionally pennitted activity.	Any Activity permitted in the CN-4 zone.
or S-3 zone.	The following such Commercial Activities where they are not a pennitted or conditionally permitted activity:	
	Research Service	(see below)
	General Wholesale Sales	(see below)
· ·	Building Material Sales	(see below)
	Automobile and Other Light Vehicle Sales and Rental	(see below)
	Automotive and Other Light Vehicle Repair and Cleaning	(see below)
	Taxi and Light Fleet-Based Service	(see below)
	Transport and Warehousing	(see-below)
	Animal Care	(see below)
	Animal Boarding	(see below)
	Undertaking Service	(see below)
	Scrap Operation	(see below)
v		Any Activity permitted in the CC-2 zone.
	The following such Commercial Activities where they are not a permitted or conditionally permitted activity:	
	General Food Sales	(see below)
	Full Service Restaurant	(see below)
	Limited Service Restaurant and Cafe	(see below)
	Fast-Food Restaurant	(see below)
	Convenience Market	(see below)
	Alcoholic Beverage Sales	(see below)
	Mechanical or Electronic Games	(see below)
	General Retail Sales	(see below)
	Consumer Service	(see below)
	Consumer Cleaning and Repair Service	(see below)
	Consumer Dry Cleaning Plant	(see below)
	Group Assembly	(see below)
	Personal Instruction and Improvement and Small Scale Entertainment	(see below)
	Business, Communication, and Media Service	(see below)
······	Broadcasting and Recording Service	(see below)
	[	Any Commercial Activity permitted in the CN-4 zone.
	The following such Commercial Activities where they are not a permitted or conditionally permitted activities:	

٤

	Medical Service	(see below)		
	Consultative and Financial Service	(see below)		
	Administrative	(see below)		
,		Administrative Civic Activities. Administrative Commercial Activities. Medical Service. Consultative and Financial Service.		
	Any other Commercial Activity where it is not a permitted or conditionally permitted activity.	Any Commercial Activity permitted in the CC-2 zone.		
Any Commercial zone.	Any Industrial Activity where it is not a permitted or conditionally permitted activity.	Any Commercial Activity permitted in the CC-2 zone.		
Any Industrial zone.	Any such Commercial Activity where it is not a permitted or conditionally permitted activity.	Any Commercial Activity permitted in the CC-2 zone.		

Changes that do not constitute substitutions may be made in any activity which is nonconforming wholly or partly because it is not itself a permitted activity where it is located. The above substitutions and other changes may be made without regard for requirements on off-street parking and loading, conduct of activities within enclosed buildings, means of customer access, and total floor area which normally apply to activities, except as otherwise provided in Section 17.116.020C. However:

- tf the nonconforming activity is itself conditionally permitted where it is located, no substitution or other change shall be made in it which would conflict with, or further conflict with, any requirement on off-street parking or loading, conduct of activities within enclosed buildings, means of customer access, or total floor area which normally applies to activities. (Changes which are allowed by Section 17.116.020B shall not be deemed to conflict or further conflict with the parking or loading requirements-).
- 2. Conversions of dwelling units to use by a nonresidential activity shall be subject, where applicable, to the provisions of Section 17.102.230.
- 3. If the nonconforming activity is located at ground level on any lot in the CN-1 or CN-2 zone, no change shall be made in the nature of the particular activity, except when the result is itself permitted in the same location, unless a conditional use permit is granted pursuant to the conditional use permit procedure. This does not restrict a change in ownership, tenancy, or management where the previous line of business or other function is not changed.
- 4. For any nonconforming Alcoholic Beverage Sales Commercial Activity presently located in any zone in which it is not a permitted activity, no change shall be made in the activity which change requires obtaining a different type of alcoholic beverage sale retail license from the state of California Department of Alcoholic Beverage Control. Further, no change shall be made in any nonconforming activity involving the sale of alcoholic beverages at a full service restaurant in any location described by Section—17-102-210\_17.103.030(B)(2), which change requires obtaining a different type of alcoholic beverage sale retail license from the state of California Department of Alcoholic beverage sale retail license from the state of california any location described by Section—17-102-210\_17.103.030(B)(2), which change requires obtaining a different type of alcoholic beverage sale retail license from the state of California Department of Alcoholic Beverage Control, unless a conditional use permit is granted pursuant to the conditional use permit procedure in Chapter 17.134.
- 5. No substitution or other change shall be made in any nonconforming activity which would conflict, or further conflict, with any applicable provision of the performance standards in Chapter 17.120, or of any kind of requirement not mentioned hereinabove which applies to activities.
- 6. In cases of discontinuance, damage, or destruction, the pertinent provisions of Sections 17.114.050 or 17.114.060 shall also apply.

If the activity resulting from a change allowed above is not a normally permitted and otherwise conforming activity, and is not authorized by a conditional use permit or other special zoning approval, it shall be deemed a nonconforming activity and changes in it shall be subject to this section.

## 17.114.080 - Nonconforming activity-Allowed alterations and extensions.

- A. Nonresidential Activity Nonconforming Because It Is Not a Permitted Activity. Except as otherwise provided in Section 17.114.060, a nonresidential activity which is nonconforming wholly or partly because it is not itself a permitted activity where it is located may be extended, and the facilities accommodating or serving such activity may be altered or otherwise changed, subject to the requirements normally applying to uses where the activity is located and subject to the following provisions and exceptions:
  - Except as otherwise provided in subsection (A)(3) of this section, the floor area and overall outside dimensions of any building, or portion thereof, devoted to such activity shall not be increased; no open parking, loading, sales, display, service, production, or storage area accommodating or serving such activity shall be relocated or increased in size; and no such building or open area shall be wholly reconstructed. However, in the case of an establishment classified as an Alcoholic Beverage Sales Commercial Activity, the total floor area, open areas, or outside building dimensions occupied by the establishment may be increased as long as the amount of space actually devoted to the sale of alcoholic beverages is not increased by more than twenty (20)-percent (20%) of that already existing. See 17.15.01 (L4), 17.17.01 (L4), and 17.19.01 (L7) for restrictions to this allowable expansion in the RD, RM, and RU zones.
  - 2. In the case of an establishment classified as an Alcoholic Beverage Sales Commercial Activity, the percentage of actual fioor area devoted to the sale of alcoholic beverages shall not be increased by more than twenty percent (20%) of that already existing, except upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.
  - 3. New, wholly reconstructed, enlarged, or relocated structures or open areas devoted to off-street parking or loading serving such activity may be provided wherever Automotive Fee Parking Commercial Activities are permitted or, upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134, wherever Automotive Fee Parking Commercial Activities are conditionally permitted. In residential zones, such facilities for off-street parking may be provided in the situations, and subject to the conditions, prescribed in Section-17-102-100 17.116.075.
  - 4. New Signs may be provided for such activity, but the aggregate area of display surface of all Signs serving such activity shall not be increased. All Signs shall be subject to the limitations, other than aggregate area of display surface, normally applying to Signs where they are located.
  - 5. During any five-year period, beginning on or after the effective date of the zoning regulations or of any subsequent rezoning or other amendment thereto which makes such activity thus nonconforming, the aggregate cost of all alterations for which a building or sign permit is required, and which are intended for any activity subject to this subsection, shall not exceed twenty-five percent (25%) of the replacement cost, as estimated by the Building Services Department, of the facilities accommodating or serving such activity at the beginning of said period. However, the cost of alterations ordered by any governmental agency or permitted by Section 17.114.060 shall be exempt from said maximum cost.
  - 6. No facility accommodating a nonconforming Automobile and Other Light Vehicle Gas Station and Servicing or Automotive and Other Light Vehicle Repair and Cleaning Commercial Activity shall be altered in exterior appearance, unless plans for the proposal have been approved pursuant to the design review procedure in Chapter 17.136.
    - 7. A nonconforming Automobile and Other Light Vehicle Gas Station and Servicing or Automotive and Other Light Vehicle Repair and Cleaning Commercial Activity in the HBX-1 zone may be extended, and the facilities accommodating or serving such activity may be altered or otherwise changed upon the granting of a conditional use permit (see Chapter 17.134) and approval pursuant to the regular design review procedure in Chapter 17.136. This conditional use permit and design review approval may be granted only upon determination that the proposal is

adequately buffered from the street and surrounding residential activities through landscaping and fencing.

17.114.100 - Nonconforming scrap operation commercial activities—Discontinuance required within one year.

Within one (1) year after the effective date of this section or of any subsequent rezoning which makes an existing Scrap Operation Commercial Activity a nonconforming activity, all nonconforming Scrap Operation Commercial Activities located within a residential zone or within one\_-hundred (100) feet of a residential zone and which wholly or partially occupy an open facility shall be discontinued or may continue only upon the granting of a conditional use permit, pursuant to the conditional use permit procedure in Chapter 17.134; provided, however, that if the proposal does not conform to the use permit criteria at Section 17.134.050, but as an alternative a finding is made that the activity involves investment of money in leasehold or improvements such that a longer period is necessary to prevent undue financial hardship, then a conditional use permit shall be granted for a period not to exceed two (2) additional years.

#### Article III - Nonconforming Facilities

### 17.114.110 - Nonconforming facility—Allowed alterations.

- A. When Occupied by Conforming Activity. Except as otherwise provided in Section 17.114.120, a nonconforming facility which accommodates or serves a conforming activity may be altered or otherwise changed, and the lot lines of the lot containing it may be changed, in any way which does not create any new nonconformity or increase the degree of any existing nonconforming with respect to any requirement applying to facilities. Any new, relocated, or wholly reconstructed part of a facility shall itself conform to all applicable such requirements. Nonconforming Residential Facilities containing a total of more than one living unit on a lot, when located in a zone where only one living unit is permitted on a lot, shall be subject to the requirements generally applying in the RU-2 zone with respect to side yards opposite living room windows; courts; and usable open space. Nonconforming Nonresidential Facilities which are not themselves permitted facility types in the zone where they are located shall not be increased in floor area or overall outside dimensions; relocated, except to remove a nonconformity; or wholly reconstructed.
- B. When Occupied by Nonconforming Activity. Except as otherwise provided in Section 17.114.120, a nonconforming facility which accommodates or serves a nonconforming activity may be altered or otherwise changed, and the lot lines of the lot containing it may be changed, subject to the conditions of Section 17.114.080 as well as those of subsection A of this section. In such a case, new Signs of a type not otherwise permitted may be developed as authorized by subsections A and B of Section 17.114.080.
- C. Conversion from Advertising Sign in the CN, CR-1, or S-15 zones. No nonconforming Advertising Sign in the CN, CR-1, or S-15 zones shall be converted, by change of copy or otherwise, to any other type of Sign unless the entire Sign as converted meets all the requirements of said zone for a new Sign, including design review approval.
- D. Conversion from Advertising Sign Within One Thousand (1,000) Feet of, and Primarily Viewable from, Rapid Transit Route. No Advertising Sign shall be converted, to any other type of Sign unless the Sign as converted is approved, in a content-neutral manner, pursuant to the design review procedure in Chapter 17.136 and the provisions of Section 17.104.040A.

## 17.114.120 - Nonconforming facility—Damage or destruction.

A. Nonconforming Nonresidential Facilities. Nonconforming nonresidential facilities which are damaged or destroyed to the extent of not more than seventy-five percent (75%) may be restored to their prior condition and occupancy. If such damage or destruction exceeds seventy-five percent (75%), the facilities may thereafter only be restored to their prior condition upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134.

17.114.130 - Nonconforming open storage on same lot as Residential Facility—Screening required within three years.

On any lot containing a Residential Facility, any open area which is devoted to storage or display of goods or materials shall, within three <u>(3)</u> years after the effective date of the zoning regulations, be either removed or made to conform to the applicable screening requirements of Section 17.110.040A.

(Prior planning code § 7432)

17.114.140 - Nonconforming open storage in CN, CR-1 and M-20 zones—Screening required within three years.

In the CN, CR-1, and M-20 zones, all open storage areas shall, within three <u>(3)</u> years after inclusion in said zones, be either removed or made to conform to the screening requirements of Section 17.110.040B.

17.114.150 - Nonconforming Sign within 1,000 feet of, and primarily viewable from, rapid transit route—Removal required for certain categories.

A. Basic Requirements. Within the indicated time periods, and except as otherwise provided in Subsection B of this Section, all nonconforming Signs in the following categories which are located within <u>one thousand (1,000)</u> feet of the centeriine of a rapid transit route shall be removed, relocated, or otherwise changed so as to conform:

Category	Time Period
Any Business which is painted, or consists of a poster affixed, directly on a building wall or fence; for which design review is prescribed by Section 17.104.040A; and which is or has become primarily Viewable by the passengers on the transit route.	Three (3) years after the effective date of Section 17.104.040 (that date was April 8, 1971) or three (3) years after the date of official determination of the transit route, whichever occurs later.

The Director of City Planning shall determine which Signs are or have become so viewable, subject to appeal pursuant to the administrative appeal procedure in Chapter 17.132.

17.114.170 - Nonconforming Signs in CN-1 zone—Removal required.

A. Basic Requirements. Within the time periods indicated below for the specified categories, and except as otherwise provided in subsection B of this section, all nonconforming Signs shall be removed, relocated, or otherwise changed so as to conform. See also Section 17.114.110C.

Category .	Time Period
	One (1) year after inclusion in the CN-1 zone.

Any other Sign which is nonconforming with respect to any	Three (3) years after inclusion in the
provision of Section 17.148.110	CN-1 zone.

The Director's determination shall be subject to appeal pursuant to the administrative appeal procedure in Chapter 17.132.

.

Chapter 17.116 - OFF-STREET PARKING AND LOADING REQUIREMENTS Sections:

**Article 1 - General Provisions** 

17.116.010 - Title, purpose, and applicability.

17.116.020 - Effect on new and existing uses.

17.116.030 - More than one activity on a lot.

17.116.040 - Determination by Director of City Planning.

Article II - Off-Street Parking Requirements

17.116.050 - Calculation rules.

17.116.060 - Off-street parking-Residential Activities.

17.116.070 - Off-street parking—Civic Activities.

17.116.080 - Off-street parking—Commercial Activities.

17.116.090 - Off-street parking-Industrial Activities.

17.116.100 - Off-street parking—Agricultural and Extractive Activities.

17.116.110 - Special exemptions to parking requirements.

Article III - Off-Street Loading Requirements

17.116.120 - Off-street loading-Residential Activities.

17.118.130 - Off-street loading—Civic Activities.

17.116.140 - Off-street loading—Commercial Activities.

17.116.150 - Off-street loading-Industrial Activities.

17.116.160 - Off-street loading—Agricultural and Extractive Activities.

Article IV - Standards for Required Parking and Loading Facilities

17.116.170 - Property on which parking and loading must be provided.

<u>17.116.175 – Standards and criteria for accessory parking that serves a prohibited activity.</u>

17.116.180 - Conditions for off-street parking or loading.

17.116.190 - Utilization of off-street parking and loading facilities.

17.116.200 - Parking space dimensions.

17.116.210 - Driveways and maneuvering aisles for parking.

17.116.220 - Loading berth dimensions.

Oakland, California, Code of Ordinances

17.116.230 - Driveways and maneuvering aisles for loading.

17.116.240 - Tandem spaces and berths.

17.116.250 - Maximum backing distance.

17.116.260 - Surtacing and grade of parking and loading facilities.

17.116.270 - Screening and setback of parking and loading areas.

17.116.280 - Control on artificial illumination of parking and loading facilities.

17.116.290 - Special requirements applying in some zones.

17.116.300 - Parking accommodation requirements for one- and two-family residential facilities.

17.116.020 - Effect on new and existing uses.

(See illustrations I-19a, b.)

A. New Parking and Loading to Be Provided for New Facilities and Additions to Existing Facilities. Except as otherwise provided in Section 6-17-102-04017.114.030 and 17-102-310 Chapter 17.138, new off-street parking and loading as prescribed hereafter shall be provided for activities occupying facilities, or portions thereof, which are constructed, established, wholly reconstructed, or moved onto a new lot after the effective date of the zoning regulations, or of a subsequent rezoning or other amendment thereto establishing or increasing parking or loading requirements for such activities, except to the extent that existing parking or loading shall be based on the cumulative increase in floor area, or other applicable unit of measurement prescribed hereafter, after said effective date; provided, however, that for an activity occupying a facility existing on said effective date, new parking shall be required for said increase to the extent that the total of such existing facility and the added facilities exceeds any minimum size hereafter prescribed for which any parking is required for such activity.

17.116.060 - Off-street parking-Residential Activities."

A. Permanent and Semi-Transient Residential Activities. Except as otherwise provided in Section 17.44.200, Chapter 17.94, Sections 17.102.300, 17.116.020, 17.116.030, and 17.116.110, and subject to the calculation rules set forth in Section 17.116.050, the following amounts of off-street parking are required for all Permanent and Semi-Transient Residential Activities when located in the indicated zones and occupying the specified facilities and shall be developed and maintained pursuant to the provisions of Article IV of this chapter:

Residential Facility Type	Zone	Requirement
One-Family Dwelling	RH and RD zones, except when combined with the S- 12 zone.	Two (2) spaces for each dwelling unit; howeVer, in the S-11 zone, the requirement shall be one space per bedroom with a minimum of two (2) spaces per dwelling unit and a maximum requirement of four (4) spaces per dwelling unit.
	RM-1, except when combined with the S-12 zone.	One and one-half (1½) spaces for each dwelling unit.
	RM-2 zone	One (1) space for each dwelling unit when lot is less than 4,000 square feet in size and/or 45 feet in width, except when combined with the S-12 zone. One and

. , -		one-half (1½) spaces for each dwelling unit when lot is 4,000 square feet or more in size and/or 45 feel in width, except when combined with the S-12 zone.
	CBD-P zone (when combined with the S-7 zone), except when combined with the S-12 zone.	No spaces required.
	S-15 zone, except when combined with the S-12 zone.	One-haif (½)-space for dwelling unit.
	Any other zone, except when combined with the S- 12 zone	One (1) space for each dwelling unit.
	Any zone combined with the S-12 zone.	See Section 17.94.040.
One-Family Dwelling with Secondary Unit.	RH, RD, RM-1, and RM-2 zones, except when combined with the S-12 zone.	One (1) space for the secondary unit unless the lot already contains a total of at least three (3) spaces; however, in the S-11 zone the requirement shall be one (1) space for each bedroom in any secondary unit, up to a maximum requirement of two (2) spaces per secondary unit. See Section $-17-102-360$ 17.103.080.
	All other zones, except when combined with the S- 12 zone.	One (1) space for the secondary unit unless the lot already contains a total of at least two (2) spaces; however, in the S-11 zone the requirement shall be one (1) space for each bedroom in any secondary unit, up to a maximum requirement of two (2) spaces per secondary unit. See Section-17-102-360 17.103.080.
	Any zone combined with the S-12 zone.	See Section 17.94.040.
Two-Family Dwelling, Multifamily	RD-2, RM-1, RM-2 zones, except when combined with the S-12 zone.	One and one-half (1½) spaces for each dwelling unit.
Dweiling.	CBD-P zone (when combined with the S-7 zone), except when combined with the S-12 zone.	No spaces required.
	S-15 zone, except when combined with the S-12 zone.	One-half (½) space for each dwelling unit.
	Any other zone, except when combined with the S- 12 zone.	One (1) space for each dwelling unit.
	Any zone combined with the S-12 zone.	See Section 17.94.040
Rooming House.	CBD-P zone (when combined with the S-7 zone).	No spaces required.
	Any other zone.	One (1) space for each two rooming units.
Mobile Home.	CBD-P zone (when combined with the S-7 zone).	No spaces required.
	Any other zone.	One (1) space for each living unit plus one (1) additional space for each four living units.
Bed and Breakfast	Any zone.	One (1) space for each two units plus the required parking for a One-Family dwelling in the underlying zone.

# 17.116.070 - Off-street parking-Civic Activities.

Except as otherwise provided in Sections 17.44.200, 17.116.020, 17.116.030, and 17.116.110, and subject to the calculation rules set forth in Section 17.116.050, the following amounts of off-street parking are required for the specified **C**ivic Activities when located in the indicated zones and occupying facilities of the specified sizes or having the indicated numbers of employees or doctors, and shall be developed and maintained pursuant to the provisions of Article IV of this chapter: (See illustration I-18.)

Civic Activity	Zone	Minimum Total Size for Which Parking Required	Requirement
A. Essential Service. Limited Childcare.	S-15 zone.		A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040
	Any other zone.		No spaces required.
B. Community Assembly and Recreational Assembly: playgrounds and playing fields; concessions	CBD-P zone (when combined with the S-7 zone).	-	No spaces required.
located in public parks; temporary nonprofit festivals.	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040
、	Any other zone.	No minimum.	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040
Private non-profit clubs and lodges.	S-15 zone.		A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040
,	Any other zone.	-	No spaces required.
Churches and all other.	CBD-P zone (when combined with the S-7 zone).	-	No spaces required.
	C-45, CBD-P (except when combined with the S-7 zone), CBD-C, CBD-X, and S-2 zones.	10,000 square feet of floor area.	One (1) space for each 20 seats or for each 150 square feet area where seats are not fixed, in principal meeting rooms.
	CN zones	Total of 75 seats or 750 square feet of floor area where seats are not fixed, in principal meeting rooms.	One (1) space for each 15 seats, or for each 100 square feel of floor area where seats are not fixed, in principa meeting rooms.
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040
	Any other zone.	Total of 75 seats, or 750 square feet of floor area where seats are not fixed in principal meeting rooms.	One (1) space for each 10 seats, or for each 100 square feet of floor area where seats are not fixed, in principal meeting rooms.
C. Community Education; high, schools,	CBD-P, CBD-C, and CBD-X zones.	-	No spaces required.
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040
	Any other zone.	No minimum.	One (1) space for each three employees plus one space for each 10 students of planned capacity.
a <u>A</u> ll others.	CBD-P, CBD-C, and CBD-X zones.	-	No spaces required.

•	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	No minimum.	One (1) space for each three employees.	
D. Nonassembly Cultural Administrative.	CBD-P, CBD-C, and CBD-X zones	-	No spaces required.	
	C-45, and S-2 zones.	10,000 square feet of floor area.	One (1) space for each 1,400 square feet of floor area.	
· · · ·	CN zones	3,000 square feet of floor area.	One (1) space for each 900 square feet of floor area.	
	S-15 zone	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	3,000 square feet of floor area.	One (1) space for each 600 square feet of floor area.	
E. Health Care: hospitals.	CBD-P zone (only when combined with the S-7 zone)	-	No spaces required.	
· · ·	C-45, CBD-P (only if not combined with the S-7 zone), CBD-C, CBD-X, and S-2 zones.	No minimum	One (1) space for each staff or regular visiting doctor.	
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	No minimum.	One (1) space for each four beds, plus one space for each four employees other than doctors, plus one space for each staff or regular visiting doctor.	
e <u>C</u> linics.	CBD-P zone (only when combined with the S-7 zone)	-	No spaces required.	
	C-45, CBD-P (only when not combined with the S-7 zone), CBD-C, CBD-X, and S- 2 zones.	No minimum.	One (1) space for each staff or regular visiting doctor.	
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	No minimum.	Three (3) spaces for each staff or regular visiting doctor plus one (1) space for reach two other employees	
a <u>A</u> ll other.	CBD-P zone (only when combined with the S-7 zone).	-	No spaces required.	
	C-45, CBD-P (only when not combined with the S-7 zone), CBD-C, CBD-X, and S- 2 zones.	No minimum.	One (1) space for each staff or regular visiting doctor.	
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	No minimum.	One (1) space for each six beds, plu	

Oakland, California, Code of Ordinances

			one space for each four employees other than doctors, plus one space for each staff or regular Visiting doctor.	
F. Utility and Vehicular.	CDB-P, CBD-C, and CBD-X zones.	-	No spaces required.	
	C-45, and S-2 zones.	10,000 square feet of floor area.	One (1) space for each vehicle used in connection with the activities.	
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	3,000 square feet of floor area.	One (1) space for each three employees plus one space for each vehicle used in connection with the activities.	
G. Extensive Impact: colleges and universities.	CBD-P, CBD-C, and CBD-X zones.	-	No spaces required,	
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section 17.116.040	
	Any other zone.	No minimum.	One (1) space for each three employees plus one space for each six students of planned capacity.	
aAll other.	CBD-P zone (only when combined with the S-7 zone)	-	No spaces required.	
	S-15 zone.	-	A number of spaces to be prescribed by the Director of City Planning pursuant to Section <b>1</b> 7.116.040	
	Any other zone.	No minimum.	A number of spaces to be prescribed by the Director of City Planning pursuant to Section <b>1</b> 7.116.040	

# 17.116.175 - Standards and criteria for accessory parking that serves a prohibited activity.

The following regulations shall apply to parking serving principal activities which are not themselves permitted, wherever such parking is listed in the applicable individual zone regulations as permitted or conditionally permitted. Approval of a permit for such accessory parking is subject to the conditions set forth in this section:

- A. General Conditions. In all zones, such parking facilities shall be used for accessory parking only, with no sales, dead storage, repair work, dismantling, or servicing of any kind.
- B. Conditions in Residential Zones. In all residential zones:
  - Such parking shall not in any case be located farther than one hundred fifty (150) feet, excluding the width of any inten/ening street, from the nearest boundary of any nonresidential zone, as measured perpendicularly from said boundary at any point; and
  - 2. Such parking shall not be so located as to extend along any one side of any street farther into any residential zone than any residentially zoned lot which is in separate ownership and which has frontage on the same side of the same street as said parking, other than a lot developed only for parking; and
  - 3. Such parking facilities shall be open only; and
  - 4. All Signs serving such parking shall be subject to the limitations set forth in Section <u>17.104.010(G)(3)</u>.

17.116.290 - Special requirements applying in some zones.

B. In the S-15 zone:

- 1. Location of Parking. All off-street parking may be provided anywhere on the lot, or on a separate lot which is not in common ownership with the subject lot, provided that a long-term lease agreement or comparable binding agreement is provided, pursuant to Section 17.116.180.
- 2. Parking Serving Nonresidential Uses. Off-street parking serving nonresidential uses may only be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedures in Chapter 17.134.
- 3. Ground Floor Parking and Loading. Off-street parking, loading, and driveway located within twenty (20) feet from all pedestrian walkways and plazas may only be permitted upon the granting of a conditional use permit pursuant to the conditional use permit procedures in Chapter 17.134 and Section 17.100.100.
- 4. Provisions for Shared Parking. Off-street parking may be shared amongst daytime activities between the hours of business operation and between the hours of nighttime activities. The number of required parking spaces for daytime use may be transferable to required parking or nighttime use, provided that a long-term lease agreement or comparable binding agreement is provided, pursuant to Section 17.116.180.
- 5. Exceptions to Parking Requirement. The number of parking spaces provided may exceed the number required upon the granting of a conditional use permit pursuant to Section 17.100.100 and the conditional use permit procedure in Chapter 17.134.

# 17.117.020 - Bicycle parking required for new and existing uses.

- B. Bicycle Parking Shall be Provided for Remodels. "Remodel" means any proposed physical improvement of an existing structure which requires a building permit but does not include New Facilities or Additions to Existing Facilities.
  - Remodel projects that are over ten thousand (10,000) square feet and have an estimated construction cost, excluding seismic retrofit costs, greater than \$250,000.00 shall provide the number of short-term bicycle parking spaces prescribed in Sections 17.117.090 to 17.117.120. This amount shall be adjusted to account for changes in the Building Cost index for the San Francisco Bay Region, as reported in the Engineering News Record. The adjustment shall be made annually, starting in 2009, no sooner than one year from adoption.
  - 2. Remodel projects that are over <u>fifty thousand (50,000)</u> square feet and have an estimated construction cost, excluding seismic retrofit costs, over \$1,000,000.00 shall provide, in addition to short-term bicycle parking, the number of long-term bicycle parking spaces and shower and locker facilities prescribed in Sections 17.117.090 to 17.117.130. This amount shall be adjusted to account for changes in the Building Cost Index for the San Francisco Bay Region, as reported in the Engineering News Record. The adjustment shall be made annually, starting in 2009, no sooner than one year from adoption.

### 17.120.020 - Existing activities.

Activities existing on the effective date of the zoning regulations, or of a subsequent rezoning or other amendment thereto applying more restrictive performance standards to such activities, shall not be required to change their operations to comply with the performance standards. However, their operations shall not be so changed as to result in a greater degree of nonconformity with respect to such standards, except as otherwise authorized under-Soction-17-102-310-and the development agreement procedure in Chapter 17.138. For existing activities meeting the definition specified in Section 17.114.080C., an expansion greater than twenty percent (20%) of production (e.g. non-administrative) floor area is one example of a change in operations that shall be considered an increase in the degree of nonconformity.

## 17.120.050 - Noise.

- I. Commercial Refrigeration Units. Stationary and mobile commercial refrigeration units shall not produce a noise level greater than the noise level standards set forth in this section. Between the hours of ten (10) p.m. and seven (7) a.m., a mobile refrigeration unit shall not be located within two hundred (200) feet of any residential zone boundary unless such unit is within an enclosure which reduces the noise level outside the enclosure to no more than sixty (60) dBA and reduces vibration to a level below the vibration perception threshold set forth in Section 17.120.060.
- J. Commercial Exhaust Systems. Unnecessary noise caused by exhaust from ventilation units, or other air control device shall not produce a noise level greater than the noise level standards set forth in this section between the hours of ten p.m. and seven a.m. and shall not be located within two hundred (200) feet of any residential zone boundary unless such unit is within an enclosure which reduces the noise level outside the enclosure to no more than sixty (60) dBA and reduces vibration to a level below the vibration perception threshold set forth in Section 17.120.060.

Chapter 17.124 - LANDSCAPING AND SCREENING STANDARDS Sections:

17.124.010 - Title, purpose, and applicability.

17.124.020 - Required landscape plan for new residential units and certain additions to Residential Facilities.

17.124.025 - Required landscape plan for new Nonresidential Facilities and certain additions to Nonresidential Facilities.

17.124.030 - Residential landscape requirements for street frontages.

17.124.040 - Residential landscape requirements for downslope lots.

17.124.045 - Trash and Utility Screening

17.124.050 - Assurance of landscaping completion.

17.124.060 - Maintenance.

17.124.070 - Required materials and opacity.

17.124,080 - Combination of materials.

17.124.090 - Reference level for prescribed heights.

17.124.100 - Exceptions to requirements.

17.124.110 - Frequently planted tree species list for Oakland.

17.124.020 - Required landscape plan for new residential units and certain additions to Residential Facilities.

Submittal and approval of a landscape plan for the entire site is required for the establishment of a new residential unit, excluding secondary units of <u>five hundred (500)</u> square feet or less, and for additions to Residential Facilities of over <u>five hundred (500)</u> square feet. The landscape plan and the plant materials installed pursuant to the plan shall conform with all provisions of this Chapter, Title 12 Street, Sidewalks and Public Spaces and the following:

A. Landscape plans for projects involving grading, rear walls on downslope lots requiring conformity with the screening requirements in Section 17.124.040, or vegetation management prescriptions in the S-11 zone shall show proposed landscape treatments for all graded areas, rear wall treatments, and vegetation management prescriptions.

17.124.025 - Required landscape plan for new Nonresidential Facilities and certain additions to Nonresidential Facilities.

Submittal and approval of a landscape plan for the entire site and street frontage is required for the establishment of a new Nonresidential Facility and for additions to Nonresidential Facilities of over <u>one</u> thousand (1,000) square feet. The landscape plan and the plant materials installed pursuant to the plan shall conform with all previsions of this Chapter, Title 12 Street, Sidewalks and Public Spaces and the standards for required landscaping and screening, including the following:-

# 17.124.045 - Trash and Utility Screening

A. Screening of Utility Meters. All utility meters shall either be located within a box set within a building. located on a non-street facing elevation, or screened with vegetation. B. Screening of Trash Containers. All trash containers shall be located in a storage area that is screened from the street and adjacent properties by a wall, fence, or dense landscaping with a minimum height of four (4) feet.

17.124.100 - Exceptions to requirements.

The landscaping and screening requirements set forth in other provisions of the zoning regulations shall be subject to the following exceptions:

- A. Equivalent Screening on Abutting Lot. Prescribed fences, walls, or dense landscaping need not be provided along a lot line if a building, fence, wall, or dense landscaping of at least equivalent height, opacity, and maintenance exists immediately abutting and on the opposite side of said lot line.
- B. Window on Abutting Lot. Prescribed fences, walls, or dense landscaping need not be higher than three and one-half (3½) feet when located opposite and within three (3) feet of any window in a Residential Facility on an abutting lot, other than a window in a basement or cellar, or within three feet of any portion of the same story of the wall containing such window and lying within ten feet in either direction from said window. Landscaping or a fence or wall shall be considered opposite such a window or portion of wall whenever it would be intersected by a horizontal plane drawn from the wall perpendicularly to the window.
- C. Adjacent to Excavated Parking or Other Area. Where a parking, loading, storage, or similar area, or usable open space, is excavated below adjoining tinished grade, the depth of excavation may be deducted there from the prescribed height of fences, walls, or landscaping required to screen the area or space.
  - D. Height Within Required Minimum Yard or Court. Required fences, walls, or dense landscaping need not be higher than three and one-half (<u>3½</u>) feet in that portion of any required minimum yard which lies within ten\_(<u>10</u>) feet of any street line. The height of fences, walls, and dense landscaping shall be limited within all required minimum yards and courts by the applicable provisions of Section 17.108.140
  - E. General Exceptions to Prescribed Heights. The prescribed heights of dense landscaping shall indicate the height to be attained within three (3) years after planting. The height at time of planting may be not more than two (2) feet lower for dense landscaping required to be taller than five (5) feet, and not more than one (1) foot lower for dense landscaping for which a height of less than tive (5) feet is prescribed. An earthen berm not taller than two (2) feet may count toward the prescribed height of any fence, wall, or dense landscaping.

17.124.110 - Frequently planted tree species list for Oakland.

			1	2
No.	Botanical Name	Common Name	Size	H × S
1.	Arbutus unedo	Strawberry Tree	s	25×25
2.	Cercis canadensis	Eastern Redbud	s	25×25
3.	Lagerstroemia indica X L faurjei	Crape Myrtle	s	30×20
4.	Photima fraseri	Photinja	s	20×15
5.	Prunus cerasifera Thundercloud'	Purple Leaf Plum	S	30×20
6.	Pyrus kawakamii	Evergreen Pear	S	25×30
7.	Rhus lancea	African Sumac	s	20×20
8.	Tristania laurina 'Elegant'	Water Gum	S	25×20
9.	Acer buergeranum	Trident Maple	М	30×25

10.	Aesculus carnea 'Briotii'	Red Horsechestnut	м	40×35
11	Eriobotrya defle×a	Bronze Loquat	м	20×20
12.	Geijera parviflora	Australian Willow	м	30×30
13	Ginkgo biloba 'Saratoga' or 'Autumn Gold'	Maidenhair Tree	М	35×30
14	Koetreuleria bipinnata	Chinese Flame Tree	M .	30×30
15.	Koelreuteria paniculata	Golden Rain Tree	м	30×30
16	Laurus nobilis 'Saratoga'	Saratoga Laurel	м	40×20
17	Magnolia grandiflora 'Saint Mary'	Saint Mary Magnolia	м	20×20
1 <b>8</b> .	Maytenus boaria 'Green Showers'	Mayten Tree	M	30×25
1 <b>9</b> .	Metrosideros excelsus	New Zealand Christmas Tree	м	30×30
20.	Olea europa 'Swan Hill'	Olive	M	40×40
21.	Pyrus calleryana 'Aristocrat'	Aristocrat Pear	. M	40×30
22.	Carpinus betulus 'Fastigiata'	European Hornbern	L	50×40
23.	Fraxinus oxycarpa 'Raywood'	Raywood Ash	Ĺ	35×25
24.	Gliditsia tríacanthos inermis 'Shademaster'	Thornless Honey Locust	L	40×30
25.	Nyssa sylvatica	Sour Gum or Tupelo	L	50×25
26.	Pistacia chinensis 'Keith Davey' or 'Pearl Street'	Chinese Pistache	L	50×30
27.	Platanus acerifolia 'Yarwood'	London Plane	L	70×50
28.	Podocarpus gracilior	African Fern Pine	L	30×20
29.	Quereus rubra	Red Oak	L	50×40
30.	Quercus coccinea	Scarlet Oak	L	75×50

1. Size: (S) Small, (M) Medium, (L) Large

2. H × S: Height by Spread

# 17.126.040 - Private usable open space.

All required private usable open space shall be permanently maintained; shall be located, except as otherwise provided in subsection B of this section, on the same lot as the living unit it serves; and shall conform to the following standards:

- A. Usability. A surface shall be provided which prevents dust and allows convenient use for outdoor activities. Such surface shall be any practicable combination of lawn, garden, flagstone, wood planking, concrete, asphalt, or other serviceable, dustfree surfacing. Slope shall not exceed ten percent (10%). Off-street parking and loading areas, driveways, and service areas shall not be counted as usable open space. Adequate safety railings or other protective devices shall be erected wherever necessary for space on a roof or balcony, but shall not be more than the minimum height required by the Oakland Building Code.
- B. Location. The space may be located anywhere on the lot, except that ground-level space shall not be located in a required minimum front yard and except that above-ground-level space shall not be located within five (5) feet of an interior side lot line. Above-ground-level space may be counted even though it projects beyond a street line. All spaces shall be adjacent to, and not more than four (4) feet above or below the floor level of, the living unit served.
- C. Size and Shape. An area of contiguous ground-level space shall be of such size and shape that a rectangle inscribed within it shall have no dimension less than ten (10) feet. An area of above-ground-level space shall be of such size and shape that a rectangle inscribed within it shall have no dimension less than five (5) feet. When space is located on a roof, the area occupied by vents or other structures which do not enhance usability of the space shall not be counted toward the above dimension.
- D. Accessibility. The space shall be accessible to only one living unit by a doorway to a habitable room or hallway.
- E. Openness. There shall be no obstructions over ground-level space except for devices to enhance its usability and except that not more than fifty percent (50%) of the space may be covered by a private balcony projecting from a higher story. Above-ground-level space shall have at least one exterior side open and unobstructed, except for incidental railings or balustrades, for eight (8) feet above its floor level.
- F. Enclosure. Ground-level space shall be screened from abutting lots, streets, alleys, and paths, from abutting private ways described in Section 17.106.020, and from other areas on the same lot by a building wall, by dense landscaping not less than five and one-half (5½) feet high and not less than three (3) feet wide, or by a solid or grille, lumber or masonry fence or wall not less than five and one-half (5½) feet high, subject to the standards for required landscaping and screening in Chapter 17.124 and the exceptions stated in said chapter. However, when such screening would impair a beneficial outward and open orientation or view, with no building located opposite and within fifty (50) feet from such required screening, as measured perpendicularly therefrom in a horizontal plane, the above prescribed height may be reduced to three and one-half (3½) feet. Fences and walls shall not be so constructed as to interfere with the access required by applicable fire prevention regulations.

Chapter 17.130 - ADMINISTRATIVE PROCEDURES GENERALLY Sections:

17.130.010 - Title, purpose, and applicability.

17.130.020 - Alternative notification procedures.

17.130.030 - Nofice by mail.

17.130.040 - Procedure for resolving tie votes.

17.130.050 - Presentation of written and documentary evidence.

17.130.060 - Obligation of applicant to defend, indemnify, and hold harmless the City of Oakland.

**17.130.070** - Uniformly applied development standards automatically imposed as standard conditions of approval for development projects.

17.130.080 - City Council consideration of legislative and adjudicatory actions.

17.130.090 - Minor land use permits considered concurrently with Major permits.

# 17.130.050 - Presentation of written and documentary evidence.

Whenever, pursuant to the Oakland Planning Code, <u>a</u> an-appeal-or-matter of original jurisdiction, for which a hearing is required, is pending before the City Council, or City Planning Commission, or-the Commission's Residential-Appeals Committee, any interested party, while the hearing is open, may submit written and/or documentary evidence to the City Council<u>or</u>, the Commission, or-the-Committee, whichever is applicable, for its consideration. Whenever, <u>pursuant to the Oakland Planning Code</u>, an appeal for which a hearing is required is pending before the City Council, City Planning Commission, or the Commission's Residential Appeals Committee, the appellant may not submit written and/or documentary evidence not previously submitted in the appeal form itself and presented: (a) prior to the close of the written public comment period for the underlying decision being appealed, in the case of appeals based on a decisions made by the City Planning for the underlying decision being appealed, in the case of appeals based on decisions made by the City Planning Commission, as applicable.

17.130.060 - Obligation of applicant to defend, indemnify, and hold harmless the City of Oakland.

- A. To the maximum extent permitted by law, the applicant shall defend (with counsel-reasonably acceptable to the City), indemnify, and hold harmless the City of Oakland, the Oakland City Council, the City of Oakland Redevelopment Agency, the City of Oakland Redevelopment Successor Agency, the Oakland City Planning Commission and its respective agents, officers, volunteers, and employees (hereafter collectively called City) from any liability, damages, claim, judgment, loss (direct or indirect), action, causes of action or proceeding (including legal costs, attorneys' fees, expert witness or consultant fees, City Attorney or staff time, expenses or costs) (collectively called "Action") against the City to attack, set aside, void or annul, (1) an approval by the City relating to a development-related application or subdivision, and/or a (Lease) Disposition and Development Agreement; or (2) implementation of such an approved development-related project. The City may elect, in its sole discretion, to participate in the defense of said Action and the applicant shall reimburse the City for its reasonable legal costs and attorneys' fees.
- B. Within ten (10) calendar days of the filing of any Action as specified in subsection A above, the applicant shall execute a Letter of Agreement with the City, acceptable to the Office of the City Attorney, which memorializes the above obligations. These obligations and the Letter of Agreement shall survive termination, extinguishment or invalidation of the approval. Failure to timely execute the

Letter of Agreement does not relieve the applicant of any of the obligations contained in this Section or any other requirements or conditions of approval that may be imposed by the City.

**17.130.080** - City Council consideration of legislative and adjudicatory actions.

When a development application requires both legislative and adjudicatory actions, the entire application shall be considered by the City Council for tinal action. The City Council has the authority to consider and revise as appropriate (accept, reject, or modify) the adjudicatory land use decisions of the Planning Commission, regardless of whether an appeal to the City Council is filed challenging such adjudicatory land use decisions.

## 17.130.090 - Minor land use permits considered concurrently with Major permits.

- A Any Minor land use related permit and/or approval that is related to a development application that also includes any Major land use related permit and/or approval shall be considered concurrently with the Major land use related permit and/or approval, and shall follow all procedural requirements associated with City Planning Commission decisions. In this case, the entire application shall initially be considered by the City Planning Commission and may be appealed to the City Council, in accordance with the requirements for Major land use related permit and/or approval or discretionary actions.
- B. Any Minor land use related permit referred to the City Planning Commission for initial decision in order to be considered concurrently with any Major land use related permit and/or approval shall still be considered a Minor land use related permit and/or approval, and the required tindings for said Minor land use related permit and/or approval shall apply.

## 17.132.020 - Appeal.

Within ten (10)\_calendar days after the date of any administrative determination or interpretation made by the Director of City Planning under the zoning regulations, an appeal from such decision may be taken to the City Planning Commission by any interested party. In the case of appeals involving one- or two-unit Residential Facilities, the appeal shall be considered by the Commission's Residential Appeals Committee. Such appeal shall be made on a form prescribed by the City Planning Department and shall be filed with such Department and shall be accompanied by such a fee as specified in the City fee schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Director or wherein his or her decision is not supported by the evidence in the record. The appeal shall be accompanied by such information as may be required to facilitate review. Upon receipt of the appeal, the Secretary of the City Planning Commission shall set the date for consideration thereof and, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal.

## 17.132.040 - Appeal to Council on transit line sign controls.

Within ten (10) calendar days after the date of a decision by the City Planning Commission on an administrative appeal invoiving the provisions of Sections 17.104.040 or 17.114.150, an appeal from said decision may be taken to the City Council by any interested party. In event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Commission and shall be filed with the City Clerk. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. Upon receipt of the appeal, the Council shall set the date for consideration thereof. After setting the hearing date, the Council, prior to hearing the appeal, may refer the matter back to the Planning Commission for further consideration and advice. Appeals referred to the Planning Commission shall be considered by the Commission at its next available meeting. Any such referral shall be only for the purpose of issue clarification and advice. In all cases, the City Council shall retain jurisdiction and, after receiving the advice of the Planning Commission, shall hold a hearing on and decide the appeal. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. In considering the appeal, the Council shall review the purpose and intent, as well as the letter, of the pertinent provisions, and shall affirm, modify, or reverse the Commission's decision. The decision of the Council shall be final.

17.134.020 - Definition of major and minor conditional use permits.

- A. Major Conditional Use Permit. A conditional use permit is considered a major conditional use permit if it involves any of the following:
  - 1. Thresholds. Any project that meets any of the following size thresholds:
    - a. The actual project site (including only portions of the lot actually affected by the project) exceeds one acre;
    - Nonresidential projects involving twenty-five thousand (25,000) square feet or more of floor area, except in the R-80, CBD-R, CBD-P (when not combined with the S-7 zone), CBD-C, CBD-X, S-2, or S-15 zones;
    - Residential projects requiring a conditional use permit for density resulting in a total number of living units as follows;
      - *i.* Three (3) or more dwelling units in the RM-2 zone,
      - ii. Seven (7) or more dwelling units in the RM-3 or RM-4 zone.
    - d. Residential projects requiring a conditional use permit to exceed the basic or permitted density which results in seven (7) or more living units in the RU or CBD-R zone.
    - e. Large Scale Developments. Any development which is located in the R-80, CBD-R, CBD-P (when not combined with the S-7 zone), CBD-C, CBD-X, S-2, or S-15 zone and results in more than one hundred thousand (100,000) square feet of new floor area.
  - Uses. Any project that involves any of the following activity or facility types except where the proposal involves only accessory parking, the resumption of a discontinued nonconforming activity, or an addition to an existing activity which does not increase the existing floor area by more than twenty (20)-percent (20%):
    - a. Activities:
      - i. Residential Care Residential,
      - ii. Service Enriched Housing Residential,
      - iii. Transitional Housing Residential,
      - iy. Emergency Shelter Residential,
      - v. Extensive Impact Civic,
      - vi. Fast-food Restaurant Commercial,
      - vii. Convenience Market Commercial,
      - viii. Alcoholic Beverage Sales Commercial or sale of alcoholic beverages at any fullservice restaurant in a location described by Section-17-102-210 17.103.030(B),
      - ix. Heavy Manufactuhng,
      - x. Small Scale Transfer and Storage Hazardous Waste Management,
      - xi. Industrial Transfer/Storage Hazardous Waste Management,
      - xii. Mining and Quarrying Extractive

xiii. Special Health Care Civic Activities.

- b. Facilities:
  - i. Drive-Through,

ii. Advertising Sign, except when the facility meets the requirements of Section 17.11.090.

iii.----Spocial-Health-Care-Civis-Activitios-

- 3. Special Situations. Any project that involves any of the following situations:
  - a. Any project that requires development of an Environmental Impact Report (EIR);
  - b. Any single establishment containing a Commercial or Industrial Activity, or portion thereof, which is located in any residential zone and occupies more than five thousand (5,000) square feet of floor area, except where the proposal involves only the resumption of a nonconforming activity;
  - c. Off-Street Parking Facilities in the C-40, CBD-P, CBD-C, CBD-X, and S-2 zones serving fifty (50) or more vehicles;
  - d. Transient Habitation Commercial Activities in the C-40 and C-45 zones;
  - e. Monopole Telecommunications Facilities in, or within three hundred (300) feet of the boundary of, any residential or HBX zone;
  - f. Any project in the OS zone listed as requiring a major conditional use permit in Chapter 17.11;
  - g. Any electroplating activity as defined in Section 17.09.040 subject to the provisions of Section 17.102.340;
  - h.-...Any-conditional-use-permit-application-referred-by-tho-Director-of-City-Planning-to-the-City-Planning-to-the-City-Planning-Commission-for-initial-decision-pursuant-to-Soction-17-134-040(B)(1);
  - hi. Any Telecommunications Facility in or within one hundred (100) feet of the boundary of any residential or HBX zone;
  - ij. Any Telecommunications Facility whose antennas and equipment are not fully concealed from view within three hundred (300) feet of the boundary of the RH, RD, RM, RU-1, or RU-2 zones, or any HBX zone.

### 17.134.040 - Procedures for consideration.

A. Major Conditional Use Permits.

In All Zones. An application for a major conditional use permit shall be considered by the City 1. Planning Commission which shall hold a public hearing on the application. Notice of the hearing shall be given by posting an enlarged notice on the premises of the subject property involved in the application. Notice of the hearing shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the city within three hundred (300) feet of the property involved; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Commission for its consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Commission shall determine whether the proposal conforms to the general use permit criteria set forth in Section 17.134.050 and to other applicable use permit criteria, and may grant or deny the application for the proposed conditional use permit or require such changes or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said criteria. The determination of the Commission shall become final ten (10) calendar days after the date of decision unless appealed to the City Council in accordance with Section 17.134.070. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Commission prior to the close of the Commission's public hearing on

the matter. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal.

- Alcoholic Beverage Sales Activities in Alcoholic Beverage Sales License Overconcentrated Areas. In addition to following the provisions of subsection (A)(1) of this section, the City Planning Commission shall also determine whether the proposal conforms to the criteria for findings of "Public Convenience and Necessity" set forth in Section-17-102-210 <u>17,103.030(B)(3)</u>.
- 3. In the OS Zone. Applications for conditional use permits in the OS zone shall be subject to the special use permit review procedure for the OS zone established in Chapter 17.135.
- B. Minor Conditional Use Permits, <sup>1</sup>
  - In All Zones. An application for a minor conditional use permit shall be considered by the 1 Director of City Planning. However, the Director may, at his or her discretion, refer the application to the City Planning Commission for decision rather than acting on it himself or herself. In this case, the application shall still be considered a minor permit, but shall be processed according to the procedure in as-a-major-conditional-use-permit-pursuant-to subsection A of this section. In these instances, any other minor permits associated with the application shall be considered concurrently by the Planning Commission, pursuant to Section 17.130.090. At-his-or-her-discrotion,--an-administrative-hearing-may-bo-held,-Notice shall be given by posting an enlarged notice on the premises of the subject property involved in the application; notice shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the city within three hundred (300) feet of the property involved; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing, if such is to be held, or, if not, for decision on the application by the Director. Any interested party must enter into the record any issues and/or evidence: (a) to the Director prior to the close of the written public comment period for his or her consideration, or (b) to the Commission while the hearing is open for its consideration, whichever is applicable; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Director shall determine whether the proposal conforms to the general use permit criteria set forth in Section 17.134.050 and to other applicable use permit criteria, and may grant or deny the application for the proposed conditional use permit or require such changes in the proposed use or impose such reasonable conditions of approval as are in his or her judgment necessary to ensure conformity to said criteria. The determination of the Director of City Planning shall become final ten (10) calendar days after the date of decision unless appealed to the City Planning Commission in accordance with Section 17.134.060. In those cases which are referred to the Commission by the Planning Director, the decision of the Commission shall become tinal ten (10) days after the date of decision unless appealed to the City Council in accordance with Section 17.134.070. Any party seeking to appeal the determination will be limited to issues and/or evidence presented (a) to the Director prior to the close of the written public comment period, or (b) to the Commission prior to the close of the Commission's public hearing on the matter, whichever is applicable. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal.
  - 2. In the OS Zone. Applications for conditional use permits in the OS zone shall be subject to the special use permit review procedure for the OS zone established in Chapter 17.135.
- C. Alternative Notification Procedures. If the conditions as set forth in Section 17.130.020 apply, alternative notification procedures discussed therein may replace or supplement the procedures set forth in subsections A and B of this section.

### 17.134.050 - General use permit criteria.

- D. That the proposal conforms to all applicable regular design review criteria set forth in the regular design review procedure at Section 17.136.050;
- F. For proposals involving a One- or Two-Family Residential Facility: If the conditional use permit concerns a regulation governing maximum height, minimum yards, maximum lot coverage, or maximum fioor area ratio, the proposal also conforms with at least one of the following additional criteria:
  - 1. The proposal when viewed in its entirety will not adversely impact abutting residences to the side, rear, or directly across the street with respect to solar access, view blockage and privacy to a degree greater than that which would be possible if the residence were built according to the applicable regulation, and, for conditional use permits that allow height increases, the proposal provides detailing, articulation or other design treatments that mitigate any bulk created by the additional height; or
  - 2. At least sixty (60)-percent (60%) of the lots in the immediate context are already developed and the proposal would not exceed the corresponding as-built condition on these lots, and, for conditional use permits that allow height increases, the proposal provides detailing, articulation or other design treatments that mitigate any bulk created by the additional height. The immediate context shall consist of the five (5) closest lots on each side of the project site plus the ten (10) closest lots on the opposite side of the street (see illustration I-4b); however, the Director of City Planning may make an alternative determination of immediate context based on specific site conditions. Such determination shall be in writing and included as part of any decision on any conditional use permit

### 17.134.060 - Appeal to Planning Commission—Minor conditional use permits.

Within ten (10) calendar days after the date of a decision by the Director of City Planning on an application for a minor conditional use permit, an appeal from said decision may be taken to the City Planning Commission by the applicant or any other interested party. In the case of appeals involving oneor two-unit Residential Facilities, the appeal shall be considered by the Commission's Residential Appeals Committee. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the City Planning Department and shall be filed with such Department, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Director or wherein his or her decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to the Director of City Planning prior to the close of the written public comment period on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of the appeal, the Secretary of the City Planning Commission shall set the date for consideration thereof; which in the case of applications limited to oneor two- unit Residential Facilities, shall be the date of the Committee's next regulariy scheduled meeting following the thirtieth day after the appeal is filed. Not less than seventeen (17) days prior to the date of the Commission's or Committee's consideration of the appeal, the Secretary shall give written notice to: the applicant; the appellant in these cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented to the Director of City Planning prior to the close of the written public comment period for the underlying decision being appealed. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Commission or, if applicable, the Committee shall determine whether the proposal conforms to the general use permit criteria set forth in Section 17.134.050 and to any other applicable use permit criteria, and may grant or deny a permit or

Oakland, California, Code of Ordinances
require such changes in the proposed use or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said criteria. The decision of the Commission or, if applicable, the Committee shall be final.

17.134.070 - Appeal to Council—Major conditional use permits.

A. With the exceptions of appeal for adult entertainment activities, appeals to the City Council shall be governed by the following:

Within ten (10) calendar days after the date of a decision by the City Planning Commission on an application for a major conditional use permit, an appeal from said decision may be taken to the City Council by the applicant, the permit holder, or any other interested party. In event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Commission and shall be filed with the City Clerk, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of the appeal, the Council shall set the date for consideration thereof. After setting the hearing date, the Council, prior to hearing the appeal, may refer the matter back to the Planning Commission for further consideration and advice. Appeals referred to the Planning Commission shall be considered by the Commission at its next available meeting. Any such referral shall be only for the purpose of issue clarification and advice. In all cases, the City Council shall retain jurisdiction and, after receiving the advice of the Planning Commission, shall hold a hearing on and decide the appeal. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the proposed use conforms to the applicable use permit criteria, and may grant or deny a permit or require such changes in the proposed use or impose such reasonable conditions of approval as are, in its judgment, necessary to ensure conformity to said criteria. The decision of the City Council shall be made by resolution and shall be final. The City-Council shall-vote-on the appeal within thirty (30) days after its first hearing of the appeal. If the Council is unable to decide the appeal at that meeting, it shall appear for a vote on each regular meeting of the Council thereafter until deoided-

B. Appeals to the City Council relating to adult entertainment activities shall be governed by the following:

Within ten (10) calendar days after the date of a decision by the City Planning Commission on an application for a major conditional use permit, an appeal from said decision may be taken to the City Council by the applicant, the permit holder, or any other interested party. In event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Commission and shall be filed with the City Clerk, along with the appropriate fees required by the City's <u>Master Fee Schedule</u>. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its

Oakland, California, Code of Ordinances

Page 274

public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of the appeal, the Council shall set the date for consideration thereof. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the council shall determine whether the proposed use conforms to the applicable special use permit criteria, and shall grant the permit if it determines that all the said criteria are present or require such chances in the proposed use or impose such reasonable conditions of approval as are, in its judgment, necessary to ensure conformity to said criteria. The decision of the City Council shall be made by resolution and shall be final. The City Council shall vote on the appeal within thirty (30) days after its first hearing of the appeal and --If-the Council-is-unable-to-decide-the-appeal at-that-meeting-it-shall-appear-for-a-vote-on each-regular-mooting-of-the-Council-thereafter-until-decided-In-any-ovont,-however,-the-City-Council-must decide the appeal within sixty (60) days of the appeal being filed.

17.134.110 - Conditional use permit related to planned unit development or subdivision.

Whenever a conditional use permit is required for a proposal also requiring a planned unit development permit, application for the use permit shall be included in the application for the planned unit development permit and shall be processed and considered as part of same. Whenever a conditional use permit is required within a proposed subdivision, the application for the use permit may be submitted with the tentative map or tentative parcel map required by the Oakland Municipal Code, and may be processed and considered therewith. In either case, however, the reviewing officer or body shall, in considering such a use permit, determine whether the proposal conforms to all the applicable use permit criteria.

17.135.020 - Exemptions.

- Projects approved by the City Council in conjunction with the public art program, Measure AA (1989), Measure K (1990), and Measure I (1996).
- B. Business and Advertising Signs. Business and Advertising Signs are exempt from these provisions only when a city agency enters into an agreement with a private enterprise to enhance public park facilities and/or programs, and the private enterprise is a principal provider of cash and/or in-kind contribution toward the enhancements. Such signs will meet the requirements of Section 17.11.090.
- 17.135.030 Procedure for consideration.
  - C. Public Hearing. A public hearing shall be required for any change in use or improvement and shall be conducted and heard by the City Planning Commission and/or the Parks and Recreation Advisory Commission, as provided by subdivisions 1 and 2 of this subsection.
    - 1. Major Conditional Use Permits.
      - An application for a major conditional use permit, as required by Sections 17.11.060 a. and 17.11.090, shall be considered first by the Parks and Recreation Advisory Commission (PRAC) and second by the City Planning Commission. Each commission shall conduct a public hearing on the application. Notice of the PRAC hearing shall follow the procedure outiined at Section 17.135.030(C)(2). Notice of the City Planning Commission hearing shall be given by posting an enlarged notice on the premises of the subject property. At the discretion of the Director, notice of the public hearing may also be provided on utility poles within three hundred (300) feet of such park or open space land. Notice of each hearing shall also be given by mail or delivery to all persons owning real property in the city of Oakland within three hundred (300) feet of the property involved; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing. Notice shall also be provided to those community or neighborhood groups included in the Planning Department database that are within the service area radius of the impacted park. Additional outreach shall be provided through press releases and other notification as warranted by the size and location of the project.
      - b. The PRAC shall schedule its public hearing within foriy-five (45) days after receiving the application for consideration. The PRAC shall make a recommendation to the Planning Commission at the conclusion of the hearing. In the event the PRAC has not acted on the application within forty-five (45) days, the project shall automatically be forwarded to the City Planning Commission.
      - c. The City Planning Commission shall determine whether the proposal conforms to the use permit criteria set forth in Section 17.11.110 and to other applicable criteria, and shall make a recommendation to grant or deny the application, or recommend such changes or impose such conditions of approval as are in its judgment necessary to ensure conformity to said criteria. The determination of the Commission shall become tinal within ten calendar days after the date of the decision unless appealed to the City Council in accordance with Section 17.134.070.
    - 2. Minor Conditional Use Permits.
      - a. An application for a minor conditional use permit, as required by Sections 17.11.060 and 17.11.090, shall be considered by the Parks and Recreation Advisory Commission prior to a final decision by the Director of City Planning. The Parks and Recreation Advisory Commission shall hold a noticed public hearing on the application and shall make a recommendation to grant or deny the application, or

recommend such changes or conditions of approval as are in its judgment necessary. Notice of the public hearing shall be provided by posting an enlarged notice on the premises of the park or open space land. At the discretion of the Director, the meeting notice may also be provided on utility poles within three hundred (300) feet of such park or open space land. Notices shall also be mailed to neighborhood organizations and individuals who have expressed an interest in the subject park or project area.

b. The Director of City Planning shall determine whether the proposal conforms to the special use permit criteria set forth in Section 17.11.110 and to other applicable criteria and shall grant, deny, or conditionally grant the permit The determination of the Director of City Planning shall become final within ten calendar days after the date of the decision unless appealed to the City Planning Commission in accordance with Section 17.134.060. If no action is taken by the Director of City Planning within thirty (30) days of the Parks and Recreation Advisory Commission's recommendation, the project shall be deemed approved.

#### 17.135.040 - Referral to Landmarks Preservation Advisory Board.

Any project in the OS zone requiring a major or minor conditional use permit shall be subject to review by the Landmarks Preservation Advisory Board if that project is located:

- A. Within the S-7 zone;
- B. On a site that could potentially impact a structure, site, or feature that is listed on the State or National Registers, or that has been formally designated as an Oakland landmark.

For projects requiring a minor conditional use permit, this review shall be made after the public hearing of the Parks and Recreation Advisory Commission and before the final decision of the Director of City Planning. For projects requiring a major conditional use permit this review shall be made after the public hearing of the Parks and Recreation Advisory Commission and before the public hearing of the City Planning Commission. The Landmarks Preservation Advisory Board may recommend modifications to the project that it deems necessary to ensure that the historic value of the structure, site, or feature is not adversely impacted. If no action is taken by the Landmarks Preservation Advisory Board within thirty (30) days of its receipt of the application, the project will be forwarded to the Planning Commission (for major conditional use permits) or Director of City Planning (for minor conditional use permits).

Chapter 17.136 - DESIGN REVIEW PROCEDURE Sections:

17.136.010 - Title, purpose, and applicability.

17.136.020 - Application.

17.136.025 - Exemptions from design review.

17.136.030 - Small project design review.

17.136.035 - Small project design review criteria.

17.136.040 - Regular design review.

17.136.050 - Regular design review criteria.

17.136.055 - Special regulations for historic properties in the central business district zones.

17.136.060 - Review by Landmarks Board in certain cases.

17.136.070 - Special regulations for designated landmarks.

**17.136.075** - Regulations for demolition or removal of designated historic properties and potentially designated historic properties.

17.136.080 - Appeal to Planning Commission—Regular design review.

17.136.090 - Appeal to City Council—Regular design review.

17.136.100 - Adherence to approved plans.

**17.136.120** - Design review related to conditional use permit, planned unit development, variance, or subdivision.

17.136.130 - Limitation on resubmission—Small project design review.

## 17.136.025 - Exemptions from design review.

- B. Definition. The following types of work are exempt from design review, pursuant to all provisions in Section 17.136.025(A):
  - 1. Additions or Alterations.
    - Projects not requiring a building permit, except if otherwise specified below;
    - b. Repair or replacement of existing building components in a manner that visually matches the existing or historical design of the structure;
    - c. After notice to the Director of City Planning, demolition or removal of either:
      - Structures declared to be unsafe by the Building Official or the City Council. "Unsafe structures" means structures found by the Building Official or the City Council, to require immediate issuance of a demolition permit to protect the public health and safety; or
      - ii) Structures declared be a public nuisance by the Building Official or City Council that are not Designated Historic Properties or Potentially Designated Historic Properties.
    - Secondary Units of five\_-hundred (500) square feet or less on a lot with only one existing or proposed primary dwelling unit, pursuant to all regulations in Section <u>17.102.360</u> <u>17.103.080;</u>

- e. Floor area additions within the existing building envelope not involving the creation of a dwelling unit;
- f. Cumulative additions over a three (3) year period not involving the creation of a dwelling unit that are outside the existing building envelope and equal no more than ten percent (10%) of the total floor area or footprint on site;
- g. For Commercial, Civic, or Industrial Facilities and the Non-residential Portions of Mixed-Use Development Projects, any addition or alteration on a roof that does not project above the existing parapet walls; and any addition or alteration not otherwise exempt which is used as a loading dock, recycling area, utility area, or similar open structure addition that is no higher than six (6) feet above finished grade, less than five hundred (500) square feet in floor area or footprint, and is visually screened from neighboring properties; such exemptions shall only permitted where the proposal conforms with all Buffering regulations in Chapter 17.110 and all Performance Standards in Chapter 17.120;
- h. Areas of porch, deck or balcony with a surface that is less than thirty (30) inches above finished grade.

# 2. Signs.

- a. A change of sign face copy or new sign face within an existing Advertisement Sign or a change of sign face copy within Business or Civic Sign structures so long as the structure and framework of the sign remain unchanged and the new sign face duplicates the colors of the original or, in the case of an internally illuminated sign, the letter copy is light in color and the background is dark;
- Installation, alteration or removal of Realty Signs, Development Signs, holiday decorations, displays behind a display window and, except as otherwise provided in Section 17.114.120(C), for mere changes of copy, including cutouts, on Signs which customarily involve periodic changes of copy;
- c. New or modified Signs conforming to an approved Master Sign Program, pursuant to Section 17.104.070.
- 3. Other Projects.
  - Sidewalk Cafes that have a maximum of five (5) tables and no more than fifteen (15) chairs and/or do not have any permanent structures in the public right of way, pursuant to Section 47-102-335\_17.103.090;
  - b. Solar Power Production Equipment The installation of Solar Power Production Equipment is exempt from design review within any zoning district

## 17.136.030 - Smalt project design review.

- B. Definition of "Small Project". Small Projects are limited to one or more of the following types of work:
  - 1. Additions or Alterations.
    - a. Repair or replacement of existing building components in a manner that is compatible with, but not necessarily identical to, the property's existing or historical design;
    - b. Except as otherwise specified in Sections 17.136.025, and 17.136.040, demolition or removal of structures not involving a Designated Historic Property or Potential Designated Historic Property, on a site where the zoning regulations require design review to alter the exterior appearance of the applicable building facility, regardless of whether the owner intends to create a surface parking lot or a vacant lot pursuant to Section 15.36.080;
    - c. Cumulative additions over a three (3) year period not involving the creation of a dwelling unit that are outside the existing building envelope and equal more than ten percent (10%) of the total fioor area or footprint on site, but do not exceed one thousand (1000) square

feet or one hundred percent (100%) of the total floor area or footprint on site, whichever is less;

- d. Secondary Units of more than five hundred (500) square feet in floor area, but not exceeding nine\_-hundred (900) square feet or fifty percent (50%) of the floor area of the primary dwelling unit whichever is less, pursuant to all regulations in Section-17-102-360 17.103.080;
- e. For commercial, civic, or industrial facilities and the non-residential portions of mixed-use development projects, changes to storefronts or street-fronting facades, such as: (i) replacement or construction of doors, windows; bulkheads and nonstructural wall infill, or (ii) restoration of documented historic fabric.
- 2. Fences, barriers, and similar freestanding walls.
  - a. For Residential Zones and Residential Facilities, any fence, barrier, or similar freestanding wall exceeding forty-two (42) inches in height in the front yard and street-side yards, but not exceeding six (6) feet in height pursuant to Section 17.108.140;
  - b. For Commercial Zones, Industrial Zones, and S-1, S-2, S-3, and S-15 Zones, any fence, barrier, or similar freestanding wall exceeding eight (8) feet in height within ten (10) feet of any abutting property in a residential zone, but not exceeding ten (10) feet in height pursuant to Section 17.108.140.
- 3. Signs.
  - a. New or modified Signs, excluding Signs requiring Regular Design Review, Conditional Use Permit or Variance, pursuant to the zoning regulations of Title 17 of the Oakland Planning Code; and Signs conforming to an approved Master Sign Program, pursuant to Section 17.104.070.
  - b. New or modified awnings or other similar facilities;
  - c. Color changes to Signs, awnings or other similar facilities;
  - d. Installation of flags or banners having any permanent structure within the public right of way, pursuant to the same regulations for sidewalk cafes in Section—17-102-335 <u>17.103.090B</u>.;
- C. Procedures for Consideration Small Project Design Review. The Director of City Planning may, at his or her discretion, consider an application for small project design review according to the following Three-Track process, or if additional consideration is required, determine that the proposal shall be reviewed according to the regular design review procedure in Section 17.136.040.
  - Track One Procedure Small Project Design Review Proposals Not Involving a Local Register Property; or an Upper-Story Addition requiring the Track Three review procedure pursuant to Subsection (C)(3):
    - a. The Director of City Planning, or his or her designee, shall determine whether the proposal meets the requirements for small project design review as set forth in this section.
    - b. Decision by the Director of City Planning. The Director, or his or her designee, may approve or disapprove a Track One proposal determined eligible for small project design review and may require such changes therein or impose such reasonable conditions of approval as are in his or her judgment necessary to ensure conformity to the applicable small project design review criteria in Section 17.136.035.
    - c. The decision by the Director, or his or her designee, shall be tinal immediately and not appealable.
  - 2. Track Two Procedure Small Project Design Review Proposals Involving a Local Register Property:

- a. The Director of City Planning, in concert with the City of Oakland's Historic Preservation staff, shall determine whether a proposed addition or alteration involving a Local Register Property will have a significant effect on the property's character-defining elements. "Character-defining elements" are those features of design, materials, workmanship, setting, location, and association that identify a property as representative of its period and contribute to its visual distinction or historical significance. Any proposed addition or alteration determined to have a significant effect on a Local Register Property's character-defining elements shall be reviewed instead according to the regular design review procedure in Section 17.136.040. Any proposed addition involving an upper-story addition of more than two hundred fifty (250) square feet in floor area or footprint to a One- or Two-Family Residential Facility or to any Building Facility in the HBX zones that is determined eligible for small project design review and to not have a significant effect on the property's character-defining elements, shall be reviewed according to the Track Three procedure in Section 17.136.030(C)(3).
- b. Decision by the Director of City Planning. The Director, or his or her designee, may approve or disapprove a Track Two proposal determined eligible for small project design review and may require such changes therein or impose such reasonable conditions of approval as are in his or her judgment necessary to ensure conformity to the applicable small project design review criteria in Section 17.136.035.
- c. The decision by the Director, or his or her designee, shall be tinal immediately and not appealable.
- 3. Track Three Procedure Small Project Design Review Proposals involving an Upper-Story Addition of More than Two Hundred Fifty (250) Square Feet in Floor Area or Footprint to a Oneor Two-Family Residential Facility or an over eight (8) foot increase in the height of any Building Facility in the HBX zones, not including allowed projections above the height limits listed in 17.108.030;
  - a. The Director of City Planning, or his or her designee, shall determine whether the proposal meets the requirements for small project design review as set forth in this section.
  - b. At the time of small project design review application, the owner of the affected property, or his or her authorized agent shall obtain from the City Planning Department a list of names and mailing addresses of all persons shown on the last available equalized assessment roll as owning the City of Oakland lot or lots adjacent to the project site and directly across the street abutting the project site; a notice poster to install on the project site; and a Notice to Neighboring Property Owners form which includes the project description and contact information.
  - c. Prior to the subject application being deemed complete, the applicant shall install the notice poster provided at the time of application at a location on the project site that is clearly visible from the street, alley, or private way providing access to the subject lot; and provide by certified mail or delivery to all persons shown on the last available equalized assessment roll as owning the City of Oakland lot or lots adjacent to the project site and directly across the street abutting the project site, a copy of the completed project notice form, as well as a set of reduced plans (consisting of at least a site plan and building elevations that show all proposed exterior work).
  - d. All required posting of the site and notification of adjacent and across the street property owners shall be completed by the project applicant not less than ten (10) days prior to the earliest date for final decision on the application. During the required noticing period, the Planning Department shall receive and consider comments from any interested party, as well as accept requests for a meeting with City Planning staff.
  - e. Decision by the Director of City Planning. Prior to final decision, City Planning staff shall hold a single meeting with interested parties whenever such a meeting request is received in writing by the Planning Department during the small project design review comment period. Following any such meeting with interested parties, the Director, or his or her

designee, may approve or disapprove a Track Three proposal determined eligible for small project design review and may require such changes therein or impose such reasonable conditions of approval as are in his or her judgment necessary to ensure conformity to the applicable small project design review criteria in Section 17.136.035.

f The decision by the Director, or his or her designee, shall be final immediately and not appealable.

### 17.136.040 - Regular design review.

- A. Applicability. "Regular design review" shall apply to proposals that require design review pursuant to the zoning regulations of Title 17 of the Oakland Planning Code, but do not qualify for a design review exemption as set forth in Section 17.136.025 or small project design review as set forth in Section 17.136.030. Projects requiring regular design review include, but are not limited to, the following types of work:
  - 1. Any proposal involving one or more of the facility, activity, building, structure, or development types that require design review pursuant to the zoning regulations of Title 17 of the Oakland Planning Code, but does not qualify for a design review exemption as set forth in Section 17.136.025, or small project design review as set forth in Section 17.136.030;
  - 2. Any construction, addition or alteration of structures requiring a conditional use permit or variance, pursuant to the zoning regulations of Title 17 of the Oakland Planning Code;
  - 3. New construction of one or two dwelling units, other than a secondary unit;
  - New construction of three or more dwelling units, or adding units to a property for a total of three or more dwelling units on site;
  - 5. New construction of principal facilities in the HBX zone;
  - The creation of any new HBX work/live unit or HBX live/work unit (see Sections 17.65.160 and 17.65.170). This requirement shall apply for both: a) conversions of existing facilities to contain either of these unit types, and b) the construction of new buildings that contain either of these unit types;
  - Cumulative additions over a three (3) year period not involving the creation of a dwelling unit that are outside the existing building envelope and exceed one thousand (1,000) square feet or one hundred percent (100%) of the total floor area or footprint on site, whichever is less;
  - 8. Exceptions to the parking accommodation requirements for one- and two-family. Residential Facilities in-Section-17-102-390- Section 17.116.075;
  - 9. New or modified Signs not qualifying for a design review exemption as set forth in Section 17.136.025 or small project design review as set forth in Section 17.136.030,
  - Proposals for new or modified Telecommunications Facilities, pursuant to Chapter 17.128, but excluding those alterations to existing Telecommunications Facilities listed as a Small Project in Section 17.136.030(B);-
  - 11. Demolition or removal of any structure, or portion thereof, where the replacement project requires Regular Design Review, Conditional Use Permit or Variance;
  - 12. Demolition or removal of any Designated Historic Property (DHP) or Potential Designated Historic Property (PDHP) pursuant to Section 17.136.075.
- B. Pre-Application Review—Regular Design Review. Prior to application for regular design review, any applicant or his or her representative seeking early project feedback may submit for a pre-application review of the proposal by a representative of the City Planning Department. For projects of a larger scale or involving a significant policy issue, the Director of City Planning may, at his or her discretion, request that an applicant or his or her representative submit for a pre-application review of the proposal. During a pre-application review, the City representative will provide information about

applicable design review criteria and pertinent procedures, including the opportunity for advice from outside design professionals. Where appropriate the City representative may also informally discuss possible design solutions, point out potential neighborhood concerns, and mention local organizations which the applicant is encouraged to contact before finalizing the proposal.

- C. Procedure for Consideration of Regular Design Review Proposals which Involve an Initial Decision by the Director of City Planning Decisions Not Ultimately Appealable to City Council.
  - Decision by the Director of City Planning. An application for regular design review that is not referred to the City Planning Commission for initial decision as specified in Section 17.136.040(D) shall be considered by the Director of City Planning.
  - 2. Notification Procedures. Notice shall be given by posting an enlarged notice at a location on the project site that is clearly visible from the street alley, or private way providing access to the subject lot. Notice shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the City within three\_-hundred (300) feet of the project site; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for decision on the application by the Director. During the required noticing period, the planning department shall receive and consider comments from any interested party.
  - 3. The Director or the applicant may seek the advice of outside design professionals. Any interested party must enter into the record any issues and/or evidence to the Director prior to the close of the written public comment period for his or her consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Director shall determine whether the proposal conforms to the applicable design review criteria, and may approve or disapprove the proposal or require such changes therein or impose such reasonable conditions of approval as are in his or her judgment necessary to ensure conformity to said criteria.
  - 4. Finality of Decision. A decision by the Director shall become final ten (10) calendar days after the date of initial decision unless appealed to the City Planning Commission or the Commission's Residential Appeals Committee in accordance with Section 17.136.080. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Director prior to the close of the written public comment period. In the event that the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal. Appeals considered by the City Planning Commission or the Commission's Residential Appeals Committee under the procedures specified In Section 17.136.080 shall be final immediately and are not ultimately appealable to the City Council.
- D. Procedure for Consideration of Regular Design Review Proposals which Involve an Initial Decision by the City Planning Commission Decisions Ultimately Appealable to City Council.
  - Decision by the City Planning Commission. The Director of City Planning may, at his or her discretion, refer an application for regular design review to the City Planning Commission for an initial decision rather than acting on it himself or herself. In this case, the application shall still be considered a minor permit, but shall be processed according to the procedure in this subsection. In these instances, any other minor permits associated with the application shall be considered concurrently by the Planning Commission, pursuant to Section <u>17.130.090</u>. <u>47-130-080</u>. However, if the project involves a major variance or major conditional use permit requires an Environmental Impact Report (EIR); or results in twenty-five thousand (25,000) square feet or more of new nonresidential floor area and is located in any zone other than the R-80, CBD-R, CBD-P (when not combined with the S-7 zone), CBD-C, CBD-X, S-2, or S-15 zones, the Director of City Planning shall refer the application to the City Planning Commission for an initial decision rather than acting on it himself or herselt
  - 2. Notification Procedures. Notice shall be given by posting an enlarged notice at a location on the project site that is clearly visible from the street, alley, or private way providing access to the

subject lot Notice shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the City within three hundred (300) feet of the project site; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for a hearing before the Commission. During the required noticing period, the planning department shall receive and consider comments from any interested party.

- 3. The Planning Commission may seek the advice of outside design professionals. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Commission for its consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Commission shall determine whether the proposal conforms to the applicable design review criteria, and may approve or disapprove the proposal or require such changes therein or impose such reasonable conditions of approval as are in his or her or its judgment necessary to ensure conformity to said criteria.
- 4. Finality of Decision. The initial decision of the Planning Commission shall become final ten (10) days after the date of decision unless appealed to the City Council in accordance with Section 17.136.090. <u>Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Commission prior to the close of the Commission's public hearing on the matter.</u> In the event that the last day of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal.
- E. Alternative Notification Procedures. If the conditions as set forth in Section 17.130.020 apply, alternative notification procedures discussed therein may replace or supplement the procedures set forth in subsections C and D of this section.

### 17.136.050 - Regular design review criteria.

- B. For Nonresidential Facilities and Signs.
  - 1. That the proposal will help achieve or maintain a group of facilities which are well related to one another and which, when taken together, will result in a well-composed design, with consideration given to site, landscape, bulk, height, arrangement texture, materials, colors, and appurtenances; the relation of these factors to other facilities in the vicinity; and the relation of the proposal to the total setting as seen from key points in the surrounding area. Only elements of design which have some significant relationship to outside appearance shall be considered, except as otherwise provided in Section 17.136.060;
  - 2. That the proposed design will be of a quality and character which harmonizes with, and serves to protect the value of, private and public investments in the area;
  - That the proposed design conforms in all significant respects with the Oakland General. Plan and with any applicable design review guidelines or criteria, district plan, or development control map which have been adopted by the Planning Commission or City Council.
- 17.136.055 Special regulations for historic properties in the central business district zones.
- A. The provisions of this Section shall only apply to proposals in the <u>Central Business District (CBD)</u> zones.
- B. Findings.
  - Any exterior alteration to a character-defining element of a Designated Historic Property (DHP) or Potentially Designated Historic Property (PDHP) that 1) does not match its exterior historical materials or appearance, and 2) is part of the existing building (not part of any proposed addition) shall be required to meet any applicable criteria in Chapter 17.136 and meet findings

(a) and (b), below. The determination of whether a project meets these findings requires consultation with Historic Preservation staff.

- a. Any replacements of exterior character-defining elements are required because repair is not feasible. "Character-defining elements" are those features of design, materials, workmanship, setting, location, and association that Identify a property as representative of its period and contribute to its visual distinction or historical significance; and
- b. Consultation with Historic Presenvation staff has determined that any replacement or repair that differs from the original feature is compatible with the character of the building, Area of Primary Importance (API) or Area of Secondary Importance (ASI), if applicable, and retains the character-defining appearance of the feature.
- 2. Approval of applications for projects in an API that require Regular Design Review approval may be granted only upon determination that the proposal conforms to any applicable criteria in Chapter 17.136 and to the following additional criteria:
  - Any proposed new construction is compatible with the existing API in terms of massing, siting, rhythm, composition, patterns of openings, quality of material, and intensity of detailing;
  - b. New street frontage has forms that reflect the widths and rhythm of the facades on the street and entrances that reflect the patterns on the street;
  - c. The proposal provides high visual interest that either reflects the level and quality of visual interest of the API contributors or otherwise enhances the visual interest of the API.
  - d. The proposal is consistent with the visual cohesiveness of the API. For the purpose of this finding, visual cohesiveness is the architectural character, the sum of all visual aspects, features, and materials that defines the API. A new structure contributes to the visual cohesiveness of a district if it relates to the design characteristics of a historic district while also conveying its own time. New construction may do so by drawing upon some basic building features, such as the way in which a building is located on its site, the manner in which it relates to the street, its basic mass, form, direction or orientation (horizontal vs. vertical), recesses and projections, quality of materials, patterns of openings and level of detailing. When some combination of these design variables are arranged in a new building to relate to those seen traditionally in the area, but integral to the design and character of the proposed new construction, visual cohesiveness results;
  - e. Where height is a character-defining element of the API there are height transitions to any neighboring contributing historic buildings. "Character-defining elements" are those features of design, materials, workmanship, setting, location, and association that identify a property as representative of its period and contribute to its visual distinction or historical significance. APIs with a character-defining height and their character-defining height level are designated on the zoning maps; and
  - f. For additions, the proposal meets either: 1) Secretary of Interior's standards for the treatment of historic resources; 2) the proposal will not adversely affect the character of the property or API; or, 3) upon the granting of a conditional use permit, (see Chapter 17.134 for the CUP procedure) and a hearing in front of the Landmarks Preservation Advisory Board for its recommendations, a project meets the additional findings in Subsection g., below.
  - g. For construction of new principal buildings:
    - i. The project will not cause the API to lose its status as an API;
    - ii. The proposal will result in a building or addition with exterior visual quality, craftsmanship, detailing, and high quality and durable materials that is at least equal to that of the API contributors; and

- iii. The proposal contains elements that relate to the character-defining height of the API, if any, through the use of a combination of upper story setbacks, window patterns, change of materials, prominent cornice lines, or other techniques. APIs with a character-defining height and their character-defining height level are designated on the zoning maps.
- 3. Approval of an application for a project that requires Regular Design Review Approval involving a DHP or PDHP outside of an API may be granted only upon determination that the proposal conforms to any applicable criteria in Chapter 17.136 and either meets each criteria (a), (b), and (c), or only (d), below:
  - Any proposed new construction is compatible with the existing district and/or building in terms of massing, siting, rhythm, composition, patterns of openings, quality of material, and intensity of detailing;
  - b. The proposal reflects the quality and visual interest of the building and/or ASI, or otherwise enhances the visual interest of the building or ASI.
  - . c. The proposal does not disqualify an ASI as an ASI; and
  - d. If a project does not meet either finding (a), (b), or (c), above, approval of applications for projects may still be granted, but only after a hearing in front of the Landmarks Preservation Advisory Board for its recommendations and determination that the proposal meets the following criteria: The proposal will result in a signature building within the neighborhood, City, or region based on qualities including, but not necessarily limited to, exterior visual quality, craftsmanship, detailing, and high quality and durable materials.
- C. Required Hearings in Front of the Landmarks Preservation Advisory Board (LPAB).
  - 1. Prior to project approval, the following projects require a hearing in front of the LPAB for its recommendations and/or advice to the decision making body:
    - a. Any construction of a new principal building in an API;
    - b. An addition to a API contributor when required by 17.136.055(B)(2)(f).
    - c. With the exception of additions that are not visible from a street or other public area, projects in an API that would result in a building taller than the character-defining height of the district, if any. Districts with a character-defining height and their character-defining height levels are designated on the zoning maps. An addition is considered "visible from a street or other public area" if it is located within the "critical design area," defined as the area within <u>forty (40)</u> feet of any street line, public alley, public path, park or other public area.
    - d. New construction or an addition to a building when required by Subsection 17.136.055 B.3.d.
    - e. Any proposal involving a Local Register Property that requires Regular Design Review approval.
- 17.136.060 Review by Landmarks Board in certain cases.
- A. Whenever an application is for regular design review in the S-7 zone, or on a designated landmark site, the Director of City Planning shall refer the proposal to the Landmarks Preservation Advisory Board for its recommendations.
- B. Whenever an application is for regular design review in the S-20 zone, and the Director of City Planning determines that a proposed addition or alteration will have a significant effect on the property's character-defining elements that are visible from a street or other public area, the Director may, at his or her discretion, refer the project to the Landmarks Preservation Advisory Board for its recommendations. "Character-defining elements" are those features of design, materials,

workmanship, setting, location, and association that identify a property as representative of its period and contribute to its visual distinction or historical significance. An addition or alteration is normally considered "visible from a street or other public area" if it affects a street face or public face of the facility or is otherwise located within the "critical design area," defined as the area within 40 feet of any street line, public alley, public path, park or other public area.

17.136.070 - Special regulations for designated landmarks.

A. Designation: In any zone, the City Council may designate as a landmark any facility, portion thereof, or group of facilities which has special character, interest, or value of any of the types referred to in Section 17.07.030P. The designating ordinance for each landmark shall include a description of the characteristics of the landmark which justify its designation and a clear description of the particular features that should be preserved. Each ordinance shall also include the location and boundaries of a landmark site, which shall be the lot, or other appropriate immediate setting, containing the landmark. Designation of each landmark and landmark site shall be pursuant to the rezoning and law change procedure in Chapter 17.144.

17.136.075 - Regulations for demolition or removal of designated historic properties and potentially designated historic properties.

- B. Regular Design Review approval for the demolition or removal of any Landmark, Heritage Property, structure rated "A" or "B" by the Oakland Cultural Heritage Survey, and structure on the City's Preservation Study List that are not in an S-7 or S-20 zone or Area of Primary Importance (API) as determined by the Oakland Cultural Heritage Survey may be granted only if the proposal conforms to the <u>regular\_general</u>-design review criteria, all other applicable design review criteria, and the following additional criteria:
  - The applicant demonstrates that. a) the existing property has no reasonable use or cannot generate a reasonable economic return and that the development replacing it will provide such use or generate such return, or b) the applicant demonstrates that the structure constitutes a hazard and is economically infeasible to rehabilitate on its present site. For this finding, a hazard constitutes a threat to health and safety that is not immediate;
  - 2. If a replacement facility is required by Section <u>17.136.075(A)</u>, tThe design quality of the replacement facility is equal <u>or</u>/superior to that of the existing facility; and
  - 3. It is economically, functionally architecturally, or structurally infeasible to incorporate the historic structure into the proposed development.
- E. For proposals that have received Design Review approval pursuant to this section, the issuance of a demolition permit for any structure or portion thereof may be postponed by the Director of City Planning for a period not to exceed one hundred twenty (120) days from the date of application for such permit. The Director may do so upon determination that the structure or portion thereof is listed as a Local Register Property, or is on a study list of facilities under serious study by the Landmarks Preservation Advisory Board, the City Planning Commission, or the Director, for possible landmark designation under Section 17.136.070 or for other appropriate action to preserve it. During the period of postponement the Board, the Commission, or the Director shall explore means for preserving or restoring the structure or portion thereof However, demolition may not be postponed under this section if, after notice to the Director of City Planning, the Building Services Department. the Housing Conservation Division, their respective appeals boards, or the City Council determines that immediate demolition is necessary to protect the public health or safety. Any determination made by the Director of City Planning under this section may be appealed pursuant to the administrative appeal procedure in Chapter 17.132.

## 17.136.080 - Appeal to Planning Commission—Regular design review.

Within ten (10) calendar days after the date of initial decision by the Director of City Planning on an application for regular design review under the procedure specified in Section 17.136.040(C), an appeal from said decision may be taken to the City Planning Commission by the applicant; the Landmarks Preservation Advisory Board, or any ether interested party. In the case of appeals involving one- or twounit Residential Facilities, the appeal shall be considered by the Commission's Residential Appeals Committee. In the event the last day of appeal falls on a weekend or holiday when City offices are closed, the next date offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the City Planning Department and shall be filed with such Department, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Director or wherein his or her decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to the Director of City Planning prior to the close of the written public comment period on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of such appeal, the Secretary of the City Planning Commission shall set the time for consideration thereot; Not less than seventeen (17) days prior to the date of the Commission's or Committee's consideration of the appeal, the Secretary shall give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented to the Director of City Planning prior to the close of the written public comment period for the underlying decision being appealed. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Commission or, if applicable, the Committee shall determine whether the proposal conforms to the applicable design review criteria, and may approve or disapprove the proposal or require such changes therein or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said criteria. The Commission or, if applicable, the Committee may seek the advice of outside design professionals. The decision of the Commission or, if applicable, the Committee on a proposal being considered under the procedure specified in Section 17.136.040(C) shall be final immediately and is not ultimately appealable to the City Council.

17.136.090 - Appeal to City Council—Regular design review.

Within ten (10) calendar days after the date of initial decision by the City Planning Commission on an application for regular design review under the procedure specified in Section 17.136.040(D), an appeal from said decision may be taken to the City Council by the applicant, the Landmarks Preservation Advisory Board, or any other interested party. In the event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal. No such appeal to the City Council is allowable under the procedure specified in Section 17.136.040(C). Such appeal shall be made on a form prescribed by the Commission and shall be filed with the City Clerk, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal: failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of the appeal, the Council shall set the date for consideration thereof. After setting the hearing date, the Council, prior to hearing the appeal, may refer the matter back to the Planning Commission for further consideration and advice. Appeals referred to the Planning Commission shall be considered by the Commission at its next available meeting. Any such referral shall be only for the purpose of issue clarification and advice. In all cases, the City Council shall

retain jurisdiction and, after receiving the advice of the Planning Commission, shall hold a hearing on and decide the appeal.

The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the proposal conforms to the applicable design review criteria, and may approve or disapprove the proposal or require such changes therein or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said criteria.

The decision of the City Council shall be made by resolution and shall be final. The-City-Council-shall vote-on-the-appeal-within-thirty-(30)-doys-after-its-first-hearing-of-the-appeal-If-the-Council-is-unable-to decide-the-appeal-ot-that-meeting, it-shall-appear-for-a-veto-on-each-regular-meeting-of-the-Council thereafter until-decided.

Chapter 17.138 - DEVELOPMENT AGREEMENT PROCEDURE Sections:

17.138.010 - Title, purposes, and applicability.

17.138.015 - Projects eligible and special regulations for projects with development agreements.

17.138.020 - Application.

17.138.030 - Planning Commission action.

17.138.040 - Council action.

17.138.050 - Criterion.

17.138.060 - Factors for consideration.

17.138.070 - Recordation.

**17.138.080 - A**dherence to development agreement, and amendment or cancellation by mutual consent.

17.138.090 - Periodic review.

17.138.100 - Development agreement related to other special zoning approval or subdivision.

### 17.138.010 - Title, purposes, and applicability.

The provisions of this chapter shall be known as the development agreement procedure. The purposes of these provisions are to prescribe the procedure for consideration of development agreements and, by encouraging appropriate projects, to strengthen the public planning process, encourage private participation in comprehensive planning, and reduce the economic costs of development This procedure shall apply to all proposals for development agreements.\_as-outherizod-by\_Section\_17,102,310,

(Prior planning code § 9350)

## 17.138.015 - Projects eligible and special regulations for projects with development agreements.

- A. Any person having a legal or equitable interest in the real property involved may, upon approval pursuant to the development agreement procedure in this chapter, enter into a development agreement with the City for any specific development project which involves either:
  - 1. Aa total of at least four (4) acres of land area; or
  - <u>2</u> <u>fFive hundred thousand (500,000) square feet of floor area; and is a project intended to be developed in stages; or which</u>
  - 3. iInvolves land sold or leased by the Redevelopment Agency or the successor to the Redevelopment Agency of the City, and is to be carried out by agreement with the Redevelopment Agency or the successor to the Redevelopment Agency.
- B. The development agreement shall not be approved unless the project has received, or simultaneously receives, whatever design review, major-conditional use permit, preliminary planned unit development plan approval, and/or major-variance it may otherwise require. For the duration of the particular agreement, and unless otherwise provided in the terms thereot there shall be a contractual guarantee that the project covered by the agreement may be pursued under the applicable procedural criteria, if any, and other zoning regulations, and plans or other documents referred to by any such criteria, as they existed when the agreement was approved and notwithstanding any subsequent changes in said zoning regulations to benefit or protect the City for

entering into the development agreement. The conditions may include, but are net limited to, supplemental restrictions on kinds of uses, floor-area ratio, or density; special conditions or criteria for required subsequent zoning approvals, if any; and requirements for the reservation, dedication, or improvement of land for public purposes or accessible to the public.

# 17.138.020 - Application.

Application for a development agreement shall be made by a person, or the authorized agent of a person, having a legal or equitable interest in the affected property. Application shall be made on a form prescribed by the City Planning Department and shall be filed with such Department. The application shall be accompanied by the fee prescribed in the fee schedule in Chapter 17.150 and by the proposed development agreement and any supporting material which, between them, shall include the following:

- A. An identification of the affected property and the proposed parties to the agreement;
- B. A description of the development project, indicating the proposed kinds of uses, floor-area ratio or density, and building height and size, and such additional information as may be required to allow the applicable criterion and factors to be applied to the proposal. Such information may include, but is not limited to, site and building plans, elevations, relationships to adjacent properties, and operational data. Where appropriate the description may distinguish between elements of the project which are proposed to be fixed under the agreement and those which may vary;
- C. An identification of any subsisting planned unit development permit or other special zoning approval which has already been obtained for the development project;
- D. The special conditions, if any, to be imposed pursuant to Soction 17-102-310 Section 17.138.015;
- E. The proposed duration of the agreement and timing of the development project
- F. A program for periodic review under Section 17.138.090.

(Prior planning code § 9351).

## 17.140.020 - Application.

- A. A preliminary development plan of the entire development showing streets, driveways, sidewalks and pedestrian ways, and off-street parking and loading areas; location and approximate dimensions of structures; utilization of structures, including activities and the number of living units; estimated population; reservations for public uses, including schools, parks, playgrounds, and other open spaces; major landscaping features; relevant operational data; and drawings and elevations cleariy establishing the scale, character, and relationship of buildings, streets, and open spaces. Such development plan shall include maps and information on the surrounding area within one hundred (100) feet of the development. All elements listed in this paragraph shall be characterized as existing or proposed, and sufficiently detailed to indicate intent and impact. In the case of a development intended to be constructed over a period of more than four years, the design and arrangement of those portions of the project to be constructed more than four (4) years in the future may be shown in generalized, schematic fashion;
- B. A tabulation of the land area to be devoted to various uses, a tabulation of gross floor area to be devoted to various uses, and a calculation of the average residential density per net acre and per net residential acre;
- C. A stage development demonstrating that the developer intends to commence construction within one year after the approval of the final development plan and will proceed diligently to completion;

D. If it is proposed that the final development plan will be submitted in stages, a schedule for submission thereof.

### 17.140.030 - Preliminary Planning Commission action.

An application for a planned unit development permit shall be considered by the City Planning Commission which shall hold a public hearing on the application. Notice of the hearing shall be given by posting an enlarged notice on the premises of the subject property. Notice of the hearing shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property within three hundred (300) feet of the property involved; provided, however, that failure to send notice to any such owner where his or her address is not shown in such records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing. If, however, the conditions as set forth in Section 17.130.020 apply, alternative notification procedures discussed therein may replace or supplement these procedures. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Commission for its consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Commission shall determine whether the proposal conforms to the permit criteria set forth in Section 17.140.080 and to the planned unit development regulations in Chapter 17.142, and may approve or disapprove the application and the accompanying preliminary development plan or require such changes therein or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said criteria and regulations. In so doing, the Commission may, in its discretion, authorize submission of the final development plan in stages corresponding to different units or elements of the development. It may do so only upon evidence assuring completion of the entire development in accordance with the preliminary development plan and stage development schedule. The determination of the Commission shall become final ten (10) calendar days after the date of decision unless appealed to the City Council in accordance with Section 17.140.070. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Commission prior to the close of the Commission's public hearing on the matter. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal.

# 17.140.060 - Final Planning Commission action.

Upon receipt of the final development plan, the City Planning Commission shall examine such plan and determine whether it conforms to all applicable criteria and standards and whether it conforms in all substantial respects to the previously approved preliminary development plan, or, in the case of the design and arrangement of those portions of the plan shown in generalized, schematic fashion, whether it conforms to applicable design review criteria. After receiving a final development plan which includes design and arrangement of portions of the project shown In generalized, schematic fashion on the preliminary development plan, the Commission shall hold a public hearing before taking action. Notice of the hearing shall be given in the same manner as set forth in Section 17.140.030. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Commission for its consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Commission may grant or deny a planned unit development permit or require such changes in the proposed development or impose such conditions of approval as are in its judgment necessary to ensure conformity to the applicable criteria and standards. In so doing, the Commission may permit the applicant to revise the plan and resubmit It as a final development plan within thirty (30) days. If the Commission does not grant such permission, the decision of the Commission shall become final ten (10) calendar days after the date of decision unless appealed to the City Council in accordance with Section 17.140.070. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Commission prior to the close of the Commission's public hearing on the matter. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal.

## 17.140.070 - Appeal to Council.

Within ten (10) calendar days after the date of a decision by the City Planning Commission on an application for approval of a preliminary or final development plan, or for modification or amendment of any such plan, an appeal from said decision may be taken to the City Council by the applicant, the permit holder, or any other interested party. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Commission and shall be filed with the City Clerk, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of such appeal, the Council shall set the date for consideration thereot; After setting the hearing date, the Council, prior to hearing the appeal, may refer the matter back to the Planning Commission for further consideration and advice. Appeals referred to the Planning Commission shall be considered by the Commission at its next available meeting. Any such referral shall be only for the purpose of issue clarification and advice. In all cases, the City Council shall retain jurisdiction and, after receiving the advice of the Planning Commission, shall hold a hearing on and decide the appeal. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the proposal conforms to the applicable criteria and standards, and may approve or disapprove the proposed development or require such changes therein or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said criteria and standards. The decision of the City Council shall be made by resolution and shall be final. The-City-Council-shall-vote-on-the-appeal-within thirty-(30)-days-after-its-first-hearing-of-the-appeal-If-the-Council-is-unable-to-decide-the-appeal-at-that meeting, it-shall-appear-for-a-vote-on-each-regular-mooting-of-the-Council-thereafter-until-decided,

# Chapter 17.142 - MINI-LOT AND PLANNED UNIT DEVELOPMENT REGULATIONS

Sections:

Article I - Title, Purposes, and Applicability.

17.142.00240 - Title and , purposes , and applicability-

17.142.004 - Applicability.

## Article II - Mini-lot Developments

<u>17.142.010 – Definition of mini-lot development.</u>

17.142.012 - Basic provisions for mini-lot developments.

17.142.014 - Zones in which requirements may be waived for a mini-iot development.

17.142.016 - Maximum size for which requirements may be waived for a mini-lot development.

# Article III – Planned Unit Developments

17.142.020 - Definition of planned unit development.

17.142.030 - Developments for which <u>planned unit development permit</u> approval Is required <u>or</u> <u>requested</u>.

17.142.040 - Ownership and division of land.

17.142.050 - Professional design.

17.142.060 - Dedication of public facilities and maintenance of open space.

17.142.070 - Performance bonds.

17.142.080 - Zones in which bonuses may be granted.

17.142.090 - Minimum size for which bonuses may be granted.

17.142.100 - Bonuses.

17.142.110 - Development standards.

#### Article I – Title, Purposes and Applicability

#### 17.142.00240 — Title and , purposes, and applicability.

The provisions of this chapter shall be known as the <u>mini\_lot\_and</u> planned unit development regulations. The purposes of these regulations are to

- A. -Encourage the comprehensive planning of the appropriate development of tracts of land; sufficiently large to allow comprehensive planning, and to
- <u>B.</u> Provide flexibility in the application of certain regulations in a manner consistent with the general purposes of the zoning regulations; and

and stable environments for living, shopping, or working. These-regulations-shall apply-to-all large, integrated-developments-for-which-a-planned-unit-development-permit-is-required-by Soction-17-142-030-

# 17.142.004 - Applicability.

These regulations shall apply to all:

- A. Mini-lot developments located on a single tract of land of less than sixty thousand (60,000) square feet, and containing lots which do not meet the minimum size or other requirements applying to individual lots in the zone where it is located; and
- B. Planned Unit Developments (PUDs) located on a single tract of land of sixty thousand (60,000) square feet or more, or on two or more tracts of land equaling sixty thousand (60,000) square feet or more in total which may be separated only by a street or other right-of-way.

## Article II – Mini-lot Developments

17.142.010 - Definition of mini-lot development.

A mini-lot development is a comprehensively designed development containing lots that do not meet the minimum size or other requirements applying to individual lots of less than sixty thousand (60,000) square feet in the zone where it is located.

17.142.012 - Basic provisions for mini-lot developments.

Subject to the provisions of this article, the maximum height and minimum yard, lot area, width, and frontage requirements otherwise applying to individual lots may be waived or modified within a mini-lot development, and floor area, parking, and other facilities may be located within said development without reference to lot lines, upon the granting of a conditional use permit pursuant to the conditional use permit procedure in Chapter 17.134 and upon determination;

- 1. That there is adequate provision for maintenance of the open space and other facilities within the development; and
- 2. That the total development meets all the requirements that would apply to it if it were a single lot.

17.142.014 - Zones in which requirements may be waived for a mini-lot development.

A conditional use permit pursuant to 17,142,012 may only be granted in the S-1 or S-2 zone, or in any residential or commercial zone other than RH or RD zones.

<u>17.142.016 – Maximum size for which requirements may be waived for a mini-lot development.</u> <u>Maximum Size for Which Requirements May Be Waived. A conditional use permit pursuant to</u> <u>17.142.012 of this section may be granted only if the total land area of the mini-lot development is less</u> than sixty-thousand (60,000) square feet.

# Article III - Planned Unit Developments

17.142.020 - Definition of planned unit development.

A "Pplanned Uunit Ddevelopment" (PUD) is a large, integrated development adhering to a comprehensive plan and located on a single tract of land of sixty thousand (60,000) square feet or more, or on two or more tracts of land equaling sixty thousand (60,000) square feet or more in total which may be separated only by a street or other right-of-way. In developments that are approved pursuant to the

<u>Planned Unit Development regulations in this Chapter, certain uses may be permitted in addition to those otherwise allowed in the underlying zone, and certain of the other regulations applying in said zone may be waived or modified.</u>

17.142.030 - Developments for which <u>planned unit development permit approval</u> is required <u>or</u> requested.

- - <u>1</u>A. Any planned unit development incorporating any of the bonuses set forth in Section 17.142.100.
    - 28. Any integrated development which is primarily designed for or occupied by Commercial Activities, which is located in any commercial zone, and which is developed under unified control, in accordance with a comprehensive plan, on a single tract with sixty thousand (60,000) square feet or more of land area, or on two or more tracts which total such area and which are separated only by a street or other right-of-way.
- B. Unless required by the Planning Director, other large, integrated developments involving the same minimum land area thresholds of a planned unit development as defined in Section 17.142.020, are permitted without such a permit. However, an applicant for such a development may request a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140, but shall be subject to all regulations generally applying in the zone in which they are located.

## 17.142.070 - Performance bonds.

The City Planning Commission or, on appeal, the City Council may, as a condition of approval of any development for which a permit is required by Section 17.142.030, require a cash bond or surety bond for the completion of all or specified parts of the development deemed to be essential to the achievement of the purposes set forth in Section 17.142.010. The bond shall be in a form approved by the City Attorney, in a sum of one hundred (400)-percent (100%) of the estimated cost of the work, and conditioned upon the faithful performance of the work specified within the time specified.

17.142.100 - Bonuses.

For planned unit developments qualifying under Sections 17.142.080 and 17.142.090, the following exceptions to otherwise applicable regulations may be permitted upon the granting of a planned unit development permit pursuant to the planned unit development procedure in Chapter 17.140:

- A. Additional Permitted Activities Where Increase in Overall Density or Floor-Area Ratio (FAR) is Proposed. Except in the RH and RD-1 zones, the following activities, as described in the use classifications in Chapter 17.10, may be permitted in a planned unit development incorporating an increase in overall density or floor-area ratio pursuant to subsection E of this section, in addition to the activities generally permitted in the zone where the development is located:
  - 1. Residential Activities:

Permanent

- 2. Civic Activities:
  - Limited Child-Care

Community Education

Oakland, California, Code of Ordinances

3. Commercial Activities, provided that such activities shall not occupy in the aggregate more than four (4)-percent (4%) of the total floor area in such development, provided that the maximum floor area devoted to such activities by any single establishment shall be three thousand hundred-(3,000) square feet, and further provided that such activities shall not be permitted at all in any such development which has an overall density in Residential Facilities of less than forty (40) living units per net residential acre (excluding streets and other rights-of-way):

General Food Sales

Full Service Restaurant

Limited Service Restaurant and Cafe

Fast-Food Restaurant

Convenience Market

Alcoholic Beverage Sales

Consumer Service (see Section 17.102.170 for special regulations relating to massage services and Section 17.102.450 for special regulations related to laundromats)

Medical Service

- B. Further Additional Permitted Activities Where No Increase in Overall Density or Floor-Area Ratio Is Proposed. Except in the RH and RD-1 zones, the following activities, as described in the use classifications, may be permitted in a planned unit development for which no increase in overall density or floor-area ratio is proposed pursuant to subsection E of this section, in addition to the activities listed in subsection A of this section and in addition to the activities generally permitted in the zone in which the development is located. The special limitations prescribed in subsection (A)(3) of this section with respect to location and amount of floor area devoted to Commercial Activities shall not apply in such a development.
  - 1. Residential Activities:

Semi-Transient

2. Civic Activities:

Nursing Home

Community Assembly

Recreational Assembly

Nonassembly Cultural

Administrative

Utility and Vehicular

3. Commercial Activities:

Mechanical or Electronic Games

General Retail Sales

Consumer Service- (see Section 17.102.170 for special regulations relating to massage services and Section 17.102.450 for special regulations related to laundromats)

Consultative and Financial Service

Consumer Cleaning and Repair Service

Consumer Dry Cleaning Plant

Group Assembly

Personal Instruction and Improvement and Small Scale Entertainment

Administrative

Business, Communication, and Media Service

Broadcasting and Recording Service

Research Service

General Wholesale Sales

Automobile and Other Light Vehicle Gas Station and Servicing

Automotive Fee Parking

Animal Care

Animal Boarding

4. Manufacturing Activities:

Custom

17.142.110 - Development standards.

The following regulations shall apply to all developments for which a permit is required by Section 17.142.030:

- A. Density and Floor-Area Ratio (FAR) Calculation. The maximum overall number of living units in Residential Facilities and the maximum overall floor-area ratio, if any, shall be based on the land area within the development, excluding the following:
  - 1. Publicly dedicated streets, freeways, alleys, and paths;
  - 2. When computing density for Residential Facilities in the RH, RD, RM, C-10, C-20, or C-60 zones, the following:
    - a. Land, other than public housing sites, which is publicly owned or reserved for public ownership,
    - b. Land which is specifically devoted to or intended for Nonresidential Facilities.
- C. Height in the RH-4 and RD-1 Zones. In the RH-4 and RD-1 zones, no building shall exceed fifty (50) feet in height except as would otherwise be allowed by Section 17.108.020(A) and except for the same projections as are allowed by Section 17.108.030.
- H. Other Regulations. Except as otherwise provided in Section 17.142.100 and in this section, and except as more restrictive regulations may be prescribed pursuant to Section 17.142.060 or otherwise as a condition of approval of a planned unit development permit pursuant to Section 17.142.030, the development shall be subject to the regulations generally applying in the zone in which it is located and the provisions of Section 17.108.080.

Developments Divided by Boundaries. Any development which is divided by a boundary between zones shall be subject as if it were a single lot to the provisions of subsections (B)(2), (3), and (4) of Section <u>17.102.070 17.154.060</u> with respect to calculation of required parking, loading, and usable open space; calculation of maximum number of living units or floor-area ratio; and distribution of the resulting number of living units or amount of floor area.

17.144.050 - Review by Landmarks Board in certain cases.

Whenever an application or proposal, other than one initiated by the Landmarks Preservation Advisory Board, is to rezone property to or from the S-7 zone or to establish, amend, or delete a designated landmark or landmark site, the City Planning Department shall promptiy refer the application or proposal to said Board for its recommendations. The City Planning Commission shall not act on the application or proposal until it has first received a report from the Board or until more than thirty (30) days have elapsed since the proposal was sent to the Board, whichever is sooner. However, the thirty (30) day period may be extended by agreement between the Commission or private party applicant as the case may be, and the Board.

#### 17.144.055 - Review by Parks and Recreation Advisory Commission in certain cases.

Whenever an application or proposal, other than one initiated by the Parks and Recreation Advisory Commission (PRAC), is made to rezone property to or from the OS zone or to establish, amend, or delete a park category designation, the City Planning Department shall promptly refer the application or proposal to the PRAC for its recommendations. The City Planning Commission shall not act on the proposal until it has heard a report from the PRAC or until more than thirty (30) days have elapsed since the proposal was sent to the PRAC, whichever is sooner. However, the thirty (30) day period may be extended by agreement between the City Planning Commission or private party applicant, as the case may be, and the PRAC.

## 17.144.090 - Council action.

Upon receipt of an appeal by a private party, or upon receipt of a recommendation from the City Planning Commission, the City Council shall set the date for consideration of the matter. After setting the hearing date, the Council, prior to hearing the appeal or recommendation, may refer the matter back to the Planning Commission for further consideration and advice. Appeals referred to the Planning Commission shall be considered by the Commission at its next available scheduled meeting. Any such referral shall be only for the purpose of issue clarification and advice. In all cases, the City Council shall retain jurisdiction and, after receiving the advice of the Planning Commission, shall hold a hearing on and decide the appeal. In the case of receipt of a recommendation from the City Planning Commission, the City Clerk shall notify the Secretary of the City Planning Commission of the date set for consideration thereof; and said Secretary shall give notice of the hearing by mail or delivery to all parties who have commented on the matter and to other interested parties as deemed appropriate. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing.

In the case of an appeal by a private party, the City Clerk shall notify the Secretary of the City Planning Commission of the receipt of the appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant the appellant in those cases where the applicant is not the appellant adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. Upon an appeal by a private party, or upon the receipt of a recommendation from the City Planning Commission, the Council may approve, modify, or reverse the decision or may approve, modify, or disapprove the Commission's recommendations, as the case may be. The decision of the City Council shall be made by resolution and shall be final. Tho-City-Council-shall-vote-on-tho-appeal-within-thirty-(30)-days-after-its-first-hearing-of-the appeal. If-the Council-is-unable-to-decide-the appeal-at-that-meeting,-it-shall-appear-for-a-vote-on-each regular-meeting-of-the-Council-thereafter-until-decided.

17.148.020 - Definition of major and minor variances.

A. Major Variance. A "major variance" is a variance which involves any of the following provisions:

- 1. Allowable activity types or facility types;
- 2. Maximum number of living units;
- 3. Minimum lot area, except in the situation mentioned in Section 17.106.010B;
- Maximum floor-area ratio, except for One-Family Dwellings, One-Family Dwellings with Secondary Unit and Two-Family Dwellings;
- 5. Maximum size of Commercial or Manufacturing establishments;
- Restriction on over-concentration of Residential Care, Service-Enriched Permanent Housing, Transitional Housing, and Emergency Shelter Residential Activities as set forth in Section <u>17-102-212\_17.103.010B\_-i</u>
- 7.—Any-variance-application-that-roquires-development-of-an-Environmental-Impact-Report;
- 8.—Any--variance-application-referred--by-the-Director--of-City-Planning-to--the--City-Planning Commission-for-initial-decision-pursuant-to-Section-17-148-040(B)(-1)-
- 17.148.040 Procedure for consideration.
- A. Major Variances.
  - In All Zones. An application for a major variance shall be considered by the City Planning 1. Commission which shall hold a public hearing on the application. Notice of the hearing shall be given by posting an enlarged notice on the premises of the subject property involved in the application. Notice of the hearing shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the City within three hundred (300) feet of the property involved; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Commission for its consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Commission shall determine whether the conditions required in Section 17.148.050 are present, and may grant or deny an application for a variance or require such changes in the proposed use or impose such reasonable conditions of approval as are in Its judgment necessary to promote the purposes of the zoning regulations. The decision of the Commission shall become final ten (10) calendar days after the date of decision unless appealed to the City Council in accordance with Section 17,148,070. Any party seeking to appeal the determination will be limited to Issues and/or evidence presented to the Commission prior to the close of the Commission's public hearing on the matter. In the event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal.
  - Alcoholic Beverage Sales Commercial Activities. In addition to following the provisions of subsection (A)(1) of this section, the City Planning Commission shall also determine whether the proposal conforms to the criteria for findings of "Public Convenience and Necessity" set forth in Section-17-102-210 17.103.030(B)(3).
- B. Minor Variances.
  - In All Zones. An application for a minor variance shall be considered by the Director of City Planning. However, the Director may, at his or her discretion, refer the application to the City Planning Commission rather than acting on it himself or herselt. In this case, the application

shall still be considered a minor permit, but shall be processed according to the procedure in subsection A of this section. In these instances, any other minor permits associated with the application shall be considered concurrently by the Planning Commission, pursuant to Section 17.130.090. At his or her discretion, an administrative hearing may be held. Notice shall be given by posting an enlarged notice on the premises of the subject property involved in the application; notice shall also be given by mail or delivery to all persons shown on the last available equalized assessment roll as owning real property in the City within three hundred (300) feet of the property involved; provided, however, that failure to send notice to any such owner where his or her address is not shown in said records shall not invalidate the affected proceedings. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing, if such is to be held, or, if not, for decision on the application by the Director. Any interested party must enter into the record any issues and/or evidence: (a) to the Director prior to the close of the written public comment period for his or her consideration, or (b) to the Commission while the hearing is open for its consideration, whichever is applicable; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Director shall determine whether the conditions required in Section 17.148.050 are present, and may grant or deny the application for a variance or require such changes in the proposed use or impose such reasonable conditions of approval as are in his or her judgment necessary to promote the purposes of the zoning regulations. The decision of the Director of City Planning shall become final ten (10) calendar days after the date of decision unless appealed to the City Planning Commission in accordance with Section 17.148.060. In those cases which are referred to the Commission by the Director, the decision of the Commission shall become final ten (10) days after the date of decision unless appealed to the City Council in accordance with Section 17.148.070. Any party seeking to appeal the determination will be limited to issues and/or evidence presented (a) to the Director prior to the close of the written public comment period, or (b) to the Commission prior to the close of the Commission's public hearing on the matter, whichever is applicable. In the event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal

# 17.148.050 - Findings required.

- A. With the exception of variances for adult entertainment activities or sign facilities, a variance may be granted only upon determination that all of the following conditions are present:
  - That strict compliance with the specified regulation would result in practical difficulty or unnecessary hardship inconsistent with the purposes of the zoning regulations, due to unique physical or topographic circumstances or conditions of design; or, as an alternative in the case of a minor variance, that such strict compliance would preclude an effective design solution improving livability, operational efficiency, or appearance.
  - 2. That strict compliance with the regulations would deprive the applicant of privileges enjoyed by owners of similarly zoned property, or, as an alternative in the case of a minor variance, that such strict compliance would preclude an effective design solution fulfilling the basic intent of the applicable regulation.
  - 3. That the variance, if granted, will not adversely affect the character, livability, or appropriate development of abutting properties or the surrounding area, and will not be detrimental to the public welfare or contrary to adopted plans or development policy.
  - 4. That the variance will not constitute a grant of special privilege inconsistent with limitations , imposed on similarly zoned properties or inconsistent with the purposes of the zoning regulations.
  - 5. That the elements of the proposal requiring the variance (e.g., elements such as buildings, walls, fences, driveways, garages and carports, etc.) conform with the regular design review criteria set forth in the design review procedure at Section 17.136.050.

- 6. That the proposal conforms in all significant respects with the Oakland General Plan and with any other applicable guidelines or criteria, district plan, or development control map which have been adopted by the Planning Commission or City Council.
- 7. For proposals involving one or two residential dwelling units on a lot: That if the variance would relax a regulation governing maximum height minimum yards, maximum lot coverage or maximum floor area ratio, the proposal also conforms with at least one of the following additional criteria:
  - a. The proposal when viewed in its entirety will not adversely impact abutting residences to the side, rear, or directly across the street with respect to solar access, view blockage and privacy to a degree greater than that which would be possible if the residence were built according to the applicable regulation and, for height variances, the proposal provides detailing, articulation or other design treatments that mitigate any bulk created by the additional height or
  - b. Over sixty (60)-percent (60%) of the lots in the immediate vicinity are already developed and the proposal does not exceed the corresponding as-built condition on these lots and, for height variances, the proposal provides detailing, articulation or other design treatments that mitigate any bulk created by the additional height. The immediate context shall consist of the five closest lots on each side of the project site plus the ten closest lots on the opposite side of the street (see illustration I-4b); however, the Director of City Planning may make an alternative determination of immediate context based on specific site conditions. Such determination shall be in writing and included as part of any decision on any variance.

# 17.148.060 - Appeal to Planning Commission---Minor variances.

Within ten (10) calendar days after the date of a decision by the Director of City Planning on an application for a minor variance, an appeal from said decision may be taken to the City Planning Commission by the applicant or any other interested party. In the case of appeals involving one- or twounit Residential Facilities, the appeal shall be considered by the Commission's Residential Appeals Committee. In the event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the City Planning Department and shall be filed with such Department along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Director or wherein his or her decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to the Director of City Planning prior to the close of the written public comment period on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of such appeal, the Secretary of the City Planning Commission shall set the date for consideration thereot. Not less than seventeen (17) days prior to the date of the Commission's or Committee's consideration of the appeal, the Secretary shall give written notice to; the applicant the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented to the Director of City Planning prior to the close of the written public comment period for the underlying decision being appealed. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Commission or, if applicable, the Committee shall determine whether the conditions required in Section 17.148.050 are present and may grant or deny an application for a variance or require such changes in the proposed use or impose such reasonable conditions of approval as are in its judgment necessary to carry out the purposes of the zoning regulations. The decision of the Commission or, if applicable, the Committee shall be final.

#### 17.148.070 - Appeal to Council—Major variances.

A. With the exceptions of appeals for adult entertainment activities or for signs, appeals to the City Council shall be governed by the following:

Within ten (10) calendar days after the date of a decision by the City Planning Commission on an application for a major variance, an appeal from said decision may be taken to the City Council by the applicant, the holder of the variance, or any other interested party. In the event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the City Planning Commission and shall be filed with the City Clerk, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of such appeal, the Council shall set the date for consideration thereof. After setting the hearing date, the Council, prior to hearing the appeal, may refer the matter back to the Planning Commission for further consideration and advice. Appeals referred to the Planning Commission shall be considered by the Commission at its next available meeting. Any such referral shall be only for the purpose of issue clarification and advice. In all cases, the City Council shall retain jurisdiction and, after receiving the advice of the Planning Commission, shall hold a hearing on and decide the appeal.

The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the conditions required by Section 17.148.050 are present, and may grant or deny an application for a variance or require such changes in the proposed use or impose such reasonable conditions of approval as are in its judgment necessary to carry out the purposes of the zoning regulations.

The decision of the City Council shall be made by resolution and shall be final. The-City-Council-shall vote-on-the-appeal-within-thirty-(30)-days-after-its-first-hearing-of-the-appeal-. If-the-Council-is-unable-to decide-tho-appeal-at-that-meeting, it-shall-appear-for-a-vote-on-each-regular-meeting-of-the-Council thereafter-until-docided.

B. Appeals to the City Council relating to adult entertainment activities or for signs shall be governed by the following:

Within ten (10) calendar days after the date of a decision by the City Planning Commission on an application for a major variance, an appeal from said decision may be taken to the City Council by the applicant; the holder of the variance, or any other interested party. In the event the last date of appeal falls on a weekend or holiday when City offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Commission and shail be filed with the City Clerk, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appealant from raising such issues during the appeal and/or in court. Upon receipt of the appeal, the

Council shall set the date for consideration thereot. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, given written notice to: the applicant, the appellant in those cases where the applicant is not the appellant, adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the Citv Planning Commission's public hearing on the item. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the conditions required by Section 17.148.050 are present, and shall grant an application for variance if it determines that all the said criteria are present or require such changes in the proposed use or impose such reasonable conditions of approval as are, in its judgment, necessary to ensure conformity to said criteria. The decision of the City Council shall be made by resolution and shall be final. The City Council shall vote on the appeal within thirty (30) days after its first hearing of the appeal and --If-the-Council-is-unablo-to-decide-the-appeal at-that-meeting-it-shall-appear-for a-vote-on-each-regulor-meeting-of-the-Council-thereafter-until-docided. In any-event, however, tho-City Council-must decide the appeal within sixty (60) days of the appeal being filed.

17.148.100 - Variance related to conditional use permit, regular design review, planned unit, development, or subdivision.

Whenever a variance is required for a proposal also requiring a conditional use permit, regular design review, or a planned unit development permit, application for the variance shall be included in the application for said conditional use permit, regular design review, or planned unit development permit, and shall be processed and considered as part of same. Whenever a variance is proposed within a proposed subdivision, the application for the variance may be submitted with the tentative map or tentative parcel map required by the Oakland Municipal Code, and may be processed and considered therewith. In either case, however, the reviewing officer or body shall, in considering such a variance, determine whether the conditions required in Section 17.148.050 are present,

# 17.152.210 - Fee schedule.

Fees and regulations pertaining to fees, including fees pertaining to complaints and appeals, shall be in accordance with the city's master fee schedule, provided that no city official or employee shall be required to pay a fee to file a complaint with the City Planning Department, There shall be no fee for lodging a revocation complaint with the city. To make an appeal, the property owner, permit holder and any individual or entity representing the interests of the project owner or permit holder shall pay one hundred (100)-percent (100%) of the appeal fee set by the city's master fee schedule. However, to make an appeal, the complainant, or anyone representing the complainant's interest shall pay fifty (50)-percent (50%) of the appeal fee established by the city's master fee schedule.

17.154.060 - Application of regulations to lots divided by zone boundaries.

Wherever-it-is-found,-pursuant-to-Section-17-154.050, that a-lot-is-divided-by a-boundary-between zones, the provisions-of-the-zoning-regulations-shall-apply-to-such-lot-as-prescribod-in-Section 17-102-070, The actual location-of-tho-zone-boundary-itself, however, shall not-be-changed-by-the provisions-of-Section-17-102-070, Wherever it is found, after applying the rules set forth in Section 17.154.050 for interpretation of zone boundaries, that any lot is divided by a boundary between zones, the provisions of the zoning regulations shall apply as follows to such lot.

- A. Application of All Regulations of One Zone to Existing Lot If Boundary Is Near Lot Line. If the lot was on the effective date of the zoning regulations, or of a subsequent rezoning or other amendment thereto resulting in division of the lot by a zone boundary, and the owner or developer of such lot, or of a portion or combination of such lot or lots, may at his or her option assume that all of the regulations applying in any zone covering fifty percent (50%) or more of the lot area apply to the entire lot or lots. However, this option shall not apply unless the entire lot or all such lots or parcel of land could be included in such zone by shifting the affected zone boundary by not more than thirty (30) feet, as measured perpendiculariy to said boundary at any point.
- B. Application of Regulations Where subsection A Is Inoperative. Wherever the provisions of subsection A of this section do not apply or the option provided therein is not exercised:
  - No activity type or facility type is allowed on any portion of the lot located in a zone where such type is not generally allowed, except for the accessory uses allowed by subsections (B)(2) and (B)(3) of this section.
  - 2. Accessory off-street parking and loading may be located on the lot without regard for zone boundaries; provided that no parking or loading shall be located on any portion of the lot located in a zone where the principal activity served is not generally allowed, except as such parking is specifically allowed by the applicable individual zone regulations subject to the conditions set forth in Section 17.116.075; and further provided that parking and loading shall be subject to a conditional use permit requirement or other special controls on any portion of the lot located in any zone where such controls generally apply to parking or loading. The total amount of required parking and loading shall be calculated separately on the basis of the amount of the served use and the requirements applying in each zone; provided that the minimum size for which any parking or loading is required shall be deemed to be exceeded if it is exceeded by the total of such use on the entire lot.
  - 3. Accessory landscaping, fences, screening or retaining walls, and usable open space may be located on the lot without regard for zone boundaries. The total amount of required usable open space shall be calculated separately on the basis of the number of living units, or amount of floor area, and the usable open space requirements in each zone; provided that where reference is made to the total number of living units on a lot, the number on the entire lot shall be considered.
  - 4. The maximum permitted or conditionally permitted number of living units or Floor-Area Ratio, if any, on the lot shall be calculated separately on the basis of the amount of lot area and the density ratio and floor-area ratio applying in each zone. The resulting maximum permitted or conditionally permitted total number of living units or amount of floor area may be distributed on the lot without regard for zone boundaries, except as otherwise provided in subsection (B)(1) of this section and except that the number of living units and amount of floor area within each zone shall not exceed the number or amount which would be allowed on the entire lot if it were completely within such zone.
  - 5. The minimum lot area, width, and frontage requirements of the zone which covers the greater or greatest portion of the lot area of the lot shall apply to the entire lot. If the lot area is divided equally between two or more zones, the owner or developer of the lot may

assume that the minimum lot area, width, and frontage requirements of either or any of such zones apply to the entire lot.

- 6. All regulations not covered above shall apply separately to the portion of the let within each zone, provided that where reference is made in such regulation to the total quantity of living units or other unit of measurement on a lot, the quantity on the entire lot shall be considered.
- C. Wherever a lot is divided by a boundary between height areas, the height area line may be moved up to thirty (30) feet in any direction upon the granting of Regular Design Review approval (see Chapter 17.136 for the Regular Design Review process) to accommodate the site plan of a proposed development project. In addition to the general Design Review Criteria contained in Chapter 17.136, the proposal must meeting the following additional criteria:
  - 1. The height area line adjustment creates a more successful site plan in terms of open space, parking, or building location; and
  - Appropriate height transitions are incorporated into the building design and site plan to adjacent lower density residential properties that either share a parcel line or are across the street from the proposal.

Chapter 17.155 - Special regulations applying to mining and guarrying extractive activities.

Sections:

17.155.010

<u>17.155.020</u>	_Definitions.
17.155.030	Incorporation by Reference.
17.155.040	_Scope.
17.155.050	Vested Rights.
17.155.060	Process.
17.155.070	Standards for Reclamation.
17.155.080	_Statement of Responsibility.
17.155.090	Findings for Approval.
17.155.100	Financial Assurances.
17.155.110	Interim Management Plans.
17.155.120	Annual Report Requirements.
17.155.130	Inspections.
17.155.140	Violations and Penalties.
17.155.150	_ Appeals.
17.155.160	Fees.

Purpose and Intent.

### 17.155.010 - Purpose and Intent.

The City of Oakland recognizes that historically, the extraction of minerals has benefited the economic well-being of the city and the needs of society and that the reclamation of mined lands is necessary to prevent or minimize adverse effects on the environment and to protect the public health and safety. The city also recognizes that surface mining within the city occurs in a diverse, established, urban environment, which presents unique health, safety and welfare issues and where geologic, topographic, climatic, biological, and other conditions are significantly different than in less urbanized areas. Therefore, reclamation operations and the applicable specifications, inspections, reporting, monitoring must be appropriate to the surrounding conditions.

The purpose and intent of this section is to regulate surface mining operations as required by California's Surface Mining and Reclamation Act of 1975 (Public Resources Code Sections 2710 et seq.), as amended, hereinafter referred to as "SMARA", Public Resources Code (PRC) Section 2207 (relating to annual reporting requirements), and State Mining and Geology Board regulations (hereinafter referred to as "State regulations") for surface mining and reclamation practice (California Code of Regulations fCCR], Title 14, Division 2, Chapter 8, Subchapter 1, Sections 3500 et seg.), to ensure that:

- (a) Reclamation activities eliminate hazards to public health and safety and restore mined lands to a standard that is safe, stable, and usable for development of reuses that will enhance the cemmunity;
- (b) Adverse environmental effects are prevented or minimized in accordance with CEQA and other applicable requirements;
- (c) Reclamation activities further adopted city goals, plans, policies, objectives and regulations, including, without limitation the city's General Plan;
- (d) Reclamation activities appropriately consider values relating to recreation, watershed, wildlife, range and forage, and aesthetic enjoyment.

# 17.155.020 - Definitions.

The definitions set forth in this section shall govern the construction of this chapter.

"Area of Regional Significance" means an area designated by the State Mining and Geology Board which is known to contain a deposit of minerals, the extraction of which is judged to be of prime importance in meeting future needs for minerals in a particular region of the State within which the minerals are located and which, if prematurely developed for alternate incompatible land uses, could result in the premature loss of minerals that are of more than local significance.

"Area of Statewide Significance" means an area designated by the board which is known to contain a deposit of minerals, the extraction of which is judged to be of prime importance in meeting future needs for minerals in the State and which, if prematurely developed for alternate incompatible land uses, could result in the permanent loss of minerals that are of more than local or regional significance.

"Approved Plan" means a land use and/or development plan and all conditions of approval and adopted mitigation measures, as approved by the city pursuant to Title Chapter-17 of the Oakland Municipal Code.

"Borrow Pits" mean excavations created by the surface mining of rock, unconsolidated geologic deposits or soil to provide material (borrow) for fill elsewhere.

"City" means City of Oakland.

"City Council" means City Council of the City of Oakland.

"Compatible Land Uses" means land uses inherently compatible with mining and/or that require a minimum public or private investment in structures, land improvements, and which may allow mining because of the relative economic value of the land and its improvements. Examples of such uses may include, but shall not be limited to, very low density residential, geographically extensive but low impact industrial, recreational, agricultural, silvicultural, grazing, and open space.

"General Plan" means the General Plan of the City of Oakland.

"Haul Road" means a road along which material is transported from the area of excavation to the processing plant or stock pile area of the surface mining operation.

"Idle" means surface mining operations curtailed for a period of one year or more, by more than ninety (90) percent of the operation's previous maximum annual mineral production, with the intent to resume those surface mining operations at a future date.

"Incompatible Land Uses" means land uses inherentiv incompatible with mining and/or that require public or private investment in structures, land improvements, and landscaping and that may prevent mining because of the greater economic value of the land and its improvements. Examples of such uses may include, but shall not be limited to, high density residential, low density residential with high unit value, public facilities, geographically limited but impact intensive industrial, and commercial.

"Mined Lands" mean the surface, subsurface, and ground water of an area in which surface mining operations will be, are being, or have been conducted, including private ways and roads appurtenant to any such area, land excavations, workings, mining waste, and areas in which structures, facilities, equipment, machines, tools or other materials or property which result from, or are used in, surface mining operations are located.

"Minerals" means any naturally occurring chemical element or compound, or groups of elements and compounds, formed from inorganic processes and organic substances, including, but not limited to, coal, peat, and bituminous rock, but excluding geothermal resources, natural gas, and petroleum. "Operator" means any person who is engaged in surface mining operations, or who contracts with others to conduct operations on his or her behalf except a person who is engaged in surface mining operations as an employee with wages as his or her sole compensation.

"Reclamation" means the combined process of land treatment that minimizes water degradation, air pollution, damage to aquatic or wildlife habitat, flooding, erosion, and other adverse effects from surface mining operations, including adverse surface effects incidental to underground mines, so that mined lands are reclaimed to a usable condition which is readily adaptable for alternate land uses and create no danger to public health or safety. The process may extend to affected lands surrounding mined lands, and may require backfilling, grading, resoiling, revegetation, soil compaction, stabilization, or other measures.

"Reclamation Plan" means a plan for reclamation of mined lands as specified by SMARA.

"Stream Bed Skimming" means excavation of sand and gravel from stream bed deposits above the mean summer water level or stream bottom, whichever is higher.

"Surface Mining Operations" means all, or any part of, the process involved in the mining of minerals on mined lands by removing overburden and mining directly from the mineral deposits, open-pit mining of minerals naturally exposed, mining by the auger method, dredging and guarrying, or surface work incident to an underground mine. Surface mining operations include, but are not limited to, in place distillation or retorting or leaching, the production and disposal of mining waste, prospecting and exploratory activities, borrow pitting, streambed skimming, and segregation and stockpiling of mined materials (and recovery of same).

"Use Permit" means a conditional use permit or other land use permit for mining activities.

#### 17.155.030 - Incorporation by Reference.

The provisions of SMARA (PRC § 2710 et seq.), PRC Section 2207, and State regulations CCR § 3500 et seg. as those provisions and regulations may be amended from time to time, are made a part this section by reference with the same force and effect as if the provisions therein were specifically and fully set out herein, excepting that when the provisions of this section are more restrictive than correlative State provisions, this section shall prevail.

#### <u>17.155.040 - Scope.</u>

Except as provided in this section, no person shall conduct surface mining operations unless a Reclamation Plan and financial assurances for reclamation have first been approved by the city. Any applicable exemption from this requirement does not automatically exempt or limit a project or activity from the application of other regulations, ordinances or policies of city, including but not limited to, the application of CEQA, the requirements of an Approved Plan or other permits, the payment of development impact fees, or the imposition of other dedications and exactions as may be permitted under the law. The provisions of this section shall apply to all lands within the city, public and private.

This section shall not apply to the following activities, subject to the above-referenced exceptions:

- (a) Excavations or grading conducted for farming or on-site construction or for the purpose of restoring land following a flood or natural disaster.
- (b) Onsite excavation and onsite earthmoving activities which are an integral and necessary part of an approved construction project that are undertaken to prepare a site for construction of structures, landscaping, or other land improvements, including the related excavation, grading, compaction, or the creation of fills, road cuts, and embankments, whether or not surplus materials are exported from the site, subject to all of the following conditions:
  - (1) All required permits for the construction, landscaping, or related land improvements have been approved by a public agency or agencies in accordance with applicable provisions of state law and locally adopted plans and ordinances, including, but not limited to, the California Environmental Quality Act ("CEQA", Public Resources Code, Division 13, § 21000 et seo.).

- (2) The city's approval and CEQA review (if applicable) of the construction project included the onsite excavation and onsite earthmoving activities.
- (3) The approved construction project is consistent with the General Plan and zoning of the site.
- (4) Surplus materials shall not be exported from the site unless and until actual construction work has commenced and shall cease if the city determines, in its discretion, that construction activities have terminated, have been indefinitely suspended, or are no longer being actively pursued.
- (c) Permitted operation of a plant site used for mineral processing, including associated onsite structures, equipment, machines, tools, or other materials, including the onsite stockpiling and onsite recovery of mined materials, subject to all of the following conditions:
  - (1) The plant site is located on lands designated for industrial or commercial uses in the city's general plan.
  - (2) The plant site is located on lands zoned industrial or commercial, or are contained within a zoning category intended exclusively for industrial activities by the city.
  - (3) None of the minerals being processed is being extracted onsite.
  - (4) All reclamation work has been completed pursuant to the approved Reclamation Plan for any mineral extraction activities that occurred onsite after January 1, 1976.
- (d) Prospecting for, or the extraction ot, minerals for commercial purposes and the removal of overburden in total amounts of less than one thousand (1,000) cubic yards in any one location of one acre or less.
- (e) Surface mining operations that are required by federal law in order to protect a mining claim, as specified in Public Resources Code section 2714(e).
- (f) Any other surface mining operations that the State Mining and Geology Board determines to be of an infrequent nature and which involve only minor surface disturbances.
- (g) The solar evaporation of sea water or bay water for the production of salt and related minerals.
- (h) Emergency excavations or grading conducted by the Department of Water Resources or the <u>Reclamation Board for the purpose of averting, alleviating, repairing, or restoring damage to</u> property due to imminent or recent floods, disasters or other emergencies.
- (i) Road construction and maintenance for timber or forest operations, as specified in Public Resources Code section 2714(i)(1); and
- (i) Excavation, grading, or other earthmoving activities in an oil or gas field, as specified in Public Resources Code section 2714(k).

#### 17.155.050 - Vested Rights.

No person who obtained a vested right to conduct surface mining operations prior to January 1, 1976, shall be required to secure a permit to mine, so long as the vested right continues and as long as no substantial changes have been made in the mining operation except in accordance with SMARA, State regulations, this section and any other applicable requirements. Where a person with vested rights has continued surface mining in the same area subsequent to January 1, 1976, he shall obtain city approval of a Reclamation Plan covering the mined lands disturbed by such subsequent surface mining. In those cases where an overlap exists (in the horizontal and/or vertical sense) between pre- and post-Act mining, the Reclamation Plan shall call for reclamation proportional to that disturbance caused by the mining after the effective date of the Act (January 1, 1976), as determined by the city to be necessary or appropriate to accommodate reuse of the proposed site according to city plans, policies, ordinances, and other applicable requirements.

All other requirements of State law, this section or an approved plan shall apply to vested mining operations.

# 17.155.060 - Process.

- (a) Applications under the requirement for an Approved Plan or Reclamation Plan for surface mining or land reclamation projects shall include, at a minimum, each of the elements required by SMARA (§ 2772-2773) and State regulations, and any other requirements determined, in the discretion of the Planning Director or designee, to be necessary or appropriate to facilitate an evaluation of the proposed Reclamation Plan.
- (b) Within thirty (30) days of the acceptance of a complete application for a Reclamation Plan or as a requirement of an Approved Plan for surface mining operations and/or a Reclamation Plan, the Planning Department shall notify the State Department of Conservation of the filing of the application(s). Whenever mining operations are proposed in the one hundred (100) year flood plain of any stream, as shown in Zone A of the Flood Insurance Rate Maps issued by the Federal Emergency Management Agency ("FEMA"), and within one mile, upstream or downstream, of any state highway bridge, the Planning Department shall also notify the State Department of Transportation ("Caltrans") that the application has been received.
- (c) The Planning Department shall process the application(s) in accordance with the California Environmental Quality Act (Public Resources Code Sections 21000 et seg.) and the City's environmental review guidelines.
- (d) Subsequent to the appropriate environmental review, the Planning Department shall prepare a staff report with recommendations for consideration by the City Planning Commission. The City Planning Commission shall hold at least one noticed public hearing on Use Permit and/or Reclamation Plan. Notice shall be given by mail or delivery to all persons shown on the last available equalized assessment role as owning real property in the city limits within three hundred feet (300 feet) of the property involved. All such notices shall be given not less than seventeen (17) days prior to the date set for the hearing. At the conclusion of such hearing or hearings, the Planning Commission shall recommend to the City Council that it should approve, approve with changes, or deny the subject Reclamation Plan and/or Use Permit.
- (e) The City Council shall hold at least one noticed public hearing on a Use Permit and/or Reclamation Plan. Notice shall be given by mail or delivery to all persons shown on the last available equalized assessment role as owning real property in the city limits within three hundred feet (300 feet) of the property involved. All such notice shall be given not less than seventeen (17) days prior to the date set for the hearing.
- (f) Prior to final approval of a Reclamation Plan, financial assurances (as provided in this Chapter), any amendments to the Reclamation Plan, existing financial assurances, or those financial assurances required as part of an Approved Plan, the City Council shall certify to the State Department of Conservation that the Reclamation Plan and/or financial assurance complies with the applicable requirements of State law, and submit the plan, assurance, or amendments to the State Department of Conservation for review.

Pursuant to PRC § 2774(d), the State Department of Consenvation shall be given thirty (30) days to review and comment on the Reclamation Plan and forty-five (45) days to review and comment on the financial assurance. The Planning Department shall evaluate written comments received, if any, from the State Department of Conservation during the comment periods. Staff shall prepare a response describing the disposition of the major issues raised by the State for the City Council's approval. In particular, when the Planning Department's position is at variance with the recommendations and objections raised in the State's comments, the response shall address, in detail, why specific comments and suggestions were not accepted. Copies of any written comments received and responses prepared by the Planning Department shall be promptly forwarded to the operator/applicant.

- (g) The City Council shall then take action to approve, conditionally approve, or deny Use Permit and/or Reclamation Plan, and to approve the financial assurances pursuant to PRC § 2770(d) or any other requirement of an Approved Plan.
- (h) The Planning Department shall forward a copy of each approved Use Permit for mining operations, an Approved Plan and/or approved Reclamation Plan, and a copy of the approved financial

assurances to the State Department of Conservation. By July 1 of each year, the Planning Department shall submit to the State Department of Conservation for each active or idle mining operation a copy of the Approved Plan, or Reclamation Plan amendments, as applicable, or a statement that there have been no changes during the previous year.

# 17.155.070 - Standards for Reclamation.

- (a) All Reclamation Plans shall comply with the provisions of SMARA (§ 2772 and § 2773) and State regulations (CCR § 3500-3505). Reclamation Plans approved after January 15, 1993, Reclamation Plans for proposed new mining operations, and any substantial amendments to previously approved Reclamation Plans, shall also comply with the requirements for reclamation performance standards (CCR § 3700-3713).
- (b) The city may impose additional performance standards as developed either in review of individual projects, as warranted, through the formulation and adoption of citywide performance standards or through an Approved Plan.
- (c) Reclamation activities shall be initiated at the earliest possible time en those portions of the mined lands that will not be subject to further disturbance. Interim reclamation may also be required for mined lands that have been disturbed and that may be disturbed again in future operations. Reclamation may be done on an annual basis, in stages compatible with continuing operations, or on completion of all excavation, removal, or fill, as approved by the city. Each phase of reclamation shall be specifically described in the Reclamation Plan and shall include (a) the beginning and expected ending dates for each phase; (b) all reclamation activities required; (c) criteria for measuring completion of specific reclamation activities; and (d) estimated costs for completion of each phase of reclamation.
- (d) The information, analysis and other specifications submitted as part of the Reclamation Plan shall demonstrate that the improvements and financial assurances are sufficient to reclaim the site in a condition that meets all applicable state and city standards, and that is appropriate for the proposed reuse of the site and consistent with the land use and other applicable policies of the General Plan.

# 17.155.080 - Statement of Responsibility.

The person submitting the Reclamation Plan shall sign a statement accepting responsibility for reclaiming the mined lands in accordance with the Reclamation Plan. Said statement shall be kept by the Planning Department in the mining operation's permanent record. Prior to sale or transfer of the operation, the new operator shall submit a signed statement of responsibility as well as evidence required to demonstrate the financial assurance requirement set forth in this section or the Planning Department record.

# 17.155.090 - Findings for Approval.

- (a) Approved Plans. In addition to any findings required by the Approved Plan or for surface mining operations, a finding shall be included that the project complies with the provisions of SMARA and State regulations.
- (b) Reclamation Plans. For Reclamation Plans, the following findings shall be required:
  - (1) That the Reclamation Plan complies with SMARA Sections 2772 and 2773, and any other applicable provisions;
  - (2) That the Reclamation Plan complies with applicable requirements of State regulations (CCR § 3500-3505, and § 3700-3713).
  - (3) That the Reclamation Plan and potential use of reclaimed land pursuant to the plan are consistent with this section, the city's General Plan and any applicable resource plan, element or an Approved Plan.
  - (4) That the Réclamation Plan has been reviewed pursuant to CEQA and the City's environmental review guidelines, and all significant adverse impacts from reclamation of the surface mining operations are mitigated to the maximum extent feasible.

- (5) That the land and/or resources such as water bodies to be reclaimed will be restored to a condition that is compatible with, and blends in with, the surrounding natural environment, topography, and other resources, or that suitable off-site development will compensate for related disturbance to resource values.
- (6) That the Reclamation Plan will restore the mined lands to a safe, stable and usable condition which is readily adaptable for alternative land uses consistent with the General Plan, and other city Approved Plans, policies, ordinances and regulations.
  - (7) That a written response to the State Department of Conservation has been prepared, describing the disposition of major issues raised by that department. Where the city's position is at variance with the recommendations and objections raised by the State Department of Conservation, said response shall address, in detail, why specific comments and suggestions were not accepted.

# 17.155.100 - Financial Assurances.

- (a) To ensure that reclamation will proceed in accordance with the approved Reclamation Plan, the city shall require as a condition of approval security which will be released upon satisfactory performance. The applicant may pose security in the form of a surety bond, trust fund, irrevocable letter of credit from an accredited financial institution, or other method acceptable to the city and the State Mining and Geology Board as specified in State regulations, and which the city determines are adequate to perform reclamation in accordance with the surface mining operation's approved Reclamation Plan and/or an Approved Plan. Financial assurances shall be made payable to city and the State Department of Conservation.
- (b) Financial assurances will be required to ensure compliance with elements of the Reclamation Plan, including but not limited to, revegetation and landscaping requirements, restoration of aquatic or wildlife habitat, restoration of water bodies and water quality, slope stability and erosion and drainage control, disposal of hazardous materials, and other measures, if determined necessary by the Planning Department to comply with the requirements of an Approved Plan.
- (c) Cost estimates for the financial assurance shall be submitted to the Planning Department for review and approval prior to the operator securing financial assurances. The Planning Director shall forward a copy of the cost estimates, together with any documentation received supporting the amount of the cost estimates, to the State Department of Conservation for review. If the State Department of Conservation does not comment within forty-five (45) days of receipt of these estimates, it shall be assumed that the cost estimates are adequate, unless the City has reason to determine that additional costs may be incurred. The Planning Director shall have the discretion to approve the financial assurance if it meets the requirements of this Chapter, SMARA, State regulations and any requirements of an Approved Plan.
- (d) The amount of the financial assurance shall be based upon the estimated costs of reclamation to a safe, stable and usable condition in accordance with an Approved Plan for the years or phases stipulated in the approved Reclamation Plan, including any maintenance of reclaimed areas as may be required, subject to adjustment for the actual amount required to reclaim lands disturbed by surface mining activities since January 1, 1976, and new lands to be disturbed in the upcoming year. Cost estimates should be prepared by a California registered Professional Engineer and/or other similarly licensed and qualified professionals retained by the operator and approved by the Planning Director. The estimated amount of the financial assurance shall be based on an analysis of physical activities necessary to implement the approved Reclamation Plan in accordance with an Approved Plan for the site, the unit costs for each of these activities, the number of units of each of these activities, and the actual administrative costs. Financial assurances to ensure Reclamation Plan implementation and compliance with revegetation, restoration of water bodies, restoration of aquatic or wildlife habitat, and any other applicable element of the approved Reclamation Plan shall be based upon cost estimates that include, but may not be limited to, labor, equipment, materials, mobilization of equipment, administration, monitoring, inspections and reasonable profit by a commercial operator other than the permittee. A contingency factor of ten percent (10%) shall be added to the cost of financial assurances.

- (e) In projecting the costs of financial assurances, it shall be assumed without prejudice or insinuation that the surface mining operation could be abandoned by the operator and, consequently, the city or State Department of Conservation may need to contract with a third party commercial company for reclamation of the site.
- (f) The financial assurances shall remain in effect for the duration of the surface mining operation and any additional period until reclamation is completed (including any maintenance required).
- (g) The amount of financial assurances required of a surface mining operation for any one year shall be adjusted annually to account for new lands disturbed by surface mining operations, infiation, and reclamation of lands accomplished in accordance with the approved Reclamation Plan. The financial assurances shall include estimates to cover reclamation for existing conditions and anticipated activities during the upcoming year, excepting that the permittee may not claim credit for reclamation scheduled for completion during the coming year.
- (h) Revisions to financial assurances shall be submitted to the Planning Director each year prior to the anniversary date for approval of the financial assurances. The financial assurance shall cover the cost of existing disturbance and anticipated activities for the next calendar year, including any required interim reclamation. If revisions to the financial assurances are not required, the operator shall explain, in writing, why revisions are not required.

# 17.155.110 - Interim Management Plans.

- (a) Within ninety (90) days of a surface mining operation becoming idle, the operator shall submit to the Planning Department a proposed Interim Management Plan (IMP). The proposed IMP shall fully comply with the requirements of SMARA, including but not limited to all Approved Plan conditions, and shall provide measures the operator will implement to maintain the site in a stable condition, taking into consideration public health and safety. The proposed IMP shall be submitted on forms provided by the Planning Department, and shall be processed as an amendment to the Reclamation Plan. IMPs shall not be considered a project for the purposes of environmental review.
- (b) Financial assurances for idle operations shall be maintained as though the operation were active, or as otherwise approved through the idle mine's IMP.
- (c) Upon receipt of a complete proposed IMP, the Planning Department shall forward the IMP to the State Department of Conservation for review. The IMP shall be submitted to the State Department of Conservation at least thirty (30) days prior to approval by the Planning Director.
- (d) Within sixty (60) days of receipt of the proposed IMP, or a longer period mutually agreed upon by the Planning Director and the operator, the Planning Director shall review and approve or deny the IMP in accordance with this Chapter. The operator shall have thirty (30) days, or a longer period mutually agreed upon by the operator and the Planning Director, to submit a revised IMP. The Planning Director shall approve or deny the revised IMP within sixty (60) days of receipt. If the Planning Director denies the revised IMP, the operator may appeal that action to the City Council. The decision of the City Council shall be final.
- (e) The IMP may remain in effect for a period not to exceed five (5) years, at which time the City Council may renew the IMP for another period not to exceed five years, or require the surface mining operator to commence reclamation in accordance with its approved Reclamation Plan.

#### 17.155.120 - Annual Report Requirements.

Surface mining operators shall forward an annual surface mining report to the State Department of Conservation and to the City Planning Department on a date established by the State Department of Conservation, on forms furnished by the State Mining and Geology Board. New mining operations shall file an initial surface mining report and any applicable filing fees with the State Department of Conservation within thirty (30) days of permit approval, or before commencement of operations, whichever is sooner. Any applicable fees, together with a copy of the annual inspection report, shall be forwarded to the State Department of Conservation at the time of filing the annual surface mining report.

17.155.130 - Inspections.

The Planning Director, through the Building Department Inspection Services Division or other agency or other designee, shall arrange for inspection of a surface mining operation within six months of receipt of the Annual Report required in Section 12, to determine whether the surface mining operation is in compliance with applicable requirements, including, without limitation, the Approved Plan, Reclamation Plan, approved financial assurances, and State regulations. In no event shall less than one inspection be conducted in any calendar year. Said inspections may be made by a state-registered geologist, stateregistered civil engineer, state-licensed landscape architect, or state-registered forester, who is experienced in land reclamation and who has not been employed by the mining operation in any capacity during the previous twelve (12) months, or other qualified specialists, as selected by the Planning Director. All inspections shall be conducted using a form approved and provided by the State Mining and Geology Board.

The Planning Department shall notify the State Department of Conservation within thirty (30) days of completion of the inspection that said inspection has been conducted, and shall forward a copy of said inspection notice and any supporting documentation to the mining operator. The operator shall be solely responsible for all costs of inspections required by the city in furtherance of this section in accordance with the city master fee schedule or other applicable fee agreements or requirements.

## 17.155.140 - Violations and Penalties.

If the Planning Director, through the Building Department Inspection Services Division or other designee, based upon an annual inspection or otherwise confirmed by an inspection of the mining operation, determines that a surface mining operation is not in compliance with this section, the Approved Plan, the Reclamation Plan or other applicable requirements, the city shall follow the procedures set forth in Public Resources Code, Sections 2774.1 and 2774.2 concerning violations and penalties.

#### 17.155.150 - Appeals.

A decision by the City Council to either approve or deny a Reclamation Plan pursuant to this section shall be considered a final agency action.

# 17.155.160 - Fees.

The city shall establish such fees as it deems necessary to cover the reasonable costs incurred in implementing this section and the State regulations, including but not limited to, processing of applications, annual reports, inspections, monitoring, enforcement and compliance. These fees may be set forth in the city master fee schedule; however, failure to include such fees in the master fee schedule shall not limit the city's ability to impose fees it determines are necessary or desirable to fulfill the purposes of this section. State regulations and other applicable requirements. Such fees shall be paid by the operator, as required by the city, at the time of filing of the Reclamation Plan application, as a part of a fee agreement through an Approved Plan or at such other times as are determined by the city to be appropriate in order to ensure that all reasonable costs of implementing this section are borne by the mining operator.

ł

# 17.156.070 - Definitions.

"Full-Service Restaurant' means a place which is regulariy and in a bona fide manner used and kept open for the serving of at least lunch and dinner to guests for compensation and which has suitable kitchen facilities connected therewith, containing conveniences for cooking an assortment of foods which may be required for such meals. The sale or service of sandwiches (whether prepared in a kitchen or made elsewhere and heated up on the premises) or snack foods shall not constitute a full-service restaurant. To be considered a Full Service Restaurant under the Deemed Approved Program, the establishment must meet the following criteria:

- A "full service restaurant" shall serve "meals" to guests at all times the establishment is open for business. An establishment shall not be considered a "full-service restaurant" if it served alcohol without "meal" service being provided with the exception that alcohol sales to restaurant patrons may continue for up to two hours after meal service has ceased to allow guests to comfortably complete their meals.
- 2. There shall be a real offer or holding out to sell "meals." Premises shall make an offer or holding out of sales of "meals" to the public by maintaining and displaying a printed menu and/or a menu board. A two-thirds majority of the items offered on the menu shall be available at any given time the establishment is open. The mere offering of "meals" without actual sales shall not be deemed sufficient
- 3. The "offer" of "meals" is not adequate to meet the above criteria. A "full service restaurant" shall make actual and substantial sales of "meals" to guests for compensation. Substantial sales shall mean that no less than sixty (60)-percent (60%) of total revenue shall be generated from food service and no more than forty (40)-percent (40%) of revenue from the sales of alcohol.
- 4. "Meals" means the usual assortment of foods commonly ordered at various times of the day for the cuisine served. The service of snack foods and/or appetizers alone shall not be deemed compliance with this requirement "Meals" shall be prepared on the premises. Heating of food prepared elsewhere shall not constitute a meal for the purposes of this policy.
- 5. Premises shall be equipped for meal service and maintained in good faith. Premises must possess and maintain appliances for the cooking of a variety of foods such as stoves, ovens, broilers, or other devices, as well as pots, pans, or containers that can be used for cooking. Premises shall possess the necessary utensils, table service, and condiment dispensers with which to serve "meals" to the public.
- 6. A full service restaurant shall comply with all local health department standards.
- 7. A full service restaurant may have a separate lounge or bar area provided that the restaurant and bar/lounge area operate as a single entity. The physical layout entry location(s), spatial connection between the areas, and operational characteristics, among other factors, shall be used to determine compliance. Any bar/lounge area cannot remain open when the dining area is closed. However, the dining area may be open while the bar/lounge area is closed.
- To the extent that <u>State Department of Alcoholic Beverage Control (ABC)</u> regulations do not conflict with the above criteria, a full service restaurant shall comply with all-State-Department-of Alcoholic-Beverage-Control-<u>ABC</u> regulations related to "Bona fide public eating place, meals."

"Restricted street" means that area applied to a depth of <u>two hundred (200)</u> feet on each side of and including the following streets and portions of streets, as measured perpendicularly from the right-of-way line thereot <u>E. 14</u>Ith-Streot<u>International Boulevard</u>; Foothill Boulevard; MacArthur Boulevard and West MacArthur Boulevard; that portion of San Pablo Avenue lying between Highway 1-980 and 1-580; that portion of Edes Avenue lying between Clara Street and Bergedo Drive. 17.156.140 - Procedure for consideration of violations to performance standards.

Upon receiving a complaint from the public, Police Department, or any other interested party that a Deemed Approved Activity is in violation of the performance standards at Section 17.156.090, and once it is determined by the city that violations appear to be occurring, then the Deemed Approved Status of the Deemed Approved Activity in question shall be reviewed by the Administrative Hearing Officer at a public hearing. Notification of the public hearing shall be in accordance with Section 17.156.180.

The purpose of the public hearing is to receive testimony on whether the operating methods of the Deemed Approved Activity are causing undue negative impacts in the surrounding area. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Administrative Hearing Officer for his or her consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. At the public hearing, the Administrative Hearing Officer shall determine whether the Deemed Approved Activity conforms to the Deemed Approved Performance Standards set forth in Section 17,156,090 and to any other applicable criteria, and may continue the Deemed Approved Status for the activity in question or require such changes or impose such reasonable Conditions of Approval as are in the judgment of the Administrative Hearing Officer necessary to ensure conformity to said criteria and such conditions shall be based on the evidence before the Officer. The decision of the Officer shall be based upon information compiled by staff and testimony from the business owner and all other interested parties. New conditions of approval shall be made a part of the Deemed Approved Status and the Deemed Approved Activity shall be required to comply with these conditions. The determination of the Officer shall become final ten (10) calendar days after the date of decision unless appealed to the City Planning Commission in accordance with Section 17.156.160. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Administrative Hearing Officer prior to the close of the Administrative Hearing Officer's public hearing on the matter.

17.156.150 - Procedure for consideration of violations to conditions of approval.

In the event of a violation of any of the provisions set forth in Sections 17.156.010 through 17.156.140 of these regulations, or upon evidence that there has been a failure to comply with any prescribed condition of approval, the Officer may hold a public hearing. Notification of the public hearing shall be in accordance with Section 17.156.180.

The purpose of this public hearing is to receive testimony and determine whether violations to any conditions of approval attached to the site have occurred. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Administrative Hearing Officer for his or her consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The Officer may add to or amend the existing conditions of approval based upon the evidence presented; or alternatively may revoke the Deemed Approved Activity's Deemed Approved Status. The determination of the Administrative Hearing Officer shall become final ten (10) calendar days after the date of decision unless appealed to the City Planning Commission in accordance with Section 17.156.160. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Administrative Hearing Officer prior to the close of the Administrative Hearing Officer's public hearing on the matter. The decision of the Planning Commission shall be final unless appealed to the City Council in accordance with Section 17.156.170.

17.156.160 - Appeal to Planning Commission.

Within ten (10) calendar days after imposition of conditions of approval on a Deemed Approved Activity or the revocation of Deemed Approved Status, an appeal may be taken to the City Planning Commission by the Deemed Approved Activity owner or any other interested party. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the city. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Officer or wherein its decision is not supported by the evidence in the record. The appeal itself

must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to the Administrative Hearing Officer prior to the close of the public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Tho-appeal shall be accompanied by such information-as-may-be required to facilitate review. Upon receipt of the appeal and the required appeal fee in accordance with Section 17.156.190, the Secretary to the Planning Commission shall set the date for consideration thereof. The Administrative Hearing Officer shall, not less than seventeen (17) days prior thereto, give written notice to: the applicant; the appellant in those cases where the applicant is not the appellant, the adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as appropriate, of the date and place of the hearing on the appeal.

During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented to the Administrative Hearing Officer prior to the close of the public hearing on the item and raised in the appeal itself. The appellant shall not be permitted to present any other evidence (written, oral, or othenwise) during the appeal process. In considering the appeal, the Planning Commission shall determine whether the established use conforms to the applicable Deemed Approved performance standards and/or conditions of approval, and may continue or revoke a Deemed Approved Status; or require such changes in the existing use or impose such reasonable conditions of approval as are, in its judgment, necessary to ensure conformity to said performance standards.

The Planning-Commission-shall-vote-on-the appeal-within-thirty-(30)-days-after-its-first-hearing-of-the appeal-lf-tho-Commission-is-unable-to-decide-the-appeal-ot-that-meeting, it-shall-appear-for-a-vote-on each-regular-meeting-of-the-Commission-thoreaftor-until-decided. The decision of the Planning Commission on the appeal to the conditions of approval imposed by the Administrative Hearing Officer shall be final.

# 17.156.170 - Appeal on the revocation of a Deemed Approved Status to the City Council.

Within ten (10) calendar days after the date of a decision by the City Planning Commission to revoke a Deemed Approved Status, an appeal from said decision may be taken to the City Council by any interested party. In the event the last date of appeal falls on a weekend or holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the Planning Commission and shall be filed with the City Clerk, along with the appropriate fees required by the City's Master Fee Schedule. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of the appeal and an appeal fee in accordance with Section 17.156.190, the Council shall set the date for consideration thereof. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and of the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the owner of the Deemed Approved Activity; the property owner; adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties, other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as the Secretary deems appropriate, of the time, date and place of the hearing on the appeal. During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item and raised in the appeal itself. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the Deemed Approved Activity conforms to the applicable Deemed Approved performance standards, and may approve or disapprove the revocation or require such changes therein or impose such reasonable conditions of approval as are in its judgment necessary to ensure conformity to said standards.

The decision of the City Council shall be made by resolution and shall be final. The City-Council shall vote on the appeal within thirty-(30) days after its first hearing of the appeal. If the Council is unable to decide the appeal at that meeting, it shall appear for a vote on each regular mooting of the Council thereafter until docided.

17.157.110 - Procedure for consideration of violations to performance standards.

As a result of an annual or bi-annual inspection pursuant to OMC Section 8.030.60B or upon receiving a complaint from the public, Police Department, or any other interested party that a Deemed Approved Hotel Activity is in violation of the performance standards at Section 17.157.060, and once it is determined by the city that violations appear to be occurring, then the Deemed Approved Status of the Deemed Approved Hotel Activity in question shall be reviewed by the Administrative Hearing Officer at a public hearing. Notification of the public hearing shall be in accordance with Section 17.157.150.

The purpose of the public hearing is to receive testimony on whether the operating methods of the Deemed Approved Hotel Activity is in violation of the performance standards at Section 17.157.060, are causing undue negative impacts in the surrounding area, and/or whether the property is not being maintained in a manner to be habitable by quests or residents. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Administrative Hearing Officer for his or her consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. At the public hearing, the Administrative Hearing Officer shall determine whether the Deemed Approved Activity conforms to the Deemed Approved Performance Standards set forth in Section 17.157.060 and to any other applicable criteria, and may continue the Deemed Approved Status for the activity in question or require such changes or impose such reasonable Conditions of Approval as are in the judgment of the Administrative Hearing Officer necessary to ensure conformity with said criteria and such conditions shall be based on the evidence before the Officer. The decision of the Officer shall be based upon information compiled by staff and testimony from the business owner and all other interested parties. New conditions of approval shall be made a part of the Deemed Approved Status and the Deemed Approved Hotel Activity shall be required to comply with these conditions. The determination of the Officer shall become final ten (10) calendar days after the date of decision unless appealed to the City Planning Commission in accordance with Section 17.157.130. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Administrative Hearing Officer prior to the close of the Administrative Hearing Officer's public hearing on the matter.

17.157.120 - Procedure for consideration of violations of conditions of approval.

In the event of a violation of any of the provisions set forth in Sections 17.157.010 through 17.157.110 of these regulations, or upon evidence that there has been a failure to comply with any prescribed condition of approval, the Officer may held a public hearing. Notification of the public hearing shall be in accordance with Section 17.157.150.

The purpose of this public hearing is to receive testimony and determine whether violations to any conditions of approval attached to the site have occurred. While the hearing is open, any interested party must enter into the record any issues and/or evidence to the Administrative Hearing Officer for his or her consideration; failure to do so will preclude the party from raising such issues during the appeal hearing and/or in court. The officer may add to or amend the existing conditions of approval based upon the evidence presented; or alternatively may revoke the Deemed Approved Hotel Activity's Deemed Approved Status. The determination of the Administrative Hearing Officer shall become final ten (10) calendar days after the date of decision unless appealed to the City Planning Commission in accordance with Section 17.157.130. Any party seeking to appeal the determination will be limited to issues and/or evidence presented to the Administrative Hearing Officer prior to the close of the Administrative Hearing Officer's public hearing on the matter. The decision of the City Planning Commission shall be final unless appealed to the City Council in accordance with Section 17.157.140.

17.157.130 - Appeal to City Planning Commission.

Within ten (10) calendar days after imposition of conditions of approval on a Deemed Approved Hotel Activity or the revocation of Deemed Approved Status, an appeal may be taken to the City Planning Commission by the Deemed Approved Activity owner or any other interested party. In the event the last date of appeal falls on a weekend or a holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the city. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Officer or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to the Administrative Hearing Officer prior to the close of the public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Tho-appeal-shall-bo-accompanied-by-such information-as-may-be-required-to-facilitate-roviow-Upon receipt of the appeal and the required appeal fee in accordance with Section 17.157.160 the Secretary of the City Planning Commission shall set a date for consideration thereof. The Secretary of the City Planning Commission shall, not less than seventeen (17) days prior thereto, give written notice to: the owner of the Deemed Approved Hotel Activity; the property owner; the appellant in those cases where the appellant is not the owner; the adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as appropriate, of the date and place of the hearing on the appeal.

During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented to the Administrative Hearing Officer prior to the close of the public hearing on the item and raised in the appeal itself. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the City Planning Commission shall determine whether the Deemed Approved Hotel Activity conforms to the applicable Deemed Approved performance standards and/or conditions of approval, and may continue or revoke a Deemed Approved Status; or require such changes in the existing use or impose such reasonable conditions of approval as are, in its judgment, necessary to ensure conformity to said performance standards.

The-City-Planning-Commission-shall-vote-on-tho-appeal-within-thirty-(30)-days-after-its-first-hearing-of the appeal. If the Commission-is unable-to-decide-on-tho-appeal-at-that-meeting, it-shall-appear-for-a-vote on-each-regular-meeting-of-the-Commission-thereafter-until-decided...The decision of the City Planning Commission on the appeal to the conditions of approval imposed by the Administrative Hearing Officer shall be final.

17.157.140 - Appeal on the revocation of a Deemed Approved Status to the City Council.

Within ten (10) calendar days after the date of a decision by the City Planning Commission to revoke a Deemed Approved Status, an appeal from said decision may be taken to the city Council by any interested party. In the event the last date of appeal falls on a weekend or a holiday when city offices are closed, the next date such offices are open for business shall be the last date of appeal. Such appeal shall be made on a form prescribed by the City Planning Commission and shall be filed with the City Clerk. The appeal shall state specifically wherein it is claimed there was an error or abuse of discretion by the Commission or wherein its decision is not supported by the evidence in the record. The appeal itself must raise each and every issue that is contested, along with all the arguments and evidence in the record, previously presented to City Planning Commission prior to the close of its public hearing on the item, which supports the basis of the appeal; failure to do so will preclude the appellant from raising such issues during the appeal and/or in court. Upon receipt of the appeal and an appeal fee in accordance with Section 17,157,160, the Council shall set the date for consideration thereof. The City Clerk shall notify the Secretary of the City Planning Commission of the receipt of said appeal and the date set for consideration thereof; and said Secretary shall, not less than seventeen (17) days prior thereto, give written notice to: the owner of the Deemed Approved Hotel Activity; the property owner; the appellant in those cases where the appellant is not the owner; the adverse party or parties, or to the attorney, spokesperson, or representative of such party or parties; other interested groups and neighborhood associations who have requested notification; and to similar groups and individuals as appropriate, of the date and place of the hearing on the appeal.

During the hearing on the appeal, the appellant will be limited to issues and/or evidence presented prior to the close of the City Planning Commission's public hearing on the item and raised in the appeal itself. The appellant shall not be permitted to present any other evidence (written, oral, or otherwise) during the appeal process. In considering the appeal, the Council shall determine whether the Deemed Approved Hofel Activity conforms to the applicable Deemed Approved performance standards and/or conditions of approval, and may approve or disapprove the revocation of the Deemed Approved Status; or require such changes to the existing use or impose such reasonable conditions of approval as are, in its judgment, necessary to ensure conformity to said performance standards.

The decision of the City Council shall be made by resolution and shall be final. Tho-City-Council-shall vote-on-the-appeal-within-thirty-(30) days-after-its-first-hearing-en-the-appeal-lf-the-Council-is-unable-to decide-the-appeal-at-that-mooting, it-shall-appear-for-a-vote-on-each-regular-meeting-of-tho-Council theroaftor-until-docided.

#### 17.158.090 - Definitions.

Historic Property. "Historic property" are those properties that are designated city landmarks; pursuant-to-Section-17.102.030-of-this-code; is-<u>are</u>listed en the National Register of Historic Places, <u>or</u> is-listed-as a California Registered Historical Landmark, or <u>as</u> is-a California Point of Historical Interest is <u>are</u> contributory to an S-7 Preservation Combining Zone pursuant to Section 17.84.010 of this code; <u>or have</u> has-received an "A" or "B" rating by the Oakland Cultural Heritage Survey.

17.158.190 - Discretionary actions.

- C. Certain permits issued under other City codes, regulations, and ordinances, including but not limited to:
  - Discretionary demolition permits, as defined in Chapter 15.36 of the Oakland Municipal Code, and as related to any demolition or removal of structures on a site where the zoning regulations requires design review to alter the exterior appearance of the applicable building facility, regardless of whether the owner intends to create a surface parking lot or a vacant lot pursuant to Section 15.36.080;
  - 2. Encroachment permits;
  - 3. Excavation permits;
  - 4. Grading permits;
  - 5. House moving permits;
  - 6. Obstruction permits;
  - 7. Permits for private construction of public improvements ("P-job" permits);
  - 8. Special activity permits issued by the City Administrator;
  - 9. Tree removal permits
  - D. Amendments to the zoning regulations, subdivision regulations, other codes and regulations governing the issuance of discretionary permits, or the Oakland General Plan.
  - E. Projects sponsored or assisted by the City or the Redevelopment Agency.

17.158.200 - Decision on projects.

The City Planning Commission or other appropriate decision-making body shall not approve a project for which the environmental review process is required without following the procedures outlined below.

- A. Exempt Projects. If the project is exempt some finding to this effect shall be included in the record. If action is by resolution, the resolution will typically contain a clause declaring that the project is exempt from the requirements of CEQA.
- B. Negative Declarations. If a negative declaration has been prepared, the decision-making body shall review this document and approve the negative declaration concurrently in approving the project's discretionary permit application. After making a decision, the decision-making body or its agent shall notify the Environmental Review Officer, to cause a <u>N</u>notice of <u>D</u>determination (NOD) to be filed with the County Clerk. If the project requires discretionary approval from any state agency, the Environmental Review Officer shall also cause the notice of determination to be filed with the State Office of Planning and Research. The filing of the notice of determination is a mandatory requirement under CEQA Guidelines Section 15075.

The NOD shall include the following information:

- 1. A project identification including its common name and its location;
- A brief description of the project;
- The date of project approval;
- 4. A statement of determination that the project would not have a significant effect on the environment;
- 5. A statement that a negative declaration or an environmental impact report has been prepared pursuant to the provisions of CEQA,
- 6. The address and location where a copy of the negative declaration may be examined.
- C. Environmental Impact Report (EIR). If an EIR has been prepared, the decision-making body shall certify the final EIR before approving the discretionary permits for the project, However, no project for which an EIR was completed and certified, and which identifies one or more significant environmental effects shall be approved without making one or more findings for each of the identified significant environmental effects. Such findings shall be supported by substantial evidence in the record. The following possible findings are identified in CEQA Guidelines Section 15091:
  - Changes or alterations have been required in, or incorporated into, the project which would avoid or substantially lessen the identified significant environmental effect as identified in the final EIR;
  - 2. Such changes or alterations are within the responsibility and jurisdiction of another public agency and not the city. Such changes have been adopted by such other agency or can and should be adopted by such other agency.
  - Specific economic, social, or other considerations make infeasible the mitigation measures or project alternatives identified in the final EIR.

Should a finding be made that mitigation measures are not feasible, the decision-making body shall adopt a "statement of overriding consideration" as described in CEQA Guidelines Section 15093.

4. If the benefits of a proposed project outweigh the unavoidable adverse environmental effects, the environmental effects may be considered acceptable. The decision-making body shall state in writing the specific reasons to support its action based on the final EIR and/or information in the record. If a statement of overriding consideration is adopted, it should be identified in the notice of determination.

After making a decision, the decision-making body or its agents shall notify the Environmental Review Officer, who shall cause a <u>Notice of Determination (NOD)</u> to be filed with the County Clerk and if required, the State Office of Planning and Research.

#### 17.158.210 - Time limits, extension or waiver of time limits.

The Review Officer shall have authority to extend or waive time limits as provided in the Guidelines. Such action is appealable to the City Planning Commission within ten (10) days of the Review Officer's decision. The decision of the City Planning Commission is final. However, failure to adhere to the prescribed time limits, shall not in and of itself, prejudice the city from requiring appropriate environmental review.

F. Projects Subject to CEQA and NEPA. If a project is subject to both CEQA and NEPA environmental processing, the one-year time limit for the preparation of an EIR, and the one-hundred\_--five (105) day time limit for a negative declaration may be waived under certain conditions as discussed in Section 15110 of the CEQA Guidelines. Specifically, this time limit waiver shall apply when additional time is needed to prepare a combined EIR/EIS or combined

negative declaration/FONSI, and if the time to prepare the combined documents would be shorter than the time required to prepare the documents separately.

Oakland, California, Code of Ordinances

FILED OFFICE OF THE CITY CLERN OAKLAND

2013 MAY 30 PM 5: 11

APPROVED AS TO FORM AND LEGALITY	
(Mugchen	
- WAGING h	
City Attor	14

City Attorney

INTRODUCED BY COUNCILMEMBER

# **OAKLAND CITY COUNCIL**

ORDINANCE NO.

C.M.S.

AN ORDINANCE, RECOMMENDED BY THE PLANNING COMMISSION, TO AMEND THE OAKLAND PLANNING CODE TO: 1) UPDATE REFERENCES, REDUCE REDUNDANCY, AND CLARIFY LANGUAGE IN VARIOUS CHAPTERS OF THE PLANNING CODE; 2) MODIFY TRUCK WEIGHT CLASSIFICATIONS LANGUAGE TO CONFORM WITH INDUSTRY STANDARDS; 3) MODIFY THE SETBACK REQUIREMENT FOR THE PORTION OF BUILDINGS ABOVE 55 FEET IN HEIGHT IN THE CBD-R ZONE; 4) ADD DESIGN REVIEW REQUIREMENT FOR RESIDENTIAL FACILITIES AND HISTORIC PROPERTIES IN THE CIX, IG, AND IO ZONES; 5) ELIMINATE THE INCREASED PARKING REQUIREMENT FOR EXPANSIONS OF ACTIVITIES ACROSS ZONE BOUNDARIES; AND 6) ADD SIDEWALK CAFES AS A PERMITTED FACILITY IN THE M- INDUSTRIAL ZONES.

WHEREAS, in March of 1998, the City adopted the Land Use and Transportation Element (LUTE) of the Oakland General Plan; and

WHEREAS, one of the objectives of the LUTE is to create a 'user friendly' Planning Code document that minimizes the complexity of regulations; and

WHEREAS, the Planning Code contains overly complex language, is difficult to navigate and needs formatting and reference updates, as well as organizational improvements; and

WHEREAS, there are inconsistencies, typographical errors and omissions in the Planning Code text; and

WHEREAS, the current zoning regulations as they relate to truck weight classifications are not in conformance with industry standards; and

WHEREAS, the current zoning regulations pertaining to the setback requirement for the portion of buildings above 55 feet in height in the Central Business District –Residential (CBD-R) Zone are potentially too restrictive in a downtown context as envisioned by the Oakland General Plan; and

WHEREAS, due to an oversight in previous text amendments, the zoning regulations do not include a requirement for design review for residential facilities and historic properties in the CIX, IG, and IO industrial zones, as is currently required in the M- industrial zones; and

1

WHEREAS, the increased parking requirement for expansions of activities across zone boundaries is potentially too restrictive in corridors where development is encouraged; and

WHEREAS, due to an oversight in a previous text amendment that added restaurants as permitted activities in the M- industrial zones, sidewalk cafes were inadvertently omitted as permitted facilities; and

WHEREAS, the City of Oakland's Planning Division staff has prepared the following amendments to the Oakland Planning Code to correct the above code issues, update references, reduce redundancy, and clarify language in various chapters of the Planning Code in order to better promote the public's health, safety and general welfare; and

WHEREAS, on April 17, 2013, at a duly noticed public hearing, the Zoning Update Committee recommended approval of the proposed Planning Code Amendments; and

WHEREAS, on May 15, 2013, at a duly noticed public hearing, the Planning Commission recommended approval of the proposed Planning Code Amendments; and

WHEREAS, after a duly noticed public meeting on June 11, 2013, the Community and Economic Development Committee voted to recommend the proposal to the City Council; and

WHEREAS, the City Council held a duly noticed public hearing on June 18, 2013 to consider the proposal; and

WHEREAS, the proposed amendments to the Planning Code rely on the previously certified Final Environmental Impact Reports for the Land Use and Transportation Element of the General Plan (1998); the Oakland Estuary Policy Plan (1998); the West Oakland Central City East, Coliseum and Oakland Army Base Redevelopment Areas; the 1998 Amendment to the Historic Preservation Element of the General Plan; the 2007-2014 Housing Element Final EIR (2010); and various Redevelopment Plan Final EIRs (collectively, "EIRs"); and

WHEREAS, the EIRs provide analysis of the environmental impacts of the proposed amendments and support all levels of approval necessary to implement the Planning Code amendments; and

WHEREAS, the proposed amendments to the Planning Code would not result in any significant effect that has not already been analyzed in the EIRs, and there will be no significant environmental effects caused by the change that have not already been analyzed in the EIRs; and

WHEREAS, the City Council hereby finds and determines on the basis of substantial evidence in the record that none of the circumstances necessitating preparation of additional environmental review, as specified in CEQA and the CEQA Guidelines, including, without limitation, Public Resources Code Section 21166 and CEQA Guidelines Sections 15162 or 15163 are present in that (1) there are no substantial changes proposed in the project or the circumstances under which the project is undertaken that would require major revisions of the EIR due to the involvement of new significant environmental effects or a substantial increase in the severity of previously

2

identified significant effects; and (2) there is no "new information of substantial importance," as defined in CEQA Guidelines Section 15162(a)(3); and

WHEREAS, each as a separate and independent basis, this action is exempt from CEQA pursuant to CEQA Guidelines Sections 15183 (projects consistent with General Plan and Zoning) and 15061(b)(3) (no significant effect on the environment); now, therefore,

# THE COUNCIL OF THE CITY OF OAKLAND DOES ORDAIN AS FOLLOWS:

Section 1. The City Council finds and determines the foregoing recitals to be true and correct and hereby makes them a pari of this Ordinance.

Section 2. The Environmental Review Officer shall file a Notice of Exemption/Notice of Determination, and an Environmental Declaration under the California Fish and Game Code (Section 711.4) with the County of Alameda.

Section 3. Title 17 of the Oakland Planning Code is hereby amended pursuant to *Exhibit A* attached hereto and hereby incorporated by reference. Additions to Title 17 of the Oakland Planning Code are shown in Exhibit A as <u>underline</u> and omissions are shown as strikethrough.

Section 4. This Ordinance shall be effective 30 days from the date of final passage by the City Council, but shall not apply to (a) building/construction related permits already issued and not yet expired, or (b) to zoning applications approved by the City and not yet expired, or to (c) zoning applications deemed complete by the City as of the date of final passage. However, zoning applications deemed complete by the City prior to the date of final passage of this Ordinance may be processed under provisions of these Planning Code amendments if the applicant chooses to do so.

Section 5. Nothing in this Ordinance shall be interpreted or applied so as to create any requirement, power, or duty in conflict with any federal or state law.

Section 6. If any section, subsection, sentence, clause or phrase of this Ordinance is held to be invalid or unconstitutional, the offending portion shall be severed and shall not affect the validity of the remaining portions which shall remain in full effect.

Section 7. That the record before this Council relating to this Ordinance includes, without limitation, the following:

1. The application, including all accompanying papers;

2. All final staff reports, decision letters and other documentation and information produced by or on behalf of the City;

3. All oral and written evidence received by the City staff, Planning Commission and City Council before and during the public hearings on the application;

4. All matters of common knowledge and all official enactments and acts of the City,

such as (a) the General Plan and the General Plan Conformity Guidelines; (b) Oakland Municipal Code, including, without limitation, the Oakland real estate regulations, Oakland Fire Code; (c) Oakland Planning Code; (d) other applicable City policies and regulations; and (e) all applicable state and federal laws, rules and regulations.

Section 8. That the custodians and locations of the documents or other materials which constitute the record of proceedings upon which the City Council's decision is based are respectively: (a) Community & Economic Development Agency, Planning & Zoning Division, 250 Frank H. Ogawa Plaza, Suite 3315, Oakland, CA; and (b) Office of the City Clerk, 1 Frank H. Ogawa Plaza, 1<sup>st</sup> floor, Oakland, CA.

IN COUNCIL, OAKLAND, CALIFORNIA, \_\_\_\_\_\_

PASSED BY THE FOLLOWING VOTE:

AYES- BROOKS, GALLO, GIBSON MCELHANEY, KALB, KAPLAN, REID, SCHAAF, and PRESIDENT KERNIGHAN

NOES-

ABSENT-

#### ABSTENTION-

.

· .

ATTEST:

LaTonda Simmons City Clerk and Clerk of the Council of the City of Oakland, California

DATE OF ATTESTATION:

# **NOTICE AND DIGEST**

AN ORDINANCE, RECOMMENDED BY THE PLANNING COMMISSION, TO AMEND THE OAKLAND PLANNING CODE TO: 1) UPDATE REFERENCES, REDUCE REDUNDANCY, AND CLARIFY LANGUAGE IN VARIOUS CHAPTERS OF THE PLANNING CODE; 2) MODIFY TRUCK WEIGHT CLASSIFICATIONS LANGUAGE TO CONFORM WITH INDUSTRY STANDARDS; 3) MODIFY THE SETBACK REQUIREMENT FOR THE PORTION OF BUILDINGS ABOVE 55 FEET IN HEIGHT IN THE CBD-R ZONE; 4) ADD DESIGN REVIEW REQUIREMENT FOR RESIDENTIAL FACILITIES AND HISTORIC PROPERTIES IN THE CIX, IG, AND IO ZONES; 5) ELIMINATE THE INCREASED PARKING REQUIREMENT FOR EXPANSIONS OF ACTIVITIES ACROSS ZONE BOUNDARIES; AND 6) ADD SIDEWALK CAFES AS A PERMITTED FACILITY IN THE M- INDUSTRIAL ZONES.

This Ordinance would amend the Oakland Planning Code in largely non-substantive ways intended to improve consistency, reduce redundancy and simplify language. Other changes include, but are not limited to: delete references to Code sections that were remnants of previous code updates and are now obsolete, e.g., references to Transport and Warehousing Commercial Activities which has been replaced by the new industrial Use Classifications adopted in 2009; add clarifying language to the method of calculating the maximum non-residential floor-area-ratio and residential density in the Central Business District; correct the density for rooming units in the 120- and 160-fool height areas listed on Table 17.35.04 (Height, Floor Area Ratio, Density, and Open Space Regulations) to correct a typographical error; increase the percentage of ground tloor commercial facade transparency in the CBD Zones (Table 17.58.03) to be consistent with similar standards in other zones; add language to various Code sections to codify and clarify the long-standing practice limiting evidence and testimony during appeals to evidence submitted in the appeal form hself and presented prior to the close of the written public comment period, for administrative decisions, or the close of public hearing, for Planning Commission decisions.